# THE GROWTH HORMONE/ INSULIN-LIKE GROWTH FACTOR AXIS DURING DEVELOPMENT

Edited by Isabel Varela-Nieto and Julie Ann Chowen ADVANCES IN EXPERIMENTAL MEDICINE AND BIOLOGY

Volume 567

# THE GROWTH HORMONE/INSULIN-LIKE GROWTH FACTOR AXIS DURING DEVELOPMENT

#### ADVANCES IN EXPERIMENTAL MEDICINE AND BIOLOGY

Editorial Board:

NATHAN BACK, State University of New York at Buffalo IRUN R. COHEN, The Weizmann Institute of Science DAVID KRITCHEVSKY, Wistar Institute ABEL LAJTHA, N.S. Kline Institute for Psychiatric Research RODOLFO PAOLETTI, University of Milan

Recent Volumes in this Series

Volume 559 CELL VOLUME SIGNALING Edited by Peter Lauf

Volume 560 MECHANIMS OF LYMPHOCYTE ACTIVATION AND IMMUNE REGULATION X:INNATE IMMUNITY Edited by Sudhir Gupta, William E. Paul, and Ralph Steinman

Volume 561 CHEMISTRY AND SAFETY OF ACRYLAMIDE IN FOOD Edited by Mendel Friedman and Don Mottram

Volume 562 CHOLINGERGIC MECHANSIMS Edited by José Gonzalez-Ros

Volume 563 UPDATES IN PATHOLOGY Edited by David C. Chhieng and Gene P. Siegal

Volume 564 GLYCOBIOLOGY AND MEDICINE Edited by John S. Axford

Volume 565 SLIDING FILAMENT MECHANISM IN MUSCLE CONTRACTION: FIFTY YEARS OF RESEARCH Edited by Haruo Sugi

Volume 566 OXYGEN TRANSPORT TO TISSUE XXVI Edited by Paul Okunieff, Jacqueline Williams, and Yuhchyau Chen

Volume 567 THE GROWTH HORMONE/INSULIN-LIKE GROWTH FACTOR AXIS DURING DEVELOPMENT Edited by Isabel Varela-Nieto and Julie A. Chowen

A Continuation Order Plan is available for this series. A continuation order will bring delivery of each new volume immediately upon publication. Volumes are billed only upon actual shipment. For further information please contact the publisher.

# THE GROWTH HORMONE/INSULIN-LIKE GROWTH FACTOR AXIS DURING DEVELOPMENT

edited by

**ISABEL VARELA-NIETO, PhD** 

Instituto de Investigaciones Biomedicas "Alberto Sols" Consejo Superior de Investigaciones Científicas (IIB-CSIC) Universidad Autonoma de Madrid (UAM) Arturo Duperier 4. 28029 Madrid, Spain

JULIE A. CHOWEN, PhD

Hospital Infantil Universitario Niño Jesús Department of Endocrinology and Investigation Menéndez Pelayo, 65, 28009 Madrid, Spain



Isabel Varela-Nieto, PhD Instituto de Investigaciones Biomedicas "Alberto Sols" Consejo Superior de Investigaciones Científicas (IIB-CSIC) Universidad Autonoma de Madrid (UAM) Arturo Duperier 4. 28029 Madrid, Spain ivarela@iib.uam.es

Julie A. Chowen, PhD Hospital Infantil Universitario Niño Jesús Department of Endocrinology and Investigation Menéndez Pelayo, 65, 28009 Madrid, Spain jachowen@hotmail.com

# THE GROWTH HORMONE/INSULIN-LIKE GROWTH FACTOR AXIS DURING DEVELOPMENT

Library of Congress Cataloging-in-Publication Data

A C.I.P. Catalogue record for this book is available from the Library of Congress.

ISBN-10: 0-387-25119-7	e-ISBN-10: 0-387-26274-1	Printed on acid-free paper.
ISBN-13: 978-0387-25119-6	e-ISBN-13: 978-0387-26274-1	

© 2005 Springer Science+Business Media, Inc.

All rights reserved. This work may not be translated or copied in whole or in part without the written permission of the publisher (Springer Science+Business Media, Inc., 233 Spring Street, New York, NY 10013, USA), except for brief excerpts in connection with reviews or scholarly analysis. Use in connection with any form of information storage and retrieval, electronic adaptation, computer software, or by similar or dissimilar methodology now known or hereafter developed is forbidden.

The use in this publication of trade names, trademarks, service marks and similar terms, even if they are not identified as such, is not to be taken as an expression of opinion as to whether or not they are subject to proprietary rights.

While the advice and information in this book are believed to be true and accurate at the date of going to press, neither the authors nor the editors nor the publisher can accept any legal responsibility for any errors or omissions that may be made. The publisher makes no warranty, express or implied, with respect to the material contained herein.

Printed in the United States of America.

9 8 7 6 5 4 3 2 1 SPIN 11053767

springeronline.com

# CONTENTS

	<b>INTRODUCTION</b> vii
1.	BASIC PHYSIOLOGY OF THE GROWTH HORMONE/ INSULIN-LIKE GROWTH FACTOR (IGF)-AXIS
2.	<b>EXPERIMENTAL MODELS FOR UNDERSTANDING THE ROLE</b> <b>OF INSULIN-LIKE GROWTH FACTOR-I AND ITS RECEPTOR</b> <b>DURING DEVELOPMENT</b>
3.	INSULIN-LIKE GROWTH FACTOR BINDING PROTEINS IN DEVELOPMENT
4.	<b>GROWTH HORMONE, INSULIN-LIKE GROWTH FACTORS</b> <b>AND THE FEMALE REPRODUCTIVE SYSTEM</b>
5.	SKELETAL MUSCLE
6.	CARDIOVASCULAR AND RESPIRATORY SYSTEMS 149 Åsa Tivesten and Jörgen Isgaard
7.	HEMATOPOIESIS

8.	BRAIN DEVELOPMENT	187
	Gregory J. Popken, Martha Dechert-Zeger, Ping Ye, A. Joseph D'Ercole	e

9.	<b>REGULATION OF VERTEBRATE SENSORY ORGAN DEVELOPMEN</b> <b>A SCENARIO FOR GROWTH HORMONE AND INSULIN-LIKE</b>	IT:
	GROWTH FACTORS ACTION	
	Amelia Diaz-Casares, Yolanda Leon, Enrique J. de la Rosa & Isabel Varela-Nie	eto

10.	ROLE OF INSULIN-LIKE GROWTH FACTORS IN NEURONAL   PLASTICITY AND NEUROPROTECTION
11.	AGING AND LIFE SPAN
12.	INTERACTIONS OF INSULIN-LIKE GROWTH FACTOR-I AND ESTROGEN IN THE BRAIN
13.	CANCER
14.	GENETIC BASIS OF PROPORTIONAL SHORT STATURE 341 Ángel Campos-Barros, Karen E. Heath, Jesús Argente
15.	FUTURE PERSPECTIVES: FROM STEM CELLS AND IGFBIOLOGY TO THE CLINIC

#### **INTRODUCTION**

Although growth hormone (GH) was discovered in 1921, it was not until the 1950s that investigators began to suspect that some of its effects on growth were not direct but involved a mediating factor. Indeed, after the discovery of insulin-like growth factor (IGF) it was thought that most, if not all, GH effects involved stimulation of IGF-I production. However, it is now accepted that GH and IGF have both common and individual roles in growth, development and metabolism.

The IGF family, comprised of insulin and two factors similar to insulin termed IGF-I and IGF-II, has pleiotropic functions in mammals. The first member of this family to be identified was insulin, with subsequent investigation resulting in the elucidation of its role in glucose metabolism and its implication in the etiology of diabetes mellitus. This discovery effected an explosion in the investigation of the structure, function, and mechanisms of action of insulin. The enormous interest in this molecule resulted in the concession of three Nobel Prizes for the investigation of insulin: in 1923 for the discovery of its capacity to treat diabetes, in 1958 for the first sequence of a protein and in 1963 for the first determination of the three-dimensional structure of a protein. Hence, the investigation of insulin has been a pioneer in many scientific fields. Likewise, the insulin receptor was the first receptor with tyrosine kinase activity to be identified and cloned, and is the prototype of membrane receptors with enzymatic activity. Consequently, the mechanisms of action of this receptor have been the focus of intense study leading to the identification of intracellular signaling pathways in response to members of this protein family and has lead, once again, to the discovery of new molecules, pathways and signaling cascades that has had important reverberations in other scientific fields.

Later, the IGFs were discovered and found to be intricately involved in embryonic development and postnatal growth. This was followed by the discovery of a family of binding proteins (IGFBPs) that not only transport the IGFs in the circulation, but alter their function and biological activity. Such widespread and intricate regulation further emphasized the importance of these growth factors in normal physiology. The enormous interest in the insulin-IGF family of proteins lead to the generation of the first double and triple knock-out animals. These knock-out animals, null for a combination of the IGFs, their receptors and the BPs, has helped to shed light on the differential roles of these molecules and their importance in growth and embryonic development. These studies also demonstrated new and diverse functions for these factors, amongst which their activities in the nervous system is noteworthy. Advances in the understanding of the expression, function, and structure of these factors has not only helped us to further our knowledge of embryonic and postnatal growth, but has also opened the door to new possibilities for their therapeutic use in the treatment of growth alterations, in neurodegeneration, and regenerative medicine.

The objective of this volume is to explain our current understanding regarding the cellular actions of these factors and how this basic knowledge has contributed to our comprehension of their implications in distinct illnesses, as well as the possibility for the development of new therapies.

Chapter 1 covers the basic physiology of the GH/IGF axis, while Chapter 2 discusses some of the most widely used experimental models that have helped to determine the roles of the various members of this axis. The role of IGFBPs in brain and other tissues is discussed in Chapter 3. The importance of IGFs in brain development and function is dealt with in various chapters, all with an individual focus. Chapter 8 covers overall brain development and the effect that each component of the GH/IGF-I system exerts in this process. The trophic effects that IGFs have on neurons in the mature brain and the protection that they exert during neurodegenerative processes or assaults by noxious substances is addressed in Chapter 10 and the interaction of IGFs with other trophic factors in Chapter 12. Chapter 9 focuses specifically on IGFs in neurosensory systems. The effect of the GH/IGF axis in the reproductive tract, skeletal muscle, cardiovascular system and hematopoeisis are addressed in Chapters 4-7. Two areas of current interest include the influence of IGF-I on aging and its possible role in cancer. These subjects are affronted in Chapters 11 and 13, respectively. In Chapter 14 the growing information regarding genetic implications in human growth, involving almost all members of the GH/IGF axis, is presented. The final chapter deals with a new and exciting area of investigation: the use of stem cells in investigation and the possible treatment of a variety of illnesses. This is a rapidly advancing field and future developments will necessitate the need for a second edition.

We would like to express our sincere gratitude to all of the authors for taking time out of their busy schedules to participate in this project. In addition, we would like to acknowledge and thank our colleagues that have generously donated their time and experience to act as reviewers in order to improve the scope and focus of this book. We are in debt to all.

Isabel Varela-Nieto and Julie A. Chowen

## Chapter 1

## BASIC PHYSIOLOGY OF THE GROWTH HORMONE/INSULIN-LIKE GROWTH FACTOR AXIS

Laura M. Frago and Julie A. Chowen

Hospital Infantil Universitario Niño Jesús, Universidad Autónoma de Madrid, Departamento de Endocrinología, Menéndez Pelayo, 65, 28009 Madrid, Spain

Key words: Binding proteins; growth hormone-releasing hormone; intracellular signaling; receptor; somatostatin.

#### **1. INTRODUCTION**

The growth hormone (GH)/insulin-like growth factor (IGF) axis is intimately involved in the integration of a multitude of signals that regulate systemic growth and metabolism throughout fetal and postnatal development. In the adult animal, maintenance of metabolic balance and cellular integrity, as well as involvement in normal tissue regeneration and in response to injury, can also be attributed to GH and the IGFs.

The classical components of this axis, including GH, the GH receptor (GHR), the GH binding proteins (GHBPs), IGF-I, IGF-II, IGF receptors and the six IGF binding proteins (IGFBPs), will be described briefly in this chapter. Although GH is produced mainly by the anterior pituitary, its receptor is expressed in many tissues throughout the organism<sup>1-3</sup>, indicating that this hormone can have widespread effects. In contrast, the IGFs are produced by most tissues during both fetal and postnatal life, with both the circulating and locally produced factors having physiological functions<sup>4,5</sup>.

Most tissues also express the IGF-I receptor, during at least one period of development. Hence, both GH and IGF-I have important widespread effects on the development and physiology of the organism.

The involvement of both GH and the IGFs in the development and functioning of distinct physiological processes and systems is addressed in greater depth in subsequent chapters.

#### 2. GH PHYSIOLOGY

#### 2.1 GH

GH was discovered in 1921 by Evans and Long when they demonstrated that extracts of the anterior pituitary administered intraperitoneally promoted growth and maturation<sup>6</sup>. Growth hormone was subsequently isolated from bovine pituitary in 1944<sup>7</sup>. Indeed, GH is produced primarily, or almost exclusively, in the somatotroph of the anterior pituitary. This hormone, belonging to the family of cytokine peptides, contains 191 amino acids, and the most common circulating form has a molecular mass of 22 kDa, while a 20-kDa form has also been detected.8 Rat GH was one of the earliest cDNAs cloned<sup>9</sup>, and human GH was one of the first loci to be assigned to a human chromosome<sup>10</sup>. The human GH gene cluster is located on the long arm of chromosome 17 (17q23) and spans 66.5 kilobases<sup>11</sup>. This cluster is comprised of five genes that from 5' to 3' include pituitary GH (GH1), the chorionic somatomammotrophin pseudogene 1 (CSHP1), the chorionic somatomammotriphin gene 1 (CSH1), the placental GH gene (GH2) and gene 2 for chorionic somatomammotrophin (CSH2). Each of these genes is comprised of five exons and four introns and all share very high homology throughout their entire sequence (92% to 98%).

In the human fetus GH can be detected in the anterior pituitary as early as the  $6^{th}$  week of gestation<sup>12</sup> and its levels gradually increase during gestation. In serum, GH is detected after the  $8^{th}$  week of gestation and reaches peak levels between 20 and 24 weeks of gestation, after which they begin to decrease. At birth, however, GH levels continue to be elevated and then gradually decline until prepubertal levels are reached at approximately 3 months of age<sup>13</sup>. These high circulating GH levels are somewhat of a paradox, as this hormone is apparently not implicated in fetal growth.

In rats, GH is first detectable on fetal day 18 or 19 and its level increases progressively until birth<sup>14,15</sup>. The total GH content of the pituitary is low at birth and increases gradually during prepubertal life. At the time of pubertal

onset, there is a significant increase in pituitary GH content and during adult life it continues to increase gradually<sup>16</sup>. These changes in pituitary GH content are similar in male and female rats although in postpubertal males, pituitary GH content is significantly higher than in females.

GH is secreted from the anterior pituitary in a pulsatile fashion and this pattern is affected by both age and sex. The physiological importance of this secretory pattern is exemplified by the correlation with its ability to promote growth and stimulate the production of growth factors and liver enzymes.<sup>17,18</sup> In the laboratory rat, the postpubertal GH secretory pattern is strikingly sexually dimorphic<sup>19</sup>. The male rat exhibits high- amplitude pulses with a periodicity of approximately 3 hours, superimposed on a low interpulse baseline. In contrast, the secretory pattern in female rats is comprised of more irregular, low-amplitude pulses with an elevated baseline level. In humans, this sexual dimorphism is less striking, with some discrepancy as to its existence<sup>20</sup>. However, in both instances, the pulsatile secretion of this hormone is fundamental for its physiological effect.

#### 2.2 Hypothalamic control of GH secretion

The first evidence for central nervous system (CNS) control of GH secretion was introduced in 1940 by Heatherington and Ranson when they noted that lesions of the entire ventral hypothalamus retarded longitudinal bone growth<sup>21</sup>. It was later demonstrated that such lesions result in the depletion of pituitary GH content<sup>22</sup> and that electrical stimulation of the ventral medial hypothalamus (VMH) increases plasma GH levels<sup>23</sup>. The race to isolate and identify the hypothalamic factors involved in the control of pituitary hormones began in the 1950s, but the identity of many of these remained elusive for many years.

We now know that GH synthesis and secretion is under the control of the hypothalamus, with the neuropeptides growth hormone-releasing hormone (GHRH) stimulating and somatostatin (SS) inhibiting GH production and release from the anterior pituitary. In 1973, somatostatin, the 14-amino acid GH-inhibiting peptide, was isolated and sequenced<sup>24</sup>. The N-terminal-extended form of the tetradecapeptide composed of 28 amino acids was later isolated<sup>25-27</sup>. A common 15-kDa polypeptide appears to serve as the precursor for both the SS-14 and SS-28 isoforms<sup>28</sup>. It is believed that SS plays an important role in determining the basal or interpulse level of GH secretion<sup>29,30</sup>. Somatostatin is widely distributed throughout the brain and in peripheral tissues. The majority of the SS projections to the median eminence arise from neurons of the periventricular nucleus (PeN)<sup>31</sup> and these neurons are thought to play a key role in the neuroendocrine control of pituitary function<sup>32</sup>.

The isolation of GHRH proved to be a formidable task, which is now attributed to its minuscule concentration in the CNS and its susceptibility to biological inactivation during purification procedures. In 1982 GHRH was ultimately purified and sequenced independently by two different groups<sup>33,34</sup>. Growth hormone-releasing hormone is produced by neurons located in the hypothalamic arcuate nucleus and ventromedial hypothalamic area, but those that project to the median eminence are primarily located in the arcuate nucleus<sup>35,36</sup> and are present in three molecular forms: GHRH(1-44), GHRH(1-40), and GHRH(1-37)<sup>33,34,37</sup>. The rat GHRH gene spans nearly 10 kilobases of rat genomic DNA, contains 5 exons, and encodes the 104-amino acid GHRH precursor<sup>38</sup>.

The pulsatility of GH release is the result of an interplay between these two hormones, with the release of GHRH increasing and that of SS decreasing into the portal system during a GH burst and the inverse occurring during a nadir<sup>39</sup>. At the level of the pituitary, GHRH and SS have noncompetitive antagonistic effects<sup>40</sup>. GHRH stimulates GH synthesis and secretion through activation of the adenylate cyclase system<sup>41</sup> and SS suppresses cAMP levels in the pituitary. However, numerous other factors are also involved in this process via modulation of GHRH or SS secretion or a direct action on the somatotroph.

GH itself has a negative feedback action at the hypothalamus, stimulating SS and inhibiting GHRH synthesis<sup>42</sup>. Somatostatin neurons in the periventricular nucleus express the GH receptor, suggesting a direct action on this neuropeptide system<sup>43,44</sup>. In contrast, GHRH neurons do not appear to express this receptor<sup>44,45</sup>. The negative feedback effect on this neuropeptide system is thought to occur, at least in part, via modulation of neuropeptide Y (NPY) neurons in the arcuate nucleus<sup>44,45</sup>. As NPY is intimately involved in metabolic control, this is one possible vinculum between the control of growth and metabolism. Other circulating factors involved in growth and metabolism, such as IGF-I, leptin, ghrelin, free fatty acids, and insulin, also modulate GH secretion at the level of the hypothalamus.

#### 2.3 The GH receptor

The growth hormone receptor (GHR) is a 620-amino-acid singletransmembrane glycoprotein that belongs to the class I cytokine receptor superfamily (M 130 000)<sup>46</sup>. This family includes receptors for prolactin, erythropoietin, thrombopoietin, leptin, ciliary neurotrophic factor, leukemia inhibitory factor, granulocyte-colony stimulating factor, granulocyte macrophage-colony stimulating factor, and several interleukins (ILs)<sup>47</sup>. The class I cytokine receptors span the membrane once and contain an extracellular region, a single hydrophobic transmembrane domain of 24 amino acids and an intracellular region. In their extracellular ligand binding domain, four conserved cysteine residues are found in the N-terminal portion and a conserved WSXWS (Trp-Ser-X-Trp-Ser) motif is present in the Cterminal and predicted to function as a ligand interaction site. In the GHR, this motif is altered to YGEFS. It does not contact the ligand, but upon mutation of the last serine both binding affinity and trafficking of the receptor are impaired<sup>48</sup>. Because this motif does not make direct contact with the ligand, the YGEFS motif is likely to serve a structural role critical for ligand binding<sup>49</sup>. In the intracellular domains of cytokine receptors, no consensus catalytic site has been identified and limited amino acid homology has been recognized in only two motifs: box-1 and box-2. These motifs are located in the cytoplasmic membrane-proximal region of several members of the cytokine family and are known to be involved in signal transduction<sup>50</sup>. Box-1, a proline-rich motif present in all members, consists of a P-X-P sequence and a preceding cluster of hydrophobic amino acids. The other motif, box-2, is present in most of these receptors and is only 50% conserved in the members of the receptor family. It is characterized as a cluster of hydrophobic amino acids, is followed by negatively charged residues, and ends with one or two basic amino acids.

The GHR was initially cloned from rabbit and human cDNA libraries and subsequently from other species and has an amino acid sequence homology of 70% between the different species<sup>51</sup>. Expression of GHR mRNA is detectable in all tissues studied to date with liver, fat, muscle and kidney showing the highest levels of expression<sup>3</sup>.

The GH receptor, which forms homodimers when ligand is bound, does not have tyrosine kinase activity, but is closely associated with protein kinases, in particular Janus kinase 2 (JAK2). When the GHR is bound by its ligand, JAK2 is autophosphorylated and then the receptor is phosphorylated<sup>52</sup>. This then triggers specific intracellular signaling cascades including mitogen-activated protein kinase (MAPK), the phosphatidylinositol-3'-kinase (PI3-kinase)-Akt, and latent transcription factors known as signal transducers and activators of transcription (STAT).

In addition to the membrane-bound form of GHR, a soluble circulating form of the receptor, named growth hormone binding protein (GHBP), has been characterized. The mechanism by which GHBPs are generated varies among species. In some species, including human, the GHBP is generated by proteolytic cleavage at the cell surface (shedding) of the membrane-bound form of the receptor by a recently identified metalloprotease called tumor necrosis factor (TNF)- $\alpha$  converting enzyme (TACE or ADAM-17)<sup>53</sup>. However, in rodents, the high-affinity GHBP and GH receptor are coded by

the same gene and produced by alternate RNA splicing to generate a truncated form  $^{54}$ .

Diverse biological functions have been attributed to GHBP. In plasma, complex formation of GHBP with GH creates a circulating reservoir, protects GH from degradation and excretion, prolongs its half-life, and may enhance its bioactivity *in vivo* through these mechanisms. The high-affinity, low-capacity GHBP has a molecular mass of 61 kDa and preferentially binds the 22-kDa form of GH. This protein is structurally identical to the extracellular portion of the GH receptor. The low-affinity, high-capacity GHBP has a molecular mass of 100 kDa and preferentially binds to the 20-kDa form of GH. Some authors suggest that serum GHBP measurement is a useful tool for measuring GHR abundance in the body<sup>55</sup>.

Studies on the crystallographic structure of GH-GHR complexes revealed a trimeric complex of two receptors and a single molecule of hormone<sup>49</sup>. Recent studies suggest that the unliganded GHR may exist as a preformed dimer that undergoes a GH-induced conformational change to achieve activation<sup>56</sup>. Further analysis of the complex showed two binding sites on hGH on opposite sides with slightly different binding affinities<sup>57</sup>. The ligand binding domain and dimerization interface at the extracellular part of the GHR lie on its cysteine pairs and YGEFS motif. The binding of GH to the two receptors is considered the first step in the action of GH. This process is initiated with GH binding to one molecule of receptor through its site 1, a functional epitope of 11 amino acids, followed by association of this complex to a second receptor molecule through GH's site 2, a 31-amino-acid region that stabilizes and, together with a 500 Å contact region between the two extracellular domains of GHR, defines the binding of the second GHR<sup>58</sup>. Receptor dimerization is crucial for signal transduction since the hGH antagonist, G120R, capable of binding to the first GHR via its site 1, but defective at site 2 for inducing dimerization, cannot transduce signals. However, this antagonist is internalized as efficiently as the wild-type ligand<sup>59,60</sup>.

Although the GHR and other cytokine receptors do not possess intrinsic tyrosine kinase activity, GH binding to its receptor results in rapid tyrosine phosphorylation of multiple cellular proteins because of the activation of GHR-associated tyrosine kinases of the JAK family. The ability to associate with and activate tyrosine kinases is essential for propagation of most intracellular signals. Several intracellular pathways are activated in response to GH (Fig. 1). These include the activation of STAT1, STAT3, STAT5, MAPKs, and PI3'-kinase<sup>61,62</sup>. Activation of JAK2 kinase activity is believed to be critical for the initiation of most, if not all, of these signaling pathways. Yet the mechanism(s) by which GH binding leads to JAK2 activation is as yet uncertain.



Figure 1. Intracellular signaling cascade for the growth hormone receptor.

The JAK kinases represent a distinct family of soluble tyrosine kinases that have been strongly implicated in signal transduction of many members of the cytokine family<sup>63</sup>. Among the known members of the JAK family, JAK2 is the main kinase activated by the GH-GHR complex and this activation is considered to be the initiating step in GHR signal transduction<sup>64</sup>. JAK2 binds to the GHR via box-1<sup>65</sup>. No specific amino acid within box-1 appears to be essential for the association between the receptor and JAK2. Mutation of each individual proline residue in box-1 or the simultaneous mutation of the first two proline residues does not impair association of the kinase. However, a specific secondary structure of the receptor is required as simultaneous mutation of the last two proline residues or of the three hydrophobic residues (isoleucine; leucine and valine into threonine) abolishes the capacity of the receptor to interact with and to activate JAK2<sup>66</sup>. The box-1 sequence of the GHR is very similar to prolinerich SH3 domains. However, no SH3 binding domain has been identified in JAK2 and the existence of an SH3-containing adapter protein could mediate the association. The first 46 residues of the GHR cytosolic tail, which contain the box-1, are sufficient to induce some JAK2 activation, but maximal activity of this kinase requires downstream residues in the proximal transmembrane part of the GHR cytoplasmic domain. In resting cells, JAK kinases are thought to be associated with the receptor cytoplasmic domains, but catalytically inactive. Upon ligand stimulation, the GHR becomes firmly dimerized, initiating enzyme activity in two JAK2 molecules, which transphosphorylate each other on one or more tyrosine residues within the kinase domain of the paired JAK2.

Upon GH binding to its receptor, activation of JAK2 is rapid and transient, reaching maximal activity at 5 to 20 min, and returns to basal level by 60 min. Once activated, JAK2 phosphorylates the GHR on multiple tyrosine residues providing docking sites for other signaling molecules<sup>67</sup>. Some of the effects of GH are mediated directly through JAK2, such as stimulation of cell proliferation<sup>65</sup>. JAK2 contains 48 tyrosine residues, all of which may become phosphorylated upon GH binding, suggesting that this kinase can interact with multiple signaling molecules<sup>68</sup>. Among the direct substrates of JAK2 are several docking proteins involved in the Ras/MAPK pathway, the STAT proteins (signal transducers and activators of transcription), as well as the insulin receptor substrate proteins IRS-1 and IRS-2, which initiate the PI-3' kinase pathway<sup>69</sup>. To date, only GHdependent calcium transport appears to involve a pathway that is activation<sup>70</sup>. JAK2 Hence. independent of regulation of JAK2 phosphorylation could regulate GH actions. Indeed, protein tyrosine phosphatase-1B (PTP-1B) has been shown to regulate GH signaling by reducing JAK2 phosphorylation, and this may be an important mechanism of limiting GH's actions during metabolic stress<sup>71</sup>. Other GH effects, such as GH-dependent expression of the insulin gene or the serine protease inhibitor (Spi) 2.1 gene require additional sequences in the GHR cytoplasmic domain<sup>72,73</sup>. In a more distal C-terminal portion of the GHR cytoplasmic domain, phosphorylation of Y487, Y534, Y566, or Y627 is required for tyrosine phosphorylation of STAT5, while phosphorylation of Y333 and/or Y338 GHR appears to be required for GH-stimulated lipid and protein synthesis<sup>74,75</sup>.

#### 2.4 GH Actions

The most well studied actions of GH are those on longitudinal growth and metabolism, although this hormone has more recently been shown to have neuromodulatory actions affecting neuronal survival and processes such as memory and cognition, which may or may not involve stimulation of IGF-I production<sup>76,77</sup>. The growth and metabolic effects of GH occur in multiple tissues, either directly or via IGF-I. These actions include stimulation of protein synthesis and carbohydrate and lipid metabolism; hence, it could be classified as an anabolic, lipolytic, and diabetogenic hormone.

GH has both insulin-like and anti-insulin-like effects on cells and tissues. The insulin-like actions include stimulation of transient increases in glucose and amino acid transport, lipogenesis, and protein synthesis. Hence, GH and insulin may activate some common signaling pathways. Indeed, GH stimulates phosphorylation of the insulin receptor substrates 1, 2, and 3 (IRS-1, -2 and -3). Tyrosine phosphorylation of IRS proteins by JAK2 provides a binding site for the SH2 domain of the 85-kDa regulatory subunit of PI-3' kinase<sup>78</sup>. Although PI-3' kinase is required for insulin stimulated glucose transport, GH-induced glucose transport has been reported to not require PI-3' kinase<sup>79</sup>. However, inhibition of PI-3' kinase blocks GH-stimulated lipid synthesis<sup>80</sup> and the anti-lipolytic action of GH<sup>81</sup>, suggesting that PI-3' kinase activity is important for the insulin-like actions of GH.

The reduction of body fat induced by GH involves both decreased lipogenesis and increased lipolysis. The signaling mechanisms by which GH regulates these processes are an active area of investigation. GH appears to inhibit adipocyte differentiation at a step preceding the induction of genes required for terminal differentiation, such as the gene encoding peroxisome proliferator activated receptor  $\gamma$  (PPAR $\gamma$ )<sup>82</sup>. One signaling event that may be important in this process is GH activation of the cAMP-specific phosphodiesterase PDE4A5 by a PI-3' kinase-dependent mechanism<sup>83</sup>.

The signaling pathways involved in GH stimulation of protein synthesis remain largely unknown. Specific tyrosines in the cytoplasmic domain of the GH receptor are required for GH stimulation of protein synthesis, suggesting that recruitment of molecules to the GH receptor is involved<sup>74</sup>. GH is known to stimulate the activity of p70<sup>S6k</sup>, an enzyme thought to regulate translational activity<sup>83</sup>.

Chronically elevated GH is "anti-insulin-like," promoting insulin resistance and possibly type 2 diabetes. GH is thought to have this effect, at least in part, by interfering with the ability of insulin to stimulate carbohydrate metabolism. Potential underlying mechanisms for this action of GH are emerging. Decreased insulin receptor, IRS-1 and IRS-2, tyrosyl phosphorylation in response to insulin has been reported in rodent models of chronic GH excess<sup>84</sup>. In contrast, GH excess can lead to chronic activation of the IRS-PI-3' kinase pathway in liver, reducing the degree of insulin induced activation<sup>85</sup>. In addition, GH inhibits the expression of the gene encoding glucose transporter 1 (*GLUT1*), also potentially modulating glucose metabolism<sup>86</sup>.

For many years it was believed that GH had no role in controlling fetal growth, because expression of the GH receptor is relatively low in fetal tissue. However, congenital GH deficiency can be associated with a reduction in birth length<sup>87</sup>. Furthermore, the GH-IGF-I axis is functional and responds normally to stimuli, such as undernutrition<sup>88</sup>. However, fetal growth is dependent mainly on the IGFs as discussed below.

#### **3.** IGF PHYSIOLOGY

#### 3.1 The IGFs

The existence of the IGFs was first proposed in 1957 on the basis of studies indicating that GH did not directly stimulate the incorporation of sulfate into cartilage, but rather acted through a serum factor<sup>89</sup>. This factor was originally termed sulfation factor, then somatomedin and ultimately, insulin-like growth factor-I and II. IGF-I was not purified and characterized until more than two decades later<sup>90</sup>. The terminology "insulin-like" was used because these factors are able to stimulate glucose uptake into fat cells and muscle, and indeed, both IGF-I and IGF-II show approximately 50% homology with insulin<sup>90,91</sup>.

Subsequent investigation demonstrated that GH, after binding to its transmembrane receptor, initiates a signaling cascade leading to transcriptional regulation of IGF-I and related genes. It was originally thought that systemic growth was promoted by GH acting mainly on the liver to stimulate IGF-I production, which then reached target tissues via the circulation to activate mechanisms involved in tissue proliferation, growth, and metabolism. It is now evident that not only does GH have independent actions that do not involve IGF-I production<sup>92</sup>, but IGF-I synthesis occurs in many tissues under the control of a variety of local and circulating factors, which may or may not include GH<sup>93-96</sup>. Furthermore, this local production of IGF-I may be directly responsible for the growth promoting effects of GH, rather than the circulating growth factor<sup>97</sup>.

The initial concept that IGF-I exerts endocrine actions was wellfounded. It was identified and purified from blood, and the best data at the time indicated that the liver was the major, if not sole, site of synthesis. In the early 1980s it was observed that explants from multiple fetal mouse tissues released immunoreactive IGF-I into the cultured media and that a large portion of this material was recognized by cell surface receptors specific for IGF-I<sup>93</sup>. It was later shown that multiple human fetal tissues synthesize IGF-I, including intestine, muscle, kidney, placenta, stomach, heart, skin, pancreas, hypothalamus, brain stem, spleen, and adrenal.<sup>95</sup> Later it was demonstrated that multiple adult rat tissues, including kidney, liver, lung, heart, and testes had IGF-I concentrations much higher than could be explained by the contribution from the blood and that IGF-I levels in many tissues were regulated by GH<sup>98</sup>. Hence, it was postulated that actions of IGF-I were predominately local, being exerted either on the cells of origin (autocrine actions) or on nearby cells (paracrine actions). Northern blot analyses and *in situ* hybridization histochemistry<sup>95,98</sup> indisputably showed that IGF-I expression was widespread in rodent and human tissues and, to a significant extent, dependent on GH, at least postnatally. Thus, ubiquitous IGF-I expression, accompanied by similarly widespread expression of type 1 IGF receptors and IGFBPs, make cellular signaling and regulation of IGF-I availability possible in the local milieu.

GH deficient and GH receptor deficient rodents and humans have given much insight into the GH-independent actions of IGF-I. Normal, or only slightly reduced, birth weights and lengths in GH-deficient or -resistant humans and mice supports the theory of GH-independent actions of IGF-I during embryonic and fetal development<sup>99,100</sup>. Although, their post-natal growth rate is greatly reduced and if untreated their final stature will be well below normal, these subjects have relatively normal tissue development, indicating that GH is not essential for the differentiation of these cell types.

The IGFs are mitogenic peptides that potently affect cell growth and differentiation<sup>101</sup>. As mentioned above, IGF-I is widely expressed both prenatally and postnatally in many tissues, including the brain<sup>95,96,101,102</sup>. Various components of the IGF-I system are highly expressed in regions of the postnatal brain undergoing remodeling or plasticity such as the cerebellum, hippocampus and olfactory bulb<sup>4,103,104</sup>. Using *in situ* hybridization, it has been demonstrated that IGF-IR colocalizes to or near sites of IGF-I expression in the olfactory bulb, the hippocampus and the cerebellum<sup>4,105,106</sup>. It has therefore been suggested that autocrine or paracrine modes of action are used by the IGF system in these brain regions.

#### **3.2** The IGF-I receptor

The IGF-I receptor (IGF-IR) is ubiquitous and contributes to the regulation of biological functions of many cell types<sup>107,108</sup>. Only mature B lymphocytes and hepatocytes do not express this receptor<sup>109</sup>. The IGF-I receptor has approximately 70% homology with the insulin receptor, with both receptors having tyrosine kinase activity. IGF-IR is initially transcribed in the cell cytoplasm as a single protein, which is processed by cellular proteases to form a heterodimer having a  $\alpha_2\beta_2$  configuration. The  $\alpha$  and  $\beta$ 

subunits are joined together by disulfide bonds, as both subunits contain several cysteine amino acid residues. Most of the extracellular domain of the IGF-IR, hence the ligand binding domain, is constituted by the  $\alpha$ -subunit, which is heavily glycosylated<sup>107-109</sup>. The  $\beta$ -subunit contains a hydrophobic moiety, which functions as the transmembrane domain of the receptor. As expected, the tyrosine kinase domain of IGF-IR is in the cell cytoplasm.

There are three known ligands for the IGF-IR: IGF-I, IGF-II and insulin at supraphysiological concentrations. Binding of the ligand to the  $\alpha$ -subunit of the IGF-I receptor triggers a series of events that lead to autophosphorylation of the receptor via its tyrosine kinase activity<sup>107-109</sup>. This, in turn, causes the interaction with certain cellular substrates that mediate the transmission of mitogenic and antiapoptotic signals to the cell nucleus (Fig. 2). The major intracellular substrates for IGF-IR are insulin receptor substrate-1 (IRS-1). Shc, and 14-3-3<sup>110</sup>. On these grounds, it is clear that IGF-IR has a relevant role in promoting cell growth, size and proliferation and in protecting cells from a variety of apoptotic stimuli<sup>107-109</sup>. It has also been demonstrated that an over-expressed and activated IGF-IR is quasi-obligatory for the establishment of a transformed cell phenotype<sup>111</sup>. Interestingly, the targeting of IGF-IR can reverse the malignant cell phenotype, without significant effects on the biology of normal cells<sup>112-115</sup>. The possible roles of IGFs in cancer are discussed in Chapter 13.



Figure 2. Intracellular signaling cascade of the IGF receptor.

A number of more recent studies have indicated that the biology of the IGF-I receptor is more complex than previously thought. In fact, in certain cellular contexts, activation of the IGF-I receptor can also promote cell differentiation in myoblasts<sup>116,117</sup>, adipocytes<sup>118,119</sup>, osteoblasts<sup>120</sup>, and in cells of the CNS<sup>121-125</sup>. Interestingly, in vitro studies have shown that the IGF-IR system can induce differentiation in murine 32D hematopoietic cells along the granulocytic pathway<sup>125</sup>. Involvement of IGF-IR in cell differentiation appears to be somewhat of a paradox, as its main function consists in promoting cell proliferation and growth<sup>108</sup> as these cellular processes oppose the cell differentiation programs. However, it must be kept in consideration that the cellular context in any given paradigm also comes into play in mediating the signaling of the IGF-IR system. For example, the murine 32D hematopoietic culture system lacks IRS-I<sup>126</sup>, which is the main intracellular mediator for mitogenic signals induced by the IGF-I receptor<sup>127,128</sup>. In the absence of IRS-I, the Shc-mediated signaling system leads to cell differentiation<sup>129</sup>.

Another important property of the IGF-IR is the control of cell adhesion and motility<sup>115,130</sup>. The IGF-I receptor system also has a dual life in these processes depending on the cellular context<sup>108</sup>. There are three main intracellular substrates that interact with the activated IGF-IR: IRS-1, Shc and 14-3-3, each of which targets a distinct domain of IGF-IR and gives rise to a specific signal transduction pathway. The tyrosine kinase domain of IGF-I receptor triggers the activation of the IRS-1/PI-3' kinase/Akt/p70<sup>S6k</sup> pathway. Tyrosine 950 of the IGF-I receptor is mainly responsible for activation of the Shc/MAPK (mitogen-activated protein kinase) pathway. The third domain of the IGF-IR is composed of a serine quartet at position 1280-1283, which interacts with 14-3-3 proteins and consequently activates Raf by causing its mitochondrial translocation. All pathways lead to the neutralization of the BAD-induced apoptotic program<sup>110,131</sup>, with the inhibition of apoptosis occurring via phosphorylation of BAD. However, just two of the three pathways can be sufficient for cell survival<sup>130</sup>.

The tyrosine kinase domain of the IGF-IR is a necessary requirement for mitogenesis that is mainly controlled by the IRS-1/PI-3'kinase/Akt/p70<sup>S6k</sup> pathway. Activation of Shc and 14-3-3 can also contribute to some extent in delivering a mitogenic signal. The IRS-1/PI-3' kinase/Akt/p70<sup>S6k</sup> pathway is also essential for the regulation of cell size increase, which is required for the predisposition of cells to mitosis<sup>108</sup>.

The protective effect of the activated IGF-IR on cell survival has been known for some time. The variety of experimental procedures used to induce apoptosis suggests that the wild-type IGF-I receptor and/or its ligands have a widespread antiapoptotic effect on many death signals. Indeed, McCarthy and colleagues<sup>132</sup> have suggested that only IGF-I and Bcl-2 truly suppress the initiation of the apoptotic program, while inhibitors of caspases can arrest the completion of the program, but have no effect on its initiation. The mechanism by which the IGF-IR protects cells from apoptosis has been the object of much investigation, culminating in a reasonable elucidation of the main pathway used by this receptor for protection against apoptotic injuries. This pathway originates with the interaction of the IGF-IR with one of its major substrates, IRS-1<sup>133</sup>, which activates PI-3' kinase, which in turn activates Akt/PKB. The concluding step is the phosphorylation of BAD, one of the members of the Bcl-2 family of proteins, by Akt. However, some investigators have suggested that this receptor must have alternative pathways<sup>134</sup>. One alternative pathway could involve MAPK, which originates, at least in part, from another major substrate of the IGF-IR, the Shc proteins<sup>135,136</sup>. Another IGF-IR mediated anti-apoptotic pathway is the activation of Raf-1 and its translocation to the mitochondria. All three pathways result in the maintenance of BAD phosphorylation.

#### **3.3** The IGF Binding Proteins

Six high-affinity IGFBPs have been characterized to date. Circulating IGFBPs act as carrier proteins to transport the IGFs out of the circulation to their target tissues. The union of the IGFs to these binding proteins also prolongs their half-life by protecting them from proteolysis. A majority of circulating IGF-I is bound in a ternary complex composed of IGFBP-3, acid labile subunit (ALS), and IGF itself. More recently it has been reported that IGFBP-5 can also form a complex with IGF and ALS<sup>137</sup>. Most target tissues also express IGFBPs, with this expression being both tissue and developmental stage dependent. These binding proteins modulate the local actions of the IGFs at the cellular level and can be either stimulatory or inhibitory. In addition, independent effects of IGFBPs are becoming more apparent<sup>138,139</sup>.

The serum profile of the IGFBPs differs in the fetus from the postnatal profile; IGFBP-1 and -2 levels are higher whereas IGFBP-3 levels are lower<sup>140</sup>. Serum IGFBP-1 is the most sensitive indicator of fetal nutrition. There is an inverse correlation between fetal IGFBP-1 concentrations and birth weight from as early as 16 weeks of gestation.

#### 3.4 IGF Actions

The most important determinant of fetal growth in both humans and other mammals is the IGF-I system<sup>88,101</sup>. The IGFs are detectable in many fetal tissues from the first trimester and IGF concentrations in the fetal circulation increase during pregnancy<sup>141</sup>. It is therefore interesting that the most important regulator of fetal IGF-I concentrations is the availability of adequate glucose across the placenta. Glucose increases IGF-I concentrations through an increase in insulin secretion, which in turn regulates fetal IGF-I secretion<sup>142</sup>.

Postnatal growth is also correlated with plasma IGF-I levels in humans<sup>140,143</sup> and IGF-I treatment increases both body weight and size, indicating a role for circulating IGF-I in body composition. Indeed, as mentioned earlier, IGF-I is the mediator of many of the growth-promoting actions of GH. This is exemplified by the fact that children with GH resistance, with high levels of GH and very low levels of circulating IGF-I, have a pathologically short stature.

IGF-I has many important physiological roles<sup>107-127,131,135,136</sup>. It is involved in cell division and differentiation and throughout postnatal life it functions as a tropic factor maintaining the integrity of many tissues. In addition, evidence continues to accumulate suggesting that IGF-I is also an important factor in the defense against noxious substances and degenerative processes. The role of IGF-I in the development of specific tissues and in preventing neurodegenerative diseases, as well as its possible role in cancer is address in depth in the subsequent chapters.

#### 4. FUTURE PERSPECTIVES

Our understanding of the physiology of the GH/IGF-I axis will continue to expand as more hormones, receptors, and factors involved in this process are identified. Of course advances in genetic and molecular techniques will play a very important role in these new discoveries. Some of the most important advances will most likely involve the intracellular signaling mechanisms employed by these factors. Not only will new proteins be identified, but elucidation of how the numerous intracellular pathways interact will allow us to further understand how the same hormone or factor can cause different effects in different tissues, how different hormones can have the same effect in a given tissue, or how various factors interact at the cellular level to produce a physiologically significant outcome.

#### REFERENCES

- 1. H.C. Mertani and G. Morel, In situ gene expression of growth hormone (GH) receptor and GH binding protein in adult male rat tissues. *Mol Cell Endocrinol*, **109**, 47-61 (1995).
- S.R. Edmonson, G.A. Werther, A. Russell, D. LeRoith and C.T. Roberts Jr, Localization of growth hormone receptor/binding protein messenger ribonucleic acid (mRNA) during rat fetal development: relationship to insulin-like growth factor-I mRNA. *Endocrinology*, **136**, 4602-4609 (1995).
- M. Ballesteros, K.C. Leung, R.J. Ross, T.P. Iismaa, and K.K. Ho, Distribution and abundance of messenger ribonucleic acid for growth hormone receptor isoforms in human tissues. *J Clin Endocrinol Metab* 85, 2865-2871 (2000).
- C.A. Bondy, H. Werner, C.T. Roberts and D. LeRoith, Cellular pattern of type-I insulin-like growth factor receptor gene expression during maturation of the rat brain: comparison with insulin-like growth factors I and II. *Neuroscience* 46, 909-923 (1992).
- A.G. Schuller, J.W. van Neck, D.J. Lindenbergh-Kortleve, C. Groffen, I. de Jon, E.C. Zwarthoff and S.L. Drop, Gene expression of the IGF binding proteins during post-implantation embryogenesis of the mouse; comparison with the expression of IGF-I and II and their receptors in rodent and human. *Adv Exp Med Biol*, 343, 267-277 (1993).
- 6. H.M. Evans and L.A. Long, The effect of the anterior lobe of the hypophysis administered intraperitoneally upon growth and maturity and estrous cycles of the rat. *Anat Record* **21**, 62 (1921).
- 7. C.H. Li and H.M. Evans, The isolation of pituitary growth hormone. *Science*, **99**, 183-185 (1944).
- 8. W.L. Miller and N.L. Eberhardt, Structure and evolution of the growth hormone gene family. *Endocr Rev*, 4, 97-130 (1983).
- 9. P.H. Seeburg, J. Shine, J.A. Martial, J.D. Baxter and H.M. Goodman, Nucleotide sequence and amplification in bacteria of the structural gene for rat growth hormone. *Nature*, **270**, 486-494 (1977).
- D. Owerbach, W.J. Rutter, J.A. Martial, J.D. Baxter and T.B. Shows, Genes for growth hormone, chorionic somatomammotropin and growth hormone-like gene on chromosome 17 in humans. *Science*, 209, 289-294 (1980).
- 11. E.Y. Chen, Y.C. Liao, D.H. Smith, H.A. Barrera-Saldana, R.E. Gelinas and P.H. Seeburg, The human growth hormone locus: nucleotide sequence, biology, and evolution. *Genomics*, **4**, 479-97 (1989).
- S.L. Asa, K. Kovacs, F.A. Laszlo, I. Domokos and C. Ezrin, Human fetal adenohypophysis: histologic and immunocytochemical analysis. *Neuroendocrinology*, 43, 308-316 (1986).
- 13. J.W. Finkelstein, T.F.Anders, E.J. Sachar, H.P. Roffwarg and L.D. Hellman LD, Behavioural state, sleep stage and growth hormone levels in human infants. *J Clin Endocrinol Metab*, **32**, 368-371 (1971).
- 14. A. Chatelain, J.P. Dupouy, and M.P. Dubois, Ontogenesis of cells producing polypeptide hormones (ACTH, MSH, LPH, GH, prolactin) in the fetal hypophysis of the rat: influence of the hypothalamus. *Cell Tissue Res* **196**, 409-427 (1979).
- L.S. Frawley, J.P. Hoeffler and F.R. Boockfor, Functional maturation of somatotropes in fetal rat pituitaries analysis by reverse hemolytic plaque assay. *Endocrinology*, 116, 2355-2360 (1985).

- C.A. Birge, G.T. Peake, I.K. Mariz and W.H. Daughaday, Radioimmunoassyable growth hormone in the rat pituitary gland: effects of age, sex and hormonal state. *Endocrinology*, 81, 195-204 (1967).
- 17. O.G.P. Isaksson, J-O. Jansson, R.G. Clark and I. Robinson, Significance of the secretory pattern of growth hormone. *NIPS*, **1**, 44-47 (1986).
- H.K. Choi and D.J. Waxman, Plasma growth hormone pulse activation of hepatic JAK-STAT5 signaling: Developmental regulation and role in male-specific liver gene expression. *Growth Horm IGF Res*, 141, 3245-3255 (2000).
- 19. J-O. Jansson, S. Edén and O. Isaksson, Sexual dimorphism in the control of growth hormone secretion. *Endocr Rev*, **6**, 128-150 (1985).
- S.M. Pincus, E.F. Gevers, I.C. Robinson, G. vanden Berg, F. Roelfsema, M.L. Hartman and J.D. Veldhuis, Females secrete growth hormone with more process irregularity than males in both humans and rats. *Am J Physiol*, 270, E107-115 (1996).
- 21. A.W. Heatherington and S.W. Ranson, Hypothalamic lesions and adiposity in the rat. *Anat Rec*, **78**, 149, (1940).
- 22. S. Reichlin, Growth hormone content of the pituitaries from rats with hypothalamic lesions. *Endocrinology*, **69**, 225-230, (1961).
- 23. L.A. Frohman, L.L. Bernardis and K.L. Kant, Hypothalamic stimulation of growth hormone secretion. *Science*, **162**, 580-582 (1968).
- P. Brazeau, W. Vale, R. Burgus, N. Ling, M. Butcher, J. Rivier and R. Guillemin, Hypothalamic polypeptide that inhibits the secretion of immunoreactive pituitary growth hormone. *Science*, **179**, 77-79 (1973).
- L. Pradayrol, H. Jörnvall, V. Mutt and V. Ribet, N-terminally extended somatostatin: the primary structure of somatostatin-28. *FEBS Lett*, **109**, 55-58 (1980).
- F. Esch, P. Bóhlen, N. Ling, R. Benoit, P. Brazeau and R. Guillemin, Primary structure of ovine hypothalamic somatostatin-28 and somatostatin-25. *Proc Natl Acad Sci (USA)*, 77, 6827-6831 (1980).
- A.V. Schally, W-Y. Huang, R.C.C. Chang, A. Arimura, T.W. Redding, R.P. Millar, M.W. Hunkapiller and L.E. Hood, Isolation and structure of pro-somatostatin: a putative somatostatin precursor from pig hypothalamus. *PNAS*, 77, 4489-4493 (1980).
- 28. H.H. Zingg and Y.C. Patel, Biosynthesis of immunoreactive somatostatin by hypothalamic neurons in culture. *J Clin Invest*, **70**, 1101-1109 (1982).
- R.A. Steiner, J.K. Stewart, J. Barber, D. Koerker, C.J. Goodner, A. Brown, P. Illner and C.C. Gale, Somatostatin: A physiological role in the regulation of growth hormone secretion in the adolescent male baboon. *Endocrinology*, **102**, 1587-1594 (1978).
- G.S. Tannenbaum, J. Epelbaum, E. Colle, P. Brazeau and J.B. Martin, Antiserum to somatostatin reverses starvation-induced inhibition of growth hormone but not insulin secretion. *Endocrinology*, **102**, 1909-1914, (1978).
- K. Ishikawa, Y. Taniguchi, K. Kurosumi, M. Suzuki and M. Shinoda, Immunohistochemical identification of somatostatin-containing neuron projections to the median eminence of the rat. *Endocrinology*, **121**, 94-97 (1987).
- 32. J.O. Willoughby and J.B. Martin, Pulsatile growth hormone secretion: inhibitory role of medial proptic area. *Brain Res*, **148**, 240-244, (1978).
- 33. R. Guillemin, P. Brazeau, P. Böhlen, F. Esch, N. Ling and W.B. Wehrenberg, Growth hormone-releasing factor from a human pancreatic tumor that caused acromegaly. *Science*, **218**, 585-587 (1982).

- 34. J. Rivier, J. Spiess, M. Thorner and W. Vale, Characterization of a growth hormone-releasing factor from a human pancreatic islet tumor. *Nature*, **300**, 276-278 (1982).
- 35. B. Bloch, R.C. Gaillard, P. Brazeau, H.D. Lin and N. Ling, Topographical and ontogenetic study of the neurons producing growth hormone-releasing factor in human hypothalamus. *Regul Pept*, **8**, 21-31, (1984).
- P.E. Sawchenko, L.W. Swanson, J. Rivier and W.W. Vale, The distribution of growth hormone-releasing factor (GRF) immunoreactivity in the central nervous system of the rat: an immunohistochemical study using antisera directed against rat hypothalamic GRF. J Comp Neurol, 237, 100-115, (1985).
- 37. J. Spiess, J. Rivier and W. Vale, Characterization of rat hypothalamic growth hormona-releasing factor. *Nature*, **303**, 532-535 (1983).
- K.E. Mayo, G.M. Cerelli, M.G. Rosenfeld and R.M. Evans, Characterization of cDNA and genomic clones encoding the precursor to rat hypothalamic growth hormone-releasing factor. *Nature*, **314**, 464-467, (1985).
- P.M. Plotsky and W. Vale, Patterns of growth hormone-releasing factor and somatostatin secretion into the hypophysial-portal circulation of the rat. *Science*, 230, 461-463, (1985).
- 40. W. Vale, J. Vaughan, G. Yamamoto, J. Spiess and J. Rivier, Effects of synthetic human pancreatic (tumor) GH releasing factor and somatostatin, triiodothyronine and dexamethasone on GH secretion *in vitro*. *Endocrinology*, **112**, 1553-1555 (1983).
- P. Brazeau, N. Ling, F. Esch, P. Bohlen, C. Mougin and G. Guillemin, Somatocrinin (growth hormone releasing factor) in vitro bioactivity; Ca<sup>++</sup> involvement, cAMP mediated action and additivity of effect with PGE2. *Biochem Biophys Res Commun*, **109**, 588-594 (1982).
- 42. P. Zeitler, L. Vician, J.A. Chowen-Breed, J. Argente, G.S. Tannenbaum, D.K. Clifton and R.A. Steiner, Regulation of somatostatin and growth hormonereleasing hormone gene expression in the rat brain. *Metabolism*, **39**, 46-49 (1990).
- K.A. Burton, E.B. Kabigting, D.K. Clifton and R.A. Steiner, Growth hormone receptor messenger ribonucleic acid distribution in the adult male rat brain and its colocalization in hypothalamic somatostatin neurons. *Endocrinology*, **131**, 958-963 (1992).
- 44. S. Minami, J. Kamegai, H. Sugihara, N. Suzuki and I. Wakabayashi, Growth hormone inhibits its own secretion by acting on the hypothalamus through its receptors on neuropeptide Y neurons in the arcuate nucleus and somatostatin neurons in the periventricular nucleus. *Endocr J*, **45**, S19-26 (1998).
- 45. Y.Y. Chan, R.A. Steiner and D.K. Clifton, Regulation of hypothalamic neuropeptide-Y neurons by growth hormone in the rat. *Endocrinology*, **137**, 1319-1325 (1996).
- 46. J.F. Bazan, Structural design and molecular evolution of a cytokine receptor superfamily. *Proc Natl Acad Sci (USA)*, 87, 6934-6938 (1990).
- 47. J. Finidori and P.A. Kelly, Cytokine receptor signaling through novel families of transducer molecules: Janus kinases and signal transducers and activators of transcription. *J Endocrinol*, **147**, 11-23 (1995).
- B. Duriez, M.L. Sobrier, P. Duquesnoy, M. Tixier-Boichard, E. Decuypere, G. Coquerelle, M. Zeman, M. Goossens and S. Amselem, A naturally occurring growth hormone receptor mutation: in vivo and in vitro evidence for the functional importance of the WS motif common to all members of the cytokine receptor superfamily. *Mol Endocrinol*, 7, 806-814 (1993).

- A.M. DeVos, M. Ultsch and A.A. Kossiakoff, Human growth hormone and the cellular domain of tissue plasminogen activgator at 24-A resolution. *Science*, 255, 306-312 (1992).
- M. Murukami, M. Narazaki, M. Hibi, H. Yawata, K. Yasukawa, M. Hamaguchi, T. Taga and T. Kishimoto, Critical cytoplasmic region of the interleukin 6 signal transducer gp130 is conserved in the cytokine receptor family. *Proc Natl Acad Sci* (USA), 88, 11349-11353 (1991).
- K.C. Leung, N. Doyle, M. Ballesteros, M.J. Waters, K.K. Ho, Insulin regulation of human hepatic growth hormone receptors: divergent effects on biosynthesis and surface translocation. J Clin Endocrinol Metab, 85, 4712-4720 (2000).
- K. He, X. Wang, J. Jiang, R. Guan, K.E. Bernstein, P.P. Sayeski and S.J. Frank, Janus kinase 2 determinants for growth hormone receptor association, surface assembly, and signaling. *Mol Endocrinol*, **17**, 2211-2227 (2003).
- 53. Y. Zhang, J. Jiang, R.A. Black, G. Baumann and S.J. Frank, Tumor necrosis factoralpha converting enzyme (TACE) is a growth hormone binding protein (GHBP) sheddase: the metalloprotease (TACE/ADAM-17) is critical for (PMA-induced) GH receptor proteolysis and GHBP generation. *Endocrinology*, **141**, 4342-4348 (2000).
- W.R. Baumbach, D.L. Horner and J.S. Logan, The growth hormone-binding protein in rat serum is an alternatively spliced form of the rat growth hormone receptor. *Genes Dev*, 3, 1199-1205 (1989).
- G. Baumann, Growth hormone binding to a circulating receptor fragment the concept of receptor shedding and receptor splicing. *Exp Clin Endocrinol Diabet*, 103, 2-6 (1995).
- J. Gent, P. van Kerkhof, M. Roza, G. Bu and G.J. Strous, Ligand independent growth hormone receptor dimerization occurs in the endoplasmic reticulum and is required for ubiquitin system-dependent endocytosis. *Proc Natl Acad Sci (USA)*, 99, 9858-9863 (2002).
- M.J. Waters, S.W. Rowlinson, R.W. Clarkson, C.M. Chen, P.E. Lobie, G. Norstedt and S. Bastiras, Signal transduction by the growth hormone receptor. *Proc Soc Exp Biol Med*, 206, 216-220 (1994).
- J.A. Wells, B.C. Cunningham, G. Fuh, H.B. Lowman, S.H. Bass, M.G. Mulkerrin, M. Ultsch and A.M. deVos, The molecular basis for growth hormone receptor interactions. *Rec Prog Horm Res*, 48, 253-275 (1993).
- P.A. Harding, X. Wang, S. Okada, W.Y. Chen, W. Wan and J.J. Kopchick, Growth hormone (GH) and a GH antagonist promote GH receptor dimerization and internalization. *J Biol Chem*, 271, 6708-6712 (1996).
- G. Fuh, B.C. Cunningham, R. Fukunaga, S. Nagata, D.V. Goeddel and J.A. Wells, Rational design of potent antagonists to the human growth hormone receptor. *Science*, 256, 1677-1680 (1992).
- 61. C. Carter-Su, J. Schwartz and L.S. Smit, Molecular mechanism of growth hormone action. *Annu Rev Physiol*, **58**, 187-207 (1996).
- 62. T. Zhu, E.L.K. Goh, R. Graichen, L. Ling and P.E. Lobie, Signal transduction via the growth hormone receptor. *Cell Signall*, **13**, 599-616 (2001).
- 63. L.S. Argetsinger and C. Carter-Su, Mechanism of signaling by growth hormone receptor. *Physiol Rev*, **76**, 1089-1107 (1996).
- L.S. Argetsinger, G.S. Campbell, X.N. Yang, B.A. Witthuhn, O. Silvennoinen, J.N. Ihle and C. Carter-Su, Identification of Jak2 as a growth hormone receptor associated tyrosine kinase. *Cell*, 74, 237-244 (1993).

- P. Colosi, K. Wong, S.R. Leong and W.I. Wood, Mutational analysis of the intracellular domain of the human growth receptor. *J Biol Chem*, 268, 12617-12623 (1993).
- H. Dinerstein, F. Lago, L. Goujon, F. Ferrag, N. Exposito, J. Finidori, P.A. Kelly and M.C. Postel-Vinay, The proline-rich region of the GH receptor is essential for Jak2 phosphorylation, activation of cell proliferation and gene transcription. *Mol Endocrinol*, 9, 1701-1707 (1995).
- X.Z. Wang, C.J. Darus, B.X.C. Xu and J.J. Kopchick, Identification of growth hormone receptor (GHR) tyrosine residues required for GHR phosphorylation and JAK2 and STAT5 activation. *Mol Endocrinol*, 10, 1249-1260 (1996).
- 68. O. Sivennoinen, J.N Ihle, J. Schlessinger and D.E. Levy, Interferon-induced nuclear signaling by Jak protein tyrosine kinases. *Nature*, **366**, 583-585 (1993).
- 69. C. Carter-Su and L.S. Smit, Signaling via Jak tyrosine kinases: growth hormone receptor as a model system. *Rec Prog Horm Res*, **53**, 61-83 (1998).
- N. Billestrup, P. Bouchelouche, G. Allevato, M. Ilondo and J.H. Nielsen, Growth hormone receptor C-terminal domains required for growth hormone induced intracellular free Ca<sup>2+</sup> oscillations and gene transcription. *Proc Natl Acad Sci* (USA), 92, 2725-2729 (1995).
- F. Gu, N. Dube, J.W. Kim, A. Cg, Mde. J. Ibarra-Sanchez and Y.R. Tremblay Boisclair, Protein tyrosine phosphatase 1B attenuates growth hormone-mediated JAK2-STAT signalling. *Mol Cell Biol*, 23, 3753-3762 (2003).
- L. Goujon, G. Allevato, G. Simonin, L. Paquereau, A. LeCam, J. Clark, J.H. Nielsen, J. Djiane, M.C. Postel-Vinay and M. Edery, Cytoplasmic sequences of the growth hormone receptor necessary for signal transduction. *Proc Natl Acad Sci* (USA), 91, 957-961 (1994).
- 73. A. Moldrup, G. Allevato, T. Dyrberg, J.H. Nielsen and N. Billestrup, Growth hormone action in rat insulinoma cells expressing truncated growth hormone receptors. *J Biol Chem*, **266**, 17441-17445 (1991).
- P.E. Lobie, G. Allevato, J.H. Nielsen, G. Norstedt and N. Billestrup, Requirement of tyrosine residues 333 and 338 of the growth hormone (GH) receptor for selected GH-stimulated function. *J Biol Chem*, 270, 21745-21750 (1995).
- B.X.C. Xu, X.Z. Wang, C.J. Darns and J.J. Kopchick, Growth hormone promotes the association of transcription factor STAT5 with the growth hormone receptor. J Biol Chem, 271, 19768-19773 (1996).
- 76. K.K. Ho, A.J. O'Sullivan and D.M. Hoffman, Metabolic actions of growth hormone in man. *Endocr J*, **43**, S57-S63 (1996).
- C. Carter-Su, M.M. Ramsey and W.E. Sonntag, A critical analysis of the role of growth hormone and IGF-1 in aging and lifespan. *Trends Genet*, 18, 295-301 (2002).
- L.S. Smit, D. Meyer, L. Argetsinger, J. Schwartz and C. Carter-Su, Molecular events in growth hormone receptor interaction and signaling. Handbook of Physiology (vol V)(Kostyo JL ed.) 445-480. Oxford University Press (1999).
- H. Sakaue, W. Ogawa, M. Takata, S. Kuroda, K. Kotani, H. Matsumoto, H. Sakaue, S. Nishio, H. Ueno and M. Kasuga, Phosphoinositide 3-kinase is required for insulin-induced but not for growth hormone or hyperosmolarity induce glucose uptake in 3T3-L1 adipocytes. *Mol Endocrinol*, 11, 1552-1562 (1997).
- M. Ridderstrale and H. Tornqvist, PI-3-kinase inhibitor wortmannin blocks the insulin-like effects of growth hormone in isolated rat adipocytes. *Biochem Biophys Res Commun*, 203, 306-310 (1994).

- T. Yamauchi, Y. Kaburagi, K. Ueki, Y. Tsuji, G.R. Stark, I.M. Kerr, T. Tsushima, T. Akanuma, Y. Akanuma, I. Komuro, K. Tobe, Y. Yakazi and T. Kadowaki, Growth hormone and prolactin stimulate tyrosine phosphorylation of insulin receptor substrate-1, -2 and -3, their association with p85 phosphatidylinositol 3kinase (PI-3kinase) and concomitantly PI-3kinase activation via JAK2 kinase. J Biol Chem, 273, 15719-15726 (1998).
- L.H. Hansen, B. Madgen, B. Teisner, J.H. Nielssen and N. Billesstrup, Characterization of the inhibitory effect of growth hormone on pituitary preadipocyte differenciation. *Mol Endocrinol*, 12, 1140-1149 (1998).
- S.J. MacKenzie, S.J. Yarwood, A.H. Peden, G.B. Bodger, R.G. Vernon and M.D. Houslay, Stimulation of p70S6Kinase via a growth hormone-controlled phosphatidylinositol 3-kinase pathway leads to the activation of a PDE4A cyclic AMP-specific phosphodiesterase in 3T3F442 A preadipocytes. *Proc Natl Acad Sci* (USA), 95, 3549-3554 (1998).
- F.P. Dominici, D. Cifone, A. Barthe and D. Turyn, Alterations in the early steps of the insulin-signaling system in skeletal muscle of GH transgenic mice. Am J Physiol, 277, 447-454 (1999).
- 85. F.P. Dominici, D. Cifone, A. Barthe and D. Turyn, Loss of sensitivity to insulin at early events of the insulin signaling pathway in the liver of growth hormone transgenic mice. *J Endocrinol*, **161**, 383-392 (1999).
- P-K. Tai, I.F. Liao, E.H. Chen, J.H. Diez, J. Schwartz and C. Carter-Su, Differential regulation of two glucose transporters by chronic growth hormone treatment of cultured ·T·-F442 A adipose cells. *J Biol Chem*, 265, 21828-21834 (1990).
- P.D. Gluckman, A.J. Gunn, A. Wray, W.S. Cutfield, P.G. Chatelain, O. Guilbaud, G.R. Ambler, P. Wilton and K. Albertsson-Wikland, Congenital idiopathic growth hormone deficiency associated with prenatal and early postnatal growth failure. The International Board of the Kabi Pharmacia International Growth Study. *J Pediatr*, 121, 920-923 (1992).
- 88. P.D. Gluckman and J.E. Harding, Fetal growth retardation: underlying endocrine mechanisms and postnatal consequences. *Acta Paediatr*, **422**, 69-72 (1997).
- W.D. Salmon Jr and W.H. Daughaday, A hormonally controlled serum factor which stimulates sulphate incorporation by cartilage in vitro. *J Lab Clin Med*, 49, 825-836 (1957).
- E. Rinderknecht and R.E. Humbel RE, The amino acid sequence of human insulinlike growth factor I and its structural homology with proinsulin. *J Biol Chem*, 253, 2769-2776 (1978).
- 91. T.L. Blundell, S. Bedarkar and R.E. Humbel, Tertiary structures, receptor binding and antigenicity of insulin-like growth factors. *Fed Proc*, **42**, 2592-2597 (1983).
- 92. J. Wang, J. Zhou and C.A. Bondy, Igf1 promotes longitudinal bone growth by insulin-like actions augmenting chondrocyte hypertrophy. *FASEB J*, **13**, 1985-1990 (1999).
- 93. A.J. DÉrcole, G.T. Applewhite and L.E. Underwood, Evidence that somatomedin is synthesized by multiple tissues in the fetus. *Dev Biol*, **75**, 315-328 (1980).
- 94. C.T. Roberts Jr, S.R. Lasky, W.L. Lowe Jr, W.T. Seaman and D. LeRoith, Molecular cloning of rat insulin-like growth factor I complementary deoxyribonucleic acids: differential messenger ribonucleic acid processing and regulation by growth hormone in extrahepatic tissues. *Mol Endocrinol*, 1, 243-248 (1987).
- 95. V.K. Han, P.K. Lund, D.C. Lee and A.J. D'Ercole, Expression of somatomedin/insulin-like growth factor messenger ribonucleic acids in the human

fetus: identification, characterization, and tissue distribution. J Clin Endocrinol Metab, 66, 422-429 (1988).

- 96. W.L. Lowe Jr, S.R. Lasky, D. LeRoith and C.T. Robertrs Jr, Distribution and regulation of rat insulin-like growth factor 1 messenger ribonucleic acids encoding alternative carboxy-terminal E-peptides: evidence for differential processing and regulation in liver. *Mol Endocrinol*, 2, 528-535 (1988).
- 97. O.G. Isaksson, J-O. Jansson and I.A. Gause, Growth hormone stimulates longitudinal bone growth directly. *Science*, **216**, 1237-1239 (1982).
- A.J. D'Ercole, A.D. Stiles and L.E. Underwood, Tissue concentration of somatomedin C: further evidence for multiple sites of synthesis and paracrine or autocrine mechanisms of action. *Proc Natl Acad Sci (USA)*, **81**, 935-939 (1984).
- 99. L.A. Perez-Jurado and J. Argente, Molecular basis of familial growth hormone deficiency. *Horm Res*, **42**, 189-197 (1994).
- 100. A. Bartke, V. Chandrashekar, D. Turyn, R.W. Steger, L. Debeljuk, T.A. Winters, J.A. Mattison, N.A. Danilovich, W. Croson, D.R. Wernsing and J.J. Kopchick, Effects of growth hormone overexpression and growth hormone resistance on neuroendocrine and reproductive functions in transgeneic and knock-out mice. *Proc Soc Exp Biol Med*, **222**, 113-123 (1999).
- 101. J.J. Jones and D.R. Clemmons, Insulin-like growth factors and their binding proteins biological actions. *Endocr Rev*, 16, 3-34 (1995).
- 102. A.J. D'Ercole, P. Ye, A.S. Calikoglu and G. Gutierrez-Ospina, The role of insulinlike growth factors in the central nervous system. *Mol Neurobiol*, **13**, 227-255 (1996).
- 103. C.A. Bondy and W.H. Lee, Correlation between insulin-like growth factor (IGF)binding protein 5 and IGF-I gene expression during brain development. *J Neurosci*, 13, 5092-5104 (1993).
- 104. W.H. Lee, K.M. Michels and C.A. Bondy, Localization of insulin-like growth factor binding protein-2 messenger RNA during postnatal brain development: correlation with insulin-like growth factors I and II. *Neuroscience*, 53, 251-263 (1993).
- 105. V.C. Russo, S.R: Edmondson, F.A. Mercuri, C.R. Buchanan and G.A. Werther, Identification, localization, and regulation of insulin-like growth factor binding proteins and their messenger ribonucleic acids in the newborn rat olfactory bulb. *Endocrinology*, 135, 1437-1446 (1994).
- 106. G.A. Werther, M. Abate, A. Hogg, H. Cheesman, B. Oldfield, D. Hards, P. Hudson, B. Power, K. Freed and A.C. Herington, Localization of insulin-like growth factor-I mRNA in rat brain by in situ hybridization-relationship to IGF-I receptors. *Mol Endocrinol*, 4, 773-778 (1990).
- 107. A. Grimberg and P. Cohen, Role of insulin-like growth factors and their binding proteins in growth control and carcinogenesis. *J Cell Physiol*, **183**, 1-9 (2000).
- 108. R. Baserga, The contradictions of the insulin-like growth factor I receptor. Oncogene, 19, 5574-5581 (2000).
- 109. B. Valentinis and R. Baserga, IGF-I receptor signaling in transformation and differentiation. *Mol Pathol*, **54**, 133-137 (2001).
- 110. F. Peruzzi, M. Prisco, M. Dews, P. Salamoni, E. Grasilli, G. Romano, B. Calabretta and R. Baserga, Multiple signaling pathways of the insulin-like growth factor 1 receptor in protection of apoptosis. *Mol Cell Biol*, **19**, 7203-7215 (1999).
- 111. H. Yu and T.J. Rohan, Role of the insulin-like growth factor family in cancer development and progression. *Natl Cancer Inst*, **92**,1472-1489 (2000).

- 112. R. Baserga, A. Hongo, M. Rubini, M. Prisco and B. Valentini, The IGF-I receptor in cell growth, transformation and apoptosis. *Biochem Biophys Res Commun*, 214, 475-481 (1995).
- 113. C. D'Ambrosio, A. Ferber, M. Resnicoff and R. Baserga, A soluble insulin-like growth factor I receptor that induces apoptosis of tumor cells in vivo and inhibits tumorigenesis. *Cancer Res*, **56**, 4013-4020 (1996).
- 114. G. Romano, K. Reiss, X. Tu, F. Peruzzi, B. Belleti, J.Y. Wang, T. Zanocco Maranni and R. Baserga, Efficient in vitro and in vivo gene regulation of a retrovirallydelivered pro-apoptotic factor under the control of the Drosophila HSP70 promoter. *Gene Ther*, 8, 600-607 (2001).
- 115. K. Reiss, X. Tu, G. Romano, F. Peruzzi, Y. Kanes and R. Baserga, Intracellular association of a mutant insulin-like growth factor receptor with endogenous receptors. *Clin Cancer Res*, 7, 2134-2144 (2001).
- 116. J.R. Florini, D.Z. Ewton, S.L. Falen and J.J. Van Wyk, Biphasic concentration dependency of stimulation of myoblast differenciation. *Am J Physiol*, **250**, 771-778 (1986).
- 117. P.J. Schmid, T. Steiner and E.R. Froesch, Preferential enhancement of myoblast differentiation by insulin-like factors (IGF-I and IGF-II) in primary cultures of chicken embryonic cells. *FEBS Lett*, **161**, 117-121 (1983).
- 118. P.J. Smith, L.S. Wise, R. Berkovitz, C. Wan and C.S. Rubin, Insulin-like growth factor-I is an essential regulator of the differentiation of 3T3-L1 adipocytes. *J Biol Chem*, **263**, 9402-9408 (1988).
- 119. M. Lorenzo, A.M. Valverde, T. Teruel and M. Benito, IGF-I is a mitogen involved in differentiation-related gene expression in fetal rat brown adipocytes. *J Cell Biol*, 123, 1567-1575 (1993).
- 120. P.J. Schmid, T. Steiner and E.R. Froesch, Insulin-like growth factor-I supports differentiation of cultured osteoblast-like cells. *FEBS Lett*, **173**, 48-52 (1984).
- 121. F.A. McMorris, T.M. Smith, S. DeSalvo and R.W. Furlanetto, Insulin-like growth factor I/somatomedin C: a potent inducer of oligodendrocyte development. *Proc Natl Acad Sci (USA)*, 83, 822-826 (1986).
- 122. J.F. Mill, M.V. Chao and D.N. Ishii, Insulin, Insulin-like growth factor-II and nerve growth effects on tubulin mRNA levels and neurite formation. *Proc Natl Acad Sci* (USA), 81, 7126-7130 (1985).
- 123. E. Recio-Pinto, F.F. Lanf and D.N. Ishii, Insulin and insulin-like growth factor II permit nerve growth factor binding and the neurite response in human cultured neuroblastoma cells. *Proc Natl Acad Sci (USA)*, **81**, 2562-2566 (1984).
- 124. A. Morrione, G. Romano, M. Navarro, K. Reiss, B. Valentinis, M. Dews, E. Eves, M.R. Rosner and R. Baserga, Insulin-like growth factor I receptor signaling in differentiation of neuronal H19-7 cells. *Cancer Res*, **60**, 2263-2272 (2000).
- 125. A. Morrione, M. Navarro, G. Romano, M. Dews, K. Reiss, B. Valentinis, B. Belletti and R. Baserga, The role of the insulin receptor substrate-1 in the differentiation of rat hippocampal neuronal cells. *Oncogene*, **20**, 4842-4852 (2001).
- 126. M. Valtieri, D.J. Tweardy, D. Caracciolo, K. Johnson, F. Mavilio, S. Altmann, D. Santoli and G. Rovera, Cytokine-dependent granulocytic differentiation. Regulation of proliferative and differentiative responses in a murine progenitor cell line. J Immnunol, 138, 3829-2835 (1987).
- 127. B. Cristofanelli, B. Vallentinis, S. Soddu, M.G. Rizzo, A. Marchetti, G. Bossi, A.R. Morena, M. Dews, R. Baserga and A. Sacchi, Cooperative transformation of 32D cells by the combined expression of IRS-1 and V-Ha-Ras. *Oncogene*, **19**, 3245-3255 (2000).

- 128. B. Valentinis, M. Navarro, T. Zanocco-Marani, P. Edmons, J. McCormick, A. Morrione, A. Sacchi, G. Romano, K. Reiss and R. Baserga, Insulin-receptor substrate-1, p70S6K, and cell size in transformation and differentiation of hemopoietic cells. *J Biol Chem*, 275, 25451-25459 (2000).
- 129. B. Valentinis, G. Romano, F. Peruzzi, A. Morrione, M. Prisco, S. Soddu, B. Cristofanelli, A. Sacchi and R. Baserga, Growth and differentiation signals by the insulin-like growth factor I receptor in hemopoietic cells are mediated through different pathways. *J Biol Chem*, 274, 12423-12430 (1999).
- 130. K. Reiss, J.Y. Wang, G. Romano, X. Tu, F. Peruzzi and R. Baserga, Mechanisms of regulation of cell adhesion and motility by insulin receptor substrate-1 in prostate cells. *Oncogene*, **20**, 490-500 (2001).
- 131. M. Navarro and R. Baserga, Limited redundancy of survival signals from the type 1-insulin-like growth factor receptor. *Endocrinology*, **142**, 1073-1081 (2001).
- 132. N.J. Mc Carthy, M.K.B. Whyte, C.S. Gilbert and G.I. Evan, Inhibition of Ced-3/ICE-related proteases does not prevent cell death induced by oncogenes, DNA damage or the Bcl-2 homologue Bak. J Cell Biol, 136, 215-227 (1997).
- 133. M.G. Myers, T.C. Grammer, L.M. Wang, X.J. Sun, J.H. Pierce, J. Blenis and M.F. White, Insulin receptor substrate-1 mediates phosphatidylinositol 3kinase and p70S6k signaling during insulin, insulin like growth factor 1 and interleukin-4 stimulation. *J Biol Chem*, 269, 28783-28789 (1994).
- 134. J.S. Greenberger, M.A. Sakakeeny, C.J. Humphries, C.J. Eaves and R.J. Eckner, Demonstration of permanent factor-dependent multipotential erythroid/neutrophil/basophil hemopoietic progenitor cell lines. *Proc Natl Acad Sci* (USA), 80, 2931-2935 (1983).
- 135. G.J. Pronk, J. McGlade, G. Pelicci, T. Pawson and J.L. Bos, Insulin-induced phosphorylation of the 46-and 52-kDa Shc proteins. *J Biol Chem*, **268**, 5748-5753 (1993).
- 136. T. Sasaoka, D.W. Rose, B.H. Jhun, A.R. Saltier, B. Draznin and J.M. Olefsky, Evidence for a functional role of Shc proteins in mitogenic signaling induced by insulin, insulin-like growth factor-1 and epidermal growth factor. *J Biol Chem*, 269, 13689-13694 (1994).
- 137. S.M. Twigg, M.C. Kiefer, J. Zapf and R.C. Baxter RC, Insulin-like growth factorbinding protein 5 complexes with the acid-labile subunit. Role of the carboxylterminal domain. *J Biol Chem*, 273, 28791-28798 (1998).
- 138. L.J. Cobb, D.A. Salih, I. Gonzalez, G. Tripathi, E.J. Carter, F. Lovett, C. Holding and J.M. Pell, Partitioning of IGFBP-5 actions in myogenesis: IGF-independent anti-apoptotic function J Cell Sci, 117, 1737-1746 (2004).
- 139. S. Mohan and D.J. Baylink, IGF-binding proteins are multifunctional and act via IGF-dependent and independent mechanisms. *J Endocrinol*, **175**, 19-31 (2002).
- 140. C. Lassarre, S. Hardouin, F. Daffos, F. Forestier, F. Frankeene and M. Binoux, Serum insulin-like growth factors and insulin-like growth factor binding proteins in the human fetus. Relationships with growth in normal subjects and in subjects with intrauterine growth retardation. *Pediatr Res*, **29**, 219-225 (1991).
- 141.R.I. Holt. Fetal programming of the growth hormone-insulin- like growth factor axis. *Trends Endocrinol Metab.* 13, 392-397 (2002).

- 142. P.D. Gluckman, J.H. Butler, R. Comline and A. Fowden, The effects of pancreatectomy on the plasma concentrations of insulin-like growth factors 1 and 2 in the sheep fetus. *J Dev Physiol*, **9**, 79-88 (1987).
- 143. M. Gourmelen, Y. Le Bouc, F. Girard and M. Binoux, Serum levels of insulin-like growth factor (IGF) and IGF binding protein in constitutionally tall children and adolescents. *J Clin Endocrinol Metab*, **59**, 1197-1203 (1984).

### Chapter 2

# EXPERIMENTAL MODELS FOR UNDERSTANDING THE ROLE OF INSULIN-LIKE GROWTH FACTOR-I AND ITS RECEPTOR DURING DEVELOPMENT

Jose Luis Trejo<sup>1</sup>, Eva Carro<sup>1</sup> and Deborah J. Burks<sup>2</sup> <sup>1</sup>Instituto Cajal, Madrid, Spain; <sup>2</sup>Instituto de Neurociencias de Castilla y Leon, Universidad de Salamanca, Salamanca, Spain

Key Words: Gene targeting; knockouts; receptor signal transduction; RNA interference; transgenes.

#### **1. INTRODUCTION**

During the last two decades, in vivo models have been employed extensively to elucidate key molecular mechanisms governing embryonic growth and development. Genetic manipulations of worms, flies, and mice provide powerful tools to unravel the complex program of development, and thus have revolutionized the study of factors and signaling pathways that regulate cellular differentiation, proliferation, and survival during embryogenesis and postnatal growth. The pioneering deletions of insulinlike growth factor (IGF)-I, IGF-II and/or the IGF-I receptor in mice have demonstrated definitively that this growth factor system provides essential signals for the regulation of embryonic and postnatal development in vertebrate species<sup>1,2</sup>. More recently, genetic ablations of signaling molecules have reinforced the role of IGF-I in development, whereas the advent of conditional gene targeting has challenged historical dogma regarding the importance of circulating IGF-I for postnatal growth. Given the critical contributions of animal models to our current knowledge of IGF-I function, this chapter reviews the phenotypes of various mice bearing genetic modifications of the IGF system. Interestingly, the use of divergent model systems has revealed that many aspects of development are evolutionarily conserved, particularly those events regulated by signals from the insulin/IGF-I receptor. Therefore, this chapter also includes relevant findings from *Caenorhabditis elegans* and *Drosophila*, with the aim of illustrating how model organisms represent indispensable scientific tools for understanding developmental pathways and ultimately, for obtaining insights into the molecular mechanisms of human disease such as cancer.

#### 2. THE TECHNOLOGY AND TERMINOLOGY OF CONTEMPORARY EXPERIMENTAL MODELS

Model systems have been used since the era of Mendel to understand basic, conserved biological processes. However, prior to the development of gene targeting, analysis of the inactivation or modification of a gene in mammals was limited to rare, spontaneous mutations that produced obvious phenotypes such as an inheritable human disease. Thus, the ability to generate genetically modified mice represents one of the major milestones of modern science as it has empowered researchers with very precise tools to determine the physiological function(s) of a gene product. Recognizing that techniques within this field progress rapidly, we will discuss superficially the more common methods for genetic manipulation of model systems.

#### 2.1 Conventional knockouts

Deletion or knockout of a gene is the definitive way to abolish its expression. Knockouts in mice have been typically accomplished by deleting or replacing one or more crucial exons with a neomycin resistance gene using *in vitro* recombination in embryonic stem (ES) cells, followed by selection for the recombined (knockout) cells<sup>3</sup>. These cells are then injected into host blastocysts to produce chimeric embryos. Chimeric adults in which the knockout cells have "gone germline" are then bred in stages to obtain homozygous knockout mice. Although the success of this technique was somewhat tentative with the onset of its use in the late 1980s, the establishment of core facilities and companies dedicated solely to the production of knockout mice has perfected the technology, and thus these methods are now considered routine. However, the basic knockout technique still represents a major investment of time and money. In addition, some
gene ablations are incompatible with development or postnatal survival. The later complication is illustrated by the conventional knockouts of IGF-I and its receptor; IGF-I null mice display extremely low survival rates during the early postnatal period and  $IGF-IR^{-/-}$  animals die at birth as a result of respiratory problems<sup>1,2</sup>.

## 2.2 Conditional knockouts

As mentioned previously, conventional germline mutations can be lethal, in which case there is no mouse for the study of gene function. This biological reality prompted the development of methods for conditional gene targeting including tissue-specific and/or inducible deletion of a gene<sup>4,5</sup>. In addition, if a gene exerts its function at several stages of development and/or in multiple cell types, a whole-body conventional knockout may produce various phenotypes, making it difficult to discern the authentic physiological role(s) of the gene product. Here again conditional knockouts offer advantages because the deletion can be restricted to a particular point of development or to a cell-type of interest.

The most typical strategy for production of conditional knockouts exploits the bacteriophage-derived Cre-loxP recombination system<sup>4</sup>. Cre recombinase recognizes a sequence motif of 34 bp, known as loxP sites<sup>6</sup>. Mice with loxP-flanked (floxed) target genes are generated in several steps<sup>7</sup>. Conditional targeting of endogenous genes consists of flanking the gene sequence with loxP sites in ES cells via classical homologous recombination and subsequently deleting the neomycin resistance marker by transient transfection with Cre-encoding plasmid. This approach yields ES cell mutants in which the gene of interest is either flanked by loxP sites or deleted<sup>6</sup>. Either mutation can be transmitted into the germline. The conditional deletion of loxP-flanked genes is finally achieved by crossing the mutant line with a Cre-transgene in which the Cre enzyme is expressed under the control of cell-type specific or inducible promoter. (Collections of Cre-mice are maintained by various sources and include the expansive list of Andras Nagy at www.mshri.on.ca/nagy.) Liver-specific deletion of the IGF-I gene has been accomplished by crossing floxed IGF-I mice with transgenics expressing Cre via the albumin promoter which is highly active in liver<sup>8,9</sup>. Note that Cre-based conditional targeting can also be used for the introduction of subtle gene mutations for example, knockins of gain of function or loss of function mutants.



*Figure 1.* The RNA interference pathway. In the initiation step, double-stranded DNA (dsDNA) is cleaved by the RNAse-III-like enzyme Dicer into small interference RNAs (siRNA). These siRNAs contain characteristic 5' phosphorylated ends and 3' overhangs of 2 nt. Next, the siRNAs are incorporated into the RNA-induced silencing complex (RISC). Unwinding of the siRNA duplex requires ATP. Once unwound, the single-stranded antisense directs RISC to mRNA bearing a complementary sequence. As a consequence of this base-pairing, the mRNA is degraded, effectively silencing the target gene.

### 2.3 RNA interference/knockdowns

RNA interference (RNAi), also termed RNA silencing or posttranscriptional gene silencing, is a biological phenomenon by which small, double-stranded RNA (dsRNA) molecules induce sequence-specific degradation of homologous single-stranded RNA<sup>11,12</sup>. From a practical perspective, RNAi can be exploited to target gene expression and has already proven itself as a very powerful technique to knock down specific genes for the assessment of their physiological roles. Interest in gene silencing expanded when it was discovered that the introduction of dsRNA into Caenorhabditis elegans silences gene expression in a homologydependent manner much more efficiently than does either sense or antisense RNA<sup>13</sup>. The technique is rapid and can inhibit the expression of most genes by more than 90%. Illustrating these advantages of RNAi technology, two groups recently reported the screen of C. elegans chromosomes I and III (a total of a third of the genome) for RNAi phenotypes<sup>14,15</sup>. Using RNAi in developing Drosophila embryos, Kim et al recently analyzed more than 5,800 genes (40% of all Drosophila genes) to identify transcription factors and cell signaling proteins required for heart development<sup>16</sup>. Thus, not surprisingly, RNAi is being proclaimed as a major scientific advance of the 2000s.

RNAi occurs in several steps that can be induced through transfection or microinjection of long dsRNA (Fig. 1). First, the dsRNA is cleaved into 19to 23-nt RNA fragments known as small interfering RNAs (siRNA) by an evolutionarily conserved ribonuclease III-like protein Dicer<sup>17</sup>. siRNAs act as intermediates in the RNAi pathway, which is thought to protect cells from harmful transposons and highly repetitive sequences by targeting their RNA transcripts for endonucleolytic cleavage and subsequent exonucleolytic degradation<sup>18</sup>. SiRNA is incorporated into the RNA-induced gene complex (RISC) that targets and cleaves mRNA complementary to the siRNA<sup>12</sup>. The siRNA must be 5' phosphorylated in order to enter the RISC. The duplex siRNA is then unwound, allowing the antisense strand to direct the RISC to homologous target for cleavage by endonucleases. Cleavage occurs at a single site in the center of the duplex region, 10 nt from the 5' end of the siRNA.

Until very recently, only researchers working with plants or *C. elegans* could utilize RNAi technology to obtain insight into gene function. This limitation was largely attributable to the fact that only RNA molecules <30 bp can be used to induce RNAi; longer nucleotide sequences in vertebrate cells induce a nonspecific, type I interferon response that triggers mRNA degradation<sup>19</sup>. However, the groundbreaking discovery that chemically synthesized, 21- to 23-nucleotide, double-stranded RNAs can function as siRNAs to silence genes in mammalian cells makes it possible now to contemplate RNAi in mammalian systems<sup>20</sup>.

Other characteristics of RNAi that have delayed its application to the manipulation of vertebrate genes are amplification activity and the duration of gene silencing. In plants and nematodes, RNAi activity is long term and disseminates throughout the organism via an unknown amplification mechanism<sup>21</sup>. However, in mammalian cells, amplification activity seems absent, and interference activity is transient, lasting for only 3 to 5 days<sup>22</sup>. To circumvent these problems with RNAi technology in mammalian systems, DNA expression vectors have been developed to express hairpin or duplex siRNA, which employ the type III class of RNA polymerase promoters to drive the expression of siRNA molecules<sup>12</sup>. The majority of these systems take advantage of the ability of Dicer to process short hairpin RNAs into siRNAs and thereby silence the target gene. These technologies make it now possible to perform gene-silencing experiments in various mammalian cell types and cell lines, as well as to create transgenic animals that stably "knock down" a target gene.

## 3. MOUSE MODELS OF IGF-I DEFICIENCY

The insulin/IGF-I family of growth factors and their receptors regulates major aspects of mammalian development including systems of growth, metabolism, and reproduction<sup>23</sup>. At present, there are nine genes that encode insulin-like ligands in vertebrates: the two non-allelic insulin genes *ins1* and *ins2*, *igf1*, *igf2*, *relaxin*, and four insulin-like molecules of undetermined function *ins3,4,5* and  $6^{24-26}$ . Mutant models of IGF action have provided

unique opportunities for *in vivo* studies of the growth-promoting actions of IGF-I in the developing mouse. Moreover, by targeting the genes encoding individual ligands of the IGF family and/or their receptors, knockout strategies have enabled researchers to distinguish the developmental function of IGF-I from those of IGF-II and insulin.

## 3.1 Null mutants of *igf1*

Homozygous disruption of the *igf1* gene (*IGF-I*<sup>-/-</sup> or IGF-I KO) produces marked intrauterine and postnatal growth retardation, and, depending upon genetic background, reduced postnatal survival<sup>1,2,27</sup>. At birth, IGF-I-deficient mice are smaller in size by 40% in comparison to their wild-type littermates but by 2 months of age, the size of the surviving IGF-I KO is only 30% of normal owing to the role of IGF-I during the postnatal period (Fig. 2). However, the hemizygous disruption of only a single IGF-I gene  $(IGF-I^{+/-})$ has little impact on embryonic development and postnatal growth. display generalized organ hypoplasia Knockouts of IGF-I with developmental defects in bone, muscle, and reproductive systems. IGF-I deficiency also produces a central nervous system phenotype that includes reduced brain size, hypomyelination, reduced axonal diameters, and loss of selective neuronal populations<sup>28-30</sup>. A more detailed description of these phenotypes can be found in subsequent chapters which are dedicated to the discussion of specific organ systems.

Ablation of the igf2 gene causes impairments of *in utero* growth (Fig. 2); similar to IGF-I-deficient mice,  $IGF-II^{/-}$  animals are 60% of normal size at birth<sup>2,31</sup>. However, in contrast to the effects of disrupting igf1, postnatal growth of IGF-II knockouts is normal, consistent with the fact that this factor in mice is expressed only during embryogenesis and has little if any influence on postnatal development<sup>32</sup>. Thus, the conventional whole-body deletions of igf1 and igf2 have been instrumental in distinguishing the roles of these growth factors in prenatal and postnatal mouse development. These original models confirmed that the development of most, if not all, tissues and organs is regulated to some degree by IGF-I.

Interestingly, the null *igf1* phenotype in mice is remarkably similar to that observed in a single individual carrying a homozygous partial deletion of the *IGF-1* gene<sup>33</sup>. This patient presented with severely impaired pre- and postnatal growth, microcephaly, mental retardation, insulin resistance, and sensorineural deafness. These defects are a direct consequence of the inability to produce IGF-I either locally or systemically. Treatment of this 17 year-old patient with recombinant human IGF-I improved linear growth and insulin sensitivity by restoring IGF-I levels and by normalizing the levels of circulating GH and insulin<sup>34</sup>.

# **3.2** Conditional knockouts of IGF-I: Assessing local vs. systemic hormonal effects

Postnatal growth and development are coordinated by genetic and environmental influences and numerous growth factors. LeRoith and colleagues have reviewed extensively the critical role of the GH/IGF-I axis in these processes<sup>35-37</sup>. The basic assumption of the somatomedin hypothesis is that circulating IGF-I is produced by the liver in response to GH and mediates the effects of GH in peripheral tissues. Although the GH/IGF-I axis is a closely coordinated system, the biological distinctions between these factors have become apparent only in recent years through the characterization of clinical syndromes and the genetic manipulation of mice which has enabled investigators to re-examine hypotheses regarding the GH/IGF-I axis. Mice lacking both IGF-I and the receptor for GH have demonstrated that these two factors act independently and synergistically; IGF-I promotes both pre- and postnatal growth while GH is required only during postnatal development<sup>38</sup>. Moreover, the fact that IGF-I is produced locally by many tissues has suggested the importance of autocrine/paracrine actions of this growth factor.

The low survival rate of *igf1* null neonates has greatly limited their use in analyzing the function of circulating IGF-I during the postnatal period and in adult animals<sup>2</sup>. Thus, to assess the importance of endocrine vs. autocrine/paracrine IGF-1 in development and growth, conditional knockouts of this hormone in mice have been generated. Using a floxed exon of *igf-1* and transgenics expressing Cre under control of the albumin promoter, Yarkar *et al.* accomplished liver-specific deletion of  $igfl^8$ . Another liver-specific ablation of *igf1* was produced via an inducible interferon promoter to drive the tissue-specific expression of Cre<sup>39</sup>. In both liverdeficient (LID) models, circulating levels of IGF-I were reduced by 75%. However, the animals were comparable in body weight, body length, and femoral length to wild-type littermates (Fig. 2). Therefore, these models demonstrate that hepatic IGF-I as a source of circulating IGF-I is not as crucial as previously believed for normal postnatal growth, indicating that kidney, spleen, fat, muscle, and bone may supply sufficient systemic IGF-I to promote normal development. Furthermore, these results suggest that the local production of IGF-I is critical for postnatal growth and that the growth retardation observed in the whole-body *igf1* knockout reflects the absence of autocrine/paracrine actions of IGF-I. Thus, the function of circulating IGF-I in adult animals may be compensated for by other hormones and/or by locally produced IGF-I.



*Figure 2.* Alterations in mouse growth resulting from various genetic manipulations of the insulin/IGF-I system in mice. Growth impairment is expressed as the percentage of wild-type control body weight.

However, the significant reduction in circulating IGF-I levels is not totally without physiological consequence as defects have been noted in discrete brain regions of LID mice. Peripheral IGF-I is known to regulate adult dentate neurogenesis in rodents<sup>40,41</sup>. Interestingly, LID knockout mice display a 40% reduction in granule neuron density within the dentate gyrus of hippocampus compared with wild-type littermates. However, no differences in the number of CA3 or cerebral cortex neurons were noted in LID mice. In addition, neurogenesis within the granule cell layer of the dentate gyrus is reduced 33% in LID mice. These IGF-I-deficient animals also display impaired spatial learning and synaptic plasticity. The reduced neurogenesis in adult hippocampus correlates with the impairment of spatial learning, as assessed by the Morris water maze test<sup>42</sup>.

Recent evidence from experiments in IGF-I mutant mice indicates that IGF-I has an important role in recovery from neural injury and neurodegenerative diseases<sup>43</sup>, demonstrating that serum IGF-I is required for neuroprotection<sup>42,44,45</sup>. IGF-I levels decline with age and recently this hormone has been implicated in the pathogenesis of Alzheimer's disease (AD). Interestingly, brains of the LID transgenic mice contain a premature increase in levels of A $\beta$ ; this abnormality is present at 3 months of age and becomes more severe with aging<sup>46</sup>. These studies demonstrate that low levels of IGF-I are linked to premature brain amyloidosis and low brain levels of transthyretin and albumin.

These findings in LID mice suggest that although liver-derived IGF-I is not crucial for normal growth and development during pre- and postnatal stages, circulating IGF-I may be required to maintain normal brain function in adults and to avoid premature aging and/or pathologic processes such as the deposition of amyloid plaques. Serum IGF-I levels may reflect a mechanism to coordinate peripheral metabolism with central processes such as the control of adult neurogenesis. Thus, IGF-I represents a potential therapeutic target in the treatment and prevention of neurodegenerative diseases and supplies a model for understanding the pathogenesis of some disorders, such as depression which have recently been associated with changes in IGF-I levels<sup>47</sup>.

## 3.3 Classical IGF-I receptor knockout mice

The receptors for insulin (IR), IGF-I (IGF-IR), and the orphan insulinlike receptor (IRR) are plasma membrane proteins that belong to the family of ligand-activated tyrosine kinases<sup>25,48,49</sup>. However, unlike other receptor tyrosine kinases, the IR and IGF-IR exist as homodimers composed of two identical  $\alpha/\beta$  monomers or as heterodimers consisting of two different receptor monomers (e.g.,  $IR_{\alpha\beta}/IGF-IR_{\alpha\beta}$ ). Ligand binding induces a conformational change that enables the intracellular domain of these receptors to bind ATP and undergo autophosphorylation. Once activated, the intrinsic tyrosine kinase activity of the IR and IGF-IR phosphorylates intracellular substrates<sup>50,51</sup>. In contrast to IR and IGF-IR, the receptor for IGF-II (also known as the mannose-6-phosphate receptor) is not a tyrosine kinase but rather consists of a large extracellular domain with 15 repeats and a small domain with homology to fibronectin<sup>52,53</sup>. Insulin and IGF-I bind to their own receptors with high-affinity, but at high hormone concentrations, may bind with low affinity to the cognate receptors. IGF-II, however, exhibits high affinity binding to both IR and IGF-IR, and thus may activate signaling cascades via either receptor. The receptor for IGF-II has no signaling capacity; the binding of IGF-II to its receptor serves to clear the hormone from circulation<sup>25</sup>.

To test the role of IGF-IR in development and postnatal growth, the laboratory of Efstratiadis generated the classical knockouts of  $igf1r^1$ . Growth retardation was observed in IGF-IR-deficient mice as early as E10.5 and at birth these animals were 45% to 55% smaller than normal. However, the  $IGF-IR^{-/-}$  mice die at birth as a result of irreversible respiratory failure. Interestingly, double knockouts of igf1 and igf1r display the same degree of growth retardation as the simple igf1 null mice<sup>1,2</sup>, demonstrating that all physiological effects of IGF-I are mediated by its own receptor. Moreover,  $IGF-IR^{-/-}$  mice exhibit more profound intrauterine growth retardation than knockouts of either IGF-I or IGF-II, suggesting that these factors via activation of the IGF-IR to compensate the absence of one another during embryonic development<sup>54</sup>.

The perinatal lethality caused by IGF-IR deficiency has restricted detailed analysis of this model to embryonic development. The brains of *igf1r* null mice are small relative to controls, although this defect is not so obvious as the reduced body weight. Moreover, the mutant mice display a striking increment in neuronal density<sup>1</sup>. Impaired brain growth is apparent from E14 to E18, as  $IGF1R^{-1}$  embryos exhibit reduced volume and increased density of cells in the mantle zone of the spinal cord and brainstem.<sup>1</sup> Interestingly, the absence of the IGF-IR appears specifically to alter certain populations of neurons and brain regions<sup>28</sup>, for example, the number of parvalbumin<sup>+</sup> neurons of the hippocampus in IGF-IR<sup>-/-</sup> embryos compared with wild-type littermates as determined by stereological counting. These observations reveal that, in addition to its global role in brain developmental, the IGF-IR exerts additional or more specialized functions in certain neuronal populations. In sharp contrast, brain-specific deletion of the insulin receptor via the use of nestin-Cre mice has no effect on brain development; the size and morphology of the brain from NIRKO mice is absolutely normal, further demonstrating that brain growth and development are regulated by the IGF-I/IGF-IR pathway<sup>55</sup>.

Recently, another line of whole-body IGF-IR knockouts has been generated by deleting exon 3 of igf1r using Crellox recombination<sup>56</sup>. This model has been used to study the role of IGF-IR in aging where the authors report that  $IGF1R^{+/-}$  females live 33% longer than wild-type females. Confirming the earlier studies of Liu *et al.*<sup>1</sup>, *IGF-IR*<sup>+/-</sup> mice are apparently normal with respect to development, energy metabolism, nutrient uptake, physical activity, fertility, and reproduction. Interestingly, female heterozygous mice display greater resistance to oxidative stress, a known determinant of ageing. These results are both interesting and promising; on the one hand, IGF-IR signaling is required for protective functions such as the survival of neurons but this recent publication suggests that reduced IGF-IR signals may increase vertebrate lifespan, analogous to effects of mutations in the insulin-like receptor system in Caenorhabditis elegans and Drosophila. However, the significance of the longer lifespan of IGF-IR heterozygous females in this study must be carefully considered since the female wild-type controls died slightly sooner than controls of similar aging studies in mice<sup>56</sup>. Nevertheless, these observations imply that reducing oxidative stress may represent an effective way to diminish aging in all tissues and suggest a novel role for the IGF-IR in regulation of lifespan.

### **3.4** Conditional gene targeting of the IGF-IR

Although the characterization of the classical knockout of the *igflr* suggested that embryonic development of brain, skin, muscle, and lung are more dependent on IGF-I than other organs<sup>1</sup>, the perinatal lethality associated with this model has hampered extensions of these studies to postnatal development. As discussed earlier, this is frequently a problem that the production of whole-body accompanies knockouts. Moreover. compensatory mechanisms and/or selective feedback mechanisms specifically related to the pleiotropic effects of the insulin/IGF-I system may complicate analysis of the phenotype produced by gene targeting. One way to circumvent these problems is to create conditional knockouts. Recently, several types of conditional IGF-IR receptor knockouts have been produced to address various biological questions.

Holzenberger and colleagues devised a very clever Cre/lox-based strategy to achieve gene dosage of the igf1r in mice<sup>58</sup>. By crossing a line of Cre transgenics that produce a pattern of mosaic-early embryonic-ubiquitous gene deletion (MeuCre) with mice bearing the floxed IGF-IR allele, conditional mutants expressing various levels (5% to 82% of normal) of the IGF-I receptor were obtained. Animals were divided into groups of either strong (>50%, XS mice) or moderate (<50%, M mice) IGF-IR deficiency. Growth impairment in XS mice was noted by 3 weeks of age and persisted throughout the lifespan of the animals as compared to the relatively normal growth of M control littermates. The defect in postnatal growth was more pronounced in males than in females; at 9 weeks of age XS males weighed about 30% less than wild-type controls whereas XS females displayed a 20% reduction in body weight. By 10 to12 months of age, most organs and tissues displayed specific weight defects, with skin, bone and connective tissue, muscle, spleen, heart, lung, and brain being the most susceptible to the reduction in IGF-IR expression.

The generation of tissue-specific knockouts of IGF-IR has served to confirm requirements for the local actions of IGF-I in some cases, while in others the phenotype resulting from the cell-type specific ablation of *igf1r* appears at odds with findings from the whole-body deletion of this receptor. Mice harboring the osteoblast-specific deletion of IGF-IR display normal size and weight but display a striking decrease in cancellous bone volume, connectivity, and trabecular number, and an increase in trabecular spacing<sup>59</sup>. These bone abnormalities correlate with a marked decrease in the rate of mineralization of osteoid, confirming that IGF-I is essential for coupling matrix biosynthesis to sustained mineralization. However, ablation of IGF-IR is provide the growth and development of insulin-producing cells but reduces expression of Glut2

and glucokinase leading to defective glucose-stimulated insulin secretion and impaired glucose tolerance<sup>60</sup>. These observations contrast findings from the classical IGF-IR knockout in which a reduction in beta cell number and morphological abnormalities have been reported<sup>57</sup>. These discrepancies between the effects of totally deleting the IGF-IR or of specifically targeting the deletion to  $\beta$ -cells may reflect the fact that (1) the IGF-IR may form a part of some systemic coordination during the course of embryonic development or (2) tissue-specific deletion may engage compensatory mechanisms.

# 4. GENETIC ABLATION OF IRS SIGNALING PROTEINS

Although a better understanding of insulin/insulin-like growth factor (IGF) action has evolved with the use of animal models, the discovery of insulin receptor substrate (IRS) proteins and their capacity to link cell surface receptors to the intracellular signaling cascades has provided an important advance in this field. The binding of insulin and IGF-I to their cell surface receptors stimulates receptor autophosphorylation and activates their intrinsic tyrosine kinase activity, which leads to the phosphorylation of intracellular substrates<sup>49,61</sup>. The IRS proteins were initially identified as large molecular weight, tyrosine-phosphorylated proteins that occur in cells during insulin or IGF-1 stimulation<sup>50,62</sup>. cDNA cloning<sup>62-64</sup> revealed that IRSproteins contain multiple tyrosine phosphorylation sites that bind and activate SH2-domain containing effector proteins<sup>65</sup>, including the regulatory subunit of the lipid kinase phosphatidylinositol-3 kinase (PI-3 kinase), Grb2, nck, and SHP2 (Fig. 3). Of these, activation of PI-3 kinase has been implicated in the action of insulin/IGF-I upon glucose transport, glycogen synthesis, protein synthesis, cell growth, and differentiation<sup>50</sup>. The lipid products of PI-3 kinase are necessary for the activation of Akt, which in turn regulates other downstream signaling molecules by phosphorylation<sup>66</sup>. The FoxO subfamily of forkhead transcription factors are targets of Akt; phosphorylation of FoxO1 by insulin/IGF-I signaling excludes this factor from the nucleus, thereby down-regulating its antiproliferative effect in cells<sup>67-69</sup>. Thus, the IRS protein signaling network mediates the pleiotropic effects of insulin and IGF-1 upon cellular function.

The first member of this family to be identified was IRS-1, which provided the initial example of a cytoplasmic docking protein coupling an activated receptor tyrosine kinase to various signaling proteins<sup>62</sup>. However, when knockouts of IRS-1 were stimulated with insulin, a large tyrosine phosphorylated protein was observed in tissue lysates, demonstrating the

#### **Experimental Models**

existence of additional IRS proteins<sup>63,64</sup>. Indeed, currently there are four members of the IRS-protein family: IRS-1 and IRS-2 are ubiquitously expressed; IRS-3 is predominantly expressed in adipose tissue<sup>70</sup>; IRS-4 is detected mainly in kidney, pituitary and thymus<sup>71,72</sup>. Given the highly conserved structure of the four IRS proteins, it was considered during the initial stages of their discovery whether these molecules might represent a system of signaling redundancy. However, generation of mouse knockout models for each IRS protein clearly reveals specific physiological functions.



Figure 3. IRS-mediated signal transduction.

# 4.1 Knockouts demonstrate that IRS-1 is the major mediator of IGF-I growth signals

Deletion of IRS-1 reduces animal size by 40% to 50%, suggesting that IRS-1 mediates the IGF-I signaling required for embryonic growth and development<sup>57,63</sup>. By contrast, IRS-2 deficient mice are only 10% smaller than control littermates but deletion of IRS-2 produces a severe diabetic phenotype owing to peripheral insulin resistance and reduced beta cell mass

and function<sup>73,74</sup>. *IRS-1*<sup>-/-</sup> mice display mild peripheral insulin resistance; however, diabetes does not occur in this model due to hyperplasia of pancreatic beta cells<sup>63,73</sup>. Deletion of *IRS-3* is without effect on growth, glucose metabolism, or other physiological parameters<sup>75</sup>. Mice deficient for IRS-4 display very mild reduction in body growth and the females are slightly less fertile than controls<sup>76</sup>. Therefore, based on the analysis to date, IRS-3 and IRS-4 do not appear to play major roles in physiological insulin/IGF-I action. However, the differences in the phenotypes of *IRS-1*<sup>-/-</sup> and *IRS-2*<sup>-/-</sup> mice strongly suggest that these proteins have unique and critical physiological roles, despite the similarity of their tissue distribution.

To further distinguish the roles of IRS-1 and IRS-2 in development and growth, carbohydrate metabolism, and  $\beta$ -cell function, knockouts of these signaling proteins were intercrossed to generate animals with various states of IRS deficiency<sup>57</sup>. Growth curves were generated based on daily weights from birth to 30 days of age. Compound heterozygous mice (*IRS1<sup>+/-</sup>/IRS2<sup>+/-</sup>*) weighed 25% less than wild-type control littermates. *IRS1<sup>+/-</sup>/IRS2<sup>-/-</sup>* mice were of similar size to *IRS1<sup>-/-</sup>* animals, whereas *IRS1<sup>-/-</sup>/IRS2<sup>+/-</sup>* were 70% to 75% smaller than normal littermates. *IRS1<sup>-/-</sup>/IRS2<sup>+/-</sup>* are amongst the smallest viable mice that have been generated by genetic manipulations. Double mutants null for both *irs1* and *irs2* were not observed in this study; however, Kubota *et al.* report the low viability of these double knockouts in lines from a different genetic background<sup>74</sup>. The findings of these intercrosses further implicate IRS-1 as the principal mediator of IGF-I-regulated somatic growth and suggest that although IRS-2 participates in development and postnatal growth, it cannot replace the function of IRS-1.

These *in vivo* observations are consistent with results from *in vitro* studies of studies of  $IRS1^{-/-}$  cells. Embryonic fibroblasts derived from IRS-1 deficient mice display a 70% to 80% reduction in IGF-1-stimulated cell growth and parallel decreases in IGF-1-stimulated S-phase entry, PI-3 kinase activity, and induction of the immediate-early genes *c-fos* and *egr-1*<sup>77</sup>. While recombinant expression of IRS-1 in this cell system fully reconstituted the IGF-I signals required for cell growth, overexpression of IRS-2 had only slight effects on cell-cycle progression. Thus, IRS-1 and IRS-2 are not functionally interchangeable despite the high degree of structural similarity and a common tissue distribution.

# 4.2 IRS-2 knockouts: IGF-I signals in the development and function of neuroendocrine tissues

Given that insulin and IGF-I have a wide range of physiological effects and use common signaling pathways to accomplish these tasks, the IRS proteins add a unique layer of specificity and control to the system. This is nicely illustrated by the neuroendocrine phenotypes produced specifically by the deletion of irs2. IRS-2-deficient mice develop diabetes because of defects in both insulin action and insulin production<sup>73</sup>. As early as 3 days postpartum male,  $IRS2^{-/-}$  animals have elevated random blood sugars and by 6 weeks of age male animals, display hyperglycemia and markedly abnormal glucose tolerance tests. Untreated male animals die of nonketotic hyperosmolar coma between 12 and 20 weeks of age. IRS-2 knockout mice are also resistant to the peripheral actions of insulin. However, this diabetic phenotype displays a sexual dimorphism: female  $IRS2^{-1}$  mice are only mildly glucose intolerant and do not develop diabetes until 6 months of  $age^{78}$ . Morphometric analysis of pancreatic islets from 4-week-old IRS-2 knockouts shows that they have 50% reduction in beta cell mass compared to control mice, demonstrating that IRS-2 signals are required for the normal development, growth, and/or differentiation of insulin-producing cells<sup>73</sup>. Indeed, more recently it was been shown the IGF-1-receptor/IRS-2 pathway is essential for normal  $\beta$ -cell development and compensation to insulin resistance<sup>57</sup>. Embryos homozygous for IGF-1-receptor-null alleles show a marked impairment of β-cell development, and animals lacking IRS-2 and heterozygous for IGF-1-receptor-null alleles display impaired β-cell development, proliferation, and survival<sup>57</sup>. Thus, the observation that deletion of IRS-2 impairs beta cell function suggests that IGF-I/IRS-2dependent signaling pathways are critical for regulation of the proliferative and neogenic responses of the beta cell.

Analysis of female *irs2* null mice has revealed that these animals are infertile and have disorders of energy homeostasis that are independent of the diabetic phenotype<sup>78</sup>. IRS-2-null females have small, anovulatory ovaries with reduced numbers of follicles and an absence of corpora lutea. Ovaries are resistant to superovulation, suggesting intrinsic defects in ovarian function. However, plasma levels of luteinizing hormone, prolactin, and sex steroids are low, which suggests defects at the hypothalamic level as well as in the gonads and pituitary. Pituitaries of IRS-2-deficient mice are decreased in size and contain reduced numbers of gonadotrophs. Interestingly, food intake, body weight, and fat deposition are also dysregulated in IRS-2 knock-out females. Taken together, these observations suggest that IRS-2 pathways might integrate reproductive function and energy homeostasis in mammalian systems. Several lines of evidence have suggested that insulin and IGFs may function in the hypothalamo-pituitary-gonadal axis. Insulin, IGF-1 receptors, and IRS proteins are present in the hypothalamus and pituitary gland<sup>26,79</sup>. Furthermore, as discussed more extensively in the text that follows, the insulin/IGF-1 receptor homolog in Caenorhabditis elegans regulates development, reproduction, and longevity in response to environmental signals such as food. Mutations in these pathways can induce

developmental arrest at the dauer stage and reduce fertility<sup>80</sup>. Similarly, deletion of CHICO, the *Drosophila* IRS protein, causes female sterility as

well as reduced somatic growth and increased lipid storage<sup>81</sup>. During development and regeneration of the mammalian brain, insulin and IGF-I have important functions in brain including metabolic, neurotrophic, and neuroendocrine actions. Interestingly, the brains of IRS-2 knockouts are significantly reduced in size (40%) compared to age-matched controls<sup>82</sup>. IRS-2-deficient mice are of normal body size, and thus display a significant reduction in body:brain ratio. Reduced brain size is apparent during early development; at e16, the brain weight of  $IRS2^{-1}$  mice is already significantly reduced in comparison to control embryos<sup>82</sup>. Analysis of neuronal proliferation and survival during embryogenesis and postnatal development has revealed that the small brain phenotype reflects impaired IGF-I-mediated proliferation rather than enhanced apoptosis. Brain-specific deletion of the insulin receptor has no effect on brain size<sup>55</sup>. Consistent with the role of IGF-I/IRS-2 signals in brain development, deletion of chico produces a pin-head phenotype, flies with reduced head size<sup>81</sup>. Thus, while IRS-1 has evolved to regulate global somatic growth in mammals, IRS-2 has acquired a more specialized function in mediating the local effects of insulin/IGF-I during the development and function of various neuroendocrine tissues.

	Development and growth	Metabolism	Neuroendocrine system
IRS-1	Severely impaired intrauterine and postnatal growth (40-50% reduction in size). Enhanced pancreatic $\beta$ -cell mass.	Mild insulin resistance in skeletal muscle, hyperinsulinemia, impaired adipocyte differentiation.	Reduced fertility.
IRS-2	Slight impairment of intrauterine growth (10%). Reduced pancreatic $\beta$ -cell development and growth. Small brain. Reduced pituitary size. Abnormal development of testes and ovary.	Profound hepatic insulin resistance, hyperinsulinemia, abnormal glucose intolerance and hyperglycemia, mild obesity, dyslipidemia, hypertension.	Female infertility, abnormal reproductive hormone levels. Mild obesity, hyperphagia, hyperleptinemia. Dysregulated Tau phosphorylation.
IRS-3	None	None	None
IRS-4	Mild growth retardation (10%) of males.	Mild glucose intolerance.	Reduced female fertility.

Table 1. Summary of phenotypes resulting from deletions of IRS signaling proteins.



Figure 4. Evolutionarily conserved elements of the insulin/IGF-I signaling system.

## 5. THE INSULIN/IGF-I PATHWAY IN THE CAENORHABDITIS ELEGANS MODEL

The extensive knowledge of the biology of the nematode *Caenorhabditis* elegans biology together with the complete and well-annotated genome sequence make these worms a very attractive experimental model<sup>83</sup>. Moreover, the ability to use forward and/or reverse genetic screens in this model provides powerful tools to identify genes required for specific cell functions. As discussed earlier in this chapter, the recent application of RNA interference technology to *C. elegans* has expanded the advantages of this model system to a new level; this simple method allows for rapid, efficient gene inactivation, and thus facilitates identification of loss-of-function phenotypes of genes. Indeed, based on this new technique, a systematic study of the function of every *C. elegans* gene is currently in progress.

The identification of an insulin-like signaling cascade in *C. elegans* has provided novel insights into the molecular mechanisms of insulin action in mammals. A neurosecretory system regulates whether these animals enter the reproductive life cycle or arrest developmentally at the dauer diapause stage<sup>84</sup>. When food is limited, young animals become dauers instead of developing to adulthood<sup>85</sup>. *daf-2*, a key gene in the genetic pathway that

mediates this endocrine signaling, encodes a homolog of the insulin/IGF-I receptor<sup>86-88</sup>. Unlike vertebrates, these organisms contain a single insulin/IGF receptor gene and many developmentally regulated insulin-like genes. However, similar to insulin signaling in mammals, DAF-2 action requires *age-1*, which encodes a PI-3 kinase<sup>80,89-91</sup> and *akt-1* encoding Akt/PKB-like proteins which are activated by the lipid products of PI3-kinase<sup>92</sup>. The DAF-2 pathway controls both reproductive development and normal senescence; loss of function mutants of *daf-2*, *age-1* or *akt-1* arrest development at the dauer stage and convey an increase in life-span<sup>80,87,89,90</sup>. Recently, orthologs of the p50/55 regulatory subunit of PI-3 kinase and IRS proteins have been identified in *C. elegans* and genetic analysis has demonstrated that *aap-1* and *ist-1* function in the *daf-2* pathway<sup>93</sup>. Mutants of *aap-1* arrest in dauer larval stage and display an increase in lifespan. IST-1 appears to be required for a pathway that may act in parallel to AGE-1<sup>93</sup>.

Some mutations suppress, to varying degrees, the effect of *daf-2* loss-offunction mutations, presumably because these genes counteract the effect of insulin signaling. Two of these genes, daf-16 and daf-18, have been implicated in PI-3 kinase signaling<sup>94.97</sup>. Mutations in *daf-16*, which encodes the C. elegans homolog of the mammalian FOXO forkhead transcription factors, completely suppress the *dauer* phenotype produced by daf-2 mutants<sup>80,89</sup>. Several lines of evidence indicate that FOXO1 is a transcriptional promoter and that its activity is inhibited by Akt and other lipid-dependent kinases; phosphorylation by these kinases excludes FOXO1 from the nucleus<sup>69,98,99</sup>. FOXO1 has been proposed to induce apoptosis, inhibit entry into the cell cycle, and stimulate glucose production<sup>67,68</sup>. Similar to observations in C. elegans, loss-of-function mutants of Foxol rescues diabetes in the IRS-2 knockout model;  $\beta$ -cell mass is restored to almost normal in  $Irs2^{-/-}Foxo1^{+/-}$  mice, demonstrating that IRS-2 signaling is required to regulate the negative impact of Foxo1 function on  $\beta$ -cell development, growth and/or survival<sup>100</sup>.

Dauer arrest coincides with major metabolic changes. DAF-2 signaling normally allows non-dauer reproductive growth which is associated with the utilization of food for growth in cell number and size and with small stores of fat<sup>87</sup>. In *daf-2* mutants which dauer arrest, metabolism is shifted to the production of fat and glycogen in intestinal cells<sup>89</sup>. Thus, signals from the DAF-2 pathway regulate metabolism, growth, reproductive development, and longevity in this invertebrate system<sup>101</sup>. The structural and functional homology of DAF-2 to the mammalian insulin receptor provides a basis for conserved mechanism that integrates metabolic signals with а neuroendocrine regulation of reproduction and perhaps even aging. The increase in longevity associated with decreased DAF-2 signaling is

analogous to enhanced mammalian longevity associated with caloric restriction<sup>97</sup>.

# 6. INSULIN RECEPTOR SIGNALING IN DROSOPHILA

As a model system, Drosophila enjoys many of the same advantages as C. elegans including ease of genetic manipulation and a known genome. The evolutionary conservation of the insulin signaling system and its role in the coordination of multiple systems is further illustrated by recent findings in Drosophila. The Drosophila insulin receptor homolog inr encodes a protein remarkably similar to its mammalian counterparts, including two  $\alpha$  and two  $\beta$  subunits, with a cytoplasmic tyrosine kinase that is activated by insulin binding.<sup>102</sup> The overall level of identity between INR and human IR and IGF1R is 32.5% and 33.3%, respectively<sup>103,104</sup>. However, one structural difference between the Drosophila and human insulin receptors is the presence of a C-terminal tail of ~400 amino acids in the dINR<sup>105</sup>. This extension contains several potential tyrosine phosphorylation sites and three NPXYXXM motifs, suggesting that it may participate in signal transduction. Indeed, the tail of INR binds IRS-1 and PI-3 kinase when the Drosophila receptor is expressed in mammalian cells<sup>106,107</sup>, but its precise function remains to be determined. Similar to results in worms and humans, partial loss of INR function reduces cell, organ and body size by decreasing both growth and proliferation<sup>108-111</sup>.

CHICO, the *Drosophila* homolog of vertebrate IRS1-4, serves as a substrate of the INR and exerts a critical role in the control of cell growth, reproduction, metabolism, and lifespan<sup>81,112,113</sup>. Animals mutant for *chico* are half the size of wild-type animals owing to a reduction in cell size and number<sup>81</sup> and null *chico* mutants live up to 48% longer than controls<sup>113</sup>. Interestingly, CHICO also regulates metabolism; even though reduced in overall size, chico mutants display a 2-fold increase in lipid levels. Moreover, female *chico* mutants are sterile, suggesting that even flies possess a mechanism for coordinating metabolism and reproduction<sup>81</sup>. Thus, in both *C. elegans* and *Drosophila*, homologs of the insulin/IGF-I/IRS signaling network are implicated not only in growth, but also in the maintenance of fuel homeostasis and in the regulation of reproduction and longevity.

#### Glossary

AAP-1: C.elegans homolog of p50/55 regulatory subunit of PI3-kinase	IR
AGE-1: C.elegans homolog of p110 catalytic subunit of PI3-kinase	IR
Akt: protein kinase B	IS
DAF-2: C.elegans homolog of insulin/IGF-I receptor	M
Dicer: RNAse-III equivalent	p6
FKHR: forkhead family of transcription factors	PI
Foxo: subfamily of forkhead transcription factors	RI
Grb-2: growth-factor receptor binding protein-2	St
GSK-3: glycogen-synthase kinase	S
IGF-I: insulin-like growth factor	S
INR: Drosoph#e insulin receptor	T

IR: mammalian insulin receptor IRS: insulin receptor substrate proteins IST-1: C.elegans homolog of IRS MAPK: mitogen-activated protein kinase 560: Drosophila homolog of p85/PI3-K PI3-Kinase: phosphatidyl-Inositol kinase RNAI: RNA interference to silence genes S6K: p70 ribosomal 6 kinase Shaggy: Drosophila homolog of GSK-3 SH2: src-homology domain 2 TOR: target of rapamycin

## 7. THE FUTURE OF EXPERIMENTAL MODELS

During the last two decades, many functions of the insulin/IGF-I system in development and postnatal growth have been revealed, chiefly by genetic modifications of various experimental model systems. However, as noted throughout this chapter, many questions remain regarding the role of this growth factor in the developmental events of mammals. The recent publication of the sequences of entire genomes<sup>114-116</sup> and the development of RNA interference will now greatly facilitate the production of new, more refined model systems for the study of IGF-I in development. One great challenge for the future is the development of technology that will enable us to apply what we learn about IGF-I in animal models to the diagnosis, treatment and prevention of human diseases such as cancer and diabetes.

#### REFERENCES

- 1. J.P. Liu, J. Baker, A.S. Perkins, E.J. Robertson and A. Efstratiadis, Mice carrying null mutations of the genes encoding insulin-like growth factor I (Igf-1) and type 1 IGF receptor (Igf1r). *Cell*,**75**, 59-72 (1993).
- 2. J. Baker, J.P. Liu, E.J. Robertson and A. Efstratiadis, Role of insulin-like growth factors in embryonic and postnatal growth. *Cell*, **75**, 73-82 (1993).
- 3. S.S. Cheah and R.R. Behringer, Contemporary gene targeting strategies for the novice. *Mol Biotechnol*, **19**, 297-304 (2001).
- H. Gu, J.D. Marth, P.C. Orban, H. Mossmann and K. Rajewsky, Deletion of a DNA polymerase beta gene segment in T cells using cell type-specific gene targeting. *Science*, 265, 103-106 (1994).
- 5. A. Nagy, Cre recombinase: the universal reagent for genome tailoring. *Genesis*, **26**, 99-109 (2000).

46

#### Experimental Models

- 6. A.K. Hadjantonakis, M. Pirity and A. Nagy, Cre recombinase mediated alterations of the mouse genome using embryonic stem cells. *Methods Mol Biol*, **97**, 101-122 (1999).
- 7. K. Rajewsky, H. Gu, R. Kuhn, U.A. Betz, W. Muller, J. Roes and F. Schwenk, Conditional gene targeting. *J Clin Invest*, **98**, 600-603 (1996).
- 8. S. Yakar, J.L. Liu, B. Stannard, A. Butler, D. Accili, B. Sauer and D. LeRoith, Normal growth and development in the absence of hepatic insulin-like growth factor I. *Proc Natl Acad Sci USA*, **96**, 7324-7329 (1999).
- 9. J.L. Liu, S. Yakar and D. LeRoith, Conditional knockout of mouse insulin-like growth factor-1 gene using the *Cre/loxP* system. *Proc Soc Exp Biol Med*, **223**, 344-351 (2000).
- N. Schultze, Y. Burki, Y. Lang, U. Certa and H. Bluethmann, Efficient control of gene expression by single step integration of the tetracycline system in transgenic mice. *Nat Biotechnol*, 14, 499-503 (1996).
- 11. G.J. Hannon, RNA interference, Nature, 418, 244-251 (2002).
- 12. D.M. Dykxhoorn, C.D. Novina and P.A. Sharp, Killing the messenger: short RNAs that silence gene expression. *Nat Rev Mol Cell Biol*, 4, 457-467 (2003).
- A. Fire, S. Xu, M.K. Montgomery, S.A. Kostas, S.E. Driver and C.C. Mello, Potent and specific genetic interference by double-stranded RNA in *Caenorhabditis elegans*. *Nature*, 391, 806-811 (1998).
- A.G. Fraser, R.S. Kamath, P. Zipperlen, M. Martinez-Campos, M. Sohrmann and J. Ahringer, Functional genomic analysis of *C. elegans* chromosome I by systematic RNA interference. *Nature* 408, 325-30 (2000).
- 15. P. Gonczy, C. Echeverri, K. Oegema, A. Coulson, S.J. Jones, R.R. Copley, J. Duperon, J. Oegema, M. Brehm, E. Cassin, E. Hannak, M. Kirkham, S. Pichler, K. Flohrs, A. Goessen, S. Leidel, A.M. Alleaume, C. Martin, N. Ozlu, P. Bork and A.A. Hyman, Functional genomic analysis of cell division in *C. elegans* using RNAi of genes on chromosome III. *Nature*, 408, 331-336 (2000).
- Y.O. Kim, S.J. Park, R.S. Balaban, M. Nirenberg and Y. Kim, A functional genomic screen for cardiogenic genes using RNA interference in developing *Drosophila* embryos. *Proc Natl Acad Sci U S A* 101, 159-64 (2004).
- 17. L.R. Saunders and G.N. Barber, The dsRNA binding protein family: critical roles, diverse cellular functions. *FASEB J*, **17**, 961-983 (2003).
- 18. Q.C. Wang, Q.H. Nie and Z.H. Feng, RNA interference: antiviral weapon and beyond. *World J Gastroenterol*, **9**, 1657-1661 (2003).
- 19. Y. Shi, Mammalian RNAi for the masses. Trends Genet, 19, 9-12 (2003).
- S.M. Elbashir, J. Harborth, W. Lendeckel, A. Yalcin, K. Weber and T. Tuschl, Duplexes of 21-nucleotide RNAs mediate RNA interference in cultured mammalian cells. *Nature*, 411, 494-498 (2001).
- J.M. Bosher and M. Labouesse, RNA interference: genetic wand and genetic watchdog. *Nat Cell Biol*, 2, E31-36 (2000).
- 22. N.R. Wall and Y. Shi, Small RNA: can RNA interference be exploited for therapy? *Lancet*, **362**, 1401-1403 (2003).
- 23. A. Efstratiadis, Genetics of mouse growth. Int J Dev Biol, 42, 955-976 (1998).
- 24. S.Y. Hsu, Cloning of two novel mammalian paralogs of relaxin/insulin family proteins and their expression in testis and kidney. *Mol Endocrinol*, **13**, 2163-2174 (1999).

- J. Nakae, Y. Kido and D. Accili, Distinct and overlapping functions of insulin and IGF-I receptors. *Endocr Rev*, 22, 818-835 (2001).
- 26. A.J. D'Ercole, P. Ye, A.S. Calikoglu and G. Gutierrez-Ospina, The role of the insulinlike growth factors in the central nervous system. *Mol Neurobiol*, **13**, 227-255 (1996).
- 27. J. Wang, J. Zhou, L. Powell-Braxton and C. Bondy, Effects of Igf1 gene deletion on postnatal growth patterns. *Endocrinology*, **140**, 3391-3394 (1999).
- K.D. Beck, L. Powell-Braxton, H.R. Widmer, J. Valverde and F. Hefti, Igf1 gene disruption results in reduced brain size, CNS hypomyelination, and loss of hippocampal granule and striatal parvalbumin-containing neurons. *Neuron*, 14, 717-730 (1995).
- C.M. Cheng, G. Joncas, R.R. Reinhardt, R. Farrer, R. Quarles, J. Janssen, M.P. McDonald, J.N. Crawley, L. Powell-Braxton and C.A. Bondy, Biochemical and morphometric analyses show that myelination in the insulin-like growth factor 1 null brain is proportionate to its neuronal composition. *J Neurosci*, **18**, 5673-5681 (1998).
- W.Q. N. Gao, N. Shinsky, G. Ingle, K. Beck, K.A. Elias and L. Powell-Braxton, IGF-I deficient mice show reduced peripheral nerve conduction velocities and decreased axonal diameters and respond to exogenous IGF-I treatment. *J Neurobiol*, **39**, 142-152 (1999).
- T.M. DeChiara, A. Efstratiadis and E.J. Robertson, A growth-deficiency phenotype in heterozygous mice carrying an insulin-like growth factor II gene disrupted by targeting. *Nature*, 345, 78-80 (1990).
- 32. T.M. DeChiara, E.J. Robertson and A. Efstratiadis, Parental imprinting of the mouse insulin-like growth factor II gene. *Cell*, **64**, 849-859 (1991).
- 33. K.A. Woods, C. Camacho-Hubner, M.O. Savage and A.J. Clark, Intrauterine growth retardation and postnatal growth failure associated with deletion of the insulin-like growth factor I gene. *N Engl J Med*, **335**, 1363-1367 (1996).
- K.A. Woods, C. Camacho-Hubner, R.N. Bergman, D. Barter, A.J. Clark and M.O. Savage, Effects of insulin-like growth factor I (IGF-I) therapy on body composition and insulin resistance in IGF-I gene deletion. *J Clin Endocrinol Metab*, 85, 1407-1411 (2000).
- D. LeRoith, C. Bondy, S. Yakar, J.L. Liu and A. Butler, The somatomedin hypothesis: 2001. Endocr Rev, 22, 53-74 (2001).
- 36. A.A. Butler and D. LeRoith, Minireview: tissue-specific versus generalized gene targeting of the *igf1* and *igf1r* genes and their roles in insulin-like growth factor physiology. *Endocrinology*, **142**, 1685-1688 (2001).
- A.A. Butler and D. LeRoith, Control of growth by the somatropic axis: growth hormone and the insulin-like growth factors have related and independent roles. *Annu Rev Physiol*, 63, 141-164 (2001).
- F. Lupu, J.D. Terwilliger, K. Lee, G.V. Segre and A. Efstratiadis, Roles of growth hormone and insulin-like growth factor 1 in mouse postnatal growth. *Dev Biol*, 229, 141-162 (2001).
- K. Sjogren, J.L. Liu, K. Blad, S. Skrtic, O. Vidal, V. Wallenius, D. LeRoith, J. Tornell, O.G. Isaksson, J.O. Jansson and C. Ohlsson, Liver-derived insulin-like growth factor I (IGF-I) is the principal source of IGF-I in blood but is not required for postnatal body growth in mice. *Proc Natl Acad Sci USA*, **96**, 7088-7092 (1999).

- J.L. Trejo, E. Carro and I. Torres-Aleman, Circulating insulin-like growth factor I mediates exercise-induced increases in the number of new neurons in the adult hippocampus. J Neurosci, 21, 1628-1634 (2001).
- 41. M.A. Aberg, N.D. Aberg, H. Hedbacker, J. Oscarsson and P.S. Eriksson, Peripheral infusion of IGF-I selectively induces neurogenesis in the adult rat hippocampus. *J Neurosci*, **20**, 2896-2903 (2000).
- 42. J.L. Trejo, E. Carro, A. Nunez and I. Torres-Aleman, Sedentary life impairs selfreparative processes in the brain: the role of serum insulin-like growth factor-I. *Rev Neurosci*, **13**, 365-374 (2002).
- 43. E. Carro, J.L. Trejo, A. Nunez and I. Torres-Aleman, Brain repair and neuroprotection by serum insulin-like growth factor I. *Mol Neurobiol*, **27**, 153-162 (2003).
- 44. E. Carro, A. Nunez, S. Busiguina and I. Torres-Aleman, Circulating insulin-like growth factor I mediates effects of exercise on the brain. *J Neurosci*, **20**, 2926-2933 (2000).
- E. Carro, J.L. Trejo, S. Busiguina and I. Torres-Aleman, Circulating insulin-like growth factor I mediates the protective effects of physical exercise against brain insults of different etiology and anatomy. *J Neurosci*, 21, 5678-5684 (2001).
- 46. E. Carro, J.L. Trejo, T. Gomez-Isla, D. LeRoith and I. Torres-Aleman, Serum insulinlike growth factor I regulates brain amyloid-beta levels. *Nat Med*, **8**, 1390-1397 (2002).
- G. Kempermann and G. Kronenberg, Depressed new neurons-adult hippocampal neurogenesis and a cellular plasticity hypothesis of major depression. *Biol Psychiatry*, 54, 499-503 (2003).
- 48. C.R. Kahn and J. Roth, Cell membrane receptors for polypeptide hormones. Application to the study of disease states in mice, and men. *Am J Clin Pathol*, **63**, 656-667 (1975).
- 49. C.R. Kahn and M.F. White, The insulin receptor and the molecular mechanism of insulin action. *J Clin Invest*, **82**, 1151-1156 (1988).
- 50. M.F. White and C.R. Kahn, The insulin signaling system. J Biol Chem, 269, 1-4 (1994).
- 51. C.R. Kahn and A.B. Goldfine, Molecular determinants of insulin action. J Diabetes Compl, 7, 92-105 (1993).
- R. Pohlmann, G. Nagel, A. Hille, M. Wendland, A. Waheed, T. Braulke and K. von Figura, Mannose 6-phosphate specific receptors: structure and function. *Biochem Soc Trans*, 17, 15-6 (1989).
- 53. T. Braulke, Type-2 IGF receptor: a multi-ligand binding protein. *Horm Metab Res*, **31**, 242-246 (1999).
- 54. A.J. D'Ercole, P. Ye and J.R. O'Kusky, Mutant mouse models of insulin-like growth factor actions in the central nervous system. *Neuropeptides*, **36**, 209-220 (2002).
- J.C. Bruning, D. Gautam, D.J. Burks, J. Gillette, M. Schubert, P.C. Orban, R. Klein, W. Krone, D. Muller-Wieland and C.R. Kahn, Role of brain insulin receptor in control of body weight and reproduction. *Science*, 289, 2122-2125 (2000).
- M. Holzenberger, J. Dupont, B. Ducos, P. Leneuve, A. Geloen, P.C. Even, P. Cervera and Y. Le Bouc, IGF-1 receptor regulates lifespan and resistance to oxidative stress in mice. *Nature*, 421, 182-187 (2003).
- D.J. Withers, D.J. Burks, H.H. Towery, S.L. Altamuro, C.L. Flint and M.F. White, Irs-2 coordinates Igf-1 receptor-mediated beta-cell development and peripheral insulin signalling. *Nat Genet*, 23, 32-40 (1999).

- M. Holzenberger, G. Hamard, R. Zaoui, P. Leneuve, B. Ducos, C. Beccavin, L. Perin and Y. Le Bouc, Experimental IGF-I receptor deficiency generates a sexually dimorphic pattern of organ-specific growth deficits in mice, affecting fat tissue in particular. *Endocrinology*, 142, 4469-4478 (2001).
- M. Zhang, S. Xuan, M.L. Bouxsein, D. von Stechow, N. Akeno, M.C. Faugere, H. Malluche, G. Zhao, C.J. Rosen, A. Efstratiadis and T.L. Clemens, Osteoblast-specific knockout of the insulin-like growth factor (IGF) receptor gene reveals an essential role of IGF signaling in bone matrix mineralization. *J Biol Chem*, 277, 44005-44012 (2002).
- R.N. Kulkarni, M. Holzenberger, D.Q. Shih, U. Ozcan, M. Stoffel, M. A. Magnuson and C.R. Kahn, beta-cell-specific deletion of the Igf1 receptor leads to hyperinsulinemia and glucose intolerance but does not alter beta-cell mass. *Nat Genet*, **31**, 111-115 (2002).
- 61. M.F. White and C.R. Kahn, Cascade of autophosphorylation in the beta-subunit of the insulin receptor. *J Cell Biochem*, **39**, 429-441 (1989).
- X.J. Sun, P. Rothenberg, C.R. Kahn, J.M. Backer, E. Araki, P.A. Wilden, D.A. Cahill, B.J. Goldstein and M.F. White, Structure of the insulin receptor substrate IRS-1 defines a unique signal transduction protein. *Nature*, 352, 73-77 (1991).
- E. Araki, B. L. Haag, 3rd and C.R. Kahn, Cloning of the mouse insulin receptor substrate-1 (IRS-1) gene and complete sequence of mouse IRS-1. *Biochim Biophys Acta*, 1221, 353-356 (1994).
- M.E. Patti, X.J. Sun, J.C. Bruening, E. Araki, M.A. Lipes, M.F. White and C. R. Kahn, 4PS/insulin receptor substrate (IRS)-2 is the alternative substrate of the insulin receptor in IRS-1-deficient mice. *J Biol Chem*, 270, 24670-24673 (1995).
- S.E. Shoelson, S. Chatterjee, M. Chaudhuri and M.F. White, YMXM motifs of IRS-1 define substrate specificity of the insulin receptor kinase. *Proc Natl Acad Sci USA*, 89, 2027-2031 (1992).
- D.P. Brazil, J. Park and B.A. Hemmings, PKB binding proteins. Getting in on the Akt. Cell, 111, 293-303 (2002).
- M. Schmidt, S. Fernandez de Mattos, A. van der Horst, R. Klompmaker, G.J. Kops, E. W. Lam, B.M. Burgering and R.H. Medema, Cell cycle inhibition by FoxO forkhead transcription factors involves downregulation of cyclin D. *Mol Cell Biol*, 22, 7842-7852 (2002).
- 68. G.J. Kops, R.H. Medema, J. Glassford, M.A. Essers, P.F. Dijkers, P.J. Coffer, E.W. Lam and B.M. Burgering, Control of cell cycle exit and entry by protein kinase B-regulated forkhead transcription factors. *Mol Cell Biol*, **22**, 2025-2036 (2002).
- G. Rena, Y.L. Woods, A.R. Prescott, M. Peggie, T.G. Unterman, M.R. Williams and P. Cohen, Two novel phosphorylation sites on FKHR that are critical for its nuclear exclusion. *EMBO J*, 21, 2263-2271 (2002).
- B.E. Lavan, W.S. Lane and G.E. Lienhard, The 60-kDa phosphotyrosine protein in insulin-treated adipocytes is a new member of the insulin receptor substrate family. J Biol Chem, 272, 11439-11443 (1997).
- B.E. Lavan, V.R. Fantin, E.T. Chang, W.S. Lane, S.R. Keller and G.E. Lienhard, A novel 160-kDa phosphotyrosine protein in insulin-treated embryonic kidney cells is a new member of the insulin receptor substrate family. *J Biol Chem*, 272, 21403-21407 (1997).

- V.R. Fantin, B.E. Lavan, Q. Wang, N.A. Jenkins, D.J. Gilbert, N.G. Copeland, S.R. Keller and G.E. Lienhard, Cloning, tissue expression, and chromosomal location of the mouse insulin receptor substrate 4 gene. *Endocrinology*, **140**, 1329-1337 (1999).
- D.J. Withers, J.S. Gutierrez, H. Towery, D.J. Burks, J.M. Ren, S. Previs, Y. Zhang, D. Bernal, S. Pons, G.I. Shulman, S. Bonner-Weir and M.F. White, Disruption of IRS-2 causes type 2 diabetes in mice. *Nature*, **391**, 900-904 (1998).
- 74. N. Kubota, K. Tobe, Y. Terauchi, K. Eto, T. Yamauchi, R. Suzuki, Y. Tsubamoto, K. Komeda, R. Nakano, H. Miki, S. Satoh, H. Sekihara, S. Sciacchitano, M. Lesniak, S. Aizawa, R. Nagai, S. Kimura, Y. Akanuma, S.I. Taylor and T. Kadowaki, Disruption of insulin receptor substrate 2 causes type 2 diabetes because of liver insulin resistance and lack of compensatory beta-cell hyperplasia. *Diabetes*, **49**, 1880-1889 (2000).
- 75. S.C. Liu, Q. Wang, G.E. Lienhard and S.R. Keller, Insulin receptor substrate 3 is not essential for growth or glucose homeostasis. *J Biol Chem*, **274**, 18093-18099 (1999).
- V.R. Fantin, Q. Wang, G.E. Lienhard and S.R. Keller, Mice lacking insulin receptor substrate 4 exhibit mild defects in growth, reproduction, and glucose homeostasis. *Am J Physiol Endocrinol Metab*, 278, E127-133 (2000).
- J.C. Bruning, J. Winnay, B. Cheatham and C.R. Kahn, Differential signaling by insulin receptor substrate 1 (IRS-1) and IRS-2 in IRS-1-deficient cells. *Mol Cell Biol*, 17, 1513-1521 (1997).
- D.J. Burks, J.F. de Mora, M. Schubert, D. J. Withers, M.G. Myers, H. H. Towery, S.L. Altamuro, C.L. Flint and M.F. White, IRS-2 pathways integrate female reproduction and energy homeostasis. *Nature*, 407, 377-382 (2000).
- 79. J.W. Unger and M. Betz, Insulin receptors and signal transduction proteins in the hypothalamo-hypophyseal system: a review on morphological findings and functional implications. *Histol Histopathol*, **13**, 1215-1224 (1998).
- 80. H.A. Tissenbaum and G. Ruvkun, An insulin-like signaling pathway affects both longevity and reproduction in *Caenorhabditis elegans. Genetics*, **148**, 703-717 (1998).
- R. Bohni, J. Riesgo-Escovar, S. Oldham, W. Brogiolo, H. Stocker, B.F. Andruss, K. Beckingham and E. Hafen, Autonomous control of cell and organ size by CHICO, a Drosophila homolog of vertebrate IRS1-4. *Cell*, 97, 865-875 (1999).
- M. Schubert, D.P. Brazil, D.J. Burks, J.A. Kushner, J. Ye, C.L. Flint, J. Farhang-Fallah, P. Dikkes, X.M. Warot, C. Rio, G. Corfas and M.F. White, Insulin receptor substrate-2 deficiency impairs brain growth and promotes tau phosphorylation. *J Neurosci*, 23, 7084-7092 (2003).
- 83. R.A. Ankeny, The natural history of *Caenorhabditis elegans* research. *Nat Rev Genet*, **2**, 474-479 (2001).
- 84. B. Gerisch, C. Weitzel, C. Kober-Eisermann, V. Rottiers and A. Antebi, A hormonal signaling pathway influencing *C. elegans* metabolism, reproductive development, and life span. *Dev Cell*, **1**, 841-851 (2001).
- A. Dillin, D.K. Crawford and C. Kenyon Timing requirements for insulin/IGF-1 signaling in C. elegans. *Science*, 298, 830-834 (2002).
- K.D. Kimura, H.A. Tissenbaum, Y. Liu and G. Ruvkun, daf-2, an insulin receptor-like gene that regulates longevity and diapause in Caenorhabditis elegans. *Science*, 277, 942-946 (1997).

- 87. J. Apfeld and C. Kenyon, Cell nonautonomy of *C. elegans* daf-2 function in the regulation of diapause and life span. *Cell*, **95**, 199-210 (1998).
- S.B. Pierce, M. Costa, R. Wisotzkey, S. Devadhar, S.A. Homburger, A. R. Buchman, K.C. Ferguson, J. Heller, D.M. Platt, A.A. Pasquinelli, L. X. Liu, S.K. Doberstein and G. Ruvkun, Regulation of DAF-2 receptor signaling by human insulin and ins-1, a member of the unusually large and diverse *C. elegans* insulin gene family. *Genes Dev*, 15, 672-686 (2001).
- S. Ogg, S. Paradis, S. Gottlieb, G.I. Patterson, L. Lee, H.A. Tissenbaum and G. Ruvkun, The Fork head transcription factor DAF-16 transduces insulin-like metabolic and longevity signals in *C. elegans. Nature*, 389, 994-999 (1997).
- J.B. Dorman, B. Albinder, T. Shroyer and C. Kenyon, The age-1 and daf-2 genes function in a common pathway to control the lifespan of *Caenorhabditis elegans*. *Genetics*, 141, 1399-13406 (1995).
- 91. T.E. Johnson, Increased life-span of age-1 mutants in *Caenorhabditis elegans* and lower Gompertz rate of aging. *Science*, **249**, 908-912 (1990).
- S. Paradis and G. Ruvkun, *Caenorhabditis elegans* Akt/PKB transduces insulin receptorlike signals from AGE-1 PI3 kinase to the DAF-16 transcription factor. *Genes Dev*, 12, 2488-2498 (1998).
- C.A. Wolkow, M.J. Munoz, D.L. Riddle and G. Ruvkun, Insulin receptor substrate and p55 orthologous adaptor proteins function in the *Caenorhabditis elegans* daf-2/insulinlike signaling pathway. *J Biol Chem*, 277, 49591-49597 (2002).
- 94. N. Libina, J.R. Berman and C. Kenyon, Tissue-specific activities of *C. elegans* DAF-16 in the regulation of lifespan. *Cell*, **115**, 489-502 (2003).
- K. Houthoofd, B.P. Braeckman, T.E. Johnson and J.R. Vanfleteren, Life extension via dietary restriction is independent of the Ins/IGF-1 signalling pathway in *Caenorhabditis elegans. Exp Gerontol*, 38, 947-954 (2003).
- C.T. Murphy, S.A. McCarroll, C.I. Bargmann, A. Fraser, R.S. Kamath, J. Ahringer, H. Li and C. Kenyon, Genes that act downstream of DAF-16 to influence the lifespan of *Caenorhabditis elegans. Nature*, 424, 277-283 (2003).
- K. Lin, H. Hsin, N. Libina and C. Kenyon, Regulation of the *Caenorhabditis elegans* longevity protein DAF-16 by insulin/IGF-1 and germline signaling. *Nat Genet*, 28, 139-145 (2001).
- 98. S. Guo, G. Rena, S. Cichy, X. He, P. Cohen and T. Unterman, Phosphorylation of serine 256 by protein kinase B disrupts transactivation by FKHR and mediates effects of insulin on insulin-like growth factor-binding protein-1 promoter activity through a conserved insulin response sequence. *J Biol Chem*, **274**, 17184-17192 (1999).
- G. Rena, S. Guo, S.C. Cichy, T.G. Unterman and P. Cohen, Phosphorylation of the transcription factor forkhead family member FKHR by protein kinase B. *J Biol Chem*, 274, 17179-17183 (1999).
- 100. T. Kitamura, J. Nakae, Y. Kitamura, Y. Kido, W.H. Biggs, 3rd, C.V. Wright, M.F. White, K. C. Arden and D. Accili, The forkhead transcription factor Foxo1 links insulin signaling to Pdx1 regulation of pancreatic beta cell growth. *J Clin Invest*, **110**, 1839-1847 (2002).
- 101. J.H. Thomas and T. Inoue, Methuselah meets diabetes. Bioessays, 20, 113-115 (1998).

#### **Experimental Models**

- 102. R. Fernandez-Almonacid and O.M. Rosen, Structure and ligand specificity of the *Drosophila melanogaster* insulin receptor. *Mol Cell Biol*, **7**, 2718-2727 (1987).
- 103. L. Petruzzelli, R. Herrera, R. Arenas-Garcia, R. Fernandez, M.J. Birnbaum and O.M. Rosen, Isolation of a *Drosophila* genomic sequence homologous to the kinase domain of the human insulin receptor and detection of the phosphorylated Drosophila receptor with an anti-peptide antibody. *Proc Natl Acad Sci USA*, 83, 4710-4714 (1986).
- 104. R. Fernandez, D. Tabarini, N. Azpiazu, M. Frasch and J. Schlessinger, The *Drosophila* insulin receptor homolog: a gene essential for embryonic development encodes two receptor isoforms with different signaling potential. *EMBO J*, **14**, 3373-3384 (1995).
- 105. Y. Ruan, C. Chen, Y. Cao and R.S. Garofalo, The *Drosophila* insulin receptor contains a novel carboxyl-terminal extension likely to play an important role in signal transduction. *J Biol Chem*, 270, 4236-4243 (1995).
- 106. T. Yamaguchi, R. Fernandez and R.A. Roth, Comparison of the signaling abilities of the Drosophila and human insulin receptors in mammalian cells. *Biochemistry*, 34, 4962-4968 (1995).
- 107. L. Yenush, R. Fernandez, M.G. Myers, Jr., T.C. Grammer, X.J. Sun, J. Blenis, J.H. Pierce, J. Schlessinger and M.F. White, The *Drosophila* insulin receptor activates multiple signaling pathways but requires insulin receptor substrate proteins for DNA synthesis. *Mol Cell Biol*, 16, 2509-2517 (1996).
- 108. J. Verdu, M.A. Buratovich, E.L. Wilder and M.J. Birnbaum, Cell-autonomous regulation of cell and organ growth in *Drosophila* by Akt/PKB. *Nat Cell Biol*, **1**, 500-506 (1999).
- 109. S. Oldham, R. Bohni, H. Stocker, W. Brogiolo and E. Hafen, Genetic control of size in Drosophila. Philos Trans R Soc Lond B Biol Sci, 355, 945-952 (2000).
- 110. S. Oldham, H. Stocker, M. Laffargue, F. Wittwer, M. Wymann and E. Hafen, The Drosophila insulin/IGF receptor controls growth and size by modulating PtdInsP(3) levels. Development, 129, 4103-4109 (2002).
- 111. E. Hafen, Interplay between growth factor and nutrient signaling: lessons from *Drosophila* TOR. *Curr Top Microbiol Immunol*, **279**, 153-167 (2004).
- 112. W. Brogiolo, H. Stocker, T. Ikeya, F. Rintelen, R. Fernandez and E. Hafen, An evolutionarily conserved function of the *Drosophila* insulin receptor and insulin-like peptides in growth control. *Curr Biol*, **11**, 213-221 (2001).
- 113. D. J. Clancy, D. Gems, L. G. Harshman, S. Oldham, H. Stocker, E. Hafen, S. J. Leevers and L. Partridge, Extension of life-span by loss of CHICO, a *Drosophila* insulin receptor substrate protein. *Science*, **292**, 104-106 (2001).
- 114. E.W. Myers, G.G. Sutton, A.L. Delcher, I.M. Dew, D.P. Fasulo, M.J. Flanigan, S.A. Kravitz, C.M. Mobarry, K.H. Reinert, K.A. Remington, E.L. Anson, R.A. Bolanos, H.H. Chou, C.M. Jordan, A.L. Halpern, S. Lonardi, E.M. Beasley, R.C. Brandon, L. Chen, P.J. Dunn, Z. Lai, Y. Liang, D.R. Nusskern, M. Zhan, Q. Zhang, X. Zheng, G.M. Rubin, M.D. Adams and J.C. Venter JC, A whole-genome assembly of *Drosophila. Science*, 287, 2196-204 (2000).
- 115. E.S. Lander *et al.* Initial sequencing and analysis of the human genome. *Nature*, **409**, 860-921 (2001).
- 116. J.C. Venter et al. The sequence of the human genome. Science, 291, 1304-51 (2001).

# Chapter 3

# INSULIN-LIKE GROWTH FACTOR BINDING PROTEINS IN DEVELOPMENT

Josef V. Silha<sup>1</sup> and Liam J. Murphy<sup>1,2</sup>

Departments of <sup>1</sup>Physiology & <sup>2</sup>Internal Medicine, University of Manitoba, Winnipeg R3E 3P4 Canada

Key words: Growth retardation; IGF; IGFBP; transgenic mice.

## 1. OVERVIEW OF THE IGFBP FAMILY

Insulin-like growth factor-I (IGF-I) and IGF-II are growth-promoting polypeptides essential for normal growth and development<sup>1</sup>. Unlike other growth factors, both IGFs and their cognate membrane receptors are expressed in variety of tissues and cell types<sup>1</sup>. In both humans and rodents, IGFs share approximately 50% homology with proinsulin<sup>2</sup>. Close homology to insulin gives IGFs a unique characteristic among other growth factors as they not only regulate cell proliferation, differentiation, and apoptosis but also exert insulin mimetic metabolic effects in tissues expressing high levels of type I IGF receptor, such as skeletal muscle and pancreatic  $\beta$ -cells<sup>3,4</sup>.

IGFs are present in plasma and other biological fluids predominantly in a complex with IGFBPs. IGFBPs comprise a family of six homologous proteins that bind IGFs with a high affinity in the range of 10<sup>9</sup> to 10<sup>11</sup> l/mol, equal or higher than the affinity of IGFs for IGF type I membrane receptors<sup>5,6</sup>. IGFBPs control bioavailability of IGFs by regulating their transport, deposition in extracellular matrix, binding to membrane receptors, and degradation<sup>5</sup>. Besides acting through IGF-dependent mechanisms,

various IGF-independent effects of IGFBPs are increasingly described in  $vitro^7$ .

The *IGFBP* genes, like those of the IGFs themselves, are believed to have emerged early in vertebrate evolution<sup>8,9</sup>. Detailed examination of *IGFBP* gene family in the human genome revealed close linkage to the homeobox (*HOX*) gene family, which encodes transcription factors important in early embryogenesis. *IGFBP* genes are co-localized to four chromosomal regions with the *HOX* gene clusters. These observations suggest that *IGFBP* and *HOX* gene clusters duplicated together from a common ancestral chromosomal region.

The *IGFBP* genes share a common structural organization being coded by four conserved exons with the exception of IGFBP-3 gene which has a fifth exon in the 3'-untranslated region<sup>5</sup>. Exon 1 of the *IGFBP* gene family is shared by several other genes encoding a variety of IGFBP-related proteins, leading to the proposal that the IGFBPs belong to a larger superfamily of cysteine-rich proteins<sup>9</sup>.

IGFBPs share a conserved structure consisting of three domains of approximately equal size: N-terminal, central, and C-terminal domains. Both N- and C-terminals contain 12 and 6 conserved cysteine residues, respectively with the exception of IGFBP-6 with 10 cysteine residues in Nterminal domain<sup>9</sup>. These cysteine residues form intermolecular cystine bridges by the pairing of adjacent cysteines within the domain<sup>9</sup>. Amino acid residues important in IGF-binding are found in the N-terminal and also in Cterminal domain<sup>10</sup>. The observation suggests the existence of an IGF-binding pocket formed by both domains. Other important subdomains have also been identified within the C-terminal region of various IGFBPs, such as Arg-Gly-Asp (RGD) integrin-binding motifs are located at residues 221-223 of IGFBP-1<sup>11</sup> and residues 265–267 of IGFBP-2<sup>12</sup>. Multifunctional 18-residue basic motifs with heparin-binding activity and nuclear localization signal (NLS), described at residues 215-232 of IGFBP-3 and residues 201-218 of IGFBP-5, are involved in interaction with the serum glycoprotein ALS (acidlabile subunit)<sup>10,13,14</sup> and other ligands such as plasminogen activator inhibitor-1 (PAI-1)<sup>15</sup> and transferrin<sup>16</sup>, vitronectin<sup>17</sup>, cell and extracellular matrix binding<sup>10,18</sup>, and nuclear transport<sup>19</sup>. The C-terminal domain is homologous to the thyroglobulin type I repeat unit.

The central domain of the IGFBPs shows no structural similarity among any members of the family. Secondary IGFBP-5 binding sites for  $ALS^{20}$  and heparin<sup>21</sup>, and a potential cell-association domain of IGFBP-3<sup>7,10,22</sup>, are also found in this region. Sites of *N*-linked glycosylation in IGFBP-3 and IGFBP-4 and *O*-glycosylation in IGFBP-5 and -6 are found in this region<sup>23-26</sup>. The extent of glycosylation of individual IGFBPs may differ between tissues, as has been shown for IGFBP-6<sup>26</sup>. Glycosylation does not appears to alter IGFBP affinity for IGFs<sup>23</sup> but it may have an effect on proteolysis, cell membrane association, and clearance from circulation and thereby secondarily influence IGFBP affinity for IGFs<sup>23,26-28</sup>.

Other sites of posttranslational modification are also found in the central domain. These include potential phosphoacceptor sites on all IGFBPs, some of which are phosphorylated in IGFBP-1, -3, and -5<sup>29</sup>, and proteolytic cleavage sites in some of the proteins<sup>26,30-32</sup>. Posttranslational modifications modulate affinity of IGFBPs to IGFs. For example, it has been shown that serine phosphorylation of IGFBP-1 enhances its affinity for IGF-1 by six- to eightfold and also increases its capacity to inhibit IGF-I actions<sup>33</sup>. Although IGFBP-3 could also be phosphorylated on serine residues, it is present in circulation predominantly in an unphosphorylated form<sup>29</sup>. Interestingly, it has been recently shown that binding of IGFBP-3 to breast cancer cells is accompanied by phosphorylation at the plasma membrane level and that both processes are inhibited by IGF-I. However, once phosphorylated, the ability of IGFBP-3 to bind IGF-I is enhanced, possibly resulting in increased association of the IGF-I with the cell membrane<sup>34</sup>.

Each of the six IGFBPs can undergo proteolysis, which decreases their affinity for IGFs. Therefore, proteolysis has a critical role in regulation of IGFs release from their binary and ternary complexes with IGFBPs. While some proteases are specific for a single IGFBP, others can use multiple IGFBP as substrates. IGFBP proteolysis has been recently reviewed extensively elsewhere<sup>35,36</sup>.

In the circulation, IGFBP-3 and to a lesser extent IGFBP-5, form a ternary complex of IGF, IGFBP, and ALS. The ternary complex is responsible for transport of the majority of the IGFs in the circulation<sup>37</sup>. Under the stimulatory effect of growth hormone (GH), ALS and IGF-I are secreted from hepatocytes and IGF-I induces IGFBP-3 expression in hepatic endothelial cells<sup>37-40</sup>. This leads to formation of the ternary complex of IGF, IGFBP-3 or IGFBP-5, and ALS in 1:1:1 molar ratio<sup>41</sup>. This mechanism of the ternary complex formation has been confirmed in mouse models of selective IGF-I nullification in hepatocytes. In these mice with minimal production of IGF-I in the liver, serum levels of IGFBP-3 are decreased to ~30% of normal controls in the presence of low serum IGF-I and elevated GH<sup>42</sup>. See Table 1 for some of the structural features of the IGFBPs and ALS.

	MM (kDa)	IGF affinity	Posttranslationa l modification	Functional subdomains	Other interacting proteins
IGFBP-1	25.3	I = II	phosphorylation	RGD	A5β1 integrin receptor <sup>11</sup>
IGFBP-2	31.4	I < II		RGD	Heparin, extracellular matrix and proteoglycans 74,75
IGFBP-3	28.7 (40-42)	I = 11	Phosphorylation N-glycosylation	NLS	ALS <sup>108</sup> , transferrin <sup>16</sup> , fibrinonectin <sup>187</sup> , importin $\beta^{19}$ , RXR $\alpha^{79}$ , PGI/AMF <sup>188</sup> , humanin <sup>189</sup>
IGFBP-4	25.9 (29)	I = II	N-glycosylation		
IGFBP-5	28.6 (34)	I < II	phosphorylation <i>O</i> -glycosylation	NLS	ALS <sup>13</sup> , PAI-1 <sup>15</sup> , vitronectin <sup>17</sup> , importin $\beta^{19}$ , FHL2 <sup>93</sup> , $\alpha$ s <sub>2</sub> -casein <sup>94</sup> , collagen III and IV, laminin, fibronectin <sup>186</sup>
IGFBP-6	22.8 (32)	I << II	O-glycosylation		
ALS	63.3 (85)		glycosylation	, <u>, , ,</u>	IGFBP-3 <sup>108</sup> , IGFBP-5 <sup>13</sup>

Table 1. Structural aspects of human IGFBPs

MM = molecular mass, \*Numbers in parentheses indicate MM of glycosylated protein

## 2. IGFBPS: EXPRESSION AND REGULATION

In this section, individual IGFBPs are reviewed briefly. For a more detailed description of the individual IGFBPs, comprehensive reviews are available<sup>5,7,9,26,43</sup>. A summary of tissue expression in embryonic and postanatal life in humans and mice is given in Table 2.

## **2.1 IGFBP-1**

IGFBP-1 was first isolated from midterm human amniotic fluid<sup>44</sup>. Brewer et al. isolated its cDNA from a human decidua library and showed that it can potentiate the effects of IGF-I *in vitro*<sup>45</sup>.

	Human embryo	Mouse embryo
IGFBP-1	Liver (a), kidney (b)	After e12, restricted to the liver (c,d)
IGFBP-2	Moderate abundance in every tissue	Widely distributed mainly in the ectodermal layer
	with the highest level observed in the	and also in the mesoderm derived part of the
	liver (a); UD, committed glomerular and	tongue (c); lung, liver, kidney, choroid plexus,
	epithelial cells, mature glomerular	floor plate (d)
	epithelial cells, decreases in proximal	
	tubule with maturation (e)	
IGFBP-3	Most abundant in the skin, muscle, and	Only weakly in late gestational embryos (c,d)
	heart (a); mature UD, epithelium of the	liver, lung, tongue, and nasal epithelium (d),
	collecting ducts and pelvicalyceal	choroids plexus, kidney (da), heart (db)
	system (e)	
IGFBP-4	Moderate abundance equally in all	Widely distributed in the mesoderm derived
	tissues (a); kidney, stomach, intestine,	tissues (c); mesencephalon, telencephalon, heart,
	and lung and least in the liver (14-18 w)	liver, lung, tongue, blood vessels (d), kidney (da)
	(f); UD glomerular epithelial cells,	
	increases in proximal tubule with	
	maturation (e)	
IGFBP-5	Muscle, skin, stomach, and intestine	Unlike other BPs only after implantation in
	(14-18w) (f), most abundantly in the	mouse embryo, widely distributed, predominantly
	skin, muscle and stomach (a),	in the meninges, vertebrae, lung, kidney (c); liver,
	mesenchyme of the vascular cleft,	gut endothelium, meninges, lung mesenchyme,
	mature mesangium and supporting	tongue (d), notochord, floor plate, muscle (da)
	mesenchyme (e)	
IGFBP-6	Low abundance in all tissues (a), heart,	Weakly expressed in late gestational embryos
	muscle, gut, lung, brain and skin and	(c,d), liver, nasal epithelium (d), hindbrain, heart
	least in the liver (f)	(da), perichondral layers of cartilage (db)

*Table 2.* Tissue expression of IGFBPs during embryonic development in human and mouse tissues. E = mouse embryonic day, W = human fetal week, UD = ureteric duct. (a) W10-16 mRNA<sup>190</sup>, (b) W15-23 mRNA<sup>191</sup>, (c) E13.5 mRNA<sup>192</sup>, (da) E13.5 protein and (db) E13.5 mRNA<sup>193</sup>,  $\notin$  mRNA<sup>194</sup>, (f) W14-18 mRNA<sup>195</sup>

IGFBP-1 is primarily expressed in liver, decidualized uterine endometrium, ovarian granulosa cells, and kidney<sup>43</sup>. Insulin inhibits IGFBP-1 expression in hepatocytes at the level of gene transcription<sup>46</sup>. IGFBP-1 expression is inhibited by GH<sup>47</sup> and also by IGF-I and -II in cells with type I IGF receptor<sup>48, 49</sup>. In these cells IGF-I may be a more potent inhibitor of IGFBP-1 expression than insulin. IGFBP-1 expression is stimulated by glucagon, glucocorticoids, T4, somatostatin, epidermal growth factor, interleukin (IL) IL-6, IL-1, and tumor necrosis factor (TNF)- $\alpha^{43,50}$ . Comparison of human, rat and mouse promoters indicates high sequence similarity in the proximal ~300 bp, where IGFBP-1 promoter activity has been shown to be regulated by HNF1, HNF3, glucocorticoids, insulin, and cAMP *in vitro*. IGFBP-1 promoter is highly homologous to a promoter of PEPCK, a key gluconeogenetic enzyme, suggesting a role for IGFBP-1 in regulation of glucose metabolism<sup>43</sup>. Levels in the circulation are regulated predominantly by insulin and IGFBP-1 levels in the circulation reflect hepatic portal insulin concentration and hepatic insulin sensitivity rather than whole body insulin sensitivity<sup>51</sup>. Human IGFBP-1 is secreted as a phosphoprotein with phosphorylated serine residues 101, 119, or 169. Phosphorylation of IGFBP-1 is associated with increased affinity for IGF peptides. It has been shown that site-directed mutagenesis of Ser-101 prevents phosphorylation of this residue and results in a three-fold decrease in affinity for IGF-I<sup>29,33</sup>. IGFBP-1 phosphorylation is developmentally and hormonally regulated<sup>52,53</sup>.

The majority of *in vitro* and *in vivo* studies support an inhibitory role of IGFBP-1 on IGF-stimulated growth, differentiation, and metabolic activity. There is evidence for an inverse relationship between IGFBP-1 and bioactive free IGF-I<sup>54</sup>. During fasting low insulin levels lead to activation of IGFBP-1 transcription in hepatocytes, and a rise in IGFBP-1 in circulation is accompanied by a decline in free IGF-I<sup>55</sup>. IGFBP-1 is reduced in obese humans, which reflects elevated insulinemia in response to insulin resistance<sup>56</sup>. Further progression of insulin resistance with onset of type 2 diabetes mellitus (T2DM) is associated with an increase of IGFBP-1 back to values observed in lean individuals<sup>56</sup>. This is in a good agreement with the increased free IGF-I and IGF-II in obese subjects. Even more profound changes in IGFBP-1 are present in type 1 diabetes mellitus (T1DM), where insufficient insulin inhibitory effect on IGFBP-1 transcription in hepatocytes leads to increased IGFBP-1 production in the presence of low IGF-I, free IGF-I, and free IGF-II levels in circulation<sup>56</sup>.

*In vivo* studies by Lewitt *et al.* demonstrated that human IGFBP-1 blocks the hypoglycemic response to intravenous IGF-I and may even cause a mild increase in glycemia when administered alone<sup>57</sup>. In a more recent study, Lang *et al.* demonstrated that acute *in vivo* elevation in IGFBP-1, of the magnitude observed in various catabolic conditions, is capable of selectively decreasing protein synthesis in fast-twitch skeletal muscle while it had no effect on glycemia in rats<sup>58</sup>. Since IGFBP-1 is upregulated by pro-inflammatory and stress-related cytokines, elevations in circulating and tissue levels of IGFBP-1 may be an important mediator of the muscle catabolism observed in these conditions.

## 2.2 IGFBP-2

Binkert first cloned the cDNA and determined primary structure of human IGFBP-2<sup>12</sup>. IGFBP-2 is highly expressed in fetal tissues, particularly in the CNS<sup>59</sup>. High levels of IGFBP-2 decrease after birth until puberty after which it gradually increases again, particularly after age of 60<sup>60</sup>. IGFBP-2 remains the most abundant protein in the CNS postnatally<sup>61</sup> and it is the predominant IGFBP in the lymph, bronchoalveolar lavage, prostate and seminal fluid of normal human subjects<sup>62,63</sup>. It is also expressed at high levels in liver,

kidney, and secretory endometrium. GH and insulin inhibit hepatic synthesis and thereby control levels in circulation<sup>64-66</sup>. Increased IGFBP-2 has been reported during prolonged fasting, and protein restriction<sup>67,68</sup>, in GH deficiency<sup>66</sup> and after IGF-I administration possibly via suppression of GH<sup>65,69</sup>. IGFBP-2 is elevated in a number of pathological conditions, such as chronic renal failure, liver cirrhosis, non-islet-cell tumor hypoglycemia, whereas decreased IGFBP-2 has been observed in response to GH therapy in GH deficient subjects<sup>65,66,70</sup>. IGFBP-2 is lower in obese than lean subjects and unlike IGFBP-1, it further declines in T2DM subjects<sup>56</sup>. Marked reduction in IGFBP-2 has been observed in stromal cells of patients with benign prostatic hyperplasia, suggesting inhibitory effect of IGFBP-2 on stromal growth<sup>71</sup>. However, in prostate cancer IGFBP-2 may also have stimulatory effects on cell growth<sup>72</sup> and IGFBP-2 gene has been recently found to be highly overexpressed in the human glioblastoma, but not lowgrade gliomas<sup>73</sup>. Similar observations have also been made for other IGFBPs and suggest an antagonistic role in modulation of IGF action, depending on the cell status.

IGFBP-2 can bind to heparin, extracellular matrix and proteoglycans in cooperation with IGFs<sup>74,75</sup>. This binding is enhanced by the IGFs, but does not change affinity for IGFs and it may inhibit IGF-II action in some cell types<sup>76</sup>.

## 2.3 IGFBP-3

Based on sequence data derived from GH-dependent insulin-like growth factor (IGF) binding protein, purified from human plasma, Wood *et al.* isolated full-length IGFBP-3 cDNA clones<sup>78</sup>. As the most abundant IGFBP in circulation and a key component of the ternary complex, IGFBP-3 has a central role in regulation of IGF transport to tissues<sup>37</sup>. IGFBP-3 can be also transported into the nucleus by an importinβ-mediated mechanism, which requires intact nuclear localization signal in heparin binding subdomain of IGFBP-3 C-terminal<sup>19</sup>. In the nucleus, it has been shown to interact with the RXR $\alpha$  and possibly other nuclear elements<sup>79</sup>. The ability of IGFBP-3 to bind to a variety of other serum proteins, including ALS, transferrin, plasminogen, fibrinogen, vitronectin, and fibrin, has been reported<sup>10,13-19</sup>. Cell surface binding sites for IGFBP-3 have also been identified in some cell types<sup>10,18,34</sup>.

The principal regulators of serum IGFBP-3 are its hepatic synthesis, which is regulated by IGF-I produced locally by hepatocytes and to a lesser extent direct regulation by GH and insulin<sup>39-41</sup>. Hepatic IGFBP-3 production is responsive to alterations in dietary intake and it parallels changes in hepatic production of IGF-I with IGFBP-3 being reduced during protein restriction or malnutrition<sup>79,80</sup>. Serum IGFBP-3 is relatively constant throughout the day<sup>80</sup>. Decreased IGFBP-3 is accompanied by increased IGF

clearance from circulation in conditions characterized by low IGFBP-3, such as hypopituitarism<sup>81</sup>. In acromegaly and GH deficient subjects, IGFBP-3 parallels changes in IGF-I in circulation. Measurement of IGFBP-3 and IGF-I in addition to GH dynamic testing is clinically useful in evaluation of GH functional status<sup>82</sup>. IGFBP-3 is also mildly elevated in obese and type II diabetic subjects<sup>56</sup>. While liver production is the most important IGFBP-3 source in circulation, IGFBP-3 is expressed in variety of other tissues such as skin, skeletal muscle, reproductive organs, bone, and endothelium<sup>5</sup>. Endothelial cells are an important source of IGFBP-3 as documented by in situ hybridization studies in the liver and ovary<sup>40,41,83</sup>. These cells also have binding sites for IGFBP-3<sup>18</sup>. Since IGF-I has been shown to regulate endothelial cell proliferation via direct effect as well as through modulation of the vascular endothelial growth factor (VEGF) system, IGFBP-3 may play a significant role in control of angiogenesis. Fraser et al. found that IGFBP-3 mRNA was expressed in the endothelium of the human corpus luteum<sup>83</sup>. The mRNA expression is high during the early luteal phase and its subsequent decline is prevented by administration of human chorionic gonadotropin (CG) and IGFBP-3 may play a role in controlling angiogenesis involution of the human corpus luteum by autocrine and paracrine mechanisms<sup>83</sup>.

### 2.4 IGFBP-4

Shimasaki et al. cloned human IGFBP4 from human placenta, liver, and ovary cDNA libraries and they demonstrated tissue expression in all tissues examined, with highest expression in liver<sup>84</sup>. The central domain of IGFBP-4 has two extra Cys residues that form a cystine bridge and this may contribute to the distinctive biological behavior of IGFBP-4, which has only inhibitory actions on IGF action in vitro. IGFBP-4 does not associate with cell surface<sup>85</sup>. IGFBP-4 can be glycosylated at Asn104, but this does not affect its binding to IGF-I<sup>85</sup>. IGFBP-4 is expressed in a variety of cell types, including hepatocytes, vascular endothelium and smooth muscle, granulosa cells, osteoblasts, fibroblasts, and articular chondrocytes<sup>86</sup>. The regulation of IGFBP-4 expression is cell type specific with IGF-I having predominantly an inhibitory effect on IGFBP-4 expression. IGFBP-4 is detected in high levels in small androgen-dominant, growth-arrested, and atretic follicles, but in low levels in estrogen-dominant growing follicles<sup>87</sup>. Loss of IGFBP-4 inhibitory action in the dominant follicle, possibly the result of increased IGFBP-4 proteolysis, increases bioavailable IGFs in follicular fluid and may be involved in selection of the dominant follicle<sup>87</sup>. IGFs are the most abundant growth factors in bone and IGFBP-4 is the major IGFBP produced microenvironment<sup>88</sup>. Parathyroid hormone in bone and 1.25dihydroxyvitamin-D3 stimulate local IGFBP-4 production in bone and this may inhibit bone formation<sup>88</sup>. Local IGFBP-4 administration inhibits bone formation whereas systemic IGFBP-4 may increase bone formation in mice, probably by increased IGF delivery into bone<sup>89</sup>. IGFBP-4 is regulated developmentally and its levels in circulation increase with age<sup>88</sup>.

# 2.5 IGFBP-5

IGFBP-5 was purified from human serum and subsequently cloned by Kiefer et  $al^{90}$ . IGFBP-5 is homologous to IGFBP-3 and ~58% is present in the circulation in the ternary complex with ALS<sup>91</sup>. It shares several other binding partners with IGFBP-3 and is also transported to the nucleus<sup>19</sup>. An IGFBP-5 membrane receptor has been described<sup>92</sup>, and using a yeast 2hybrid screen, Amaar et al. found strong and specific interaction between IGFBP-5 and four and a half LIM 2 (FHL-2), a nuclear coactivator of the androgen receptor<sup>93</sup>. Purified FHL-2 did not interact with any other IGFBPs tested<sup>93</sup>. These data support the claim for IGF-independent effects of IGFBP-5. IGFBP-5 also binds to  $\alpha$ s<sub>2</sub>-casein, types III and IV collagen, laminin, and fibronectin and IGFBP-5 bound to ECM may potentiate the IGF effect<sup>94,95</sup>. Augmentation of IGF action typically involves proteolysis or reduction in the affinity of IGFBP-5 for IGF-1 after interaction with components of the extracellular matrix<sup>95</sup>. IGFBP-5 expression is not acutely altered by nutrients or insulin, and the most important determinants of levels in the circulation are protection by IGF-I from proteolysis and stimulation of transcription by IGF-I and GH, whereas IGFBP-5 expression is inhibited by dexamethasone<sup>88,96,97</sup>. Levels in the circulation increase during puberty and decline with aging<sup>88,98</sup>.

IGFBP-5 plays a role in bone, ovary, kidney, and skeletal muscle homeostasis. Increased expression has been observed during involution in mammary gland<sup>99</sup> and prostate<sup>100</sup>, and in ovarian follicles undergoing atresia, implying proapoptotic, antiproliferative effects<sup>87</sup>. IGFBP-5 concentrations in milk may reach 50 mg/l during rat mammary gland involution, the highest level of IGFBP reported in any biological fluid<sup>99</sup>. Prolactin may act by suppressing the production of IGFBP-5 from the mammary epithelium, thereby preventing the involution<sup>99</sup>.

# 2.6 IGFBP-6

Martin *et al.* first purified IGFBP-6 from transformed human fibroblasts and Shimasaki *et al.* determined the human cDNA sequence<sup>101,102</sup>. IGFBP-6 binds IGF-II with marked preferential affinity over IGF-I and it has the strongest affinity for IGF-II among the IGFBPs<sup>101</sup>. Human and rat IGFBP-6 lack two N-terminal cysteines and therefore the Gly-Cys-Gly-Cys-Cys motif, present in IGFBPs 1 through  $5^{102}$ . This is accompanied by preferential binding of IGF-II<sup>101</sup>. IGFBP-6 is *O*-glycosylated, and five serine/threonine glycosylation sites in the non-conserved mid-region of human IGFBP-6 have been identified<sup>103</sup>. Glycosylation modulates the half-life in the circulation, cell association, and proteolysis of IGFBP-6<sup>26-28</sup>. IGFBP-6 is a relatively specific inhibitor of IGF-II actions and it has not been shown to potentiate IGF actions. It is often expressed in non-proliferative, quiescent state cells *in vitro* and differentiating agents increase its expression, invoking a role for IGFBP-6 as an autocrine inhibitor of cell proliferation. In this respect, it is interesting that IGFBP-6 expression is uniformly induced by retinoic acid, a potent differentiating agent<sup>26,104</sup>.

IGFBP-6 and IGFBP-2 are the major IGFBPs in the cerebrospinal fluid<sup>105</sup>. IGFBP-6 is synthesized by the choroid plexus and astroglial cells<sup>106</sup>. It is also expressed in other human tissues such as placenta, lung, vascular smooth muscles, bone, skeletal muscle, prostate, and ovary<sup>26</sup>. Serum IGFBP-6 increases with age and it is higher in adult men than in women<sup>107</sup>. Elevated levels were observed in non-islet cell tumor induced hypoglycemia and in chronic renal failure<sup>107</sup>.

## 2.7 Acid-labile subunit

Baxter has shown that ALS forms a ternary complex in the presence of IGF-I and IGFBP-3<sup>108</sup>. Leong et al. isolated full-length clones encoding the human ALS from human liver cDNA libraries<sup>109</sup>. ALS expression is developmentally and hormonally regulated and the majority of ALS is produced by hepatocytes under stimulatory control of GH, suggesting an important role in regulating the availability and action of the IGFs<sup>40</sup>. Association of IGFs into the ternary complex stabilizes levels of IGFs in the circulation and prevents rapid IGF efflux from the circulation into tissues<sup>40,81,110</sup>. In the presence of low GH levels before birth or during the early postnatal period in rodents, the ternary complex is not detectable in the circulation and most of the IGFs present are bound to other IGFBPs as binary complexes<sup>111</sup>. Under the influence of GH, hepatic expression of ALS and IGFBP-3 is enhanced postnatally, and the abundance of the ternary complex increases dramatically in the first few weeks of rodent life<sup>111,112</sup>. Unlike the binary complexes that can traverse the endothelial capillary barrier, the ternary complex is thought to be confined to the circulation 113.

The enhanced expression of ALS, with an eventual two- to three-fold molar excess over IGFBP-3<sup>110</sup>, results in the gradual appearance of the ternary complex in the circulation in the first few weeks of rodent postnatal life. This change in the compartmentalization of IGFs in the circulation ensures formation of stable IGF reservoirs in the circulation, which appears to be important for normal postnatal growth<sup>114</sup>. In the rodent, IGF-II declines in importance during this period because of developmental down-regulation of expression<sup>115</sup>, and IGF-I becomes the most abundant IGF in adult life. A

similar developmental pattern of ALS and ternary complex formation has been observed in humans. There are markedly lower levels of the ternary complex in second- and third-trimester fetal serum compared to adult serum<sup>116</sup>. In addition to low ALS levels, IGFBP-3 in present predominantly as a 29-kDa proteolytic fragment rather than an intact molecule during human fetal development, which further compromises ternary complex formation<sup>116</sup>. In postnatal life, serum ALS follows GH level increases during childhood, reaches a peak during puberty and declines in adult life<sup>117</sup>. Glycosylation and sialylation patterns of ALS carbohydrate moieties influence ALS binding to IGFBP-3 and association to ternary complex<sup>118</sup>.

## 3. ALTERATIONS IN IGFBPS IN DEVELOPMENT

## **3.1 IGFBP-1**

A physiologic role for IGFBP-1 during fetal development is suggested by the demonstrated inverse relationship between fetal serum or cord blood IGFBP-1 and fetal size in mid and late gestation<sup>119-121</sup>. IGFBP-1 is elevated in the fetal circulation of human and animal pregnancies complicated by intrauterine growth retardation (IUGR) caused by placental insufficiency and in utero hypoxia <sup>122,123</sup>. It is believed to restrict fetal growth by sequestering free IGFs.

Popovici *et al.* established highly pure primary cultures of human fetal hepatocytes *in vitro* and investigated the expression of IGFBP-1 and the effects of hypoxia on expression of IGFBP-1 mRNA and protein<sup>124</sup>. The study has shown that hypoxia up-regulates fetal hepatocyte IGFBP-1 protein and mRNA levels and IGFBP-1 is the major IGFBP derived from these fetal hepatocytes. These data also support a role for the fetal liver as a source of elevated circulating levels of IGFBP-1 in fetuses with *in utero* hypoxia and IUGR.

Barreca *et al.* reported a child with approximately 20-fold elevation of IGFBP-1 above levels normal for age, which manifested with prenatal growth retardation that further deteriorated postanatally<sup>125</sup>. The child had no detectable endocrine or metabolic disorders and normal IGF-I and ALS, lower IGFBP-3 and IGF-II and mildly elevated GH serum concentration, compared to age appropriate controls. Almost all IGF-I activity, basally as well as after the GH stimulation test, was in the 35- to 45-kDa complex, corresponding to the IGFBP-1 peak. GH therapy resulted in increased growth velocity in this child<sup>125</sup>. This report supports a growth inhibitory
effect of IGFBP-1 *in vivo* and is in agreement with phenotype of IGFBP-1 transgenic (Tg) mice discussed later.

The insulin receptor is a ligand-activated tyrosine kinase. Mutations in the corresponding gene cause rare inherited insulin-resistant disorders in humans, leprechaunism, and Rabson-Mendenhall syndrome (RMS). Patients with the most severe syndrome, leprechaunism, manifest intrauterine growth retardation, hypoglycemia, and early death (usually before 1 year of age). RMS is caused by mutations in the insulin receptor gene with less severe disruption of insulin signaling. RMS manifests clinically with severe insulin resistance with hyperglycemia, short stature, and survival up to 5-15 years of age. Neonates with RMS are small for gestational age and infants are characterized by absence of subcutaneous fat, premature dentition, acanthosis nigricans, fasting hypoglycemia in infancy, and postprandial hyperglycemia in the presence of extremely high insulinemia<sup>126,127</sup>. Levels of IGFBP-1 in early infancy are higher than in adult human subjects and the high IGFBP-1 plasma levels in newborns could be important in protecting them from hypoglycemia<sup>128</sup>. Interestingly, low IGF-I, low IGFBP-3, high GH, and high IGFBP-1 have been reported in children with leprechaunism and RMS<sup>126,129,130</sup>. High IGFBP-1 levels may further attenuate biologic activity of low IGF-I in this condition. The actual levels of free IGFs have not been reported in RMS or leprechaunism to our knowledge; however, high IGFBP-1 is likely to contribute to the observed growth and metabolic abnormalities.

Therapy with rhGH (up to 0.5 mg/kg/week) did not improve growth and failed to increase the levels of circulating IGF-I and IGF-binding protein-3 over a 14-month period in a child with RMS. Subsequently, rhIGF-I was given by daily subcutaneous injection and no increase in growth velocity was observed over a 14-month period<sup>131</sup>. The lack of efficacy of IGF-I treatment may be related to several factors, such as the poor metabolic state of the patient, the observed deficiency of IGFBP-3, an increased clearance of the IGFs, IGF-I resistance in target cells at a receptor or postreceptor level, or an inhibitory action of the mutant insulin receptors on IGF-I receptor signaling through IGF/insulin hybrid receptors<sup>131,132</sup>. Indeed, therapy of an infant with leprechaunism with combined subcutaneous and intravenous administration of rhIGF-I, which resulted in sustained high IGF-I in circulation, caused a substantial improvement of growth velocity and improved metabolic control<sup>129</sup>.

In accordance with observations in leprechaunism and RMS are recent seminal observations of Accili and coworkers in mice with variable cellular mosaicism for null insulin receptor (*Insr*) allele<sup>133</sup>. A complete targeted ablation of *Insr* causes early postnatal lethality and precludes investigation of growth and metabolic phenotype of *Insr* deficiency in adult life<sup>134</sup>. *Insr*  ablation in approximately 80% of cells caused extreme growth retardation, lipoatrophy, and hypoglycemia, a phenotypic constellation that resembles the human leprechaunism. *Insr* ablation in 98% of cells also resulted in growth retardation and lipoatrophy. In addition, it caused diabetes without  $\beta$ -cell hyperplasia. In accordance with the data from children with RMS and leprechaunism, the growth retardation in the presence of disrupted insulin signaling in these mosaic mice was associated with a greater than 60-fold increase in the expression of hepatic IGFBP-1 and elevated IGFBP-1 in the circulation<sup>134</sup>.

## **3.2 IGFBP-3**

IGFBP-3 is an integral component of IGF ternary complex<sup>37</sup>. As the ternary complex formation is strongly GH dependent, IGFBP-3 levels in circulation are decreased in GH-deficient states and increased in the presence of GH excess<sup>82</sup>. GH deficiency or resistance to GH caused by an aberrant GH receptor signaling both lead to low abundance of ternary complex components and increased IGF disappearance from circulation<sup>82</sup>. Low levels of the ternary complex in circulation are associated with impaired postnatal growth in both humans and rodents<sup>114, 135</sup>.

#### 3.3 ALS

Recently, Domene *et al.* reported a patient with inherited absence of  $ALS^{135}$ . Sequencing of the *ALS* gene in this patient revealed a 1-bp deletion, resulting in the substitution E35K and the appearance of a premature stop codon. They described a 17-year-old boy with delayed onset of puberty and slow pubertal progress, but with normal psychomotor and neurologic development. At 14 years of age, the boy was referred for evaluation of growth pubertal delay and at this time his height was -2.05 SD. Growth hormone responses to provocative tests were normal, but there was marked reduction of both IGF-1 and IGFBP-3, which remained unchanged after stimulation with growth hormone. The ALS was undetectable in the serum before and after GH stimulation. After 6 months of treatment with recombinant human growth hormone, reduction in the subscapular skinfold thickness was observed but there was no beneficial effect on either the velocity of growth or the serum levels of IGF-I, IGFBP-3, and the acid-labile subunit.

## 4. DISORDERS WITH COMPLEX DISTURBANCES IN IGFBP SYSTEM

## 4.1 Chronic renal failure

Chronic renal failure (CRF) is a state of GH resistance and children with CRF often fail to achieve adult height consistent with their genetic potential and develop progressive growth retardation. CRF is characterized by simultaneous increase in IGFBP-1, -2, -4, -6, and proteolytic fragments of IGFBP-3 and -5 in serum<sup>70,136,137,138</sup>. Increased levels of IGFBPs, in the presence of relatively normal total IGFs, lead to decreased free IGF-I may contribute to decreased skeletal growth, peripheral substrate utilization, and tissue metabolism in CRF<sup>136</sup>. Improvement of linear growth with growth hormone (GH) treatment of uremic children is thought to be due to increased IGF-I/IGFBP ratio, resulting in increased free IGF-I levels during treatment<sup>139,140</sup>.

## 4.2 Duchenne muscular dystrophy

Duchenne muscular dystrophy (DMD) is an X-linked hereditary disease characterized by progressive loss of muscle ability and concomitantly increasing fibrosis. When supplemented or overexpressed in the Tg mice, IGFs have been shown to improve skeletal muscle functioning in a mouse model of DMD<sup>141,142</sup>. IGFBP-5, produced in excess by DMD fibroblast, is a potent inhibitor of proliferation in myoblast from DMD patients<sup>143</sup>. Bakay *et al.* used microarray studies to demonstrate an increase in mRNA levels of IGFs as well as IGFBPs in DMD skeletal muscle<sup>144</sup>. Therefore, the beneficial effects of IGF-I or IGF-II supplementation in dystrophic muscle could result from prevention of dose-dependent sequestration of the IGFs by inhibitory IGFBPs<sup>144</sup>.

## 4.3 Hypodactyly

Hypodactyly (Hd) mice manifest an impaired limb development with abnormal digit formation<sup>145</sup>. The distinctive pointed limb buds have been shown to be the result of a massive increase in apoptosis. Allan *et al.* examined IGF-1, IGFBP-2, and IGFBP-5 expression patterns during limb development in Hd mice and found down-regulated IGF-I expression in Hd limb buds in regions undergoing high levels of cell death, while IGFBP-5 was increased in the regions of cell death, consistent with reports of its association with apoptosis in adult tissues. This study suggested that IGF and

its binding proteins may play an important role in the process of limb formation<sup>145</sup>.

#### 4.4 Cornelia de Lange syndrome

The Cornelia de Lange syndrome (CdLS) is a multisystem congenital disorder characterized by typical facial features, IUGR, short stature, microbrachycephaly, hirsutism, limb anomalies, cognitive retardation, and other organ system involvement<sup>146</sup>. Pregnancy associated plasma protein-A (PAPP-A) is an IGF-dependent IGFBP-4 degrading enzyme, which has been shown to enhance IGF activity by IGFBP-4 degradation<sup>147</sup>. Interestingly, PAPP-A levels are significantly reduced in second-trimester maternal serum from women with CdLS newborns<sup>146</sup>. Low maternal levels of PAPP-A at 8-14 wk gestation are associated with an increased risk of IUGR<sup>148</sup>. Alteration in IGF system, caused by low levels of PAPP-A, may contribute to pathogenesis of CdLS.

## 5. INSIGHTS FROM IGFBP AND ALS TRANSGENIC AND KO MICE

In vivo studies of IGFBP action have been limited until recently by the lack of sufficient quantities of purified IGFBPs. Some insights into the physiological role of the IGFBPs are now emerging with development of genetically modified mice with either absence or increased levels of IGFBPs achieved through gene inactivation (knockout, KO) or overexpression of additional gene copies (Tg mice), respectively. KO models exist for all six IGFBPs in mice<sup>149</sup>. Tg mouse models have been developed with either targeted or generalized overexpression of IGFBPs. Targeted and generalized IGFBP overexpression address different questions. Generalized IGFBP overexpression ensures that both high circulating and tissue levels are achieved and can determine the overall effect of increased binding protein on the whole animal. Targeted IGFBP overexpression, achieved by the use of tissue specific promoters, allows for the investigation of local effects of IGFBP expression in specific tissues without perturbing circulating levels of the IGFBPs, IGFs, GH, and other components of the IGF system. Phenotypic manifestation of IGFBP transgenic mice are summarized in Table 3. The predominant effect of generalized overexpression has been growth retardation as would be anticipated from inhibitory effect of IGFBP excess on the actions of IGFs<sup>150-154</sup>. In addition, impaired glucose homeostasis has been observed in both IGFBP-1- and IGFBP-3 Tg mice<sup>150,155,156</sup>. Similarly, reduced fecundity has been observed in IGFBP-1, -3, -5, and -6 Tg mice<sup>152-</sup> <sup>154,157</sup>. Localized tissue overexpression results in tissue inhibition of IGF action in most, but not all studies (Table 3).

## 5.1 IGFBP-1

IGFBP-1 KO mice demonstrated normal development<sup>149</sup>. The absence of a marked phenotype in this, and other IGFBP KO mice, may result from compensatory mechanisms, although the nature of these mechanisms have not been elucidated. After a two-thirds hepatectomy, one of the most rapidly and highly induced genes and proteins in regenerating liver is IGFBP-1<sup>158</sup>. Leu et al. studied the influence of IGFBP-1 nullification on hepatic necrosis and regeneration. They used IGFBP-1 KO mice, control mice and mice with immunoneutralized IGFBP-1, to compare effect of IGFBP-1 deficiency on hepatic necrosis after of a sublethal dose of Fas agonist <sup>159</sup>. IGFBP-1 deficiency was associated with more massive hepatocyte apoptosis and caspase activation, which was attenuated by pretreatment with IGFBP-1. IGFBP-1-deficient livers had enhanced signaling via the integrin receptor after Fas agonist treatment. Elevated expression of active TGF- $\alpha$ 1, a hepatocyte apoptogen, was observed in IGFBP-1-deficient livers that correlated with the appearance of the apoptotic process. IGFBP-1 KO mice also displayed increased injury in a toxic hepatic injury model caused by carbon tetrachloride. IGFBP-1 KO mice had abnormal liver regeneration after partial hepatectomy, characterized by liver necrosis and reduced and delayed hepatocyte DNA synthesis<sup>160</sup>. These findings indicate that IGFBP-1 functions as an important hepatic survival factor in the regenerating liver by reducing the level of proapoptotic signals. This IGFBP-1 effect may be IGFindependent, possibly mediated most via inhibition of integrin signaling, or IGF-dependent. Although mature hepatocytes have a low abundance of type I IGF receptors<sup>161</sup>, it is possible that hepatocyte progenitor cells have more abundant IGF receptors. Since IGFBP-1 appears to be important for hepatocyte regeneration it is possibley that under these circumstances IGFBP-1 may function to enhance IGF-I delivery to hepatocyte progenitor cells.

Phenotypic manifestations of IGFBP-1 overexpression in Tg mice suggest that excess IGFBP-1 consistently inhibits IGF action *in vivo*. Ubiquitous overexpression of IGFBP-1 at high levels is associated with pre- and post-natal retardation and reduced brain size<sup>150</sup>. Impaired brain development is consistently observed also in other IGFBP Tg mice and it is thought to be a particularly sensitive indicator of impaired IGF action caused by IGFBP excess<sup>152-154,162,163</sup>.

	·····			
IGFBP	Transgene construct <sup>1</sup>	Transgene expression	Phenotypic manifestation	Reference
IGFBP-1	MT- hIGFBP-1	Brain, heart, kidney, liver, lung, testes	Abnormal brain development, marked inhibition of hepatic preneoplasia	161,196
	mPGK- rIGFBP-1	Ubiquitous	Reduction in birth weight, postnatal growth retardation, reduction in litter size, impaired glucose tolerance	150,157, 164
	α1AT- hIGFBP-1	Liver	Reduced brain weight with structural alteration, reduced body weight gain, impaired glucose tolerance, reduced fecundity, proteinuria and glomerular lesions, impaired spermatogenesis, delay I bone mineralization	
	hIGFBP-1	Liver	Impaired glucose tolerance, abnormalities in insulin action, growth retardation in early postnatal life	155,165
IGFBP-2	CMV- mIGFBP-2	Ubiquitous	Reduced postnatal body weight gain, reduced bone size and mineral content, inhibition of hypertrophic and hyperplastic effects of GH/IGF-I excess on adrenocortical cells	151,168, 169
IGFBP-3	mPGK- hIGFBP-3	Ubiquitous	Reduction in birthweight, postnatal growth retardation, reduction in litter size, impaired glucose tolerance, significant negative effect on bone formation	152,156, 170
	CMV- hIGFBP-3	Ubiquitous	Reduction in birth weight, postnatal growth retardation, catch-up growth after puberty, reduction in litter size, increased adiposity, impaired glucose tolerance and insulin sensitivity, significant negative effect on bone formation	152,156, 170
	MT- hIGFBP-3	Kidney, intestine	Selective organomegaly, spleen, liver, heart	201
	WAP- hIGFBP-3	Mammary epithelium in late pregnancy and lactation	Reduced size of alveoli at peak lactation, inhibition of the gland from undergoing programmed remodeling and apoptosis after weaning	171
IGFBP-4	mSMP- rIGFBP-4	Smooth muscle cells (SMC)	Smooth muscle hypoplasia, significant reduction in wet weight of SMC-rich tissues, including bladder, intestine, aorta, uterus and stomach, with no change in total body or carcass weight	172,173
	hOCN- IGFBP-4	Osteoblasts	Markedly reduced cancellous bone formation and turnover and severely impaired overall postnatal skeletal and somatic growth	174
IGFBP-5	Cβ-actin- IGFBP-5	Ubiquitous	Pre-and postnatal growth retardation, neonatal mortality, reduced female fecundity, impaired myogenesis	153
	β-Ig-	Mammary	Impaired mammary development during	175

IGFBP	Transgene	Transgene	Phenotypic manifestation	Reference
	construct-	expression		
	rIGFBP-5	epithelium	pregnancy	
	rOCN-	Osteoblasts	Transient decrease in trabecular bone volume,	177
	rIGFBP-5		impaired osteoblstic function and osteopenia	
IGFBP-6	mGFAP-	CNS, gonads	Pre-and postnatal growth retardation, reduced	154
	hIGFBP6		brain size, reduced female fecundity	
ALS	CMV-	Ubiquitous	Modest postnatal growth retardation, reduction	185
	hALS		in litter size	

*Table 3.* Phenotypic manifestation of IGFBP overexpression in transgenic mice.  $^{1}h = human$ , m = mouse, r = rat, c = chicken, MT = metallothionein promoter, PGK = phosphoglycerate kinase I promoter,  $\alpha 1AT = \alpha 1$  antitrypsin promoter, CMV = cytomegalovirus promoter, WAP = whey acidic protein promoter, SMP = smooth muscle actin promoter, OCN = osteocalcin promoter,  $\beta$ -lg =  $\beta$ -lactoglobin promoter

Initial observation of hyperglycemic effects of acute IGFBP-1 administration in rats<sup>57</sup> has not been reproduced recently with IGFBP-1 elevated to levels observed in various catabolic conditions in humans<sup>58</sup>. However, the latter study showed inhibition of protein synthesis in the fast-twitch skeletal muscle, thus supporting a role for IGFBP-1 in muscle catabolism under stress conditions<sup>58</sup>. In two different models of IGFBP-1 Tg mice with high level of expression, chronic exposure to elevated IGFBP-1 levels leads to development of impaired glucose tolerance, which supports the notion of IGFBP-1 inhibitory role of IGFBP-1 in regulation of IGF-I and insulin action *in vivo*<sup>150,155,164</sup>.

In addition to impaired glucose tolerance, Crossey *et al.* studied effect of decidual IGFBP-1 excess on fetoplacental growth in Tg mice overexpressing human IGFBP-1 gene. In this model, human IGFBP-1 expression has the same pattern as in human tissues<sup>155</sup>. Endogenous fetal IGFBP-1 overexpression was associated with an impairment of fetal growth in midgestation. Maternal decidual IGFBP-1 excess was also associated with impaired fetal growth in independent of fetal genotype<sup>165</sup>. The authors further showed that decidual IGFBP-1 overexpression had a marked effect on placental development and demonstrated a predominantly maternal origin of IGFBP-1 in amniotic fluid. Placental morphology was abnormal in Tg females due to altered trophoblast invasion and differentiation<sup>165</sup>. These data are in agreement with prenatal growth retardation observed with ubiquitous overexpression of IGFBP-1<sup>150</sup> and also with human studies showing association between IUGR and elevated IGFBP-1<sup>119-123</sup>.

#### 5.2 IGFBP-2

IGFBP-2 KO mice express no IGFBP-2 mRNA and have no detectable IGFBP-2 in the adult circulation are viable and fertile<sup>165</sup>. Heterozygous and homozygous animals showed no significant differences in prenatal or postnatal body growth. Analyses of organ weights in adult males, however, revealed that spleen weight was reduced and liver weight was increased in the absence of IGFBP-2. In addition, ligand blot analyses of sera from adult IGFBP-2 null males showed that IGFBP-1, IGFBP-3, and IGFBP-4 levels were increased relative to wild-type mice. These results demonstrate that up-regulation of multiple IGFBPs accompanies the absence of IGFBP-2 and this may represent a partial compensation for IGFBP-2 deficiency. However, IGFBP-2 has a critical role, either directly or indirectly, in modulating spleen and liver size<sup>166</sup>.

IGFBP-2 has higher affinity for IGF-II than IGF-I and IGFBP-2 levels are increased in IGF-II Tg mice suggesting regulation of IGFBP-2 levels by its ligand<sup>167</sup>. IGFBP-2 Tg mice with ubiquitous **IGFBP-2** primary overexpression had ~3-fold elevation of serum IGFBP-2 levels whereas normal serum levels of IGF-I<sup>151</sup>. There were no apparent abnormalities of glucose metabolism. These mice demonstrated mild growth retardation (~10% by day 23) with mild impairment of linear growth and with reduced carcass weight and reduced brain size in later postnatal life. In an extension of their study, they overexpressed a mutated form of IGFBP-2 lacking RGD motif and showed that this domain is not essential for IGFBP-2 interaction with cell membrane in vivo and that the mutation of this sequence does not alter growth inhibitory effects of IGFBP-2<sup>168</sup>. Interestingly crossbreeding of giant PEPCK-bGH Tg mice overexpressing GH with IGFBP-2 Tg mice resulted in significantly lower body weight (by 17%), attenuated linear growth, and reduced organ size, compared to the giant mice<sup>169</sup>. These data indicate importance of tissue concentration of inhibitory IGFBPs for regulation of IGF and GH effects on tissue growth.

#### 5.3 IGFBP-3

IGFBP-3 KO mice are viable and fertile. Ubiquitous overexpression of IGFBP-3 leads to an extensive phenotype with prenatal and postnatal growth retardation in the presence of elevated IGFBP-3 and IGF-I levels in circulation<sup>152</sup>. Significant reduction of organ size has been observed. The elevated IGFBP-3 levels perturbed glucose metabolism as evidenced by increased fasting glycemia and insulinemia and impaired glucose tolerance<sup>156</sup>. Glucose uptake into skeletal muscle *in vivo* was significantly decreased under both fasting and insulin-stimulated conditions. IGFBP-3

females had mildly reduced fecundity. IGFBP-3 overexpression had significant negative effect on bone formation *in vivo*, inhibited osteoblast proliferation derived from transgenic mice *in vitro*, and accelerated bone resorption in IGFBP-3 transgenic mice. Total and cortical bone mineral density has been significantly decreased in IGFBP-3 transgenic mice<sup>170</sup>.

On the contrary, ubiquitous overexpression of mutant form of IGFBP-3, devoid of IGF binding affinity, had no measurable effect on pre- or postnatal linear growth or body weight acquisition. No alteration of glucose metabolism or organogenesis with exception of mildly reduced brain weight or glucose metabolism was observed in these mice (Silha & Murphy, unpublished data). These data support the notion that IGFBP-3 functions predominantly by attenuating IGF activity through IGF-dependent mechanisms *in vivo*.

In contrast, in Tg mice expressing human IGFBP-3 specifically in the mammary gland tissue, delayed involution of the gland with decreased number of apoptotic cells was observed. However, mammary gland development during puberty and pregnancy was normal<sup>171</sup>. This study demonstrated anti-apoptotic effects of IGFBP-3 *in vivo* and provides evidence that IGFBPs may act as a bifunctional, pro- and anti-apoptotic agents depending on tissue and developmental stage.

## 5.4 IGFBP-4

IGFBP-4 KO mice are viable, fertile and have no marked phenotype<sup>149</sup>. Fagin and coworkers used Tg mice with selective IGFBP-4 overexpression in smooth muscle to demonstrate importance of IGFBP-4 protease activity for regulation of IGF system *in vivo*.

Targeted expression of a protease-resistant IGFBP-4 mutant in smooth muscle resulted in greater growth inhibition of smooth muscle rich organs than equivalent levels of native IGFBP-4<sup>172,173</sup>. Targeted overexpression of IGFBP-4 to osteoblasts using a human osteocalcin promoter markedly reduced cancellous bone formation and turnover and severely impaired overall postnatal skeletal and somatic growth while intrauterine development was normal. These data are in accordance with inhibitory role of IGFBP-4 on IGF action *in vivo*<sup>174</sup>.

## 5.5 IGFBP-5

IGFBP-5 KO mice are also viable and fertile and lack a significant phenotype<sup>149</sup>. Ubiquitous overexpression of IGFBP-5 resulted in IUGR, postnatal growth inhibition, reduced female fertility and retarded skeletal

muscle development in Tg mice which is consistent with ubiquitous overexpression of IGFBP-3, the most closely related IGFBP<sup>152,153</sup>.

IGFBP-5 production by mammary epithelial cells increases dramatically during an involution of the mammary gland. In contrast to IGFBP-3, IGFBP-5 overexpression in mammary gland revealed decreased duct branching, cellularity, proliferation and increased apoptosis. These abnormalities could be corrected by treatment with IGF analogue with no affinity for IGFBP-5, suggesting that inhibition of IGF action in mammary gland is responsible for the phenotype of this transgenic model<sup>175</sup>.

Targeted overexpression of transcription factor AP- $2\gamma$ , which is involved in embryonic development but also in tumorigenesis, to mammary gland and seminal vesicles epithelium, identified IGFBP-5 as a target gene involved in coupling of increased proliferation and apoptosis in both tissues<sup>176</sup>.

Transgenic mice overexpressing IGFBP-5 in the bone microenvironment have impaired osteoblastic function, osteopenia, and a transient decrease in trabecular bone volume secondary to reduced trabecular number and thickness and a transient decrease in bone mineral apposition rate<sup>177</sup>.

#### 5.6 IGFBP-6

IGFBP-6 KO mice are viable and fertile and appear normal<sup>149</sup>. IGFBP-6 Tg mice overexpressing IGFBP-6 under the control of GFAP have high levels of transgene in brain and gonads. These mice show marked reduction in brain size and reduced female fecundity, which is consistent with inhibitory function of IGFBP-6 *in vivo*<sup>154</sup>. Reduced fecundity of IGFBP-6 Tg females lends support to earlier studies suggesting antigonadotropin activity of IGFBP-6<sup>154,178</sup>.

#### 5.7 ALS

Inactivation of the ALS gene in mice results in mild retardation of postnatal growth despite profound disruptions in the circulating insulin-like growth factor system<sup>114</sup>. A modest growth deficit was observed in ALS KO mice after 3 weeks of life and reached 13% reduction in body weight by 10 weeks. This modest phenotype was observed despite reductions of 62 and 88% in the concentrations of plasma IGF-I and IGFBP-3, respectively. Increased IGF turnover accounted for these reductions because indices of IGF-I synthesis in liver and kidney were not decreased<sup>114</sup>. This study demonstrates importance of ALS for postnatal accumulation of IGF-I and IGFBP-3 in the circulation. However, ALS is not essential for growth. This phenotype of mild growth retardation in the presence of profoundly disrupted IGF system in circulation is in agreement with the manifestation of

ALS absence in humans and also with the phenotype of genetic nullification of hepatic IGF-I expression (LID mice)<sup>135,179,180</sup>.

Genetic nullification of hepatic IGF-I expression using cre/lox system in LID mice leads to a 75% decrease in IGF-I, decreased IGFBP-3, and normal free IGF-I in circulation while IGF-I expression in peripheral tissues remains normal<sup>179</sup>. Although there are no significant differences reported on body weight acquisition in these mice, development of skeletal muscle insulin resistance<sup>43,181</sup>, decreased adiposity<sup>181</sup>, increased amyloid deposition in the brain<sup>182</sup>, decreased myelopoiesis<sup>183</sup>, and impaired cortical bone formation in these mice do suggest an important role for circulating IGF-I in these tissues<sup>184</sup>. Some of these effects, such as increased skeletal muscle insulin resistance and decreased adiposity, are also mediated by reciprocal increase in GH secretion in LID mice<sup>43,181</sup>. Therefore, these studies support a role of circulating IGF-I and its intact ternary complex in physiologic functioning of IGF system.

In ALS Tg mice, overexpressing ALS ubiquitously under the control of CMV promoter, mild growth retardation was observed<sup>185</sup>. The mild growth deficit in the absence of alterations in IGF system in circulation suggests a role for tissue-specific restriction of ALS expression in postnatal life. The growth deficit was probably caused by decreased IGF-I availability in tissues, possibly due to formation of the ternary complex in tissues. Indeed, double transgenic mice overexpressing ubiquitously in tissues both human ALS and IGFBP-3 manifested more pronounced growth retardation compared to IGFBP-3 Tg mice<sup>185</sup>. These data suggest an inhibitory role for tissue ternary complex in growth regulation and restriction of the ternary complex into circulation may therefore confer growth advantage.

## 6. SUMMARY AND CONCLUSION

IGFBPs regulate growth and development by regulating IGF transport to tissues and IGF bioavailability to IGF receptors at cell membrane level. IGFBP excess leads predominantly to inhibition of IGF action and growth retardation with impaired organogenesis. Absence of human and also mouse ALS leads to decreased IGF-I levels in circulation and causes mild growth retardation. Although IGFBP KO mice demonstrate relatively minor phenotypes, the possibility of compensatory mechanisms that mask the phenotypic manifestation of lack of individual binding proteins needs to be further investigated. Recent studies of hepatic regeneration in IGFBP-1 KO mice and also with mutant IGFBP-3 Tg mice provide some limited support for the existence of IGF-independent mechanism of action *in vivo*.

#### ACKNOWLEDGMENTS

This research was supported by funds from the Canadian Institutes of Health Research. L.J.M is a recipient of the Henry G. Friesen Chair in Endocrine and Metabolic Research. J.V.S is a recipient of Canadian Diabetes Association Postdoctoral Fellowship.

#### REFERENCES

- W.S. Cohick and D.R. Clemmons, The insulin-like growth factors. Annu Rev Physiol, 55, 131-153 (1993).
- 2. E. Rinderknecht and R.E. Humbel, The amino acid sequence of human insulin-like growth factor-I and its structural homology with proinsulin. *J Biol Chem*, **253**, 2769-2776 (1978).
- 3. E.R: Froesch, H. Burgi, E.B. Ramseier, P. Ball and A. Labhart, Antibody –suppressible and nonsuppressible insulin-like activities in human serum and their physiologic significance. An insulin assay with adipose tissue of increased precision and specificity. *J Clin Invest*, **42**, 1816-1834 (1963).
- 4. J. Zapf, The IGF-insulin relationship. J Pediatr Endocrinol and Metab, 10, 87-95 (1997).
- 5. J.I. Jones and D.R. Clemmons, Insulin-like growth factors and their binding proteins: biological actions. *Endocr Rev*, **16**, 3-34 (1995).
- R.C. Baxter, M.A. Binoux, D.R. Clemmons, C.A. Conover, S.L. Drop, J.M. Holly, S. Mohan, Y. Oh and R.G. Rosenfeld, Recommendations for nomenclature of the insulinlike growth factor binding protein superfamily. *Endocrinology*, 139, 4036 (1998).
- 7. S.M. Firth and R.C. Baxter, Cellular actions of the insulin-like growth factor binding proteins. *Endocr Rev*, 23, 824-854 (2002).
- Z. Upton, S.J. Chan, D.F. Steiner, J.C. Wallace and F.J. Ballard, Evolution of insulinlike growth factor binding proteins. *Growth Regul*, 3, 29–32 (1993).
- 9. V. Hwa, Y. Oh and R.G. Rosenfeld, The insulin-like growth factor-binding protein (IGFBP) superfamily. *Endocr Rev*, **20**, 761-87 (1999).
- S.M. Firth, U. Ganeshprasad and R.C. Baxter, Structural determinants of ligand and cell surface binding of insulin-like growth factor binding protein-3. J Biol Chem, 273, 2631–2638 (1998).
- 11. J.I. Jones, A. Gockerman, W.H. Busby, G. Wright and D.R. Clemmons, Insulin-like growth factor binding protein 1 stimulates cell migration and binds to the  $\alpha_5\beta_1$  integrin by means of its Arg-Gly-Asp sequence. *Proc Natl Acad Sci USA*, **90**, 10553–10557 (1993).
- C. Binkert, J. Landwehr, J.L. Mary, J. Schwander and G. Heinrich, Cloning, sequence analysis and expression of a cDNA encoding a novel insulin-like growth factor binding protein (IGFBP-2). *EMBO J*, 8, 2497–2502 (1989).
- 13. S.M. Twigg, M.C. Kiefer, J. Zapf and R.C. Baxter, Insulin-like growth factor-binding protein 5 complexes with the acid-labile subunit: role of the carboxyl-terminal domain. *J Biol Chem*, **273**, 28791–28798 (1998).
- 14. S.M. Firth, D.R. Clemmons and R.C. Baxter, Mutagenesis of basic amino acids in the carboxyl-terminal region of insulin-like growth factor binding protein-5 affects acid-labile subunit binding. *Endocrinology*, **142**, 2147–2150 (2001).
- 15. T.J. Nam, W. Busby and D.R. Clemmons, Insulin-like growth factor binding protein-5 binds to plasminogen activator inhibitor-I. *Endocrinology*, **138**, 2972-2978 (1997).

- S.A. Weinzimer, T.B. Gibson, P.F. Collett-Solberg, A. Khare, B. Liu and P. Cohen, Transferrin is an insulin-like growth factor-binding protein-3 binding protein. J Clin Endocrinol Metab, 86, 1806–1813 (2001).
- 17. T. Nam, A. Moralez and D. Clemmons, Vitronectin binding to IGF binding protein-5 (IGFBP-5) alters IGFBP-5 modulation of IGF-I actions. *Endocrinology*, **143**, 30–36 (2002).
- B.A. Booth, M. Boes, D.L. Andress, B.L. Dake, M.C. Kiefer, C. Maack, R.J. Linhardt, K. Bar, E.E. Caldwell, J. Weiler and R.S. Bar, IGFBP-3 and IGFBP-5 association with endothelial cells: role of C-terminal heparin binding domain. *Growth Regul*, 5, 1–17 (1995).
- 19. L.J. Schedlich, S.L. Le Page, S.M. Firth, L.J. Briggs, D.A. Jans and R.C. Baxter, Nuclear import of insulin-like growth factor-binding protein-3 and -5 is mediated by the importin β subunit. *J Biol Chem*, **275**, 23462–23470 (2000).
- S.M. Twigg, M.C. Kiefer, J. Zapf and R.C. Baxter, A central domain binding site in insulin-like growth factor-binding protein 5 for the acid-labile subunit. *Endocrinology*, 141, 454–457 (2000).
- H. Song, J.H. Shand, J. Beattie, D.J. Flint and G.J. Allan, The carboxy-terminal domain of IGF-binding protein-5 inhibits heparin binding to a site in the central domain. J Mol Endocrinol, 26, 229–239 (2001).
- Y. Yamanaka, J.L. Fowlkes, E.M. Wilson, R.G. Rosenfeld and Y. Oh, Characterization of insulin-like growth factor binding protein-3 (IGFBP-3) binding to human breast cancer cells: kinetics of IGFBP-3 binding and identification of receptor binding domain on the IGFBP-3 molecule. *Endocrinology*, 140, 1319–1328 (1999).
- 23. S.M. Firth and R.C. Baxter, Characterisation of recombinant glycosylation variants of insulin-like growth factor binding protein-3. *J Endocrinol*, **160**, 379–387 (1999).
- 24. P.T. Cheung, E.P. Smith, S. Shimasaki, N. Ling and S.D. Chernausek, Characterization of an insulin-like growth factor binding protein (IGFBP-4) produced by the B104 rat neuronal cell line: chemical and biological properties and differential synthesis by sublines. *Endocrinology*, **129**, 1006–1015 (1991).
- 25. C.A. Conover and M.C. Kiefer, Regulation and biological effect of endogenous insulinlike growth factor binding protein-5 in human osteoblastic cells. *J Clin Endocrinol Metab*, **76**, 1153-1159.
- 26. L.A. Bach, Insulin-like growth factor binding protein-6: the "forgotten" binding protein? *Horm Metab Res.* **31**, 226-234 (1999).
- 27. J.A. Marinaro, D.J. Casley and L.A. Bach, O-glycosylation delays the clearance of human IGF-binding protein-6 from the circulation. *Eur J Endocrinol.* **142**, 512-516 (2000).
- J.A. Marinaro, G.M. Neumann, V.C. Russo, K.S. Leeding and L.A. Bach, Oglycosylation of insulin-like growth factor (IGF) binding protein-6 maintains high IGF-II binding affinity by decreasing binding to glycosaminoglycans and susceptibility to proteolysis. *Eur J Biochem*, 267, 5378-5386 (2000).
- 29. J.A. Coverley and R.C. Baxter, Phosphorylation of insulin-like growth factor binding proteins. *Mol Cell Endocrinol*, **128**, 1-5 (1997).
- H. Salahifar, S.M. Firth, R.C. Baxter and J.L. Martin, Characterization of an aminoterminal fragment of insulin-like growth factor binding protein-3 and its effects in MCF-7 breast cancer cells. *Growth Horm IGF Res*, 10, 367–377 (2000).
- C.A. Conover, S.K. Durham, J. Zapf, F.R. Masiarz and M.C. Kiefer, Cleavage analysis of insulin-like growth factor (IGF)-dependent IGF-binding protein-4 proteolysis and expression of protease-resistant IGF-binding protein-4 mutants. J Biol Chem, 270, 4395–4400 (1995).
- 32. Y. Imai, W.H. Busby, C.E. Smith, J.B. Clarke, A.J. Garmong, G.D. Horwitz, C. Rees and D.R. Clemmons, Protease-resistant form of insulin-like growth factor-binding

78

protein 5 is an inhibitor of insulin-like growth factor-I actions on porcine smooth muscle cells in culture. *J Clin Invest*, **100**, 2596–2605 (1997).

- J.I. Jones, W.H. Busby Jr, G. Wright, C.E. Smith, N.M. Kimack and D.R. Clemmons, Identification of the sites of phosphorylation in insulin-like growth factor binding protein-1. Regulation of its affinity by phosphorylation of serine 101. *J Biol Chem*, 268, 1125-1131 (1993).
- S. Mishra and L.J. Murphy, Phosphorylation of insulin-like growth factor (IGF) binding protein-3 by breast cancer cell membranes enhances IGF-I binding. *Endocrinology*, 144, 4042-4050 (2003).
- 35. L.A. Maile and J.M. Holly, Insulin-like growth factor binding protein proteolysis: occurrence, identification, role and regulation. *Growth Horm IGF Res*, **9**, 85-95 (1999).
- 36. R.C. Bunn and J.L. Fowlkes, Insulin-like growth factor binding protein proteolysis. *Trends Endocrinol Metab*, 14, 176-181 (2003).
- 37. Y.R. Boisclair, R.P. Rhoads, I. Ueki, J. Wang and G.T. Ooi, The acid-labile subunit (ALS) of the 150 kDa IGF-binding protein complex: an important but forgotten component of the circulating IGF system. J Endocrinol, 170, 63-70 (2001).
- L.J. Murphy, G.I. Bell and H.G. Friesen, Growth hormone stimulates sequential induction of c-myc and insulin-like growth factor I expression *in vivo*. *Endocrinology*, 120, 1806-1812 (1987).
- B.C. Villafuerte, W.N. Zhang and L.S. Phillips, Insulin and insulin-like growth factor-I regulate hepatic insulin-like growth factor binding protein-3 by different mechanisms. *Mol Endocrinol*, 10, 622-630 (1996).
- 40. E. Chin, J. Zhou, J. Dai, R.C. Baxter and C.A. Bondy, Cellular localization and regulation of gene expression for components of the insulin-like growth factor ternary binding protein complex. *Endocrinology*, **134**, 2498-2504 (1994).
- 41. E. Arany, S. Afford, A.J. Strain, P.J. Winwood, M.J. Arthur and D.J. Hill, Differential cellular synthesis of insulin-like growth factor binding protein-1 (IGFBP-1) and IGFBP-3 within human liver. *J Clin Endocrinol Metab*, **79**, 1871-1876 (1994).
- S. Yakar, J.L. Liu, A.M. Fernandez, Y. Wu, A.V. Schally, J. Frystyk, S.D. Chernausek, W. Mejia, D. LeRoith, Liver-specific igf-1 gene deletion leads to muscle insulin insensitivity. *Diabetes*, 50, 1110-1118 (2001).
- 43. P.D. Lee, L.C. Giudice, C.A. Conover and D.R. Powell, Insulin-like growth factor binding protein-1: recent findings and new directions. *Proc Soc Exp Biol Med*, **216**, 319-357 (1997).
- G. Povoa, G. Enberg, H. Jornvall and K. Hall, Isolation and characterization of a somatomedin-binding protein from mid-term human amniotic fluid. *Eur J Biochem*, 144, 199-204 (1984).
- 45. M.T. Brewer, G.L. Stetler, C.H. Squires, R.C. Thompson, W.H. Busby and D.R. Clemmons, Cloning, characterization, and expression of a human insulin-like growth factor binding protein. *Biochem Biophys Res Commun*, **152**, 1289-1297 (1988).
- S.K. Durham, A. Suwanichkul, A.O. Scheimann, D. Yee, J.G. Jackson, F.G. Barr and D.R. Powell, FKHR binds the insulin response element in the insulin-like growth factor binding protein-1 promoter. *Endocrinology*, **140**, 3140-3146 (1999).
- 47. C. Seneviratne, J.M. Luo and L.J. Murphy, Transcriptional regulation of rat IGFbinding protein-1 expression by growth hormone. *Mol Endocrinol*, **4**, 1199-1204 (1990).
- 48. L. Poretsky, B. Chun, H.C. Liu and Z. Rosenwaks, Insulin-like growth factor II (IGF-II) inhibits insulin-like growth factor binding protein I (IGFBP-1) production in luteinized human granulosa cells with a potency similar to insulin-like growth factor I (IGF-I) and higher than insulin. J Clin Endocrinol Metab, 81, 3412-3414 (1996).
- P.D. Lee, A. Suwanichkul, L.A. DePaolis, M.B. Snuggs, S.L. Morris and D.R. Powell, Insulin-like growth factor (IGF) suppression of IGFBP-1 production: evidence for mediation by the type I IGF receptor. *Regul Pept*, 48, 199-206 (1993).

- 50. C.A. Benbassat, D.D. Lazarus, S.B. Cichy, T.M. Evans, L.L. Moldawer, S.F. Lowry and T.G. Unterman, Interleukin-1 alpha (IL-1 alpha) and tumor necrosis factor alpha (TNF alpha) regulate insulin-like growth factor binding protein-1 (IGFBP-1) levels and mRNA abundance *in vivo* and *in vitro*. *Horm Metab Res*, **31**, 209-215 (1999).
- H. Yki-Jarvinen, S. Makimattila, T. Utriainen and E.M. Rutanen, Portal insulin concentrations rather than insulin sensitivity regulate serum sex hormone-binding globulin and insulin-like growth factor binding protein 1 in vivo. J Clin Endocrinol Metab, 80, 3227-3232 (1995).
- 52. M. Westwood, J.M. Gibson, A.C. Williams, P.E. Clayton, O. Hamberg, A. Flyvbjerg and A. White, Hormonal regulation of circulating insulin-like growth factor-binding protein-1 phosphorylation status. *J Clin Endocrinol Metab*, **80**, 3520-3527 (1995).
- 53. M. Westwood, J.M. Gibson, A.J. Davies, R.J. Young and A. White, The phosphorylation pattern of insulin-like growth factor-binding protein-1 in normal plasma is different from that in amniotic fluid and changes during pregnancy. J Clin Endocrinol Metab, **79**, 1735-1741 (1994).
- B.L. Nyomba, L., Berard and L.J. Murphy, Free insulin-like growth factor I (IGF-I) in healthy subjects: relationship with IGF-binding proteins and insulin sensitivity. J Clin Endocrinol Metab, 82, 2177-2181 (1997).
- 55. H. Norrelund, J. Frystyk, J.O. Jorgensen, N. Moller, J.S. Christiansen, H. Orskov and A. Flyvbjerg, The effect of growth hormone on the insulin-like growth factor system during fasting. *J Clin Endocrinol Metab*, **88**, 3292-3298 (2003).
- 56. J. Frystyk, C. Skjaerbaek, E. Vestbo, S. Fisker and H. Orskov, Circulating levels of free insulin-like growth factors in obese subjects: the impact of type 2 diabetes. *Diabetes Metab Res Rev*, **15**, 314-322 (1999).
- M.S. Lewitt, G.S. Denyer, G.J. Cooney and R.C. Baxter, Insulin-like growth factorbinding protein-1 modulates blood glucose levels. Endocrinology. 129:2254-2256 (1991).
- 58. C.H. Lang, T.C. Vary and R.A. Frost, Acute *in vivo* elevation of insulin-like growth factor (IGF) binding protein-1 decreases plasma free IGF-I and muscle protein synthesis. *Endocrinology*,**144**, 3922-3933 (2003).
- 59. G. Lamson, H. Pham, Y. Oh, I. Ocrant, J. Schwander and R.G. Rosenfeld, Expression of the BRL-3A insulin-like growth factor binding protein (rBP-30) in the rat central nervous system. *Endocrinology*, **125**, 1100-1102 (1989).
- W.F. Blum, N. Horn, J. Kratzsch, J.O. Jorgensen, A. Juu, D. Teale, K. Mohnike and M.B. Ranke, Clinical studies of IGFBP-2 by radioimmunoassay. *Growth Regul*, 3, 100-104 (1993).
- 61. R.G. Rosenfeld, H. Pham, C.A. Conover, R.L. Hintz and R.C. Baxter, Structural and immunological comparison of insulin-like growth factor binding proteins of cerebrospinal and amniotic fluids. *J Clin Endocrinol Metab*, **68**, 638-646 (1989).
- K. Chadelat, M. Boule, S. Corroyer, B. Fauroux, B. Delaisi, G. Tournier and A. Clement, Expression of insulin-like growth factors and their binding proteins by bronchoalveolar cells from children with and without interstitial lung disease. *Eur Respir J*, 11, 1329-1336 (1998).
- 63. R.G. Rosenfeld, H. Pham, Y. Oh, G. Lamson and L.C. Giudice, Identification of insulin-like growth factor-binding protein-2 (IGF-BP-2) and a low molecular weight IGF-BP in human seminal plasma. *J Clin Endocrinol Metab*, **70**, 551-553 (1990).
- 64. G.T. Ooi, C.C. Orlowski, A.L. Brown, R.E. Becker, T.G. Unterman and M.M. Rechler, Different tissue distribution and hormonal regulation of messenger RNAs encoding rat insulin-like growth factor-binding proteins-1 and -2. *Mol Endocrinol*, **4**, 321-328 (1990).
- 65. J. Zapf, C. Schmid, H.P. Guler, M. Waldvogel, C. Hauri, E. Futo, P. Hossenlopp, M. Binoux and E.R. Froesch, Regulation of binding proteins for insulin-like growth factors in humans. Increased expression of IGF binding protein 2 during IGF I treatment of

healthy adults and in patients with extrapancreatic tumor hypoglycemia. J Clin Invest, 86, 952-961 (1990).

- 66. D.R. Clemmons, D.K. Snyder and W.H. Busby Jr, Variables controlling the secretion of IGFBP-2 in normal human subjects. *J Clin Endocrinol Metab*, **73**, 727-733 (1991).
- D.R. Counts, H. Gwirtsman, L.M. Carlsson, M. Lesem and G.B. Cutler Jr, The effect of anorexia nervosa and refeeding on growth hormone-binding protein, the insulin-like growth factors (IGFs), and the IGF-binding proteins. J Clin Endocrinol Metab, 75, 762-767 (1992).
- W.J. Smith, L.E. Underwood and D.R. Clemmons, Effects of caloric or protein restriction on insulin-like growth factor-I (IGF-I) and IGF-binding proteins in children and adults. J Clin Endocrinol Metab, 80, 443-449 (1995).
- P.V. Carroll, M. Umpleby, E.L. Alexander, V.A. Egel, K.V. Callison, P.H. Sonksen and D.L. Russell-Jones, Recombinant human insulin-like growth factor-I (rhIGF-I) therapy in adults with type 1 diabetes mellitus: effects on IGFs, IGF-binding proteins, glucose levels and insulin treatment. *Clin Endocrinol (Oxf)*, **49**, 739-746 (1998).
- 70. W.F. Blum, M.B. Ranke, K. Kietzmann, B. Tonshoff and O. Mehls, Growth hormone resistance and inhibition of somatomedin activity by excess of insulin-like growth factor binding protein in uraemia. *Pediatr Nephrol*, **5**, 539-544 (1991).
- 71. P. Cohen, D.M. Peehl, B. Baker, F. Liu, R.L. Hintz and R.G. Rosenfeld, IGF axis abnormalities in prostatic stromal cells from patients with benign prostatic hyperplasia. *J Clin Endocrinol Metab*, **79**, 1410-1415 (1994).
- 72. M.G. Moore, L.A. Wetterau, M.J. Francis, D.M. Peehl, P. Cohen, Novel stimulatory role for IGFBP-2 in prostate cancer cells. *Int J Cancer*, **105**, 14-19 (2003).
- 73. G.N. Fuller, C.H. Rhee, K.R. Hess, L.S. Caskey, R. Wang, J.M. Bruner, W.K. Yung and W. Zhang, Reactivation of IGFBP-2 expression in glioblastoma multiforme: a revelation by parallel gene expression profiling. *Cancer Res*, **59**, 4228-4232 (1999).
- 74. T. Arai, W. Busby Jr and D.R. Clemmons, Binding of insulin-like growth factor (IGF) I or II to IGF-binding protein-2 enables it to bind to heparin and extracellular matrix. *Endocrinology*, **137**, 4571-4575 (1996).
- V.C. Russo, G. Rekaris, N.L. Baker, L.A. Bach and G.A. Werther, Basic fibroblast growth factor induces proteolysis of secreted and cell membrane-associated insulin-like growth factor binding protein-2 in human neuroblastoma cells. *Endocrinology*, 140, 3082-3090 (1999).
- C.A. Conover and S. Khosla, Role of extracellular matrix in insulin-like growth factor (IGF) binding protein-2 regulation of IGF-II action in normal human osteoblasts. *Growth Horm IGF Res*, 13, 328-335 (2003).
- W.I. Wood, G. Cachianes, W.J. Henzel, G.A. Winslow, S.A. Spencer, R. Hellmiss, J.L. Martin and R.C. Baxter, Cloning and expression of the GH dependent IGF binding protein. *Molec Endocr*, 2, 1176-1185 (1988).
- B. Liu, H.Y. Lee, S.A. Weinzimer, D.R. Powell, J.L. Clifford, J.M. Kurie and P. Cohen, Direct functional interactions between insulin-like growth factor-binding protein-3 and retinoid X receptor-alpha regulate transcriptional signaling and apoptosis. *J Biol Chem*, 275, 33607-33613 (2000).
- J.P. Thissen, L.E. Underwood, D. Maiter, M. Maes, D.R. Clemmons and J.M. Ketelslegers, Failure of insulin-like growth factor-I (IGF-I) infusion to promote growth in protein-restricted rats despite normalization of serum IGF-I concentrations. *Endocrinology*, **128**, 885-890 (1991).
- R.C. Baxter and C.T. Cowell, Diurnal rhythm of growth hormone-independent binding protein for insulin-like growth factors in human plasma. *J Clin Endocrinol Metab*, 65, 432-640 (1987).
- 81. K.L. Cohen and S.P. Nissley, The serum half-life of somatomedin activity: evidence for growth hormone dependence. *Acta Endocrinol*, **83**, 243-258 (1976).

- P. Marzullo, C. Di Somma, K.L. Pratt, J. Khosravi, A. Diamandis, G. Lombardi, A. Colao and R.G. Rosenfeld, Usefulness of different biochemical markers of the insulin-like growth factor (IGF) family in diagnosing growth hormone excess and deficiency in adults. *J Clin Endocrinol Metab*, 86, 3001-3008 (2001).
- H.M. Fraser, S.F. Lunn, H. Kim, W.C. Duncan, F.E. Rodger, P.J. Illingworth and G.F. Erickson, Changes in insulin-like growth factor-binding protein-3 messenger ribonucleic acid in endothelial cells of the human corpus luteum: a possible role in luteal development and rescue. *J Clin Endocr Metab*, 85, 1672-1677 (2000).
- S. Shimasaki, F. Uchiyama, M. Shimonaka and N. Ling, Molecular cloning of the cDNAs encoding a novel insulin-like growth factor-binding protein from rat and human. *Mol Endocrinol*, 4, 1451-1458 (1990).
- 85. D. Chelius, M.A. Baldwin, X. Lu and E.M. Spencer, Expression, purification and characterization of the structure and disulfide linkages of insulin-like growth factor binding protein-4. *J Endocrinol*, **168**, 283-296 (2001).
- 86. R. Zhou, D. Diehl, A. Hoeflich, H. Lahm and E. Wolf, IGFBP-4: biochemical characteristics and functional consequences. *J Endocrinol*, **178**, 177-193 (2003).
- 87. L. Poretsky, N.A. Cataldo, Z. Rosenwaks and L.C. Giudice, The insulin-related ovarian regulatory system in health and disease. *Endocr Rev*, **20**, 535-582 (1999).
- 88. S. Mohan, J.R. Farley and D.J. Baylink, Age-related changes in IGFBP-4 and IGFBP-5 levels in human serum and bone: implications for bone loss with aging. *Prog Growth Factor Res*, 6, 465-473 (1995).
- N. Miyakoshi, X. Qin, Y. Kasukawa, C.Richman, A.K. Srivastava, D.J. Baylink and S. Mohan, Systemic administration of insulin-like growth factor IGFBP-4 increases bone formation parameters in mice by increasing IGF bioavailability via an IGFBP-4 protease-dependent mechanism. *Endocrinology*, **142**, 2641-2648 (2001).
- M.C. Kiefer, F.R. Masiarz, D.M. Bauer and J. Zapf, Identification and molecular cloning of two new 30-kDa insulin-like growth factor binding proteins isolated from adult human serum. *J Biol Chem*, 266, 9043-9049 (1991).
- S.M. Twigg, M.C. Kiefer, J. Zapf and R.C. Baxter, Insulin-like growth factor-binding protein 5 complexes with the acid-labile subunit. Role of the carboxyl-terminal domain. *J Biol Chem*, 273, 28791-28798 (1998).
- 92. D.L. Andress, Insulin-like growth factor-binding protein-5 (IGFBP-5) stimulates phosphorylation of the IGFBP-5 receptor. *Am J Physiol*, **274**, E744-50 (1998).
- Y.G. Amaar, G.R. Thompson, T.A. Linkhart, S.T. Chen, D.J. Baylink and S. Mohan S, Insulin-like growth factor-binding protein 5 (IGFBP-5) interacts with a four and a half LIM protein 2 (FHL2). *J Biol Chem*, 277, 12053-12060 (2002).
- 94. E. Tonner, G. Allan, L. Shkreta, J. Webster, C.B. Whitelaw and D.J. Flint, Insulin-like growth factor binding protein-5 (IGFBP-5) potentially regulates programmed cell death and plasminogen activation in the mammary gland. *Adv Exp Med Biol*, **480**, 45-53 (2000).
- 95. D.R. Clemmons, Role of insulin-like growth factor binding proteins in controlling IGF actions. *Mol Cell Endocrinol*, **140**, 19-24 (1998).
- 96. C. Camacho-Hubner, W.H. Busby Jr, R.J. McCusker, G. Wright and D.R. Clemmons, Identification of the forms of IGF-binding proteins produced by human fibroblasts and the mechanisms that regulate their secretion. *J Biol Chem*, **26**7, 11949-11956 (1992).
- C.A. Conover, J.T. Clarkson and L.K. Bale, Effect of glucocorticoid on insulin-like growth factor (IGF) regulation of IGF-binding protein expression in fibroblasts. *Endocrinology*, 136, 1403-1410 (1995).
- 98. M.R. Schneider, E. Wolf, A. Hoeflich and H. Lahm, IGF-binding protein-5: flexible player in the IGF system and effector on its own. *J Endocrinol*, **172**, 423-440 (2002).
- 99. E. Tonner, G.J. Allan and D.J. Flint, Hormonal control of plasmin and tissue-type plasminogen activator activity in rat milk during involution of the mammary gland. J Endocrinol, 167, 265-273 (2000).

- 100. L.N. Thomas, P. Cohen, R.C. Douglas, C. Lazier and R.S. Rittmaster, Insulin-like growth factor binding protein 5 is associated with involution of the ventral prostate in castrated and finasteride-treated rats. *Prostate*, **35**, 273-278 (1998).
- 101. J.L. Martin, K.E. Willetts and R.C. Baxter, Purification and properties of a novel IGF-II binding protein from transformed human fibroblasts. *J Biol Chem*, **26**5, 4124-4130 (1990).
- S. Shimasaki, L. Gao, M. Shimonaka and L. Ling, Isolation and molecular cloning of insulin-like growth factor-binding protein-6. *Mol Endocrinol*, 5, 938-948 (1991).
- 103. L.A. Bach, N.R. Thotakura and M.M. Rechler, Human insulin-like growth factor binding protein-6 is O-glycosylated. *Biochem Biophys Res Commun*, **186**, 301-307 (1992).
- 104. B. Gabbitas and E. Canalis, Retinoic acid stimulates the transcription of insulin-like growth factor binding protein-6 in skeletal cells. *J Cell Physiol*, **169**, 15-22 (1996).
- 105. M. Roghani, C. Lassarre, J. Zapf, G. Povoa and M. Binoux, Two insulin-like growth factor (IGF)-binding proteins are responsible for the selective affinity for IGF-II of cerebrospinal fluid binding proteins. *J Clin Endocrinol Metab*, **73**, 658-666 (1991).
- 106. E.J. Beilharz, V.C. Russo, G. Butler, N.L. Baker, B. Connor, E.S. Sirimanne, M. Dragunow, G.A. Werther, P.D. Gluckman, C.E. Williams and A. Scheepens, Coordinated and cellular specific induction of the components of the IGF/IGFBP axis in the rat brain following hypoxic-ischemic injury. *Brain Res Mol Brain Res*, **59**, 119-134 (1998).
- 107. J. Van Doorn, A.M. Ringeling, S.S. Shmueli, M.C. Kuijpers, A.C. Hokken-Koelega, S.C. van Buul-Offers and M. Jansen, Circulating levels of human insulin-like growth factor binding protein-6 (IGFBP-6) in health and disease as determined by radioimmunoassay. *Clin Endocrinol (Oxf)*, **50**, 601-609 (1999).
- R.C. Baxter, Characterization of the acid-labile subunit of the growth hormonedependent insulin-like growth factor binding protein complex. J Clin Endocrinol Metab, 67, 265-272 (1988).
- S.R. Leong, R.C. Baxter, T. Camerato, J. Dai and W.I. Wood, Structure and functional expression of the acid-labile subunit of the insulin-like growth factor-binding protein complex. *Molec Endocr*, 6, 870-876 (1992).
- 110. C.Y. Lee and M.M. Rechler, Formation of 150-kDa binary complexes of insulin-like growth factor binding protein-3 and the acid-labile subunit *in vitro* and *in vivo*. *Prog Growth Factor Res*, 6, 241-251 (1995).
- 111. C.Y. Lee, F.J. Cohen, H.B. Wu, G.T. Ooi and M.M. Rechler, The absence of 150-kDa insulin-like growth factor complexes in fetal rat serum is not due to a lack of functional acid-labile subunit. *Horm Metab Res*, **31**, 182–185 (1999).
- 112. G.T. Ooi, F.J. Cohen, L.Y. Tseng, M.M. Rechler and Y.R. Boisclair, Growth hormone stimulates transcription of the gene encoding the acid-labile subunit (ALS) of the circulating insulin-like growth factor-binding protein complex and ALS promoter activity in rat liver. *Mol Endocrinol*, **11**, 997–1007 (1997).
- 113. M. Binoux and P. Hossenlopp, Insulin-like growth factor (IGF) and IGF-binding proteins: Comparison of human serum and lymph. J Clin Endocrinol Metab, 134, 2404–2409 (1988).
- 114. I. Ueki, G.T. Ooi, M.L. Tremblay, K.R. Hurst, L.A. Bach and Y.R. Boisclair, Inactivation of the acid labile subunit in mice results in mild retardation of postnatal growth despite profound disruptions in the circulating insulin-like growth factor system. *Proc Natl Acad Sci USA*, 97, 6868–6873 (2000).
- 115. A.L. Brown, D.E. Graham, S.P. Nissley, D.J. Hill, A.J. Strain and M.M. Rechler, Developmental regulation of insulin-like growth factor-II mRNA in different rat tissues. *J Biol Chem*, 261, 13144–13151 (1986).

- 116. P. Bang, M. Stangenberg, M. Westgren and R.G. Rosenfeld, Decreased ternary complex formation and predominance of a 29 kDa IGFBP-3 fragment in human fetal serum. *Growth Regul*, 4, 68-76 (1994).
- 117. A. Juul, S. Moller, E. Mosfeldt-Laursen, M.H. Rasmussen, T. Scheike, S.A. Pedersen, K.W. Kastrup, H. Yu, J. Mistry, S. Rasmussen, J. Muller, J. Henriksen and N.E. Skakkebaek, The acid-labile subunit of human ternary insulin-like growth factor binding protein complex in serum: hepatosplanchnic release, diurnal variation, circulating concentrations in healthy subjects, and diagnostic use in patients with growth hormone deficiency. J Clin Endocrinol Metab, 83, 4408-4415 (1998).
- 118. J.B. Janosi, S.M. Firth, J.J. Bond, R.C. Baxter and P.J. Delhanty, N-Linked glycosylation and sialylation of the acid-labile subunit. Role in complex formation with insulin-like growth factor (IGF)-binding protein-3 and the IGFs. *J Biol Chem*, **274**, 5292-5298 (1999).
- S. Baldwin, T. Chung, M. Rogers, T. Chard and H.S. Wang, Insulin-like growth factorbinding protein-1, glucose tolerance and fetal growth in human pregnancy. J Endocrinol, 136, 319–325 (1993).
- 120. M.S. Boyne, M. Thame, F.I. Bennett, C. Osmond, J.P. Miell and T.E. Forrester, The relationship among circulating insulin-like growth factor (IGF)-I, IGF-binding proteins-1 and -2, and birth anthropometry: a prospective study. *J Clin Endocrinol Metab*, **88**, 1687-1691 (2003).
- 121. L.C. Giudice, F. de Zegher, S.E. Gargosky, et al. Insulin-like growth factors and their binding proteins in the term and preterm human fetus and neonate with normal and extremes of intrauterine growth. *J Clin Endocrinol Metab*, **80**, 1548–1555 (1995).
- 122. J. Verhaeghe, R. Loos, R. Vlietinck, E.V. Herck, R. van Bree and A.M. Schutter, C-peptide, insulin-like growth factors I and II, IGFBP-1 in cord serum of twins: genetic versus environmental regulation. *Am J Obstet Gynecol*, **175**, 1180-1188 (1996).
- 123. J. Verhaeghe, J. Billen and L.C. Giudice, Insulin-like growth factor-binding protein-1 in umbilical artery and vein of term fetuses with signs suggestive of distress during labor. *J Endocrinol*, **170**, 585–590 (2001).
- 124. R.M. Popovici, M. Lu, S. Bhatia, G.H. Faessen, A.J. Giaccia and L.C. Giudice, Hypoxia regulates insulin-like growth factor-binding protein 1 in human fetal hepatocytes in primary culture: suggestive molecular mechanisms for in utero fetal growth restriction caused by uteroplacental insufficiency. J Clin Endocr Metab 86, 2653-2659 (2001).
- 125. A. Barreca, M. Bozzola, A. Cesarone, P.H. Steenbergh, P.E. Holthuizen, F. Severi, G. Giordano and F. Minuto, Short stature associated with high circulating insulin-like growth factor (IGF)-binding protein-1 and low circulating IGF-II: effect of growth hormone therapy. *J Clin Endocrinol Metab*, **83**, 3534-3541 (1998).
- 126. A. Krook and S. O'Rahilly, Mutant insulin receptors in syndromes of insulin resistance. *Baillieres Clin Endocrinol Metab*, **10**, 97-122 (1996).
- 127. N. Longo, Y. Wang and M. Pasquali, Progressive decline in insulin levels in Rabson-Mendenhall syndrome. *J Clin Endocrinol Metab*, **84**, 2623-2629 (1999).
- 128. S. Bernardini, G.L. Spadoni, G. Povoa, B. Boscherini and K. Hall, Plasma levels of insulin-like growth factor binding protein-1, and growth hormone binding protein activity from birth to the third month of life. *Acta Endocrinol (Copenh)*, **127**, 31331-8 (1992).
- 129. J. Nakae, M. Kato, M. Murashita, N. Shinohara, T. Tajima and K. Fujieda, Long-term effect of recombinant human insulin-like growth factor I on metabolic and growth control in a patient with leprechaunism. *J Clin Endocrinol Metab*, **83**, 54254-9 (1998).
- J.D. Quin, A. Checkley, A. Gallagher, J. Jones, A.C. MacCuish and J.P. Miell, Response of insulin-like growth factor (IGF)-binding protein-1 (IGFBP-1) and IGFBP-3 to IGF-I treatment in severe insulin resistance. *J Endocrinol*, 141, 177-182 (1994).

- 131. N. Longo, R. Singh, L.D. Griffin, S.D. Langley, J.S. Parks and L.J. Elsas, Impaired growth in Rabson-Mendenhall syndrome: lack of effect of growth hormone and insulinlike growth factor-I. *J Clin Endocrinol Metab*, **79**, 799-805 (1994).
- 132. N. Longo, R. Singh and L.J. Elsas, Decreased half-life of insulin-like growth factor I in Rabson-Mendenhall syndrome. *J Inherit Metab Dis*, **24**, 546-550 (2001).
- 133. T. Kitamura, Y. Kitamura, J. Nakae, A. Giordano, S. Cinti, C.R. Kahn, A. Efstratiadis and D. Accili, Mosaic analysis of insulin receptor function. *J Clin Invest*, **113**, 209-219 (2004).
- 134. D. Accili, J. Drago, E.J. Lee, M.D. Johnson, M.H. Cool, P. Salvatore, L.D. Asico P.A. Jose, S.I. Taylor and H. Westphal, Early neonatal death in mice homozygous for a null allele of the insulin receptor gene. *Nat Genet*, **12**, 106-109 (1996).
- H.M. Domene, S.V. Bengolea, A.S. Martinez, M.G. Ropelato, P. Pennisi, P. Scaglia, J.J. Heinrich and H.G. Jasper, Deficiency of the circulating insulin-like growth factor system associated with inactivation of the acid-labile subunit gene. *NEJM*, 350, 570-577 (2004).
- 136. D.R. Powell, Effects of renal failure on the growth hormone-insulin-like growth factor axis. *J Pediatr*, **131**, S13-16 (1997).
- 137. M. Houang, S. Cabrol, L. Perin, B. Ducos, A. Bensman and Y. Le Bouc, Insulin-like growth factor-I (IGF-I), IGF binding proteins (IGFBP) and IGF type I receptor in children with various status of chronic renal failure. *Growth Horm IGF Res*, 10, 332-341 (2000).
- 138. T. Ulinski, S. Mohan, D. Kiepe, W.F. Blum, A.M. Wingen, O. Mehls and B. Tonshoff, Serum IGFBP-4 and IGFBP-5 in children with chronic renal failure: relationship to growth and glomerular filtration rate. The European Study Group for Nutritional Treatment of Chronic Renal Failure in Childhood. German Study Group for Growth Hormone Treatment in Chronic Renal Failure. *Pediatr Nephrol*, 14, 589-597 (2000).
- 139. D.R. Powell, F. Liu, B.K. Baker, R.L. Hintz, P.D. Lee, S.K. Durham, E.D. Brewer, J.W. Frane, S.L. Watkins and R.J. Hogg, Modulation of growth factors by growth hormone in children with chronic renal failure. The Southwest Pediatric Nephrology Study Group. *Kidney Int*, **51**, 1970-1979 (1997).
- 140. D.R. Powell, S.K. Durham, E.D. Brewer, J.W. Frane, S.L. Watkins, R.J. Hogg and S. Mohan, Effects of chronic renal failure and growth hormone on serum levels of insulin-like growth factor-binding protein-4 (IGFBP-4) and IGFBP-5 in children: a report of the Southwest Pediatric Nephrology Study Group. J Clin Endocrinol Metab, 84, 596-601 (1999).
- 141. G.S. Lynch, S.A. Cuffe, D.R. Plant and P. Gregorevic, IGF-I treatment improves the functional properties of fast- and slow-twitch skeletal muscles from dystrophic mice. *Neuromuscul Disord*, 11, 260-268 (2001).
- 142. J. Smith, C. Goldsmith, A. Ward and R. LeDieu, IGF-II ameliorates the dystrophic phenotype and coordinately down-regulates programmed cell death. *Cell Death Differ*, 7, 1109-1118 (2000).
- 143. M.A. Melone, G. Peluso, U. Galderisi, O. Petillo and R. Cotrufo, Increased expression of IGF-binding protein-5 in Duchenne muscular dystrophy (DMD) fibroblasts correlates with the fibroblast-induced downregulation of DMD myoblast growth: an *in vitro* analysis. J Cell Physiol, 185, 143-153 (2000).
- 144. M. Bakay, P. Zhao, J. Chen and E.P. Hoffman, A web-accessible complete transcriptome of normal human and DMD muscle. *Neuromuscul Disord*, **12**, S125-141 (2002).
- 145. G.J. Allan, D.J. Flint, S.M. Darling, J. Geh and K. Patel, Altered expression of insulinlike growth factor-1 and insulin like growth factor binding proteins-2 and 5 in the mouse mutant Hypodactyly (Hd) correlates with sites of apoptotic activity. *Anat Embryol (Berl)*, **202**, 1-11 (2000).

- 146. D.A. Aitken, M. Ireland, E. Berry, J.A. Crossley, J.N. Macri, J. Burn and J.M. Connor, Second-trimester pregnancy associated plasma protein-A levels are reduced in Cornelia de Lange syndrome pregnancies. *Prenat Diagn*, **19**, 706-710 (1999).
- 147. D. Byun, S. Mohan, M. Yoo, C. Sexton, D.J. Baylink and X. Qin, Pregnancy-associated plasma protein-A accounts for the insulin-like growth factor (IGF)-binding protein-4 (IGFBP-4) proteolytic activity in human pregnancy serum and enhances the mitogenic activity of IGF by degrading IGFBP-4 in vitro. J Clin Endocrinol Metab, 86, 847-854 (2001).
- 148. G.C. Smith, E.J. Stenhouse, J.A. Crossley, D.A. Aitken, A.D. Cameron and J.M. Connor, Early pregnancy levels of pregnancy-associated plasma protein a and the risk of intrauterine growth restriction, premature birth, preeclampsia, and stillbirth. J Clin Endocrinol Metab, 87, 1762-1767 (2002).
- 149. J.E. Pintar, Single and multiple knockouts of the IGFBPs. Program of the 83rd Annual Meeting of The Endocrine Society, Denver, CO, p 44 (Abstract S33-1) (2001)
- 150. R. Kadaba, D. Barron, M.S. Lewitt and L.J. Murphy, Growth retardation and hyperglycemia in insulin-like growth factor binding protein-1 transgenic mice. *Endocrinology*, **136**, 4029–4034 (1995).
- 151. A. Hoeflich, M. Wu, S. Mohan, J. Foll, R. Wanke, T. Froehlich, G.J. Arnold, H. Lahm, H.J. Kolb and E. Wolf, Overexpression of insulin-like growth factor-binding protein-2 in transgenic mice reduces postnatal body weight gain. *Endocrinology*, **140**, 5488–5496 (1999).
- 152. T. Modric, J. Silha, Z. Shi, Y. Gui, A. Suwanichkul, S.K. Durham, D.R. Powell and L.J. Murphy, Phenotypic manifestations of insulin-like growth factor binding protein-3 overexpression in transgenic mice. *Endocrinology*, **142**, 1958–1967 (2001).
- 153. D.A. Salih, G. Tripathi, C. Holding, T.A. Szestak, M.I. Gonzalez, E.J. Carter, L.J. Cobb, J.E. Eisemann and J.M. Pell, Insulin-like growth factor-binding protein 5 (Igfbp5) compromises survival, growth, muscle development, and fertility in mice. *Proc Natl Acad Sci USA*, **101**, 4314-4319 (2004)
- 154. G. Bienvenu, D. Seurin, P. Grellier, P. Froment, M. Baudrimont, P. Monget, Y. Le Bouc and S. Babajko, IGFBP-6 transgenic mice: post-natal growth, brain development and reproduction abnormalities. *Endocrinology*, **145**, 2412-2420 (2004).
- 155. P.A. Crossey, J.S. Jones and J.P. Miell, Dysregulation of the insulin/IGF binding protein-1 axis in transgenic mice is associated with hyperinsulinemia and glucose intolerance. *Diabetes*, **49**, 457–465 (2000).
- 156. J.V. Silha, Y. Gui and L.J. Murphy, Impaired glucose homeostasis in insulin-like growth factor-binding protein-3-transgenic mice. *Am J Physiol Endocrinol Metab*, **283**, E937-45 (2002).
- 157. H. Huang, K. Rajkumar and L.J. Murphy, Reduced fecundity in insulin-like growth factor-binding protein-1 transgenic mice. *Biol Reprod*, 56, 284-289 (1997).
- 158. A. Ghahary, G.Y. Minuk, J. Luo, T. Gauthier and L.J. Murphy, Effects of partial hepatectomy on hepatic insulin-like growth factor binding protein-1 expression. *Hepatology*, **15**, 1125-1131 (1992).
- 159. J.I. Leu, M.A. Crissey and R. Taub, Massive hepatic apoptosis associated with TGFbeta-1 activation after Fas ligand treatment of IGF binding protein-1-deficient mice. *J Clin Invest*, **111**, 129-139 (2003).
- 160. J.I. Leu, M.A. Crissey, L.E. Craig, R. Taub, Impaired hepatocyte DNA synthetic response posthepatectomy in insulin-like growth factor binding protein 1-deficient mice with defects in C/EBP beta and mitogen-activated protein kinase/extracellular signalregulated kinase regulation. *Mol Cell Biol*, 23, 1251-1259 (2003).
- 161. J.F. Caro, J. Poulos, O. Ittoop, W.J. Pories, E.G. Flickinger and M.K. Sinha, Insulin-like growth factor I binding in hepatocytes from human liver, human hepatoma, and normal, regenerating, and fetal rat liver. *J Clin Invest*, **81**, 976-981 (1988).

86

- 162. A.J. D'Ercole, Z. Dai, Y. Xing, C. Boney, M.B. Wilkie, J.M. Lauder, V.K. Han and D.R. Clemmons, Brain growth retardation due to the expression of human insulin like growth factor binding protein-1 in transgenic mice: an in vivo model for the analysis of IGF function in the brain. *Brain Res Dev Brain Res*, 82, 213-222 (1994).
- 163. E. Gay, D. Seurin, S. Babajko, S. Doublier, M. Cazillis and M. Binoux, Liver-specific expression of human insulin-like growth factor binding protein-1 in transgenic mice: repercussions on reproduction, ante- and perinatal mortality and postnatal growth. *Endocrinology*, **138**, 2937–2947 (1997).
- K. Rajkumar, M. Krsek, S.T. Dheen and L.J. Murphy, Impaired glucose homeostasis in insulin-like growth factor binding protein-1 transgenic mice. *J Clin Invest*, **98**, 1818-1825 (1996).
- 165. P.A. Crossey, C.C. Pillai and J.P. Miell, Altered placental development and intrauterine growth restriction in IGF binding protein-1 transgenic mice. *J Clin Invest*, **110**, 411-418 (2002).
- 166. T.L. Wood, L.E. Rogler, M.E. Czick, A.G.P. Schuller and J.E. Pintar, Selective alterations in organ sizes in mice with a targeted disruption of the insulin-like growth factor binding protein-2 gene. *Mol Endocrinol*, **14**, 1472–1482 (2000).
- 167. A. Blackburn, R.A. Dressendorfer, W.F. Blum, M. Erhard, G. Brem, C.J. Strasburger and E. Wolf, Interactions of insulin-like growth factor (IGF)-II and growth hormone *in vivo*: circulating levels of IGF-I and IGF-binding proteins in transgenic mice. Eur J Endocrinol, 137, 701-708 (1997).
- 168. A. Hoeflich, R. Reisinger, G.A. Vargas, M.W. Elmlinger, B. Schuett, P.M. Jehle I. Renner-Muller, H. Lahm, V.C. Russo and E. Wolf, Mutation of the RGD sequence does not affect plasma membrane association and growth inhibitory effects of elevated IGFBP-2 *in vivo. FEBS Lett*, **523**, 63-67 (2002).
- 169. A. Hoeflich, S. Nedbal, W.F. Blum, M. Erhard, H. Lahm, G. Brem, H.J. Kolb, R. Wanke and E. Wolf, Growth inhibition in giant growth hormone transgenic mice by overexpression of insulin-like growth factor-binding protein-2. *Endocrinology*, 142, 1889–1898 (2001).
- 170. J.V. Silha, S. Mishra, C.J. Rosen, W.G. Beamer, R.T. Turner, D.R. Powell and L.J. Murphy, Perturbations in bone formation and resorption in insulin-like growth factor binding protein-3 transgenic mice. *J Bone Miner Res*, **18**, 1834-1841 (2003).
- S. Neuenschwander, A. Schwartz, T.L. Wood, C.T. Roberts Jr, L. Hennighausen and D. LeRoith, Involution of the lactating mammary gland is inhibited by the IGF system in a transgenic mouse model. *J Clin Invest*, 97, 2225-2232 (1996).
- 172. J. Wang, W. Niu, D.P. Witte, S.D. Chernausek, Y.E. Nikiforov, T.L. Clemens, B. Sharifi, A.R. Strauch and J.A. Fagin, Overexpression of insulin-like growth factorbinding protein-4 (IGFBP-4) in smooth muscle cells of transgenic mice through a smooth muscle α-actin-IGFBP-4 fusion gene induces smooth muscle hypoplasia. *Endocrinology*, **139**, 2605–2614 (1998).
- 173. M. Zhang, E.P. Smith, H. Kuroda, W. Banach, S.D. Chernausek and J.A. Fagin, Targeted expression of a protease-resistant IGFBP-4 mutant in smooth muscle of transgenic mice results in IGFBP-4 stabilization and smooth muscle hypotrophy. *J Biol Chem*, 277, 21285-21290 (2002).
- 174. M. Zhang, M.C. Faugere, H. Malluche, C.J. Rosen, S.D. Chernausek and T.L. Clemens, Paracrine overexpression of IGFBP-4 in osteoblasts of transgenic mice decreases bone turnover and causes global growth retardation. *J Bone Miner Res*, **18**, 836-843 (2003).
- 175. E. Tonner, M.C. Barber, G.J. Allan, J. Beattie, J. Webster, C.B. Whitelaw and D.J. Flint, Insulin-like growth factor binding protein-5 (IGFBP-5) induces premature cell death in the mammary glands of transgenic mice. *Development*, **129**, 4547-4557 (2002).
- 176. R. Jager, U. Werling, S. Rimpf, A. Jacob and H. Schorle, Transcription factor AP-2gamma stimulates proliferation and apoptosis and impairs differentiation in a transgenic model. *Mol Cancer Res*, **1**, 921-929 (2003).

- 177. R.D. Devlin, Z. Du, V. Buccilli, V. Jorgetti and E. Canalis, Transgenic mice overexpressing insulin-like growth factor binding protein-5 display transiently decreased osteoblastic function and osteopenia. *Endocrinology*, **143**, 3955-3962 (2002).
- 178. R.M. Rohan, E. Ricciarelli, M.C. Kiefer, C.E. Resnick and E.Y. Adashi, Rat ovarian insulin-like growth factor-binding protein-6: a hormonally regulated theca-interstitialselective species with limited antigonadotropic activity. *Endocrinology*, **132**, 2507-2512 (1993).
- 179. S. Yakar, J.L. Liu, B. Stannard, A. Butler, D. Accili, B. Sauer and D. LeRoith, Normal growth and development in the absence of hepatic insulin-like growth factor I. *Proc Natl Acad Sci USA*, 96, 7324–7329 (1999).
- 180. K. Sjogren, J.L. Liu, K. Blad, S. Skrtic, O. Vidal, V. Wallenius, D. LeRoith, J. Tornell, O.G. Isaksson, J.O. Jansson and C. Ohlisson, Liver-derived insulin-like growth factor-I (IGF-I) is the principal source of IGF-I in blood but is not required for postnatal body growth in mice. *Proc Natl Acad Sci USA*, **96**, 7088–7092 (1999)
- 181. K. Sjogren, K. Wallenius, J.L. Liu, Y.M. Bohlooly, G. Pacini, L. Svensson, J. Tornell, O.G. Isaksson, B. Ahren, J.O. Jansson and C. Ohlsson, Liver-derived IGF-I is of importance for normal carbohydrate and lipid metabolism. *Diabetes*, 50, 1539-1545 (2001).
- 182. E. Carro, J.L. Trejo, T. Gomez-Isla, D. LeRoith and I. Torres-Aleman, Serum insulinlike growth factor I regulates brain amyloid-beta levels. *Nat Med*, **8**, 1390-1397 (2002).
- 183. L.A. Welniak, M. Karas, S. Yakar, M.R. Anver, W.J. Murphy and D. LeRoith, Effects of organ-specific loss of insulin-like growth factor-I production on murine hematopoiesis. *Biol Blood Marrow Transplant*, **10**, 32-39 (2004).
- 184. S. Yakar, C.J. Rosen, W.G. Beamer, C.L. Ackert-Bicknell, Y. Wu, J.L. Liu, G.T. Ooi, J. Setser, J. Frystyk, Y.R. Boisclair and D. LeRoith, Circulating levels of IGF-1 directly regulate bone growth and density. *J Clin Invest*, **110**, 771-781 (2002).
- J.V. Silha, T. Modric, Y. Gui, A. Suwanichkul, S.K. Durham, D.R. Powell and L.J. Murphy, Overexpression of the acid-labile subunit of IGF ternary complex in transgenic mice. *Endocrinology*, **142**, 4305–4313 (2001).
- J.I. Jones, A. Gockerman, W.H. Busby Jr, C. Camacho-Hubner and D.R. Clemmons, Extracellular matrix contains insulin-like growth factor binding protein-5: potentiation of the effects of IGF-I. J Cell Biol, 121, 679-687 (1993).
- 187. Y. Gui and L.J. Murphy, Insulin-like growth factor (IGF)-binding protein-3 (IGFBP-3) binds to fibronectin (FN): demonstration of IGF-I/IGFBP-3/fn ternary complexes in human plasma. J Clin Endocrinol Metab, 86, 2104-2110 (2001).
- 188. S. Mishra and L.J. Murphy, Insulin-like growth factor binding protein-3 interacts with autocrine motility factor/phosphoglucose isomerase (AMF/PGI) and inhibits the AMF/PGI function. *Cancer Res*, **64**, 2516-2522 (2004).
- 189. M. Ikonen, B. Liu, Y. Hashimoto, L. Ma, K.W. Lee, T. Niikura, I. Nishimoto and P. Cohen, Interaction between the Alzheimer's survival peptide humanin and insulin-like growth factor-binding protein 3 regulates cell survival and apoptosis. *Proc Natl Acad Sci USA*, **100**, 13042-13047 (2003).
- 190. V.K. Han, D.G. Matsell, P.J. Delhanty, D.J. Hill, S. Shimasaki and K. Nygard, IGFbinding protein mRNAs in the human fetus: tissue and cellular distribution of developmental expression. *Horm Res*, **45**, 160-166 (1996).
- 191. A.M. Suikkari, I. Leivo, M. Kamarainen, H. Holthofer, M. Seppala, M. Julkunen and R. Koistinen, Expression of insulin-like growth factor binding protein-1 mRNA in human fetal kidney. *Kidney Int*, 42, 749-754 (1992).
- 192. A.G. Schuller, J.W. van Neck, D.J. Lindenbergh-Kortleve, C. Groffen, I. de Jong, E.C. Zwarthoff and S.L. Drop, Gene expression of the IGF binding proteins during post-implantation embryogenesis of the mouse; comparison with the expression of IGF-I and -II and their receptors in rodent and human. Adv Exp Med Biol, 343, 267-277 (1993).

- 193. M. van Kleffens, C.A. Groffe, N.F. Dits, D.J. Lindenbergh-Kortleve, A.G. Schuller, S.L. Bradshaw, J.E. Pintar, E.C. Zwarthoff, S.L. Drop and J.W. van Neck, Generation of antisera to mouse insulin-like growth factor binding proteins (IGFBP)-1 to -6: comparison of IGFBP protein and messenger ribonucleic acid localization in the mouse embryo. *Endocrinology*, **140**, 5944-5952 (1999).
- D.G. Matsell, P.J. Delhanty, O. Stepaniuk, C. Goodyear and V.K. Han, Expression of insulin-like growth factor and binding protein genes during nephrogenesis. *Kidney Int*, 46, 1031-1042 (1994).
- 195. P.J. Delhanty, D.J. Hill, S. Shimasaki and V.K. Han, Insulin-like growth factor binding protein-4, -5 and -6 mRNAs in the human fetus: localization to sites of growth and differentiation? *Growth Regul*, **3**, 8-11 (1993).
- 196. S. Lu and M.C. Archer, Insulin-like growth factor binding protein-1 over-expression in transgenic mice inhibits hepatic preneoplasia. *Mol Carcinog*, **36**, 142-146 (2003).
- 197. P. Froment, D. Seurin, S. Hembert, J.E. Levine, C. Pisselet, D. Monniaux, M. Binoux and P. Monget, Reproductive abnormalities in human IGF binding protein-1 transgenic female mice. *Endocrinology*, **143**, 1801-1808 (2002).
- S. Doublier, D. Seurin, B. Fouqueray, M.C. Verpont, P. Callard, L.J. Striker, G.E. Striker, M. Binouxand L. Baud. Glomerulosclerosis in mice transgenic for human insulin-like growth factor-binding protein-1. *Kidney Int*, 57, 2299-2307 (2000).
- 199. P. Froment, C. Staub, S. Hembert, C. Pisselet, M. Magistrini, B. Delaleu, D. Seurin, J.E. Levine, L. Johnson, M. Binoux and P. Monget, Reproductive abnormalities in hIGFBP-1 transgenic male mice. *Endocrinology*, **145**, 2080-2091 (2004).
- 200. N. Ben Lagha, P. Menuelle, D. Seurin, M. Binoux, Y. Lebouc and A. Berdal, Bone formation in the context of growth retardation induced by hIGFBP-1 overexpression in transgenic mice. *Connect Tissue Res*, **43**, 515-519 (2002).
- 201. L.J. Murphy, P. Molnar, X. Lu and H.H. Huang, Expression of human insulin-like growth factor binding protein-3 in transgenic mice. *J Mol Endocrinol*, **15**, 293–303 (1995).

## Chapter 4

# GROWTH HORMONE, INSULIN-LIKE GROWTH FACTORS AND THE FEMALE REPRODUCTIVE SYSTEM

Carolyn A. Bondy and Jian Zhou

Developmental Endocrinology Branch National Institute of Child Health and Human Development Bethesda, MD, USA

Key words: Endometrium; IGFBP; IGF receptor; oocyte; ovary, steriodogenesis; uterus.

#### **1. INTRODUCTION**

Insulin-like growth factors (IGFs) promote coordinated somatic growth during fetal and postnatal development. These growth-promoting effects are mediated by the IGF-I receptor, which binds both IGF-I and IGF-II with high affinity. The IGF-I receptor is a membrane bound, ligand-activated tyrosine kinase that is structurally and functionally homologous to the insulin receptor and is very widely expressed in animal tissues. The two IGFs are abundant in the circulation and synthesized locally in many different tissues, including the reproductive system of most mammals. Given the widespread availability of the ligands and receptor, it is thought that interactions with diverse family of IGF binding proteins (IGFBPs 1-6) may modulate local IGF effects. These proteins bind IGF-I and II with high affinity, regulating their transport and presentation to the receptor and prolonging their half-life in the circulation and other fluid compartments<sup>1</sup>.

Growth hormone (GH) stimulates the synthesis and release of IGF-I from the liver and some local tissues, and this circulating IGF1 mediates most of GH's effects on somatic growth. It appears that GH's major effects on reproduction are mediated by IGF-I, but evidence for some facilitory effects by GH on ovarian function, mediated directly by the GH receptor, will be reviewed briefly. This chapter concentrates in main on the IGF system in mammalian ovary and uterus, comparing observations in rodent models, in large animals such as cow, pig and sheep and in the primate. Rodent models have been very informative because of the power of transgenics and targeted deletions, although their reproductive habits are very specialized and differ substantially from those of larger species, especially humans. Because of agricultural interest in promoting fertility and fecundity, there is abundant information on the IGF system in farm animals. Finally, we review somewhat more limited data available for humans and nonhuman primates. The rhesus monkey serves as a useful model for IGF actions in the human reproductive system<sup>2</sup>. While we are focused on mammals, it is interesting to note that a connection between insulin-like signaling and reproductive capacity is observed in evolutionarily distant species such as *C. elegans* and *D. melanogaster*, in which interruptions in insulin-like signaling pathways result in infertility<sup>3</sup>.

The IGF system is apparently involved in all aspects of female reproduction, including ovarian, uterine, and placental growth and function and the neuroendocrinological integration of nutritional status and reproductive competency. This chapter focuses on IGFs in the ovary and uterus (the IGF system and placenta is the subject of another chapter in this book). The IGF system is also important in mammary gland development, but that subject is beyond the scope of the present effort. Finally, there is a huge body of literature on *in vitro* effects of IGFs on various types of reproductive tissues that have been reviewed<sup>4</sup>, and this chapter concentrates primarily on *in vivo* studies.

#### 2. THE OVARY

#### 2.1 GH

Sexual maturation is delayed in states of GH deficiency or resistance, in both mice and men. This delay may be attributed to direct or indirect effects of GH or IGF-I on gonadotropin release as well as gonadal function. Ovarian development appears normal, however, in GH-deficient/resistant females, and they are generally fertile, although in mice, litter size is reduced due to reduced ovulatory rate<sup>5,6</sup>. Evaluation of ovaries of adult females at estrus revealed that follicles were normal in appearance and steroidogenic apparatus, but preovulatory follicles and corpora lutea were significantly reduced in number in GH receptor null mice. One of these studies reported decreased IGF1 mRNA levels in the GH receptor null ovaries<sup>5</sup> while the other found no decrease<sup>6</sup>. We found no alteration in ovarian IGF-I mRNA concentration after hypophysectomy, using *in situ* hybridization to directly measure hybrids in defined granulosa cell populations<sup>7</sup>. Thus it remains somewhat unsettled as to whether the reduced follicular growth in GH deficient or resistant females is due to reduced circulating IGF-I, reduced local IGF-I or to a non-IGF-I mediated effect of GH. While most of the anabolic effects of GH on ovarian function appear secondary to IGF-I, the presence of GH receptor expression in the ovary<sup>8,9</sup> leaves open the possibility for direct effects by GH on ovarian function not mediated by IGF-I.

GH treatment has been found to reduce the dose of exogenous gonadotrophin required to enhance folliculogenesis in women with hypogonadotrophic hypogonadism, but there is little evidence that GH is of clinical benefit in the management of patients with other ovulatory disorders, including polycystic ovary syndrome (PCOS), or in superovulation protocols for in *vitro* fertilization<sup>10</sup>.

#### 2.2 IGFs and IGF-I receptor

IGFs appear to be important regulators of ovarian follicular growth and selection, and luteal steroidogenesis. The rat and mouse have served as important experimental models for elucidation of IGF-I's role in folliculogenesis. The IGF-I receptor is highly expressed in the oocyte, in granulosa cells of follicles at all stages, from small primary to mature preovulatory follicles, and corpora luteal cells in both rat and mouse<sup>7,11</sup>. IGF-I expression is more selective, however, being limited to the granulosa of healthy, growing and selected follicles<sup>7,12</sup>. Ovarian IGF-I expression is selectively localized in follicle-stimulating hormone (FSH) receptor expressing follicles but does not appear to be regulated by gonadotropin or GH, while IGF-I receptor expression is dramatically increased by gonadotropin<sup>7,13</sup>. Local IGF-I expression is significantly correlated with granulosa cell proliferation in the murine ovary<sup>14</sup> and IGF-I receptor expression is correlated with high-level metabolic activity, reflected by glucose utilization (Fig. 1). It seems likely that increased energy requirements, associated with rapid follicular growth, in the case of preovulatory follicles, and steroidogenesis, in the case of corpora lutea, are met by enhanced "insulin-like" IGF-I effects promoting nutrient acquisition at the follicular level.



*Figure 1.* Ovarian IGF-I receptor expression reflects metabolic activity. Panel **A** shows an H&E stained section from a mature rat ovary and panels **B-D** are film autoradiographs from sequential ovary sections. **B** shows C-14-labeled 2-deoxyglucose uptake; **C** shows IGF-I and **D** IGF-I receptor mRNA by *in situ* hybridization. cl; Corpus luteum.

The cellular patterns of IGF-I and IGF-I receptor expression in the normal murine ovary correlate well with *in vitro* studies showing that IGF-I promotes proliferation in granulosa cells from the smallest follicles and steroid production in luteinized granulosa cells<sup>4,15</sup>. If, as we suggest, IGF-I's fundamental effects are to promote nutrient utilization, then IGF-I should effectively enhance whatever cellular processes are in progress. In the case of the murine ovary, local, granulosal IGF-I production promotes granulosa proliferation and hence follicular growth, while circulating IGF1 may act on the corpus luteum (CL) to promote steroidogenesis (Fig. 1). IGF-II is minimally expressed in the murine ovary, primarily in perifollicular blood vessels<sup>2</sup>, and IGF-II deletion does not impair fertility in the mouse. It should be noted that IGF-II is much less abundant in the circulation and tissues postnatally in murine species than in other mammals, where IGF-II is significantly more abundant than IGF-I.



*Figure 2.* Schematic illustrating IGF-I's proposed role in follicular development. Oocyte autonomous signals are hypothesized to initiate granulosa cell proliferation and IGF-I production. IGF-I augments FSHR expression and FSH augments both IGFIR and FSHR expression. This mutual complementary positive feedback loop within the follicle is hypothesized to be critical for the amplification of FSH action to induce the formation of mature graafian follicles. In the absence of IGF-I or FSH, follicles arrest at a preantral/early antral stage of development. Maximal FSH action leads to mature antrum formation and granulosa cell aromatase and LHR expression. The resulting peak in follicular E2 synthesis stimulates an LH surge, which in turn stimulates ovulation. For the sake of simplicity, thecal layer development, which is also impaired in the IGF-I knockout mouse, has been omitted from the diagram. From ref. 13.

Targeted IGF-I gene deletion mice have small ovaries, consistent with their dwarf size; these ovaries appear anatomically normal with a full complement of primordial follicles, as determined for two independent deletions in different mouse lines<sup>13,16</sup>. Thus, IGF-I apparently is not essential for the normal development or differentiation of the ovary. IGF-I null mice are infertile, however, owing to impaired follicular growth, since although primordial follicle numbers are normal in IGF- $\Gamma'$ -mice, few antral follicles are evident, no pregnancies occur and no corpora lutea are formed<sup>13</sup>. Moreover, these IGF-I null mice do not ovulate in response to gonadotropin hyperstimulation<sup>16</sup>. We showed that glucose transporter, FSH receptor and aromatase expression were all significantly reduced in granulosa cells of the IGF-I null ovary<sup>13,17</sup>, and that administration of IGF-I systemically restored their expression to wild-type levels, suggesting that the reduced gonadotropin responsiveness of these mice may be at least partially explained by reduced FSH receptor expression. Based on these observations, we proposed that local IGF-I production and reception promotes an amplification cycle as illustrated in Fig. 2. Local IGF-I produced by granulosa cells in small growing follicles enhances FSH receptor expression and thus follicle response to FSH. FSH, in turn, stimulates more IGF-I receptor expression providing a positive feedback loop allowing rapid follicular growth. The specific molecular pathways engaged by IGF-I and its cognate receptor in the granulosa cell have been elucidated and reviewed by JoAnne Richards<sup>18</sup>. She has identified major signaling targets of IGF-I receptor action in the ovary, including the PI-3 kinase cascade, PKB/Akt and SGK, and members of the forkhead transcription factor family. She has also demonstrated that estradiol participates in the growth amplification cycle we have outlined earlier by augmenting IGF-I receptor expression.

The ovarian IGF system has been investigated in large farm animals such as pig, cow, and sheep as recently reviewed<sup>19</sup>. Interestingly, the sow is the only large animal that selectively expresses IGF-I in growing follicles similar to the murine example<sup>20</sup>, and is also similar to murine species in being polytocous, in contrast to cattle and sheep. IGF-I and IGF-I receptor are selectively concentrated in healthy follicles, characterized by expression of aromatase and the gonadotropin receptors, in the sow ovary (Fig. 3). IGF-I expression does not characterize the emergence of ovulatory follicles in any of the other species that have been examined to date, although increasing IGF-I concentration, presumably derived from the circulation, is described for dominant follicle formation in cattle<sup>21</sup>. IGF-II is expressed in the thecainterstitial compartment of porcine, bovine, and ovine and primate ovaries and is abundant in the fluid obtained from large follicles of these species, as is IGF-I, apparently derived from the systemic circulation<sup>19,22</sup>. Evidence suggests that circulating IGF-I plays a role in follicular growth in domestic animals, since reduction in GH secretion and hence IGF-I levels leads to reduced follicular development, while increased IGF-I levels enhance follicular growth<sup>23-25</sup>. This may be relevant to nutritional aspects of reproduction in farm animals, since nutritional deficiencies are associated reduced follicular with reduced circulating IGF-I levels and maturation/ovulation<sup>24,26</sup>

We have investigated IGF system expression in the nonhuman primate (rhesus macaque) and human ovary. As reported above for monotocous farm animals, IGF-I expression is not normally detected in rhesus or human growing follicles while IGF-II is expressed in the theca-interstitial compartment without apparent selectivity for follicle or cycle stage (Fig. 4B)<sup>27-29</sup>. IGF-II is, however, expressed in the membrana granulosa of large pre-ovulatory and ovulatory follicles in the monkey (Fig. 4J). We are not sure whether this IGF-II is expressed by granulosa cells or by invading stromal elements in the periovulatory follicle. As in all the other species, the IGF-I receptor is abundantly expressed in developing follicles and corpora lutea<sup>28</sup>.



*Figure 3.* IGF system gene expression in the mid-follicular (day 16) gilt ovary. Panel **A** is an H & E-stained section and panels **B–I** are film autoradiographs of sequential sections hybridized to antisense RNA probes for the indicated mRNAs. Follicles 1-6 are healthy growing follicles of various sizes. Follicles 7-9 are clearly attetic as demonstrated by shredding of the membrana granulosa and by loss of the follicle's oval shape. Follicles 10 and 11 appear healthy by morphological criteria but are presumed to be destined for atresia based on absence of gonodotropin receptor or aromatase mRNA. ARO, aromatase; FSHR, FSH receptor; IGFIR,IGF-I receptor; LHR, LH receptor. Scale bar = 2.5 mm. From ref. 20.



*Figure 4.* IGF-II and IGFBP4 gene expression in normal mid-cycle (day 14) rhesus monkey ovaries. On the extreme left of each row are H&E-stained ovary sections and the subsequent panels show film autoradiographs of sequential sections hybridized to cRNA probes for IGF-II, IGFBP4 and aromatase, or in panel L, the IGFBP4 protease, PAPP-A. A single dominant follicle (DF), defined by maximum cross-sectional diameter >2 mm and aromatase expression, is apparent in each ovary. The emerging DF in A–D has the highest aromatase expression, but other, smaller follicles still have discernible aromatase and IGFBP4 expression are restricted to the DF, while IGF-II mRNA is still evident in numerous smaller follicles. The DF in panel I is in the process of ovulating, and IGF-II, IGFBP4, and PAPP-A mRNAs are abundant in the exfoliating membrana granulose (J–L). This ovary is cut through the ovarian pedicle and the profusion of engorged blood vessels (BV) is quite impressive. DF, dominant follicle; CL, Corpus luteum. Scale bar = 0.8 mm. From ref. 29.

## 2.3 Androgen and the ovarian IGF system

While IGF-I is generally below detection limits in the normal primate ovary, increased IGF-I and IGF-I receptor expression are found in ovaries from androgen-treated monkeys<sup>30</sup>. We investigated the effects of androgen

on the monkey ovary to test the hypothesis that androgens may contribute to the development of the polycystic ovary syndrome<sup>31</sup>. Small preantral and antral follicles were significantly and progressively increased in number and thecal layer thickness in testosterone and dihydrotestoserone treated monkeys over 3-10 days of treatment. Granulosa and thecal cell proliferation, as determined by immunodetection of the Ki67 antigen, are significantly increased in these follicles. Androgens appear to boost the recruitment of resting, primordial follicles into the actively growing pool, with the number of primary follicles significantly increasing over time during androgen treatment<sup>32</sup>. Androgen treatment results in a three-fold increase in IGF-I and five-fold increase in IGF-I receptor mRNA in primordial follicle oocytes<sup>32</sup>, suggesting that androgens may promote initiation of primordial follicle growth by augmenting IGF-I activity in the primordial follcile. In addition, androgen treatment results in significant, 3-4-fold increases in IGF-I and IGF-I receptor expression in granulosa, the cal and interstitial compartments (Fig. 5). Interestingly, FSH receptor expression was increased in parallel with IGF-I receptor in these androgen-treated animals<sup>33</sup> (Fig. 6), suggesting that in primates as in rodents, IGFs act to enhance FSH receptor expression and responsiveness (see Fig 2). Follicular atresia was not increased and there were actually significantly fewer apoptotic granulosa cells in the androgentreated groups<sup>31</sup>. These findings show that, over the short term at least, androgens enhance IGF-I and IGF-I receptor expression along with follicular growth and survival in the primate ovary, suggesting that IGF-I and its cognate receptor may mediate androgen-induced ovarian follicular and thecal-interstitial growth.

The mechanism whereby androgens promote primordial follicle development appears to involve activation of oocyte IGF1 signaling. The IGF-I receptor is abundant in oocytes from all species investigated<sup>7,11,20,28</sup> and IGF-I stimulates oocyte metabolic activity and maturation in vitro<sup>34,35</sup>. Thus, enhanced IGF-I signaling through increased expression of both the peptide and its receptor in oocytes from androgen-treated monkeys may trigger oocyte activation and initiation of follicle growth. These observations provide a plausible explanation for the pathogenesis of "polycystic" ovaries in hyperandrogenism. Pathological studies show increased numbers of in ovaries including primary follicles, growing follicles. from hyperandrogenemic women with polycystic ovary syndrome<sup>36</sup>. Increased follicle numbers are also seen in testosterone-treated women<sup>37,38</sup> showing that androgens, whether derived from ovary, adrenal or exogenous sources, may stimulate excessive ovarian follicular growth. Further support for this notion is provided by the observation that treatment of women with PCOS with androgen receptor blockade results in diminution of follicle numbers<sup>39</sup>.





Figure 5. Effect of androgens on IGF-I receptor mRNA in the monkey ovary. Representative film autoradiographs illustrate in *situ* hybridization results from control, placebo-treated ( $\mathbf{A}$ ), testosterone-treated ( $\mathbf{B}$ ), and DHT-treated ( $\mathbf{C}$ ) monkeys. The IGF-I receptor films were exposed for 5 days. From ref. 30.

*Figure 6.* Increased FSHR gene expression in follicles from testosterone-treated monkeys. Representative film autoradiographs from follicular phase control (A), 3 day testosterone-treated (B), and 10 day testosterone-treated (C) monkeys are shown. Note the increased number of follicles in the androgen-treated ovaries. Bar =

In summary, although very little IGF-I is detected in the normal primate ovary, both IGF-I and IGF-I receptor expression are greatly increased by exposure to elevated androgen levels. In the model for generation of PCOS illustrated in Fig. 7, the polycystic ovarian phenotype may result from at least 2 different pathways. Excess systemic androgen levels, whether from adrenal hyperplasia, tumors or exogenous administration may stimulate increased follicle numbers and stromal growth, presumably acting through augmented IGF-I and IGF-I receptor activation. Alternatively, insulin resistance and hyper-insulinism may increase bioavailable IGF-I and testosterone by reducing the hepatic production of IGFBP1 and SHBG, so that increases in circulating free testosterone and free IGF-I may stimulate ovarian growth. Increased follicle numbers and stromal tissue result in increased production of inhibin and androgen, suppressing gonadotropins and impairing cyclic function. This scenario may explain why some women with PCOS respond to gonadotropins with hyperstimulation, since they have increased numbers of small follicles and enhanced FSH responsiveness. It also helps to understand why wedge resection of the ovary may be effective in normalizing cycles in women with this ovarian phenotype, presumably reduction in follicle number and stromal volume reduces excessive inhibin and androgen production, allowing resumption of normal gonadotropin entrainment of the ovary.



*Figure 7.* Potential routes to the polycystic ovary phenotype. Increased circulating insulin secondary to adiposity or "extreme" insulin resistance may have direct effects on the ovary to promote follicular growth and androgen production, and also suppresses hepatic production of SHBG and IGFBP1, resulting in increased free androgen and IGF-I levels in the circulation, which may then act on the ovary to promote follicular growth and androgen production. High circulating androgen derived from adrenal or exogenous sources stimulate the ovary directly, promoting early follicle growth and stromal hypertrophy through enhanced local IGF-I/IGF-I receptor expression. By either pathway, increased numbers of immature follicles produce androgen and inhibin, interfering with gonadotropin dynamics.

#### 2.4 IGFBPs

IGFBPs 2–5 are expressed in diverse patterns in the rat and mouse ovary<sup>11,40,41</sup>. For example, IGFBP2 expression is confined to the ovarian stroma in the rat<sup>11,40</sup> but is expressed by granulosa cells in healthy follicles of mice<sup>11,41</sup>. IGFBP4 is found in atretic follicles in both rat and mouse<sup>40,41</sup>, while IGFBP5 is reportedly localized in atretic follicles in rats<sup>40,42</sup> but not in mice<sup>41</sup>. Based on *in vitro* studies showing that IGFBPs inhibit IGF1's effects, it was thought that follicular IGFBP expression may lead to atresia. However, mice with targeted gene deletions for IGFBPs 2-5 display no obvious reproductive phenotype (43 and unpublished observations, J.E. Pintar). Given the variability in expression patterns and apparent insensitivity to deletion, it seems most likely that IGFBPs play dispensable, modulatory roles in murine ovarian function that are compensated by other IGFBPs or that are too subtle to be easily detected in the knockout models.

In larger animals and humans, there is also considerable variability is ovarian synthesis and circulating levels of IGFBPs. For example, equine granulosa cells produce only IGFBPs 2 and 5<sup>44</sup>, while porcine granulosa produce IGFBPs 2, 3, 4 and 5<sup>45</sup>. Furthermore, physiological regulation of ovarian IGFBP expression also appears to be species specific<sup>22</sup>. IGFBP levels have been intensely investigated in follicular fluid, mainly by ligand binding studies. IGFBP levels generally are reported to decrease in dominant follicles and increase in atretic follicles, although IGFBP4 is found to increase markedly in ovulatory follicles in the sow<sup>46</sup>. In the sow, IGFBP2 mRNA is abundant in granulosa cells of small and preovulatory follicles and less abundant in atretic-appearing follicles (Fig. 3). IGFBP4 is concentrated in the theca interna of healthy, medium to large growing follicles, characterized by IGF-I, aromatase, and gonadotropin receptor expression, but is not detected in attric follicles (Fig. 3). IGFBP4 appears in granulosa cells of peri-ovulatory follicles showing evidence of luteinization and luteinizing hormone (LH) receptor expression, consistent with the increase in IGFBP4 found in follicular IGFBP4 levels<sup>46</sup>. IGFBP4 is also abundant in the sow corpora lutea<sup>20</sup> and most other animals.

In the rhesus monkey, IGFBP4 expression is selective for LH receptorexpressing, steroidogenic thecal and granulosa cells of healthy follicles (Fig. 4) and is stimulated by the LH analog human chorionic gonadoropin (hCG) (Fig. 8). LH/hCG also increases IGFBP4 expression in the bovine and murine ovary<sup>28,41,47</sup>. The observation that IGFBP4 gene expression is selective for healthy follicles is supported by demonstration of thecal and granulosa cell proliferation, LH, and FSH receptor and aromatase expression in these follicles (Fig. 4). IGFBP4 is also expressed by healthy theca interna and corpora lutea in human ovaries<sup>27,28</sup>. Thus LH/hCG stimulates ovarian IGFBP4 expression in rodents, large animals, and primates, and IGFBP4 expression closely parallels LHR expression in healthy developing follicles
and corpora lutea. These observations suggest a role for IGFBP4 in LHinduced steroidogenesis or luteinization, rather than in inhibiting follicular development.



*Figure* 8. hCG increases IGFBP4 gene expression. Representative film autoradiographs comparing LHR, IGFBP4, and PAPP-A mRNA localization in sequential sections from an untreated mid-cycle monkey ovary (**A-C**) and from an hCG-treated mid-cycle monkey ovary (**D-F**). Both are nondominant ovaries with relatively "fresh" corpus lutea from the previous cycle. IGFBP4 and PAPP-A mRNAs are selectively localized in LHR-expressing follicles but little IGFBP4 or PAPP-A mRNA is detected in the corpus luteum (CL) of the untreated monkey ovary (as also seen in Fig. 1C). After hCG treatment, however, IGFBP4 and PAPP-A expression are abundant in the CL, and increased also in the antral follicles (**E and F**). There are a couple of small scars of old CL at the lower pole of the ovary which are also demonstrating heightened IGFBP4 and PAPP-A expression. Bar = 1 mm. From ref. 29.

The monkey ovary is replete with IGFBPs in addition to IGFBP4, although none of the others demonstrates an obvious connection with follicular development or sensitivity to gonadotropin regulation<sup>48</sup>. IGFBP1 is detected in theca-interstitial cells and at low levels in granulosa cells of atretic follicles. IGFBP2 is abundant in the ovarian surface epithelium and in granulosa cells of all antral follicles, including obviously atretic as well as dominant follicles. IGFBP3 is localized in oocytes, in the ovarian vascular endothelium, and the superficial cortical stroma where it is distinctly more

abundant in the nondominant ovary. IGFBP5 is selectively expressed by granulosa cells of mature ovulatory follicles in the monkey ovary<sup>48</sup>. IGFBP5 is also widely expressed in the ovarian stroma, where, in contrast to IGFBP3, it is distinctly more abundant in dominant compared with non-dominant ovary. IGFBP6 mRNA is detected at low levels in the ovary interstitium and theca externa, and in the ovary surface epithelium. These complex, differential patterns of IGFBP expression in the primate ovary certainly suggest that IGFBPs play some role in ovarian function, but this role(s) remain poorly understood.

A recent refinement of the notion that IGFBPs promote follicular atresia involves the idea that selective IGFBP proteolysis may promote formation of ovulatory follicles by increasing "free" or bioavailable IGFs<sup>47,49,50</sup>. This theory is based in large part upon findings of IGFBP fragments and increased free IGF-I in the follicular fluid of dominant follicles of cattle. IGFBP4 has been intensely studied in this context. One enzyme implicated in IGFBP4 proteolysis in follicular fluid is pregnancy-associated protein (PAPP)-A<sup>47,51</sup>. This enzyme displays proteolytic activity against IGFBP4 only in the presence of IGF-I/-II and is present in the circulation of pregnant women bound to a protein known as the proform of the major eosinophil basic protein (proMBP), which inhibits its proteolytic activity. The source and function of this protein complex in the circulation is unclear. The PAPP-A in ovarian follicular fluid is produced by granulosa cells and may be associated with proMBP. A complex series of inhibitory interactions has been proposed to explain the role of IGFBPs, in particular IGFBP4, in folliculogenesis, as illustrated in Fig. 9. IGF activation of its receptor is thought to be inhibited by IGFBP4 until such time as an IGFBP4 protease is unleashed from its inhibitor, allowing binding protein degradation. Although PAPP-A has received much attention, many other proteases and protease inhibitors are found in follicular fluid, particularly around the time of ovulation.

There are some unresolved issues related to this theory. If IGFs are in fact important effectors of follicular selection, then selective regulation of IGF expression would seem to be a more direct, economical mechanism than installing multiple layers of inhibitory factors. In fact, IGF-I expression appears to be selective for healthy follicles in murine species and swine, although not in ruminants or primates. It seems unlikely that IGFBP4's highly selective expression by steroidogenic cells in healthy follicles and corpora lutea and regulation by LH are consistent with a role in follicular demise. Why is IGFBP4 gene expression concentrated in steroidogenic cells of healthy developing follicles and corpora lutea (at least in the sow and rhesus monkey), if its role is to inhibit follicle function? Why would LH stimulate the production of a factor that inhibits steroidogenesis and follicular selection? Finally, IGFBP2 is abundantly expressed by follicles in all stages of development in the sow, mouse, and monkey ovary, not particularly suggestive of an inhibitory role.



*Figure 9.* Illustration of the proposed IGFBP inhibitory network governing follicular IGF activity. The presence of intact IGFBPs, IGFBP4 in this example, is expected to inhibit IGF activation of the IGF-I receptor. IGFBP proteases such as PAPP-A are also present in the ovary and circulation, but may be inhibited by proteins such as proMBP. Follicular selection is proposed to occur when proteolysis of IGFBP liberates IGFs to interact with their receptor.

Most support for the view of IGFBPs as inhibitors of follicular selection is derived from studies of follicular fluid contents; however, interactions between IGFs and their receptor take place at the plasma membrane. In the developing follicle, IGF-I target cells exist in complex, interlocking local microenvironments comprised by the membrane granulosa and theca interna. IGFBPs produced by granulosa and thecal cells are expected to decorate cell membranes and the extracellular matrix within these highly specialized compartments. Thus the scene of action for IGFs produced by granulosa cells or entering the follicle from the blood stream will be in the membrane granulosa or theca interna, where binding proteins may reside to protect them from proteolysis, provide local storage sites, present the peptides to the receptor, or selectively inhibit their interaction with the receptor. It is not clear that the contents of the follicular fluid accurately reflect these interactions between IGF system components and cells of the functional follicular compartments. The increased proteolysis of IGFBPs noted in periovulatory follicles may simply reflect the general increase in protease activity as follicles prepare for rupture and transformation to corpora lutea. Nevertheless, given the convincing correlations between accumulation of free IGF-I and loss of intact IGFBPs in dominant follicle fluid from wellstudied species such as cattle, it seems quite possible that there are important species differences in the mechanisms employed for emergence of ovulatory

follicles. Species that depend more on circulating than locally produced IGF for follicular growth may control emergence of a single dominant follicle through local regulation of IGFBP and protease production.

## 2.5 Summary on IGFs and ovarian function

Our understanding of the IGF role in ovarian function is most advanced in murine species, where IGF-I expression selectively characterizes growing, healthy follicles, and in which IGF-I gene deletion results in critically impaired follicular growth and infertility. While granulosa proliferation, glucose transport, and FSH receptor and aromatse expression are reduced in these mice, the fundamental molecular defects due to loss of IGF-I effect in follicular development remain unknown. Although IGFBPs are abundantly expressed in the murine ovary, targeted deletions of IGFBPs 2-5 have no apparent effect on female fertility, although situational roles, e.g., reproductive adaption to stress, cannot be excluded. In domestic animals and primates, including humans, the story is less clear. The IGF-I receptor is highly expressed in developing follicles and corpora lutea of all species examined, suggesting that IGF-I or -II, whether produced locally in the ovary, or derived from abundant circulating reservoirs, effect ovarian follicular development and luteal function. However, in the absence of IGF-I or IGF-II gene deletions, we do not know if these are critical or simply facilitory effects. Further studies are required to elucidate the role of IGFBPs in ovarian function.

## 3. UTERUS

## 3.1 IGF-I as "estromedin"?

The uterus undergoes dramatic, sex-steroid regulated growth at puberty, during each menstrual cycle, and with pregnancy. Estrogen stimulates the production of IGF-I by uterine stromal cells<sup>52-54</sup>, and the IGF-I receptor is highly expressed by uterine epithelium<sup>54,55</sup>, making local IGF-I a likely candidate as mediator of estrogen's mitogenic effects in the uterus. This hypothesis has recently been examined in the IGF1 gene deletion mouse. The uterus is anatomically normal but extremely small in the IGF-I null mouse, as described in two independent deletions in different strains of mice<sup>16,56</sup>. This uterine hypoplasia could be due to diminished estrogen effect, since, as noted above, IGF1 null ovaries show poor follicle development and deficient estrogen production. Treatment of these mice with exogenous estrogen, however, failed to produce uterine growth<sup>56,57</sup>. Interestingly, DNA

synthesis was increased in the IGF-I null uterus comparable to that in similarly treated wildtype mice (Fig. 10), but cell size was smaller and mitotic figures were fewer in the IGF1 null mice<sup>56</sup>. These observations are especially interesting because IGF-I is usually regarded as a  $G_1$  phase promoter, but uterine cells were arrested in  $G_2$  in the estrogen-treated IGF1 null mice<sup>56</sup>. The retardation in mitoses was correlated with delayed cellular somatic growth in the absence of IGF-I, suggesting that a  $G_2$  checkpoint based on cell size, similar to that seen in yeast, may explain the mitotic delay in the IGF-I null mice.

A recent study investigated whether transplantation of IGF-I null uteri into athymic female hosts could restore normal growth patterns<sup>57</sup>. In effect, the experiment tested whether exposure to circulating IGF-I in the normal host could normalize growth in an IGF1 null uterus. In fact it could, since the IGF-I null uteri reached normal size that matched the host uterine weight and epithelial cell size, 4 weeks after transplantation. Thus it appears that absence of locally produced IGF-I in the IGF-I null uteri did not prevent normal uterine growth in the context of normal circulating IGF-I1 and estrogen in the hosts. In addition, normal wild-type uteri were transplanted into estrogentreated IGF-I null mice, where they demonstrated very little growth, suggesting that local, uterine IGF-I production is not sufficient to allow normal growth in the absence of circulating IGF-I<sup>57</sup>. These interesting observations suggest that circulating rather than locally produced IGF-I may be critical for mediation of estrogen's growth promoting effects upon the uterus in the mouse.

Another recent study approached the elucidation of estrogen-IGF-I interaction in uterine growth from another angle. These experiments treated ovariectomized, estrogen receptor-alpha null (ERKO) and wild-type (WT) mice with IGF-I to determine if IGF-I could stimulate uterine growth in the absence of estrogen effect<sup>58</sup>. IGF-I administration was reported to stimulate proliferation in WT, but not in ERKO uteri, suggesting that IGF-I's effects on uterine growth require ER- $\alpha$ . An issue with these observations, however, is that the degree of uterine proliferation induced by IGF-I treatment of WT mice appears to have been very low, and moreover was not quantified, so that it is unknown whether there were significant differences in IGF-Iinduced proliferation between WT and ERKO uteri. We have not found any effect of IGF-I administration on uterine proliferation in WT or IGF-I null mice, and others have reported that IGF-I does not stimulate proliferation of uterine tissue in the absence of sex steroids<sup>59</sup>. Thus one possible explanation for these observations in different knockout models is that estrogen, acting through ER-a, triggers entry into the cell cycle and progression through Sphase in uterine cells - independent of IGF-I, but that local or systemic IGF-I is required to complete mitosis in a timely manner. The IGF-I requirement



Figure 10. Autoradiographic visualization of E2-induced DNA synthesis in WT (A, C, E) and Igf1 null (B, D, F) uteri. A and C and B and D are paired bright- and dark-field photomicrographs. E and F are higher magnification views showing localization of tritium-positive red nuclei in epithelium (ep), stroma (st) and myometrium (my). Mice were given a single intraperitoneal injection of 17-beta-estradiol (1  $\mu g/g$ ) at 0 h and injected with 3H-thymidine (2  $\mu$ Ci/g) one hour before sacrifice at 21 h. From ref. 56.

may, as a matter of speculation, be due to its anabolic effects on uterine cell growth required for cytokinesis. Previous *in vitro* work in cell lines has shown that IGF-I and other peptide growth factors may activate ER- $\alpha$  by phosphorylation, in the absence of ligand<sup>60</sup>. The recent observations in ERKO mice are taken as evidence that such cross-talk also takes place *in vivo*, *i.e.*, that IGF-I activation of ER- $\alpha$  is required for IGF-I's mitogenic effects<sup>58</sup>. However, as noted above, IGF-I does not have obvious mitogenic effects in the absence of estrogen in WT, ER- $\alpha$  expressing animals.

IGFBPs are also expressed in the murine uterus, but there has not been extensive study of this aspect of the IGF system, and as noted above, mice with targeted gene deletions for all the major IGFBPs have been fertile and without obvious abnormal reproductive phenotype, so will focus on IGFBPs in the primate uterus.

### **3.2** IGF system in the primate uterus

The IGF system (including IGF-I, IGF-II, the IGF-I receptor and IGFBPs) is highly expressed in the human<sup>61-63</sup> and rhesus monkey uterus<sup>54,64,65</sup>. Uterine IGF system expression is found early in fetal development<sup>66,67</sup>, and during maturity is regulated by menstrual cycle stage<sup>62</sup> and exogenous steroids<sup>54,56,65,68</sup>. The IGF system is believed to be important in endometrial growth and remodeling during the menstrual cycle and during implantation and pregnancy<sup>69</sup>. As in the rat and mouse, IGF-I is expressed by the human endometrial stroma and increased in concert with increasing estrogen levels during the menstrual cycle<sup>64</sup>. IGF-II is also present in the human endometrial stroma, but is relatively more abundant in the secretory phase of the cycle. The IGF-I receptor is preferentially expressed in the uterine epithelium<sup>64</sup>, although low level expression is detected in the stroma and myometrium. Several IGFBPs are detected in the human endometrium. IGFBP1 is abundant during the mid-late secretory phase and occurs in abundance in the decidua<sup>68</sup>. IGFBPs 2, 4, 5, and 6 are expressed throughout endometrial stroma<sup>64</sup>. IGFBP3, in contrast, is focally concentrated in endometrial capillaries and is increased in the secretory phase largely due to the intense vascularization of endometrial glands during this phase. IGFBP5 expression is more abundant in the proliferative phase, but all other IGFBP are relatively increased in the secretory phase of the menstrual cycle<sup>64</sup>. These findings support the view that the IGF system plays a fundamental role in endometrial biology, acting via autocrine and/or paracrine mechanisms, with IGF-I and IGFBP5 being dominant in the proliferative phase and IGF-II and the other IGFBPs predominant in the secretory phase of the menstrual cycle.

The IGF system is also highly expressed in the uterine muscle, where IGF overactivity has been implicated in the growth of uterine fibroids<sup>61,68,70</sup>. Studies in the ovariectomized rhesus monkey showed that IGF-I, IGF-II, and

the IGF-I receptor are coexpressed by uterine smooth muscle cells, supporting the possibility of autocrine/paracrine IGF action in stimulating myometrial growth<sup>54</sup>. IGF-I mRNA, which is barely detected in control myometrium, is significantly increased by E2 treatment and augmented even more by the combination of E2 and P4 treatment, with little change noted in IGF-II or IGF-I receptor expression. There is a significant correlation between local IGF-I concentration and local Ki-67-positive myometrial cells (r = 0.891; P = 0.003), implicating local IGF-I in both E2- and P4-induced myometrial growth<sup>54</sup>. IGFBPs 2-5 are all expressed by myometrial smooth muscle cells but each displays distinctive patterns of regulation by sex steroids. IGFBP2 is significantly increased by E2 and even more significantly by E2 and P4. IGFBP4 and 5 are readily detectable in control myometrium and significantly increased by E2 treatment, with no alteration by addition of P4. IGFBP3 is abundant in the control myometrium, but in contrast to other IGFBPs, it is significantly reduced by approximatley 75% by E2 and by E2 and P4 treatment. Interestingly, however, IGFBP3 is increased in the uterine vascular endothelium by both E2 and E2/P4 treatment. IGF-I and IGFBP2 demonstrate similar cellular and sex steroidregulated patterns of myometrial expression. Both are dependent on E2 for significant expression and both are further augmented by the addition of P4 to E2 treatment. These observations indicate the potential for complex local interactions between IGF system components regulated by estrogen and progesterone

## 4. CONCLUSION

Reproduction in mammals is very energy expensive business, at least for females. The energy costs are so high that reproductive competency normally shuts down in times of nutrient scarcity or excessive energy output (e.g., extreme exercise, lactation). IGFs and IGFBPs are strongly regulated by nutritional status, with scarcity leading to suppression of free circulating IGF, as well as reduced IGF production in some tissues. While we usually think of nutritional cues acting at the neuroendocrine level, there is abundant evidence from domestic animals that the systemic IGF-I level is highly and positively correlated with the rate of ovarian follicle growth. Just as reduced energy stores and low insulin and IGF-I levels arrest follicular growth, nutrient excess elevates insulin and IGF levels, resulting in enhanced follicle recruitment and growth, sometimes to excess, as in the PCOS.

While local sources of IGF peptides and local distribution of IGFBPs in the ovary and uterus vary considerably between species, IGF-I receptor expression is quite constant. This receptor is most highly expressed in growing follicles and in corpora lutea, in parallel with the level of metabolic activity in these structures, as reflected in glucose utilization. So whatever the source of the IGF ligand, and however the ligands are affected by the IGFBPs, the ultimate effect of IGF-I receptor activation is to promote anabolic processes required to support rapid cellular growth (follicle) and steroid biosynthesis (corpora lutea). IGF-I receptor expression is also highly abundant in the endometrial epithelium, which is mitotically and biosynthetically extremely active, undergoing rapid cycles of growth and differentiation and requiring abundant supplies of fuel. Somewhat beyond the scope of the present chapter, IGF-II serves a similar role in placental and fetal development, controlling the placental supply of nutrients available for fetal growth, irrespective of the maternal nutritional state. Thus, the IGF family appears to play an important role in nutrient partitioning, supporting the extraordinary, energy-expensive processes of follicular growth, steroidogenesis, and uterine hyperplasia and remodeling and fetal growth.

#### REFERENCES

- 1. S.M. Firth and R.C. Baxter, Cellular actions of the insulin-like growth factor binding proteins. *Endocr Rev*, 23, 824-854 (2002).
- 2. C.A. Bondy, E. Chin and J. Zhou, Significant species differences in local IGF-I and -II gene expression. *Adv Exp Med Biol*, **343**, 73-77 (1993).
- 3. M. Tatar, A. Bartke and A. Antebi, The endocrine regulation of aging by insulin-like signals. *Science*, **299**, 1346-1351 (2003).
- 4. E.Y. Adashi, The IGF family and folliculogenesis. J Reprod Immunol, 39,13-19 (1998).
- D. Zaczek, J. Hammond, L. Suen, S. Wandji, D. Service, A. Bartke, V. Chandrashekar, K. Coschigano and J. Kopchick, Impact of growth hormone resistance on female reproductive function: new insights from GH receptor knockout mice. *Biol Reprod*, 67, 1115-1124 (2002).
- A. Bachelot, P. Monget, P. Imbert-Bollore, K. Coshigano, J.J. Kopchick, P.A. Kelly and N. Binart, Growth hormone is required for ovarian follicular growth. *Endocrinology*, 143, 4104-4112 (2002).
- 7. J. Zhou, E. Chin and C. Bondy, Cellular pattern of insulin-like growth factor-I (IGF-I) and IGF-I receptor gene expression in the developing and mature ovarian follicle. *Endocrinology*, **129**, 3281-3288 (1991).
- 8. F.I. Sharara and L.K. Nieman, Identification and cellular localization of growth hormone receptor gene expression in the human ovary. *J Clin Endocrinol Metab*, **79**, 670-672 (1994).
- 9. B. Carlsson, A. Nilsson, O.G. Isaksson and H. Billig, Growth hormone-receptor messenger RNA in the rat ovary: regulation and localization. *Mol Cell Endocrinol*, **95**, 59-66 (1993).
- 10. S. Franks, Growth hormone and ovarian function. *Baillieres Clin Endocrinol Metab*, **12**, 331-340 (1998).
- 11. E.Y. Adashi, C.E. Resnick, D.W. Payne, R.G. Rosenfeld, T. Matsumoto, M.K. Hunter, S.E. Gargosky, J. Zhou and C.A. Bondy, The mouse intraovarian insulin-like growth

factor I system: departures from the rat paradigm. *Endocrinology*, **138**, 3881-3890 (1997).

- J.E. Oliver, T.J. Aitman, J.F. Powell, C.A. Wilson and R.N. Clayton, IGF I gene expression in the rat ovary is confined to the granulosa cells of developing follicles. *Endocrinology*, **124**, 2671-2679 (1989).
- 13. J. Zhou, T.R. Kumar, M.M. Matzuk and C. Bondy, Insulin-like growth factor I regulates gonadotropin responsiveness in the murine ovary. *Mol Endocrinol*, **11**, 1924-1933 (1997).
- J. Zhou, J. Refuerzo and C. Bondy, Granulosa cell DNA synthesis is strictly correlated with the presence of IGF I and absence of c-fos/c-jun expression. *Mol Endocrinol*, 9, 924-931 (1995).
- 15. P. Monget, and C. Bondy, Importance of the IGF system in early folliculogenesis. *Mol Cell Endocrinol* 163:89-93 (2000).
- J. Baker, M.P. Hardy, J. Zhou, C. Bondy, F. Lupu, A.R. Bellve and A. Efstratiadis, Effects of an Igf1 gene null mutation on mouse reproduction. *Mol Endocrinol*, 10, 903-918 (1996).
- 17. J. Zhou, M. Bievre and C.A. Bondy, Reduced GLUT1 expression in Igf1-/- null oocytes and follicles. *Growth Horm IGF Res*, 10, 111-117 (2000).
- J.S. Richards, D.L. Russell, S. Ochsner, M. Hsieh, K.H. Doyle, A.E. Falender, Y.K. Lo and S.C. Sharma, Novel Signaling Pathways That Control Ovarian Follicular Development, Ovulation, and Luteinization. *Recent Prog Horm Res*, 57, 195-220 (2002).
- S. Mazerbourg, C. Bondy, J. Zhou and P. Monget, The Insulin-like Growth Factor System: a Key Determinant Role in the Growth and Selection of Ovarian Follicles? A Comparative Species Study. *Reprod Domest Anim*, 38, 247-258 (2003).
- J. Zhou, O.O. Adesanya, G. Vatzias, J.M. Hammond and C.A. Bondy, Selective expression of IGF system components during porcine ovary follicular selection. *Endocrinology*, 137, 4893-4901 (1996).
- 21. O.J. Ginther, M.A. Beg, D.R. Bergfelt, F.X. Donadeu and K. Kot, Follicle selection in monovular species. *Biol Reprod*, **65**, 638-647 (2001).
- 22. L.J. Spicer, Proteolytic Degradation of Insulin-Like Growth Factor Binding Proteins by Ovarian Follicles: A Control Mechanism for Selection of Dominant Follicles. *Biol Reprod* (2003).
- M.C. Lucy, C.R. Bilby, C.J. Kirby, W. Yuan and C.K. Boyd, Role of growth hormone in development and maintenance of follicles and corpora lutea. *J Reprod Fertil Suppl*, 54, 49-59 (1999).
- 24. D.G. Armstrong, J.G. Gong and R. Webb, Interactions between nutrition and ovarian activity in cattle: physiological, cellular and molecular mechanisms. *Reprod Suppl*, **61**, 403-414 (2003).
- W.S. Cohick, J.D. Armstrong, M.D. Whitacre, M.C. Lucy, R.W. Harvey and R.M. Campbell, Ovarian expression of insulin-like growth factor-I (IGF-I), IGF binding proteins, and growth hormone (GH) receptor in heifers actively immunized against GHreleasing factors. *Endocrinology*, 137, 1670-1677 (1996).
- 26. V.C. Zulu, T. Nakao and Y. Sawamukai, Insulin-like growth factor-I as a possible hormonal mediator of nutritional regulation of reproduction in cattle. *J Vet Med Sci*, **64**, 657-665 (2002).
- A. el-Roeiy, X. Chen, V.J. Roberts, S. Shimasakai, N. Ling, D. LeRoith, C.T. Roberts Jr. and S.S. Yen, Expression of the genes encoding the insulin-like growth factors (IGF-I and II), the IGF and insulin receptors, and IGF-binding proteins-1-6 and the localization of their gene products in normal and polycystic ovary syndrome ovaries. *J Clin Endocrinol Metab*, **78**, 1488-1496 (1994).

- 28. J. Zhou and C. Bondy, Anatomy of the human ovarian insulin-like growth factor system. *Biol Reprod*, 48, 467-482 (1993).
- J. Zhou, J. Wang, D. Penny, P. Monget, J.A. Arraztoa, L.J. Fogelson and C.A. Bondy, Insulin-like growth factor binding protein 4 expression parallels luteinizing hormone receptor expression and follicular luteinization in the primate ovary. *Biol Reprod*, 69, 22-29 (2003).
- K. Vendola, J. Zhou, J. Wang and C.A. Bondy, Androgens promote insulin-like growth factor-I and IGF-I receptor gene expression in the primate ovary. *Hum Reprod*, 14, 2328-2332 (1999).
- K.A. Vendola, J. Zhou, O.O. Adesanya, S.J. Weil and C.A. Bondy, Androgens stimulate early stages of follicular growth in the primate ovary. *J Clin Invest*, **101**, 2622-2629 (1998).
- K. Vendola, J. Zhou, J. Wang, O.A., Famuyiwa, M. Bievre and C.A. Bondy, Androgens promote oocyte insulin-like growth factor I expression and initiation of follicle development in the primate ovary. *Biol Reprod*, 61, 353-357 (1999).
- S. Weil, K. Vendola, J. Zhou and C.A. Bondy, Androgen and follicle-stimulating hormone interactions in primate ovarian follicle development. *J Clin Endocrinol Metab*, 84, 2951-2956 (1999).
- Y. Yoshimura, M. Ando, S. Nagamatsu, M. Iwashita, T. Adachi, K. Sueoka, T. Miyazaki, N. Kuji and M.Tanaka, Effects of insulin-like growth factor-I on follicle growth, oocyte maturation, and ovarian steroidogenesis and plasminogen activator activity in the rabbit. *Biol Reprod*, 55, 152-160 (1996).
- 35. F.J. Thomson, T.J. Jess, C. Moyes, R. Plevin and G.W. Gould, Characterization of the intracellular signalling pathways that underlie growth-factor-stimulated glucose transport in Xenopus oocytes: evidence for ras- and rho-dependent pathways of phosphatidylinositol 3-kinase activation. *Biochem J*, **325**, 637-643 (1997).
- 36. P.E. Hughesdon, Morphology and morphogenesis of the Stein-Leventhal ovary and of socalled "hyperthecosis". *Obstet Gynecol Surv*, **37**, 59-77 (1982.)
- 37. W. Futterweit and L. Deligdisch, Histopathological effects of exogenously administered testosterone in 19 female to male transsexuals. *J Clin Endocrinol Metab*, **62**, 16-21 (1986).
- T.D. Pache, S. Chadha, L.J. Gooren, W.C. Hop, K.W. Jaarsma, H.B. Dommerholt and B.C. Fauser, Ovarian morphology in long-term androgen-treated female to male transsexuals. A human model for the study of polycystic ovarian syndrome? *Histopathology*, **19**, 445-452 (1991).
- V. De Leo, D. Lanzetta, D. D'Antona, A. la Marca and G. Morgante, Hormonal effects of flutamide in young women with polycystic ovary syndrome. *J Clin Endocrinol Metab*, 83, 99-102 (1998).
- A. Nakatani, S. Shimasaki, G.F. Erickson and N. Ling, Tissue-specific expression of four insulin-like growth factor-binding proteins (1, 2, 3, and 4) in the rat ovary. *Endocrinology*, **129**, 1521-1529 (1991).
- S.A. Wandji, T.L. Wood, J. Crawford, S.W. Levison and J.M. Hammond, Expression of mouse ovarian insulin growth, factor system components during follicular development and atresia. *Endocrinology* 139:5205-5214 (1998).
- G.F. Erickson, A. Nakatani, N. Ling and S. Shimasaki, Localization of insulin-like growth factor-binding protein-5 messenger ribonucleic acid in rat ovaries during the estrous cycle. *Endocrinology*, **30**, 1867-1878 (1992).
- T.L. Wood, L.E. Rogler, M.E. Czick, A.G. Schuller and J.E. Pintar, Selective alterations in organ sizes in mice with a targeted disruption of the insulin-like growth factor binding protein-2 gene. *Mol Endocrinol*, 14, 1472-1482 (2000).

- T.R. Davidson, C.S. Chamberlain, T.S. Bridges and L.J. Spicer, Effect of follicle size on in vitro production of steroids and insulin-like growth factor (IGF)-I, IGF-II, and the IGFbinding proteins by equine ovarian granulosa cells. *Biol Reprod*, 66, 1640-1648 (2002).
- 45. J.K. Leighton, R.W. Grimes, S.F. Canning and J.M. Hammond, IGF-binding proteins are differentially regulated in an ovarian granulosa cell line. *Mol Cell Endocrinol*, **106**, 75-80 (1994).
- 46. J. Liu, A.T. Koenigsfeld, T.C. Cantley, C.K. Boyd, Y. Kobayashi and M.C. Lucy, Growth and the initiation of steroidogenesis in porcine follicles are associated with unique patterns of gene expression for individual components of the ovarian insulin-like growth factor system. *Biol Reprod*, **63**, 942-952 (2000).
- 47. S. Mazerbourg, M.T. Overgaard, C. Oxvig, M. Christiansen, C.A. Conover, I. Laurendeau, M. Vidaud, G. Tosser-Klopp, J. Zapf and P. Monget, Pregnancy-associated plasma protein-A (PAPP-A) in ovine, bovine, porcine, and equine ovarian follicles: involvement in IGF binding protein-4 proteolytic degradation and mRNA expression during follicular development. *Endocrinology*, **142**, 5243-5253 (2001).
- J.A. Arraztoa, P. Monget, C. Bondy and J. Zhou, Expression patterns of insulin-like growth factor-binding proteins 1, 2, 3, 5, and 6 in the mid-cycle monkey ovary. J Clin Endocrinol Metab, 87, 5220-5228 (2002).
- 49. L.C. Giudice, Insulin-like growth factor family in Graafian follicle development and function. *J Soc Gynecol Investig*, **8**, S26-29 (2001).
- 50. P. Monget, N. Besnard, C. Huet, C. Pisselet and D. Monniaux, Insulin-like growth factorbinding proteins and ovarian folliculogenesis. *Horm Res*, **45**, 211-217 (1996).
- C.A. Conover, C. Oxvig, M.T. Overgaard, M. Christiansen and L.C. Giudice, Evidence that the insulin-like growth factor binding protein-4 protease in human ovarian follicular fluid is pregnancy associated plasma protein-A. *J Clin Endocrinol Metab*, 84, 4742-4745 (1999).
- 52. L.J. Murphy, L.C. Murphy and H.G. Friesen, Estrogen induces insulin-like growth factor-I expression in the rat uterus. *Mol Endocrinol*, **1**, 445-450 (1987).
- 53. G. Norstedt, A. Levinovitz and H. Eriksson, Regulation of uterine insulin-like growth factor I mRNA and insulin-like growth factor II mRNA by estrogen in the rat. *Acta Endocrinol (Copenh)*, **120**, 466-472 (1989).
- O.O. Adesanya, J. Zhou and C.A. Bondy, Sex steroid regulation of insulin-like growth factor system gene expression and proliferation in primate myometrium. J Clin Endocrinol Metab, 81, 1967-1974 (1996).
- 55. K.R. Stevenson, R.S. Gilmour and D.C. Wathes, Localization of insulin-like growth factor-I (IGF-I) and -II messenger ribonucleic acid and type 1 IGF receptors in the ovine uterus during the estrous cycle and early pregnancy. *Endocrinology*, **134**, 1655-1664 (1994).
- O.O. Adesanya, J. Zhou, C. Samathanam, L. Powell-Braxton and C.A. Bondy, Insulinlike growth factor 1 is required for G2 progression in the estradiol-induced mitotic cycle. *Proc Natl Acad Sci U S A*, 96, 3287-3291 (1999).
- T. Sato, G. Wang, M.P. Hardy, T. Kurita, G.R. Cunha and P.S. Cooke, Role of Systemic and Local IGF-I in the Effects of Estrogen on Growth and Epithelial Proliferation of Mouse Uterus. *Endocrinology*, 143, 2673-2679 (2002).
- D.M. Klotz, S.C. Hewitt, P. Ciana, M. Raviscioni, J.K. Lindzey, J. Foley, A. Maggi, R.P. DiAugustine and K.S. Korach, Requirement of Estrogen Receptor-alpha in Insulin-like Growth Factor-1 (IGF-1)-induced Uterine Responses and in Vivo Evidence for IGF-1/Estrogen Receptor Cross-talk. *J Biol Chem*, 277, 8531-8537 (2002).
- 59. A. Ghahary, S. Chakrabarti and L.J. Murphy, Localization of the sites of synthesis and action of insulin-like growth factor-I in the rat uterus. *Mol Endocrinol*, 4, 191-195 (1990).

- 60. C.L. Smith, Cross-talk between peptide growth factor and estrogen receptor signaling pathways. *Biol Reprod*, **58**, 627-632 (1998).
- 61. B.J. Vollenhoven, A.C. Herington and D.L. Healy, Messenger ribonucleic acid expression of the IGFs and their binding proteins in uterine fibroids and myometrium. J Clin Endocrinol Metab, 76, 1106-1110 (1993).
- 62. L.C. Giudice, D.A. Milkowski, G. Lamson, R.G. Rosenfeld and J.C. Irwin, Insulin-like growth factor binding proteins in human endometrium: steroid-dependent messenger ribonucleic acid expression and protein synthesis. *J Clin Endocrinol Metab*, **72**, 779-787 (1991).
- X.M. Tang, M.J. Rossi, B.J. Masterson and N. Chegini, Insulin-like growth factor I (IGF-I), IGF-I receptors, and IGF binding proteins 1-4 in human uterine tissue: tissue localization and IGF-I action in endometrial stromal and myometrial smooth muscle cells in vitro. *Biol Reprod*, 50, 1113-1125 (1994).
- 64. J. Zhou, B.A. Dsupin, L.C. Giudice and C.A. Bondy, Insulin-like growth factor system gene expression in human endometrium during the menstrual cycle. *J Clin Endocrinol Metab*, **79**, 1723-1734 (1994).
- 65. O.O. Adesanya, J. Zhou and C.A. Bondy, Cellular localization and sex steroid regulation of insulin-like growth factor binding protein messenger ribonucleic acids in the primate myometrium. *J Clin Endocrinol Metab*, **81**, 2495-2501 (1996).
- 66. J.L. Shifren, R. Osathanondh and J. Yeh, Human fetal ovaries and uteri: developmental expression of genes encoding the insulin, insulin-like growth factor I, and insulin-like growth factor II receptors. *Fertil Steril*, **59**, 1036-1040 (1993).
- 67. M.P. Thiet, R. Osathanondh and J. Yeh, Localization and timing of appearance of insulin, insulin-like growth factor-I, and their receptors in the human fetal mullerian tract. *Am J Obstet Gynecol*, **170**, 152-156 (1994).
- E.M. Rutanen, A. Salmi and T. Nyman, mRNA expression of insulin-like growth factor-I (IGF-I) is suppressed and those of IGF-II and IGF-binding protein-1 are constantly expressed in the endometrium during use of an intrauterine levonorgestrel system. *Mol Hum Reprod*, 3, 749-754 (1997).
- 69. N.R. Nayak and L.C. Giudice, Comparative biology of the IGF system in endometrium, decidua, and placenta, and clinical implications for foetal growth and implantation disorders. *Placenta*, 24, 281-296 (2003).
- T.H. Vu, C. Yballe, S. Boonyanit and A.R. Hoffman, Insulin-like growth factor II in uterine smooth-muscle tumors: maintenance of genomic imprinting in leiomyomata and loss of imprinting in leiomyosarcomata. *J Clin Endocrinol Metab*, 80, 1670-1676 (1995).

## Chapter 5

## SKELETAL MUSCLE

Ana M. Fernandez<sup>1</sup> and Derek LeRoith<sup>2</sup>

<sup>1</sup>Unidad de Plasticidad Neural, Instituto Cajal, CSIC, Madrid, <sup>2</sup>Diabetes Branch, NIDDK, National Institutes of Health, Maryland, USA

Key words: Myoblast; myotube; myofiber; regeneration; targeted knockout.

### **1. INTRODUCTION**

Somatic growth is determined by several control mechanisms that integrate environmental signals with neuroendocrine cues. Growth factors are produced by many tissues and their effects are mainly local (autocrine/paracrine), whereas hormones exert their actions systemically, often far away from their tissue of origin. The growth hormone (GH)/insulin-like growth factor-I (IGF-I) system is essential for normal growth and development during embryonic and postnatal periods. Although GH is important for postnatal growth, IGF-I has been found to be critical for both stages of development. The original somatomedin hypothesis was initially postulated to demonstrate that GH stimulates growth through a factor produced by the liver (somatomedin C, called later IGF-I)<sup>1</sup>. Later, it was demonstrated that multiple tissues from the fetal mouse synthesized IGF-I<sup>2,3</sup> and that GH stimulated local production of IGF-I in bone<sup>4,5</sup>. This raised the importance of locally produced IGF-I in mediating the growth effects of GH. A revision of the original somatomedin hypothesis came with the results of two independent mouse models of liver IGF-I disruption <sup>6,7</sup>. Target deletion of liver IGF-I synthesis produced normal growth in mice leading to the suggestion that autocrine/paracrine IGF-I may be quite important in mediating postnatal somatic growth. More recent studies have, however, demonstrated that endocrine IGF-I is of major importance<sup>8</sup>.

Taking together the extensive literature, it appears that whole body growth may be the sum of endocrine and autocrine/paracrine actions of the GH/IGF-I axis. Thus, the endocrine function of somatic growth may correspond to the overlapping activity of GH and liver IGF-I, and possibly GH-dependent IGF-I production in a few extrahepatic tissues. Paracrine/autocrine effects may represent the GH-independent role of IGF-I in the extrahepatic tissues.

During the development of skeletal muscle, myoblasts cease to proliferate and differentiate into myotubes by activating a number of musclespecific genes and are further remodeled into myofibers. In adult muscle a class of cells with embryonic characteristics named satellite cells remains. These cells are responsible for the regenerating capacity of the adult skeletal muscle after injury<sup>9</sup>. Myofibers are grouped into fiber types, type I (slow-twitch fibers) and type II, with subgroups IIA, IIX, and IIB (fast-twitch fibers). The different types of myofibers exhibit distinctly different contractile and metabolic properties. Type I and type IIA demonstrate oxidative metabolism whereas type IIB fibers are more glycolytic<sup>10</sup>.

The IGF system plays an essential role in the formation and maintenance of skeletal muscle. IGF-I and IGF-II and their cognate tyrosine kinase receptor (IGF-I receptor) have been shown to elicit major growth-promoting effects<sup>11</sup>. While IGF-II is essential for normal embryonic growth <sup>12</sup>, IGF-I operates throughout prenatal and postnatal development<sup>13</sup>. Unlike other growth factors, IGF-I stimulates both myoblast proliferation and differentiation<sup>14,15</sup>. In proliferating myoblasts, IGF-I increases the expression of cell cycle progression factors<sup>14</sup>. After withdrawal of myoblasts from the cell cycle, IGF-I promotes muscle differentiation by inducing the expression or activity of myogenic regulatory factors<sup>16</sup>.

There are still controversial points of view on how GH regulates growth in skeletal muscle *in vivo*. The relatively scarce literature on this issue suggests that GH apparently has no direct growth-promoting effects in the skeletal muscle. In this chapter we summarize the overall literature of the GH/IGF-I system in the development of the skeletal muscle and its implications in aging, regeneration, diabetes, and myopathies.

### 2. THE GH SYSTEM

### 2.1 Growth hormone

GH and the GH receptor (GHR) belong to the superfamily of cytokine peptides<sup>17</sup> and receptors<sup>18</sup>. Somatotroph cells of the adenohypohysis

synthesize and store GH. Rodents and humans show a sexual dimorphism in the pulsatile secretion of  $GH^{19}$ . GH secretion is tightly regulated by two hypothalamic factors, the stimulating GH releasing hormone (GHRH)<sup>20,21</sup> and the inhibitory hormone, somatostatin (SS)<sup>22</sup>. An inhibitory feedback loop on GH secretion is provided by circulating IGF-I that regulates GH actions in peripheral tissues. Other metabolic factors also regulate GH release; free fatty acids (FFAs)<sup>23</sup> inhibiting it, whereas leptin, neuropeptide Y (NPY), hexarelin and ghrelin have been shown to stimulate GH release<sup>24</sup>.

### 2.2 The GHR and GHR signal transduction

The GHR was the first member of the type I cytokine receptor family to be cloned. GHR is a single-transmembrane receptor that mediates the actions of GH. It is now well established that binding of one molecule to two adjacent receptors is required for optimal activity<sup>30,31</sup>. GHR is present on the surface of most cells<sup>18</sup>. GH-binding protein (GHBP) is formed either by proteolytic cleavage in humans and rabbits<sup>32-34</sup> or alternative splicing in mice and rats<sup>35,36</sup> of the GHR extracellular ligand binding domain. GHBP stabilizes GH in plasma acting as a physiological buffer<sup>37</sup>.

As with most cytokine receptors, GHR utilizes the JAK-STAT signal transduction pathway<sup>38</sup>. Upon GH binding, the GHR associates with the tyrosine kinase janus kinase 2 (JAK2). JAK2 phosphorylates STATs -1, -3, -5a, and -5b on tyrosine residues<sup>38</sup> allowing their translocation to the nucleus where they activate gene transcription upon binding to DNA sequences<sup>39,40</sup> (Fig. 1).

Gene expression of IGF-I and various IGF-binding proteins (IGFBPs) is regulated by GHR stimulation although a GH-responsive element has been found only in the promoter for the acid labile subunit (ALS) of the IGFBP-3 complex<sup>41</sup>.

It has been proposed that there may be cross-talk between GH and IGF-I at the level of their signal transduction pathways. Upon GHR activation, JAK2 induces tyrosine phosphorylation of the insulin receptor substrate and their association  $(\mathbb{IRS})$ proteins IRS-1 IRS-2 and with phosphatidylinositol-3'-kinase (PI3K)<sup>42,43</sup> (Fig. 1). As mentioned later, IRS proteins are tyrosine phosphorylated by the activation of the IGF-I receptor (as well as by the insulin receptor). Effects of GH on insulin receptor-related metabolic signal transduction have been described<sup>42,43</sup>. However, there are no studies relating GH-induced IRS tyrosylphosphorylation to IGF-Imediated mitogenic and metabolic effects.



*Figure 1.* Signal transduction pathways used by GH and convergence between IGF-I and GH signaling. GH-induced dimerization of GHR drives activation of JAK2 and subsequent phosphorylation of STAT proteins which translocate to the nucleus to activate gene transcription. Crosstalk (dotted lines) between IGF-I and GH signaling is thought to be mediated by JAK2 tyrosyl-phosphorylation of SHC and IRS proteins leading, in turn, to activation of MAPK and PI3K pathways involved in growth and metabolism actions, respectively.

### 2.3 GHR in skeletal muscle

Although reports of direct effects of GH on muscle are rare, *in situ* hybridization studies in rats showed that both the GHR and GHBP genes are expressed in a number of tissues, including skeletal muscle<sup>44</sup>. At least five variant forms of the GHR mRNA were found in rats with the V3 variant being found in skeletal muscle. In human skeletal muscle, GHR transcripts are at low and variable levels among individuals<sup>45</sup>. Although there are very few reports of GH-GHR direct effects, expression of the GHR in many cell types has led to the suggestion that they might all be responsive to GH. However, there has been a great degree of difficulty in demonstrating GHR binding, particularly in skeletal muscle<sup>46</sup>.

Expression of the GHR and GHBP are apparently under both tissuespecific and developmental controls. For example, there is a developmental switch from a nonbinding variant of the GHR in the embryo to a functioning form shortly before birth in sheep and rats<sup>47,48</sup>. This was in agreement with the general view that GH has little or no effect on fetal growth (see later). During human development, GHR and GHBP mRNAs are present at low levels in fetal liver, increasing considerably after birth<sup>44</sup>. It was concluded that levels of liver GHR mRNA probably correlate with the influence of GH on growth of the animals. These levels were positively correlated with plasma IGF-I and growth rate. On the contrary, muscle GHR mRNA levels correlate with metabolic control<sup>49</sup>. Based on these studies, it appears that the GHR gene is expressed in skeletal muscle, but whether it might stimulate myogenesis either directly or by inducing autocrine expression of the IGFs remains to be elucidated.

## 2.4 The physiological role of GH on skeletal muscle

### 2.4.1 Effects of GH on muscle growth

GH is a pleitropic hormone with physiological actions that involve multiple organs and physiological systems<sup>50</sup>. Studies on the effects of GH in skeletal muscle have been performed in a variety of models, from cultured cells to intact and hypophysectomized animals and in humans. Hypophysectomy in young animals causes a substantial decrease in fast type 1 fibers in both soleus (primarily slow) and extensor digitorum longus (primarily fast) muscles, and treatment with GH reverses this effect<sup>51</sup>. Administration of GH to hypophysectomized animals causes an increase in muscle mass<sup>52</sup>, although GH has no effect in intact young rats. Treatment of intact rats with a polyclonal antiserum to rat GH substantially reduced the weight, total protein and RNA content of hind limb muscles<sup>53</sup>, but probably because there was a marked decrease in serum IGF-I levels.

It is well known that adults with GH deficiency have a generalized reduction in skeletal muscle mass<sup>54,55</sup>. Treatment of such patients with recombinant human GH (rhGH) increases lean tissue and skeletal muscle mass at all ages<sup>56,57</sup>, although long-term exposure produces potential drawbacks<sup>58</sup>. However, other studies have reported beneficial effects of extended GH treatment on the musculature of GH-deficient humans<sup>59,60</sup>. Taken together these studies suggest that GH has the same effects in both human and animal muscles, but they do not prove that these effects do or do not involve elevation of circulating or muscle-expressed IGFs. Indeed, GH stimulates both circulating and local levels of IGF-I in various tissues<sup>50</sup>. Hence, whether the effects of GH on skeletal muscle growth are direct or IGF-I dependent remains to be elucidated.

Evidence for the synthesis of GH in a number of extrapituitary sites suggests that GH may have local paracrine/autocrine effects that might be distinct from the known classic circulating IGF-I-dependent effects. These local effects may be mediated either by local production of IGF-I or by other additional factors<sup>61</sup>.

### 2.4.2 Effects of GH on muscle metabolism

GH is primarily an anabolic hormone and in skeletal muscle it induces positive nitrogen balance and protein synthesis<sup>62</sup>. Treatment with GH reduces fat mass and increases lean body weight by enhancing protein synthesis, with no effect on protein degradation<sup>63,64</sup>. In elderly patients, rhGH treatment had similar effects. Infusion of GH, IGF-I, and insulin into the forearm showed that GH and IGF-I increased amino acid uptake, whereas IGF-I and insulin, but not GH, inhibited amino acid release. Moreover, IGF-I is capable of inhibiting proteolysis<sup>65</sup>. Although there is no consensus as to whether IGF-I may or may not mediate the effects of GH on nitrogen balance and protein synthesis in muscle<sup>66</sup>, these studies strongly support the suggestion that many of the anabolic effects of GH in muscle may be IGF-I dependent.

GH has a lipolytic action on fat and muscle. Acute GH administration causes a rise in the levels of circulating FFAs and glycerol by inhibition of lipoprotein lipase<sup>67</sup>. Long-term effects of GH include decreased deposition of fat and increased fat mobilization. The acute insulin-like activity of GH on carbohydrate metabolism has been seen both in vitro and in vivo and is insulin<sup>68</sup>. IGF-I and GH-induced tyrosine independent of both phosphorylation of IRS-1 and/or IRS-2 may be responsible for this effect<sup>69</sup>. Prolonged GH stimulation also enhances hepatic gluconeogenesis and glycogenolysis, resulting ultimately in hyperglycemia. This inhibitory effect on insulin activity may be indirectly caused by the GH-induced lipolysis and elevated plasma FFA levels that inhibit insulin activity at its target tissues $^{70}$ .

### 2.4.3 GH actions on skeletal muscle not mediated by IGF-I

Fetal growth is not dependent on GH, as demonstrated by the fact that children with congenital absence of the pituitary and GH or GHR deletions are born near normal in size. However, postnatal growth and development are dependent on normal pulsatile secretion of GH<sup>71</sup>, an effect that is especially prominent during puberty. There is extensive evidence for the effects of IGFs on isolated skeletal muscle cells, and a general lack of such

evidence for GH. Although there is evidence for a growth-promoting effect of GH that is independent of IGF-I in some tissues such as bone<sup>72</sup>, as to date there is no evidence of such effect in skeletal muscle development.

Most studies of GH actions on isolated muscle cells have given negative results<sup>46</sup>. A report of a direct effect of GH on muscle cells involved BC3HI cells. These cells exhibited metabolic responses to GH (*i.e.*, the insulin-like stimulation of glucose metabolism), but the authors did not show any effect on either growth or myogenesis in these cells<sup>73</sup>. Therefore, there are no reports establishing a direct effect of GH in stimulating the growth or differentiation of muscle cells. A clinical observation suggesting a direct GH effect on muscle is that local infusion of GH into the forearm of humans synthesis<sup>74</sup> and stimulated protein myosin heavy chain mRNA accumulation<sup>75</sup>. These studies do not exclude the possibility that GH may be acting by inducing expression of IGF-I in the musculature.

A more convincing observation for a direct effect of GH on skeletal muscle is a study showing that tyrosine phosphorylation of JAK2 and STAT5 occurs in skeletal muscle and liver of rats after intravenous administration of  $GH^{76}$ . Although this is a more convincing report for some direct effects of GH on skeletal muscle, again it does not demonstrate that an effect on growth or myogenesis is or is not mediated by induction of IGF-I expression in skeletal muscle.

# 2.4.4 Effects of GH on expression of the IGF-I gene in muscle

The role of IGF gene expression in mediating the effects of GH on skeletal muscle growth is not clear. It has been known for more than a decade that cultured myoblasts release IGFs<sup>77</sup>, but the relationship of this effect to GH treatment is ambiguous. There are a number of reports of GH-independent expression of IGF-I mRNA in skeletal muscle. Levels of IGF-I mRNA in extrahepatic tissues in the GH-deficient dwarf chicken are close to normal<sup>78</sup>. In most studies there was a similar lack of correlation between GH levels, growth, and muscle IGF-I mRNA levels<sup>79-81</sup>. However, in exercized rats, the anabolic response to training involved both GH-dependent increases in liver IGF-I mRNA and GH-independent increases in muscle IGF-I mRNA. This study also reported that the GH dependence/independence was related to different spliced variants of IGF-I mRNA<sup>81</sup> (see later). However, in one study, it was reported that GH does affect IGF-I mRNA gene expression in muscle. GH treatment of hypophysectomized rats caused an increase in IGF-I mRNA content in the gastrocnemius muscle<sup>82,83</sup>.

It can be concluded that IGF-I gene expression can occur in muscle with or without GH stimulation. The possibility remains that at least some of the reported direct effects of GH might be mediated by autocrine/paracrine actions of IGF-I, although it has not been easy to demonstrate functioning GH receptors on skeletal muscle cells (see earlier).

# 2.4.5 Effects of GH on circulating levels of IGFs and IGFBPs

The stimulation by GH on liver secretion of IGF-I has been widely reported in a number of reviews and it is well established<sup>84,85</sup>. There is a general consensus that the liver is the primary source of circulating IGF-I (and IGFBP-3), but there is considerable uncertainty about the extent to which circulating IGF-I (in contrast to autocrine/paracrine secretions) affects specific tissues in specific situations.

### **3.** THE IGF SYSTEM

The IGF family of growth factors is composed of three ligands: IGF-I, IGF-II, and insulin; their cognate receptors, IGF-I receptor, insulin receptor, and the IGF-II mannose-6-phosphate receptor (IGF-II/M-6-P receptor); and six high-affinity binding proteins (IGFBPs)<sup>11,86</sup>. Gene targeting studies revealed that the IGF-I receptor mediates the cellular responses of IGF-I and IGF-II. The insulin receptor mediates the biological (metabolic) functions of insulin (and some of the mitogenic actions of IGF-II in the embryo)<sup>87</sup>. A subset of alternately spliced insulin receptors with increased affinity for IGF-II has been described<sup>88</sup>. In contrast, the IGF-II/M-6-P receptor has no major role in IGF signal transduction, but is primarily responsible for clearing the extracellular IGF-II during fetal development<sup>89</sup>. The IGF-II/M-6-P receptor also plays a major role in transporting lysosomal enzymes between intracellular compartments.

The IGF-I receptor is a member of the tyrosine kinase growth factor receptors with high homology to the insulin receptor, especially in the tyrosine kinase domain<sup>11</sup>. Both receptors are formed by a heterotetrameric configuration of two  $\alpha$  and two  $\beta$  subunits. The extracellular region is composed of the  $\alpha$  subunits and forms the ligand binding domain. The  $\beta$  subunits form the transmembrane region of the receptor and contain the cytoplasmic tyrosine kinase domain. The tyrosine kinase activity of the

receptor is activated upon binding of the ligand, initiating various signaling cascades that result primarily in cellular proliferation or differentiation. In a similar way, insulin binds the insulin receptor to mediate primarily metabolic responses in insulin-responsive cells. Upon tyrosine phosphorylation of the IGF-I receptor, multiple endogenous substrates bind to its phosphotyrosine domain<sup>11</sup>. These substrates include the insulin related substrate (IRS) family of proteins (IRS-1 through -4) and the SHC family of adapter proteins, with SH2 domains<sup>90,91</sup>. Both docking proteins bind to the IGF-I receptor at the juxtamembrane region and then are able to recruit other substrates that lead to activation of a two principal signaling pathways, the mitogen-activated protein kinase (MAPK) pathway and the phosphoinositide 3'-kinase (PI3K) pathway<sup>92,93</sup> (Fig. 2).

The IGF-I receptor mediates most actions of IGFs in skeletal muscle. In vitro and in vivo, the IGF-I receptor mediates proliferation and differentiation, and several anabolic actions such as stimulation of protein synthesis and glucose uptake and inhibition of protein degradation<sup>46</sup> (Fig. 2). Owing to the close homology of the IGF-I and insulin receptors, hybrid receptors consisting of one IGF  $\alpha\beta$ -half receptor and one insulin  $\alpha\beta$ -half receptor have been described in many tissues, including skeletal muscle, that bind with high affinity to IGF-I, but not insulin<sup>94</sup>.

Unlike insulin, regulation of IGF-I and IGF-II signaling is further complicated by their union with six high-affinity binding proteins (IGFBP-1 to -6)<sup>86</sup>. IGFBP-3 was the first IGFBP identified and forms part of the ternary complex. Recent data suggest that IGFBP-5 also forms a complex with IGF-I and ALS<sup>95,96</sup>. The IGFBPs function as carrier proteins for IGFs in the circulation, protecting them from proteolytic degradation, transporting them to specific tissues, and also modulate IGF actions<sup>86</sup>. In addition to their major roles in the circulation, most target tissues also express IGFBPs where they further regulate the local action of IGFs. Most IGFBPs can either inhibit or potentiate IGF-stimulated actions of IGF-I receptors, depending on the cell type. In addition, IGFBPs may have IGF-independent effects on cell function. Whereas IGFBP-3 functions primarily as the major transporter of the IGFs in the circulation, all six IGFBPs have been shown to be produced by individual tissues in a cell type specific combination.



*Figure 2.* IGF-I receptor signaling pathways in skeletal muscle. Following IGF-I binding, IGF-I receptor autophosphorylates and recruits IRS and SHC proteins to activate two divergent downstream signaling cascades, MAPK and PI3K pathways. Grb2 linked to Sos activates Ras-Raf with subsequent MAPK activation of cell proliferation. p38 MAPK is also involved in both proliferation and differentiation of muscle cells (dashed lines). Similarly, PI3K activation is mediated upon tyrosylphosphorylate AKT leading to cell differentiation, hypertrophy, resistance to apoptosis, and metabolism of muscle cells. A balance between MAPK and PI3K marks the proliferation and differentiation state of skeletal muscle growth (dotted lines). Myoblast proliferation is mediated by MAPK signaling. Moreover, active Raf also suppress IGF-I mediated PI3K signaling, favoring proliferation. In contrast, myotube differentiation is driven by PI3K and AKT can also bind and inactivate Raf, inhibiting MAPK signaling and thus suppressing proliferation.

Of the six IGFBPs, four have been shown to be secreted by various skeletal muscle cell lines. These include IGFBP-2, -4, -5, and -6<sup>97-99</sup>. IGFBP-1 and -3 were not secreted by any of the cultured muscle cell types examined. IGFBP-4 secretion appears to be associated with proliferation and IGFBP-5 is induced during differentiation<sup>100,101</sup>. IGFBP-2, -4, -5, and -6 can all act to inhibit muscle differentiation, but under proper conditions IGFBP-5 has the additional capability of stimulating myogenesis. IGFBP-6 also plays a role in quiescence<sup>97</sup>. *In vivo*, mRNAs of IGFBP-2, -4 and -5 have been shown to be highly expressed in skeletal muscles of rodents<sup>102-104</sup>.

### **3.1 IGF-I in skeletal muscle**

The mature IGF-I polypeptide is highly conserved across species and has many structural similarities to the proinsulin molecule. IGF-I is a singlechain polypeptide composed of 70 amino acid residues in four highly conserved domains (A,B,C,D) (Fig. 3). The IGF-I gene sequence is comprised of six exons. IGF-I premRNA can be alternatively spliced leading to five known mRNA precursors. IGF-I contains two promoters initiating at the 5'end of exons 1 and 2. IGF-I transcripts initiating at exon 1 are widely expressed in all tissues. Transcripts initiating at exon 2 predominate in the liver and are the major endocrine effectors of GH<sup>105</sup>. Three different E peptides are generated by alternative splicing at the 3' end of IGF-I transcripts. These E peptides contain a common N-terminal 16-amino-acid sequence and alternative C-terminal sequences. Transcripts containing Eb sequence are more abundant in liver, whereas Ea-containing transcripts are widespread in extrahepatic tissues<sup>105,106</sup>. A third E peptide splice variant (Ec) has been described that results in the inclusion of a novel splicing site in the intron preceding exon 6. This novel transcript was found in muscles subjected to stretch and was called "mecano growth factor" (MGF)<sup>107,108</sup>. Exons 3 and 4 encode for the entire sequence of the mature polypeptide. Circulating IGF-I isoforms synthesized in the liver are very GH-responsive and therefore the principal endocrine effectors of GH.



*Figure 3.* Schematic representation of *IGF-I* gene and peptide structure. BCAD represent the various domains of the mature peptide. The *IGF-I* gene contains two promoters initiating at exons 1 and 2 leading to two different tissue-specific transcripts. Moreover, tissue specificity is also determined by alternative splicing at 3' end of IGF-I transcripts (Ea or Eb). 1Ea IGF-I transcripts are predominately expressed in resting skeletal muscle whereas MGF is found in skeletal muscles subjected to stretch.

The class of 1/Ea IGF-I transcripts is expressed in all tissues during development<sup>109-112</sup>. At embryonic and early postnatal ages, the 1/Ea IGF-I mRNAs are highly expressed in the liver in a GH-independent manner<sup>113-115</sup>. In contrast, class 2Eb transcripts are detected in the liver postnatally and are tightly GH-regulated<sup>113,114,116</sup>. Other IGF-I isoforms predominately synthesized in skeletal muscle tissue presumably operate in an autocrine or paracrine manner and are less affected by circulating GH levels. The class of 1/Ea IGF-I transcripts have been found in the resting skeletal muscle, whereas the 1/Ec or MGF isoforms were prominent in stretched skeletal muscle<sup>107,108</sup>.

### **3.2** Physiological effects of the IGFs in skeletal muscle

# **3.2.1** Biological significance of the IGFs demonstrated by knockout and transgenic tools

Naturally occurring mutations in the IGF system are extremely rare. A mutation of the *IGF-I* gene in a single patient has shown both intrauterine and postnatal growth retardation<sup>117</sup>. Congenital absence of IGF-II or IGF-IR has not been reported, suggesting that these mutations may be lethal *in utero* or in infancy. In humans, plasma IGF-I levels correlate with body size. Tall children have elevated plasma IGF-I levels<sup>118</sup> and mice selected for high IGF-I levels show increased body weight<sup>119</sup>. Moreover, infusions of rhIGF-I also enhance body weight and size in a number of models, further suggesting a role for circulating IGF-I in growth.

Homologous recombination techniques have allowed the biological significance of the IGF system to be elucidated<sup>120</sup> (Table 1). IGF-I and IGF-II and their cognate IGF-I receptor have been shown to elicit major growth promoting effects<sup>11</sup>. While IGF-II is essential for normal embryonic growth<sup>12</sup>, IGF-I operates throughout prenatal and postnatal development<sup>13</sup>. The results of these studies have demonstrated that the IGFs are very important in muscle growth and development. Mouse lines lacking IGF-I or IGF-II or IGF-II or IGF-I receptor show impaired embryonic development. Although the embryos are viable, they die immediately after birth because of breathing failure<sup>12,13,121</sup>. Mice with a targeted IGF-I deletion show generalized organ hypoplasia, including the muscles<sup>13,121</sup>. Mice lacking the IGF-I receptor exhibit marked skeletal muscle hypoplasia and die soon after birth because they fail to breathe<sup>13</sup>. In contrast, mice lacking functional IGF-II/M-6-P receptors also die around birth, but exhibit fetal overgrowth. These mutants

have elevated circulating levels of IGF-II. These findings support the theory that one function of the IGF-II/M-6-P receptor is to remove excess IGF-II from the circulation<sup>89</sup>.

Mouse	IGF alteration	Skeletal muscle	Reference
		phenotype	
IGF-I <sup>-/-</sup>	Global KO of IGF-I	Hypoplasia	13,121
IGF-I receptor	Global KO of IGF-I receptor	Hypoplasia	13
IGF-II -/-	Global KO of IGF-II	Hypoplasia	12
IGF-II/M-6-P	Global KO of IGF-II/M-6-P	Muscle overgrowth	89
receptor	receptor		
IGF-I Tg	Global IGF-I overexpression	Muscle overgrowth	122
IGF-II Tg	Global IGF-II overexpression	Decreased muscle mass	123
IGF-I Tg targeted	Viral over-expression of IGF-I	Hypertrophy	124-126
	in skeletal muscle		
Dom-neg IGF-I	Overexpression of dom-neg	Hypoplasia at birth/	127
receptor Tg	IGF-I receptor in skeletal	Hyperplasia in adulthood	
targeted	muscle		

*Table 1.* Genetically altered IGF mouse models and their skeletal muscle phenotypes. Domneg, dominant-negative; -/-, homozygous KO; KO, knockout; Tg, transgenic.

Using the opposite approach, transgenic mice overexpressing IGF-I have increased circulating IGF-I levels and increased weight and DNA synthesis in most organs, including skeletal muscle<sup>122</sup>. In contrast, mice expressing very high levels of IGF-II grow more slowly than their control animals<sup>123</sup> (Table 1).

Likewise, mice overexpressing IGF-I in muscle develop increased muscle mass and muscle hypertrophy<sup>124-126</sup>. We developed a transgenic mouse overexpressing a dominant-negative IGF-I receptor in skeletal muscle. The results showed marked hypoplasia in skeletal muscles just after birth, although a compensatory muscle hyperplasia was undergone by the satellite cells in adulthood<sup>127</sup> (Table 1). Thus, the IGF system plays an important role in both embryonic and postnatal growth and is essential for the formation and maintenance of skeletal muscle.

# **3.2.2 IGF-I effects on cell growth (proliferation and differentiation)**

The development of skeletal muscle is a multistep process wherein pluripotent mesodermal cells give rise to myoblasts. Myoblasts permanently withdraw from the cell cycle and begin to express muscle-specific genes and proteins. These processes lead to the fusion of myoblasts into multinucleated differentiated myotubes. Prior to the onset of muscle differentiation, proliferating myoblasts express two muscle-specific basic helix-loop-helix (bHLH) transcription factors, MyoD and Myf5. Upon activation, MyoD and Myf5 induce myoblast withdrawal from the cell cycle and the expression of other myogenic bHLH regulatory transcription factors, including myogenin and MRF4<sup>128-132</sup>. Unlike other growth factors, IGF-I stimulates both myoblast proliferation and differentiation<sup>14,15</sup>. These effects are temporally separated. In proliferating myoblasts, IGF-I increases the expression of cell cycle progression factors<sup>14</sup>. After withdrawal of myoblasts from the cell cycle, IGF-I promotes muscle differentiation by inducing the expression or activity of myogenic regulatory factors<sup>16,133-136</sup>.

### 3.2.3 IGF-I effects on skeletal muscle metabolism

IGF-I and insulin have both shared and unique actions on muscle metabolism. In skeletal muscle and satellite cells, IGF-I increases protein synthesis and inhibits proteolysis. These actions are distinct from the metabolic effects of insulin, which acts primarily by inhibition of proteolysis. This evidence strongly supports the idea that IGF-I is acting via the IGF-I receptor and not the insulin receptor in muscle<sup>46,65,137</sup>. IGF-I also enhances glucose uptake into peripheral tissues, which is an insulin-like effect<sup>138</sup>. It has been proposed that this effect could be mediated by hybrid IGF-I/insulin receptors in both rodents and humans<sup>94,139,140</sup>. Indeed, we showed that in transgenic mice overexpressing a dominant-negative IGF-I receptor specifically in skeletal muscle the diabetic phenotype observed may be mediated by the formation of hybrids between IGF-I and insulin receptors<sup>141</sup>. In contrast, long-term administration of GH inhibits glucose uptake by inducing a state of insulin resistance at the liver and peripheral tissue level as a result of a postreceptor effect<sup>142,143</sup>. These findings further suggest that certain actions of GH occur independently from IGF-I.

### **3.2.4** Mechanism of IGF actions in skeletal muscle

Mice with targeted disruption of the IGF-I receptor exhibit marked skeletal muscle hypoplasia and die soon after birth<sup>13</sup>. In contrast, mice overexpressing IGF-I specifically in muscle develop increased muscle mass and hypertrophy<sup>124-126</sup>. Conversely, overexpression of human IGF-I receptors in rodent skeletal muscle cells enhances IGF-I–induced proliferation and differentiation<sup>15</sup>. Furthermore, functional specific inactivation of the IGF-I receptor in skeletal muscle of mice results in a delay in the proliferation and differentiation processes in skeletal muscle cells<sup>127</sup>. In accordance with these

*in vivo* results, transfection with a kinase-inactive IGF-I receptor abolished IGF-I-induced proliferation and delayed differentiation in mouse myoblasts in culture<sup>144</sup>.

Unlike other growth factors, IGF-I stimulates proliferation, as well as differentiation of myoblasts acting on myogenic regulating factors and effectors of the cell cycle<sup>14,15,127</sup>. The proliferative versus the differentiating functions of IGF-I appear to be mediated by distinct intracellular signaling pathways<sup>145</sup>. Studies suggest that the MAPK pathway mediates cellular proliferation, whereas the PI3K pathway is activated during differentiation<sup>145-148</sup>. In addition to its function on fusion of myoblasts into myotubes, PI3K signaling also mediates IGF-I's anabolic effects on protein and glucose uptake, and cell survival or resistance to apoptosis in skeletal muscle<sup>148-151</sup> (Fig. 2).

However, recent studies have reported that ERK1/2 MAP kinases are differentially regulated in mitogenesis and myogenesis and that the stressactivated protein kinase p38 is critical for regulation of terminal differentiation<sup>152-155</sup>. All of these studies were carried out in cultured muscle cells. The discrepancy might be due to the distinct types of culture cells used in the various studies. The mechanism by which IGF-I-induced activation of these signaling pathways regulates muscle differentiation in vivo has not been clearly determined. Indeed, in mice with an inactivated IGF-I receptor in skeletal muscle (MKR mice) there is multistep regulation of the myogenic program by the p38 and ERK MAP kinase pathways; wherein ERKs have a biphasic profile during myogenesis and p38 is particularly activated for terminal differentiation<sup>127</sup>. These results indicate that the two MAP kinase signaling pathways (ERK and p38) must act in parallel and that both are required for the compensatory myogenic proliferation and differentiation in MKR mice. However, the PI3K pathway may also be important for muscle cell survival, since even though adult MKR mice experience compensatory muscle hyperplasia, they still have reduced muscle mass<sup>127</sup>.

The exact mechanism by which myoblasts switch from a proliferative to a differentiation signal is yet unknown. A crosstalk mechanism between the two opposing pathways has been postulated, depending on the differentiation state of myocytes. Constitutively active Raf-1 promotes proliferation through Erk, and also inhibits IGF-I-induced myoblast fusion<sup>156</sup>. Thus, the MAPK pathway seems to predominate in proliferating myoblasts. In contrast, initiation of myotube fusion is subjected to an increase in Akt <sup>157</sup>. Postmitotic myotubes show increased metabolism, hypertrophy, and survival that is mediated by the PI3K–Akt signaling pathway<sup>149-151,158-160</sup>. Moreover, Akt is able to inhibit Erk activation by binding to and inactivating Raf in myotubes, but not myoblasts<sup>159,160</sup>. It has been suggested that IGF-I protects myoblasts from apoptosis during fusion into mature myotubes by activation of Akt through PI3K<sup>150</sup> (Fig. 2). Taken together, these studies establish that the IGF-I receptor acts as an important regulator of muscle cell differentiation by regulating muscle-specific genes under tight control of its signaling pathways.

# **3.3** Autocrine function of the IGF system in skeletal muscle

Two mice models of liver IGF-I deficiency, in which body growth remained unaltered<sup>6,7</sup>, challenged the somatomedin hypothesis that circulating IGF-I levels mediated by GH secretion are the principal effectors of body growth. These results pointed out the importance of autocrine/paracrine IGF-I secretion in extrahepatic tissues to mediate growth. These data may be also interpreted to suggest that IGF-I may function alone in certain tissues, whereas it may mediate the effects of GH in other tissues as in bone<sup>161,162</sup>. As we mentioned earlier, gene targeting studies of the IGF-I system in skeletal muscle have clearly shown the role of skeletal muscle IGF-I and its receptor in mediating proliferation and differentiation as well as anabolic actions. Since GH binding in skeletal muscle cells has been very difficult to assess<sup>46</sup>, it may be possible that the autocrine/paracrine effects of IGF-I and its receptor are exerted in a GH-independent manner.

## 3.3.1 IGF-I and diabetes

Diabetes results from the inability of insulin secretion from pancreatic beta cells to control blood glucose levels. Under normal conditions, insulin regulates hepatic glucose production and stimulates glucose uptake into peripheral tissues. A primary defect in type 2 diabetes mellitus is insulin resistance, since the liver continues to produce glucose and the uptake of glucose into muscle is impaired.

In humans, rhIGF-I has been used successfully as a coadjuvant to insulin therapy in patients with type 1 and type 2 diabetes. Plasma glucose concentrations are reduced after acute or chronic administration of rhIGF-I <sup>163-165</sup>. By contrast, GH therapy leads to increased insulin resistance. In type 1 patients with poorly controlled diabetes, levels of circulating IGF-I are reduced and GH levels are elevated. Administration of rhIGF-I often restores the high levels of GH to normal, leading to improved insulin sensitivity and enhance glucose uptake in peripheral tissues<sup>166</sup>. In type 2 diabetic patients, who are commonly insulin-resistant, IGF-I therapy leads to enhance muscle glucose uptake <sup>167-170</sup>.

In mice, genetic manipulations were aimed to elucidate insulinstimulated glucose uptake into muscle as potential therapeutic targets for type 2 diabetes. Skeletal muscle expresses substantial levels of both the insulin receptor and the insulin-like growth factor I receptor (IGF-IR). The inhibition of insulin receptor function in skeletal muscle using a dominant negative insulin receptor transgene failed to exhibit severe insulin resistance or diabetes<sup>171</sup>. A similar conclusion was reached in mice with musclespecific deletion of the insulin receptor which developed very mild insulin resistance and remained normoglycemic<sup>172</sup>. One possible explanation for the failure of these latter two mouse models to show insulin resistance and diabetes is that IGF-IRs remained functional in these animals and can compensate for the loss or inactivity of the insulin receptor<sup>173</sup>. Indeed, in our MKR model, both insulin receptor and IGF-IR were affected, because of the formation of hybrid receptors. These mice showed a clearly type 2 diabetic phenotype. Further support for this hypothesis came from the studies of muscle GLUT-4 knockout mice<sup>174</sup>, in which both insulin- and IGF-I-induced glucose uptake in muscle were abrogated, and this resulted in severe insulin resistance and mild diabetes. In humans, a reduced responsiveness of muscle to insulin has also been related to the formation of IGF-I/insulin hybrid receptors in the skeletal muscle of obese and diabetic patients<sup>139,140</sup>. Thus, those findings showed that the IGF-I receptor may be essential to control glucose overall metabolism.

### 3.3.2 IGF-I in regeneration, aging and myopathies

IGF-I plays a critical role in muscle regeneration. It promotes the proliferation and differentiation of satellite cells in the muscle, enabling them to fuse to existing muscle fibers and repair damaged regions<sup>14,46,105,175</sup>.

Normal muscle responds to exercise or stretch with hypertrophy accompanied by a rapid increase in IGF-I and markers of satellite cell differentiation<sup>176</sup>. In response to mechanical stretch, myocytes produce both IGF-I and a splice variant of IGF-I, the mechano-growth factor (MGF)<sup>108</sup>. MGF is proposed to exert a paracrine effect which is essential for myocytoskeletal repair<sup>9</sup>. IGF-I thus serves as a link between muscle mechanical stress and antiapoptotic and hypertrophic repair response<sup>9</sup>.

Cytoskeletal remodeling and proliferation and fusion of satellite cells to existing myotubes occur in muscle degeneration by denervation or structural injury. Following denervation IGF-I, IGF-II, and IGF-IR expression is upregulated in satellite cells localized to the area of regeneration<sup>177,178</sup>.

Numerous studies in which IGF-I has been introduced by recombinant virus or overexpressed in transgenic animals have demonstrated that IGF-I causes significant muscle hypertrophy and strength in postnatal skeletal muscle<sup>16,124-126,179,180</sup>. The IGF-I hypertrophic response is mediated by the activation of the serine/threonine protein phosphatase calcineurin in muscle cells<sup>181,182</sup>. Moreover, recent data using transplanted bone marrow cells in damaged muscle of mice overexpressing local IGF-I have shown that IGF-I enhanced muscle regeneration by promoting the recruitment of these stem cells to sites of muscle damage<sup>183</sup>.

It has clearly been described that in aging a loss of lean body mass and increased adipose tissue occurs. GH/IGF-I axis function declines with aging. Thus, GH secretion and plasma IGF-I levels decline steadily after age 30 in humans. In rat muscle, the cell surface IGF-IR expression decreases between infancy and young adulthood with a subsequent decrease during senescence<sup>184,185</sup>. Moreover, increased insulin resistance in aging is associated with dramatically decreased IGF-I-stimulated muscle protein synthesis and glucose uptake<sup>184</sup>. In patients with GH deficiency, treatment with GH or IGF-I clearly increases lean body weight, muscle mass, muscle strength, and muscle protein synthesis. Exercise partially restores muscle anabolic responsiveness to IGF-I<sup>185,186</sup> in aging muscle. In young men, exercise alone is sufficient to stimulate muscle anabolism, and GH treatment does not augment either strength or protein synthesis<sup>187</sup>. In GH-treated elderly men, resistance exercise did not improve strength or increase muscle IGF-I or IGF-IR mRNA expression and muscle mass<sup>188</sup>. In another study, there is no clear correlation between IGF-I serum concentrations and strength in older men<sup>189</sup>.

During aging, declines in muscle mass and strength are due to impaired regenerative capacity. Studies have shown that postmitotic expression of a local isoform of IGF-I preserves regenerative capacity in senescence via the stimulation of satellite cell proliferation and differentiation<sup>124,126</sup>. Furthermore, IGF-I promotes protein synthetic pathways that can override muscle degeneration<sup>179,190</sup>. These data clearly show that IGF-I can prevent aging-related loss of muscle function<sup>124,191</sup>.

Clinical experience with IGF-I was described for treatment of diabetic patients with severe insulin resistance. IGF-I improves glucose control in these patients, at least in part by decreasing secretion of GH, which has been shown to exacerbate insulin resistance<sup>164,192</sup>. IGF-I myotherapy in humans and animals has focused on IGF-I's anabolic effects on skeletal muscle. Administration of IGF-I, GH, or both increased muscle mass and/or strength in burn injury<sup>193</sup>. Myopathic injury is reported from direct inhibition of postreceptor IGF-I signaling through PI3K in skeletal muscle. IGF-I largely prevents muscle atrophy associated with acute glucocorticoid (acute quadriplegic myopathy) use in rodents and when injected intramuscularly

causes potent localized muscle hypertrophy<sup>179,194</sup>. However, IGF-I treatment of structural myopathy is poorly tested. IGF-I is of clear benefit in murine muscular dystrophy. In dystrophic mice IGF-I together with a high-protein diet significantly reduced protein degradation, enhanced protein synthesis, and increased hindlimb strength<sup>195</sup>. Moreover, postmitotic expression of muscle IGF-I preserves regenerative capacity in dystrophic mice<sup>196</sup>. However, analogous trials for human myopathies have not been reported. From these data, it is possible that IGF-I can prevent or diminish muscle loss associated with disease.

At present, systemic use of IGF-I is unconceivable due to elevated cost and poor absorption of IGF-I. Recently, administration of IGF-I in a stable binary complex with IGFBP-3 has reported to minimize side effects in diabetic patients<sup>192</sup>. Although, muscular injection of a recombinant adenoassociated virus directing overexpression of IGF-I into aged rats and in mice with Duchenne muscular dystrophy has shown to ameliorate muscle decline<sup>124,196</sup>, muscle injection in humans remains impractical for structural myopathies.

### 4. CONCLUSIONS

Many of the effects of GH on various systems maybe affected by IGF-I produced by the liver and reaching that organ via the circulation. On the other hand GH may have direct effects on skeletal muscle for example since GHRs are expressed in those tissues. The source of IGF-I could also be local autocrine/paracrine in origin. While the studies presented in this chapter suggest that all these possibilities are at work when considering the role of the GH/IGF system on skeletal muscle growth and differentiation, more work remains to firmly establish these concepts. With regard to the effects of the GH/IGF system it is apparent that it forms the basis for an important element in skeletal muscle growth and differentiation, given the results of gene-deletion and transgenic mouse models as well as gene therapy with IGF-I. The metabolic effects of the GH/IGF-I system on muscle are also interesting and require further experimentation. GH generally causes insulin resistance in muscle and IGF-I enhances glucose uptake; the overall effect on insulin action and diabetes needs clarification.

#### ACKNOWLEDGEMENTS

We thank Alberto Rodriguez, Chari Garcia-Galiano and Alejandro Rodriguez-Fernandez for technical assistance on preparing this manuscript.

### REFERENCES

- 1. W.D. Salmon Jr. and W.H. Daughaday, A hormonally controlled serum factor which stimulates sulfate incorporation by cartilage in vitro. *J Lab Clin Med*, **49**, 825-836 (1957).
- 2. A.J. D'Ercole, G.T. Applewhite and L.E. Underwood, Evidence that somatomedin is synthesized by multiple tissues in the fetus. *Dev Biol*, **75**, 315-328 (1980).
- 3. A.J. D'Ercole, A.D. Stiles and L.E. Underwood, Tissue concentrations of somatomedin C: further evidence for multiple sites of synthesis and paracrine or autocrine mechanisms of action. *Proc Natl Acad Sci USA*, **81**, 935-939 (1984).
- 4. O.G. Isaksson, J.O. Jansson, & I.A. Gause, Growth hormone stimulates longitudinal bone growth directly. *Science* **216**, 1237-1239 (1982).
- 5. J. Isgaard, A. Nilsson, A. Lindahl, J.O. Jansson and O.G. Isaksson, Effects of local administration of GH and IGF-1 on longitudinal bone growth in rats. *Am J Physiol*, **250**, E367-372 (1986).
- K. Sjogren, J.L. Liu, K. Blad, S. Skrtic, O. Vidal, V. Wallenius, D. LeRoith, J. Tornell, O.G. Isaksson, J.O. Jansson and C. Ohlsson, Liver-derived insulin-like growth factor I (IGF-I) is the principal source of IGF-I in blood but is not required for postnatal body growth in mice. *Proc Natl Acad Sci USA*, **96**, 7088-7092 (1999).
- S. Yakar, J.L. Liu, B. Stannard, A. Butler, D. Accili, B. Sauer and D. LeRoith, Normal growth and development in the absence of hepatic insulin-like growth factor I. *Proc Natl Acad Sci USA*, 96, 7324-7329 (1999).
- S. Yakar, C.J. Rosen, W.G. Beamer, C.L. Ackert-Bicknell, Y. Wu, J.L. Liu, G.T. Ooi, J. Setser, J. Frystyk, Y.R. Boisclair and D. LeRoith, Circulating levels of IGF-1 directly regulate bone growth and density. *J Clin Invest*, 110, 771-781 (2002).
- 9. G. Goldspink, Changes in muscle mass and phenotype and the expression of autocrine and systemic growth factors by muscle in response to stretch and overload. *J Anat*, **194**, 323-334 (1999).
- 10. R. Bassel-Duby and E.N. Olson, Role of calcineurin in striated muscle: development, adaptation, and disease. *Biochem Biophys Res Commun*, **311**, 1133-1141 (2003).
- 11. D. LeRoith, H. Werner, D eitner-Johnson and C.T. Roberts Jr., Molecular and cellular aspects of the insulin-like growth factor I receptor. *Endocr Rev*, **16**, 143-163 (1995).
- 12. T.M. DeChiara, A. Efstratiadis and E.J. Robertson, A growth-deficiency phenotype in heterozygous mice carrying an insulin-like growth factor II gene disrupted by targeting. *Nature*, **345**, 78-80 (1990).
- 13. J.P. Liu, J. Baker, A.S. Perkins, E.J. Robertson and A. Efstratiadis, Mice carrying null mutations of the genes encoding insulin-like growth factor I (Igf-1) and type 1 IGF receptor (Igf1r). *Cell*, **75**, 59-72 (1993).
- 14. J.C. Engert, E.B. Berglund and N. Rosenthal, Proliferation precedes differentiation in IGF-I-stimulated myogenesis. *J Cell Biol*, **135**, 431-440 (1996).
- 15. L.S. Quinn, B. Steinmetz, A. Maas, L. Ong and M. Kaleko, Type-1 insulin-like growth factor receptor overexpression produces dual effects on myoblast proliferation and differentiation. *J Cell Physiol*, **159**, 387-398 (1994).
- 16. A. Musaro and N. Rosenthal, Maturation of the myogenic program is induced by postmitotic expression of insulin-like growth factor I. *Mol Cell Biol*, **19**, 3115-3124 (1999).
- 17. W.L. Miller and N.L. Eberhardt, Structure and evolution of the growth hormone gene family. *Endocr Rev*, **4**, 97-130 (1983).

- D. Cosman, S.D. Lyman, R.L. Idzerda, M.P. Beckmann, L.S. Park, R.G. Goodwin and C.J. March, A new cytokine receptor superfamily. *Trends Biochem Sci*, 15, 265-270 (1990).
- 19. S.M. Pincus, E.F. Gevers, I.C. Robinson, G. van den Berg, F. Roelfsema, M.L. Hartman and J.D. Veldhuis. Females secrete growth hormone with more process irregularity than males in both humans and rats. *Am J Physiol*, **270**, E107-115 (1996).
- 20. J. Spiess, J. Rivier and W. Vale, Characterization of rat hypothalamic growth hormone-releasing factor. *Nature*, **303**, 532-535 (1983).
- 21. N. Ling, F. Esch, P. Bohlen, P. Brazeau, W.B. Wehrenberg and R. Guillemin, Isolation, primary structure, and synthesis of human hypothalamic somatocrinin: growth hormone-releasing factor. *Proc Natl Acad Sci USA*, **81**, 4302-4306 (1984).
- 22. P. Brazeau, W. Vale, R. Burgus, N. Ling, M. Butcher, J. Rivier and R. Guillemin. Hypothalamic polypeptide that inhibits the secretion of immunoreactive pituitary growth hormone. *Science*, **179**, 77-79 (1973).
- T. Imaki, T. Shibasaki, K. Shizume, A. Masuda, M. Hotta, Y. Kiyosawa, K. Jibiki, H. Demura, T. Tsushima and N. Ling, The effect of free fatty acids on growth hormone (GH)-releasing hormone-mediated GH secretion in man. *J Clin Endocrinol Metab*, 60, 290-293 (1985).
- 24. G.S. Tannenbaum, W. Gurd and M. Lapointe, Leptin is a potent stimulator of spontaneous pulsatile growth hormone (GH) secretion and the GH response to GH-releasing hormone. *Endocrinology*, **139**, 3871-3875 (1998).
- 25. B.A. Vuagnat, D.D. Pierroz, M. Lalaoui, P. Englaro, F.P. Pralong, W.F. Blum and M.L. Aubert, Evidence for a leptin-neuropeptide Y axis for the regulation of growth hormone secretion in the rat. *Neuroendocrinology*, **67**, 291-300 (1998).
- 26. Y.Y. Chan, R.A. Steiner and D.K. Clifton, Regulation of hypothalamic neuropeptide-Y neurons by growth hormone in the rat. *Endocrinology*, **137**, 1319-1325 (1996).
- 27. E. Ghigo, E. Arvat, L. Gianotti, B.P. Imbimbo, V. Lenaerts, R. Deghenghi and F. Camanni, Growth hormone-releasing activity of hexarelin, a new synthetic hexapeptide, after intravenous, subcutaneous, intranasal, and oral administration in man. *J Clin Endocrinol Metab*, **78**, 693-698 (1994).
- 28. R. Deghenghi, M.M. Cananzi, A. Torsello, C. Battisti, E.E. Muller and V. Locatelli, GH-releasing activity of Hexarelin, a new growth hormone releasing peptide, in infant and adult rats. *Life Sci*, **54**, 1321-1328 (1994).
- 29. M. Kojima, H. Hosoda, Y. Date, M. Nakazato, H. Matsuo and K. Kangawa, Ghrelin is a growth-hormone-releasing acylated peptide from stomach. *Nature*, **402**, 656-660 (1999).
- M.M. Ilondo, A.B. Damholt, B.A. Cunningham, J.A. Wells, P. De Meyts and R.M. Shymko, Receptor dimerization determines the effects of growth hormone in primary rat adipocytes and cultured human IM-9 lymphocytes. *Endocrinology*, 134, 2397-2403 (1994).
- 31. J.J. Kopchick, History and future of growth hormone research. *Horm Res*, **60 Suppl 3**, 103-112 (2003).
- D.W. Leung, S.A. Spencer, G. Cachianes, R.G. Hammonds, C. Collins, W.J. Henzel, R. Barnard, M.J. Waters and W.I. Wood, Growth hormone receptor and serum binding protein: purification, cloning and expression. *Nature*, 330, 537-543 (1987).
- A. Sotiropoulos, L. Goujon, G. Simonin, P.A. Kelly, M.C. Postel-Vinay and J. Finidori, Evidence for generation of the growth hormone-binding protein through proteolysis of the growth hormone membrane receptor. *Endocrinology*, 132, 1863-1865 (1993).

- T.E. Adams, Differential expression of growth hormone receptor messenger RNA from a second promoter. *Mol Cell Endocrinol*, **108**, 23-33 (1995).
- 35. W.R Baumbach, D.L. Horner and J.S. Logan, The growth hormone-binding protein in rat serum is an alternatively spliced form of the rat growth hormone receptor. *Genes Dev*, **3**, 1199-1205 (1989).
- 36. W.C. Smith, J. Kuniyoshi and F. Talamantes, Mouse serum growth hormone (GH) binding protein has GH receptor extracellular and substituted transmembrane domains. *Mol Endocrinol*, **3**, 984-990 (1989).
- 37. R.J. Ross, Truncated growth hormone receptor isoforms. *Acta Paediatr Suppl*, **88**, 164-166; discussion 167 (1999).
- C. Carter-Su and L.S. Smit, Signaling via JAK tyrosine kinases: growth hormone receptor as a model system. *Recent Prog Horm Res*, 53, 61-82; discussion 82-83 (1998).
- 39. K. Takeda and S. Akira, STAT family of transcription factors in cytokine-mediated biological responses. *Cytokine Growth Factor Rev*, **11**, 199-207 (2000).
- 40. W.J. Leonard and J.J. O'Shea, Jaks and STATs: biological implications. Annu Rev Immunol, 16, 293-322 (1998).
- 41. Y.R. Boisclair, D. Seto, S. Hsieh, K.R. Hurst and G.T. Ooi, Organization and chromosomal localization of the gene encoding the mouse acid labile subunit of the insulin-like growth factor binding complex. *Proc Natl Acad Sci USA*, **93**, 10028-10033 (1996).
- 42. L.S. Argetsinger, G. Norstedt, N. Billestrup, M.F. White and C. Carter-Su, Growth hormone, interferon-gamma, and leukemia inhibitory factor utilize insulin receptor substrate-2 in intracellular signaling. *J Biol Chem*, **271**, 29415-29421 (1996).
- 43. S.C. Souza, G.P. Frick, R. Yip, R.B. Lobo, L.R. Tai and H.M. Goodman, Growth hormone stimulates tyrosine phosphorylation of insulin receptor substrate-1. *J Biol Chem*, **269**, 30085-30088 (1994).
- 44. H.C. Mertani and G. Morel, In situ gene expression of growth hormone (GH) receptor and GH binding protein in adult male rat tissues. *Mol Cell Endocrinol*, **109**, 47-61 (1995).
- J.F. Martini, S.M. Villares, M. Nagano, M.C. Delehaye-Zervas, B. Eymard, P.A. Kelly and M.C. Postel-Vinay Quantitative analysis by polymerase chain reaction of growth hormone receptor gene expression in human liver and muscle. *Endocrinology*, 136, 1355-1360 (1995).
- 46. J.R. Florini, D.Z. Ewton and S.A. Coolican, Growth hormone and the insulin-like growth factor system in myogenesis. *Endocr Rev*, **17**, 481-517 (1996).
- 47. S.L. Pratt and R.V. Anthony, The growth hormone receptor messenger ribonucleic acid present in ovine fetal liver is a variant form. *Endocrinology*, **136**, 2150-2155 (1995).
- 48. H. Kaji, S. Ohashi, H. Abe and K. Chihara, Regulation of the growth hormone (GH) receptor and GH-binding protein mRNA. *Proc Soc Exp Biol Med*, **206**, 257-262 (1994).
- M.J. Dauncey, K.A. Burton, P. White, A.P. Harrison, R.S. Gilmour, C. Duchamp and D. Cattaneo, Nutritional regulation of growth hormone receptor gene expression. *FASEB J*, 8, 81-88 (1994).
- 50. D. Le Roith, C. Bondy, S. Yakar, J.L. Liu and A. Butler, The somatomedin hypothesis: 2001. *Endocr Rev*, 22, 53-74 (2001).

#### Skeletal Muscle

- 51. C.M. Ayling, B.H. Moreland, J.M. Zanelli and D. Schulster, Human growth hormone treatment of hypophysectomized rats increases the proportion of type-1 fibres in skeletal muscle. *J Endocrinol*, **123**, 429-435 (1989).
- 52. H.P. Guler, J. Zapf, E. Scheiwiller and E.R. Froesch, Recombinant human insulin-like growth factor I stimulates growth and has distinct effects on organ size in hypophysectomized rats. *Proc Natl Acad Sci USA*, **85**, 4889-4893 (1988).
- 53. R.M. Palmer, N. Loveridge, B.M. Thomson, S.C. Mackie, E. Tonner and D.J. Flint, Effects of a polyclonal antiserum to rat growth hormone on circulating insulin-like growth factor (IGF)-I and IGF-binding protein concentrations and the growth of muscle and bone. *J Endocrinol*, **142**, 85-91 (1994).
- 54. R.C. Cuneo, F. Salomon, C.M. Wiles and P.H. Sonksen, Skeletal muscle performance in adults with growth hormone deficiency. *Horm Res*, **33 Suppl 4**, 55-60 (1990).
- A. Sartorio, M. Narici, A. Conti, M.Monzani and G. Faglia, Quadriceps and hand-grip strength in adults with childhood-onset growth hormone deficiency. *Eur J Endocrinol*, 132, 37-41 (1995).
- 56. R.C. Cuneo, F. Salomon, C.M. Wiles, R. Hesp and P.H. Sonksen, Growth hormone treatment in growth hormone-deficient adults. I. Effects on muscle mass and strength. *J Appl Physiol*, **70**, 688-694 (1991).
- 57. D. Rudman, A.G. Feller, H.S. Nagraj, G.A. Gergans, P.Y. Lalitha, A.F. Goldberg, R.A. Schlenker, L. Cohn, I.W. Rudman and D.E. Mattson, Effects of human growth hormone in men over 60 years old. *N Engl J Med*, **323**, 1-6 (1990).
- J.S. Christiansen, J.O. Jorgensen, S.A. Pedersen, J. Muller, J. Jorgensen, J. Moller, L. Heickendorf and N.E. Skakkebaek, GH-replacement therapy in adults. *Horm Res*, 36 Suppl 1, 66-72 (1991).
- 59. A. Sartorio and M.V. Narici, Growth hormone (GH) treatment in GH-deficient adults: effects on muscle size, strength and neural activation. *Clin Physiol*, **14**, 527-537 (1994).
- 60. S.A. Beshyah, C. Freemantle, M. Shahi, V. Anyaoku, S. Merson, S. Lynch, E. Skinner, P. Sharp, R. Foale and D.G. Johnston, Replacement treatment with biosynthetic human growth hormone in growth hormone-deficient hypopituitary adults. *Clin Endocrinol (Oxf)*, **42**, 73-84 (1995).
- 61. M.J. Waters, C.A. Shang, S.N. Behncken, S.P. Tam, H. Li, B. Shen and P.E. Lobie, Growth hormone as a cytokine. *Clin Exp Pharmacol Physiol*, **26**, 760-4 (1999).
- 62. J.L. Kostyo, Rapid effects of growth hormone on amino acid transport and protein synthesis. *Ann N Y Acad Sci*, **148**, 389-407 (1968).
- 63. R.F. Wolf, M.J. Heslin, E. Newman, D.B. Pearlstone, A. Gonenne and M.F. Brennan, Growth hormone and insulin combine to improve whole-body and skeletal muscle protein kinetics. *Surgery*, **112**, 2842-91; discussion 291-292 (1992).
- 64. D.L. Russell-Jones, A.J. Weissberger, S.B. Bowes, J.M. Kelly, M. Thomason, A.M. Umpleby, R.H. Jones and P.H. Sonksen, The effects of growth hormone on protein metabolism in adult growth hormone deficient patients. *Clin Endocrinol (Oxf)*, **38**, 427-431 (1993).
- 65. D.A. Fryburg, Insulin-like growth factor I exerts growth hormone- and insulin-like actions on human muscle protein metabolism. *Am J Physiol*, **267**, E331-336 (1994).
- 66. D.L. DeVol, P. Rotwein, J.L. Sadow, J.Novakofski and P.J. Bechtel, Activation of insulin-like growth factor gene expression during work-induced skeletal muscle growth. *Am J Physiol*, **259**, E89-95 (1990).
- 67. M. Ottosson, K. Vikman-Adolfsson, S. Enerback, A. Elander, P. Bjorntorp and S. Eden Growth hormone inhibits lipoprotein lipase activity in human adipose tissue. *J Clin Endocrinol Metab*, **80**, 936-941 (1995).
- 68. H.M. Goodman, Biological activity of bacterial derived human growth hormone in adipose tissue of hypophysectomized rats. *Endocrinolog*, y **114**, 131-135 (1984).
- 69. C. Carter-Su, A.P. King, L.S. Argetsinger, L.S. Smit, J. Vanderkuur and G.S. Campbell, Signalling pathway of GH. *Endocr J*, 43 Suppl, S65-70 (1996).
- 70. P.J. Randle, P.B. Garland, C.N. Hales and E.A. Newsholme, The glucose fatty-acid cycle. Its role in insulin sensitivity and the metabolic disturbances of diabetes mellitus. *Lancet*, **1**, 785-789 (1963).
- P.D. Gluckman, M.M. Grumbach and S.L. Kaplan, The neuroendocrine regulation and function of growth hormone and prolactin in the mammalian fetus. *Endocr Rev*, 2, 363-395 (1981).
- 72. J. Wang, J. Zhou and C.A. Bondy, Igf1 promotes longitudinal bone growth by insulinlike actions augmenting chondrocyte hypertrophy. *FASEB J*, **13**, 1985-1990 (1999).
- 73. N.A. Adamafio, R.J. Towns and J.L. Kostyo, Growth hormone receptors and action in BC3H-1 myocytes. *Growth Regul*, 1, 17-22 (1991).
- 74. D.A. Fryburg, R.A. Gelfand and E.J. Barrett, Growth hormone acutely stimulates forearm muscle protein synthesis in normal humans. *Am J Physiol*, **260**, E499-504 (1991).
- 75. Y. Fong, M. Rosenbaum, K.J. Tracey, G. Raman, D.G. Hesse, D.E. Matthews, R.L. Leibel, J.M. Gertner, D.A. Fischman and S.F. Lowry, Recombinant growth hormone enhances muscle myosin heavy-chain mRNA accumulation and amino acid accrual in humans. *Proc Natl Acad Sci USA*, **86**, 3371-3374 (1989).
- 76. J.C. Chow, P.R. Ling, Z. Qu, L. Laviola, A. Ciccarone, B.R. Bistrian and R.J. Smith, Growth hormone stimulates tyrosine phosphorylation of JAK2 and STAT5, but not insulin receptor substrate-1 or SHC proteins in liver and skeletal muscle of normal rats in vivo. *Endocrinology*, **137**, 2880-2886 (1996).
- 77. D.J. Hill, C.J. Crace, L. Fowler, A.T. Holder and R.D. Milner, Cultured fetal rat myoblasts release peptide growth factors which are immunologically and biologically similar to somatomedin. *J Cell Physiol*, **119**, 349-358 (1984).
- M. Tanaka, Y. Hayashida, K. Sakaguchi, T. Ohkubo, M. Wakita, S. Hoshino and K. Nakashima, Growth hormone-independent expression of insulin-like growth factor I messenger ribonucleic acid in extrahepatic tissues of the chicken. *Endocrinology*, 137, 30-34 (1996).
- 79. J.M. Pell, J.C. Saunders and R.S. Gilmour, Differential regulation of transcription initiation from insulin-like growth factor-I (IGF-I) leader exons and of tissue IGF-I expression in response to changed growth hormone and nutritional status in sheep. *Endocrinology*, **132**, 17971-807 (1993).
- M.E. Coleman, L. Russell and T.D. Etherton, Porcine somatotropin (pST) increases IGF-I mRNA abundance in liver and subcutaneous adipose tissue but not in skeletal muscle of growing pigs. *J Anim Sci*, 72, 918-24 (1994).
- 81. K. Hannon, A. Gronowski and A. Trenkle, Relationship of liver and skeletal muscle IGF-1 mRNA to plasma GH profile, production of IGF-1 by liver, plasma IGF-1 concentrations, and growth rates of cattle. *Proc Soc Exp Biol Med*, **196**, 155-163 (1991).
- P.C. Bates, P.T. Loughna, J.M. Pell, D. Schulster and D.J. Millward, Interactions between growth hormone and nutrition in hypophysectomized rats: body composition and production of insulin-like growth factor-I. *J Endocrinol*, **139**, 117-126 (1993).
- 83. P.T. Loughna, P. Mason and P.C. Bates, Regulation of insulin-like growth factor 1 gene expression in skeletal muscle. *Symp Soc Exp Biol*, **46**, 319-330 (1992).

- D.R. Clemmons, Structural and functional analysis of insulin-like growth factors. Br Med Bull, 45, 465-480 (1989).
- 85. W.H. Daughaday and P. Rotwein, Insulin-like growth factors I and II. Peptide, messenger ribonucleic acid and gene structures, serum, and tissue concentrations. *Endocr Rev*, **10**, 68-91 (1989).
- 86. J.I. Jones and D.R. Clemmons, Insulin-like growth factors and their binding proteins: biological actions. *Endocr Rev*, **16**, 3-34 (1995).
- 87. A. Louvi, D. Accili and A. Efstratiadis, Growth-promoting interaction of IGF-II with the insulin receptor during mouse embryonic development. *Dev Biol*, **189**, 33-48 (1997).
- F. Frasca, G. Pandini, P. Scalia, L. Sciacca, R. Mineo, A. Costantino, I.D. Goldfine, A. Belfiore and R. Vigneri, Insulin receptor isoform A, a newly recognized, highaffinity insulin-like growth factor II receptor in fetal and cancer cells. *Mol Cell Biol*, 19, 3278-3288 (1999).
- 89. J. Baker, J.P. Liu, E.J. Robertson and A. Efstratiadis, Role of insulin-like growth factors in embryonic and postnatal growth. *Cell*, **75**, 73-82 (1993).
- A.A. Butler, S. Yakar, I.H. Gewolb, M. Karas, Y. Okubo and D. LeRoith, Insulin-like growth factor-I receptor signal transduction: at the interface between physiology and cell biology. *Comp Biochem Physiol B Biochem Mol Biol*, **121**, 19-26 (1998).
- 91. M.F. White, The IRS-signaling system: a network of docking proteins that mediate insulin and cytokine action. *Recent Prog Horm Res*, **53**, 119-138 (1998).
- 92. A. Craparo, T.J. O'Neill and T.A. Gustafson, Non-SH2 domains within insulin receptor substrate-1 and SHC mediate their phosphotyrosine-dependent interaction with the NPEY motif of the insulin-like growth factor I receptor. *J Biol Chem*, **270**, 15639-15643 (1995).
- S. Tartare-Deckert, D. Sawka-Verhelle, J. Murdaca and E. Van Obberghen, Evidence for a differential interaction of SHC and the insulin receptor substrate-1 (IRS-1) with the insulin-like growth factor-I (IGF-I) receptor in the yeast two-hybrid system. *J Biol Chem*, 270, 23456-23460 (1995).
- 94. C.P. Moxham, V. Duronio and S. Jacobs, Insulin-like growth factor I receptor betasubunit heterogeneity. Evidence for hybrid tetramers composed of insulin-like growth factor I and insulin receptor heterodimers. *J Biol Chem*, **264**, 13238-13244 (1989).
- 95. S.M. Twigg and R.C. Baxter, Insulin-like growth factor (IGF)-binding protein 5 forms an alternative ternary complex with IGFs and the acid-labile subunit. *J Biol Chem*, **273**, 6074-6079 (1998).
- S.M. Twigg, M.C. Kiefer, J. Zapf and R.C. Baxter, A central domain binding site in insulin-like growth factor binding protein-5 for the acid-labile subunit. *Endocrinology*, 141, 454-457 (2000).
- D.Z. Ewton and J.R. Florini, IGF binding proteins-4, -5 and -6 may play specialized roles during L6 myoblast proliferation and differentiation. *J Endocrinol*, 144, 539-553 (1995).
- 98. R.H. McCusker and D.R. Clemmons, Insulin-like growth factor binding protein secretion by muscle cells: effect of cellular differentiation and proliferation. *J Cell Physiol*, **137**, 505-512 (1988).
- 99. R.H. McCusker, C. Camacho-Hubner and D.R. Clemmons, Identification of the types of insulin-like growth factor-binding proteins that are secreted by muscle cells in vitro. *J Biol Chem*, **264**, 7795-800 (1989).
- 100. D.D. Sarbassov, R. Stefanova, V.G. Grigoriev and C.A. Peterson, Role of insulin-like growth factors and myogenin in the altered program of proliferation and

differentiation in the NFB4 mutant muscle cell line. Proc Natl Acad Sci USA, 92, 10874-10878 (1995).

- P. Rotwein, P.L. James and K. Kou, Rapid activation of insulin-like growth factor binding protein-5 gene transcription during myoblast differentiation. *Mol Endocrinol*, 9, 913-923 (1995).
- P.L. James, S.B. Jones, W.H. Busby Jr., D.R. Clemmons and P.A. Rotwein, highly conserved insulin-like growth factor-binding protein (IGFBP-5) is expressed during myoblast differentiation. *J Biol Chem*, 268, 22305-22312 (1993).
- 103. B.N. Green, S.B. Jones, R.D. Streck, T.L. Wood, P. Rotwein and J.E. Pintar, Distinct expression patterns of insulin-like growth factor binding proteins 2 and 5 during fetal and postnatal development. *Endocrinology*, **134**, 954-962 (1994).
- 104. M.A. Gosteli-Peter, K.H. Winterhalter, C. Schmid, E.R. Froesch and J. Zapf, Expression and regulation of insulin-like growth factor-I (IGF-I) and IGF-binding protein messenger ribonucleic acid levels in tissues of hypophysectomized rats infused with IGF-I and growth hormone. *Endocrinology*, **135**, 2558-2567 (1994).
- 105. C.E. Stewart and P. Rotwein, Growth, differentiation, and survival: multiple physiological functions for insulin-like growth factors. *Physiol Rev*, **76**, 1005-1026 (1996).
- 106. M.L. Adamo, S. Neuenschwander, D. LeRoith and C.T. Roberts Jr., Structure, expression, and regulation of the IGF-I gene. *Adv Exp Med Biol*, **343**, 1-11 (1993).
- 107. G. McKoy, W. Ashley, J. Mander, S.Y. Yang, N. Williams, B. Russell and G. Goldspink, Expression of insulin growth factor-1 splice variants and structural genes in rabbit skeletal muscle induced by stretch and stimulation. J Physiol, 516 (Pt 2), 583-592 (1999).
- S. Yang, M. Alnaqeeb, H. Simpson and G. Goldspink, Cloning and characterization of an IGF-1 isoform expressed in skeletal muscle subjected to stretch. J Muscle Res Cell Motil, 17, 487-495 (1996).
- 109. C.T. Roberts Jr., S.R. Lasky, W.L. Lowe Jr., W.T. Seaman and D. LeRoith, Molecular cloning of rat insulin-like growth factor I complementary deoxyribonucleic acids: differential messenger ribonucleic acid processing and regulation by growth hormone in extrahepatic tissues. *Mol Endocrinol*, 1, 243-248 (1987).
- 110. P. Rotwein, K.M. Pollock, D.K. Didier and G.G. Krivi, Organization and sequence of the human insulin-like growth factor I gene. Alternative RNA processing produces two insulin-like growth factor I precursor peptides. *J Biol Chem*, **261**, 4828-4832 (1986).
- 111. A. Shimatsu and P. Rotwein, Sequence of two rat insulin-like growth factor I mRNAs differing within the 5' untranslated region. *Nucleic Acids Res*, **15**, 7196 (1987).
- E.A. Wong, S.M. Ohlsen, J.A. Godfredson, D.M. Dean and J.E. Wheaton, Cloning of ovine insulin-like growth factor-I cDNAs: heterogeneity in the mRNA population. DNA, 8, 649-657 (1989).
- 113. M.L. Adamo, H. Ben-Hur, C.T Roberts Jr. and D. LeRoith, Regulation of start site usage in the leader exons of the rat insulin-like growth factor-I gene by development, fasting, and diabetes. *Mol Endocrinol*, **5**, 1677-1686 (1991).
- 114. E.C. Hoyt, J.J. Van Wyk and P.K. Lund, Tissue and development specific regulation of a complex family of rat insulin-like growth factor I messenger ribonucleic acids. *Mol Endocrinol*, **2**, 1077-1086 (1988).
- 115. K. Kikuchi, D.P. Bichell and P. Rotwein, Chromatin changes accompany the developmental activation of insulin-like growth factor I gene transcription. *J Biol Chem*, 267, 21505-21511 (1992).

- 116. L.J. Hall, Y. Kajimoto, D. Bichell, S.W. Kim, .PL. James, D. Counts, L.J. Nixon, G. Tobin and P. Rotwein, Functional analysis of the rat insulin-like growth factor I gene and identification of an IGF-I gene promoter. *DNA Cell Biol*, **11**, 301-313 (1992).
- 117. K.A. Woods, C. Camacho-Hubner, M.O. Savage and A.J. Clark, Intrauterine growth retardation and postnatal growth failure associated with deletion of the insulin-like growth factor I gene. *N Engl J Med*, **335**, 1363-1367 (1996).
- 118. M. Gourmelen, Y. Le Bouc, F. Girard and M. Binoux, Serum levels of insulin-like growth factor (IGF) and IGF binding protein in constitutionally tall children and adolescents. *J Clin Endocrinol Metab*, **59**, 1197-1203 (1984).
- 119. H.T. Blair, S.N. McCutcheon, D.D. Mackenzie, J.E. Ormsby, R.A. Siddiqui, B.H. Breier, P.D. Gluckman, Genetic selection for insulin-like growth factor-1 in growing mice is associated with altered growth. *Endocrinology*, **123**, 1690-1692 (1988).
- 120. T.L. Wood, Gene-targeting and transgenic approaches to IGF and IGF binding protein function. *Am J Physiol*, **269**, E613-622 (1995).
- L. Powell-Braxton, P. Hollingshead, C. Warburton, M. Dowd, S. Pitts-Meek, D. Dalton, N. Gillett and T.A. Stewart. IGF-I is required for normal embryonic growth in mice. *Genes Dev*, 7, 2609-2617 (1993).
- 122. L.S. Mathews, R.E. Hammer, R.R. Behringer, A.J. D'Ercole, G.I. Bell, R.L. Brinster and R.D. Palmiter, Growth enhancement of transgenic mice expressing human insulin-like growth factor I. *Endocrinology*, **123**, 2827-2833 (1988).
- 123. C.E. Rogler, D. Yang, L. Rossetti, J. Donohoe, E. Alt, C.J. Chang, R. Rosenfeld, K. Neely and R. Hintz, Altered body composition and increased frequency of diverse malignancies in insulin-like growth factor-II transgenic mice. *J Biol Chem*, 269, 13779-13784 (1994).
- 124. E.R. Barton-Davis, D.I. Shoturma, A. Musaro, N. Rosenthal and H.L. Sweeney, Viral mediated expression of insulin-like growth factor I blocks the aging-related loss of skeletal muscle function. *Proc Natl Acad Sci USA*, **95**, 15603-15607 (1998).
- 125. M.E. Coleman, F. DeMayo, K.C. Yin, H.M. Lee, R. Geske, C. Montgomery and R.J. Schwartz, Myogenic vector expression of insulin-like growth factor I stimulates muscle cell differentiation and myofiber hypertrophy in transgenic mice. *J Biol Chem*, 270, 12109-12116 (1995).
- 126. A. Musaro, K. McCullagh, A. Paul, L. Houghton, G. Dobrowolny, M. Molinaro, E.R. Barton, H.L. Sweeney and N. Rosenthal, Localized Igf-1 transgene expression sustains hypertrophy and regeneration in senescent skeletal muscle. *Nat Genet*, 27, 195-200 (2001).
- 127. A.M. Fernandez, J. Dupont, R.P. Farrar, S. Lee, B. Stannard and D. Le Roith, Musclespecific inactivation of the IGF-I receptor induces compensatory hyperplasia in skeletal muscle. *J Clin Invest*, **109**, 347-355 (2002).
- 128. H.H. Arnold and T. Braun, Targeted inactivation of myogenic factor genes reveals their role during mouse myogenesis: a review. *Int J Dev Biol*, **40**, 345-353 (1996).
- 129. A.B. Lassar, S.X. Skapek and B. Novitch, Regulatory mechanisms that coordinate skeletal muscle differentiation and cell cycle withdrawal. *Curr Opin Cell Biol*, **6**, 788-794 (1994).
- 130. H. Weintraub, The MyoD family and myogenesis: redundancy, networks, and thresholds. *Cell*, **75**, 1241-1244 (1993).
- 131. E.N. Olson and W.H. Klein, bHLH factors in muscle development: dead lines and commitments, what to leave in and what to leave out. *Genes Dev*, **8**, 1-8 (1994).
- 132. D.C. Ludolph and S.F. Konieczny, Transcription factor families: muscling in on the myogenic program. *FASEB J*, **9**, 1595-1604 (1995).

- 133. J.R. Florini, D.Z. Ewton and S.L. Roof, Insulin-like growth factor-I stimulates terminal myogenic differentiation by induction of myogenin gene expression. *Mol Endocrinol*, **5**, 718-724 (1991).
- L.S. Quinn, M. Ehsan, B. Steinmetz and M. Kaleko, Ligand-dependent inhibition of myoblast differentiation by overexpression of the type-1 insulin-like growth factor receptor. J Cell Physiol, 156, 453-461 (1993).
- 135. E.N. Olson, Signal transduction pathways that regulate skeletal muscle gene expression. *Mol Endocrinol*, 7, 1369-1378 (1993).
- 136. A. Rawls, J.H. Morris, M. Rudnicki, T. Braun, H.H. Arnold, W.H. Klein and E.N. Olson, Myogenin's functions do not overlap with those of MyoD or Myf-5 during mouse embryogenesis. *Dev Biol*, **172**, 37-50 (1995).
- 137. D.A Fryburg, L.A. Jahn, S.A. Hill, D.M. Oliveras and E.J. Barrett, Insulin and insulinlike growth factor-I enhance human skeletal muscle protein anabolism during hyperaminoacidemia by different mechanisms. *J Clin Invest*, **96**, 1722-1729 (1995).
- 138. R. Jacob, E. Barrett, G. Plewe, K.D. Fagin and R.S. Sherwin, Acute effects of insulinlike growth factor I on glucose and amino acid metabolism in the awake fasted rat. Comparison with insulin. *J Clin Invest*, **83**, 1717-1723 (1989).
- M. Federici, L. Zucaro, O. Porzio, R. Massoud, P. Borboni, D. Lauro and G. Sesti Increased expression of insulin/insulin-like growth factor-I hybrid receptors in skeletal muscle of noninsulin-dependent diabetes mellitus subjects. *J Clin Invest*, 98, 2887-2893 (1996).
- 140. M. Federici, O. Porzio, D. Lauro, P. Borboni, B. Giovannone, L. Zucaro, M.L. Hribal and G. Sesti, Increased abundance of insulin/insulin-like growth factor-I hybrid receptors in skeletal muscle of obese subjects is correlated with in vivo insulin sensitivity. *J Clin Endocrinol Metab*, **83**, 2911-2915 (1998).
- 141. A.M. Fernandez, J.K. Kim, S. Yakar, J. Dupont, C. Hernandez-Sanchez, A.L. Castle, J. Filmore, G.I. Shulman and D. LeRoith, Functional inactivation of the IGF-I and insulin receptors in skeletal muscle causes type 2 diabetes. *Genes Dev*, 15, 1926-1934 (2001).
- 142. L.R. MacGorman, R.A. Rizza and J.E. Gerich, Physiological concentrations of growth hormone exert insulin-like and insulin antagonistic effects on both hepatic and extrahepatic tissues in man. *J Clin Endocrinol Metab*, **53**, 556-559 (1981).
- 143. R.G. Rosenfeld, D.M. Wilson, L.A. Dollar, A. Bennett and R.L. Hintz, Both human pituitary growth hormone and recombinant DNA-derived human growth hormone cause insulin resistance at a postreceptor site. *J Clin Endocrinol Metab*, 54, 1033-1038 (1982).
- 144. Z.Q. Cheng, S. Adi, N.Y. Wu, D. Hsiao, E.J. Woo, E.H. Filvaroff, T.A. Gustafson and S.M. Rosenthal, Functional inactivation of the IGF-I receptor delays differentiation of skeletal muscle cells. *J Endocrinol*, **167**, 175-182 (2000).
- 145. S.A. Coolican, D.S. Samuel, D.Z. Ewton, F.J. McWade and J.R. Florini, The mitogenic and myogenic actions of insulin-like growth factors utilize distinct signaling pathways. *J Biol Chem*, **272**, 6653-6662 (1997).
- 146. Y. Tamir and E. Bengal, Phosphoinositide 3-kinase induces the transcriptional activity of MEF2 proteins during muscle differentiation. *J Biol Chem*, **275**, 34424-34432 (2000).
- N.C Jones, Y.V. Fedorov, R.S. Rosenthal and B.B. Olwin, ERK1/2 is required for myoblast proliferation but is dispensable for muscle gene expression and cell fusion. J Cell Physiol, 186, 104-115 (2001).

- 148. J.R. Singleton and E.L. Feldman, Insulin-like growth factor-I in muscle metabolism and myotherapies. *Neurobiol Dis*, **8**, 541-554 (2001).
- M.A. Lawlor and P. Rotwein, Coordinate control of muscle cell survival by distinct insulin-like growth factor activated signaling pathways. J Cell Biol, 151, 1131-1140 (2000).
- 150. M.A. Lawlor and P. Rotwein, Insulin-like growth factor-mediated muscle cell survival: central roles for Akt and cyclin-dependent kinase inhibitor p21. *Mol Cell Biol*, **20**, 8983-8995 (2000).
- 151. M.A. Lawlor, X. Feng, D.R. Everding, K. Sieger, C.E. Stewart and P. Rotwein, Dual control of muscle cell survival by distinct growth factor-regulated signaling pathways. *Mol Cell Biol*, **20**, 3256-3265 (2000).
- E. Gredinger, A.N. Gerber, Y. Tamir, S.J. Tapscott and E. Bengal, Mitogen-activated protein kinase pathway is involved in the differentiation of muscle cells. *J Biol Chem*, 273, 10436-10444 (1998).
- 153. Z. Wu, P.J. Woodring, K.S. Bhakta, K. Tamura, F. Wen, J.R. Feramisco, M. Karin, J.Y. Wang and P.L. Puri, p38 and extracellular signal-regulated kinases regulate the myogenic program at multiple steps. *Mol Cell Biol* 20, 3951-3964 (2000).
- 154. A. Zetser, E. Gredinger and E. Bengal, p38 mitogen-activated protein kinase pathway promotes skeletal muscle differentiation. Participation of the Mef2c transcription factor. *J Biol Chem*, **274**, 5193-5200 (1999).
- 155. T.P. Garrington and G.L. Johnson, Organization and regulation of mitogen-activated protein kinase signaling pathways. *Curr Opin Cell Biol*, **11**, 211-218 (1999).
- 156. D.S. Samuel, D.Z. Ewton, S.A. Coolican, T.D. Petley, F.J. McWade and J.R. Florini Raf-1 activation stimulates proliferation and inhibits IGF-stimulated differentiation in L6A1 myoblasts. *Horm Metab Res*, **31**, 55-64 (1999).
- 157. M.R. Calera and P.F. Pilch, Induction of Akt-2 correlates with differentiation in Sol8 muscle cells. *Biochem Biophys Res Commun*, **251**, 835-841 (1998).
- 158. S.C. Bodine, T.N. Stitt, M. Gonzalez, W.O. Kline, G.L. Stover, R. Bauerlein, E. Zlotchenko, A. Scrimgeour, J.C. Lawrence, D.J. Glass and G.D. Yancopoulos, Akt/mTOR pathway is a crucial regulator of skeletal muscle hypertrophy and can prevent muscle atrophy in vivo. *Nat Cell Biol*, **3**, 1014-1019 (2001).
- 159. C. Rommel, B.A. Clarke, S. Zimmermann, L. Nunez, R. Rossman, K. Reid, K. Moelling, G.D. Yancopoulos and D.J. Glass, Differentiation stage-specific inhibition of the Raf-MEK-ERK pathway by Akt. *Science*, **286**, 1738-1741 (1999).
- 160. S. Zimmermann and K. Moelling, Phosphorylation and regulation of Raf by Akt (protein kinase B). *Science*, **286**, 1741-1744 (1999).
- 161. S. Yakar and C.J. Rosen, From mouse to man: redefining the role of insulin-like growth factor-I in the acquisition of bone mass. *Exp Biol Med (Maywood)*, **228**, 245-252 (2003).
- J. Wang, J. Zhou, C.M. Cheng, J.J. Kopchick and C.A. Bondy, Evidence supporting dual, IGF-I-independent and IGF-I-dependent, roles for GH in promoting longitudinal bone growth. *J Endocrinol*, 180, 247-255 (2004).
- M.A. Bach, E. Chin and C.A. Bondy, The effects of subcutaneous insulin-like growth factor-I infusion in insulin-dependent diabetes mellitus. *J Clin Endocrinol Metab*, 79, 1040-1045 (1994).
- 164. K. Cusi and R. DeFronzo, Recombinant human insulin-like growth factor I treatment for 1 week improves metabolic control in type 2 diabetes by ameliorating hepatic and muscle insulin resistance. *J Clin Endocrinol Metab*, **85**, 3077-3084 (2000).
- 165. T.P. Ciaraldi, L. Carter, N. Rehman, P. Mohideen, S. Mudaliar and R.R. Henry, Insulin and insulin-like growth factor-1 action on human skeletal muscle: preferential

effects of insulin-like growth factor-1 in type 2 diabetic subjects. *Metabolism*, **51**, 1171-1179 (2002).

- 166. T.D. Cheetham, J.M. Holly, K. Clayton, S. Cwyfan-Hughes and D.B. Dunger, The effects of repeated daily recombinant human insulin-like growth factor I administration in adolescents with type 1 diabetes. *Diabet Med*, **12**, 885-892 (1995).
- 167. A.C. Moses, L.A. Morrow, M. O'Brien, D.E. Moller and J.S. Flier, Insulin-like growth factor I (rhIGF-I) as a therapeutic agent for hyperinsulinemic insulin-resistant diabetes mellitus. *Diabetes Res Clin Pract*, **28 Suppl**, S185-194 (1995).
- 168. A.C. Moses, S.C. Young, L.A. Morrow, M. O'Brien and D.R. Clemmons, Recombinant human insulin-like growth factor I increases insulin sensitivity and improves glycemic control in type II diabetes. *Diabetes*, **45**, 91-100 (1996).
- 169. D.S. Schalch, N.J. Turman, V.S. Marcsisin, M. Heffernan and H.P. Guler, Short-term effects of recombinant human insulin-like growth factor I on metabolic control of patients with type II diabetes mellitus. J Clin Endocrinol Metab, 77, 1563-1568 (1993).
- 170. P.D. Zenobi, S.E. Jaeggi-Groisman, W.F. Riesen, M.E. Roder and E.R. Froesch, Insulin-like growth factor-I improves glucose and lipid metabolism in type 2 diabetes mellitus. *J Clin Invest*, **90**, 2234-2241 (1992).
- 171. P.Y. Chang, L.J. Goodyear, H. Benecke, J.S. Markuns and D.E. Moller, Impaired insulin signaling in skeletal muscles from transgenic mice expressing kinase-deficient insulin receptors. *J Biol Chem*, **270**, 12593-12600 (1995).
- 172. J.C. Bruning, M.D. Michael, J.N. Winnay, T. Hayashi, D. Horsch, D. Accili, L.J. Goodyear and C.R. Kahn, A muscle-specific insulin receptor knockout exhibits features of the metabolic syndrome of NIDDM without altering glucose tolerance. *Mol Cell*, 2, 559-69 (1998).
- 173. D. Le Roith, H. Kim, A.M. Fernandez and D. Accili, Inactivation of muscle insulin and IGF-I receptors and insulin responsiveness. *Curr Opin Clin Nutr Metab Care*, **5**, 371-375 (2002).
- 174. A. Zisman, O.D. Peroni, E.D. Abel, M.D. Michael, F. Mauvais-Jarvis, B.B. Lowell, J.F. Wojtaszewski, M.F. Hirshman, A. Virkamaki, L.J. Goodyear, C.R. Kahn and B.B. Kahn, Targeted disruption of the glucose transporter 4 selectively in muscle causes insulin resistance and glucose intolerance. *Nat Med*, 6, 924-928 (2000).
- 175. P. Seale and M.A. Rudnicki, A new look at the origin, function, and "stem-cell" status of muscle satellite cells. *Dev Biol*, **218**, 115-124 (2000).
- G.R. Adams, F. Haddad and K.M. Baldwin, Time course of changes in markers of myogenesis in overloaded rat skeletal muscles. *J Appl Physiol*, 87, 1705-1712 (1999).
- 177. G.W. Glazner and D.N. Ishii, Insulin-like growth factor gene expression in rat muscle during reinnervation. *Muscle Nerve*, **18**, 1433-1442 (1995).
- 178. H.L. Keller, B. St Pierre Schneider, L.A. Eppihimer and J.G. Cannon, Association of IGF-I and IGF-II with myofiber regeneration in vivo. *Muscle Nerve*, **22**, 347-354 (1999).
- 179. G.R. Adams and S.A. McCue, Localized infusion of IGF-I results in skeletal muscle hypertrophy in rats. *J Appl Physiol*, **84**, 1716-1722 (1998).
- G.S. Lynch, S.A. Cuffe, D.R. Plant and P. Gregorevic, IGF-I treatment improves the functional properties of fast- and slow-twitch skeletal muscles from dystrophic mice. *Neuromuscul Disord*, 11, 260-268 (2001).
- A. Musaro, K.J. McCullagh, F.J. Naya, E.N. Olson and N. Rosenthal, IGF-1 induces skeletal myocyte hypertrophy through calcineurin in association with GATA-2 and NF-ATc1. *Nature*, 400, 581-585 (1999).

- 182. C. Semsarian, M.J. Wu, Y.K. Ju, T. Marciniec, T. Yeoh, D.G. Allen, R.P. Harvey and R.M. Graham, Skeletal muscle hypertrophy is mediated by a Ca2+-dependent calcineurin signalling pathway. *Nature*, **400**, 576-581 (1999).
- 183. A. Musaro, C. Giacinti, G. Borsellino, G. Dobrowolny, L. Pelosi, L. Cairns, S. Ottolenghi, G. Cossu, G. Bernardi, L. Battistini, M. Molinaro and N. Rosenthal, Stem cell-mediated muscle regeneration is enhanced by local isoform of insulin-like growth factor 1. *Proc Natl Acad Sci USA*, **101**, 1206-1210 (2004).
- 184. D. Dardevet, C. Sornet, D. Attaix, V.E. Baracos and J. Grizard, Insulin-like growth factor-1 and insulin resistance in skeletal muscles of adult and old rats. *Endocrinology*, 134, 1475-1484 (1994).
- P.E. Willis, S. Chadan, V. Baracos and W.S. Parkhouse, Acute exercise attenuates age-associated resistance to insulin-like growth factor I. Am J Physiol, 272, E397-404 (1997).
- 186. P.E. Willis, S.G. Chadan, V. Baracos and W.S. Parkhouse, Restoration of insulin-like growth factor I action in skeletal muscle of old mice. *Am J Physiol*, **275**, E525-530 (1998).
- 187. K.E. Yarasheski, J.J. Zachweija, T.J. Angelopoulos and D.M. Bier, Short-term growth hormone treatment does not increase muscle protein synthesis in experienced weight lifters. *J Appl Physiol*, **74**, 3073-3076 (1993).
- 188. D.R. Taaffe, I.H. Jin, T.H. Vu, A.R. Hoffman and R. Marcus, Lack of effect of recombinant human growth hormone (GH) on muscle morphology and GH-insulinlike growth factor expression in resistance-trained elderly men. J Clin Endocrinol Metab, 81, 421-425 (1996).
- 189. D.P. Kiel, J. Puhl, C.J. Rosen, K. Berg, J.B. Murphy and D.B. MacLean, Lack of an association between insulin-like growth factor-I and body composition, muscle strength, physical performance or self-reported mobility among older persons with functional limitations. *J Am Geriatr Soc*, **46**, 822-828 (1998).
- 190. D.L. Russell-Jones, A.M. Umpleby, T.R. Hennessy, S.B. Bowes, F. Shojaee-Moradie, K.D. Hopkins, N.C. Jackson, J.M. Kelly, R.H. Jones and P.H. Sonksen, Use of a leucine clamp to demonstrate that IGF-I actively stimulates protein synthesis in normal humans. *Am J Physiol*, **267**, E591-598 (1994).
- 191. M. Renganathan, M.L. Messi and O. Delbono, Overexpression of IGF-1 exclusively in skeletal muscle prevents age-related decline in the number of dihydropyridine receptors. *J Biol Chem*, **273**, 28845-28851 (1998).
- 192. D.R. Clemmons, A.C. Moses, M.J. McKay, A. Sommer, D.M. Rosen and J. Ruckle, The combination of insulin-like growth factor I and insulin-like growth factor-binding protein-3 reduces insulin requirements in insulin-dependent type 1 diabetes: evidence for in vivo biological activity. *J Clin Endocrinol Metab*, **85**, 1518-1524 (2000).
- 193. C.H. Fang, B.G. Li, J.J. Wang, J.E. Fischer and P.O. Hasselgren, Treatment of burned rats with insulin-like growth factor I inhibits the catabolic response in skeletal muscle. *Am J Physiol*, **275**, R1091-1098 (1998).
- 194. F. Kanda, K. Takatani, S. Okuda, T. Matsushita and K. Chihara, Preventive effects of insulinlike growth factor-I on steroid-induced muscle atrophy. *Muscle Nerve*, 22, 213-217 (1999).
- M.M. Zdanowicz, J. Moyse, M.A. Wingertzahn, M. O'Connor, S. Teichberg and A.E. Slonim Effect of insulin-like growth factor I in murine muscular dystrophy. *Endocrinology*, 136, 4880-4886 (1995).
- 196. E.R. Barton, L. Morris, A. Musaro, N. Rosenthal and H.L. Sweeney, Muscle-specific expression of insulin-like growth factor I counters muscle decline in mdx mice. *J Cell Biol*, 157, 137-148 (2002).

#### Chapter 6

## CARDIOVASCULAR AND RESPIRATORY SYSTEMS

Åsa Tivesten and Jörgen Isgaard

Research Center for Endocrinology and Metabolism, Department of Internal Medicine, Sahlgrenska Academy at the University of Göteborg, Göteborg, Sweden

Key words: Cardiac output; cardiomyocytes; myocardium; peripheral resistance; sleep disorders

#### 1. STRUCTURAL AND FUNCTIONAL ASPECTS OF THE CARDIOVASCULAR SYSTEM

#### **1.1** The myocardium and cardiac hypertrophy

The myocardium is the muscle layer of the wall of the heart. It consists of cardiac muscle fibres (cardiomyocytes), connective tissue (connective tissue cells and extracellular matrix), and blood vessels in the capillary microcirculation. Approximately 90% of left ventricular tissue volume in small rodents is occupied by cardiomyocytes, while 85% of the cells are interstitial cells.<sup>1</sup> The interstitium is comprised of more than 95% of type I and type III collagens<sup>2</sup>.

The term hypertrophy defines an increase in cellular size based on a concordant increase in all of the cellular constituents<sup>3</sup>. Cardiac hypertrophy is an important compensatory mechanism in left ventricular pressure or volume overload states. The hypertrophy is then induced locally, triggered by the increased wall tension<sup>3</sup>. Besides increased load, extrinsic trophic hormonal or neurogenic factors may trigger cardiac hypertrophy. Load-dependent and load-independent hypertrophy share some common

intracellular as well as autocrine-paracrine mediators<sup>3</sup>. Moreover, there are common transcriptional regulatory pathways during cardiac hypertrophy, including re-expression of protein isoforms normally expressed in the developing heart (such as skeletal  $\alpha$ -actin, atrial and brain natriuretic factor, and  $\beta$ -isoform of myosin heavy chain), as well as immediate-early proto-oncogenes. Besides control of gene transcription, increased translation efficiency represents a means of initiating the hypertrophic response<sup>3</sup>.

# **1.2** Calcium in the contractile process and in hypertrophysignaling

Calcium ions (Ca<sup>2+</sup>) have multiple functions within the cardiomyocyte, although the most well-studied is their role in the contractile process<sup>4</sup>. Following a membrane potential, Ca<sup>2+</sup> enter the cardiomyocyte through the L-type Ca<sup>2+</sup> channels and this Ca<sup>2+</sup> influx in turn triggers release of Ca<sup>2+</sup> from the sarcoplasmic reticulum through the ryanodine receptor. The Ca<sup>2+</sup> then bind to troponin C and the contractile elements are activated. The sarcoplasmic reticulum ATPase (SERCA) then rapidly pumps back Ca<sup>2+</sup> into the sarcoplasmic reticulum. Ca<sup>2+</sup> are also excreted from the cell, mainly through the Na<sup>+</sup>/Ca<sup>2+</sup>Exchanger, which is driven by the electrochemical gradients for Na<sup>+</sup> and Ca<sup>2+</sup>.

 $Na^+/K^+$ -ATPase has a key role in the active transport of Na<sup>+</sup> and K<sup>+</sup> across the cell membrane and establishing a negative electrical potential inside cells. Cardiac glycosides, *e.g.* digoxin, increase cardiac contractility by an inhibition of Na<sup>+</sup>/K<sup>+</sup>-ATPase, which results in an increased intracellular concentration of Na<sup>+</sup> ions<sup>5</sup>. As the Na<sup>+</sup>/Ca<sup>2+</sup>-Exchanger transports Na<sup>+</sup> and Ca<sup>2+</sup> in both directions, increased Na<sup>+</sup> levels are followed by increased Ca<sup>2+</sup> levels within the cell and thereby contractility is enhanced.

Besides being involved in the contractile process,  $Ca^{2+}$  acts as an activator of signal transduction pathways responsible for hypertrophic cardiac growth<sup>4</sup>. Proteins involved in Ca<sup>2+</sup> handling, such as calmodulin and calsequestrin, seem to play important roles as overexpression of these proteins may cause cardiac hypertrophy<sup>4</sup>. Recent studies have reported increased expression and/or activity of the Na<sup>+</sup>/Ca<sup>2+</sup>-Exchanger in experimental models of hypertrophy<sup>6</sup>, and inhibition of Na<sup>+</sup>/K<sup>+</sup>-ATPase has also been linked to cardiac hypertrophy<sup>7</sup>.

#### **1.3 Peripheral resistance and vascular endothelial function**

Arterial blood pressure is determined by two prime factors: cardiac output and peripheral resistance. Cardiac output is almost exclusively regulated by changes in total peripheral resistance, and is adjusted in order to maintain arterial pressure at a normal level<sup>8</sup>. In hypertension, increased peripheral resistance is a consistent finding, as it may be either a primary (vasoconstrictor hypertension) or secondary (volume-loading hypertension) event<sup>8</sup>. The small prearteriolar arteries are the major determinants of local blood flow as well as peripheral resistance. They have a thick smooth muscle layer and a high myogenic basal tone that is modulated by local regulatory mechanisms (mechanical and chemical) as well as higher control systems such as hormonal factors and sympathetic activity.

The endothelium, a continuous cellular monolayer lining the entire vascular tree, is highly involved in the regulation of vascular tone by transmitting signals to the vascular smooth muscle. The endothelium also has a wide range of other homeostatic functions, including blood coagulation, fibrinolysis, and platelet and leukocyte interactions. Endothelial dysfunction has been reported to exist in numerous pathological conditions, such as insulin resistance and congestive heart failure, and it is generally regarded as important in the pathogenesis of atherosclerosis<sup>9</sup>. Endothelial function is impaired in most studies of laboratory animals and humans with chronic hypertension, and it is plausible that endothelial dysfunction is both a cause and an effect of hypertension. According to one definition, endothelial dysfunction implies an imbalance between vasoconstricting/prothrombotic factors (e.g., angiotensin II, endothelin-1) on the one hand, and vasodilating/fibrinolytic factors (e.g., nitric oxide (NO), endothelium-derived hyperpolarizing factor, prostaglandin  $I_2$ ) on the other<sup>10</sup>.

NO is produced mainly in the endothelium, by endothelial NO-synthase from the precursor L-arginine<sup>9</sup>. The production of NO is triggered by various factors, including acetylcholine stimulation and shear stress. Among other functions, NO has a relaxant effect on vascular smooth muscle cells and thereby causes vasodilatation. Functional inhibition of NO-synthase *in vivo* (*e.g.*, with N[w]-Nitro-L-arginine methylester hydrochloride; L-NAME) increases blood pressure, illustrating the importance of NO in the regulation of vascular tone<sup>11</sup>.

#### 1.4 Factors determining cardiac output

Cardiac output, the volume of blood ejected from the heart in 1 minute, is the product of stroke volume and heart rate. Stroke volume is determined by both systolic (ejecting) function and diastolic (relaxing) function. Left ventricular systolic function, in turn, is the product of interaction of four variables<sup>12</sup>:

1. *Preload*, *i.e.*, the hemodynamic force exerted on the ventricular wall during filling. In practice, end-diastolic volumes/dimensions are

considered reliable indices of ventricular preload. By intrinsic regulation of cardiac pumping through the Frank–Starling mechanism (*i.e.*, stretched muscle contracts with greatly increased force), cardiac output increases with increased preload. Preload is largely regulated by venous return to the heart.

- 2. *Afterload*, *i.e.*, the load against which the muscle exerts its contractile force during systole.
- 3. *Left ventricular* myocardial mass, *i.e.*, more muscle generates more force. With hypertrophy of the left ventricular wall, wall stress also decreases, and systolic function is thereby enhanced.
- 4. *Myocardial contractility*, *i.e.*, the contractile state of the cardiac muscle.

#### 1.5 Cardiac remodeling

Left ventricular remodeling is the process by which ventricular size, shape, and function are regulated by mechanical, neurohormonal, and genetic factors<sup>13</sup>. It may be physiological/adaptive or pathological, *e.g.*, due to myocardial infarction or hypertension. Post-infarction remodeling may be divided into an early phase (infarct expansion) and a late phase with global left ventricular dilatation and mural hypertrophy<sup>13</sup>. During infarct expansion and ventricular dilatation, cardiac pumping function is partly restored through the Frank–Starling mechanism and this process is associated with alterations in the structure and function of the extracellular matrix. Hypertrophy of the surviving myocardium occurs as a counterregulatory mechanism to decrease wall stress. In the case of a large myocardial injury, the remodeling fails to normalize wall stress, resulting in a progressive dilatation and deterioration of contractile function and subsequently heart failure.

#### 2. GH/IGF-I AND THE CARDIOVASCULAR SYSTEM

Besides its growth-promoting and metabolic effects, the GH/IGF-I axis has an important role during cardiac development and in maintaining the structure and function of the heart<sup>13,14</sup>. This arrangement is well suited to its purpose, GH/IGF-I increases cardiac output to meet the peripheral metabolic demands elicited by its own actions. GH/IGF-I influences the vascular system and may have a role in the regulation of vascular tone and thereby peripheral resistance. It was recently shown that the *IGF-I* gene locus is linked to both systolic blood pressure and cardiac dimensions<sup>15</sup>.

The myocardium and vessels express  $IGF-I^{16-18}$  and functional receptors for both  $GH^{19-21}$  and  $IGF-I^{22,23}$  and IGF-I production is up-regulated in

response to GH<sup>17</sup>. Thus, there are possibilities of direct actions of GH as well as endocrine or autocrine/paracrine effects of IGF-I on the cardiovascular system. However, although interaction of the GH/IGF-I axis and the cardiovascular system has been extensively studied, the relative importance of direct effects of GH and local and endocrine IGF-I remains unclear.

#### 2.1 GH/IGF-I and peripheral resistance

In normal rats, intravenous administration of IGF-I acutely decreases mean arterial blood pressure within a few minutes<sup>24,25</sup>. In humans, stroke volume and cardiac output were increased but blood pressure remained unchanged a few hours after a single injection of IGF-I<sup>26</sup>. In heart failure patients, IGF-I infusion increased cardiac output and decreased peripheral resistance within approximately 2 h<sup>27</sup>, while a similar effect of GH was seen after 24 h when serum IGF-I was concomitantly increased<sup>28</sup>. These data suggest that IGF-I has potent, acute functional effects on the cardiovascular system. In a recent study, it was shown in healthy subjects that GH directly increased forearm blood flow in parallel with a decrease in peripheral resistance<sup>29</sup>. Moreover, these changes were suggested to be mediated by stimulation of endothelial function through the NO system. However, the effect was not seen until 4 h after GH infusion and it is possible that the GH effect was mediated by local production of IGF-I in the periphery.

Several studies have shown that the vascular actions of IGF-I may be exerted by stimulation of release of NO and/or other vasodilators from the endothelium. IGF-I stimulates NO release from cultured endothelial cells<sup>30</sup>, vascular smooth muscle cells<sup>31</sup>, as well as aortic preparations<sup>25</sup>. Pretreatment with the NO-synthase inhibitor L-NAME may abolish the vasodilating effect of IGF-I, as shown in large arteries<sup>24,32-34</sup>. There are also reports of eicosanoids as mediators of the vasodilating effects of IGF-I and that indomethacin may prevent vasodilatation by IGF-I<sup>32,34,35</sup>. Moreover, IGF-I may cause vasorelaxation by non-endothelium-dependent actions<sup>36</sup>, possibly by increasing the activity of the Na<sup>+</sup>,K<sup>+</sup>-ATPase in vascular smooth muscle cells<sup>37</sup>. Despite the accumulating data on IGF-I as a vasodilator, there have only been a few studies addressing the role of IGF-I in physiological regulation of peripheral resistance. In 1996, Lembo et al. showed that mice with a mutant IGF-I allele and 30% of wild-type IGF-I levels present in all tissues and serum have elevated blood pressure<sup>38</sup>. In a recent study from our laboratory of transgenic mice with a liver-specific knockout of IGF-I (LI-*IGF-I*<sup>/-</sup> mice), resulting in a 80% decrease of circulating IGF-I, we found a similar elevation of blood pressure as in the IGF-I mutant mice with a similar decrease in serum IGF-I, indicating that the decrease in endocrineacting IGF-I is responsible for this blood pressure-elevating effect<sup>39</sup>.

#### 2.2 GH/IGF-I and cardiac hypertrophy

*In vitro*, the results of many studies have shown that IGF-I increases protein synthesis<sup>40,41</sup> and the size of cardiomyocytes<sup>42</sup>. Many studies have failed to show direct, IGF-I-independent hypertrophic effects of GH on cardiomyocytes<sup>41,42</sup>, but it has been suggested that GH may cause alterations in cardiomyocyte metabolism and stimulate hypertrophy independently of IGF-I<sup>43,44</sup>.

Genes involved in cardiac hypertrophy. It has previously been shown that expression of muscle-specific genes<sup>42</sup>. IGF-I stimulates the hypophysectomized (hx) rats, GH and IGF-I treatment stimulated similar cardiac gene expression (skeletal a-actin, atrial natriuretic factor), but neither IGF-I nor GH caused a shift of myosin heavy chain isoforms<sup>45</sup>. Moreover, altered Na<sup>+</sup>, Ca<sup>2+</sup>-exchanger expression<sup>6</sup> and inhibition of Na<sup>+</sup>, K<sup>+</sup>-ATPase<sup>7</sup> have been associated with development of hypertrophy. Interestingly, it has been shown that expression of activated calcineurin mimics the hypertrophic effects of IGF-I in skeletal muscle<sup>46</sup>. Taken together, these data suggest a role for calcium as a mediator of GH-induced hypertrophy.

Effects on extracellular matrix. Cardiac hypertrophy requires concomitant remodeling of the heart and attention has recently been focused on the role of the extracellular matrix in the remodeling process<sup>47</sup>. Matrix metalloproteinases, which are capable of degrading all the matrix components of the heart, are the driving force behind myocardial matrix remodeling<sup>47</sup>. It has been shown that IGF-I promotes collagen synthesis by fibroblasts<sup>48</sup> and that GH increases the collagen deposition rate in the heart<sup>49</sup>, while in GH-induced hypertrophy the volume fraction of collagen is normal<sup>50,51</sup>. Thus, available data are in line with an increased synthesis, as well as breakdown of extracellular matrix with a finally unchanged cardiac collagen concentration by GH.

*Increased number of cardiomyocytes?* Besides promoting cardiomyocyte growth, GH/IGF-I may influence the trophic status of the myocardium by preventing cardiomyocyte loss through apoptosis. It has been shown that apoptosis is ongoing even in the normal heart<sup>52</sup> and there is substantial evidence that IGF-I acts as an inhibitor of apoptosis and that GH/IGF-I may be "cardioprotective" <sup>53-56</sup>. However, although an increased number of nuclei in the heart of GH-treated adult rats has been reported<sup>51</sup>, there is still no evidence that GH or IGF-I affects cardiac myocyte number in the normal heart unless overexpressed from the embryonic cell stage<sup>52</sup>.

#### 2.3 GH/IGF-I and cardiac contractility

In line with the findings on cardiac hypertrophy, there are several *in vitro* studies demonstrating direct effects of IGF-I on intrinsic cardiac contractility<sup>53-55</sup>, while there is thus far no evidence of direct, IGF-I-independent effects of GH on cardiac contractility. However, if animals are treated with GH *in vivo*, allowing stimulation of IGF-I, subsequent *in vitro* assessment shows improved contractility<sup>56,57</sup>. Accordingly, decreased contractility has been demonstrated in dwarf rats with GH/IGF-I deficiency<sup>58-60</sup>. In contrast, a paradoxical enhancement of cardiac contractility was observed in the IGF-I mutant mouse<sup>38</sup>.

At least three different mechanisms have been suggested for the GH/IGF-I-induced increase in cardiac contractility: (1) altered intracellular  $Ca^{2+}$  transients, (2) increased myofilament  $Ca^{2+}$  sensitivity, and (3) a myosin isoform shift.

Intracellular Ca-transients. IGF-I has been shown to acutely affect Ca<sup>2+</sup> currents within the cardiomyocyte, with increased peak Ca<sup>2+</sup> levels<sup>54,61</sup>, and an altered time course of the current<sup>54</sup> in association with increased contractility. Specifically, the activity of L-type Ca<sup>2+</sup>channels was acutely increased by IGF-I in vitro<sup>62</sup>. In cardiomyocytes from acromegalic rats, the action potential duration was increased as a result of a decrease in density of a transient outward current carried by K<sup>+</sup>, which, in turn prolongs the Ca<sup>2+</sup> influx through L-type  $Ca^{2+}$  channels<sup>63</sup>. In contrast to findings in other in vitro studies, the acute increase of inotropy by IGF-I was associated with decreased peak  $Ca^{2+}$  levels but increased  $Ca^{2+}$  sensitivity of the contractile elements in isolated whole heart preparations<sup>55</sup>. No influence of GH on  $Ca^{2+}$ currents has been seen in acute settings<sup>55,61</sup>, while after more long-term treatment in vivo, GH has been suggested to increase peak intracellular Ca<sup>2+</sup> levels measured ex vivo<sup>56,57</sup>. Accordingly, reduced peak intracellular Ca<sup>2+</sup> levels as well as slowed intracellular Ca<sup>2+</sup>clearing has been demonstrated in GH/IGF-I deficiency<sup>60</sup>, while others report peak intracellular Ca<sup>2+</sup> levels to be unchanged in GH/IGF-I deficiency<sup>59</sup>.

To date, little is known about possible genes involved in the action of GH/IGF-I in altering Ca<sup>2+</sup> handling. An up-regulation of SERCA levels has been suggested to contribute to the increased contractile function elicited by GH after myocardial infarction<sup>57</sup> and in rapid pacing heart failure<sup>54</sup>, while another study<sup>64</sup>, did not detect any change in SERCA expression. SERCA may increase contractility by enhancement of the so-called contractile reserve, *i.e.*, the Ca<sup>2+</sup> storage within the sarcoplasmic reticulum, allowing higher peak Ca<sup>2+</sup> levels on stimulation. Ueyama *et al.* also suggested that

GH treatment in cardiomyopathic, but not normal, hamsters preserved cardiac ryanodine receptor density<sup>65</sup>.

Inhibition of Na<sup>+</sup>,K<sup>+</sup>-ATPase increases intracellular Na, which causes a net Ca<sup>2+</sup> influx through the Na<sup>+</sup>,Ca<sup>2+</sup> exchange system and thereby enhances cardiac contractility<sup>5</sup>, in addition to a possible role in the development of hypertrophy<sup>7</sup>. However, the activity of cardiac Na<sup>+</sup>,K<sup>+</sup>-ATPase has previously been reported to not be significantly changed by GH treatment in pigs<sup>66</sup> or GH-deficient dwarf rats<sup>58</sup>.

*Myofilament Ca-sensitivity and myosin isoform shift.* GH/IGF-I has been suggested to increase myofilament  $Ca^{2+}$  sensitivity<sup>55,67,68</sup> and maximum  $Ca^{2+}$  activated force<sup>55,56,67</sup>. However, data are conflicting, and others report unchanged<sup>61</sup> or even decreased<sup>56</sup> myofilament  $Ca^{2+}$  sensitivity by GH/IGF-I. In dwarf rats, unchanged myofilament  $Ca^{2+}$  sensitivity has been reported<sup>59,60</sup>, although maximum  $Ca^{2+}$  activated tension was less<sup>59</sup>. In animal models of GH excess, a shift toward a myosin isoform with lower ATPase activity has been demonstrated, which may decrease the energy demand of the contractile process<sup>67,69</sup>.

Taken together, available data suggest that GH/IGF-I may increase cardiac contractility through alterations of intracellular  $Ca^{2+}$  transients, myofilament  $Ca^{2+}$  sensitivity and myosin isoform expression, although the findings vary between different settings and between studies. Besides regulation of ion channel activity, GH/IGF-I may regulate the expression of ion channels.

#### 2.4 GH deficiency

In line with a trophic effect of GH/IGF-I on the heart, GH deficiency has been associated with a reduction of left ventricular mass and cardiac output in both experimental<sup>58,70</sup> and clinical<sup>71</sup> studies. GH substitution may improve cardiac output, as well as left ventricular mass and/or wall thickness<sup>58,70-72</sup>. Both experimental and clinical studies have shown that GH deficiency is associated with an increased total peripheral resistance that may be reduced by GH substitution<sup>70,72</sup>. Despite increased total peripheral resistance, blood pressure may be unchanged or even decreased in experimental models of GH deficiency<sup>58,70,73</sup>, as well as in studies of young adults with GH deficiency<sup>71,74,75</sup>.

Hypopituitary patients on routine pituitary hormone replacement therapy, excluding GH, have an increased mortality from cardiovascular disease<sup>76</sup> and several cardiovascular risk factors (abdominal obesity, insulin resistance, decreased fibrinolysis, disturbed lipoprotein pattern, atherosclerosis) have been identified in GH-deficient patients<sup>77</sup>. These factors may underlie the increased prevalence of treated hypertension reported in a cohort of patients with adult onset GH deficiency<sup>77</sup>, a notion

that is supported by a recent study showing increased systolic blood pressure in familial isolated GH deficiency<sup>78</sup>. Taken together, the reports on blood pressure levels in GH deficiency are somewhat conflicting, and it is likely that the time of onset, as well as the duration of the GH deficiency, determines both its cardiac and vascular consequences.

#### 2.5 GH/IGF-I excess

Some of the first experimental studies of the hemodynamic effects of chronic GH hypersecretion utilized a rat model with an implantable GHsecreting tumor<sup>79</sup>. These rats displayed cardiomegaly and increased cardiac contractility and output, while peripheral resistance was decreased<sup>79</sup>. In human acromegalics two stages of cardiovascular disease have been identified, an early "hyperkinetic" stage with increased cardiac output and decreased total peripheral resistance, and a late stage characterized by cardiomyopathy, hypertension, atherosclerosis, and ischemic heart disease<sup>13</sup>. This is in agreement with experimental studies suggesting beneficial effects initially, but later impaired cardiac performance in mice with overexpression of IGF-I in the heart<sup>80</sup> or overexpression of bovine GH (bGH) in mice<sup>81,82</sup>. The acromegalic cardiomyopathy is characterized by cardiomegaly, ventricular hypertrophy, fibrosis, and degeneration of cardiomyocytes<sup>83</sup>. Hypertension has been reported in the bGH mouse model<sup>84</sup> and different studies report a prevalence of hypertension of 20-50% of acromegalic patients<sup>83</sup>. Although some of these late complications can be ascribed to the metabolic derangements in acromegaly rather than the GH hypersecretion per se<sup>85</sup>, the prevalent opinion today is that there is an independent detrimental cardiovascular effect of chronic exposure to excessive GH levels<sup>83</sup>.

#### 2.6 The GH/IGF-I axis in cardiovascular pathology

Considering the importance of the GH/IGF-I axis in cardiovascular physiology, one might expect that the GH/IGF-I axis plays a role, either mediary or compensatory, in cardiovascular pathophysiology. Indeed, local IGF-I, systemic IGF-I, and GH levels may all be affected in different cardiovascular pathological states. IGF-I expression is known to increase in cardiomyocytes<sup>86</sup>, vascular smooth muscle cells<sup>87</sup>, and whole heart preparations<sup>86</sup> in response to stretch *in vitro*. Moreover, cardiac IGF-I expression is up-regulated in both pressure and volume overload states<sup>20,88</sup>.

*Hypertension.* In accordance with the aforementioned, hypertensive rats have increased vascular and cardiac IGF-I expression<sup>21,89</sup>, while serum IGF-I has been reported to be unchanged<sup>90</sup>. Serum IGF-I has been reported to be

increased<sup>91,92</sup> and to be an independent determinant of cardiac mass<sup>93</sup> in human hypertension. Others have suggested that serum IGF-I correlates negatively to blood pressure in borderline hypertension<sup>94</sup>. Decreased IGF-I in serum, with a good negative correlation to mean arterial pressure, has also been reported in obese hypertensive patients<sup>95</sup>. Taken together, data on the relationship between serum IGF-I levels and blood pressure are conflicting, possibly reflecting alternatively causative or compensatory roles of IGF-I in different cohorts.

*Myocardial infarction.* As in all conditions with acute stress, serum levels of GH and IGF-I are elevated in acute myocardial infarction<sup>96,97</sup>. It has been shown that patients with poor outcomes have lower serum IGF-I in the acute infarction phase<sup>97</sup> and that low serum IGF-I is associated with increased the risk of ischemic heart disease<sup>98</sup>. Accordingly, a polymorphism in the IGF-I gene causing low serum IGF-I levels increases risk of myocardial infarction<sup>99</sup>. Following the increased load, auto/paracrine IGF-I is overexpressed within 6 h after experimental myocardial infarction, especially in the zone of viable myocardium immediately adjacent to the infarct<sup>100</sup>. After the acute infarction phase, local IGF-I remains elevated for at least 20 days<sup>100</sup>.

#### 2.7 GH/IGF-I in the treatment of heart failure

Several research groups have studied the effects of GH and/or IGF-I in states of impaired cardiac function. In an established rat model of congestive heart failure following ligation of the left coronary artery, GH and IGF-I has been found to increase stroke volume and cardiac output<sup>101,102</sup>, also in the presence of angiotensin-converting enzyme (ACE) inhibition<sup>103</sup>. GH treatment of rats with experimental myocardial infarction has also been found to improve myocardial bioenergetics<sup>104</sup> and long-term survival<sup>105</sup>.

In studies of patients with congestive heart failure due to ischemic or dilated cardiomyopathy, both acute administration of GH and more chronic treatment evoked beneficial effects<sup>106,107</sup>. Recently, however, randomised, placebo-controlled studies have failed to show any significant GH-mediated improvement of cardiac performance in patients with heart failure, despite significant increases in IGF-1<sup>108,109</sup>. Acquired GH resistance might be an important feature of severe heart failure and may explain the diverse responses to GH therapy observed in different patients<sup>110</sup>. Moreover, the duration of treatment has been relatively short (3 months) and the studies include a small number of patients, which may also contribute to the negative results. Hence, whether GH treatment will finally find a place in the treatment of heart failure remains to be established.

#### **3. GH/IGF-I AND THE RESPIRATORY SYSTEM**

Although traditionally the respiratory system is not considered to be under major endocrine regulation, it has become clear that hormones can influence respiration in several ways. In principle, hormones can influence breathing at the level of CNS, at peripheral chemoreceptors, or by altering basic metabolic rate. Long-term indirect effects of hormones on the respiratory system include alterations of acid/base balance, body temperature and body composition (for review, see ref. 111). Studies related to the GH/IGF-I axis and respiration are essentially limited to sleep-disorder conditions such as obstructive sleep apnea syndrome (OSAS), chronic obstructive pulmonary disease (COPD) and Prader–Willi syndrome (PWS).

Patients with OSAS have decreased GH secretion that can be normalized with continuous positive airway pressure (CPAP) treatment<sup>112</sup>. There are also a few reports suggesting a beneficial effect of GH treatment of patients with COPD on peak expiratory flow, respiratory muscle strength<sup>113</sup> and maximal inspiratory pressure<sup>114</sup>.

PWS is a neurogenetic disorder in which clinical characteristics include short stature, mental retardation, behavior disorders, muscular hypotonia, hypogonadism and hyperphagia often resulting in severe obesity. In addition, many PWS patients frequently suffer from excessive daytime somnolens and apnea during sleep. These disturbancies could be in part related to obesity, but PWS patients also have decreased sensitivity of peripheral chemoreceptors to changes in blood oxygen and  $CO_2^{115}$ . GH treatment of patients with PWS results in increased ventilation, central inspiratory drive, airway occlusion pressure and hypercapnic respiratory response<sup>116</sup>.

#### REFERENCES

- 1. P.A. Doevendans, M.J. Daemen, E.D. de Muinck and J.F. Smits, Cardiovascular phenotyping in mice. *Cardiovasc Res*, **39**, 34-49 (1998).
- 2. J. D'Armiento, Matrix metalloproteinase disruption of the extracellular matrix and cardiac dysfunction. *Trends Cardiovasc Med*, **12**, 97-101 (2002).
- 3. G.T. Copper, Basic determinants of myocardial hypertrophy: a review of molecular mechanisms. *Annu Rev Med*, **48**, 13-23 (1997).
- 4. Frey N, McKinsey TA, Olson EN. Decoding calcium signals involved in cardiac growth and function. Nat Med;6:1221-1227 (2000).
- 5. K. Kjeldsen, A. Norgaard and M. Gheorghiade, Myocardial Na<sup>+</sup>,K<sup>+</sup>-ATPase: the molecular basis for the hemodynamic effect of digoxin therapy in congestive heart failure. *Cardiovasc Res*, **55**, 710-713 (2002).
- 6. K.R. Sipido, P.G. Volders, M.A. Vos and F. Verdonck, Altered Na<sup>+</sup>/Ca<sup>2+</sup> exchange activity in cardiac hypertrophy and heart failure: a new target for therapy? *Cardiovasc Res*, **53**, 782-805 (2002).

- Z. Xie, P. Kometiani, J. Liu, J. Li, J.I. Shapiro and A. Askari, Intracellular reactive oxygen species mediate the linkage of Na<sup>+</sup>/K<sup>+</sup>-ATPase to hypertrophy and its marker genes in cardiac myocytes. *J Biol Chem*, 274, 19323-19328 (1999).
- 8. A.C. Guyton, Textbook of medical physiology, Seventh ed. W B Saunders company, Philadelphia, PA, USA (1986).
- 9. Behrendt D, Ganz P. Endothelial function. From vascular biology to clinical applications. *Am J Cardiol*, **90**, 40L-48L (2002).
- 10. Luscher TF, Barton M. Biology of the endothelium. Clin Cardiol, 20, II-3-10 (1997).
- 11. M. Lavallee, M. Takamura, R. Parent and E. Thorin, Crosstalk between endothelin and nitric oxide in the control of vascular tone. *Heart Fail Rev*, **6**, 265-76 (2001).
- 12. E.B. Braunwald and J. Ross Jr, Control of cardiac performance. In: Handbook of Physiology. American Physiological Society, Bethesda, pp 533-580 (1979).
- 13. L Sacca, A Cittadini, S. Fazio Growth hormone and the heart. *Endocr Rev*, **15**, 555-573 (1994).
- J Ren, WK Samson, JR.Sowers Insulin-like growth factor I as a cardiac hormone: physiological and pathophysiological implications in heart disease. J Mol Cell Cardiol, 31, 2049-2061 (1999).
- Z. Nagy, A. Busjahn, S. Bahring, H.D. Faulhaber, H.D. Gohlke, H. Knoblauch, M. Rosenthal, B. Muller-Myhsok, H. Schuster and F.C. Luft, Quantitative trait loci for blood pressure exist near the IGF-1, the Liddle syndrome, the angiotensin II-receptor gene and the renin loci in man. *J Am Soc Nephrol*, **10**, 1709-16 (1999).
- J. Isgaard, A. Nilsson, K. Vikman and O.G. Isaksson Growth hormone regulates the level of insulin-like growth factor-I mRN, A in rat skeletal muscle. J Endocrinol, 120,107-112 (1989).
- A.J. D'Ercole, A.D. Stiles and L.E. Underwood, Tissue concentrations of somatomedin C: further evidence for multiple sites of synthesis and paracrine or autocrine mechanisms of action. *Proc Natl Acad Sci USA*, 81,935-9399 (1984).
- 18. P. Delafontaine, K.E. Bernstein and R.W. Alexander, Insulin-like growth factor I gene expression in vascular cells. *Hypertension*, **17**, 693-699 (1991).
- 19. L.S. Mathews, B. Enberg, G. Norstedt, Regulation of rat growth hormone receptor gene expression. *J Biol Chem*, **264**, 9905-9910 (1989).
- J. Isgaard, H. Wåhlander, M.A. Adams and P. Friberg, Increased expression of growth hormone receptor mRNA and insulin-like growth factor-I mRNA in volumeoverloaded hearts. *Hypertension*, 23, 884-888 (1994).
- A. Wickman, P. Friberg, M.A. Adams, G.L. Matejka, C. Brantsing, G. Guron, J. Isgaard, Induction of growth hormone receptor and insulin-like growth factor-I mRNA in aorta and caval vein during hemodynamic challenge. *Hypertension*, 29, 123-130 (1997).
- H.P. Guler, J. Zapf, E. Scheiwiller, E.R. Froesch, Recombinant human insulin-like growth factor I stimulates growth and has distinct effects on organ size in hypophysectomized rats. *Proc Natl Acad Sci USA*, 85, 4889-4893 (1988).
- A. Wickman, J. Isgaard, M.A. Adams, P. Friberg, Inhibition of nitric oxide in rats. Regulation of cardiovascular structure and expression of insulin-like growth factor I and its receptor messenger RNA. J Hypertens, 15, 751-759 (1997).
- G. Pete, Y. Hu, M. Walsh, J. Sowers and J.C. Dunbar, Insulin-like growth factor-I decreases mean blood pressure and selectively increases regional blood flow in normal rats. *Proc Soc Exp Biol Med*, 213, 187-192 (1996).
- M.F. Walsh, M. Barazi, G. Pete, R. Muniyappa, J.C. Dunbar and J.R. Sowers, Insulinlike growth factor I diminishes *in vivo* and *in vitro* vascular contractility: role of vascular nitric oxide. *Endocrinology*, **137**, 1798-1803 (1996).

160

- M.Y. Donath, R. Jenni, H.P. Brunner, M. Anrig, S. Kohli, Y. Glatz and E.R. Froesch, Cardiovascular and metabolic effects of insulin-like growth factor I at rest and during exercise in humans. *J Clin Endocrinol Metab*, 81, 4089-4094 (1996).
- M.Y. Donath, G. Sutsch, X.W. Yan, B. Piva, H.P. Brunner, Y. Glatz, J. Zapf, F. Follath, E.R. Froesch and W. Kiowski, Acute cardiovascular effects of insulin-like growth factor I in patients with chronic heart failure. *J Clin Endocrinol Metab*, 83, 3177-3183 (1998).
- F. Manelli, M. Volterrani, R. Lorusso, G. Romanelli and A. Giustina. Acute growth hormone effects in human subjects with heart failure. In: *Growth hormone and the heart*, F. Manelli (ed) Kluwer Academic Publishers, Boston, pp 87-96 (2001).
- R. Napoli, V. Guardasole, V. Angelini, F. D'Amico, E. Zarra, M. Matarazzo and L. Saccà, Acute effects of growth hormone on vascular function in human subjects. J Clin Endocrinol Metab, 88, 2817-2820 (2003).
- H. Tsukahara, D.V. Gordienko, B. Tonshoff, M.C. Gelato, M.S. and Goligorsky Direct demonstration of insulin-like growth factor-I-induced nitric oxide production by endothelial cells. *Kidney Int*, 45, 598-604 (1994).
- R. Muniyappa, M.F. Walsh, J.S. Rangi, R.M. Zayas, P.R. Standley, J.L. Ram and J.R. Sowers, Insulin like growth factor 1 increases vascular smooth muscle nitric oxide production. *Life Sci*, 61, 925-931 (1997).
- 32. J. Haylor, I. Singh and A.M. el Nahas, Nitric oxide synthesis inhibitor prevents vasodilation by insulin-like growth factor I. *Kidney Int*, **39**, 333-335 (1991).
- D.A. Fryburg, NG-monomethyl-L-arginine inhibits the blood flow but not the insulinlike response of forearm muscle to IGF- I: possible role of nitric oxide in muscle protein synthesis. J Clin Invest, 97, 1319-1328 (1996).
- 34. C.L. Oltman, N.L. Kane, D.D. Gutterman, R.S. Bar and K.C. Dellsperger, Mechanism of coronary vasodilation to insulin and insulin-like growth factor I is dependent on vessel size. *Am J Physiol Endocrinol Metab*, **279**, E176-181 (2000).
- 35. R. Hirschberg and J.D. Kopple, Evidence that insulin-like growth factor I increases renal plasma flow and glomerular filtration rate in fasted rats. *J Clin Invest*, **83**, 326-330 (1989).
- 36. D. Hasdai, R.A. Rizza, D.R. Holmes Jr., D.M. Richardson, P. Cohen and A. Lerman, Insulin and insulin-like growth factor-I cause coronary vasorelaxation *in vitro*. *Hypertension*, **32**, 228-34 (1998).
- P.R. Standley, F. Zhang, R.M. Zayas, R. Muniyappa, M.F. Walsh, E. Cragoe and J.R. Sowers, IGF-I regulation of Na<sup>+</sup>/K<sup>+</sup>-ATPase in rat arterial smooth muscle. *Am J Physiol*, 273, E113-121 (1997).
- G. Lembo, H.A. Rockman, J.J. Hunter, H. Steinmetz, W.J. Koch, L. Ma, M.P. Prinz, J. Ross Jr., K.R. Chien and L. Powell-Braxton, Elevated blood pressure and enhanced myocardial contractility in mice with severe IGF-1 deficiency. J Clin Invest, 98, :2648-2655 (1996).
- Å. Tivesten, E. Bollano, I. Andersson, S. Fitzgerald, K. Caidahl, K. Sjögren, J. Skøtt, J-L. Liu, D. LeRoith, R. Mobini, O.G.P. Isaksson, J-O. Jansson, C. Ohlsson, G. Bergström and J. Isgaard, Liver-derived insulin-like growth factor-I is involved in the regulation of blood pressure in nice. *Endocrinology*, 143, 4235-4242 (2002).
- 40. S.J. Fuller, J.R. Mynett and P.H. Sugden Stimulation of cardiac protein synthesis by insulin-like growth factors. *Biochem J*, **282**, 85-90 (1992).
- M.Y. Donath, J. Zapf, M. Eppenberger-Eberhardt, E.R. Froesch and H.M. Eppenberger Insulin-like growth factor I stimulates myofibril development and decreases smooth muscle alpha-actin of adult cardiomyocytes. *Proc Natl Acad Sci* USA, 91, 1686-1690 (1994).

- 42. H. Ito, M. Hiroe, Y. Hirata, M. Tsujino, S. Adachi, M. Shichiri, A. Koike, A. Nogami and F. Marumo Insulin-like growth factor-I induces hypertrophy with enhanced expression of muscle specific genes in cultured rat cardiomyocytes. *Circulation*, **87**, 1715-1721 (1993).
- C. Lu, G. Schwartzbauer, M.A. Sperling, S.U. Devaskar, S. Thamotharan, P.D. Robbins, C.F. McTiernan, J.L. Liu, J. Jiang, S.J. Frank and R.K. Menon. Demonstration of direct effects of growth hormone on neonatal cardiomyocytes. J Biol Chem, 276, 22892-900 (2001).
- 44. M.Y. Donath, M.A. Gosteli-Peter, C. Hauri, E.R. Froesch and J. Zapf, Insulin-like growth factor-I stimulates myofibrillar genes and modulates atrial natriuretic factor mRNA in rat heart. *Eur J Endocrinol*, **137**, 309-315 (1997).
- 45. A. Musaro, K.J. McCullagh, F.J. Naya, E.N. Olson, N. Rosenthal, IGF-1 induces skeletal myocyte hypertrophy through calcineurin in association with GATA-2 and NF-ATc1. *Nature*, **400**, 581-585 (1999).
- 46. E.E. Creemers, J.P. Cleutjens, J.F. Smits and M.J. Daemen, Matrix metalloproteinase inhibition after myocardial infarction: a new approach to prevent heart failure? *Circ Res*, **89**, 201-210 (2001).
- 47. RP Butt, GJ Laurent and JE Bishop, Collagen production and replication by cardiac fibroblasts is enhanced in response to diverse classes of growth factors. *Eur J Cell Biol*, **68**, 330-335 (1995).
- 48. A .Bruel and H. Oxlund, Biosynthetic growth hormone increases the collagen deposition rate in rat aorta and heart. *Eur J Endocrinol*, **132**, 195-199 (1995).
- 49. P. Schnabel, F. Mies, T. Nohr, M. Geisler and M. Bohm, Differential regulation of phospholipase C-beta isozymes in cardiomyocyte hypertrophy. *Biochem Biophys Res Commun*, **275**, 1-6 (2000).
- 50. A. Cittadini, H. Stromer, S.E. Katz, R. Clark, A.C. Moses, J.P. Morgan and P.S. Douglas, Differential cardiac effects of growth hormone and insulin-like growth factor-1 in the rat. A combined *in vivo* and *in vitro* evaluation. *Circulation*, **93**, 800-809 (1996).
- 51. A. Bruel, H. Oxlund and J.R. Nyengaard, Growth hormone increases the total number of myocyte nuclei in the left ventricle of adult rats. *Growth Horm* IGF Res, **12**, 106-115 (2002).
- 52. J. Kajstura, W. Cheng, R. Sarangarajan, P. Li, B. Li, J.A. Nitahara, S. Chapnick, K. Reiss, G. Olivetti and P. Anversa, Necrotic and apoptotic myocyte cell death in the aging heart of Fischer 344 rats. *Am J Physiol*, **271**, H1215-1228 (1996).
- 53. U. Vetter, C. Kupferschmid, D. Lang and S. Pentz, Insulin-like growth factors and insulin increase the contractility of neonatal rat cardiocytes *in vitro*. *Basic Res Cardiol*, **83**, 647-654 (1988).
- 54. N.S. Freestone, S. Ribaric and W.T. Mason, The effect of insulin-like growth factor-1 on adult rat cardiac contractility. *Mol Cell Biochem*, **163-164**, 223-229 (1996).
- 55. A. Cittadini, Y. Ishiguro, H. Stromer, M. Spindler, A.C. Moses, R. Clark, P.S. Douglas, J.S. Ingwall and J.P. Morgan, Insulin-like growth factor-1 but not growth hormone augments mammalian myocardial contractility by sensitizing the myofilament to Ca<sup>2+</sup> through a wortmannin-sensitive pathway: studies in rat and ferret isolated muscles. *Circ Res*, 83, 50-59 (1998).
- 56. H. Stromer, A. Cittadini, P.S. Douglas, J.P. Morgan, Exogenously administered growth hormone and insulin-like growth factor-I alter intracellular Ca<sup>2+</sup> handling and enhance cardiac performance. *In vitro* evaluation in the isolated isovolumic bufferperfused rat heart. *Circ Res*, **79**, 227-236 (1996).

162

#### Cardiovascular and Respiratory Systems

- 57. M. Tajima, E.O. Weinberg, J. Bartunek, H. Jin, R. Yang, N.F. Paoni and B.H. Lorell, Treatment with growth hormone enhances contractile reserve and intracellular calcium transients in myocytes from rats with postinfarction heart failure. *Circulation*, **99**, 127-134 (1999).
- A .Cittadini, H. Stromer, D.E. Vatner, J.D. Grossman, S.E. Katz, R. Clark, J.P. Morgan and P.S. Douglas, Consequences of growth hormone deficiency on cardiac structure, function, and beta-adrenergic pathway: studies in mutant dwarf rats. *Endocrinology*, 138, 5161-5169 (1997).
- 59. H. Stromer, A. Cittadini, J.D. Grossma, P.S. Douglas and J.P. Morgan, Intrinsic cardiac muscle function, calcium handling and beta -adrenergic responsiveness is impaired in rats with growth hormone deficiency. *Growth Horm IGF Res*, **9**, 262-271 (1999).
- J. Ren and H. Brown-Borg, Impaired cardiac excitation-contraction coupling in ventricular myocytes from Ames dwarf mice with IGF-I deficiency. Growth Horm IGF Res;12:99-105 (2002).
- 61. S. Kinugawa, H. Tsutsui, T. Ide, R. Nakamura, K. Arimura, K. Egashira and A. Takeshita, Positive inotropic effect of insulin-like growth factor-1 on normal and failing cardiac myocytes. *Cardiovasc Res*, **43**, 157-164 (1999).
- 62. M.L. Solem, A.P. Thomas, Modulation of cardiac Ca<sup>2+</sup> channels by IGF1. *Biochem Biophys Res Commun*, 252, 151-155 (1998).
- X.P. Xu and P.M. Best, Decreased transient outward K<sup>+</sup>current in ventricular myocytes from acromegalic rats. Am J Physiol 1991;260:H935-942
- 64. H. Jin, R. Yang, H. Lu, A.K. Ogasawara, W. Li, A. Ryan, F. Peale and N.F. Paoni, Effects of early treatment with growth hormone on infarct size, survival, and cardiac gene expression after acute myocardial infarction. *Growth Horm IGF Res*, **12**, 208-215 (2002).
- 65. T. Ueyama, T. Ohkusa, M. Yano and M. Matsuzaki, Growth hormone preserves cardiac sarcoplasmic reticulum Ca<sup>2+</sup> release channels (ryanodine receptors) and enhances cardiac function in cardiomyopathic hamsters. *Cardiovasc Res*, **40**, 64-73 (1998).
- T.J. Caperna and D. Gavelek, Influences of somatotropin on Na<sup>+</sup>/K<sup>+</sup>-ATPase, Mg(2<sup>+</sup>)-ATPase and Ca(2<sup>+</sup>)-ATPases of Porcine visceral tissues. *Comp Biochem Physiol B Biochem Mol Biol*, 115, 13-18 (1996).
- E. Mayoux, R. Ventura-Clapier, J. Timsit, F. Behar-Cohen, C. Hoffmann and J.J. Mercadier, Mechanical properties of rat cardiac skinned fibers are altered by chronic growth hormone hypersecretion. *Circ Res*, 72, 57-64 (1993).
- 68. J. Ren, Short-term administration of insulin-like growth factor I (IGF-1) does not induce myocardial IGF-1 resistance. *Growth Horm IGF Res*, **12**, 162-168 (2002).
- 69. J. Timsit, B. Riou, J. Bertherat, C. Wisnewsky, N.S. Kato, A.S. Weisberg, J. Lubetzki, Y. Lecarpentier, S. Winegrad and J.J. Mercadier, Effects of chronic growth hormone hypersecretion on intrinsic contractility, energetics, isomyosin pattern, and myosin adenosine triphosphatase activity of rat left ventricle. *J Clin Invest*, **86**, 507-515 (1990).
- S. Longobardi, A. Cittadini, H. Stromer, S.E. Katz, J.D. Grossman, R.G. Clark, J.P. Morgan and P.S. Douglas, Echocardiographic assessment of cardiac morphology and function in mutant dwarf rats. *Growth Horm IGF Res*, 10, 242-247 (2000).
- A. Cittadini, A. Cuocolo, B. Merola, S. Fazio, D. Sabatini, E. Nicolai, A. Colao, S. Longobardi, G. Lombardi and L. Sacca, Impaired cardiac performance in GH-deficient adults and its improvement after GH replacement. *Am J Physiol*, 267, E219-225 (1994).

- 72. K. Caidahl, S. Eden and B.A. Bengtsson, Cardiovascular and renal effects of growth hormone. *Clin Endocrinol (Oxf)*, **40**, 393-400 (1994).
- 73. L. Brown, B. Wyse and C. Sernia, Adrenoceptor-mediated cardiac and vascular responses in genetically growth hormone-deficient rats. *Biochem Pharmacol*, **45**, 2223-2229 (1993).
- 74. G. Amato, C. Carella, S. Fazio, G. La Montagna, A. Cittadini, D. Sabatini, C. Marciano-Mone, L. Sacca and A. Bellastella, Body composition, bone metabolism, and heart structure and function in growth hormone (GH)-deficient adults before and after GH replacement therapy at low doses. J Clin Endocrinol Metab, 77, 1671-1676 (1993).
- S. Fazio, A. Cittadini, D. Sabatini, B. Merola, A. Colao, B. Biondi, S. Longobardi, G. Lombardi and L. Sacca, Growth hormone and heart performance. A novel mechanism of cardiac wall stress regulation in humans. *Eur Heart J*, 18, 340-347 (1997).
- 76. T. Rosen and B.A. Bengtsson, Premature mortality due to cardiovascular disease in hypopituitarism. Lancet;336:285-288 (1990).
- T. Rosen, S. Eden, G. Larson, L. Wilhelmsen and B.A. Bengtsson, Cardiovascular risk factors in adult patients with growth hormone deficiency. *Acta Endocrinol (Copenh)*, 129, 195-200 (1993).
- 78. J.A. Barreto-Filho, M.R. Alcantara, R. Salvatori, M.A. Barreto, A.C. Sousa, V. Bastos, A.H. Souza, R.M. Pereira, P.E. Clayton, M.S. Gill and M.H. Aguiar-Oliveira, Familial isolated growth hormone deficiency is associated with increased systolic blood pressure, central obesity, and dyslipidemia. *J Clin Endocrinol Metab*, 87, 2018-2023 (2002).
- 79. D.G. Penney, J.C. Dunbar, Jr. and M.S. Baylerian, Cardiomegaly and haemodynamics in rats with a transplantable growth hormone-secreting tumour. *Cardiovasc Res*, **19**, 270-277 (1985).
- M.C. Delaughter, G.E. Taffet, M.L. Fiorotto, M.L. Entman and R.J. Schwartz, Local insulin-like growth factor I expression induces physiologic, then pathologic, cardiac hypertrophy in transgenic mice. *Faseb J*, 13, 1923-1929 (1999).
- E. Bollano, E. Omerovic, M. Bohlooly-y, V. Kujacic, B. Madhu, J. Tornell, O. Isaksson, B. Soussi, W. Schulze, M.L. Fu, G. Matejka, F. Waagstein and J. Isgaard, Impairment of cardiac function and bioenergetics in adult transgenic mice overexpressing the bovine growth hormone gene. *Endocrinology*, 141, 2229-2235 (2000).
- 82. E Omerovic, M Basetti, E Bollano, M Bohlooly, J Tornell, J Isgaard, A Hjalmarson, B Soussi and F. Waagstein, *In vivo* metabolic imaging of cardiac bioenergetics in transgenic mice. *Biochem Biophys Res Commun*, **271**, 222-228 (2000).
- 83. M. Losa and K. von Werder, The heart in acromegaly. In: F. Manelli, ed. Growth hormone and the heart. Kluwer Academic Publishers, Boston, pp 33-43 (2001).
- Y.M. Bohlooly, L. Carlson, B. Olsson, H. Gustafsson, I.J. Andersson, J. Tornell and G. Bergstrom. Vascular function and blood pressure in GH transgenic mice. *Endocrinology*, 142, 3317-3323 (2001).
- M.L. Jaffrain-Rea, C. Moroni, R. Baldelli, C. Battista, P. Maffei, M. Terzolo, M. Correra, M.R. Ghiggi, E. Ferretti, A. Angeli, N. Sicolo, V. Trischitta, A.Liuzzi, R. Cassone and G. Tamburrano 2001 Relationship between blood pressure and glucose tolerance in acromegaly. *Clin Endocrinol (Oxf)*, 54, 189-195 (2001).
- E.A. Palmieri, G. Benincasa, F. Di Rella, C. Casaburi, M.G. Monti, G. De Simone, L. Chiariotti, L. Palombini, C.B. Bruni, L. Sacca and A. Cittadini, Differential expression of TNF-alpha, IL-6, and IGF-1 by graded mechanical stress in normal rat myocardium. *Am J Physiol Heart Circ Physiol*, 282, H926-934 (2002).

164

- P.R. Standley, T.J. Obards and C.L. Martina, Cyclic stretch regulates autocrine IGF-I in vascular smooth muscle cells: implications in vascular hyperplasia. *Am J Physiol*, 276, E697-705 (1999).
- H. Wahlander, J. Isgaard, E. Jennische and P. Friberg, Left ventricular insulin-like growth factor I increases in early renal hypertension. *Hypertension*, **19**, 25-32 (1992).
- G. Guron, P. Friberg, A. Wickman, C. Brantsing, B. Gabrielsson and J. Isgaard, Cardiac insulin-like growth factor I and growth hormone receptor expression in renal hypertension. *Hypertension*, 27, 636-642 (1996).
- R. Ebensperger, E. Acevedo, J. Melendez, R. Corbalan, M. Acevedo, M. Sapag-Hagar, J.E. Jalil and S. Lavandero, Selective increase in cardiac IGF-1 in a rat model of ventricular hypertrophy. *Biochem Biophys Res Commun*, 243, 20-24 (1998).
- G. Andronico, M.T. Mangano, E. Nardi, G. Mule, G. Piazza and G. Cerasola, Insulinlike growth factor 1 and sodium-lithium countertransport in essential hypertension and in hypertensive left ventricular hypertrophy. *J Hypertens*, **11**, 1097-1101 (1993).
- M. Galderisi, G. Vitale, G. Lupoli, M. Barbieri, G. Varricchio, C. Carella, O. de Divitiis, G. Paolisso Inverse association between free insulin-like growth factor-1 and isovolumic relaxation in arterial systemic hypertension. *Hypertension*, 38, 840-845 (2001).
- P. Verdecchia, G. Reboldi, G. Schillaci, C. Borgioni, A. Ciucci, M.P. Telera, F. Santeusanio, C. Porcellati and P. Brunetti, Circulating insulin and insulin growth factor-1 are independent determinants of left ventricular mass and geometry in essential hypertension. *Circulation*;100:1802-1807 (1999).
- C. Lemne and K. Brismar, Insulin-like growth factor binding protein-1 as a marker of the metabolic syndrome-a study in borderline hypertension. *Blood Press*, 7, 89-95 (1998).
- E. Dall'Aglio, I. Salimbeni, A. Rocci, S. Mazzoni, F. Corradi, E. Cattadori, S. Visioli,
  A. Banchini, G. Valenti and G.P. Ceda, Modifications of blood pressure and IGF-I levels after weight loss in obesity. *J Endocrinol Invest*, 25, 107-109 (2002).
- 96. L. Friberg, S. Werner, G. Eggertsen and S. Ahnve, Growth hormone and insulin-like growth factor-1 in acute myocardial infarction. *Eur Heart J*, **21**, 1547-1554 (2000).
- W.L. Lee, J.W. Chen, C.T. Ting, S.J. Lin and P.H. Wang, Changes of the insulin-like growth factor I system during acute myocardial infarction: implications on left ventricular remodeling. *J Clin Endocrinol Metab*, 84, 1575-1581 (1999).
- A. Juul, T. Scheike, M. Davidsen, J. Gyllenborg and T. Jorgensen, Low serum insulinlike growth factor I is associated with increased risk of ischemic heart disease: a population-based case-control study. *Circulation*, **106**, 939-944 (2002).
- N. Vaessen, P. Heutink, J.A. Janssen, J.C. Witteman, L. Testers, A. Hofman, S.W. Lamberts, B.A. Oostra, H.A. Pols and C.M. van Duijn, A polymorphism in the gene for IGF-I: functional properties and risk for type 2 diabetes and myocardial infarction. *Diabetes*, 50, 637-642 (2001).
- 100. R.V. Krishnamurthi, L. Maxwell, N.S. Bassett, J.B. Gavin, P.D. G and B.M. Johnston, The spatial and temporal distribution of insulin-like growth factor-1 following experimental myocardial infarction in the rat. *Cardiovasc Pathol*, **6**, 197-203 (1997).
- 101. R. Yang, S. Bunting, N. Gillett, R. Clark and H. Jin, Growth hormone improves cardiac performance in experimental heart failure. *Circulation*, **92**, 262-267 (1995).
- 102. J. Isgaard, V. Kujacic, E. Jennische, A. Holmäng, X.Y. Sun, T. Hedner, Å. Hjalmarson and B-Å. Bengtsson, Growth hormone improves cardiac function in rats with experimental myocardial infarction. *Eur J Clin Invest*, **27**, 517-525 (1997).

- H. Jin, R. Yang, N. Gillett, R.G. Clark, A. Ko and N.F. Paoni, Beneficial effects of growth hormone and insulin-like growth factor-1 in experimental heart failure in rats treated with chronic ACE inhibition. *J Cardiovasc Pharm*, 26, 420-425 (1995).
- E. Omerovic, R. Fahrami, M. Basetti, B. Soussi, Å. Hjalmarson, F. Waagstein and J.Isgaard, Growth hormone improves bioenergetics and decreases catecholamines in postinfarct rat hearts. *Endocrinology*, **141**, 4592-4599 (2000).
- A. Cittadini, J. Isgaard, M.G. Monti, C. Casaburi, A.D. Di Gianni, R. Serpico and R. Saccà, Growth hormone prolongs survival and attenuates interstitial remodeling in postinfarction heart failure. *J Am Coll Cardiol*, 41, 2154-2163 (2003).
- M. Volterrani, P. Desenzani, R. Lorusso, A. d'Aloia, F. Manelli and A. Giustina, Haemodynamic effects of intravenous growth hormone in congestive heart failure. *Lancet*, 349, 1067-1068 (1997).
- S. Fazio, D. Sabatini, B. Capaldo, C. Vigorito, A. Giordano, R. Guida, F. Pardo, B. Biondi and L. Sacca, A preliminary study of growth hormone in the treatment of dilated cardiomyopathy. *N Engl J Med*, 334, 809-814 (1996).
- 108. K.J. Osterziel, O. Strohm, J. Schuler, M. Friedrich, D. Hanlein, R. Willenbrock, S.D. Anker, P.A. Poole-Wilson, M.B. Ranke and R. Dietz, Randomised, double-blind, placebo-controlled trial of human recombinant growth hormone in patients with chronic heart failure due to dilated cardiomyopathy. *Lancet*, **351**, 1233-1237 (1998).
- J. Isgaard, C.H. Bergh, K. Caidahl, M. Lomsky, A. Hjalmarson and B.A. Bengtsson, A placebo-controlled study of growth hormone in patients with congestive heart failure. *Eur Heart J*, 19, 1704-1711 (1998).
- S.D. Anker, M. Volterrani, C.D. Pflaum, C.J. Strasburger, K.J. Osterziel, W. Doehner, M.B. Ranke, P.A. Poole-Wilson, A. Giustina, R. Dietz and A.J. Coats, Acquired growth hormone resistance in patients with chronic heart failure: implications for therapy with growth hormone. *J Am Coll Cardiol*, 38, 443-452 (2001).
- 111. T. Saaresranta and O. Polo, Hormones and breathing. Chest, 122, 2165-82 (2002).
- 112. B.G. Cooper, J.E. White, L.A. Ashworth, K.G. Alberti and G.J. Gibson, Hormonal and metabolic profiles in subjects with obstructive sleep apnea syndrome and the effects of nasal continuous positive airway pressure (CPAP) treatment. *Sleep*, **18**, 172-183 (1995).
- 113. T.W. Felbinger, U. Suchner, A.E. Goetz, J. Briegel and K. Peter, Recombinant human growth hormone for reconditioning of respiratory muscle after lung volume reduction surgery. *Crit Care Med*, **27**, 1634-1638 (1999).
- 114. G.S. Pape, M. Friedman, L.E. Underwood and D.R. Clemmons, The effect of growth hormone on weight gain and pulmonary function in patients with chronic obstructive lung disease. *Chest*, **99**, 1495-1500 (1991).
- 115. R. Arens, D. Gozal, K.J. Omlin, J. Vega, K.P. Boyd, T.G. Keens and M.S. Woo, Comparison of high frequency chest compression and conventional chest physiotherapy in hospitalized patients with cystic fibrosis. Am J Respir Crit Care Med, 150, 1154-1157 (1994).
- 116. A.C. Lindgren, L.G. Hellstrom, E.M. Ritzen and J. Milerad, Growth hormone treatment increases CO(2) response, ventilation and central inspiratory drive in children with Prader-Willi syndrome. *Eur J Pediatr*, **158**, 936-940 (1999).

166

# Chapter 7

### **HEMATOPOIESIS**

Wilson Savino<sup>1,2</sup>, Salete Smaniotto<sup>1,2,3</sup> and Mireille Dardenne<sup>2</sup> <sup>1</sup>Laboratory on Thymus Research, Department of Immunology, Oswaldo Cruz Institute, Oswaldo Cruz Foundation, Rio de Janeiro, Brazil; <sup>2</sup>CNRS UMR-8147, Université Paris V, Hôpital Necker, Paris, France; <sup>3</sup>Department of Morphology, Federal University of Alagoas, Maceió, Brazil

Key words: Apoptosis; bone marrow; chemokines; cytokines; extracellular matrix; growth hormone; hematopoiesis; insulin-like growth factors; integrins; thymic epithelial cells; thymic nurse cells; thymocytes; thymulin; thymus.

#### 1. **INTRODUCTION**

The crosstalk between the neuroendocrine and hematopoietic systems is now well established. Similar ligands and receptors are used by these systems, providing physiological intra- and intersystem communication circuitry, which seems to play a relevant role in homeostasis. Increasing evidence has placed growth hormone (GH) and insulin-like growth factors (IGFs) among potent modulators of the hematopoietic system, in both health and disease (reviewed by Venters<sup>1</sup> and Zumkeller<sup>2</sup>). Herein, we focus on the effects of GH and IGF-I upon hematopoietic organs including bone marrow, thymus, and peripheral lymphoid organs. Nevertheless, before discussing such effects, it seems worthwhile to summarize background data concerning the general features of hematopoiesis, in terms of developing cells as well as the corresponding microenvironments in which differentiation takes place.

#### 2. HEMATOPOIETIC MICROENVIRONMENTS AND THEIR ROLE IN HEMATOPOIESIS

Bone marrow and thymus are the primary hematopoietic organs responsible for the generation of myeloid and lymphoid precursors of the various cellular blood components. As briefly described in the following, these series of complex proliferation/differentiation and death processes occur in the context of specific microenvironments in each organ.

In the bone marrow, cell differentiation occurs from a common stem cell that progressively gives rise to further precursors, now committed with the distinct cell lineages, as illustrated in Fig. 1.



*Figure 1.* Hematopoiesis in the bone marrow. In this simplified scheme, we can see that starting from a common hemopoietic stem cell (HSC), both lymphoid and non-lymphoid blood components are generated in the bone marrow. As regards lymphoid cells, a common progenitor differentiates into a T-cell precursor (that leaves the organ toward the thymus) and a B-lymphoblast that will eventually differentiate within the bone marrow into a mature B lymphocyte. Intermediate precursors for the monocyte/granulocyte series will further differentiate and give rise to monocytes and neutrophils. Other precursors are committed to the generation of eosinophils or basophils (not shown). In addition, distinct HSC-derived precursors can differentiate toward the megakaryocytic series, eventually giving rise to platelets or the erythroid precursors that will differentiate into erythrocytes. Once each mature cell in the bone marrow is formed, it will leave the organ to gain the blood vessel.

#### *Hematopoiesis*

Such progressive differentiation can be traced morphologically, phenotypically, and functionally. For example, CD34 is a membrane protein typically found in bone marrow precursors of the various lineages, including lymphoid, macrophage/granulocytic, megakaryocytic, and erythrocytic. By contrast, coexpression of CD34 plus CD2 denotes a pre-T commitment, whereas CD34<sup>+</sup>CD19<sup>+</sup> precursors will progress toward B lymphocyte differentiation. Immature B cells will rearrange their immunoglobulin genes and the resulting products will form the core of the B-cell receptor. Ultimately, cells formed in the bone marrow leave the organ migrating through a sinusoid capillary wall, and gaining the blood flow.

Maturation of the distinct cell lineages in the bone marrow largely depends on the local interaction with microenvironmental components, which comprise reticular cells, endothelial cells, fibroblasts. and macrophages, as well as adipocytes and osteoblasts. Accordingly, hematopoiesis is influenced by a large number of soluble products secreted by the microenvironmental cells, including typical cytokines, growth factors and chemokines, comprising among others, interleukin (IL)-1, IL-3, stem cell factor, colony stimulating factors for granulocytes (G-CSF), and granulocyte/macrophages (GM-CSF), and the chemokine CXCL12 (reviewed by Metcalf<sup>3</sup>). In addition, the hormone erythropoietin, normally produced in the kidney, is crucial for the development of the erythroid lineage, whereas thrombopoietin is important for the production of megacaryocytes, and ultimately for platelet differentiation.

Important is the concept that distinct combinations of cytokines in the milieu generate the preferential expansion and differentiation of specific cell lineages. For example, the presence of IL-3, stem cell factor, plus GM-CSF directs differentiation toward the granulocyte/macrophage lineages, whereas IL-3 plus GM-CSF alone tends to induce proliferation/commitment of eosinophilic precursors. In addition, chemokines are involved in bone marrow physiology, as exemplified by the role of CXCL12 and its receptor CXCR4 in stem cell mobilization<sup>4</sup>.

Cell–cell interactions are also crucial for normal hematopoiesis in the bone marrow. For example, membrane receptors such as the syalomucin CD34; the proteoglycan CD44 (that binds to hyaluronic acid and fibronectin); and the integrin  $\alpha 4\beta 1$  (CD49d/CD29), which binds fibronectin and the vascular cell adhesion molecule, are relevant in the mobilization of CD34<sup>+</sup> hematopoietic precursors<sup>4,5</sup>. It is noteworthy that such an important event is also under control of matrix proteinases, including MMP2 and MMP9 (see review<sup>4</sup>).

In the thymus, bone marrow-derived T-cell precursors undergo differentiation (Fig. 2A), ultimately leading to migration of positively selected thymocytes to the T-cell-dependent areas of peripheral lymphoid organs<sup>6</sup>. Such a process involves sequential expression of various proteins and rearrangements of the T-cell receptor (TCR) genes. The most immature thymocytes express neither the TCR complex nor the CD4 or CD8 accessory molecules, being called double-negative thymocytes. Maturation progresses with the initiation of rearrangements of TCR genes and then acquisition of CD4 and CD8 markers, yielding the CD4<sup>+</sup>CD8<sup>+</sup> cells. Productive rearrangements result in the membrane expression of the TCR (complexed with the CD3) in low densities (TCR<sup>low</sup>). Thymocytes that do not undergo a productive TCR gene rearrangement die by apoptosis, whereas those expressing productive TCR will interact with peptides presented by molecules of the major histocompatibility complex (MHC), expressed on microenvironmental cells. This interaction will determine the positive and negative selection events, crucial for normal thymocyte differentiation. Positively selected thymocytes progress to the mature TCR<sup>high</sup>CD4<sup>+</sup>CD8<sup>-</sup> or TCR<sup>high</sup>CD4<sup>-</sup>CD8<sup>+</sup> single positive stage, and ultimately leave the organ to form the large majority of the peripheral T cell repertoire<sup>6</sup>.

Thymocyte differentiation occurs as cells migrate within the thymic lobules: TCR<sup>-</sup>CD4<sup>-</sup>CD8<sup>-</sup> and TCR<sup>+</sup>CD4<sup>+</sup>CD8<sup>+</sup> are cortically located, whereas mature TCR<sup>+</sup>CD4<sup>+</sup>CD8<sup>-</sup> and TCR<sup>+</sup>CD4<sup>-</sup>CD8<sup>+</sup> cells are found in the medulla.

Along their way, thymocytes interact with components of the thymic microenvironment, a tridimensional network formed of thymic epithelial cells (TEC), macrophages, dendritic cells, fibroblasts, and extracellular matrix<sup>6,7</sup>. The TEC network is the major component of the thymic microenvironment, being heterogeneous in morphology and phenotype. One lymphoepithelial complex, the thymic nurse cell, corresponds to a multicellular structure formed by one epithelial cell and variable numbers of thymocytes (mostly CD4<sup>+</sup>CD8<sup>+</sup> cells), being located in the cortical region of thymic lobules. Once obtained TNCs can be used as an in vitro model of thymocyte migration in the TEC context. In addition to the interaction involving the TCR/peptide-MHC, in the context of CD8 or CD4 molecules, the thymic microenvironment influences thymocyte maturation via adhesion molecules (Fig. 2B). For example, TEC express intercellular adhesion molecule (ICAM)-1 and lymphocyte function-associated antigen (LFA)-3, which respectively bind to LFA-1 and CD2 present on thymocyte membranes<sup>6</sup>. Moreover, TEC-thymocyte interactions can be mediated by various extracellular matrix (ECM) moieties, and this is relevant for thymocyte migration $^{7,9}$ .



Figure 2. Thymopoiesis and the thymic microenvironment. Panel A shows, on the left side, one thymic lobule with a simplified view of thymocyte differentiation. Bone marrow-derived precursor cells enter the organ and localize in the outer cortex. At this stage they do not express the CD3-T-cell receptor complex, or the accessory molecules CD4 or CD8. As they differentiate, they acquire all these markers and become CD<sup>+</sup>CD4<sup>+</sup>CD8<sup>+</sup> cells, being located in the inner cortex. Those cells undergoing positive selection migrate from the cortex (C) to the medulla (M) and become either CD4<sup>+</sup> or CD8<sup>+</sup> single positive cells. These cells are able to leave the organ. The right part of panel A shows that thymocyte differentiation occurs in the context of the thymic microenvironment, composed by various epithelial cells, including the thymic nurse cells, as well as macrophages, dendritic cells, and fibroblasts (see the legend underneath the figure for each cell type). When migrating, differentiating thymocytes encounter microenvironmental cells with which they interact. Some of these cellular interactions are seen in panel **B**, where we can see that they interact with each other via soluble products, produced either by the microenvironmental cell (a) or the developing thymocyte (b). In addition, membrane interactions take place, such as those involving the TCR/CD3 complex with the major histocompatibility complex coupled with an endogenous peptide (c), crucial for defining positive versus negative selection. In addition, they can interact by means of classical adhesion molecules (d) and through extracellular matrix ligands and receptors (e).

Microenvironmental cells also modulate thymocyte differentiation by soluble polypeptides, comprising: (1) cytokines, such as interleukin-1 (IL-1), IL-3, IL-6, IL-7, IL-8, and stem cell factor; (2) chemokines, as for example CXCL12, CXCL10 and CCL25; and (3): thymic hormones, including thymulin, thymopoietin and thymosin- $\alpha 1^{6,10-12}$ .

In addition to terminal differentiation of B cells (including hypermutation of immunoglobulin genes and further negative selection events), activationinduced proliferation and plasma cell differentiation occur in the lymphoid follicles of the various peripheral lymphoid organs, whereas antigen-induced T-cell proliferation takes place in the so-called T-cell zones of these organs, for instance, in the paracortical region of lymph nodes. Again, microenvironmental cells of these tissues are important in lymphocyte proliferation, differentiation, and migration.

#### 3. GH AND IGF-I ENHANCE PRODUCTION OF SOLUBLE PEPTIDES BY HEMATOPOIETIC MICROENVIRONMENTS

Particularly in the thymus, accumulating data indicate that the GH/IGF-I axis influences the production of a variety of cytokines, including the chemokine family. Exogenous GH increases the *in vitro* production of IL-1 $\alpha$  and IL-1 $\beta$  by bovine thymic microenvironmental cells<sup>13</sup>. In addition, IL-6 secretion is augmented by GH, an effect partially exerted through the IL-1 production pathway. Interestingly, IL-6 production by thymocytes is also upregulated by *in vivo* injection of GH in aging animals<sup>14</sup>. More recently, we showed that CXCL12 production by TEC is enhanced, at both mRNA and protein levels, in GH-transgenic mice<sup>15</sup>.

*In vivo* and *in vitro* fluctuations in GH or IGF-I levels modulate thymic hormone secretion. Thymulin levels are low in dwarf mice and in children bearing deficient GH production, whereas GH treatment consistently restored this thymic endocrine function, as early as 24 hours after injection<sup>16-18</sup>. On the other hand, elevated thymulin serum levels have been detected in acromegalic patients, and were correlated with the circulating IGF-1 levels<sup>19</sup>. It is interesting to note that both thymulin and GH serum levels decrease along with age. In this respect, *in vivo* injection of GH in ageing animals significantly increases thymulin serum levels.<sup>14</sup>

Peripheral blood cells can also be modulated by the GH/IGF-I axis, as regards secretory products. One example is the GH or IGF-I-induced enhancement of secretion of superoxide anion by neutrophils and macrophages, which is important for their ability to kill bacteria (reviewed by Venters<sup>1</sup>).

#### 4. GH/IGF-I EFFECTS ON PROLIFERATION AND DEATH OF HEMATOPOIETIC CELLS

Several data indicate that the GH/IGF-I axis stimulates erythropoiesis (see review Halvorsen and Bechesteen<sup>20</sup>), in particular when the levels of erythropoietin are low or absent<sup>21</sup>. Such effects are actually abrogated with anti-IGF-I receptor antibodies<sup>22,23</sup>, which is in keeping with the demonstration of IGF-I receptors in cells of the erythroid lineage<sup>24</sup>. Interestingly, administration of IGF-I in patients suffering from Laron syndrome (primary GH insensitivity) results in a strong stimulatory effect on erythropoiesis<sup>25</sup>. In the same vein, erythropoiesis, which is impaired in adult GH-deficiency, is stimulated once the patients undergo GH treatment<sup>26</sup>. The monocytic lineage is also influenced by the GH/IGF-I axis, since *in vitro* proliferation of bone marrow-derived macrophages is enhanced by exogenous IGF-I<sup>27</sup>.

The expression of proliferation-related genes such as cyclins A and E (respectively controlling the entry into and transition through the S phase of the cell cycle) is enhanced in GH receptor expressing Ba/F3 pro-B cell line after GH treatment<sup>28</sup>. Furthermore, it has been shown that in 32D myeloid precursor cell line, as well as in Ba/F3 cells, IGF-I synergizes with IL-4 to enhance cell proliferation, an effect that requires MAPK and STAT-6 signalling pathways<sup>29</sup>. Furthermore, it has been shown that IGF-I synergizes with IL-7 in inducing proliferation of pro-B cells<sup>30</sup>. These data are in keeping with those showing that IGF-I bone marrow stimulates B lymphopoiesis<sup>31</sup>.

Studies performed in 32D cells also revealed that in some conditions, activation of the IGF-I receptor results in differentiation towards the granulocytic pathway similar to the granulocyte colony-stimulating factor<sup>32</sup>.

Proliferation of thymocytes was enhanced when they were exposed *exvivo* to supernatants from GH-treated TEC cultures<sup>33</sup>. In addition, GH itself synergizes with anti-CD3 in its direct stimulatory effect on thymocyte proliferation<sup>34</sup>. In the same vein, an enhanced concanavalin-A mitogenic response and IL-6 production by thymocytes is observed in GH-treated aging rats<sup>14</sup>, as well as in IGF-I-treated mice<sup>35</sup>. Finally, it is noteworthy that human thymocyte-derived GH-induced proliferation in thymocyte suspensions is apparently mediated by IGF-I<sup>36</sup>. As further discussed in the following sections, these findings indicate an autocrine/paracrine influence of the GH/IGF-I axis in the physiology of the thymus.

In addition to the data obtained in cell culture or *ex-vivo* conditions, a series of *in vivo* observations further supports the notion that GH and IGF-I exert enhancing effects upon hematopoiesis, and more particularly, thymopoiesis. Injections of GH3 cells (a pituitary adenoma cell line, able to produce GH) to old rats reversed the age-dependent thymic atrophy, with a consequent increase in thymocyte numbers<sup>37</sup>. Similar results were obtained with injection of IGF-I<sup>38</sup>. In addition, it was shown that cyclosporin A-induced thymic atrophy was restored by *in vivo* treatment with recombinant GH or IGF-I<sup>39</sup>, and that IGF-I was able to induce repopulation of the atrophic thymus from diabetic rats<sup>38</sup>. These findings are supported by a clinical case of an acromegalic patient with high circulating levels of GH and IGF-I associated with thymic hyperplasia<sup>40</sup>. Moreover, as seen in Fig. 3, transgenic mice overexpressing GH or GH-releasing hormone exhibit overgrowth of the thymus and peripheral lymphoid organs.



*Figure 3.* Increase of cell numbers in primary and secondary lymphoid organs in GH-transgenic mice. Considering that GH-transgenic animals are larger in size and weight, we calculated cell numbers per gram of body weight of each group. In thymus, spleen, and subcutaneous lymph nodes (SCLN) relative cells numbers of GH transgenic mice (GH tg) were higher than the corresponding wild-type controls (Wt). \*\*p < 0.01, as ascertained by unpaired Student's t-test.

Furthermore, short-term intrathymic injection of GH in mice, as well as long-term treatment with GH (in association with classical therapy) in HIV-positive patients, promoted an increase in the size of the thymus<sup>41,15</sup>. Also, GH or IGF-I administration together with bone marrow cells into syngeneic old recipients resulted in an increase of thymus cellularity, when compared

to the transfer of bone marrow cells alone<sup>42</sup>. In the same study, the authors showed that IGF-I potentiates the colonization of fetal thymus organ cultures with T-cell precursors, clearly indicating that IGF-I enhances the entrance of cell precursors into the thymus. This is in keeping with the replenishment of the thymus in cats infected with feline immunodeficiency virus. These animals exhibit a severe cortical thymocyte depletion, which is restored after long term treatment with IGF-I<sup>43</sup>. GH/IGF-I-induced cell proliferation is not restricted to primary lymphoid organs: exogenous GH and IGF-I increase cellularity in the spleen, an effect that can be observed not only in fetal age but also in aging animals<sup>37,44</sup>. This is in keeping with the specific increase in cell numbers in spleen and lymph nodes of GH-transgenic mice, as compared to their age-matched wild-type counterparts (Fig. 3).

In keeping with this, GH exhibits a synergistic proliferative effect on concanavalin A- or anti-CD3-stimulated spleen-derived T cells<sup>34</sup>, thus telling us that the periphery of the hematopoietic system is also a target for the GH/IGF-I axis. IGF-I also stimulates peripheral CD4<sup>+</sup> T cells as well as B lymphocytes in the spleen (reviewed by Zumkeller<sup>2</sup>).

In addition to enhancing cell proliferation, a role of the GH/IGF-I in modulating apoptosis has been documented. For example, it has been shown that GH prevents apoptosis in the Ba/F3 cell line<sup>45,46</sup>, and so has been shown for IGF-I, which actually enhanced the expression of the anti-apoptotic molecule Bcl-xL<sup>47</sup>. Moreover, in the rat Nb2 T cell lymphoma, GH inhibits the pro-apoptotic effect of dexamethasone, in a dose-dependent manner. Lastly, DNA breakdown in human erythroid colony-forming cells is largely reduced by IGF-I. Taken together, the data summarized in the foregoing strongly indicate that the ultimate enhancing impact of the GH/IGF-I axis in hematopoiesis is a combined effect of increase in cell proliferation plus decrease in programmed cell death.

#### 5. THE GH/IGF-I AXIS MODULATE THE TRAFFIC OF HEMOPOIETIC CELLS?

The traffic of hematopoietic cells is also a target for the GH/IGF-I axis. Although the literature remains poor concerning normal bone marrow cells, it has been shown that exogenous recombinant GH in aging or irradiated mice not only induces hematopoiesis, but also mobilizat on of hematopoietic precursors and significant increase in blood cell counts<sup>48</sup>.

Bone marrow stromal cell-derived IGF-I can play a role of chemoattractant for multiple myeloma cells<sup>49</sup>. In addition, IGF-I induces multiple myeloma cell adhesion to and migration through fibronectin via integrin-type fibronectin receptor stimulation<sup>50</sup>, which fits with the data

showing that IGF-I enhanced transendothelial migration of these cells<sup>51</sup>. Recombinant GH increases human T-cell engraftment into the thymus of severe combined immunodeficiency (SCID) mice, an effect that seems to be mediated by adhesion molecules and extracellular matrix, since it can be abrogated with anti- $\beta$ 1 or anti- $\beta$ 2 integrin antibodies<sup>52</sup>. Moreover, as mentioned earlier, it has been demonstrated that IGF-I potentiates the colonization of fetal thymus organ cultures with T cell precursors<sup>42</sup>, clearly indicating that IGF-I also enhances the entrance of cell precursors into the thymus. We showed that intrathymic lymphocyte traffic is enhanced in vitro by GH and IGF-I, via modulation of extracellular matrix ligands and receptors<sup>53</sup>. It is noteworthy that insulin does not promote significant effects upon the intrathymic nurse cell traffic<sup>54</sup>, suggesting that IGF-I effects are triggered by the IGF-I receptor itself, and not via the insulin receptor. In vivo, we showed that, 16 h following a single intrathymic injection of GH in normal mice, there is an increase in numbers of CD4<sup>+</sup> recent thymic emigrants in the lymph nodes<sup>15</sup>. Moreover, increasing numbers of circulating CD4<sup>+</sup> recent thymic emigrants were seen in the blood of HIV patients receiving GH therapy<sup>41</sup>.

Recent data from our laboratory indicate that GH also influences migration of lymphocytes from peripheral lymphoid organs. As depicted in Table 1, ECM-driven, as well as chemokine-driven lymphocyte migration is enhanced in GH transgenic animals, as compared to corresponding agematched wild-type controls. Since the density of the respective receptors in peripheral lymphocyte surface does not change, these experiments also indicate that GH overexpression yields an activation status of ECM and chemokine receptors, above the levels seen in normal individuals.

	<b>Wild type</b> (x10 <sup>4</sup> )	<b>GH-Tg</b> (x10 <sup>4</sup> )
BSA	$2.53 \pm 0.18$	$2.38 \pm 0.21^{NS}$
LN	$4.85 \pm 0.17$	$6.45 \pm 0.43*$
CCL21	$7.55 \pm 0.16$	$11.30 \pm 0.46*$
LN + CCL21	$23.90 \pm 1.74$	$29.60 \pm 2.26^*$

*Table 1.* Enhancement of migratory capacity of peripheral cells from mesenteric lymph nodes of GH transgenic mice, as compared to wild-type controls\*
#### 6. GH/IGF-I CIRCUITS IN HEMATOPOIETIC TISSUES

To further understand the various endocrine effects of GH and IGF-I in hematopoietic tissues, it is necessary to define the molecular mechanisms involved in triggering such biological responses. In this respect, expression of receptors for both hormones has been detected in both hematopoietic and microenvironmental cells of bone marrow, thymus (Fig. 4), as well as secondary lymphoid organs.



*Figure 4.* Flow cytometry analysis of GH receptor expression by thymic lymphocytes. The upper left panel shows the whole labeling of thymocytes, whereas the others depict GH receptor in CD4/CD8-defined subsets. Mean percentage levels are shown within each profile. Since the CD4<sup>-</sup>CD8<sup>-</sup> double-negative compartment also contains non-T lymphocytes, we first gated double-negative cells in the positive region for CD90w (a pan T-cell marker) and negative region for CD4 plus CD8, as seen in the bottom left panel. The cells within the square, thus corresponding to CD4<sup>-</sup>CD8<sup>-</sup>CD90<sup>+</sup> thymocytes, were then evaluated for the presence of GH receptor, as seen in the bottom right panel.

Expression of GH receptor by human TEC has been demonstrated by means of ligand-specific binding assay, immunocytochemistry, *in situ* hybridization and reverse transcriptase-polymerase chain reaction (RT-PCR) (reviewed by Savino and Dardenne<sup>6</sup>) Additionally, GH receptor is present on thymocytes, particularly in the CD4<sup>-</sup>CD8<sup>-</sup> immature subset<sup>55,56</sup>. Together, these findings point to the possibility of indirect as well as direct role of GH in thymocyte differentiation.

In peripheral lymphoid organs as well as in circulating blood cells, we also detected GH receptors in both T and B lymphocytes, as well as monocytes and granulocytes<sup>57</sup>, thus indicating that hematopoietic cells may be responsive to GH, not only in early phases of their differentiation but also once they are mature, which fits, for example, with the data showing that GH improves resistance to bacterial infections, acting directly on phagocytic cells<sup>58,59</sup>.

IGF-I receptor has been found in hematopoietic tissues. In the bone IGF-I receptor been detected erythroid, marrow. has in granulocytic/macrophage, as well as lymphoid lineages (reviewed by Venters<sup>1</sup> and Zumkeller<sup>2</sup>). IGF-I receptor is also expressed in thymocytes and TEC<sup>60-63</sup>. Interestingly, we found that IGF-I receptor density on TEC membranes increases following in vitro GH treatment<sup>62</sup>, clearly suggesting that in hematopoietic tissues, IGF-I receptor expression is also partially under GH control. Lastly, functional IGF-I receptor is also present in peripheral lymphocytes, monocytes, and granulocytes<sup>59,63</sup>.

In addition to the GH/IGF-I endocrine loops discussed earlier, these molecules likely act on an autocrine/paracrine way. In fact, it has been largely shown that cells of hematopoietic tissues produce GH, as well as IGF-I. Bone marrow stromal cells, as well as hematopoietic precursors in the organ can secrete both hormones<sup>1</sup>. Actually, IGF-I is secreted by human CD34<sup>+</sup> precursors, as well as *ex vivo* expanded granulocyte-macrophage, erythroid, and megakaryocyte precursors<sup>63</sup>.

Moreover, both TEC and thymocytes can produce GH and IGF-I<sup>35,53,56</sup>. From a functional point of view, it was found that thymocyte proliferation induced by thymocyte-derived GH occurs via IGF-I production by these cells<sup>64</sup>, indicating an autocrine/paracrine circuit involving the GH/IGF-I axis. In fact, thymocytes and epithelial cells constitutively express IGF-I, and such expression is enhanced by GH<sup>62</sup>. Accordingly, anti-IGF-I receptor blocking antibodies block thymocyte differentiation and expansion in fetal thymus organ cultures<sup>65</sup>.

Overall, at least concerning lymphopoietic tissues, we can hypothesize that the GH/IGF-I axis exerts its modulatory effects simultaneously by means of endocrine and autocrine/paracrine loops.

# 7. HEMATOPOIESIS IN GH GENETICALLY ENGINEERED ANIMALS

The various aspects discussed in the preceeding quite strongly point to a role of GH and IGF-I in regulating various aspects of hematopoiesis, not only *in vitro* but, most importantly, in normal adult life. A putative mandatory role of GH and IGF-I upon the early development of hematopoietic organs has been investigated using genetically engineered mice, either for GH, GH receptor, or IGF-I.

In both GH receptor knockout and GH transgenic mice, we can see progression from immature CD4<sup>-</sup>CD8<sup>-</sup> thymocytes through CD4<sup>+</sup>CD8<sup>+</sup>, ending up with CD4 and CD8 simple positive cells<sup>66</sup> (our unpublished results). Nevertheless, the thymic microenvironment of GH transgenic and GH receptor knockout animals is altered. Taking thymulin serum levels as one parameter for TEC activity, we found an enhancement in GH-transgenic and a decrease in GR receptor knockout mice, as compared to age-matched wild-type counterparts<sup>54</sup>. In addition, recent data revealed an increase in laminin production and CXCL12 expression by TEC from GH transgenic animals that parallels increased laminin- or CXCL12-driven thymocyte migration in these mice (our unpublished results).

Studies have also been conducted in IGF-I knockout mice<sup>6,7</sup>, and the relative numbers of CD4/CD8 defined thymocyte subsets appear normal, which tells us that thymocyte development can occur in the absence of IGF-I. Yet, in this study several other parameters of both the lymphoid and microenvironmental compartments were not dissected. So, we cannot discard the possibility that some thymic functions are altered in these animals.

# 8. CONCLUDING REMARKS AND MAJOR ISSUES TO BE ADDRESSED IN THE FUTURE

The findings discussed herein clearly indicate that GH and IGF-I participate in the physiology of hematopoiesis. Although data from genetically engineered mice indicate that these molecules are not mandatory for initial development of hematopoietic tissues, it is apparent that circulating levels of these hormones in adult individuals regulate various biological functions related to both microenvironmental and lymphoid cells.

Also necessary for a better understanding of the role of the GH/IGF axis in hematopoiesis will be the dissection of the intracellular signaling pathways triggered by the GH/GH receptor ligation and activation of the IGF-I circuit in distinct cell types of hematopoietic organs. This issue, together with an accurate gene profiling study for determining which genes are modulated by these molecules, will provide a more solid molecular background for the biological responses elicited by GH/IGF-I axis.

The possibility that GH and IGF-I improve some thymic functions, including thymocyte proliferation and migration, places this molecule as a potential adjuvant therapeutic in immunodeficiency conditions associated with thymocyte decrease and loss of peripheral T cells, as in HIV infection. Peripheral blood mononuclear cells from HIV-infected patients treated with GH for 3 months exhibited an upward trend in the IL-2 production after in vitro stimulation with HIV-HIV-1 envelope peptides<sup>8</sup>. More recently, 6month treatment of AIDS patients with GH or IGF-I plus anti-virus therapy resulted in an increase of the thymus, with augmentation of CD4<sup>+</sup> recent thymic emigrants and total numbers of circulating  $CD4^+$  T cells<sup>41</sup>. In addition, it has been recently shown that IGF-I was able to enhance lymphoid as well as myeloid cell reconstitution in mice, following allogeneic bone marrow transplantation, without side effects in terms of aggravating graft versus host disease<sup>69</sup>. Overall, the potential therapeutic use of GH and IGF-I fits with the hypothesis for a general anti-stress effect of these hormones upon the hematopoietic system $^{70,71}$ .

Lastly, it should be pointed out that the effects of the GH/IGF-I axis upon hematopoiesis should be placed in a broader context. Actually, the neuroendocrine control of hematopoiesis appears to be far more complex, with other hormones, neuropeptides and neurotransmitters being involved<sup>6,70,72</sup>.

#### **ACKNOWLEDGEMENTS**

This work was partially supported with grants from: Inserm/Fiocruz and Capes/Cofecub France/Brazil conjoint Programmes, PICS/CNRS grant (France), CNPq and Capes (Brazil).

#### REFERENCES

- H.D. Venters, R. Dantzer, G.G. Freund, S.R. Broussard and K.K. Kelley, Growth hormone and insulin-like growth factor as cytokines in the immune system. In. *Psychoneuroimmunology*, R. Ader, D.L. Felten and N. Cohen, Eds., 3rd Ed. Academic Press, San Diego, pp. 339-62 (2001).
- 2. W. Zumkeller, The insulin-like growth factor system in hematopoietic cells. Leukemia & Lymphoma, 43: 487-491(2002).
- 3. M. Metcalf, Regulatory mechanisms controlling hematopoiesis: principles and problems. *Stem cells*, **16 Suppl 1**: 3-11 (1998).
- 4. T. Lapidot and I. Petit, Current understanding of stem cell mobilization: the roles of chemokines, proteolytic enzymes, adhesion molecules, cytokines, and stromal cells. *Exp Hematol*, **30**, 973-981 (2002).

- 5. F. Prosper and C.M. Verfaillie, Regulation of hematopoiesis through adhesion receptors. *J Leukocyte Biol*, **69**, 307-316 (2001).
- 6. G. Anderson and E.J. Jenkinson, Lymphostromal interactions in thymic development and function. *Nat Rev Immunol*, **1**, 31-40 (2001).
- 7. W. Savino and M. Dardenne, Neuroendocrine control of the thymus. Endocrine Rev, 21, 412-443 (2000).
- W. Savino, D.A. Mendes-da-Cruz, J.S. Silva, M.Dardenne and V. Cotta-de-Almeida, Intrathymic T cell migration: a combinatorial interplay of extracellular matrix and chemokines? *Trends Immunol*, 23, 305-313 (2002).
- 9. W. Savino, S.R. Dalmau and V. Cotta-de-Almeida, Role of extracellular matrix mediated interactions in thymocyte migration. *Dev Immunol*, 7, 19-28 (2000).
- F. Annunziato, P. Romagnani, L. Cosmi, E. Lazzeri and S. Romagnani, Chemokines and lymphopoiesis in the thymus. *Trends Immunol*, 22, 277-281 (2001).
- 11. K.M. Ansel and J.G. Cyster, Chemokines in lymphopoiesisand lymphoid organ development. *Curr Opin Immunol*, **13**, 172-179 (2001).
- S. Ben-Efraim, Y. Keisari, R. Ophir, M. Pecht, N Trainin and L. Burstein, Immunopotentiating and immunotherapeutic effects of thymic hormones and factors with special emphasis on thymic humoral factor THF-γ2. *Crit Rev Immunol*, 19, 261-284 (1999).
- Y.H. Tseng, M.A. Kessler and L.A. Schuler, Regulation of interleukin (IL)-1α, IL-1β and IL-6 expression by growth hormone and prolactin in bovine thymic stromal cells, *Mol Cell Endocrinol*, **128**, 117-127 (1997).
- R.G. Goya, M.C. Gagnerault, M.C Leite-de-Moraes, W. Savino and M. Dardenne, In vivo effects of growth hormone on thymus function in aging mice. Brain Behav Immun, 6, 341-354 (1992).
- S. Smaniotto, M.M. Ribeiro-Carvalho, M. Dardenne, W. Savino and V. Mello-Coelho, Growth hormone stimulates the selective trafficking of thymic CD4<sup>+</sup>CD8<sup>-</sup> emigrants to peripheral lymphoid organs. *Neuroimmunomodulation*, **11**, 299-306 (2004).
- 16. M. Pelletier, S. Montplaisir, M. Dardenne and J.F. Bach, Thymic hormone activity and spontaneous autoimmunity in dwarf mice and their littermates. *Immunology*, **30**, 783-788 (1976).
- E. Mocchegiani, P. Paolucci, A. Balsamo, E. Cacciari and N.Fabris, Influence of growth hormone on thymic endocrine activity in humans. *Horm Res*, 33, 248-255 (1990).
- E. Mocchegiani, A. Sartorio, L. Santarelli, S. Ferrero and N. Fabris, Thymulin, zinc and insulin-like growth factor-I (IGF-1) activity before and during recombinant growth hormone (rec-GH) therapy in children and adults with GH deficiency. J Endocrinol Invest, 19, 630-637 (1996).
- 19. J. Timsit, B. Safieh, M.C. Gagnerault, W. Savino, J. Lubetzki, J.F. Bach and M. Dardenne, Increase of circulating levels of thymulin in hyperprolactinemia and acromegaly. *C R. Acad Sci (Paris) Serie III*, **310**, 7-13 (1991).
- 20. S. Halvorsen and A.G. Bechensteen, Physiology of erythropoietin during mammalian development. *Acta Paediatr*, **91 Suppl**, 17-26 (2002).
- S.H. Boyer, T.R. Bisho, O.C. Rogers, A.N. Noyes, L.P. Frelin and S. Hobbs, Roles of erythropoietin, insulin-like growth factor 1, and unidentified serum factors in promoting maturation of purified murine erythroid colony-forming units. *Blood*, 80, 2503-2512 (1992).

- 22. S. Merchav, I. Silvian-Drachsler, I. Tatarsky, M. Lake and A. Skottner, Comparative studies of the erythroid-potentiating effects of biosynthetic human insulin-like growth factor-I and -II. J Clin Endocrinol Metabol, 74, 447-452 (1992).
- S. Miyagawa, M. Kobayashi, N. Konishi, T. Sato, and K. Ueda, Insulin and insulinlike growth factor I support the proliferation of erythroid progenitor cells in bone marrow through the sharing of receptors. *Brit J Hematol*, 109, 555-562 (2000).
- N. Hizuka, K. Takano, I. Tanaka, N. Honda, T. Tsushima and K. Shizume, Characterization of insulin-like growth factor 1 receptor on human erythrocytes. J Clin Endocrinol Metabol, 61, 1066-1070 (1985).
- B. Sivan, P. Lilos and Z. Laron, Effects of insulin-like growth factor-I deficiency and replacement therapy on the hematopoietic system in patients with Laron syndrome (primary growth hormone insensitivity). *J Pediatr Endocrinol Metabol*, 16, 509-520 (2003).
- E.R. Christ, M.H. Cunnings, N.B. Westwood, B.M. Sawyer, T.C. Pearson, P.H. Sönksen, and D.J. Russel-Jones, The importance of growth hormone in the regulation of erythropoiesis, red cell mass and plasma volume in adult with growth hormone deficiency. *J Clin Endocrinol Metabol*, 82, 2985-2990 (1997).
- 27. E. Long, H.T. Huynh and X. Zhao, Involvement of insulin-like growth factor-1 and its binding proteins in proliferation and differentiation of murine bone marrowderived macrophage precursors. Endocrine, **9**, 185-192 (1998).
- 28. S. Jeay, G.E. Sonenshein, P.A. Kelly, M.C. Postel-Vinay and E. Baixeras, Growth hormone exerts antiapoptotic and proliferative effects through two different pathways involving nuclear factor-kappaB and phosphatidylinositol 3-kinase. *Endocrinology*, 142, 147-156 (2001).
- L. Soon, L. Flechner, J.S. Gutkind, L.H. Wang, R. Baserga, J.H. Pierce and W. Li, Insulin-like growth factor I synergises with interleukin 4 for hematopoietic cell proliferation independent of insulin receptor substrate expression. *Mol Cell Biol*, 19, 3816-3828 (1999).
- 30. L.F. Gibson, D. Pitkel and K.S. Landreth, Insulin-like growth factor-I potentiates expansion of interleukin-7 dependent pro-B cells. *Blood*, **82**, 3005-3011 (1993).
- P. Jardieu, R. Clark, D. Mortensen and K. Dorskind, *In vivo* administration of insulin-like growth factor-I stimulates primary-B lymphopoiesis and enhances lymphocyte recovery after bone marrow transplantation. *J Immunol*, **152**, 4320-4327 (1994).
- B. Valentinis, G. Romano, F. Peruzzi, A. Morrione, M. Risco, S. Soddu, B. Cristofanelli, A. Sacchi and R. Baserga, Growth and differentiation signals by the insulin-like growth factor 1 receptor in hemopoietic cells are mediated through different pathways. *J Biol Chem*, **30**, 12423-12430 (1999).
- 33. B. Lin, Y. Kinoshita, F. Hato and Y. Tsuji, Enhancement of DNA synthetic activity of thymic lymphocytes by the culture supernatant of thymus epithelial cells stimulated by growth hormone. *Cell Mol Biol (Noisy-Le-Grand)*, **43**, 351-359 (1999).
- 34. M.C. Postel-Vinay, V. Mello-Coelho, M.C. Gagnerault and M. Dardenne, Growth hormone stimulates the proliferation of activated mouse T lymphocytes. *Endocrinology*, **138**, 1816-1820 (1997).
- 35. R. Clark, The somatogenic hormones and insulin-like growth factor-1: stimulators of lymphopoiesis and immune function. *Endocrine Rev*, **18**, 157-179 (1997).
- 36. P. Sabharwal and S.Varma, Growth hormone synthesized and secreted by human thymocytes acts via insulin-like growth factor I as an autocrine and paracrine growth factor. *J Clin Endocrinol Metab*, **81**, 2663-2669 (1996).

- K.W. Kelley, S. Brief, H.J. Westly, T. Novakofski, P.J. Bechtel, J. Simon and E.B. Walker, GH3 pituitary adenoma cells can reverse thymic aging in rats. *Proc Natl Acad Sci USA*, 83, 5663-5667 (1986).
- K. Binz, P. Joller, P. Froesch, H. Binz, J. Zapf and E.R. Froesch, Repopulation of the atrophied thymus in diabetic rats by insulin-like growth factor I. *Proc Natl Acad Sci USA*, 87, 3690-3694 (1990).
- W.E. Beschorner, J. Divic, H. Pulido, X. Yao, P. Kenworthy and G. Bruce, Enhancement of thymic recovery after cyclosporine by recombinant human growth hormone and insulin-like growth factor I. *Transplantation*, 52, 879-884 (1991).
- N. Bazzoni, B. Ambrosi, M. Arosio, M. Barbareschi, M. Losa and G. Faglia, Acromegaly and thymic hyperplasia: a case report. *J Endocrinol Invest*, 11, 931-935 (1990).
- L.A. Napolitano, J.C. Lo, M.B. Gotway, K. Mulligan, J.D. Barbour, D. Schmidt, R.M. Grant, R.A. Halvorsen, M. Schambelan and J.M. McCune, Increased thymic mass and circulating naïve CD4 T cells in HIV-1-infected adults treated with growth hormone. *AIDS*, 16, 1103-1110 (2002).
- E. Montecino-Rodriguez, R. Clark and K. Dorshkind, Effects of insulin-like growth factor administration and bone marrow transplantation on thymopoiesis in aged mice. *Endocrinology*, 139, 4120-4126 (1998).
- J.C. Woo, G.A. Dean, R. Clark and P.F. Moore, Investigation of recombinant human insulin-like growth factor type I in thymus regeneration in the acute stage of experimental FIV infection in juvenile cats. *AIDS Res Hum Retroviruses*, **10**, 1377-1388 (1999).
- A.F. Tarantal, M.K. Hunter and S.E. Garkosky, Direct administration of insulin-like growth factor to fetal rhesus monkeys (*Macaca mulatta*). *Endocrinology*, **138**, 3349-3358 (1997).
- 45. S. Jeay, G.E. Sonenshein, M.C. Postel-Vinay and E. Baixeras, Growth Hormone prevents apoptosis through activation of nuclear factor-kB in interleukin-3 dependent Ba/F3 cell line. *Mol Endocrinol*, **14**, 650-661 (2000).
- E. Baixeras, S. Jeay, P. Kelly and M.C. Postel-Vinay, The proliferative and antiapoptotic actions of growth hormone and insulin-like growth factor-1 are mediated through distinct signaling pathways in the pro-B Ba/F3 cell line. *Endocrinology*, 142, 2968-2977 (2001).
- J. Suzuki, Y. Kaziro and H. Koide, Synergistic action of R-Ras and IGF-1 on BclxL expression and caspase inhibition in BaF3 cells: R-Ras and IGF-1 control distinct anti-apoptotic kinase pathways. *FEBS Lett*, 437, 112-116 (1998).
- C. Carlo-Stella, M. Di Incola, R. Milani, P. Longoni, M. Milanesi, A. Bifulo, C. Stucchi, A. Gidetti, L.Cleris, F. Formelli, G. Garotta and A.M. Gianni, Age- and irradiation-associated loss of bonne marrow hematopoietic function in mice is reversed by recombinant human growth hormone. *Exp Hematol*, 32, 171-178 (2004).
- 49. K. Vanderkerken, K. Asosingh, F. Braet, I. van Riet and B. van Camp, Insulin-like growth factor-1 acts as a chemoattractant factor for 5T2 multiple myeloma cells. *Blood*, **93**, 235-241(1999).
- Y.T. Tai, K. Podar, L. Atley, Y.H. Tseng, M. Akiyama, R. Shringarpure, R. Burger, T. Hideshima, D. Chauhan, N. Mitsiades, P. Richardson, N.C. Munshi, C.R. Kahun, C. Mitsiades and K.C. Anderson, Insulin-like growth factor-1 induces adhesion and migration in human multiple myeloma cells via activation of beta1integrin and phosphatidylinositol 3'-kinase/AKT signaling. *Cancer Res*, 15, 5850-5858 (2003).

- Y.W. Qiang, L. Yao, G. Tosato and S. Rudikoff, Insulin-like growth factor I induces migration and invasion of multiple myeloma cells. *Blood*, **103**, 301-308 (2004).
- D.D. Taub, G. Tsarfaty. A.R. Lloyd, S.K. Durum, D.L. Longo and W.J. Murphy, Growth hormone promotes human T cell adhesion and migration to both human and murine matrix proteins in vitro and directly promotes xenogeneic engraftment. J Clin Invest, 94, 293-300 (1994).
- 53. V. Mello-Coelho, D.M.S. Villa-Verde, M. Dardenne and W. Savino, Pituitary hormones modulate cell-cell interactions between thymocytes and thymic epithelial cells. *J Neuroimmunol*, **76**, 39-49 (1997).
- 54. W. Savino, S. Smaniotto, N. Binart, M.C. Postel-Vinay and M. Dardenne, *In vivo* effects of growth hormone on thymic cells. *Ann NY Acad Sci*, **992**, 179-185 (2003).
- 55. M.C. Gagnerault, M.C. Postel-Vinay and M. Dardenne, Expression of growth hormone receptors in murine lymphoid cells analyzed by flow cytofluorometry. *Endocrinology*, **137**, 1719-1726 (1996).
- V. Mello-Coelho, M.C. Gagnerault, J.C. Souberbielle, C.J.Strasburger, W. Savino, M. Dardenne and M.C. Postel-Vinay, Growth hormone and its receptor are expressed in human thymic cells. *Endocrinology*, **139**, 3837-3842 (1998).
- 57. C.K. Edwards, L.M. Yunger, R.M. Lorence, R. Dantzer and K.W. Kelley, The pituitary gland is required for protection against lethal effects of *Salmonella typhimurium*. *Proc Natl Acd Sci USA*, **88**, 2274-2277 (1991).
- T. Inoue, H. Saito, T. Matsuda, K. Fukatsu, I. Han, S. Furukawa, S. Ikeda and T. Muto, Growth hormone and insulin-like growth factor-I augment bactericidal capacity of human polymorphonuclear neutrophils. *Shock*, 10, 278-284 (1998).
- S. Verland and S. Gammeltoft, Functional receptors for insulin-like growth factors I and II in rat thymocytes and mouse thymoma cells. *Mol Cell Endocrinol*, 67, 207-216 (1989).
- R. Kooijman, L.E. Scholtens, G.T. Rijkers and B.J. Zegers, Type I insulin-like growth factor receptor expression in different developmental stages of human thymocytes. *J Endocrinol*, 147, 203-209 (1995).
- V. Mello-Coelho, D.M.S. Villa-Verde, D.A. Farias-de-Oliveira, J.M. Brito, M. Dardenne and W. Savino, Functional growth hormone-controlled IGF-1/IGF-1 receptor autocrine/ paracrine circuit in human and murine thymic epithelial cells. *Neuroendocrinology*, **75**, 139-150 (2002).
- 62. X. Xu, C. Mardell, C.J. Xian, H. Zola and L.C. Read, Expression of functional IGF-1 receptor on lymphoid cell subsets of rats. *Immunology*, **85**, 394-399 (1995).
- 63. M. Majka, A. Janowska-Wieczorek, J. Ratajczak, K. Ehrenman, Z. Pietrzkowski, M.A. Kawalka, A.M. Gewirtz, S.G.Emerson and M.Z. Ratajczak, Numerous growth factors, cytokines, and chemokines are secreted by human CD34+ cells, myeloblats, and megakaryoblasts and regulate normal hematopoiesis in an autocrine/paracrine manner. *Blood*, 15, 3075-3085 (2001).
- 64. P.S. Hinton, C.A. Peterson, E.M. Dahly and D.M. Ney, IGF-1 alters lymphocyte survival and regeneration in thymus and spleen after dexamethasone treatment. *Am J Physiol*, **274**, R912-20(1998).
- O. Kecha, F. Brilot, H. Martens, N. Franchimont, C.Renard, R. Greimers, M.P. Defresne, R. Winkler and V. Geneen, Involvement of insulin-like growth factors in early T cell development: a study using fetal thymic organ cultures. *Endocrinology*, 141, 1209-1917 (2000).
- 66. K. Dorskind and N.D. Horseman, The roles of prolactin, growth hormone, insulinlike growth factor-I, and thyroid hormones in lymphocyte development and

function: insights from genetic models of hormone and hormone receptor deficiency. *Endocrine Rev*, **21**, 292-312 (2000).

- E. Montecino-Rodriguez, R.G. Clark, L. Power-Braxton and K. Dorshkind, Primary B cell development is impaired in mice with defects of the pituitary/thyroid axis. J Immunol, 159, 2712-2719 (1997).
- B.Y. Nguyen, M. Clerici, D.J. Venzon, S. Bauza, W.J. Murphy, D. Longo, M. Baseler, N. Gesundheit, S. Broder, G. Shearer and R. Yarchoan, Pilot study of the immunologic effects of recombinant human growth hormone and insulin-like growth factor in HIV-infected patients. *AIDS Infection*, **12**, 895-904 (1998).
- O. Alpdogan, S.J. Muriglan, B.J. Kappel, E. Doubrovina, C. Schmaltz, R. Schiro, J.M. Eng, A.S. Greenberg, L.M. Willis, J.A. Rotolo, R.J. O'Reilly and M.R.M. Van Den Brink, Insulin-like growth factor-1 enhances lymphoid and myeloid reconstitution after allogeneic bone marrow transplantation. *Transplantation*, 75, 1977-1983 (2003).
- 70. K. Dorshkind and N.D. Horseman, Anterior pituitary hormones, stress, and immune system homeostasis. *Bioessays*, 23, 288-294 (2001).
- 71. L.A. Welniak, R. Sun and W.J. Murphy, The role of growth hormone in T-cell development and reconstitution. *J Leukocyte Biol*, **71**, 381-387 (2002).

# Chapter 8

# **BRAIN DEVELOPMENT**

Gregory J. Popken, Martha Dechert-Zeger, Ping Ye, A. Joseph D'Ercole Division Pediatric Endocrinology, Department of Pediatrics, University of North Carolina at Chapel Hill, Chapel Hill, NC 27599-7039, USA

Key words: Apoptosis; astrocyte; brain injury; differentiation; glia; growth hormone; IGF; IGF-I; IGF-II; IGF1R; IGFBP; neuron; oligodendrocyte; transgenic.

## **1. INTRODUCTION**

A relatively large body of evidence accumulated in recent years establishes an essential role for insulin-like growth factor-I (IGF-I) in central nervous system (CNS) development. Evidence that growth hormone (GH) has a role in brain development, both by stimulating IGF-I expression and through its direct actions, also is emerging. This chapter focuses on recent studies of the CNS that have contributed to our understanding of the roles played by IGFs and GH on neural proliferation, differentiation, survival, and maturation.

The IGFs, IGF-I and IGF-II, IGF receptors, and IGF binding proteins (IGFBPs), often are collectively termed the IGF system. Compared to GH, much more research has focused on the roles of IGFs in CNS development. Both IGF-I and IGF-II appear to signal their actions predominantly, if not exclusively, by interacting with the type I IGF receptor (IGFIR)<sup>1,2</sup>. Furthermore, each of their actions is modulated by IGFBPs that can either inhibit or augment their activity<sup>3,4</sup>. In addition, the cation-independent mannose-6-phosphate receptor (ciM6PR) binds IGF-II (but not IGF-I) with high affinity, and therefore, also is termed the type II IGF receptor (IGFIIR)<sup>1,2</sup>. It is, however, not clear whether the ciM6PR/IGFIIR transduces IGF-II growth-promoting signals. Because it traffics mannose-6-phosphate-

containing proteins and IGF-II to lysosomes for degradation, the ciM6PR/IGFIIR is capable of regulating IGF-II abundance and in this way likely alters IGF-II's influence. A wide array of studies, including those of genetically engineered mutant mouse lines<sup>5</sup>, has demonstrated the pivotal role IGFs play in development and maintenance of the CNS. Components of the IGF system have been implicated in virtually every aspect of neural development including proliferation, survival, differentiation, and maturation (see later).

Although IGF-I and IGF-II are transported across the blood-brain barrier (BBB)<sup>6,7</sup> the relative contribution of peripherally produced IGFs to CNS development, as compared to that produced locally, remains to be entirely clarified. Nonetheless, several observations make it seem certain that IGF developmental influences derive from the actions of IGFs expressed within the CNS: (1) Each component of the IGF system is expressed in the CNS during development; and (2) Circulating IGFs do not appear to contribute significantly to somatic growth until a time in postnatal life when CNS development is nearly complete. Evidence from mutant mouse lines engineered to have low blood IGF-I levels, but normal tissue IGF-I expression, do not begin to exhibit growth retardation until the 3 to 4 weeks of age<sup>8,9</sup>.

# 2. EXPRESSION

## 2.1 GH and GH receptor

GH is produced in the anterior pituitary gland and is believed to reach the CNS via receptor-mediated transport across the BBB<sup>10,11</sup>. Chronic administration of GH results in an increased GH concentration in CSF<sup>10</sup>. There is, however, evidence that GH can be synthesized within the CNS, specifically in the lateral hypothalamus, caudate putamen, striatum, ventral thalamus, formation reticularis, and the basal cortex <sup>12,13</sup>.

The cognant GH receptor (GHR) is expressed in a number of brain regions during development. Studies assessing GHR expression during early development are limited. GHR immunoreactivity is detected as early as embryonic day 3 (E3) in all regions (diencephalon, telencephalon, mesencephalon) of chick brain<sup>14</sup>. By E14, however, GHR expression appears to be restricted to the molecular and pyramidal cell layers of the cerebral cortex, the grey matter of the cerebellum, the walls of the ventricles, and the choroid plexus<sup>14</sup>. In rats, GHR immunoreactivity can be detected between E12-E14 in derivatives of neural plate ectoderm, and in most brain regions by E18<sup>15</sup>. Lobie *et al.*<sup>16</sup> carried out an extensive postnatal ontological

analysis of GHR expression in the early postnatal rat and rabbit. Both neuronal and glial cells demonstrated GHR immunoreactivity. In rats, the strongest immunoreactivity was found in layers 2, 3, 5 and 6 of the cerebral cortex; neurons of the hypothalamus; retinal ganglion cells; and Purkinje cells of the cerebellum. GHR immunoreactivity then gradually decreased from postnatal day (P)2 to P90. GHR mRNA in rabbits exhibits a similar distribution pattern and developmental change<sup>16</sup>.

In the adult rats, GHR expression is highest in the choroid plexus although it can be found in other areas, including the hippocampus, hypothalamus, and pituitary<sup>17-19</sup>. Zhai *et al.*<sup>19</sup> also report moderate expression in the cerebral cortex. These results are consistent with those observed in human studies<sup>20,21</sup>.

## 2.2 IGF system proteins

CNS expresses most components of the IGF system, including IGF-I, IGF-II; their receptors; and IGFBP-2, -4, -5, and -6 during development. Each of these proteins is expressed during development in spatially specific patterns that are temporally coordinated. These patterns suggest important and dynamic roles during CNS development. IGFBP-1 and IGFBP-3 appear not to be expressed in the brain during normal development, although IGFBP-3 expression is induced during injury<sup>22</sup>. Each, however, appears to be transported to the CNS and exert biologic actions. For example, IGFBP-1 has been identified in the rat cerebellum<sup>23</sup>, and IGFBP-1 when ectopically expressed in the brains of transgenic (Tg) mice markedly retards brain development<sup>24-27</sup>.

#### 2.2.1 IGF-I and IGFIR

IGF-I and IGFIR expression are widespread both during development and in the adult. Table 1 summarizes the temporal and region distribution of IGF-I in rodent CNS during development. Brain IGF-I is expressed in a region-specific pattern. While neurons are the major cells of IGF-I expression, IGF-I appears to be expressed by all other major neural cell types, including astrocytes, oligodendrocytes, and microglial<sup>28-32</sup>, as well as neural stem cells (NSC) and their lineage-restricted progenitors<sup>33,34</sup>. During rat early postnatal development high IGF-I expression is found localized to somatosensory and brain stem nuclei including intralaminar nucleus, anterior pretectal nucleus, parafascicular nucleus and retroflexus nucleus<sup>35</sup> (Table 1). The timing, magnitude, and location of IGF-I expression closely correlate with dendritic growth and maturation, and with synapse formation in projecting neurons in these regions.

Region	Prenatal	Perinatal	Adult
Carabrum	Л	σ	Л
Meninges			
Carebral cortax			D
Diriform cortex	ND	ND	D
Corpus callosum	ΝΔ	NA	
Olfactory			
Mitral cell laver	D	D	D
Glomerular layer	D	D	D
Hinnesempus	D	<u>D</u>	D
Dentate gyrus	NΛ	D	D
Stratum orains	NA NA	D	D
Stratum radiatum	INA NA	D	D
Stratum lug mol	NA NA	D D	D
Stratum pyramidal	INA NA		ND
Corobollum			<u>ND</u>
Cerebellar cortex	ND	D	D
Durkinia cell laver	NL NA	D	D
Fulkinge cen layer	NA NA		
Int granular cell laver	NA NA	ND	ND
White matter	ND	ND	ND
Deep puclei	ND	D	
Red nucleus	ND	D	D
Other		<u>D</u>	D
Inferior olive	р	D	D
Inferior colliculus		D	D
Medial habenula		D	D
Pretectal nucleus			
Intralaminar nuc			D
Parafascicular nuc		D	
Retroflexus nuc	D	D	ND

*Table 1.* Temporal and regional IGF-I mRNA expression during development. **D**, mRNA expression detected. **ND**, mRNA expression not detected. **NA**, area not assessed. Adapted from refs. 35-37.

The temporal expression of IGF-I also is associated with neural cell proliferation. IGF-I mRNA has been detected as early as E11 in most areas of murine brain, including olfactory bulb<sup>28,38,39</sup>. As judged by Northern blot analysis of rodent whole brain, IGF-I expression increases rapidly during late prenatal and early postnatal development and peaks in the first week of

postnatal life, times preceding and during rapid proliferation, differentiation and maturation of neural cells<sup>40</sup>. IGF-I expression then gradually decreases to adult levels by the second postnatal week<sup>40</sup>. In general, it appears that IGF-I expression in specific brain regions follows a similar temporal pattern<sup>36,37,41</sup>.

Similar to IGF-I expression, brain IGFIR exhibits widespread but distinct temporal expression pattern. IGFIR expression in the CNS begins during embryonic development, peaks perinatally, and then gradually decreases to adult levels in both rodents<sup>42,43</sup> and rabbits<sup>44</sup>. While the IGFIR appears to be expressed in all neural cell types and at each stage of development, its expression is concentrated in distinct areas. In rodents the areas that exhibit relatively high levels of expression include the cerebral cortex, hippocampus, olfactory bulb, cerebellum and choroid plexus<sup>28,45,46</sup>. In addition to the cell body, the IGFIR also is expressed on the growth cone<sup>47</sup> and synapses<sup>48</sup>, consistent with a role for IGF-I signaling in dendritic growth and maturation, and synaptogenesis. In the cerebellum the IGFIR is localized to the postsynaptic membrane in the dendrites and soma of Purkinje cells and the presynaptic membrane of axon terminals associated with Purkinje cell soma, as well as mossy fiber rosettes in the cerebellar glomeruli<sup>48</sup>.

#### 2.2.2 IGF-II and ciM6PR/IGFIIR

IGF-II likely has a role in early CNS development. In contrast to IGF-I, which exhibits greater magnitude and widespread expression during postnatal life, IGF-II is expressed less broadly in the prenatal CNS. In rats, IGF-II mRNA has been found in the neuroectoderm as early as  $E10^{49}$ . In addition to choroid plexus and meninges, IGF-II mRNA also is detected in the hippocampus, hypothalamus, brain stem, and floor of the 3rd ventricle from E13 to E18<sup>38,50</sup>. Unlike IGF-I, which is predominantly expressed by neurons during development, IGF-II appears to be primarily expressed in glia<sup>40,50</sup>. IGF-II expression in most brain regions gradually declines with maturation, and by the time of birth its expression appears to be restricted to the choroid plexus and meninges, where it continues to be expressed at relatively high levels throughout life<sup>51-56</sup>. Whether this pattern of IGF-II expression is replicated in humans is unclear, as reports are somewhat conflicting. One study of human fetal tissues reported that IGF-II mRNA and protein is restricted to the choroid plexus and meninges<sup>57</sup>. However, several studies using Northern blot hybridization and reverse transcriptasepolymerase chain reaction (RT-PCR) analysis of human fetal tissue report widespread IGF-II mRNA expression within both fetal and adult brain<sup>58-60</sup>. The relatively high magnitude of IGF-II expression in CNS is likely due to loss of parental IGF-II gene imprinting, *i.e.*, expression of both parental alleles<sup>59-61</sup>. McKelvic *et al.*<sup>62</sup> report detection of IGF-II mRNA in cerebellum and hypothalamus and note that this expression diminishes or disappears at later stages. In the chicken, IGF-II expression is widespread and can be found in the brain stem nuclei, optic tectum, and cerebellum<sup>63</sup>. IGF-II Purkinje cell expression noted in embryonic stages is later replaced by IGF-I.

In contrast to the restricted expression of IGF-II, the ciM6PR/IGFIIR can be detected in a wide range of CNS regions during development and in the adult. Analysis of mRNA concentrations in whole brain from rats revealed that peak expression occurs prenatally and gradually tapers off during development to reach adult levels by P7<sup>64</sup>. A more detailed analysis of the distribution pattern of ciM6PR/IGFIIR protein done by Valentino et al.<sup>65</sup> confirmed this temporal pattern. In addition, transiently high levels of the ciM6PR/IGFIIR are found prenatally in the cortex, hypothalamus, hippocampus, subventricular zone, and the olfactory bulb<sup>50,65,66</sup>. Hawkes and Kar<sup>67</sup>, using immunoblotting and immunohistochemistry methods, reported a distinct and widespread expression in adult rats, which includes expression in the mitral layer of the olfactory bulb, neuronal cell bodies of the striatum, lavers IV and V of the cortex, Purkinje cells of the cerebellum, and granular and pyramidal layers of the hippocampus. In addition, substantial labeling in the neurophil suggests localization on neuron fibers. Couce et al.<sup>68</sup> report that they found no glial immunoreactivity, although they confirmed pyramidal cell expression in the hippocampus, along with reactivity in the polymorphic layer of the hippocampus and granular cell layer of the dentate gyrus.

In conclusion, the overall pattern of expression of IGF-II suggests that in early prenatal development IGF-II, although produced mainly by the meninges, has widespread influence over the CNS. The extensive expression of the IGF-II receptor in prenatal and early postnatal stages and subsequent reduction indicates that this protein plays an important, but transient, role in the development of the CNS. Whether this role is simply supplanted by IGF-I at later postnatal stages or whether each IGF serves a unique function remains unclear.

#### 2.2.3 IGFBPs

The majority of the data on IGFBP expression come from studies during early postnatal development and in adult animals; but prenatal expression patterns also have been documented. The distinct spatial pattern of IGFBP expression suggests that each may play unique and regionally specific roles in development. Because IGFBPs have functions independent of their interactions with IGFs, such as stimulation of cell migration<sup>3,4</sup>, they may have distinct roles during CNS development. Expression patterns of various IGFBPs are shown in Table 2. At E15, IGFBP-4 mRNA is found in the hippocampus, choroid plexus, and the meningeal layer surrounding the cerebellum<sup>69</sup>. In the adult, expression of IGFBP-4 is more widespread. In addition to the meninges, it can be found in layers II and IV of the cerebral cortex, the olfactory peduncle, amygdala, thalamus, basal ganglia<sup>69</sup>, and the pyramidal cell layer of the hippocampus<sup>70</sup>. The early expression pattern of IGFBP-6 contrasts with embryonic expression of IGFBP-4 in that expression appears much more restricted<sup>71</sup>. IGFBP-6 mRNA has been localized to the trigeminal ganglia at E16. During later stages of development (P21), is associated mainly gamma-aminobutyric expression with acid (GABA)ergic interneurons in the forebrain and cerebellum.

Region	Prenatal	Perinatal	Adult
Cerebrum	2	2	2,4,6
Cerebral cortex	2	2	2,4,6
Piriform cortex			2
Choroid plexus	2,4		
Olfactory bulb	2,5	5	2,4,5
Mitral cell layer		5	2,5
Glomerular layer			
Hippocampus	4	4,5	2,4,5
Dentate gyrus			5
Ammon's horn		5	5
Cerebellum	5	5	2,5,6
Cerebellar cortex			2
Meninges	4		
Deep nuclei		5	
Other			
Superior olive		5	2,5
Inferior olive		5	5
Inferior colliculus		5	2
Pretectal nucleus			2
Trigeminal ganglia	6		
Basal ganglia	4	4	4
Amygdala			4
Thalamus		5	4,5

*Table 2.* Temporal and regional distribution of IGFBPs 2, 4, 5, 6. BP numbers in **bold refer to** entire structure, those not in **bold depict substructural localization** where available. Information from refs. 28,37,50,54,69-77.

Studies addressing expression patterns of IGFBP-2 and IGFBP-5 have focused on adult animals. Both of these binding proteins demonstrate widespread expression. IGFBP-2 was found in the mitral layer of olfactory bulb<sup>76</sup> and in the choroids plexus<sup>54,75</sup>. Sullivan and Feldman<sup>50</sup> also report expression in these regions prenatally. A more detailed study of mRNA expression in early postnatal rats by Lee *et al.*<sup>37</sup> noted IGFBP-2 expression in astroglia in the cerebellum, hippocampus, and neocortex.

IGFBP-5 mRNA can be detected as early as E10 in neuroepithelium and its derived structures, such as notochord and the floor plate<sup>73</sup>, and the epithelium that gives rise to the pituitary gland<sup>74</sup>. Later in prenatal development IGFBP-5 mRNA is found in the proliferative zone of the external granule layer of the cerebellum and the mitral neurons of the olfactory bulb<sup>74</sup>. IGFBP-5 expression is widespread in the postnatal brain, although cerebellum exhibits the highest levels of expression<sup>70,72,77</sup>.

## 3. GH ACTIONS IN THE CNS

GH has been shown to influence brain growth and function (see reviews<sup>78,79</sup>). Whether GH actions are direct or mediated by IGF-I remains to be clarified; however, as with other organs, both mechanisms of action are likely. GH synthesis in the CNS appears to be restricted (see earlier), whilst GH immunoreactivity can be readily identified in most regions of the brain. GH is transported to the CNS either by crossing the BBB or by entry through the circumventricular organs<sup>79</sup>. Chronic administration of GH has been shown to result in increased GH in the cerebral spinal fluid (CSF)<sup>10,11,80</sup>. The presence of GH in the brain and the widespread expression of the GHR in the brain strongly indicates a role for GH in the CNS<sup>78</sup>.

GH deficient mice, such as Snell dwarfs and *little* mice, exhibit reduced cerebral size with altered neuron size and number, and decreased myelin<sup>81</sup>. While Snell dwarfs mice carry a mutation in the pituitary transcription factor Pit-1, the *little* mice do not express the GH-releasing hormone receptor. Whether *little* mice have a reduction in brain myelin concentration, or reduced total myelin content because of their smaller brain size, however, has been debated<sup>82</sup>. In contrast, GH overexpressing transgenic mice exhibit brain weights that are not significantly heavier, although they have somewhat greater weights (7% increased)<sup>83</sup>. Because the magnitude of IGF-I expression has a major influence on brain size, neuron number and myelination (see later), the latter findings are consistent with GH having a regulatory influence on CNS IGF-I expression.

Several reports provide evidence that IGF-I mediates GH effects on the brain: (1) In mice carrying a fusion gene linking the 5'flank of the IGF-I

gene to a luciferase reporter gene, systemic GH administration stimulates luciferase expression in the CNS, but only modestly<sup>41</sup>. Nonetheless, this experiment suggests that GH can exert a degree of control over IGF-I transcription in the brain *in vivo*; (2) Peripheral administration of GH to rats increases IGF-I mRNA expression in the CNS and activates signaling pathways thought to be involved in mediating IGF-I anti-apoptotic effects<sup>84</sup> and (3) In a study of cultured embryonic rat cerebral cortical cells, Ajo *et. al.*<sup>85</sup> demonstrated that GH stimulates neuron progenitor proliferation and differentiation, and that these effects are blocked by antibodies to IGF-I. Finally, there is a significant body of evidence that GH is involved in a wide range of cognitive and behavioral CNS functions including memory, appetite, mood and sleep (see reviews in refs.17,80,86-88). Whether IGF-I mediates these GH actions, however, is not known.

# 4. IGF-I ACTIONS IN CNS

Much evidence indicates that IGFs play an important role in CNS development. While it appears that IGFs promote growth and development of all neural cell types, their specific actions likely depend on the developmental stage of each cell lineage and the local cellular milieu. IGF-I actions on neuron and oligodendrocyte genesis are relatively well studied, but there are few studies, especially *in vivo* studies, investigating other types of CNS neural cells, *i.e.*, astrocytes, microglia, radial glial, and ependymal cells. The paucity of data about IGF actions on the development of these cells is primarily attributable to the lack of a precise understanding of their development and developmental stage-specific markers.

## 4.1 IGF-I actions on neural stem cells

IGFs and IGF receptors are expressed in neural stem cells during early development, and appear to be essential for their proliferation, survival, and differentiation. In culture, IGF-I and/or IGF-II increase neural stem cell number of embryonic rat hypothalamic cells<sup>89,90</sup>, chick immature sympathetic neurons<sup>91</sup>, and neural stem cells derived from E14 mouse striatum<sup>92,93</sup>. It is not clear, however, whether this increase in cell number is the result of IGF's proliferative and/or anti-apoptotic action. Exposure to IGF-I also appears to be important for other growth factors. When IGF-I is removed from culture medium, epidermal growth factor (EGF) and fibroblast growth factor (FGF) are neither able to stimulate proliferation of neural stem cells nor to promote their differentiation into mature neural cells<sup>92,93</sup>.

Recently, neural stem cells have been found to exist in the CNS of the adult human and rodent (see review in ref. 94). These cells are termed "adult neural stem cells", and their embryonic counterparts called "young or embryonic neural stem cells". Although there is no convincing evidence showing that adult neural stem cells act the same as their embryonic counterparts, adult neural stem cells are capable of proliferating *in vivo* and differentiating into each major neural cell type in culture. This raises the possibility of a therapeutic means to treat brain injuries and disorders by promoting adult neural stem cell growth.

IGF-I appears to promote both proliferation<sup>95</sup> and survival<sup>33</sup> of cultured adult neural stem cells in a fashion similar to its effects on young neural stem cells. Reports on IGF-I actions in adult neural cell differentiation, however, have not proved consistent. In 10-day cultures of FGF2-responsive adult stem cells derived from rat forebrain<sup>96</sup> or hippocampus<sup>95</sup>, IGF-I stimulates adult neural stem cells to become neurons, as evidenced by an increase in the number of neurons without apparent changes in the number of GFAP<sup>+</sup> astrocytes and GC<sup>+</sup> oligodendrocytes<sup>95</sup>. Using a similar culture system, Gage and coworkers97, however, reported that when hippocampal FGF2-responsive adult stem cells were cultured with IGF-I or IGF-II for 4 days, the number of cells in oligodendrocyte lineage was markedly increased, while neuron number was only minimally increased. The reasons that account for these discrepancies are not clear. One possibility is that the cells used before IGF-I treatment in these experiments were at different developmental stages and/or that IGF-I has differential effects on cells at different developmental stages. Other factors, such as animal species and the brain regions from which adult stem cells were isolated, may also contribute.

IGF *in vivo* actions on neural stem cells and fate-committed early neural precursors during development and in adults have not been studied in detail. While all neural cell types apparently can develop from neural stem cells in mice with blunted expression of *Igf-1*, *Igf-2*, or *Igfr* genes, the CNS size of these mutant mice is markedly reduced both prenatally and postnatally<sup>98-103</sup>. Fewer oligodendrocytes also develop when cells from forebrains of E18 *Igfr* KO mice are grown in culture<sup>101</sup>. These data suggest that proliferation and maturation of neural stem cells and/or oligodendrocyte precursors in *Igfr* KO mice is reduced. Consistent with this notion, Ni *et al.*<sup>104</sup> showed that in P2 Tg mice that overexpress IGFBP-1, an IGF inhibitor, the number of BrdU-labeled cells is decreased in the hippocampus and in the cerebral cortical ventricular (VZ) and subventricular zones (SVZ), the locales of dividing neural stem cells and precursors. These data, therefore, implicate IGFs in maintaining the number of neural cells with proliferative potential.



*Figure 1.* Cresyl violet staining of postnatal hippocampus from (**a**) control and (**b**) nestin-IGF-I overexpressing transgenic mouse. Note the dramatic increase in overall size of the hippocampus as well as the increase in white matter. Scale bar =  $500 \,\mu\text{m}$ . See Popken *et al.*<sup>39</sup> for more information on nestin-IGF-I mice.

To directly examine IGF-I actions on neural stem cells and fatecommitted precursors, Popken et al.<sup>39</sup> recently generated a line of transgenic (Tg) mice that overexpress a nestin promoter-driven IGF-I transgene in brain (nestin/IGF-I Tg mice) as early as E11. They showed that IGF-I significantly increases brain size, and the number of proliferating cells (by BrdU labeling) in cortical VZ and SVZ at both E14 and E16. The IGF-I overexpression resulted in an increased neuron number (27-69%), depending on brain region) and glial cell number (by ~37%) in cerebral cortex of postnatal mice, indicating that IGF-1 promotes development of cells in both the neuronal and glial lineage. Fig. 1 depicts postnatal changes in hippocampal size. It remains to be elucidated, however, whether IGF-I increases survival of young neural stem cells and precursors.

#### 4.2 IGF actions on neurons

#### 4.2.1 Neurogenesis

Among the major CNS cell types, neurons are the first to develop. In mammals, neurogenesis is completed before birth in nearly all CNS regions, with the exception of the VZ, olfactory bulb, cerebellum, and hippocampus. Multipotent neural stem cells in the VZ, as well as those in the cerebellum and hippocampus, continue to proliferate and generate new neurons throughout postnatal life. It appears that IGFs play an important role in these processes.

A large amount of in vitro evidence demonstrates that IGFs enhance neuronal proliferation<sup>91,105</sup>, survival<sup>106-109</sup>, and maturation<sup>92,96,110</sup>. Postnatal overexpression of IGF-I in brain results in large brains with increased neuron number<sup>111-114</sup>. This increase in neuron number appears to be predominantly due to the capacity of IGF-I to promote neuron survival<sup>114-116</sup>, but increased neuronal proliferation also occurs<sup>114</sup>. IGF-I also may influence neuron number by promoting neuronal differentiation<sup>117</sup>. It appears, however, that IGFs act in vivo in a regionally and developmentally dependent manner. Ye et al.<sup>114</sup> found that overexpression of IGF-I in Tg mice during postnatal life significantly increases the number of cerebellar neurons. At P50 the numbers of granule cells and Purkinje cells are increased by 82% and 20%, respectively. Using the same Tg mice, Dentremont et al.<sup>111</sup> reported the number of neurons is significantly increased in the nucleus of the solitary tract and dorsal motor nucleus of the vagus, but not in the hypoglossal and facial nuclei. In addition, peripheral injection of IGF-I for 20 days in wild type adult rats increases hippocampal granular neurons by 78%, but does not alter the number of astrocytes<sup>118</sup>. Conversely, ablation of IGF-I expression during development results in reduced brain size and a selective loss of neurons, such as striatal parvalbumin-containing neurons, but dopaminergic neurons and spinal cord motorneurons appear to be unaffected in IGF-I KO mice<sup>99</sup>.

IGF-I may also promote recruitment of neuronal cells during development. Recently IGF-I was reported to be associated with radial cells

198

and to promote neuronal recruitment from the ependyma/subependyma in the adult songbird<sup>119</sup>. Whether IGF-I also serves a similar function in mammals needs to be confirmed.

### 4.2.2 Neurite growth and synaptogenesis

Both *in vitro* and *in vivo* data show that IGF-I promotes neurite growth. In culture, IGF-I not only increases the length of neurites<sup>120-125</sup>, but also enhances the initiation of neurite outgrowth and neurite branching<sup>122</sup>. Unlike brain-derived neurotrophic factor (BDNF) and neurotrophin-3 (NT-3), which only increase the basal dendrites of pyramidal cells in rat primary somatosensory cortex, IGF-I increased the branching and total extent of both apical and basal dendrites<sup>124</sup>. Similarly, overexpression of IGF-I in the CNS greatly increases neuropil, which is occupied primarily by neuronal processes<sup>27,112</sup>. Reducing IGF-I availability by blunting its expression<sup>126</sup> or overexpressing IGFBP-1<sup>27</sup> decreases dendrite and axon growth. These IGF-I actions are associated with the stabilization of tubulin mRNAs during neurite formation<sup>127</sup>and an increase in neurofilament mRNA expression (our unpublished data).

IGF-I also promotes synaptogenesis during postnatal development<sup>113,128</sup>. In the hippocampal dentate gyrus and hypoglossal nucleus, the total number of synapses is greatly increased in IGF-I Tg mice from P14 to P130<sup>113</sup>. The synapse-to-neuron ratio, however, is increased only at P28 and P35, and returns to normal values by P130<sup>113</sup>. The latter indicates that synapse elimination, a normal event during the course of brain development, remains intact. Because increasing myelination appears to play a role in synapse elimination and IGF-I is a potent stimulator of myelination (see later), IGF-I may be indirectly involved in controlling the normal number of synapses. Regardless of the latter speculation, these data indicate that IGF-I promotes formation and maturation of synapses during development. This finding is supported by the data of Camarero *et al.*<sup>117</sup>, which show that synapse distribution in the cochlear ganglion persists at an immature stage in the absence of IGF-I

# 4.3 IGF actions on oligodendrocytes and myelination

## 4.3.1 Oligodendrocyte lineage

Oligodendrocytes, arguably the most abundant CNS cell type, produce the myelin sheaths that surround axons and facilitate neuronal electrical transduction. As with the neuronal lineage, IGFs appear to affect cells of oligodendrocyte lineage during each stage of their development, and in turn

myelination. IGF-I and IGF-II significantly increase oligodendrocyte number in cultures derived both from rodents<sup>31</sup> and human<sup>129</sup> by promoting proliferation of oligodendrocyte precursors<sup>130,131</sup>, survival of both mature oligodendrocytes and their precursors<sup>132-134</sup>, and maturation of precursors<sup>31,129,134-136</sup>. Addition of IGFBP-1 or -2, two IGF inhibitory binding proteins, decreases oligodendrocyte differentiation in culture<sup>137</sup>. IGF-I actions on oligodendrocyte lineage seem to be developmental-stage specific. Mason and Goldman<sup>138</sup> reported that while in the presence of insulin at concentrations sufficiently high to interact with the IGFIR, IGF-I promotes differentiation of O4<sup>+</sup> oligodendrocyte precursors, but not A2B5<sup>+</sup> precursors, which are at an earlier development stage than O4<sup>+</sup> oligodendrocyte precursors.

Although the total number of oligodendrocytes is increased in IGF-I Tg mice<sup>27,139,140</sup> and decreased in mutant mice with reduced IGF-I availability<sup>27,99,100</sup>, there are little *in vivo* data that directly address whether IGF-I acts to promote proliferation and/or survival of oligodendrocyte lineage cells during development. The lack of data stems from: (1) the relative paucity of stage-specific markers for cells in oligodendrocyte lineage and (2) the difficulty in ascertaining the identity of apoptotic cells during development. For example, the cell membrane lipid A2B5 is a wellrecognized oligodendrocyte precursor marker in culture, but its rapid disintegration and removal of apoptotic cells abrogates its use as a marker of apoptotic oligodendrocytes in vivo. Nonetheless, several recent studies clearly demonstrate that IGF-I promotes survival of mature oligodendrocytes in IGF-I Tg mice during demyelinating injury induced by cuprizone<sup>141</sup> and axonectomy<sup>142</sup>. Importantly, in the presence of IGF-I, surviving oligodendrocytes have the capacity to induce remyelination<sup>141</sup>.

#### 4.3.2 Myelination

One of the most dramatic effects of IGF-I is its stimulation of increased myelin content. Myelin content is increased by four-fold in the brain of IGF-I Tg mice<sup>143</sup>, and dramatically reduced in mice with reduced IGF-I availability (IGF-I KO and IGFBP-1 Tg mice)<sup>25,27,104,139,144</sup>. This increase in content results from IGF-I-induced increases in myelin sheath thickness (more sheaths enwrapped around each axon) and in more myelinated axons<sup>27,139</sup> (Fig. 2). As would be expected from these findings, the expression of myelin basic protein (MBP) and proteolipid protein (PLP), two proteins expressed only in mature oligodendrocytes and myelin, is increased in IGF-I Tg mice and reduced in mutant mice with reduced IGF-I availability<sup>27,140,143</sup>. Because the magnitude of changes in expression of

myelin-related protein genes cannot be fully accounted for by the changes in oligodendrocyte number, it appears that IGF-I specifically stimulates oligodendrocyte myelin synthesis.



*Figure 2.* Electron microphotographs of posterior anterior commissure of a wild-type control mouse (a), a IGF-I Tg mouse (b), and a IGFBP-1 Tg mouse (c). Tg and wild-type control mice were sacrificed and perfusion-fixed at postnatal day 35. Scale bars = 1  $\mu$ m. See Ye *et al.* <sup>139</sup> for more information on this study.

In the absence of IGF-I expression, IGF-II seems to be able to compensate, at least in part, for IGF-I actions on oligodendrocyte development and myelination in murine. Ye *et al.*<sup>140</sup> reported that the number of oligodendrocytes and the concentration of myelin-specific proteins are reduced during early postnatal development, but approach normal in adult IGF-I KO mice. The latter observation correlates with an increase in IGF-II abundance. A similar increase in serum IGF-II also was observed in a young man with an IGF-I null gene mutation<sup>145</sup>.

## 4.4 IGF actions on astrocytes and microglia

Little is known about IGF actions in the development of astrocytes and microglia, although many studies have demonstrated that IGFs increase the number of astrocytes and microglia. In culture, addition of IGF-I significantly increases proliferation of astrocytes<sup>146,147</sup> and microglia<sup>148</sup>. Although astrocytes comprise their normal percentage of neural cells in adult IGF-I Tg mice, their total number is increased as is the total number of brain cells (our unpublished data).

During injury and in a variety of CNS disorders, both astrocytes and microglia become reactive and produce IGFs. While active astrocytes appear to produce only IGF-I<sup>149-152</sup>, microglia secrete both IGF-I and IGF-II<sup>148-150,152,153</sup>. Because increased IGF expression by astrocytes and microglia occurs at or near sites of injury, the IGFs produced by these cells appear to act locally in an autocrine or paracrine fashion. Ectopic expression of IGFBP-1 retards brain development and reduces astrocyte response to injury<sup>104</sup>. In contrast, when IGF-I Tg mice are subjected to treatment with cuprizone, a copper chelator that induces oligodendrocyte damage, there is a dramatic increase in the number of microglia<sup>141</sup>.

# 4.5 Mechanisms of IGF-I actions in the CNS

It is likely IGF-I acts in CNS primarily in a paracrine and autocrine fashion. As indicated earlier, neural stem cell precursors in the VZ and SVZ cells express IGF-I<sup>154</sup>. Addition of IGF-I antibody or inhibitory IGFBPs, or an IGFIR antagonist peptide (JB1) blunts IGF-I actions on proliferation<sup>93,155</sup>, survival<sup>107</sup>, and differentiation <sup>96</sup>.

The intracellular signaling pathways that mediate IGF-I actions in CNS development remain to be precisely elucidated. Two major signaling pathways, the mitogen activated protein (MAP) kinase and the phosphatidylinositol-3 (PI-3) kinase pathways clearly play an important role in mediating IGF-I actions in neural cell development. IGFs bind and

activate the IGFIR, which in turn activates multiple signaling pathways including MAP and PI-3 kinase pathways. MAP kinase pathway appears to be involved in progenitor proliferation in both young and adult mice. Inhibition of MAP kinase activity ablates IGF-I-stimulated stem cell proliferation<sup>95</sup>. In mature neurons the IGF-I receptor may be coupled to an inhibitory G protein, which mediates G-dependent MAP kinase activation<sup>156</sup>.

The PI3 kinase-Akt pathway plays a key role in the promotion of survival of neural cells, including neural stem cells, precursors, and mature neurons, and oligodendrocytes. In cultured dentate gyrus<sup>107</sup> and cerebellar granule cells<sup>107,157</sup> and immature oligodendrocytes<sup>158,159</sup>, IGF-I activates PI-3 kinase and its downstream serine/threonine kinases, Akt and p70 S6 kinase. Activated Akt in turn acts by inhibiting forkhead transcription factor activity<sup>160</sup>. Another Akt-mediated pro-survival pathway leads to decreased caspase-3<sup>115,161</sup> and caspase-9<sup>162</sup> activity by its capacity to phosphorylate these caspases<sup>162</sup>, which reduces their activity, and by reducing their expression<sup>115</sup>.

Akt kinase activity also appears to be involved in IGF-I-mediated proliferation and growth of neural cells. For example, IGF-I-stimulated Akt activation enhances dendritic growth in cultured Purkinje cells<sup>116</sup> and proliferation of rat hippocampal adult stem cells<sup>95</sup>. These Akt effects are likely associated with its ability to regulate cell glucose transportation and energy production. IGF-I-stimulated Akt activation also significantly increases glucose transporter (GLUT4) expression and glucose uptake in mouse brain neurons<sup>163,164</sup>. In addition, this pathway appears to enhance Na<sup>+</sup>K<sup>+</sup>ATPase activity cultured rat astrocytes, which in turn may promote mitogenesis<sup>146</sup>.

The upstream and further downstream events in these IGF-I signaling pathways are less clear. Blunting the expression of either IRS-1 or IRS-2 expression reduces brain growth<sup>103,165</sup>, myelination<sup>166</sup>, and neuronal proliferation (by 50%)<sup>165</sup>, suggesting that both IRSs play an important role in mediating IGF actions on brain development. Further studies, however, show that blunting IRS-1 only minimally blocks IGF-I-stimulated brain growth and has no effects on myelin-associated protein expression<sup>103</sup>, suggesting that IRS-1 is not essential for IGF-I signaling in oligodendrocytes, and that other IRSs, such as IRS-2 and IRS-4, likely play an important role in oligodendrocyte and brain development. Recent studies indicate that IRS-2 is essential for IGF-I stimulated neuron progenitor proliferation, but not for IGF-I anti-apoptotic action<sup>165</sup>.

Activation of protein kinase C and stabilization of intracellular calcium levels also plays a role in mediating IGF effects on rat astrocyte proliferation<sup>167</sup>, c-*fos* expression in postmitotic neurons<sup>168</sup>, and neuron survival<sup>169</sup>.

## 5. IGF SYSTEM AND INJURY

Much experimental evidence indicates that the IGFs, primarily IGF-I, have a central role in repair, recovery and/or amelioration, following a wide variety of brain injuries. This data comes both from experiments in cultured neural cells and in animals. Table 3 lists a number of studies performed in animals that support an *in vivo* role for IGF-I. As during development, IGF-I appears to ameliorate CNS injury by at least two mechanisms: (1) inhibition of apoptosis and (2) stimulating the proliferation either of neural stem cells and/or committed neural progenitors.

IGF-I's capacity to inhibit apoptosis both in cultured cells and during development is well established (see earlier). While most *in vivo* studies of IGF-I and injury have evaluated neuroprotection, relatively few have specifically addressed apoptosis. Nonetheless, there is clear evidence that IGF-I inhibits apoptosis in neurons<sup>170-172</sup> and in oligodendrocytes<sup>141,161</sup>. Presumably much, if not most, of IGF-I neuroprotection, such as reduction in infarct size following hypoxic-ischemic (H/I) injury, is due to its antiapoptotic actions<sup>173</sup>. More recently, IGF-I has been shown to stimulate neuron progenitor proliferation in the dentate gyrus following middle cerebral artery ligation<sup>174</sup>. It is not clear, however, whether IGF-I is primarily stimulating proliferation and development in neural stem cells (NSC) or in neuron progenitors. Given that IGF-I appears to stimulate proliferation of cultured uncommitted NSC<sup>33,93</sup>, both seem possible<sup>173</sup>.

IGF neuroprotective actions appear to apply to virtually all brain injuries (Table 3), including ischemia, H/I, chemically induced injuries, cryo-injury, electrolytic injury, and so forth. The relationship of IGF system expression and action to H/I injury has been extensively studied, and may be applicable to many CNS injuries (see review in ref.173). Immediately after H/I injury, IGF-I expression falls, and this correlates inversely with the magnitude of apoptosis<sup>170</sup>. Beginning about 2 days after injury, IGF-I expression increases, peaking 5-7 days following injury (more slowly in adult animals). Astrocytes and reactive microglia appear to be the major sites of this expression<sup>152,173-175</sup>, although expression by neurons, oligodendrocytes, and NSCs also seems possible and has not been excluded. Following other injuries, astrocytes

possible and has not been excluded. Following other injuries, astrocytes appear to be the major site of IGF-I expression. Examples include stereotaxic injury<sup>176</sup>, electrolytic lesion to the hippocampus<sup>177</sup>, cuprizone<sup>151</sup>, and experimental autoimmune encephalomyelitis<sup>178</sup>. Increased IGF-II expression appears to occurs much later following H/I (7-10 days), and appears to be derived from microglia and macrophages<sup>152,173</sup>, but could also be derived from its usual sites of synthesis, such as the choroids plexus, etc.<sup>175</sup>. Changes in the expression of the IGFIR following H/I have not been clearly demonstrated.

Injury	IGF-I	Animal	Observation
	Treatment		
Amyloid-β injury	Systemic	Rat	Reduces amyloid-β burden <sup>179</sup>
Cerebral artery	Intranasal	Rat	Reduced infarct volume <sup>180</sup>
occlusion	I.C.V,	Rat	Reduced infarct volume <sup>181</sup>
	I.C.V.	Sheep (fetal)	Reduced oligodendrocyte loss <sup>161</sup>
	I.C.V.	Rat	Increased neuron progenitor cell
			proliferation <sup>174</sup>
Cerebellar deafferentation	Systemic	Rat	Restored motor function; decrease apoptosis <sup>171</sup>
Colchicine	None	Rat	Decreased IGF-II binding; increased IGF-
injection in			Il gene expression; increased IGF-I binding
dentate gyrus			and IGFBP-3 gene expression <sup>175</sup>
Contusion	None	Rat	Up-regulation IGFBP-2, -4 and IGF-1 mRNA <sup>182</sup>
Cuprizone	None	Rat	Increased IGF-I expression in astocytes <sup>151</sup>
	Transgene	IGF-I over-	Inhibition of oligodendrocyte apoptosis <sup>141</sup>
	expression	expression, mice	
Cryogenic	None	Rat	Increase in IGF-I mRNA <sup>183</sup>
Electrolytic lesion	None	Rat	Increase IGF-I peptide <sup>177</sup>
to hippocampus			
Experimental	None	Rat	Increased IGF-I and IGFBP-2 expression
Autoimmune			surrounding lesions <sup>178</sup>
encephalomyelitis	Systemic	Rat	IGF-I ameliorates injury <sup>184,185</sup>
Hypoxia/	None	Rat	Increased serum IGF-I, increase IGF-I in
ischemia			astrocytes and reactive microglia; increases
		*************	in IGFBPs; IGF-IR unchanged <sup>173</sup>
		Rat	Increased IGF-I and IGFBP-5 expression <sup>186</sup>
	None	Rat	Increase IGF-I and GH receptor mRNA
	Systemic GH		Moderate neuroprotection <sup>16</sup>
	None	Mouse	Induction IGF, IGFBP-2, -3, and -5
			4 days post-injury <sup>446</sup>
Ischemia	None	Rat	Increased IGF-1 expression <sup>100</sup>
			Immediate decreased IGF-1 mRNA that
	<u>ът.</u>	<b>D</b>	correlates with increased apoptosis."
Penetrating	None	Rat	induction of IGF-1 expression in
		Pat	Increased comm ICE II protein, chronic
		Kai	increase in IGF-II mRNA <sup>56</sup>
	Injection into	Rat	Reduced apoptosis; reduced post-trauma
	injured area		weight loss; increased post-trauma motor
	post-injury		activity <sup>1/2</sup>
Quinolinic acid	Intrastriatal	Rat	Rescued projection, cholinergic NADPH
	injection of		interneurons."
	IGF N-		
	tripentido		
	urpepude	1	

*Table 3.* Evidence from experimental animals that IGF system proteins are involved in recovery and/or amelioration of CNS injury.

Expression of IGFBPs is coordinated with that of IGF-I. The precise sites of expression appear to depend upon the area of the lesion<sup>173</sup>. In general, the

normally abundant expression of IGFBP-2 in the brain is increased further in the area surrounding the injury 3-5 days following H/I<sup>152</sup>. IGFBP-3, which does not appear to be widely expressed in the brain under homeostatic conditions, is induced in a similar temporal fashion and in association with the vasculature<sup>22</sup>. IGFBP-4 expression also is increased in the first 3-5 days following H/I, but it is located predominantly within the injured area. Later, IGFBP-5 expression increases in the injured area. IGFBPs modulate IGF actions, both by inhibiting and augmenting IGF activity, but they also have independent actions<sup>3</sup>. IGFBPs, therefore, may have role following injury that do not involve modulating IGF actions. Such IGFBP action remains to be identified.

injury<sup>173</sup>. IGF-I treatment clearly is protective against H/I Intracerebroventricular (ICV) IGF-I has dose- and time- dependent effects on H/I injury. ICV IGF-I administered within hours of injury can dramatically ameliorate neuron loss and infarct size. These effects, however, differ according to region, with the greatest IGF-I effect being in the lateral parietal cortex, followed by the striatum, the dentate gyrus, and thalamus, with the least amelioration being in the hippocampus. Specific neuron types appear to be differentially protected<sup>189</sup>. Other studies have demonstrated that systemic or intranasal IGF-I administration also can be effective in treatment of neural injury<sup>184,185,190,191</sup>, including ischemia<sup>191</sup>. As these results strongly indicate, IGF-I can cross the blood-brain barrier<sup>7</sup>.

IGF-I also has actions that may protect against chronic and inherited neurologic diseases. For example, IGF-I has been shown to decrease βamyloid burden in Tg mice that overexpress  $\beta$ -amyloid<sup>179</sup>. This is supported by data showing that IGF-I protects cultured cells from β-amyloid peptide induced neuron death<sup>192</sup>. Taking together these results, it appears that IGF-I may ameliorate Alzheimer's disease. Studies in cultured cells also indicate that IGF-I can protect against the neurotoxicity caused by dopamine<sup>193</sup>, glutaric acid<sup>194</sup> and the mutant Huntington protein<sup>195</sup>, findings that suggest that IGF-I might be helpful in the treatment of early Parkinson's disease, glutaric acidemia type I, and Huntington's disease. In addition, systemic IGF-I infusion delays Purkinje cell degeneration in "shaker" mutant rats<sup>196</sup>. Finally, there is strong evidence that IGF-I promotes oligodendrocyte survival and myelination, both in culture<sup>131</sup> and *in vivo*<sup>27,103,139,143</sup>, and may protect against a variety of demyelinating injuries<sup>131,134,140,141</sup>. These findings suggest that IGF-I may be useful in the treatment of a wide variety of demyelinating diseases and injuries.

### 6. HUMAN CLINICAL CORRELATIONS

A few mutations in IGF-I and IGFIR genes have now been identified in humans. Each of these mutations is characterized by severe intrauterine growth retardation (IUGR) and poor postnatal growth. Although highly variable, each also has been associated with findings consistent with CNS abnormalities seen in mutant animals. Woods *et al.*<sup>145</sup> reported a teenager with a homozygous deletion in the *IGF-I* gene. His head circumference at birth was 4.9 SD below the mean, commensurate with the magnitude of length (-5.4 SD) and weight retardation (-3.9 SD). He also exhibited delays in his developmental milestones and subsequent mild mental retardation, as well as sensorineural deafness. More recently, another child has been found to have very low IGF-I expression resulting from a point mutation in the polyadenylation sequence of the *IGF-I* gene<sup>197</sup>. This child also had severe IUGR, a small head circumference, and sensorineural deafness. Brain magnetic resonance imaging (MRI) studies, however, were reported as normal in both.

Two IGFIR gene mutations have been reported by Abuzzahab et al.<sup>198</sup>. One was identified as a compound heterozygous missense mutation in exon 2. This exon encodes the portion of the IGFIR alpha subunit, which binds IGF-I. When studied in the patient's cultured skin fibroblasts, this mutation was found to result in decreased IGF-I receptor affinity and decreased IGF-Istimulated IGFIR phosphorylation (but with an apparent normal receptor number). At age 12 years her verbal IQ was 134, but performance IQ was 89 and she exhibited multiple behavioral abnormalities, such as anxious affect, psychomotor agitation, obsessive tendencies, and social phobias. Neither head circumference nor head imaging studies were reported. The second child reported manifested a heterozygous point mutation resulting in a stop codon in exon 2. Further evaluation of the patient's fibroblasts by flow cytometry indicated that the number of IGFIRs expressed was decreased. At birth this child had microcephaly (head circumference = -4.6 SD), again similar in magnitude to the IUGR (-3.5 SD). He is reported to have mild retardation of motor and speech development. Okubo et al. identified another child with a heterozygous IGFIR deletion on the basis of a 15q26.2 chromosomal deletion, an IGFIR specific fluorescence in situ hybridization (FISH) analysis indicating a single IGF-I gene, and reduced IGFIR expression and IGF-I-stimulated receptor phosphorylation of the patients cultured fibroblasts<sup>199</sup>. This child had moderately delayed motor and speech development, and at 10 years of age had learning difficulties.

The neurological and psychological manifestations exhibited by these children indicate that IGF signaling has a role in human brain development and function. In other words, it seems clear that either a decrease in IGF-I expression or decreased IGF signaling, whether it be secondary to decreased IGFIR number or binding affinity, leads to brain abnormalities. Nonetheless, the nature of IGF actions in the human CNS is far from being defined. More precise brain imagining and functional studies performed during the course of development may help to better define brain abnormalities in such children. Regardless, these clinical situations raise numerous questions. As appears to be the case in mice<sup>103</sup>, can increased brain IGF-II expression compensate in part for the absence of IGF-I in the children with *IGF-I* gene mutations, and thus, mask some IGF-I actions? Does IGF-I or IGF-II expression increase in the CNS in response to decreased IGFIR signaling, as is the case in disorders resulting from resistance to other hormones?

# 7. FUTURE DIRECTIONS

While major insights into the actions of GH and the IGF system on CNS growth and development continue to accumulate, it is clear our understanding remains primitive. The fundamental questions and issues that remain are many, but some that seem particularly important follow. Distinguishing direct GH actions on the brain from those that result from GH stimulation of IGF expression is problematic. IGFs stimulate an extraordinarily wide variety of influences on neural cells, including neural progenitor proliferation, inhibition of neuron and oligodendrocyte apoptosis, neuritic outgrowth, oligodendrocyte myelin synthesis, and so forth. It is unlikely that IGF is sending instructive signals for each of these events. Rather, it seems certain that IGFs are acting in concert with other growth factors and signaling molecules that give instructive signals. The nature of these interactions, the signaling pathways they utilize, and their developmental context are crucial to understanding the roles of IGFs in CNS development. Some other mechanistic questions relevant to CNS development are: (1) Do IGF-I and IGF-II have distinct actions/roles? (2) What signaling pathways are activated by IGFs to determine their apparent myriad of distinctly different effects? (3) Does the ciM6PR/IGFIIR serve a role in IGF-II signaling? and (4) What are precise actions of IGFBPs, both those in modulating IGF actions and those that are independent of IGFs? Cognitive consequences of GH and IGF actions, which seem likely to occur, are poorly defined, and for the most part, not well documented. Finally and perhaps most importantly, will IGFs prove useful in the treatment of neurological injury and disease?

#### ACKNOWLEDGEMENTS

The research of the authors reported in this chapter was supported by research grants RO1-08299 and RO1-038891 from NICHD and NINDS, respectively, and training grant T32 DK007129 from NIDDK of the U.S. National Institutes of Health.

### REFERENCES

- 1. C.E.H. Stewart and P. Rotwein, Growth, differentiation, and survival: Multiple physiological functions for insulin-like growth factors. *Physiol Rev*, **76**, 1005-1026 (1996).
- 2. J. Nakae, Y. Kido and D. Accili, Distinct and overlapping functions of insulin and IGF-I receptors. *Endocrinol Rev*, **22**, 818-835 (2001).
- 3. J.I. Jones and D.R. Clemmons, Insulin-like growth factors and their binding proteins: Biological actions. *Endocrinol Rev*, **16**, 3-34 (1995).
- 4. D.R. Clemmons, Role of insulin-like growth factor binding proteins in controlling IGF actions. *Mol Cell Endocrinol*, **140**, 19-24 (1998).
- 5. A.J. D'Ercole, P. Ye and J.R. O'Kusky, Mutant mouse models of insulin-like growth factor actions in the central nervous system. *Neuropeptides*, **36**, 209-220 (2002).
- 6. R.R. Reinhardt and C.A. Bondy, Insulin-like growth-factors cross the blood-brainbarrier. *Endocrinology*, **135**, 1753-1761 (1994).
- 7. B.E. Pulford and D.N. Ishii, Uptake of circulating insulin-like growth factors (IGFs) into cerebrospinal fluid appears to be independent of the IGF receptors as well as IGF-binding proteins. *Endocrinology*, **142**, 213-220 (2001).
- S. Yakar, J.L. Liu, B. Stannard, A. Butler, D. Accili, B. Sauer and D. LeRoith, Normal growth and development in the absence of hepatic insulin- like growth factor I. *Proc Natl Acad Sci USA*, 96, 7324-7329 (1999).
- S. Yakar, C.J. Rosen, W.G. Beamer, C.L. Ackert-Bicknell, Y. Wu, J.L. Liu, G.T. Ooi, J. Setser, J. Frystyk, Y.R. Boisclair and D. LeRoith, Circulating levels of IGF-1 directly regulate bone growth and density. *J Clin Invest*, **110**, 771-781 (2002).
- 10. M. Coculescu, Blood-brain barrier for human growth hormone and insulin-like growth factor-I. J Ped Endocrino Met, 12, 113-124 (1999).
- 11. Z.N. Lai, M. Emtner, P. Roos and F. Nyberg, Characterization of putative growthhormone receptors in human choroid-plexus. *Brain Res*, **546**, 222-226 (1991).
- 12. F. Gossard, F. Dihl, G. Pelletier, P.M. Dubois and G. Morel, *In situ* hybridization to rat-brain and pituitary-gland of growth-hormone cDNA. *Neurosci Lett*, **79**, 251-256 (1987).
- H. Yoshizato, T. Fujikawa, H. Soya, M. Tanaka and K. Nakashima, The growth hormone (GH) gene is expressed in the lateral hypothalamus: Enhancement by GHreleasing hormone and repression by restraint stress. *Endocrinology*, 139, 2545-2551 (1998).
- 14. S. Harvey, C.D.M. Johnson and E.J. Sanders, Growth hormone in neural tissues of the chick embryo. *J Endocrinol*, **169**, 487-498 (2001).
- J. Garciaaragon, P.E. Lobie, G.E.O. Muscat, K.S. Gobius, G. Norstedt and M.J. Waters, Prenatal expression of the growth-hormone (GH) receptor-binding protein in the rat - A role for GH in embryonic and fetal development. *Development*, **114**, 869-876 (1992).

- P.E. Lobie, J. Garciaaragon, D.T. Lincoln, R. Barnard, J.N. Wilcox and M.J. Waters, Localization and ontogeny of growth-hormone receptor gene expression in the central nervous system. *Dev Brain Res*, 74, 225-233 (1993).
- 17. F. Nyberg and P. Burman, Growth hormone and its receptors in the central nervous system Location and functional significance. *Horm Res*, **45**, 18-22 (1996).
- 18. F. Nyberg, Aging effects on growth hormone receptor binding in the brain. *Exper Gerontol*, **32**, 521-528 (1997).
- 19. Q. Zhai, Z. Lai, P. Roos, and F. Nyberg, Characterization of growth-hormone binding sites in rat brain. *Acta Paediatrica*, **83**, 92-95 (1994).
- Z.N. Lai, P. Roos, Q.Z. Zhai, Y. Olsson, K. Fholenhag, C. Larsson and F. Nyberg, Age-related reduction of human growth hormone-binding sites in the human brain. *Brain Res*, 621, 260-266 (1993).
- J.R. Castro, J.A. Costoya, R. Gallego, A. Prieto, V.M. Arce and R. Senaris, Expression of growth hormone receptor in the human brain. *Neurosci Lett*, 281, 147-150 (2000).
- 22. W.H. Lee, G.M. Wang, X.L. Yang, L.B. Seaman and S.I. Vannucci, Perinatal hypoxia-ischemia decreased neuronal but increased cerebral vascular endothelial IGFBP3 expression. *J Endocrinol*, **11**, 181-188 (1999).
- S. Busiguina, J.A. Chowen, J. Argente and I. Torres-Aleman, Specific alterations of the insulin-like growth factor I system in the cerebellum of diabetic rats. *Endocrinology*, 137, 4980-4987 (1996).
- A.J. D'Ercole, Z. Dai, Y. Xing, C. Boney, M.B. Wilkie, J.M. Lauder, V.K. Han and D.R. Clemmons, Brain growth retardation due to the expression of human insulin like growth factor binding protein-1 in transgenic mice: an in vivo model for the analysis of IGF function in the brain. *Brain Res Dev Brain Res*, 82, 213-222 (1994).
- L.J. Murphy, K. Rajkumar and P. Molnar, Phenotypic manifestations of insulin-like growth factor binding protein-1 (IGFBP-1) and IGFBP-3 overexpression in transgenic mice. *Prog Growth Factor Res*, 6, 425-432 (1996).
- 26. L.J. Murphy, Overexpression of insulin-like growth factor binding protein-1 in transgenic mice. *Ped Nephol*, **14**, 567-571 (2000).
- P. Ye, J. Carson and A.J. D'Ercole, In vivo actions of insulin-like growth factor-I (IGF-I) on brain myelination: studies of IGF-I and IGF binding protein-1 (IGFBP-1) transgenic mice. *J Neurosci*, **15**, 7344-7356 (1995).
- 28. C.A. Bondy and W.H. Lee, Patterns of insulin-like growth factor and IGF receptor gene expression in the brain. Functional implications. *Ann NY Acad. Sci*, **692**, 33-43 (1993).
- 29. I.K. Andersson, D. Edwall, G. Norstedt, B. Rozell, A. Skottner and H.-A. Hansson, Differing expression of insulin-like growth factor I in the developing and in the adult rat cerebellum. *Acta Physiol Scand*, **132**, 167-173 (1988).
- R. Ballotti, F.C. Nielsen, N. Pringle, A. Kowalski, W.D. Richardson, E. Van Obberghen and S. Gammeltoft, Insulin-like growth factor I in culture rat astrocytes: expression of the gene, and receptor tyrosine kinase. *EMBO J*, 6, 2620-2626 (1987).
- F.A. McMorris, T.M. Smith, S. DeSalvo and R.W. Furlanetto, Insulin-like growth factor I/somatomedin C: A potent inducer of oligodendrocyte development. *Proc Natl Acad Sci USA*, 83, 822-826 (1986).
- Y. Shinar and F.A. McMorris, Developing oligodendroglia express mRNA for insulin-like growth factor-I, a regulator of oligodendrocyte development. *J Neurosci Res*, 42, 516-527 (1995).
- M. Perez-Martin, M. Cifuentes, J.M. Grondona, F.J. Bermudez-Silva, P.M. Arrabal, J.M. Perez-Figares, A.J. Jimenez, L.M. Garcia-Segura and P. Fernandez-Llebrez, Neurogenesis in explants from the walls of the lateral ventricle of adult bovine brain: role of endogenous IGF-1 as a survival factor. *Eur J Neurosci*, **17**, 205-211 (2003).

- C. Vicario-Abejon, M.J. Yusta-Boyo, C. Fernandez-Moreno and F. De Pablo, Locally born olfactory bulb stem cells proliferate in response to insulin-related factors and require endogenous insulin-like growth factor-I for differentiation into neurons and glia. J Neurosci, 23, 895-906 (2003).
- C. Bondy and E. Chin, IGF-I mRNA localization in trigeminal and sympathetic nerve target zones during rat embryonic development. *Adv Exp Med*, 293, 431-437 (1991).
- W.P. Bartlett, X.S. Li, M. Williams and S. Benkovic, Localization of insulin-like growth factor-1 mRNA in murine central nervous system during postnatal development. *Dev Biol*, 147, 239-250 (1991).
- W.-H. Lee, K.M. Michels and C.A. Bondy, Localization of insulin-like growth factor binding protein- 2 messenger RNA during postnatal brain development: Correlation with insulin-like growth factors I and II. *Neuroscience*, 53, 251-265 (1993).
- C. Ayer-LeLievre, P.A. Stahlbom and V.R. Sara, Expression of IGF-I and -II mRNA in the brain and craniofacial region of the rat fetus. *Development*, **111**, 105-115 (1991).
- G.J. Popken, R.D. Hodge, P. Ye, J. Zheng, W. Ng, J.R. O'Kusky, and A.J. D'Ercole, *In vivo* effects of insulin-like growth factor-I (IGF-I) on prenatal and early postnatal development of the central nervous system. *Eur J Neurosci*, **19**, 2056-68 (2004).
- P. Rotwein, S.K. Burgess, J.D. Milbrandt and J.E. Krause, Differential expression of insulin-like growth factor genes in rat central nervous system. *Proc Natl Acad Sci* USA, 85, 265-269 (1988).
- P. Ye, Y. Umayahara, D. Ritter, T. Bunting, H. Auman, P. Rotwein and A.J. D'Ercole, Regulation of insulin-like growth factor-I (IGF-I) gene expression in brain of transgenic mice expressing an IGF-I-luciferase fusion gene. *Endocrinology*, 138, 5466-5475 (1997).
- A. Baron-Van Evercooren, C. Olichon-Berthe, A. Kowalski, G. Visciano and E. Van Obberghen, Expression of IGF-I and insulin receptor genes in the rat central nervous system: A developmental, regional, and cellular analysis. *J Neurosci Res*, 28, 244-253 (1991).
- H. Werner, M. Woloschak, M. Adamo, Z. Shen-Orr, C.T. Roberts, Jr. and D. LeRoith, Developmental regulation of the rat insulin-like growth factor I receptor gene. *Proc Natl Acad Sci USA*, 86, 7451-7455 (1989).
- S.U. Devaskar, H.F. Sadiq, L. Holtzclaw and M. George, The developmental pattern of rabbit brain insulin and insulin-like growth-factor receptor expression. *Brain Res*, 605, 101-109 (1993).
- 45. J.L. Marks, D. Porte and D.G. Baskin, Localization of type-I insulin-like growthfactor receptor messenger-RNA in the adult rat brain by *in situ* hybridization. *Mol Endocrinol*, **5**, 1158-1168 (1991).
- 46. S. Kar, J.-G. Chabot and R. Quirion, Quantitative autoradiographic localization of [<sup>125</sup>I]insulin- like growth factor I, [<sup>125</sup>I]insulin-like growth factor II, and [<sup>125</sup>I]insulin receptor binding sites in developing and adult rat brain. J Comp Neurol, 333, 375-397 (1993).
- S. Quiroga, R.S. Garofalo and K.H. Pfenninger, Insulin-like growth-factor-I receptors of fetal brain are enriched in nerve growth cones and contain a betasubunit variant. *Proc Natl Acad Sci U SA*, **92**, 4309-4312 (1995).
- 48. L.M. Garcia-Segura, J.R. Rodriguez and I. Torres-Aleman, Localization of the insulin-like growth factor I receptor in the cerebellum and hypothalamus of adult rats: an electron microscopic study. *J Neurocyt*, **26**, 479-490 (1997).
- 49. R.S.K. Florance, P.V. Senior, S. Byrne and F. Beck, The expression of IGF-II in the early post-implantation rat conceptus. *J Anat*, **175**, 169-179 (1991).

- 50. K.A. Sullivan and E.L. Feldman, Immunohistochemical localization of insulin-like growth factor- II (IGF-II) and IGF-binding protein-2 during development in the rat brain. *Endocrinology*, **135**, 540-547 (1994).
- 51. F. Beck, F. Samani, J.D. Penchow, B. Thorley, G.W. Tregear and J.P. Coghlan, Histochemical localization of IGF-I and -II mRNA in the developing rat embryo. *Development*, **101**, 175-184 (1987).
- 52. T. Cavallaro, R.L. Martone, F. Stylianopoulou and J. Herbert, Differential expression of the insulin-like growth factor II and transthyretin genes in the developing rat choroid plexus. *J Neuropath Exp Neurol*, **52**, 153-162 (1993).
- 53. M.A. Hynes, P.J. Brooks, J.J. Van Wyk and P.K. Lund, Insulin-like growth factor II messenger ribonucleic acids are synthesized in the choroid plexus of the rat brain. *Mol Endocrinol*, **2**, 47-54 (1988).
- A. Logan, A.-M. Gonzalez, D.J. Hill, M. Berry, N.A. Gregson, and A. Baird, Coordinated pattern of expression and localization of insulin- like growth factor-II (IGF-II) and IGF-binding protein-2 in the adult rat brain. *Endocrinology*, 135, 2255-2264 (1994).
- F. Stylianopoulou, A. Efstratiadis, J. Herbert and J. Pintar, Pattern of the insulinlike growth factor II gene expression during rat embryogenesis. *Development*, 103, 497-506 (1988).
- H.J. Walter, M. Berry, D.J. Hill, S. Cwyfan-Hughes, J.M.P. Holly and A. Logan, Distinct sites of insulin-like growth factor (IGF)-II expression and localization in lesioned rat brain: Possible roles of IGF binding proteins (IGFBPs) in the mediation of IGF-II activity. *Endocrinology*, 140, 520-532 (1999).
- 57. R. Birnbacher, G. Amann, H. Breitschopf, H. Lassmann, G. Suchanek and P. Heinz-Erian, Cellular localization of insulin-like growth factor II mRNA in the human fetus and the placenta: Detection with a digoxigenin-labeled cRNA probe and immunocytochemistry. *Ped Res*, **43**, 614-620 (1998).
- G.K. Haselbacher, M.E. Schwab, A. Pasi and R.E. Humbel, Insulin-like growth factor II (IGF II) in human brain: Regional distribution of IGF-II and of higher molecular mass forms. *Proc Natl Acad Sci USA*, 82, 2153-2157 (1985).
- 59. J.-F. Hu, T.H. Vu and A.R. Hoffman, Differential biallelic activation of three insulin-like growth factor II promoters in the mouse central nervous system. *Mol Endocrinol*, **9**, 628-636 (1995).
- 60. N.V. Pham, M.T. Nguyen, J.F. Hu, T.H. Vu, and A.R. Hoffman, Dissociation of *IGF2* and *H19* imprinting in human brain, *Brain Res*, **810**, 1-8 (1998).
- 61. J.F. Hu, K.A. Balaguru, R.D. Ivaturi, H. Oruganti, T. Li, B.T. Nguyen, T.H. Vu and A.R. Hoffman, Lack of reciprocal genomic imprinting of sense and antisense RNA of mouse insulin-like growth factor II receptor in the central nervous system. *Biochem Biophys Res Comm*, 257, 604-608 (1999).
- 62. P.A. McKelvie, K.M. Rosen, H.C. Kinney and L. Villa-Komaroff, Insulin-like growth factor II expression in the developing human brain. *J Neuropath Exp Neurol*, **51**, 464-471 (1992).
- 63. M. Holzenberger and F. Lapointe, Expression of insulin-like growth factor-I (IGF-I) and IGF-II in the avian brain: relationship of *in situ* hybridization patterns with IGF type 1 receptor expression. *Int J Dev Neurosci*, **18**, 69-82 (2000).
- 64. M.M. Sklar, W. Kiess, C.L. Thomas and S.P. Nissley, Developmental expression of the tissue insulin-like growth factor II/mannose 6-phosphate receptor in the rat. Measurement by quantitative immunoblotting. *J Biol Chem*, **264**, 16733-16738 (1989).
- 65. K.L. Valentino, I. Ocrant and R.G. Rosenfeld, Developmental expression of insulinlike growth factor- II receptor immunoreactivity in the rat central nervous system. *Endocrinology*, **126**, 914-920 (1990).

- C.C. Ferrari, B.A. Johnson, M. Leon and S.K. Pixley, Spatiotemporal distribution of the insulin-like growth factor receptor in the rat olfactory bulb. *Neurochem Res*, 28, 29-43 (2003).
- 67. C. Hawkes and S. Kar, Insulin-like growth factor-II/mannose-6-phosphate receptor: Widespread distribution in neurons of the central nervous system including those expressing cholinergic phenotype. *J Comp Neurol*, **458**, 113-127 (2003).
- M.E. Couce, A.J. Weatherington and J.F. McGinty, Expression of insulin-like growth factor-II (IGF-II) and IGF-II/mannose-6-phosphate receptor in the rat hippocampus: An *in situ* hybridization and immunocytochemical study. *Endocrinology*, **131**, 1636-1642 (1992).
- 69. A.K. Brar and S.D. Chernausek, Localization of insulin-like growth factor binding protein- 4 expression in the developing and adult rat brain: Analysis by *in situ* hybridization. *J Neurosci Res*, **35**, 103-114 (1993).
- K.L. Stenvers, E.M. Zimmermann, M. Gallagher and P.K. Lund, Expression of insulin-like growth factor binding protein-4 and -5 mRNAs in adult rat forebrain. J Comp Neurol, 339, 91-105 (1994).
- G.S. Naeve, A.M. Vana, J.R. Eggold, G. Verge, N. Ling, and A.C. Foster, Expression of rat insulin-like growth factor binding protein-6 in the brain, spinal cord, and sensory ganglia. *Mol Brain Res*, **75**, 185-197 (2000).
- H.-L. Cheng, K.A. Sullivan and E.L. Feldman, Immunohistochemical localication of insulin-like growth factor binding protein-5 in the developing rat nervous system. *Dev Brain Res*, 92, 211-218 (1996).
- 73. B.N. Green, S.B. Jones, R.D. Streck, T.L. Wood, P. Rotwein and J.E. Pintar, Distinct expression patterns of insulin-like growth factor binding proteins 2 and 5 during fetal and postnatal development. *Endocrinology*, **134**, 954-962 (1994).
- 74. S.J. Holley, S.B. Hall and P.L. Mellon, Complementary expression of IGF-II and IGFBP-5 during anterior pituitary development. *Dev Biol*, **244**, 319-328 (2002).
- S. Kubo, S. Ogino, T. Fukushima, P.R. Olson, M. Kida, M. Maruno, T. Yoshimine and T. Hyakawa, Immunohistochemical study of insulin-like growth factor II (IGF-II) and insulin-like growth factor binding protein-2 (IGFBP-2) in choroid plexus papilloma. *Neurol Res*, 21, 339-344 (1999).
- V.C. Russo, L.A. Bach, A.J. Fosang, N.L. Baker and G.A. Werther, Insulin-like growth factor binding protein-2 binds to cell surface proteoglycans in the rat brain olfactory bulb. *Endocrinology*, **138**, 4858-4867 (1997).
- 77. P. Ye and A.J. D'Ercole, Insulin-like growth factor I (IGF-I) regulates IGF-I binding protein-5 gene expression in the brain. *Endocrinology*, **139**, 65-71 (1998).
- S. Harvey, M. Kakebeeke, A.E. Murphy and E.J. Sanders, Growth hormone in the nervous system: autocrine or paracrine roles in retinal function? *Can J Physiol Pharm*, 81, 371-384 (2003).
- 79. F. Nyberg, Growth hormone in the brain: Characteristics of specific brain targets for the hormone and their functional significance. *Front Neuroendo*, **21**, 330-348 (2000).
- P. Burman, J.E. Broman, J. Hetta, I. Wiklund, E.M. Erfurth, E. Hagg and F.A. Karlsson, Quality of life in adults with growth-hormone (GH) deficiency Response to treatment with recombinant human GH in a placebo-controlled 21-month trial. *J Clin Endocrinol Metab*, **80**, 3585-3590 (1995).
- 81. T. Noguchi, Effects of growth hormone on cerebral development: Morphological studies. *Horm Res*, **45**, 5-17 (1996).
- D.M. Lehman, D.E. Hale, J.T. Cody, J.M. Harrison and R.J. Leach, Molecular, morphometric and functional analyses demonstrate that the growth hormone deficient little mouse is not hypomyelinated. *Dev Brain Res*, **116**, 191-199 (1999).
- 83. B.T. Shea, R.E. Hammer and R.L. Brinster, Growth allometry of organs in giant transgenic mice. *Endocrinology*, **121**, 1924-1930 (1987).
- L.M. Frago, C. Pañeda, S.L. Dickson, A.K. Hewson, J. Argente and J.A. Chowen, Growth hormone (GH) and GH-releasing peptide-6 increase brain insulin-like growth factor-I expression and activate intracellular signaling pathways involved in neuroprotection. *Endocrinology*, 143, 4113-4122 (2002).
- R. Ajo, L. Cacicedo, C. Navarro and F. Sanchez-Franco, Growth hormone action on proliferation and differentiation of cerebral cortical cells from fetal rat. *Endocrinology*, 144, 1086-1097 (2003).
- B.A. Bengtsson, S. Eden, L. Lonn, H. Kvist, A. Stokland, G. Lindstedt, I. Bosaeus, J. Tolli, L. Sjostrom and O.G.P. Isaksson, Treatment of adults with growth hormone (GH) deficiency with recombinant human GH. J Clin Endocrinol Metab, 76, 309-317 (1993).
- J. Gibney, J.D. Wallace, T. Spinks, L. Schnorr, A. Ranicar, R.C. Cuneo, S. Lockhart, K.G. Burnand, F. Salomon, P.H. Sonksen and D. Russell-Jones, The effects of 10 years of recombinant human growth hormone (GH) in adult GH-deficient patients. *J Clin Endocrinol Metab*, 84, 2596-2602 (1999).
- 88. G.A. McGauley, R.C. Cuneo, F. Salomon and P.H. Sonksen, Psychological well being before and after growth hormone treatment in adults with growth hormone deficiency. *Horm Res*, **33**, 52-54 (1990).
- I. Torres-Aleman, F. Naftolin and R.J. Robbins, Trophic effects of insulin-like growth factor-I on fetal rat hypothalamic cells in culture. *Neuroscience*, **35**, 601-608 (1990).
- 90. I. Torres-Aleman, F. Naftolin and R.J. Robbins, Trophic effects of basic fibroblast growth factor on fetal rat hypothalamic cells: Interactions with insulin-like growth factor I. *Dev Brain Res*, **52**, 253-257 (1990).
- K. Zackenfels, R.W. Oppenheim and H. Rohrer, Evidence for an important role of IGF-I and IGF-II for the early development of chick sympathetic neurons. *Neuron*, 14, 731-741 (1995).
- 92. Y. Arsenijevic and S. Weiss, Insulin-like growth factor-I is a differentiation factor for postmitotic CNS stem cell-derived neuronal precursors: Distinct actions from those of brain-derived neurotrophic factor. *J Neurosci*, **18**, 2118-2128 (1998).
- 93. Y. Arsenijevic, J.G. Villemure, J.F. Brunet, J.J. Bloch, N. Deglon, C. Kostic, A. Zurn and P. Aebischer, Isolation of multipotent neural precursors residing in the cortex of the adult human brain. *Exper Neurol*, **170**, 48-62 (2001).
- 94. N.L. Kennea and H. Mehmet, Neural stem cells. J Pathol, 197, 536-550 (2002).
- M.A.I. Aberg, N.D. Aberg, T.D. Palmer, A.M. Alborn, C. Carlsson-Skwirut, P. Bang, L.E. Rosengren, T. Olsson, F.H. Gage and P.S. Eriksson, IGF-I has a direct proliferative effect in adult hippocampal progenitor cells. *Mol Cell Neurosci*, 24, 23-40 (2003).
- G.J.F. Brooker, M. Kalloniatis, V.C. Russo, M. Murphy, G.A. Werther and F.F. Bartlett, Endogenous IGF-1 regulates the neuronal differentiation of adult stem cells. *J Neurosci Res*, **59**, 332-341 (2000).
- 97. J. Hsieh, J.B. Aimone, B.K. Kaspar, T. Kuwabara, K. Nakashima and F.H. Gage, IGF-1 instructs multipotent adult neural progenitor cells to become oligodendrocytes. *J Cell Biol*, **164**, 111-122 (2004).
- 98. J. Baker, J.-P. Liu, E.J. Robertson and A. Efstratiadis, Role of insulin-like growth factors in embryonic and postnatal growth. *Cell*, **75**, 73-82 (1993).
- 99. K.D. Beck, L. Powell-Braxton, H.-R. Widmer, J. Valverde and F. Hefti, Igf1 gene disruption results in reduced brain size, CNS hypomyelination, and loss of hippocampal and striatal parvalbumin-containing neurons. *Neuron*, **14**, 717-730 (1995).
- 100. C.M. Cheng, G. Joncas, R.R. Reinhardt, R. Farrer, R. Quarles, J. Janssen, M.P. McDonald, J.N. Crawley, L. Powell-Braxton and C.A. Bondy, Biochemical and morphometric analyses show that myelination in the insulin-like growth factor 1

null brain is proportionate to its neuronal composition. *J Neurosci*, **18**, 5673-5681 (1998).

- J.-P. Liu, J. Baker, A.S. Perkins, E.J. Robertson and A. Efstratiadis, Mice carrying null mutations of the genes encoding Insulin-like Growth Factor I (Igf-1) and type 1 IGF receptor, *Cell* **75**, 59-72 (1993).
- L. Powell-Braxton, P. Hollingshead, C. Warburton, M. Dowd, S. Pitts-Meek, D. Dalton, N. Gillett, and T. A. Stewart, IGF-I is required for normal embryonic growth in mice, *Genes Dev* 7, 2609-2617 (1993).
- 103. P. Ye, L.Q. Li, R.G. Richards, R.P. DiAugustine and A.J. D'Ercole, Myelination is altered in insulin-like growth factor-I null mutant mice, *J Neurosci* 22, 6041-6051 (2002).
- W. Ni, K. Rajkumar, J.I. Nagy and L.J. Murphy, Impaired brain development and reduced astrocyte response to injury in transgenic mice expressing IGF binding protein-1, *Brain Res* 769, 97-107 (1997).
- 105. D.M. Martin and E.L. Feldman, Reversibility of serum removal effects on IGF-II mRNA in human neuroblastoma cells, *Ann NY Acad Sci* **692**, 259-261 (1993).
- B. Cheng and M.P. Mattson, IGF-I and IGF-II protect cultured hippocampal and septal neurons against calcium-mediated hypoglycemic damage, *J Neurosci* 12, 1558-1566 (1992).
- C.M. Cheng, M. Cohen, V. Tseng and C. A. Bondy, Endogenous IGF1 enhances cell survival in the postnatal dentate gyrus, *J Neurosci Res* 64, 341-347 (2001).
- S.R. D'Mello, C. Galli, T. Ciotti and P. Calissano, Induction of apoptosis in cerebellar granule neurons by low potassium: Inhibition of death by insulin-like growth factor I and cAMP, *Proc Natl Acad Sci USA* **90**, 10989-10993 (1993).
- 109. I. Torres-Aleman, S. Pons and F.F. Santos-Benito, Survival of Purkinje cells in cerebellar cultures is increased by insulin-like growth factor I, *Eur J Neurosci* 4, 864-869 (1992).
- S. Pahlman, G. Meyerson, E. Lindgren, M. Schalling and I. Johansson, Insulin-like growth factor-I shifts from promoting cell division to potentiating maturation during neuronal differentiation, *Proc Natl Acad Sci USA* 88, 9994-9998 (1991).
- 111. K.D. Dentremont, P. Ye, A.J. D'Ercole and J.R. O'Kusky, Increased insulin-like growth factor-I (IGF-I) expression during early postnatal development differentially increases neuron number and growth in medullary nuclei of the mouse, *Dev Brain Res* **114**, 135-141 (1999).
- 112. G. Gutiérrez-Ospina, A.S. Calikoglu, P. Ye and A.J. D'Ercole, *In vivo* effects of insulin-like growth factor-I on the development of sensory pathways: Analysis of the primary somatic sensory cortex (S1) of transgenic mice, *Endocrinology* **137**, 5484-5492 (1996).
- 113. J.R. O'Kusky, P. Ye and A.J. D'Ercole, Insulin-like growth factor-I promotes neurogenesis and synaptogenesis in the hippocampal dentate gyrus during postnatal development, *J Neurosci* 20, 8435-8442 (2000).
- 114. P. Ye, Y.Z. Xing, Z.H. Dai and A.J. D'Ercole, In vivo actions of insulin-like growth factor-I (IGF-I) on cerebellum development in transgenic mice: Evidence that IGF-I increases proliferation of granule cell progenitors, *Dev Brain Res* 95, 44-54 (1996).
- 115. D. Chrysis, A.S. Calikoglu, P. Ye and A.J. D'Ercole, Insulin-like growth factor-I overexpression attenuates cerebellar apoptosis by altering the expression of Bcl family proteins in a developmentally specific manner, *J Neurosci* **21**, 1481-1489 (2001).
- 116. Y. Fukudome, T. Tabata, T. Miyoshi, S. Haruki, K. Araishi, S. Sawada and M. Kano, Insulin-like growth factor-I as a promoting factor for cerebellar Purkinje cell development, *Eur J Neurosci* 17, 2006-2016 (2003).

- 117. G. Camarero, C. Avendaño, C. Fernández-Moreno, A. Villar, J. Contreras, F. De Pablo, J.G. Pichel and I. Varela-Nieto, Delayed inner ear maturation and neuronal loss in postnatal *Igf*-1-deficient mice, *J Neurosci* **21**, 7630-7641 (2001).
- 118. M.A.I. Aberg, N.D. Aberg, H. Hedbacker, J. Oscarsson and P.S. Eriksson, Peripheral infusion of IGF-I selectively induces neurogenesis in the adult rat hippocampus, *J Neurosci* **20**, 2896-2903 (2000).
- 119. J.W. Jiang, J. McMurtry, D. Niedzwiecki and S.A. Goldman, Insulin-like growth factor-1 is a radial cell-associated neurotrophin that promotes neuronal recruitment from the adult songbird ependyma/subependyma, *J Neurobiol* **36**, 1-15 (1998).
- L.C. Ang, B. Bhaumick and B.H.J. Juurlink, Neurite promoting activity of insulin, insulin-like growth factor I and nerve growth factor on spinal motoneurons is astrocyte dependent, *Dev Brain Res* 74, 83-88 (1993).
- P. Fernyhough, G.B. Willars, R.M. Lindsay and D.R. Tomlinson, Insulin and insulin-like growth factor I enhance regeneration in cultured adult rat sensory neurones, *Brain Res* 607, 117-124 (1993).
- 122. D.M. Jones, B.A. Tucker, M. Rahimtula and K.M. Mearow, The synergistic effects of NGF and IGF-1 on neurite growth in adult sensory neurons: convergence on the PI 3-kinase signaling pathway, *J Neurochem* 86, 1116-1128 (2003).
- 123. K. Kimpinski and K. Mearow, Neurite growth promotion by nerve growth factor and insulin-like growth factor-1 in cultured adult sensory neurons: Role of phosphoinositide 3-kinase and mitogen activated protein kinase, *J Neurosci Res* 63, 486-499 (2001).
- M.M. Niblock, J.K. Brunso-Bechtold and D.R. Riddle, Insulin-like growth factor I stimulates dendritic growth in primary somatosensory cortex, *J Neurosci* 20, 4165-4176 (2000).
- H.B. Rind and C.S. Von Bartheld, Target-derived cardiotrophin-1 and insulin-like growth factor-I promote neurite growth and survival of developing oculomotor neurons, *Mol Cell Neurosci* 19, 58-71 (2002).
- 126. C.M. Cheng, R.F. Mervis, S.L. Niu, N. Salem, Jr., L.A. Witters, V. Tseng, R. Reinhardt and C.A. Bondy, Insulin-like growth factor 1 is essential for normal dendritic growth, *J Neurosci Res* 73, 1-9 (2003).
- 127. P. Fernyhough, J.F. Mill, J.L. Roberts and D.N. Ishii, Stabilization of tubulin mRNAs by insulin and insulin-like growth factor I during neurite formation, *Mol Brain Res* 6, 109-120 (1989).
- 128. B.H. Poe, C. Linville, D.R. Riddle, W.E. Sonntag and J.K. Brunso-Bechtold, Effects of age and insulin-like growth factor-1 on neuron and synapse numbers in area CA3 of hippocampus, *Neuroscience* **107**, 231-238 (2001).
- 129. J. Satoh and S.U. Kim, Proliferation and differentiation of fetal human oligodendrocytes in culture, *J Neurosci Res* **39**, 260-272 (1994).
- F.A. McMorris and M. Dubois-Dalcq, Insulin-like growth factor I promotes cell proliferation and oligodendroglial commitment in rat glial progenitor cells developing in vitro, *J Neurosci Res* 21, 199-209 (1988).
- P. Ye and A.J. D'Ercole, Insulin-like growth factor I protects oligodendrocytes from tumor necrosis factor-α-induced injury, *Endocrinology* 140, 3063-3072 (1999).
- 132. B.A. Barres, I.K. Hart, H.S.R. Coles, J.F. Burne, J.T. Voyyodic, W.D. Richardson and M.C. Raff, Cell death and control of cell survival in the oligodendrocyte lineage, *Cell* **70**, 31-46 (1992).
- 133. G.S. Vemuri and F.A. McMorris, Oligodendrocytes and their precursors require phosphatidylinositol 3-kinase signaling for survival, *Development* **122**, 2529-2537 (1996).
- 134. P. Ye, G. Kassiotis, G. Kollias and A.J. D'Ercole, Insulin-like grwoth factor (IGF)-I ameliorates demyelination induced by tumor necrosis factor (TNF)α in transgenic mice, *J Neurochem* **78** (suppl), 48 (2001).

- 135. R.L. Mozell and F.A. McMorris, Insulin-like growth factor I stimulates oligodendrocyte development and myelination in rat brain aggregate cultures, *J Neurosci Res* **30**, 382-390 (1991).
- 136. R.P. Saneto, K.G. Low, M.H. Melner and J. De Vellis, Insulin/insulin-like growth factor I and other epigenetic modulators of myelin basic protein expression in isolated oligodendrocyte progenitor cells, *J Neurosci Res* **21**, 210-219 (1988).
- 137. N.M. Kühl, J. De Keyser, H. De Vries and D. Hoekstra, Insulin-like growth factor binding proteins-1 and-2 differentially inhibit rat oligodendrocyte precursor cell survival and differentiation in vitro, *J Neurosci Res* **69**, 207-216 (2002).
- 138. J.L. Mason and J.E. Goldman, A2B5<sup>+</sup> and O4<sup>+</sup> cycling progenitors in the adult forebrain white matter respond differentially to PDGF-AA, FGF-2, and IGF-1, *Mol Cell Neurosci* **20**, 30-42 (2002).
- 139. P. Ye, J. Carson and A.J. D'Ercole, Insulin-like growth factor-I influences the initiation of myelination: studies of the anterior commissure of transgenic mice, *Neurosci Lett* **201**, 235-238 (1995).
- P. Ye, K. H. Lee and A.J. D'Ercole, Insulin-like growth factor-I (IGF-I) protects myelination from undernutritional insult: Studies of transgenic mice overexpressing IGF-I in brain, *J Neurosci Res* 62, 700-708 (2000).
- 141. J.L. Mason, P. Ye, K. Suzuki, A.J. D'Ercole and G.K. Matsushima, Insulin-like growth factor-1 inhibits mature oligodendrocyte apoptosis during primary demyelination, *J Neurosci* 20, 5703-5708 (2000).
- 142. B.A. Barres, M.D. Jacobson, R. Schmid, M. Sendnter and M.C. Raff, Does oligodendrocyte survival depend on axons, *Curr Biol* **3**, 489-497 (1993).
- M.J. Carson, R.R. Behringer, R.L. Brinster and F.A. McMorris, Insulin-like growth factor I increases brain growth and central nervous system myelination in transgenic mice, *Neuron* 10, 729-740 (1993).
- 144. P. Ye, L. Li, P.K. Lund and A.J. D'Ercole, Deficient expression of insulin receptor substrate-1 (IRS-1) fails to block insulin-like growth factor-I (IGF-I) stimulation of brain growth and myelination, *Brain Res Dev Brain Res* **136**, 111-121 (2002).
- 145. K.A. Woods, C. Camacho-Hubner, D. Barter, A.J.L. Clark and M.O. Savage, Insulin-like Growth Factor I gene deletion causing intrauterine growth retardation and severe short stature, *Acta Paediatr Scand Suppl* **423**, 39-45 (1997).
- 146. T. Matsuda, Y. Murata, K. Tabaks, R. Hosoi, M. Hayashi, K. Tamada, K. Takuma and A. Baba, Involvement of Na+, K(+) ATPase in the mitogenic effect of insulinlike growth factor-I on cultured rat astrocytes, *J Neurochem* 66, 511-516 (1996).
- 147. A.K. Snyder, S.P. Singh and S. Ehmann, Effects of ethanol on DNA, RNA, and protein synthesis in rat astrocyte cultures, *Alcoholism Clin Exp Res* **16**, 295-300 (1992).
- 148. S.L. O'Donnell, T.J. Frederick, J.K. Krady, S.J. Vannucci and T.L. Wood, IGF-I and microglia/macrophage proliferation in the ischemic mouse brain, *Glia* **39**, 85-97 (2002).
- 149. E.J. Beilharz, N.S. Bassett, E.S. Sirimanne, C.E. Williams and P.D. Gluckman, Insulin-like growth factor II is induced during wound repair following hypoxicischemic injury in the developing rat brain, *Mol Brain Res* 29, 81-91 (1995).
- 150. D. Gveric, M.L. Cuzner and J. Newcombe, Insulin-like growth factors and binding proteins in multiple sclerosis plaques, *Neuropathol Appl Neurobiol* **25**, 215-225 (1999).
- 151. S. Komoly, L.D. Hudson, H.D. Webster and C.A. Bondy, Insulin-like growth factor I gene expression is induced in astrocytes during experimental demyelination, *Proc Natl Acad Sci USA* **89**, 1894-1898 (1992).
- 152. W-H. Lee, J.A. Clemens and C.A. Bondy, Insulin-like growth factors in the response to cerebral ischemia, *Mol Cell Neurosci* (1992).

- 153. R.S. Nicholas, S. Stevens, M.G. Wing and D.A. Compston, Microglia-derived IGF-2 prevents TNFα induced death of mature oligodendrocytes in vitro, J *Neuroimmunol* **124**, 36-44 (2002).
- 154. W.P. Bartlett, X.-S. Li and M. Williams, Expression of IGF-I mRNA in the murine subventricular zone during postnatal development, *Mol Brain Res* **12**, 285-291 (1992).
- 155. A. Wilkins, S. Chandran and A. Compston, A role for oligodendrocyte-derived IGF-1 in trophic support of cortical neurons, *Glia* **36**, 48-57 (2001).
- 156. J.F. Keummerle and K.S. Murthy, Coupling of the insulin-like growth factor-I receptor tyrosine kinase to  $G_{12}$  in human intestinal smooth muscle  $G_{\beta gamma}$ -Dependent mitogen-activated protein kinase activation and growth, *J Biol Chem* **276**, 7187-7194 (2001).
- 157. H. Dudek, S.R. Datta, T.F. Franke, M.J. Birnbaum, R.J. Yao, G.M. Cooper, R.A. Segal, D.R. Kaplan and M.E. Greenberg, Regulation of neuronal survival by the serine-threonine protein kinase Akt, *Science* **275**, 661-665 (1997).
- 158. J.K. Ness and T.L. Wood, Insulin-like growth factor I, but not neurotrophin-3, sustains Akt activation and provides long-term protection of immature oligodendrocytes from glutamate-mediated apoptosis, *Mol Cell Neurosci* **20**, 476-488 (2002).
- 159. G.S. Vemuri, E. Boyle-Walsh, M. Holgado-Madruga, A. Wong and F.A. McMorris, Association of phosphatidylinositol 3-kinase with insulin-like growth factor receptor decking proteins in oligoendroglial cells, *J Neurochem* 69 (suppl), S9 (1997).
- A. Brunet, A. Bonni, M.J. Zigmond, M.Z. Lin, P. Juo, L.S. Hu, M.J. Anderson, K.C. Arden, J. Blenis and M.E. Greenberg, Akt promotes cell survival by phosphorylating and inhibiting a forkhead transcription factor, *Cell* 96, 857-868 (1999).
- 161. Y. Cao, A.J. Gunn, L. Bennet, D. Wu, S. George, P.D. Gluckman, X.M. Shao and J. Guan, Insulin-like growth factor (IGF)-1 suppresses oligodendrocyte caspase-3 activation and increases glial proliferation after ischemia in near-term fetal sheep, J Cereb Blood Flow Metab 23, 739-747 (2003).
- 162. D. Nijhawan, N. Honarpour and X.D. Wang, Apoptosis in neural development and disease, *Annu Rev Neurosci* 23, 73-87 (2000).
- 163. C.M. Cheng, R.R. Reinhardt, W.H. Lee, G. Joncas, S.C. Patel and C.A. Bondy, Insulin-like growth factor 1 regulates developing brain glucose metabolism, *Proc Natl Acad Sci USA* 97, 10236-10241 (2000).
- 164. G. Gutiérrez-Ospina, L. Saum, A.S. Calikoglu, S. Díaz-Cintra, F.A. Barrios and A.J. D'Ercole, Increased neural activity in transgenic mice with brain IGF-I overexpression: a [<sup>3</sup>H]2DG study, *Neuroreport* 8, 2907-2911 (1997).
- M. Schubert, D.P. Brazil, D.J. Burks, J.A. Kushner, J. Ye, C.L. Flint, J. Farhang-Fallah, P. Dikkes, X.M. Warot, C. Rio, G. Corfas and M.F. White, Insulin receptor substrate-2 deficiency impairs brain growth and promotes tau phosphorylation, J *Neurosci* 23, 7084-7092 (2003).
- M.J. Engleka, F. Folli, J. Winnay, C.R. Kahn and F.A. McMorris, Insulin-receptor substrate-1 is required for normal myelination, *J Neurochem* 66, S20 (1996).
- 167. P.A. Tranque, R. Calle, F. Naftolin and R. Robbins, Involvement of protein kinase-C in the mitogenic effect of insulin-like growth factor-I on rat astrocytes, *Endocrinology* 131, 1948-1954 (1992).
- K.A. Heidenreich, T. Zeppelin and L.J. Robinson, Insulin and insulin-like growth factor I induce c-fos expression in postmitotic neurons by a protein kinase Cdependent pathway, J Biol Chem 268, 14663-14670 (1993).

- 169. M.P. Mattson and B. Cheng, Growth factors protect neurons against excitotoxic/ischemic damage by stabilizing calcium homeostasis, *Stroke* 12, 136 (1993).
- 170. T.F. Clawson, S.J. Vannucci, G.M. Wang, L.B. Seaman, X.L. Yang and W.H. Lee, Hypoxia-ischemia-induced apoptotic cell death correlates with IGF-I mRNA decrease in neonatal rat brain, *Biol Signals Recept* 8, 281-293 (1999).
- 171. A.M. Fernandez, A.G. De la Vega, B. Planas and I. Torres-Aleman, Neuroprotective actions of peripherally administered insulin- like growth factor I in the injured olive-cerebellar pathway, *Eur J Neurosci* **11**, 2019-2030 (1999).
- 172. I. Kazanis, E. Bozas, H. Philippidis and F. Stylianopoulou, Neuroprotective effects of insulin-like growth factor-I (IGF-I) following a penetrating brain injury in rats, *Brain Res* **991**, 34-45 (2003).
- 173. J. Guan, L. Bennet, P.D. Gluckman and A.J. Gunn, Insulin-like growth factor-1 and post-ischemic brain injury, *Prog Neurobiol* **70**, 443-462 (2003).
- 174. R.J. Dempsey, K.A. Sailor, K.K. Bowen, K. Tureyen and R. Vemuganti, Strokeinduced progenitor cell proliferation in adult spontaneously hypertensive rat brain: effect of exogenous IGF-1 and GDNF, *J Neurochem* 87, 586-597 (2003).
- 175. C.R. Breese, A.D.Costa, Y.D. Rollins, C. Adams, R.M. Booze, W.E. Sonntag and S. Leonard, Expression of insulin-like growth factor-1 (IGF-1) and IGF-binding protein 2 (IGF-BP2) in the hippocampus following cytotoxic lesion of the dentate gyrus, *J Comp Neurol* **369**, 388-404 (1996).
- J. Garcia-Estrada, L.M. Garcia-Segura, and I. Torres-Aleman, Expression of insulin-like growth factor I by astrocytes in response to injury, *Brain Res* 592, 343-347 (1992).
- 177. F. Yamaguchi, T. Itano, O. Miyamoto, N.A. Janjua, T. Ohmoto, K. Hosokawa and O. Hatase, Increase of extracellular insulin-like growth factor I (IGF- I) concentration following electrolytical lesion in rat hippocampus, *Neurosci Lett* 128, 273-276 (1991).
- 178. X. Liu, D.L. Yao, C. Bondy, M. Brenner, L. Hudson, J. Zhou and H.D.F. Webster, Astrocytes express insulin-like growth factor-I (IGF-I) and its binding protein, IGFBP-2, during demyelination induced by experimental autoimmune encephalomyelitis, *Mol. Cell. Neurosci.* 5, 418-430 (1994).
- E. Carro, J.L. Trejo, T. Gomez-Isla, D. LeRoith and I. Torres-Aleman, Serum insulin-like growth factor I regulates brain amyloid-beta levels, *Neurochem Res* 8, 1390-1397 (2002).
- 180. X.F. Liu, J.R. Fawcett, R.G. Thorne, T.A. DeFor and W.H. Frey, II, Intranasal administration of insulin-like growth factor-I bypasses the blood-brain barrier and protects against focal cerebral ischemic damage, *J Neurol Sci* 187, 91-97 (2001).
- 181. W.R. Schäbitz, T.T. Hoffmann, S. Heiland, R. Kollmar, J. Bardutzky, C. Sommer and S. Schwab, Delayed neuroprotective effect of insulin-like growth factor-I after experimental transient focal cerebral ischemia monitored with MRI, *Stroke* 32, 1226-1233 (2001).
- 182. A.C.S. Nordqvist, H. vonHolst, S. Holmin, V.R. Sara, B.M. Bellander and M. Schalling, Increase of insulin-like growth factor (IGF)-1, IGF binding protein-2 and -4 mRNAs following cerebral contusion, *Mol Brain Res* 38, 285-293 (1996).
- J.L. Cook, V. Marcheselli, J. Alam, P.L. Deininger and N.G. Bazan, Temporal changes in gene expression following cryogenic rat brain injury, *Mol Brain Res* 55, 9-19 (1998).
- 184. X. Liu, D.L. Yao and H. d. F. Webster, Insulin-like growth factor I treatment reduces clinic deficits and lesion severity in acute demyelinating experimental autoimmune encephalomyelitis, *Mult Scler* **1**, 2-9 (1995).
- 185. D.L. Yao, X. Liu, L.D. Hudson and H.D. Webster, Insulin-like growth factor I treatment reduces demyelination and up-regulates gene expression of myelin-related

proteins in experimental autoimmune encephalomyelitis, *Proc Natl Acad Sci USA* **92**, 6190-6194 (1995).

- W.H. Lee, G.M. Wang, L.B. Seaman and S.J. Vannucci, Coordinate IGF-I and IGFBP5 gene expression in perinatal rat brain after hypoxia-ischemia, *J Cereb Blood Flow &. Metab* 16, 227-236 (1996).
- 187. K. Gustafson, H. Hagberg, B.A. Bengtsson, C. Brantsing and J. Isgaard, Possible protective role of growth hormone in hypoxia-ischemia in neonatal rats, *Ped Res* 45, 318-323 (1999).
- 188. T. Alexi, P.E. Hughes, W.M.C. Van Roon-Mom, R.L.M. Faull, C.E. Williams, R.G. Clark and P.D. Gluckman, The IGF-I amino-terminal tripeptide glycine-proline-glutamate (GPE) is neuroprotective to striatum in the quinolinic acid lesion animal model of Huntington's disease, *Exper Neurol* **159**, 84-97 (1999).
- J. Guan, T.L. Bennet, S. George, H.J. Waldvogel, R.L. Faull, P.D. Gluckman, H. Keunen and A.J. Gunn, Selective neuroprotective effects with insulin-like growth factor-1 in phenotypic striatal neurons following ischemic brain injury in fetal sheep, *Neuroscience* 95, 831-839 (2000).
- 190. A.M. Fernandez, A.G. De la Vega and I. Torres-Aleman, Insulin-like growth factor I restores motor coordination in a rat model of cerebellar ataxia, *Proc Natl Acad Sci* USA 95, 1253-1258 (1998).
- 191. X.F. Liu, J.R. Fawcett, R.G. Thorne, T.A. DeFor and W.H. Frey, Intranasal administration of insulin-like growth factor-I bypasses the blood-brain barrier and protects against focal cerebral ischemic damage, *J Neurol Sci* **187**, 91-97 (2001).
- 192. W.L. Wei, X.T. Wang and J.W. Kusiak, Signaling events in amyloid  $\beta$ -peptideinduced neuronal death and insulin-like growth factor I protection, *J Biol Chem* **277**, 17649-17656 (2002).
- 193. D. Offen, B. Shtaif, D. Hadad, A. Weizman, E. Melamed and I. Gil-Ad, Protective effect of insulin-like-growth-factor-1 against dopamine-induced neurotoxicity in human and rodent neuronal cultures: possible implications for Parkinson's disease, *Neurosci Lett* **316**, 129-132 (2001).
- 194. K.B. Bjugstad, W.M. Zawada, S.I. Goodman and C.R. Freed, IGF-1 and bFGF reduce glutaric acid and 3-hydroxyglutaric acid toxicity in striatal cultures, *J Inherit Metab Dis* 24, 631-647 (2001).
- 195. S. Humbert, E.A. Bryson, F.P. Cordelieres, N.C. Connors, S.R. Datta, S. Finkbeiner, M.E. Greenberg, and F. Saudou, The IGF-I/Akt pathway is neuroprotective in Huntington's disease and involves Huntingtin phosphorylation by Akt, *Dev Cell* 2, 831-837 (2002).
- 196. D.L. Tolbert and B.R. Clark, GDNF and IGF-I trophic factors delay hereditary Purkinje cell degeneration and the progression of gait ataxia, *Exper Neurol* 183, 205-219 (2003).
- 197. G. Bonapace, D. Concolino, S. Formicola and P. Strisciuglio, A novel mutation in a patient with insulin-like growth factor 1 (IGF1) deficiency, *J Med Genet* 40, 913-917 (2003).
- 198. M.J. Abuzzahab, A. Schneider, A. Goddard, F. Grigorescu, C. Lautier, E. Keller, W. Kiess, J. Klammt, J. Kratzsch, D. Osgood, R. Pfaffle, K. Raile, B. Seidel, R.J. Smith and S.D. Chernausek, IGF-I receptor mutations resulting in intrauterine and postnatal growth retardation, *N Engl J Med* **349**, 2211-2222 (2003).
- 199. Y. Okubo, K. Siddle, H. Firth, S. O'Rahilly, L.C. Wilson, L. Willatt, T. Fukushima, S. Takahashi, C.J. Petry, T. Saukkonen, R. Stanhope and D.B. Dunger, Cell proliferation activities on skin fibroblasts from a short child with absence of one copy of the type 1 insulin-like growth factor receptor (IGF1R) gene and a tall child with three copies of the IGF1R gene, J Clin Endocrinol Metab **88**, 5981-5988 (2003).

# Chapter 9

# **REGULATION OF VERTEBRATE SENSORY ORGAN DEVELOPMENT: A SCENARIO FOR GROWTH HORMONE AND INSULIN-LIKE GROWTH FACTORS ACTION**

Amelia Diaz-Casares<sup>1</sup>, Yolanda Leon<sup>1,2</sup>, Enrique J. de la Rosa<sup>3</sup> and Isabel Varela-Nieto<sup>1</sup>

<sup>1</sup>Instituto de Investigaciones Biomedicas "Alberto Sols," Consejo Superior de Investigaciones Cientificas (IIB-CSIC) – Universidad Autonoma de Madrid (UAM). Arturo Duperier 4. 28029 Madrid, Spain; <sup>2</sup>Departamento de Biología, Unidad de Fisiología Animal, UAM. Carretera de Colmenar km 15, Cantoblanco, 28049, Madrid, Spain; <sup>3</sup>Centro de Investigaciones Biologicas, CSIC. Ramiro de Maeztu 9. 28040 Madrid, Spain

Key words: Inner ear; neurogenesis; neurotrophic factors; retina; sensorineural; sensory receptors.

### **1. INTRODUCTION**

Sensorineural organs act as an interface between the nervous system and the world. Diverse sensorineural organs detect different kinds of environmental events using specialized sensorial receptor cells classified as chemoreceptor, thermoreceptor, mechanoreceptor, and photoreceptor cells. Chemoreceptor cells are associated with the senses of taste and smell and are sensitive to chemicals in the environment. The taste buds of mammals detect dissolved food molecules and are located primarily on papillae of the tongue<sup>1</sup>. The olfactory epithelium is designed for recognition of volatile molecules and is located in the nose in terrestrial vertebrates. The olfactory cells present olfactory cilia that specifically respond to a large number of different chemicals in the air, including as many as 1000 types of different smells in humans<sup>1,2</sup>. Thermoreceptor cells are associated with the sense of touch. In mammals, changes in temperature are detected by cold, warmth and pain thermoreceptors<sup>1</sup>. Mechanoreceptor cells respond to many different stimuli, such as sound, touch, pressure, gravity, stretch, or movement. Touch receptors are located in the skin and perceive touch, pressure, and pain<sup>1</sup>. Other mechanoreceptors include those located at the base of hairs and bristles that respond to motion and inner ear receptor cells that sense sound and balance<sup>1</sup>. Finally, photoreceptor cells located in the retina use pigments to absorb and detect light. The sensitive rods are specialized in low-intensity light detection, while the cones work better with daylight and are able to analyze the different light wavelengths responsible for the colors<sup>1</sup>. Upon specific activation, all types of receptor cells generate changes in membrane potentials that initiate the transmission of information to the brain that then reads this information and generates the adequate body responses.

In this chapter we summarize the available studies concerning the development of sense organs under physiological and pathological conditions, with respect to GH/IGF action.

### 2. DEVELOPMENT OF SENSORINEURAL ORGANS

The vertebrate sense organs arise at early stages of embryonic development and have distinct developmental origins. A variety of extrinsic signals control sensorineural development, including members of different families of cytokines, and growth and neurotrophic factors. The insulin-like growth factor (IGF)/growth hormone (GH) axis plays a central role in the development of the sensory organs, where these proteins and their receptors are widely expressed (Table 1)<sup>3-26</sup>. A detailed description of the elements of the IGF system and signaling pathways is provided in other chapters of this book<sup>27, 28</sup>.

GH availability, the expression of its receptor (GHR) and some actions on the sensorineural organs have also been reported, however, there is less information available than for the IGF system.

	IGFI	IGFIR	IR	GHR	IGFII	<b>BP-1</b>	BP-2	BP-3	<b>BP-4</b>	<b>BP-5</b>	BP-6
Early	OV	OlfP	OB	OpV	Ret	n.d.	OtP	n.d.	n.d.	n.d.	n.d.
Embr.	CVG	Ret					NP				
Dev.	Ret										
	Lens										
Late	OB	OE	OB	Ret	Ret	n.d.	Ret	n.d.	n.d.	OB	n.d.
Embr.	Ret	Ret		Lens							
Dev.											
Early	OB	OE	n.d.	n.d.	n.d.	n.d.	Ret	Ср	OB	OB	Ret
Post.	Ret	Ret					OB	Chr	Ср	Ret	Ср
age	Ср	Lens					Corn	Corn	Chr	Ср	Chr
	Corn	Conj					Iris		Corn	Chr	Corn
		Corn					Conj			Corn	
							Scl				
Adult	OB	OE	n.d.	n.d.	n.d.	n.d.	Corn	Ср	BG	Ret	OB
	Ret	Ret					Iris	Chr	Ср	Ср	Ret
	Ср	Lens					Conj	Corn	Chr	Chr	Ср
	Corn	Conj					Scl		Corn	Corn	Chr
		Corn									Corn

*Table 1.* Expression of factors, their receptors and BPs during sensory organ development and postnatal growth. This table summarizes available data on mRNA expression of IGF-I, IGF-IR, IR, GHR, IGF-II and IGFBPs 1-6 (refs. 3-26). **Vision**: Chr, chromatophores; Conj, conjunctiva; Corn, cornea; Cp, choroid plexus; OpV, optic vesicle; Ret, retina; Scl, sclera; **Inner ear**: CVG, cochleovestibular ganglion; OtP, otic placode; OV, otic vesicle; **Olfaction**: NP, nasal placode; OB, olfactory bulb; OE, olfactory epithelium; OlfP, olfactory placode. Emb, embryonic; Dev, development; Post, postnatal; n.d., not determined.

Many sense organs have a placodial developmental origin<sup>29,30</sup>. The inner ear and the olfactory system arise from the otic and olfactory placodes, respectively. Placodal development starts as discrete ectodermal thickenings that, through complex morphogenetic and neurogenetic processes, generate the adult structures. Furthermore, placodes can modulate the development of related structures. The olfactory placode is essential for normal forebrain development<sup>30</sup> and the olfactory and otic epithelia induce chondrogenesis in the surrounding mesenchyme, providing a protective, rigid structure for these sensorineural organs, which also contributes to the structural scaffold of the head. The optic lens also has a placodial origin and it is essential for normal development of the retina and other adjacent structures, such as the iris, the ciliary body and the overlying cornea<sup>30-34</sup>.

Effect	Neural Target	Factor
Increase cell size	Neurons (CG, OB) <sup>36, 53</sup>	IGF-I
Enhance proliferation	Neuronal precursors (Ret) <sup>37-39</sup>	IGF-I, IGF-II, Insulin
	Stem cells (OB) <sup>64</sup>	IGF-I, Insulin
	Neuronal precursors (CVG, ONE) <sup>19, 40-43</sup>	IGF-I, IGF-II, Insulin
	Schwann cells (OEG) <sup>44,45</sup>	IGF-I, IGF-II, Insulin
Decrease	Neurons (OMN, Ret) <sup>46-52</sup>	IGF-I, IGF-II, Insulin
apoptosis/enhance	Neurons (CVG, ONE, CG) <sup>42,54,63,68,69</sup>	IGF-I
survival	Schwann cells (CG) <sup>36,55-59</sup>	IGF-I
Differentiation	Neurons (Ret) <sup>38,60-62</sup>	IGF-I, IGF-II, Insulin
	Stem cells (OB) <sup>64</sup>	IGF-I, Insulin
	Neurons (CG, ONE, OB) <sup>36,42,64</sup>	IGF-I
	Schwann cells (CG) <sup>45,65,66</sup>	IGF-I, IGF-II, Insulin
Neuritogenesis/	Neurons (ION, OMN, Ret) <sup>15,24,48,67</sup>	IGF-I, Insulin
Axogenesis	Neurons (OB) <sup>19,53,70,75</sup>	IGF-I, IGF-II
	Olygodendrocytes (OB) <sup>53,70,75</sup>	IGF-I
Myelination	Neurons (CG) <sup>36,71</sup>	IGF-I, IGF-II
	Schwann cells <sup>45</sup>	IGF-I, IGF-II, Insulin
	Olygodendrocytes (OB) <sup>53</sup>	IGF-I
Synaptogenesis	Neurons (OC) <sup>36</sup>	IGF-I
Neuromodulation	Taurine (Ret) <sup>73</sup>	IGF-I, Insulin
	Calcium channels (Ret) <sup>72,74</sup>	IGF-I, Insulin

*Table 2.* IGFs actions on the sense organs. **Vision:** ION, Isthmo-optic nucleus; OMN, ocularmotoneurons; Ret, retina; **Inner ear:** CG, cochlear ganglion; CVG, cochleovestibular ganglion; OC, organ of Corti; **Olfaction:** OB, olfactory bulb; OEG, olfactory ensheating glia; ONE, olfactory neuroepithelium.

During the last decade, progress has been made in the understanding of the molecular basis of sensorineural organ development<sup>5,16,29,30,35</sup>, as well as in the role of IGFs in neural development. IGFs are fundamental for the development, maturation, and functionality of the nervous system, being involved in the regulation of processes such as cell proliferation, survival, and differentiation<sup>3</sup>. Studies on sensorineural development (Table 2)<sup>12,16,19,21,36-75</sup>, although less abundant, point in the same direction.

Diseases and	Sensory System	GH/IGF axis Dysfunction		
Syndromes	Dysfunction			
Idiopathic late onset	Reduction of olfactory	Reduced IGF-I levels		
cerebellar ataxia <sup>75</sup>	function			
Amyotrophic lateral	Reduction of olfactory	Reduced IGF-I levels		
sclerosis <sup>75</sup>	function			
Turner's syndrome <sup>78</sup>	Otitis media	Reduced IGF-I levels		
	Sensorineural hearing loss			
Laron's syndrome <sup>80,81</sup>	Deafness	GHR defects		
	Retinitis pigmentosa	Abnormalities of GH signal		
		transduction		
		Primary defects on IGF-I synthesis		
		or secretion		
Usher's syndrome <sup>83</sup>	Deafness	n.d.		
	Retinitis pigmentosa			
Igf-1 mutations <sup>76,77,82</sup>	Deafness	Elevated GH secretion		
		Undetectable serum IGF-I		
KOE1D		Normal serum IGFBP-3		
IGFIR mutations"	n.d.	Decreased IGFTR levels		
		Reduced affinity for IGF-I		
		Increased concentrations of IGF-I		
		and IGFBP-3		
		Decreased concentrations of		
		IGFBP-2		
Leber congenital	Abnormal eye movement	GH insufficiency		
amaurosis syndrome <sup>85</sup>	Vision defects			
Septo-optic dysplasia <sup>86</sup>	Optic nerve hypoplasia and	GH insufficiency		
	pituitary dysfunction			

*Table 3.* Clinical alterations associated with defects in the GH/IGF axis. This table summarizes human diseases and syndromes that present a dysfunction of the sense system associated with alterations in the GH/IGF axis<sup>75-86</sup>. n.d., not determined.

Alterations in the GH/IGF axis are causal to defects in the development of human sense organs (Table 3)<sup>75-86</sup>. The study of animal models that present altered levels of expression of one or more elements of the GH/IGF axis has, indeed, provided further insight into some of these human diseases and syndromes<sup>28,71</sup>. An understanding of the regulation of sensorineural development may contribute to the elucidation of the origin or progression of human diseases such us blindness or deafness, and to the development of novel therapeutic strategies.

#### **3.** THE VISUAL SYSTEM

Although the retina is a part of the central nervous system (CNS), because of its sensorial function and peripheral location it is considered as a sensorineural organ. The sensorial attributes reside in the innermost layer of the eye, the neuroretina. The neuroretina and the adjacent pigmented epithelium, together with part of the iris and ciliary body, are derived from the neuroepithelial wall via optic vesicle and optic cup. The lens, and to a lesser extent the cornea, is generated from the placoda formed in the overlying surface ectoderm (Fig. 1A). The vertebrate neuroretina is a complex neuronal network comprising six neuronal types: the rod and cone photoreceptors, responsible for the absorption of various wavelengths of light; the bipolar, horizontal, and amacrine interneurons, which integrate photoreceptor information; and the retinal ganglion cells, which transmit this information to the brain for cognitive processing. In addition to these neuronal types, Müller glial cells also originate within the retina. The outer most layer of the eve is the sclera, which at the front of the eve is transformed into the transparent cornea that permits light rays to enter the eye. The middle layer includes the iris, the ciliary body, and the choroid.

GH gene expression occurs in extrapituitary tissues prior to, and even after, the organogenesis of the pituitary gland<sup>87-90</sup>. GH immunoreactivity is detected early in development in, among other nervous system locations, the chick otic and optic vesicles. Later in development, but still prior to the differentiation of pituitary somatotrophs, the chick neuroretina, the pigmented epithelium, and the epithelial lens fiber cells show intense GH immunoreactivity<sup>88,91</sup>. The distribution of the GH receptor mirrors that of GH. Since GH is absent from the circulation of early chick embryos <sup>87,88</sup>, these observations suggest that extrapituitary GH expression has an autocrine/paracrine role during early embryogenesis, in particular in the development of the ear and the eye<sup>87-90,92</sup>. The presence of GH and its receptor correlates with a suggested involvement of GH in the regulation of ocular development by acting on the intraocular melanocortin system in the chick<sup>91</sup> or inducing retinal angiogenesis. Indeed, GH deficiency in humans is associated with reduced retinal vascularization<sup>93</sup>, whereas exogenous GH promotes retinal angiogenesis<sup>94</sup>.

The actions of GH during embryonic development could be mediated by other growth factors, particularly IGF-I<sup>95,96</sup>. The widespread expression of mRNAs for IGF-I, IGF-1R, and IGFBP-2 to IGFBP-6 in specific histological layers of the retina, choroids, ciliary body and cornea in the rat suggests specific roles of the IGF axis in the eye<sup>3,9,10</sup> (Table 1). The developmental expression of most IGF family members has been described in the eye of birds<sup>3,5</sup>, mammals<sup>3</sup> and fish<sup>6,7</sup> when proliferation and

differentiation of neuroretinal cells occur. IGF is also involved in lens differentiation. The presence of IGF-I mRNA in ocular embryonic tissues suggests an autocrine/paracrine function of the IGFs.

In addition to GH/IGF-I axis involvement in normal development, its deregulation has pronounced physiological effects such as dwarfism (associated with low levels of GH), gigantism, and acromegaly (associated with high levels of GH)<sup>96,97</sup>. More specifically in the visual system, Leber congenital amaurosis syndrome is associated with short stature, growth hormone insufficiency and vision defects<sup>85</sup>. Septo-optic dysplasia (SOD), a disorder of multifactorial etiology that includes gestational diabetes, is characterized by optic nerve hypoplasia with pituitary dysfunction<sup>86</sup>. Children with SOD may manifest a variety of visual and/or physical symptoms that range from mild to severe<sup>86, 98-100</sup>. A case of a 14-year-old boy with optic hypoplasia and pituitary dwarfism due to a complete deficiency in GH has also been reported<sup>101</sup>. A rare case of GH and gonadotropin deficiency associated with dysmorphic features has been reported in a 16year-old boy with clinical characteristics that included left anophthalmia (absence of the left eye), microphallus, bilateral cryptorchidism, and mental retardation<sup>102</sup>. All of these features are attributed to the hypothalamic dysfunction and are very similar to the features of septo-optic dysplasia, but mutation analyses revealed no mutations or polymorphisms in the SOD associated gene HESX1<sup>102</sup>.

Vascularization of the retina normally occurs during fetal development, with little or no vascularization after birth<sup>103</sup>. A role for GH in normal retinal vascular development has been suggested because children with congenital GH deficiency have reduced retinal vascularization<sup>93</sup>. GH may exert its effects through circulating or locally produced IGF-I since the lack of IGF-I during the early neonatal period is associated with lack of vascular growth and retinopathy of prematurity, a blinding disease initiated by lack of retinal vascular growth after premature birth<sup>104,105</sup>. IGF-I probably influences angiogenesis and the development of retinal neovascularization through interaction with locally produced factors such as vascular endothelial growth factor (VEGF), by acting as a permissive factor for maximum VEGF stimulation of angiogenesis<sup>104</sup>. IGF-I above a specific threshold level is necessary for maximum VEGF activation of the MAPK and Akt pathways, pathways important for endothelial cell proliferation and survival<sup>104</sup>. However, ischemia-induced retinal neovascularization is only partially suppressed in transgenic mice expressing a GH antagonist gene<sup>106</sup>.



Figure 1. (A). Neuroretina development. (B). Inner ear early development.

#### 4. THE OLFACTORY SYSTEM

Most elements of the olfactory system are generated from the most rostrally placed placodes, the olfactory placodes. In mammals, these placodes give rise to four types of neurons: olfactory, sense or vomeronasal, nervus terminalis, and migratory gonadotrophin-releasing hormone (GnRH) neurons<sup>107</sup>, plus other cell types such as receptor cells or glial and Schwann cells. The olfactory neurons follow distinct patterns of cell differentiation and migration and, *pari pasu* the nasal epithelium involutes and induces the chondrogenic capsule in the surrounding mesenchyme. A characteristic of GnRH neurons is that they are generated outside the CNS and migrate toward it<sup>30,108,109</sup>. Also remarkable is the fact that olfactory sense neurons are renewed from progenitor cells present in the olfactory neuroepithelium in adult mammals<sup>110,111</sup>. Therefore, the olfactory system is a source of adult stem cells<sup>112</sup>. The axons of the olfactory sensory neurons are surrounded by a special type of glial cell, called olfactory ensheathing glia, which are also derived from the olfactory placode<sup>113-116</sup>. The olfactory epithelium has intrinsic growth factors that regulate development and can support the genesis and survival of new neurons<sup>18</sup>. The members of the IGF system, in particular IGF-I, are expressed in the olfactory system cells and can influence neuronal generation, survival, and/or differentiation (Tables 1 and 2). Expression of both IGF-I and its high-affinity receptor IGF-1R decreases and/or becomes more restricted to specific cell types in the olfactory structures with age in different vertebrates<sup>16,17,24</sup>. However, it is important to note that the olfactory bulb is one of the areas in the CNS that retains relatively high expression of IGF-I and the IGF-1R, with this higher expression generally being associated with areas of adult neurogenesis<sup>3</sup>. Regarding the function of IGFs during olfactory system development, expression patterns in the olfactory bulb suggest a role for IGF-I from early neurogenesis onwards and, accordingly, *Igf-1* knockout mice present severe alterations in the formation of the mitral cell layer of the olfactory bulb, as well as altered morphology of radial glia<sup>64</sup>.

### 5. THE GUSTATORY SYSTEM

The development of the vertebrate taste system has been relatively understudied compared with other sensory systems. Tongue development starts when the first pharyngeal arch forms a swelling called the median tongue bud, shortly afterwards another pair of lateral swellings, the distal tongue buds or lateral lingual swellings, is formed, which rapidly overgrow the median tongue bud. These structures continue to grow throughout development and form the anterior two-thirds of the tongue. From the second pharyngeal arch develops a midline swelling called the copula that is later overgrown by the hypopharyngeal eminence, which gives rise to the posterior one-third of the tongue. The hypopharyngeal eminence expands mainly by the growth of the endoderm of the third pharyngeal arch, with a small contribution from the fourth pharyngeal arch to the most posterior aspect of the tongue<sup>117</sup>. The capacity of the pharyngeal endoderm to generate taste buds is probably determined by the end of gastrulation, but the molecular bases of this process are yet to be elucidated. The sensory afferent axons from the VIIth, IXth, and Xth cranial nerves invade the lingual epithelium before taste bud differentiation and it has been proposed that innervation may play a role in taste buds differentiation<sup>118</sup>. The neurotrophic factors brain-derived neurotrophic factor (BDNF) and neurotrophin-3 (NT-3) participate in the innervation of the developing tongue in mouse and humans. It has been reported that mice with deficits in BDNF or NT-3 present gustatory or somatosensory alterations, respectively<sup>119-122</sup>. In addition, alterations in size, number and morphology of gustatory papillae and taste buds present in BDNF null mutant mice demonstrate the neural dependence of developing taste organs<sup>123</sup>.

To our knowledge, there are no studies on the actions of the GH-IGF axis on the development of the gustatory system.

#### 6. THE TACTILE SYSTEM

The sense of touch allows the perception of stimuli such us contact, pressure, temperature or pain. Its sensory organ is the skin. The perception of stimuli is carried out by skin specific sensory mechanoreceptors, thermoreceptors and nociceptors located throughout the lavers of the skin (epidermis, dermis and hypodermis). For example, Meissner's corpuscles sense the onset and end of continuous light pressure, Pacinian's corpuscles sense firm pressure, Ruffini's corpuscles and Krause's corpuscles respond to changes in temperature and pressure on the skin. There are four types of mechanoreceptive afferent neurons that innervate the skin and respond to cutaneous motion and deformation in different ways<sup>124</sup>. The association of specialized receptor cells with peripheral terminals of sensory axons forms the sensory receptors in the skin. The specificity of the receptor is determined by the subtype of transducer that is stimulated and by the structure of the receptor that surrounds each of these nerve terminals<sup>124, 125</sup>. The development of the innervating sensory nerve cell occurs pari pasu with that of the receptor cell. The neurons promote receptor cell differentiation, which in turn provide neurotrophic support to the sensory neurons. The signals that mediate these actions are not known in detail, but the influence of neurotrophic factors on the development of the tactile system has been confirmed by the study of transgenic mice lacking neurotrophins or their receptors<sup>126-128</sup>.

Actions of the IGFs in the regulation of tactile corpuscles development or in the perception of the stimuli have not yet been reported.

### 7. THE AUDITORY AND VESTIBULAR SYSTEMS

The vertebrate inner ear is derived from a thickening of the head surface ectoderm, adjacent to rhombomeres 5 and 6, named the otic placode that invaginates and pinches off the ectoderm to form the otic vesicle or otocyst. Figure 1B shows the early stages of vertebrate inner ear development. The otic vesicle is a transient structure that undergoes multiple morphogenetic movements and developmental changes associated with cell proliferation, differentiation and cell-death. This results in the ear labyrinth: the cochlea, the utricle and saccule, and the semicircular canals, each containing their corresponding sensory organs<sup>35,129,130</sup> with the mechano-transducing hair cells and the neurons that connect them with the central nervous system. Local mechanical perturbations are transduced by hair cells into synaptic potentials, which elicit the activation of auditory (cochlear) and vestibular neurons, that project towards the homonymous central nuclei, the first input station of the auditory pathway. Auditory neurons inform the brain of the intensity and spectral properties of sound and vestibular neurons carry information about position, velocity and acceleration. Generation of otic neurons is a sequential process. First, otic neurons are specified in the otic epithelium, neuronal precursors then delaminate to form the cochleovestibular ganglion (CVG), where they proliferate and differentiate. The CVG neurons project extensions back to innervate the vestibular and cochlear (auditory) sensory epithelium.

Diffusible factors like fibroblast growth factors, IGF-I and the nerve growth factor family of neurotrophins are locally synthesized during development and elicit a network of interconnected signaling pathways that finally instructs the cell to proliferate, die or differentiate. The elements of the IGF system are expressed in the early developing chicken inner ear<sup>3,14</sup> (Table 1). GH immunoreactivity has also been detected during early development of the chicken otic vesicle<sup>88,90-92</sup>. In the developing vertebrate inner ear, IGF-I acts as a survival and growth factor <sup>135</sup>. IGF-I is also expressed during maturation of the rodent auditory system and in adult hair cells<sup>131,132</sup>. During the early postnatal period of mouse inner ear development, from postnatal day (P)5 to P20, IGF-I is expressed in the Organ of Corti and cochlear ganglia<sup>36</sup>. The cochlear and vestibular ganglia also express insulin, IGF-II and their receptors (Fig. 2A). Recently, the analysis of a human fetal cochlear cDNA library indicated the presence of IGF-I and IGFBP-1, -3 and -5. Analysis of gene expression profiles of the rat cochlea demonstrated the presence of IGF-II and IGFBP-2 and  $-6^{133,13}$ . Studies on primary cell cultures and genetically modified animal models have shown that IGF-I is essential for the normal development and function of the vertebrate inner ear<sup>135</sup> (Table 2). Endogenous IGF-I is essential for generation of the CVG in chicken embryos, with the blockade of IGF-I actions being associated with an increase in cell death, a reduction in cell proliferation and a reduction in the levels of expression of neuroblasts and neuronal markers<sup>140</sup>.



*Figure 2.* A. Expression of insulin-related factors and receptors in the cochlear and vestibular ganglia in P5 mice. Immunolocalization of IGF-I, IGF-II, and IGF1R was positive in the cochlear (CG) and vestibular ganglia (VG) of P5 mouse. Microphotographs show representative 25  $\mu$ m paraffin sections from at least four mice studied in three different assays. Scale bars, 50  $\mu$ m. **B.** Auditory thresholds of wild type (n=25), heterozygous (n=30) and knockout *Igf-1* mice (n=7) at P30. Statistical analysis reveal that the *Igf-1* deficient (-/-) mouse has a two-fold increase in auditory thresholds (\*\*\* P<0,001) compared to control (+/+) and heterozygous (+/-) mice.

We have explored the signaling mechanisms that mediate IGF-I proliferative responses in otic cells. IGF-I binding to IGF1R increases the levels of inositol lipid mediators, activates the Raf/MAPK cascade, and induces the expression of the transcription factor AP-1 and proliferating cell nuclear antigen (PCNA)<sup>40,43,136-138</sup>. Anti-apoptotic effects of IGF-I are mediated by the activation of the Akt/protein kinase B pathway and modulation of the levels of the pro-apoptotic lipid mediator ceramide<sup>139</sup>.

Interestingly, the early actions of IGF-I on cell proliferation are mimicked by other factors of the family, insulin and IGF-II, but later actions on cell differentiation are specific to IGF-I<sup>43,140</sup>. Accordingly, during the mouse inner ear postnatal period of maturation, from birth to P20, the neurons of the cochlear ganglia become strictly IGF-I dependent and its deficit causes retarded maturation and decreased cell survival<sup>36,71</sup>. Moreover, IGF-I deficiency causes decreased neuronal differentiation in the mouse cochlear ganglion, a sustained deficit in the cochlear nerve and ganglia myelination, and aberrant synaptogenesis at the Organ of Corti, suggesting that IGF-I is required to reach full auditory function<sup>36,71</sup>. We do not yet know whether the alterations observed in cochlear neurons of Igf-1 null mice could compromise mid-term survival of hair cells of the organ of Corti. We have confirmed by using auditory brainstem response tests that Igf-1 knockout mice at day P30 have an increase in auditory thresholds compared to control  $(Igf-1^{+/+})$  and heterozygous  $(Igf-1^{+/-})$  mice<sup>135</sup> (Fig. 2B). These results confirm the key role of IGF-I in the development and maintenance of auditory function. Indeed, a deficiency in IGF-I results in sensory-neural deafness in humans<sup>76,77,82</sup> (Table 3), but a partial deficiency in IGFR1 is not associated with deafness in humans, although no detailed study of the auditory function of these patients has been reported<sup>84</sup>.

Hearing and balance impairment caused by hair cell loss or dysfunction is a high-prevalence multifactorial disease that currently has no restorative treatment available. IGF-I and insulin, alone or in combination with other growth and neurotrophic factors, protect otic cells from ototoxic damage and promote *in vitro* hair cell regeneration<sup>141-149</sup>. The potential of IGF-I in the treatment of neurodegenerative diseases, together with the reported actions of IGF-I in inner ear development and function, support the hypothesis that this factor is a good candidate for inner ear regeneration therapy.

#### ACKNOWLEDGEMENTS

This work was supported in part by grants BMC2003-07751, SAF2001-1038, and G03/203 from the Dirección General de Investigación, Ciencia y Tecnología and Instituto de Salud Carlos III to IVN and EJdlR. A. Diaz-Casares holds a postdoctoral CSIC Research I3P contract supported by European Social Fund. We thank Dr. Julio Contreras for useful comments on the manuscript, José Ángel Morales-García for sharing unpublished information, and Yolanda Rico and the IIB-CSIC Scientific Image Service for technical help.

#### REFERENCES

- 1. S.H. Hendry, S.S. Hsiao and M.C. Brown, in: *Fundamental Neuroscience*, edited by L.R. Squire, F.E. Bloom, S.K. McConnell, J.L. Roberts, N.C. Spitzer, M.J. Zigmond (Academic Press, San Diego, California, USA, 2003), pp. 577-589.
- 2. P. Mombaerts, Odorant receptor gene choice in olfactory sensory neurons: the one receptor-one neuron hypothesis revisited. *Curr Opin Neurobiol*, **14**, 31-36 (2004).
- 3. I. Varela-Nieto, E.J. de la Rosa, A.I. Valenciano, and Y. Leon, Cell death in the nervous system: lessons from insulin and insulin-like growth factors. *Mol Neurobiol*, **28**, 23-50 (2003).
- 4. P. Moriarty, M. Boulton, A. Dickson and D. McLeod, Production of IGF-I and IGF binding proteins by retinal cells in vitro. *Br J Ophthalmol*, **78**, 638-642 (1994).
- G. Calvaruso, R. Vento, M. Giuliano, M. Lauricella, E. Gerbino and G. Tesoriere, Insulin-like growth factors in chick embryo retina during development. *Regul Pept*, 61, 19-25 (1996).
- 6. E. Ayaso, C.M. Nolan and L. Byrnes, Zebrafish insulin-like growth factor-I receptor: molecular cloning and developmental expression. *Mol Cell Endocrinol*, **191**, 137-148 (2002).
- 7. B. Funkenstein, R. Almuly and S.J. Chan, Localization of IGF-I and IGF-I receptor mRNA in Sparus aurata larvae. *Gen Comp Endocrinol*, **107**, 291-303 (1997).
- 8. D.C. Otteson, P.F. Cirenza and P.F. Hitchcock, Persistent neurogenesis in the teleost retina: evidence for regulation by the growth-hormone/insulin-like growth factor-I axis. *Mech Dev*, **117**, 137-149 (2002).
- 9. C.P. Burren, J.L. Berka, S.R. Edmondson, G.A. Werther and J.A. Batch, Localization of mRNAs for insulin-like growth factor-I (IGF-I), IGF-I receptor, and IGF binding proteins in rat eye. *Invest Ophthalmol Vis Sci*, **37**, 1459-1468 (1996).
- 10. C.P. Burren, J.L. Berka and J.A. Batch, Localization studies of IGFBP-2 and IGFBP-5 in the anterior compartment of the eye. *Curr Eye Res*, **16**, 256-262 (1997).
- J. Serna, P.R. Gonzalez-Guerrero, C.G. Scanes, M. Prati, G. Morreale and F. de Pablo, Differential and tissue-specific regulation of (pro)insulin and insulin-like growth factor-I mRNAs and levels of thyroid hormones in growth-retarded embryos. *Growth Regul*, 6, 73-82 (1996).
- 12. L.E. Politi, N.P. Rotstein, G. Salvador, N.M. Giusto and M.F. Insua, Insulin-like growth factor-I is a potential trophic factor for amacrine cells. *J Neurochem*, **76**, 1199-1211 (2001).
- 13. W.H. Lee, S. Javedan and C.A. Bondy, Coordinate expression of insulin-like growth factor system components by neurons and neuroglia during retinal and cerebellar development. *J Neurosci*, **12**, 4737-4744 (1992).
- 14. Y. Leon, C. Sanz, L.M. Frago, G. Camarero, S. Canon, I. Varela-Nieto and F. Giraldez, Involvement of insulin-like growth factor-I in inner ear organogenesis and regeneration. *Horm Metab Res*, **31**, 126-132 (1999).
- T.J. Schoen, C.A. Bondy, J. Zhou, R. Dhawan, K. Mazuruk, D.R. Arnold, I.R. Rodriguez, R.J. Waldbillig, D.C. Beebe and G.J. Chader, Differential temporal and spatial expression of insulin-like growth factor binding protein-2 in developing chick ocular tissues. *Invest Ophthalmol Vis Sci*, 36, 2652-2662 (1995).

- 16. S.K. Pixley, N.S. Dangoria, K.K. Odoms and L. Hastings, Effects of insulin-like growth factor 1 on olfactory neurogenesis in vivo and in vitro. *Ann N Y Acad Sci*, **855**, 244-247 (1998).
- 17. M. Holzenberger, F. Lapointe and C. Ayer-LeLievre, Expression of insulin-like growth factor-I (IGF-I) and IGF-II in the avian brain: relationship of in situ hybridization patterns with IGF type 1 receptor expression. *Int J Dev Neurosci*, **18**, 69-82 (2000).
- 18. J. Plendl, B. Stierstorfer and F. Sinowatz, Growth factors and their receptors in the olfactory system. *Anat Histol Embryol*, **28**, 73-79 (1999).
- 19. F. de Pablo and E.J. de la Rosa, The developing CNS: a scenario for the action of proinsulin, insulin and insulin-like growth factors. *Trends Neurosci*, **18**, 143-150 (1995).
- K.A. Sullivan and E.L. Feldman, Immunohistochemical localization of insulin-like growth factor-II (IGF-II) and IGF-binding protein-2 during development in the rat brain. *Endocrinology*, **135**, 540-547 (1994).
- V.C. Russo, S.R. Edmondson, F.A. Mercuri, C.R. Buchanan and G.A. Werther, Identification, localization, and regulation of insulin-like growth factor binding proteins and their messenger ribonucleic acids in the newborn rat olfactory bulb. *Endocrinology*, 135, 1437-1446 (1994).
- 22. V.C. Russo, L.A. Bach, A.J. Fosang, N.L. Baker and G.A. Werther, Insulin-like growth factor binding protein-2 binds to cell surface proteoglycans in the rat brain olfactory bulb. *Endocrinology*, **138**, 4858-4867 (1997).
- M.M. Giacobini, R.H. Zetterstrom, D. Young, B. Hoffer, V. Sara and L. Olson, IGF-1 influences olfactory bulb maturation. Evidence from anti-IGF-1 antibody treatment of developing grafts in oculo. *Brain Res Dev Brain Res*, 84, 67-76 (1995).
- C.A. Bondy and W.H. Lee, Patterns of insulin-like growth factor and IGF receptor gene expression in the brain. Functional implications. *Ann N Y Acad Sci*, 692, 33-43 (1993).
- E.J. de la Rosa, C.A. Bondy, C. Hernandez-Sanchez, X. Wu, J. Zhou, A. Lopez-Carranza, L.M. Scavo and F. de Pablo, Insulin and insulin-like growth factor system components gene expression in the chicken retina from early neurogenesis until late development and their effect on neuroepithelial cells. *Eur J Neurosci*, 6, 1801-1810 (1994).
- 26. J.L. Marks, D. Porter Jr. and D.G. Baskin, Localization of type I insulin-like growth factor receptor messenger RNA in the adult rat brain by in situ hybridization. *Mol Endocrinol*, **5**, 1158-1168 (1991).
- 27. L. Frago and J. Chowen, Physiology of GH-IGF axis. Chapter 1, this book.
- 28. J.L. Trejo, E. Carro and D.J. Burks, Title experimental models for understanding the role of Igf-I and its receptor during development (Chapter 3, this book).
- M. Torres and F. Giraldez, The development of the vertebrate inner ear. *Mech Dev*, 71, 5-21 (1998).
- C.V. Baker and M. Bronner-Fraser, Vertebrate cranial placodes I. Embryonic induction. Dev Biol, 232, 1-61 (2001).
- M.L. Breitman, D.M. Bryce, E. Giddens, S. Clapoff, D. Goring, L.C. Tsui, G.K. Klintworth and A. Bernstein, Analysis of lens cell fate and eye morphogenesis in transgenic mice ablated for cells of the lens lineage. *Development*, **106**, 457-463 (1989).
- 32. L. Harrington, G.K. Klintworth, T.E. Secor and M.L. Breitman, Developmental analysis of ocular morphogenesis in alpha A-crystallin/diphtheria toxin transgenic mice undergoing ablation of the lens. *Dev Biol*, **148**, 508-516 (1991).
- 33. D.C. Beebe and J.M. Coats, The lens organizes the anterior segment: specification of neural crest cell differentiation in the avian eye. *Dev Biol*, **220**, 424-431 (2000).
- 34. C.J. Thut, R.B. Rountree, M. Hwa and D.M. Kingsley, A large-scale in situ screen provides molecular evidence for the induction of eye anterior segment structures by the developing lens. *Dev Biol*, **231**, 63-76 (2001).

- B. Alsina, F. Giráldez and I. Varela-Nieto, in: Growth factors and early development of otic neurons: interactions between intrinsic and extrinsic signals. Edited by R Romand and I. Varela-Nieto (Elsevier Academic Press, San Diego, California, USA, 2003) pp. 177-206.
- G. Camarero, C. Avendano, C. Fernandez-Moreno, A. Villar, J. Contreras, F. de Pablo, J.G. Pichel and I. Varela-Nieto, Delayed inner ear maturation and neuronal loss in postnatal Igf-1-deficient mice. *J Neurosci*, **21**, 7630-7641 (2001).
- 37. S.E. Boucher and P.F. Hitchcock, Insulin-related growth factors stimulate proliferation of retinal progenitors in the goldfish. *J Comp Neurol*, **394**, 386-394 (1998).
- C. Hernandez-Sanchez, A. Lopez-Carranza, C. Alarcon, E.J. de La Rosa and F. de Pablo, Autocrine/paracrine role of insulin-related growth factors in neurogenesis: local expression and effects on cell proliferation and differentiation in retina. *Proc Natl Acad Sci U S A*, 92, 9834-9838 (1995).
- 39. A.F. Mack and R.D. Fernald, Regulation of cell division and rod differentiation in the teleost retina. *Brain Res Dev Brain Res*, **76**, 183-187 (1993).
- Y. Leon, E. Vazquez, C. Sanz, J.A. Vega, J.M. Mato, F. Giraldez, J. Represa and I. Varela-Nieto, Insulin-like growth factor-I regulates cell proliferation in the developing inner ear, activating glycosyl-phosphatidylinositol hydrolysis and Fos expression. *Endocrinology*, **136**, 3494-3503 (1995).
- 41. N. Lu, I.B. Black and E. DiCicco-Bloom, A paradigm for distinguishing the roles of mitogenesis and trophism in neuronal precursor proliferation. *Brain Res Dev Brain Res*, **94**, 31-36 (1996).
- 42. M. Mathonnet, P. Cubertafond, A. Gainant and C. Ayer-Le Lievre, The avian peripheral olfactory system: model for study of apoptosis and cellular regeneration. *Ann Chir*, **126**, 888-895 (2001).
- 43. Y. Leon, C. Sanz, F. Giraldez and I. Varela-Nieto, Induction of cell growth by insulin and insulin-like growth factor-I is associated with Jun expression in the otic vesicle. *J Comp Neurol*, **398**, 323-332 (1998).
- 44. M. Sondell, A. Fex-Svenningsen and M. Kanje, The insulin-like growth factors I and II stimulate proliferation of different types of Schwann cells. *Neuroreport*, **8**, 2871-2876 (1997).
- 45. H.J. Stewart, F. Bradke, A. Tabernero, D. Morrell, K.R. Jessen and R. Mirsky, Regulation of rat Schwann cell Po expression and DNA synthesis by insulin-like growth factors in vitro. *Eur J Neurosci*, **8**, 553-564 (1996).
- C. Alarcon, J. Serna, B. Perez-Villamil and F. de Pablo, Synthesis and differentially regulated processing of proinsulin in developing chick pancreas, liver and neuroretina. *FEBS Lett*, **436**, 361-366 (1998).
- 47. B. Diaz, B. Pimentel, F. de Pablo and E.J. de La Rosa, Apoptotic cell death of proliferating neuroepithelial cells in the embryonic retina is prevented by insulin. *Eur J Neurosci*, **11**, 1624-1632 (1999).
- H.B. Rind and C.S. von Bartheld, Target-derived cardiotrophin-1 and insulin-like growth factor-I promote neurite growth and survival of developing oculomotor neurons. *Mol Cell Neurosci*, **19**, 58-71 (2002).
- 49. B.R. Ryu, H.W. Ko, I. Jou, J.S. Noh and B.J. Gwag, Phosphatidylinositol 3-kinasemediated regulation of neuronal apoptosis and necrosis by insulin and IGF-I. *J Neurobiol*, **39**, 536-546 (1999).
- 50. P. Kermer, R. Ankerhold, N. Klocker, S. Krajewski, J.C. Reed and M. Bahr, Caspase-9: involvement in secondary death of axotomized rat retinal ganglion cells in vivo. *Brain Res Mol Brain Res*, **85**, 144-150 (2000).
- 51. P. Kermer, N. Klocker, M. Labes and M. Bahr, Insulin-like growth factor-I protects axotomized rat retinal ganglion cells from secondary death via PI3-K-dependent Akt phosphorylation and inhibition of caspase-3 In vivo. *J Neurosci*, **20**, 2-8 (2000).

236

- 52. B. Diaz, J. Serna, F. De Pablo and E.J. de la Rosa, In vivo regulation of cell death by embryonic (pro)insulin and the insulin receptor during early retinal neurogenesis. *Development*, **127**, 1641-1649 (2000).
- 53. C.M. Cheng, G. Joncas, R.R. Reinhardt, R. Farrer, R. Quarles, J. Janssen, M.P. McDonald, J.N. Crawley, L. Powell-Braxton and C.A. Bondy, Biochemical and morphometric analyses show that myelination in the insulin-like growth factor 1 null brain is proportionate to its neuronal composition. *J Neurosci*, **18**, 5673-5681 (1998).
- 54. R.W. Oppenheim, Cell death during development of the nervous system. Annu Rev Neurosci, 14, 453-501 (1991).
- 55. W.M. Campana, S.J. Darin and J.S. O'Brien, Phosphatidylinositol 3-kinase and Akt protein kinase mediate IGF-I- and prosaptide-induced survival in Schwann cells. *J Neurosci Res*, **57**, 332-341 (1999).
- H.L. Cheng, M.L. Steinway, X. Xin and E.L. Feldman, Insulin-like growth factor-I and Bcl-X(L) inhibit c-jun N-terminal kinase activation and rescue Schwann cells from apoptosis. *J Neurochem*, 76, 935-943 (2001).
- 57. C.L. Delaney, J.W. Russell, H.L. Cheng and E.L. Feldman, Insulin-like growth factor-I and over-expression of Bcl-xL prevent glucose-mediated apoptosis in Schwann cells. *J Neuropathol Exp Neurol*, **60**, 147-160 (2001).
- 58. C.L. Delaney, H.L. Cheng and E.L. Feldman, Insulin-like growth factor-I prevents caspase-mediated apoptosis in Schwann cells. *J Neurobiol*, **41**, 540-548 (1999).
- 59. C. Meier, E. Parmantier, A. Brennan, R. Mirsky and K.R. Jessen, Developing Schwann cells acquire the ability to survive without axons by establishing an autocrine circuit involving insulin-like growth factor, neurotrophin-3, and platelet-derived growth factor-BB. *J Neurosci*, **19**, 3847-3859 (1999).
- J.M. Frade, E. Marti, P. Bovolenta, M.A. Rodriguez-Pena, D. Perez-Garcia, H. Rohrer, D. Edgar and A. Rodriguez-Tebar, Insulin-like growth factor-I stimulates neurogenesis in chick retina by regulating expression of the alpha 6 integrin subunit. *Development*, 122, 2497-2506 (1996).
- 61. R.E. Hausman, G.D. Sagar and B.H. Shah, Initial cholinergic differentiation in embryonic chick retina is responsive to insulin and cell-cell interactions. *Brain Res Dev Brain Res*, **59**, 31-37 (1991).
- 62. B.H. Shah and R.E. Hausman, Effects of cell signaling on the development of GABA receptors in chick retina neurons. *Neurochem Res*, **18**, 957-964 (1993).
- 63 B. Pettmann and C.E. Henderson, Neuronal cell death. *Neuron*, 20, 633-647 (1998).
- C. Vicario-Abejon, M.J. Yusta-Boyo, C. Fernandez-Moreno and F. de Pablo, Locally born olfactory bulb stem cells proliferate in response to insulin-related factors and require endogenous insulin-like growth factor-I for differentiation into neurons and glia. *J Neurosci*, 23, 895-906 (2003).
- 65. H.L. Cheng and E.L. Feldman, Insulin-like growth factor-I (IGF-I) and IGF binding protein-5 in Schwann cell differentiation. *J Cell Physiol*, **171**, 161-167 (1997).
- H.L. Cheng, M. Shy and E.L. Feldman, Regulation of insulin-like growth factor-binding protein-5 expression during Schwann cell differentiation. *Endocrinology*, 140, 4478-4485 (1999).
- 67. T.A. Janiga, H.B. Rind and C.S. von Bartheld, Differential effects of the trophic factors BDNF, NT-4, GDNF, and IGF-I on the isthmo-optic nucleus in chick embryos. *J Neurobiol*, **43**, 289-303 (2000).
- 68. K.A. Roth and C. D'Sa, Apoptosis and brain development. *Ment Retard Dev Disabil Res Rev*, **7**, 261-266 (2001).
- 69. A.L. Calof, N. Hagiwara, J.D. Holcomb, J.S. Mumm and J. Shou, Neurogenesis and cell death in olfactory epithelium. *J Neurobiol*, **30**, 67-81 (1996).
- 70. H.A. Cameron, T.G. Hazel and R.D. McKay, Regulation of neurogenesis by growth factors and neurotransmitters. *J Neurobiol*, **36**, 287-306 (1998).

- 71 G. Camarero, M.A. Villar, J. Contreras, C. Fernandez-Moreno, J.G. Pichel, C. Avendano and I. Varela-Nieto, Cochlear abnormalities in insulin-like growth factor-1 mouse mutants. *Hear Res*, **170**, 2-11 (2002).
- 72. L.A. Blair and J. Marshall, IGF-1 modulates N and L calcium channels in a PI 3-kinasedependent manner. *Neuron*, **19**, 421-429 (1997).
- 73. R. Salceda, Insulin-stimulated taurine uptake in rat retina and retinal pigment epithelium. *Neurochem Int*, **35**, 301-306 (1999).
- 74. S.L. Stella, E.J. Bryson Jr. and W.B. Thoreson, Insulin inhibits voltage-dependent calcium influx into rod photoreceptors. *Neuroreport*, **12**, 947-951(2001).
- 75 G. Federico, C. Maremmani, V. Cinquanta, G.I. Baroncelli, G. Fattori and V. Saggese, Mucus of the human olfactory epithelium contains the insulin-like growth factor-I system which is altered in some neurodegenerative diseases. *Brain Res*, 835, 306-314 (1999).
- K.A. Woods, C. Camacho-Hubner, M.O. Savage and A.J. Clark, Intrauterine growth retardation and postnatal growth failure associated with deletion of the insulin-like growth factor I gene. *N Engl J Med*, 335, 1363-1367 (1996).
- .77. K.A. Woods, C. Camacho-Hubner, D. Barter, A.J. Clark and M.O. Savage, Insulin-like growth factor I gene deletion causing intrauterine growth retardation and severe short stature. *Acta Paediatr Suppl*, **423**, 39-45 (1997).
- 78. M. Barrenasa, K. Landin-Wilhelmsenb and C. Hansonc, Ear and hearing in relation to genotype and growth in Turner syndrome. *Hear Res*, **144**, 21-28 (2000).
- M. Holzenberger, P. Leneuve, G. Hamard, B. Ducos, L. Perin, M. Binoux and Y. Le Bouc, A targeted partial invalidation of the insulin-like growth factor I receptor gene in mice causes a postnatal growth deficit. *Endocrinology*, 141, 2557-2566 (2000).
- Z. Laron, Growth hormone insensitivity (Laron syndrome). *Rev Endocr Metab Disord*, 3, 347-355 (2002).
- 81. Z. Laron, Laron syndrome (primary growth hormone resistance or insensitivity): the personal experience 1958-2003. *J Clin Endocrinol Metab*, **89**, 1031-1044 (2004).
- G. Bonapace, D. Concolino, S. Formicola and P. Strisciuglio, A novel mutation in a patient with insulin-like growth factor 1 (IGF1) deficiency. *J Med Genet*, 40, 913-917 (2003).
- X.M. Ouyang, D. Yam, J.F. Hejtmancik, S.G. Jacobson, A.R. Li, L.L. Du, S. Angeli, M. Kaiser, T. Balkany and X.Z. Liu, Mutational spectrum in Usher syndrome type II. *Clin Genet*, 65, 288-293 (2004).
- M..J. Abuzzahab, A. Schneider, A. Goddard, F. Grigorescu, C. Lautier, E. Keller, W. Kiess, J. Klammt, J.Kratzsch, D. Osgood, R. Pfaffle, K. Raile, B. Seidel, R.J. Smith, and S.D. Chernausek, IGF-I receptor mutations resulting in intrauterine and postnatal growth retardation. *N Engl J Med*, 349, 2211-2222 (2003).
- 85. H. Ehara, C. Nakano, K. Ohno, Y.I. Goto and K. Takeshita, New autosomal-recessive syndrome of Leber congenital amaurosis, short stature, growth hormone insufficiency, mental retardation, hepatic dysfunction, and metabolic acidosis. *Am J Med Genet*, **71**, 258-266 (1997).
- 86. C.L. Campbell, Septo-optic dysplasia: a literature review. *Optometry*, **74**, 417-426 (2003).
- 87. S. Harvey, C.D. Johnson and E.J. Sanders, Extra-pituitary growth hormone in peripheral tissues of early chick embryos. *J Endocrinol*, **166**, 489-502 (2000).
- S. Harvey, C.D. Johnson and E.J. Sanders, Growth hormone in neural tissues of the chick embryo. J Endocrinol, 169, 487-498 (2001).
- 89. S. Harvey and K. Hull, Neural growth hormone: an update. J Mol Neurosci, 20, 1-14 (2003).
- 90. S. Harvey, M. Kakebeeke, A.E. Murphy and E.J. Sanders, Growth hormone in the nervous system: autocrine or paracrine roles in retinal function? *Can J Physiol Pharmacol*, **81**, 371-384 (2003).

- 91. S. Takeuchi, M. Haneda, K. Teshigawara and S. Takahashi, Identification of a novel GH isoform: a possible link between GH and melanocortin systems in the developing chicken eye. *Endocrinology*, **142**, 5158-5166 (2001).
- M.L. Baudet, E.J. Sanders and S. Harvey, Retinal growth hormone in the chick embryo. Endocrinology, 144, 5459-5468 (2003).
- A. Hellstrom, E. Svensson, B. Carlsson, A. Niklasson and K. Albertsson-Wikland, Reduced retinal vascularization in children with growth hormone deficiency. *J Clin Endocrinol Metab*, 84, 795-798 (1999).
- 94. I. Struman, F. Bentzien, H. Lee, V. Mainfroid, G. D'Angelo, V. Goffin, R.I. Weiner and J.A. Martial, Opposing actions of intact and N-terminal fragments of the human prolactin/growth hormone family members on angiogenesis: an efficient mechanism for the regulation of angiogenesis. *Proc Natl Acad Sci U S A*, **96**, 1246-1251 (1999).
- 95. F. de Pablo, L.A. Scott and J. Roth, Insulin and insulin-like growth factor I in early development: peptides, receptors and biological events. *Endocr Rev*, **11**, 558-577 (1990).
- 96. J.J. Kopchick and S. Okada, Growth hormone receptor antagonists: discovery and potential uses. *Growth Horm IGF Res*, **11** Suppl A, S103-109 (2001).
- 97. S. Okada and J.J. Kopchick, Biological effects of growth hormone and its antagonist. *Trends Mol Med*, **7**, 126-132 (2001).
- 98. A. Bereket, C.H. Lang, M.E. Geffner and T.A. Wilson, Normal growth in a patient with septo-optic dysplasia despite both growth hormone and IGF-I deficiency. *J Pediatr Endocrinol Metab*, **11**, 69-75 (1998).
- 99. E.H. Hathout, D.J. Baylink and S. Mohan, Normal growth despite GH, IGF-I and IGF-II deficiency. *Growth Horm IGF Res*, **9**, 272-277 (1999).
- 100. L. Lazar, S. Dan and M. Phillip, Growth without growth hormone: growth pattern and final height of five patients with idiopathic combined pituitary hormone deficiency. *Clin Endocrinol (Oxf)*, **59**, 82-88 (2003).
- 101. M. Ishihara, Optic hypoplasia with pituitary dwarfism (Kaplan-Grumbach-Hoyt syndrome, or DeMorsier syndrome). *Endocrinol Jpn*, **30**, 7-14 (1983).
- 102. K. Miyako, M. Takemoto, K. Ihara, R. Kuromaru, H. Kohno and T. Hara, A case of growth hormone and gonadotropin deficiency associated with unilateral anophthalmia, microphallus, cryptorchidism, and mental retardation. *Endocr J*, **49**, 15-20 (2002).
- 103. A.M. Roth, Retinal vascular development in premature infants. Am J Ophthalmol, 84, 636-640 (1977).
- 104. A. Hellstrom, C. Perruzzi, M. Ju, E. Engstrom, A.L. Hard, J.L. Liu, K. Albertsson-Wikland, B. Carlsson, A. Niklasson, L. Sjodell, D. LeRoith, D.R. Senger and L.E. Smith, Low IGF-I suppresses VEGF-survival signaling in retinal endothelial cells: direct correlation with clinical retinopathy of prematurity. *Proc Natl Acad Sci U S A*, 98, 5804-5808 (2001).
- 105. A. Hellstrom, B. Carlsson, A. Niklasson, K. Segnestam, M. Boguszewski, L. de Lacerda, M. Savage, E. Svensson, L. Smith, D. Weinberger, K. Albertsson Wikland and Z. Laron, IGF-I is critical for normal vascularization of the human retina. *J Clin Endocrinol Metab*, 87, 3413-3416 (2002).
- L.E. Smith, J.J. Kopchick, W. Chen, J. Knapp, F. Kinose, D. Daley, E. Foley, R.G. Smith and J.M. Schaeffer, Essential role of growth hormone in ischemia-induced retinal neovascularization. *Science*, 276, 1706-1709 (1997).
- 107. P.H. Francis-West, R.K. Ladher and G.C. Schoenwolf, Development of the sensory organs. *Sci Prog*, **85**, 151-173 (2002).
- 108. M. Schwanzel-Fukuda and D.W. Pfaff, The migration of luteinizing hormone-releasing hormone (LHRH) neurons from the medial olfactory placode into the medial basal forebrain. *Experientia*, **46**, 956-962 (1990).

- H.L. Eisthen, R.J Delay, C.R. Wirsig-Wiechmann and V.E. Dionne, Neuromodulatory effects of gonadotropin releasing hormone on olfactory receptor neurons. *J Neurosci*, 20, 3947-3955 (2000).
- 110. A.I. Farbman, Olfactory neurogenesis: genetic or environmental controls? *Trends* Neurosci, 13, 362-365 (1990).
- 111. R. Doucette, Glial cells in the nerve fiber layer of the main olfactory bulb of embryonic and adult mammals. *Microsc Res Tech*, 24, 113-130 (1993).
- 112. I. Arsenejevic, Future perspectives-stem cells (Chapter 17, this book).
- M.T. Moreno-Flores, J. Diaz-Nido, F. Wandosell and J. Avila, Olfactory Ensheathing Glia: Drivers of Axonal Regeneration in the Central Nervous System? J Biomed Biotechnol, 2, 37-43 (2002).
- 114. A.G. Monti-Graziadei, Cell migration from the olfactory neuroepithelium of neonatal and adult rodents. *Brain Res Dev Brain Res*, **70**, 65-74 (1992).
- 115. M. Caggiano, J.S. Kauer and D.D Hunter, Globose basal cells are neuronal progenitors in the olfactory epithelium: a lineage analysis using a replication-incompetent retrovirus. *Neuron*, **13**, 339-352 (1994).
- 116. M. Schwartz Levey, D.M. Chikaraishi and J.S. Kauer, Characterization of potential precursor populations in the mouse olfactory epithelium using immunocytochemistry and autoradiography. *J Neurosci*, **11**, 3556-3564 (1991).
- 117. J.L. Willian, Development of the head and neck, in: *Human Embryology*, third edition, edited by L.S. Sherman, S.S. Potter, W.J. Scott (Churchill Livingstone, New York, Edinburg, London, Philadelphia, 2001), pp. 351-378.
- 118. M. Brown, R. Keynes and A. Lumsden, Development of sense organs, in: *The Developing Brain*, (Oxford University Press, New York, 2001) pp. 194-217.
- 119. I.V. Nosrat, S. Lindskog, A. Seiger and C.A. Nosrat, Lingual BDNF and NT-3 mRNA expression patterns and their relation to innervation in the human tongue: similarities and differences compared with rodents. *J Comp Neurol*, **417**, 133-152 (2000).
- 120. C.A. Nosrat, D.K. MacCallum and C.M. Mistretta, Distinctive spatiotemporal expression patterns for neurotrophins develop in gustatory papillae and lingual tissues in embryonic tongue organ cultures. *Cell Tissue Res*, **303**, 35-45 (2001).
- 121. I. Nosrat, A. Seiger, L. Olson and C.A. Nosrat, Expression patterns of neurotrophic factor mRNAs in developing human teeth. *Cell Tissue Res*, **310**, 177-187 (2002).
- 122. T. Ringstedt, C.F. Ibanez and C.A. Nosrat, Role of brain-derived neurotrophic factor in target invasion in the gustatory system. *J Neurosci*, **19**, 3507-3518 (1999).
- 123. C.M. Mistretta, K.A. Goosens, I. Farinas and L.F. Reichardt, Alterations in size, number, and morphology of gustatory papillae and taste buds in BDNF null mutant mice demonstrate neural dependence of developing taste organs. *J Comp Neurol*, **409**, 13-24 (1999).
- 124. K.O. Johnson, The roles and functions of cutaneous mechanoreceptors. *Curr Opin Neurobiol*, **11**, 455-461 (2001).
- 125. J. Zelena and I. Jirmanova, Reinnervation of rat Pacinian corpuscles after nerve crush during the postcritical period of development. *J Neurocytol*, **24**, 955-964 (1995).
- 126. B.T. Fundin, I. Silos-Santiago, P. Ernfors, A.M. Fagan, H. Aldskogius, T.M. DeChiara, H.S. Phillips, M. Barbacid, G.D. Yancopoulos and F.L. Rice, Differential dependency of cutaneous mechanoreceptors on neurotrophins, trk receptors, and P75 LNGFR. *Dev Biol*, **190**, 94-116 (1997).
- 127. P. Emfors, K.F. Lee and R. Jaenisch, Mice lacking brain-derived neurotrophic factor develop with sensory deficits. *Nature*, **368**, 147-150 (1994).
- 128. P. Ernfors, K.F. Lee, J. Kucera and R. Jaenisch, Lack of neurotrophin-3 leads to deficiencies in the peripheral nervous system and loss of limb proprioceptive afferents. *Cell*, **77**, 503-512 (1994).
- 129. R. Cantos, L.K. Cole, D. Acampora, A. Simeone and D.K. Wu, Patterning of the mammalian cochlea. *Proc Natl Acad Sci U S A* 97, 11707-13 (2000).

- 130. D.M. Fekete and D. K. Wu, Revisiting cell fate specification in the inner ear. *Curr* Opin Neurobiol, **12**, 35-42 (2002).
- 131. L.D. Saffer, R. Gu and J.T. Corwin, An RT-PCR analysis of mRNA for growth factor receptors in damaged and control sensory epithelia of rat utricles. *Hear Res*, **94**, 14-23 (1996).
- 132. K.H. Lee and D.A. Cotanche, Localization of the hair-cell-specific protein fimbrin during regeneration in the chicken cochlea. *Audiol Neurootol*, **1**, 41-53 (1996).
- 133. B.L. Resendes, N.G. Robertson, J.D. Szustakowski, R.J. Resendes, Z. Weng and C.C. Morton, Gene discovery in the auditory system: characterization of additional cochlear-expressed sequences. *J Assoc Res Otolaryngol*, **3**, 45-53 (2002).
- 134. Y. Cho, T.W. Gong, T. Stover, M.I. Lomax and R.A. Altschuler, Gene expression profiles of the rat cochlea, cochlear nucleus, and inferior colliculus. J Assoc Res Otolaryngol, **3**, 54-67 (2002).
- 135. I. Varela-Nieto, J.A. Morales-Garcia, P. Vigil, A. Diaz-Casares, I. Gorospe, S. Sánchez-Galiano, S. Cañon, G. Camarero, J. Contreras, R. Cediel and Y. Leon, Trophic effects of Insulin-like growth factor-I (IGF-I) in the inner ear. *Hearing Research* (in press) (2004).
- 136. L.M. Frago, G. Camarero, S. Canon, C. Paneda, C. Sanz, Y. Leon, F. Giraldez and I. Varela-Nieto, Role of diffusible and transcription factors in inner ear development: implications in regeneration. *Histol Histopathol*, **15**, 657-666 (2000).
- 137. C. Sanz, Y. Leon, S. Canon, L. Alvarez, F. Giraldez, and I. Varela-Nieto, Pattern of expression of the jun family of transcription factors during the early development of the inner ear: implications in apoptosis. *J Cell Sci*, **112** (**Pt 22**), 3967-3974 (1999).
- C. Sanz, Y. Leon, J. Troppmair, U. R. Rapp and I. Varela-Nieto, Strict regulation of c-Raf kinase levels is required for early organogenesis of the vertebrate inner ear. *Oncogene*, 18, 429-437 (1999).
- 139. L. M. Frago, S. Canon, E. J. de la Rosa, Y. Leon, and I. Varela-Nieto, Programmed cell death in the developing inner ear is balanced by nerve growth factor and insulin-like growth factor I. *J Cell Sci*, **116**, 475-486 (2003).
- 140. G. Camarero, Y. Leon, I. Gorospe, F. De Pablo, B. Alsina, F. Giraldez, and I. Varela-Nieto, Insulin-like growth factor 1 is required for survival of transit-amplifying neuroblasts and differentiation of otic neurons. *Dev Biol*, **262**, 242-253 (2003).
- 141. E.C. Oesterle, T.T. Tsue and E.W. Rubel, Induction of cell proliferation in avian inner ear sensory epithelia by insulin-like growth factor-I and insulin. *J Comp Neurol*, **380**, 262-274 (1997).
- 142. A.L. Kuntz and E.C. Oesterle, Transforming growth factor alpha with insulin stimulates cell proliferation in vivo in adult rat vestibular sensory epithelium. *J Comp Neurol*, **399**, 413-423 (1998).
- 143. H. Staecker and T.R. Van De Water, Factors controlling hair-cell regeneration/repair in the inner ear. *Curr Opin Neurobiol*, **8**, 480-487 (1998).
- 144. R. Romand and S. Chardin, Effects of growth factors on the hair cells after ototoxic treatment of the neonatal mammalian cochlea in vitro. *Brain Res*, **825**, 46-58 (1999).
- 145. M. Duan, K. Agerman, P. Ernfors and B. Canlon, Complementary roles of neurotrophin 3 and a N-methyl-D-aspartate antagonist in the protection of noise and aminoglycosideinduced ototoxicity. *Proc Natl Acad Sci USA*, **97**, 7597-7602 (2000).
- 146. D.J. Stacey and W.G. McLean, Cytoskeletal protein mRNA expression in the chick utricle after treatment in vitro with aminoglycoside antibiotics: effects of insulin, iron chelators and cyclic nucleotides. *Brain Res*, **871**, 319-332 (2000).
- 147. R.D. Kopke, R.L. Jackson, G. Li, M.D. Rasmussen, M.E. Hoffer, D.A. Frenz, M. Costello, P. Schultheiss and T.R. Van De Water, Growth factor treatment enhances vestibular hair cell renewal and results in improved vestibular function. *Proc Natl Acad Sci USA*, **98**, 5886-5891 (2001).

- 148. S. Kanzaki, T. Stover, K. Kawamoto, D.M. Prieskorn, R.A. Altschuler, J.M. Miller and Y. Raphael, Glial cell line-derived neurotrophic factor and chronic electrical stimulation prevent VIII cranial nerve degeneration following denervation. *J Comp Neurol*, **454**, 350-360 (2002).
- 149. B. Malgrange, J.M. Rigo, P. Coucke, M. Thiry, G. Hans, L. Nguyen, T.R. van de Water, G. Moonen and P.P. Lefebvre, Identification of factors that maintain mammalian outer hair cells in adult organ of Corti explants. *Hear Res*, **170**, 48-58 (2002).

242

## Chapter 10

#### **ROLE OF INSULIN-LIKE GROWTH FACTORS IN NEURONAL PLASTICITY AND NEUROPROTECTION**

Ignacio Torres Aleman Laboratory of Neuroendocrinology, Instituto Cajal, CSIC, Madrid, Spain

Key words: Brain function; nerve cell plasticity; neurodegenerative diseases; neuro-repair; synaptic contacts.

#### **1. INTRODUCTION**

The notion that insulin-like growth factor (IGF-I) is an important modulator of brain function is now reasonably well established. The classical role assigned to IGF-I in the brain is to promote neuronal survival during development. Indeed, the canonical pro-survival pathway in developing neurons; which includes the ser-kinase Akt, was first described downstream of the IGF-I receptor<sup>[31]</sup>. But the biological role of IGF-I is not circumscribed to the developmental period; accumulating evidence indicates that this growth factor exerts a wide-spectrum protective action in the adult brain<sup>[19]</sup>. The last decade has witnessed a spur of studies on the neurobiological actions of IGF-I. The result is that we now may consider IGF-I as a prototypical neuroprotective factor. Much less is known however about the potential role of the other known member of the family, IGF-II. This is mostly attributable to the fact that the reported actions of IGF-II in the brain are mediated by the IGF-I receptor, and since IGF-II has a lower affinity for this receptor, it is widely accepted that the major player in brain physiology is IGF-I. This situation has led to a reduced interest in determining the role of IGF-II in the brain, and for that matter, elsewhere. However, high levels of IGF-II in particular brain locations<sup>11</sup> and widespread distribution of its receptor<sup>42</sup> support the idea that this growth factor has a role in brain physiology other than to mimic IGF-I actions.

IGFs are synthesized by the developing brain<sup>10,53,54</sup>, with mRNA levels becoming much lower in the adult tissue, except for very specific locations such as the choroid plexus and brain vessels for IGF-II<sup>45,63,80</sup> or a few scattered brain regions for IGF-I<sup>3,6,36</sup>. However, IGF protein levels, in particular IGF-I, do not decrease in many brain regions as much as mRNA levels, which suggested the existence of an extracerebral source of IGFs<sup>84</sup>. Recent studies have confirmed that a substantial amount of brain IGF-I is taken up from the circulation and that this uptake is modulated by physiological changes<sup>16,17</sup>. The case of IGF-II is easier to explain because cerebrospinal fluid (CSF)-derived IGF-II produced by the choroid plexus can reach the brain parenchyma. This may suggest that there is a switch from a paracrine/autocrine pathway during development to an endocrine pathway in the adult in the action of IGFs in the brain. Nevertheless this idea is tempered by the fact that extracerebral IGFs can similarly access the brain during development because they are abundant in the periphery, and locally synthesized IGFs, albeit at lower levels than during embryogenesis, may also be important in local modulation of adult brain function.

At any rate, IGFs are present in the developing and adult brain and modulate a wide variety of brain functions in a developmental and anatomically regulated fashion. In the present chapter we discuss the contribution of IGF-I to two major brain properties that are intimately interconnected, neural plasticity and neuroprotection. The two processes underscore major functional traits in brain organization.

Neural plasticity is a broad, widely used concept that refers to an intrinsic property of the brain: the ability to adapt to a constantly changing environment. Although it is quite likely that all cell constituents within brain tissue participate in its plastic properties, the current bias favoring neurons as the cells that confer the nervous system its functional properties has led to analysis of mechanisms underlying neuronal plasticity in much greater detail than plastic processes in other cell types such as glia or endothelia. Fortunately, this trend is changing, and glial cells are becoming accepted partners in neural plasticity processes<sup>8</sup>. Since neuronal function means cell communication, neuronal plasticity translates into regulated variability in neuronal transmission. Because neuron-to-neuron communication takes place at the synapse, in the majority of cases changes in synaptic plasticity. But changes in the size of neuronal populations, in neuronal excitability, or in the

metabolic status of neurons, to name a few, are undoubtedly affecting the ability of neurons to cope to changing demands.

Neuroprotection is also a comprehensive concept that includes a wide variety of processes. Usually we use the term neuroprotection when referring to therapeutic maneuvers to counteract brain disease; but we can also employ this term when analyzing endogenous mechanism that allow the brain to cope with potentially pathological disturbances. In the latter sense, IGF-I is indeed a neuroprotective signal<sup>83</sup>. A functional link between neuroprotection and neuronal plasticity is easily inferred; neuroprotective processes must aim plastic processes if they are meant to preserve neuronal function.

#### 2. NEURAL PLASTICITY

The developing brain is considered to exhibit a high level of cell plasticity. Both genetic and epigenetic cues allow a substantial degree of flexibility in the final number of cells formed and in the establishment of connections among cell populations. Abundant literature documents in detail the role of growth factors in controlling these aspects of cell plasticity during brain formation<sup>81</sup>. Although in this regard, IGF-I is not usually included as a neurotrophic factor in brain development, a field dominated by the neurotrophin family of growth factors, genetic analysis has firmly established an essential role for IGF-I in controlling brain size by modulating number of cells produced, degree of fiber myelination, and connectivity<sup>7,21</sup>.

Once brain development is completed, plasticity will rely on the resources acquired during development, *i.e.*, cell numbers and connectivity, as well as on emerging properties inherent to functional brain cells. As commented earlier, these plastic properties have been studied in neurons with a prominent focus on synaptic plasticity mechanisms because the level of activity of neural circuitries relies to a great extent in the strength of the synapses involved. In turn, synaptic strength will be determined by the amount of neurotransmitter released by the presynaptic terminal, the abundance of functionally active postsynaptic neurotransmitter receptors, and by the intrinsic excitability of the postsynaptic neuron. The number of synapses recruited, which in turn is related to the size of the neuronal population involved, will also determine overall activity of the circuitry. All these parameters are modulated by IGF-I.

Release of several types of neurotransmitters has been reported to be regulated by IGF-I under experimental conditions. These include major neurotransmitter systems such as acetylcholine<sup>57,76</sup>, glutamate<sup>34</sup>, and GABA<sup>22,76</sup>. Although modulation of GABA release by IGF-I has been documented to take place also *in vivo*, we still do not know the physiological

importance of these observations. In the case of acetylcholine, both stimulatory and inhibitory effects have been reported, while glutamate levels at the synaptic cleft are regulated indirectly by IGF-I through regulation of the expression of glutamate transporters.

The number of postsynaptic neurotransmitter receptors is also controlled by IGF-I by modulating their membrane trafficking and insertion. These effects are independent of the classical trophic actions of IGF-I in controlling the abundance of neurotransmitter receptors through modulation of their synthesis<sup>50,78,92</sup>. Insertion or internalization of neurotransmitter receptors by IGF-I allows for a rapid modulation of synaptic strength. Specifically, IGF-I modulates the cell surface distribution of AMPA and kainate subtype of glutamate receptors, albeit in opposite ways<sup>38,60</sup>, and possibly also of GABA receptors<sup>88</sup>. Although *in vivo* evidence supports a physiological role for IGF-I modulation of glutamate receptor availability on synaptic activity<sup>20</sup>, it is still premature to conclude what are the functional consequences *in vivo*, mostly because different functional outcomes can be envisaged.

By far, the best documented action of IGF-I as a modulator of neuronal plasticity is through modulation of neuronal excitability by interacting with different types of neuronal membrane ion channels. These include Cl<sup>-</sup>, Ca<sup>++</sup>, and K<sup>+</sup> channels<sup>9,47,70,73,74</sup>. In all instances, IGF-I potentiates the activity of these currents through mechanisms involving allosteric modification of ion channel function or translocation to the membrane. Although the net effect on neuronal excitability is difficult to anticipate because stimulation of these broad set of ion channels will have a disparate effect on membrane potential, *in vivo* recordings after systemic IGF-I administration reveals increased neuronal excitability in very specific brain nuclei<sup>16</sup>. Intriguingly, parallel expansion in the receptive fields of stimulated neurons is observed<sup>70</sup>.

Although still speculative, coupling of increased neuronal excitability to enlargement of the receptive field after IGF-I exposure may contribute to learning-related plastic mechanisms operating in neurons. Modifications in the size of neuronal receptive fields are postulated to underlie learning processes<sup>13</sup>. Indeed, training in a particular task results in enlargement of the neuronal population involved in task performance<sup>69</sup>. This enlargement is possible through recruitment of new synapses within the circuitry and recruitment of new synapses may be brought about through increased excitability of the neurons in the circuit. This increased in neuronal excitability may develop in the following sequence (Fig. 1): 1) engagement of a neuronal population in a given task will increase their activity; 2) through coupling of increased neuronal activity to increased blood flow, a phenomenon that forms the basis of fMRI analysis, blood-borne IGF-I will accumulate into active neurons (this activity-dependent uptake of systemic IGF-I is theoretically possible because entrance of circulating IGF-I into the brain is stimulated by physiological stimuli such as physical exercise that stimulates brain blood flow); 3) in turn, neurons accumulating IGF-I will become more excitable. Therefore, circulating IGF-I may constitute a physiological mediator of activity-dependent changes in the size of the receptive field of neurons. This introduces a fascinating new role of blood-borne IGF-I as an important modulator of cognition. Notably, serum IGF-I levels are correlated with cognitive status in humans<sup>56,86</sup>.



*Figure 1.* A role for circulating IGF-I in learning processes. Expansion of the receptive field of neurons brought about by accumulation of serum IGF-I by neurons may underlie changes in circuitry properties associated to the learning process. Stimulation of neuronal activity after behavioral engagement (as during learning) leads to increased blood flow to the stimulated area. In turn, serum IGF-I will accumulate in activated neurons which will increase their excitability<sup>16</sup> associated with an increase in the size of their receptive field<sup>70</sup>.

Synaptic remodeling refers to the dynamic status of synapses. Recent work indicates that synaptic turnover at dendritic spines is more frequent and faster than previously thought. It is easy to infer that regulation of this event will contribute to neuronal plasticity processes as the number of synapses engaged will heavily determine the activity of the circuitry. Ultrastructural studies indicate that the density and size of dendritic spines depends on IGF-I input to target neurons such as Purkinje cells in the cerebellum<sup>67</sup>. As commented earlier, the expansion of the receptive field on exposure to IGF-I

may theoretically involve not only strengthening of existing synapses, but also the formation of new ones. We hypothesize that IGF-I acts as a synaptic "stabilizer," favoring the stable formation of synapses through mechanisms involving synaptic-specific expression of proteins involved in pre- and postsynaptic interactions such as membrane adhesion molecules. This is based on the observation that IGF-I modulates the polarity and motility of many different types of cells by controlling the expression of proteins involved in cell-to-cell interactions<sup>61</sup>. However, this possibility remains entirely speculative at present.

### **3. NEUROPROTECTION**

Physiological neuroprotective mechanisms aim to maintain neurons alive and functional. Owing to the restricted neuronal turnover capacity of the adult brain, maintenance of neuronal populations appears to be crucial for proper function. A widely used strategy in the mammalian nervous system is to contain functionally overlapping cell populations. In this way, function is compromised only after extensive neuronal loss occurs, such as observed in Parkinson patients, in whom motor deficits appear only after more than 80% of the dopaminergic neurons in the substantia nigra are dead or atrophied<sup>25</sup>. Genetic manipulation has shown that IGF-I governs the size of neuronal populations. While excess IGF-I leads to bigger brains, its absence produces microcephaly<sup>27</sup>. Therefore, a first protective action of IGF-I already takes place during brain development by controlling the final size of neuronal populations.

Albeit limited in the adult period in mammals, another neuroprotective strategy is to produce new neurons, as seen in the hippocampus and olfactory bulb. Because the functional significance of adult neurogenesis is not yet established<sup>51</sup>, we can only speculate about its role as a neuroprotective strategy. However, recent work indicates that the ability of IGF-I to control adult neurogenesis is markedly neuroprotective. In the absence of appropriate IGF-I input, IGF-I-deficient mutant mice show substantial decreases in new neuron formation together with enhanced susceptibility to injury<sup>85</sup>. Hence, a reduced capacity to produce new neurons affects the ability of the brain to cope with insults<sup>4</sup>.

Apart from these preventive measures oriented to produce excess neurons and replace lost ones, the brain has developed self-reparative mechanisms against insults. Therefore, loss of function will eventually develop only if preventive and reparative processes are overcome. IGF-I participates also in this aspect of neuroprotection as a potent reparative agent. The best characterized action of IGF-I is as an antiapoptotic signal<sup>29</sup>. However, IGF-I participates also in other reparative processes, acting as an anti-inflammatory agent by counteracting the actions of pro-inflammatory cytokines, for instance<sup>28,71</sup>. IGF-I also participates in reactive angiogenesis occuring after injury and in this way ensures proper nutrient supply to damaged areas (unpublished observations). Re-establishment of lost connections is also an important aspect of IGF-I-dependent repair<sup>33,41</sup>. The cellular and molecular pathways recruited by IGF-I in these processes are still poorly characterized.

IGF-I protects against almost any kind of *in vitro* or *in vivo* injury to brain cells, from those of ample occurrence such as oxidative damage to disease-specific derangements such as  $\beta$ -amyloidosis (Table 1). This remarkable ability to protect against so widely different types of insults suggests that neurorepair by IGF-I proceeds through a fundamental protective pathway, such as activation of Akt, the canonical pro-survival kinase downstream of the IGF-I receptor<sup>31</sup>.

In vitro	Reference	In vivo	Reference
Oxidative stress	43	Ischemic insult	40
Neurotoxic drugs	12,26	Neurotoxins	33,72
Hypoxia	90	Trauma	49
β Amyloid	68	Alzheimer amyloidosis	18
Huntingtin	44	Motor neuron diseases	48,6
Hypoglycemia	23	Cerebellar ataxia	17,32
		Brain senescence	62
		Nerve transaction	46
		Diabetic neuropathy	91

Table 1. In vitro and in vivo neuroprotection by IGF-I

Alternatively, we recently suggested that the reason underlying the ability of IGF-I to protect against so many different types of insults is that pathological changes develop only after a tonic protective action of IGF-I upon neurons is compromised<sup>35</sup>. Specifically, we speculate that common pathological processes such as inflammation, excitotoxicity, or accumulation of deleterious molecules such as  $\beta$ -amyloid or glycated products lead to neuronal death, at least in part through their ability to counteract the protective actions of IGF-I<sup>85</sup> (Fig. 2).


Figure 2. Pathological derangements lead to neuronal demise at least in part by inhibiting prosurvival actions of IGF-I.

What are these tonic protective actions of IGF-I on the adult brain? Indeed, fundamental homeostatic processes are modulated by this multifaceted growth factor. IGF-I ensures adequate nutrient sources to neurons by increasing neuronal glucose uptake and lactate production by astrocytes<sup>24,77</sup>. Albeit still controversial, lactate from astrocytes has been proposed to contribute to metabolic demands in active neurons<sup>87</sup>. In this way IGF-I would stimulate neuronal excitability and at the same time ensure proper nutrient supply to stimulated neurons. In this vein, IGF-I also increases brain blood flow and the formation of new brain vessels<sup>37</sup> (and unpublished observations), allowing nutrient supply on demand together with proper oxygen levels. Conceivably, the latter is achieved through an activitydependent process similar to that already described for IGF-I-mediated changes in the receptive field size of activated neurons (see Fig.1). Since a low rate of basal angiogenesis in the adult brain is thought to be maintained to match functional demands, activity-dependent uptake of serum IGF-I, a potent angiogenic promoter, would result in local increases in angiogenesis on demand.

More specialized processes regulated by IGF-I include brain clearance of potentially hazardous substances such as  $\beta$ -amyloid peptides through a complex regulatory mechanism involving a network of  $\beta$ -amyloid transport proteins interacting at the blood-brain interface<sup>18</sup>. This novel mechanism of IGF-I neuroprotection has been discussed in detail recently<sup>15</sup> and points to an intriguing possibility: Alzheimer's neuropathology may develop as a consequence of a gradual loss of sensitivity to IGF-I during aging. However, this remains to be shown.

As suggested earlier, IGF-I may also be necessary to maintain appropriate synaptic contacts. This is also a fundamental way to maintain healthy neuronal populations since changes in the number of synaptic contacts lead to changes in neuronal excitation which in turn modulates neuronal performance through changes in synaptic gain<sup>55</sup>. Therefore, the ability of systemic IGF-I to increase neuronal excitability may contribute to strengthening synaptic contacts between stimulated partners.

## 4. PATHOLOGICAL SIGNIFICANCE AND THERAPEUTIC POTENTIAL OF IGF-I IN BRAIN DISEASE

Genetic ablation of IGF-I has shown that this growth factor is important for appropriate organ growth, including the brain<sup>7</sup>. Although brain-specific null IGF-I mutants have not been yet developed, genetic models interfering with IGF-I actions on the brain have shown that insufficient IGF-I input to the brain affects its size. This reduction may be due to decreased generation of neurons, reduced neuron survival or both. Probably because of its critical role, null IGF-I mutants have not been found among the human population (only a small proportion of null IGF-I mice survive for several weeks). However, children with very low levels of IGF-I has been reported in whom brain deficits accompanied mental retardation<sup>14</sup>. Although a few recent examples of IGF-I receptor defects have been documented in children with growth retardation<sup>1</sup>, these have not yet been analyzed for possible brain anomalies, which can be expected. At any rate, lack of IGF-I during brain development will compromise cell resources and adequate connectivity in the adult, leading to functional derangements.

As for the adult brain, we speculate that if a constant trophic input to the brain through IGF-I is required for adequate function, abnormal IGF-I signaling will have pathological consequences. Restoration of adequate IGF-I signaling through novel therapeutic strategies can be anticipated to be beneficial. However, these two notions are far from being proved yet. Albeit compelling evidence indicates that low IGF-I input probably contributes and even leads in specific cases to brain disease<sup>85</sup>, this possibility needs to be further substantiated. While IGF-I, as well as complementary strategies such as potentiation of endogenous IGF-I signaling<sup>59</sup>, have been tested as therapeutic tools in various brain diseases, a combination of shortcomings and distorting factors precludes any firm conclusion about the true therapeutic potential of IGF-I for neurodegenerative diseases<sup>2,32,39,40,48,52,58,64,65,72,75,82</sup>. The fact that controversial epidemiological data suggests that IGF-I may have tumorigenic effects<sup>5,30,79,89</sup> has considerably hindered further progress with additional clinical assays. Because new evidence supports a therapeutic use of IGF-I in diseases such as Alzheimer's dementia<sup>18</sup> or Huntington's chorea<sup>44</sup>, we can anticipate a deeper and more detailed insight into this largely neglected area of research with IGF-I. To circumvate potential carcinogenic actions of IGF-I, the pharmaceutical industry is actively looking into synthetic compounds mimicking neuroprotective pathways used by IGF-I. This is a very logical and promising strategy that will hopefully yield new useful drugs for neurodegenerative diseases.

#### 5. FUTURE PERSPECTIVES

We anticipate that the therapeutic use of IGF-I in neurodegenerative diseases is going to be one "hot-spot" in research in this field. Relying on ongoing clinical trials with IGF-I in diabetes and other endocrine diseases, soon-to-come assays in amyotrophic lateral sclerosis and hopefully in cerebellar ataxia, analysis of the therapeutic potential of IGF-I in Alzheimer's disease and in Huntington's chorea will be very likely implemented. Insight into the cellular and molecular pathways used by IGF-I for its extraordinarily ample protective action on the brain will also be developed. This aspect is particularly relevant because knowledge of physiological neuroprotective mechanisms will allow targeting of new molecules for drug research.

#### ACKNOWLEDGEMENTS

Work in the author's lab has been possible through grants from the Spanish Ministry of Science (DGCYT), Health Research Fund (FIS), Community of Madrid (CAM) and Neuropharma.

#### REFERENCES

- 1. M.J. Abuzzahab, A. Schneider, A. Goddard, F. Grigorescu, C. Lautier, E. Keller, W. Kiess, J. Klammt, J. Kratzsch, D. Osgood, R. Pfaffle, K. Raile, B. Seidel, R.J. Smith and S.D. Chernausek, IGF-I receptor mutations resulting in intrauterine and postnatal growth retardation. *N Engl J Med*, **349**, 2211-2222 (2003).
- 2. P. Aebischer and J. Ridet, Recombinant proteins for neurodegenerative diseases: the delivery issue. Trends Neurosci, 24, 533-540 (2001).
- 3. I.K. Andersson, D. Edwall, G. Norstedt, B. Rozell, A. Skottner and H.A. Hansson, "Differing expression of insulin-like growth factor I in the developing and in the adult rat cerebellum," *Acta Physiol Scand.* **132**, 167-173 (1988).
- 4. R.J. Armstrong and R.A. Barker, Neurodegeneration: a failure of neuroregeneration? *Lancet*, **358**, 1174-1176 (2001).

- R. Baffa, K. Reiss, E.A. El Gabry, J. Sedor, M.L. Moy, D. Shupp-Byrne, S.E. Strup, W.W. Hauck, R. Baserga and L.G. Gomella, Low serum insulin-like growth factor 1 (IGF-1): a significant association with prostate cancer. *Tech Urol*, 6, 236-239 (2000).
- W.P. Bartlett, X.S. Li, M. Williams and S. Benkovic, Localization of insulin-like growth factor-1 mRNA in murine central nervous system during postnatal development. *Dev Biol*, 147, 239-250 (1991).
- K.D. Beck, L. Powell-Braxton, H.R. Widmer, J. Valverde and F. Hefti, Igf1 gene disruption results in reduced brain size, CNS hypomyelination, and loss of hippocampal granule and striatal parvalbumin-containing neurons. *Neuron*, 14, 717-730 (1995).
- 8. P. Bezzi and A. Volterra, A neuron-glia signalling network in the active brain. *Curr Opin Neurobiol*, **11**, 387-394 (2001).
- 9. L.A. Blair and J. Marshall, IGF-1 modulates N and L calcium channels in a PI 3kinase-dependent manner. *Neuron*, **19**, 421-429 (1997).
- 10. C.A. Bondy, Transient IGF-I gene expression during the maturation of functionally related central projection neurons. *J Neurosci*, **11**, 3442-3455 (1991).
- 11. C.A. Bondy and W.H. Lee, Patterns of insulin-like growth factor and IGF receptor gene expression in the brain. Functional implications. *Ann NY Acad Sci*, **692**, 33-43 (1993).
- 12. D.J. Bonthius, B. Karacay, D. Dai and N.J. Pantazis, FGF-2, NGF and IGF-1, but not BDNF, utilize a nitric oxide pathway to signal neurotrophic and neuroprotective effects against alcohol toxicity in cerebellar granule cell cultures. *Brain Res Dev Brain Res*, **140**, 15-28 (2003).
- D.V. Buonomano and M.M. Merzenich, Cortical plasticity: from synapses to maps. Annu Rev Neurosci, 21, 149-186 (1998).
- 14. C. Camacho-Hubner, K.A. Woods, A.J. Clark and M.O. Savage, Insulin-like growth factor (IGF)-I gene deletion. *Rev Endocr Metab Disord*, **3**, 357-361 (2002).
- 15. I. Torres-Alemán and Carro E, "Insulin-like growth factor I and Alzheimer's disease: therapeutic prospects? *Exp Rev Neurotherapy*, (in press).
- E. Carro, A. Nunez, S. Busiguina and I. Torres-Aleman, Circulating insulin-like growth factor I mediates effects of exercise on the brain. *J Neurosci*, 20, 2926-2933 (2000).
- 17. E. Carro, J.L. Trejo, S. Busiguina and I. Torres-Aleman, Circulating insulin-like growth factor I mediates the protective effects of physical exercise against brain insults of different etiology and anatomy. *J Neurosci*, **21**, 5678-5684 (2001).
- 18. E. Carro, J.L. Trejo, T. Gomez-Isla, D. LeRoith, and I. Torres-Aleman, Serum insulin-like growth factor I regulates brain amyloid-beta levels. *Nat Med*, (2002).
- E. Carro, J.L. Trejo, A. Nunez and I. Torres-Aleman, Brain repair and neuroprotection by serum insulin-like growth factor I. *Mol Neurobiol*, 27, 153-162 (2003).
- R.C. Carroll, E.C. Beattie, M. Von Zastrow and R.C. Malenka, Role of AMPA receptor endocytosis in synaptic plasticity. *Nat Rev Neurosci*, 2, 315-324 (2001).
- M.J. Carson, R.R. Behringer, R.L. Brinster and F.A. McMorris, Insulin-like growth factor I increases brain growth and central nervous system myelination in transgenic mice. *Neuron*, 10, 729-740 (1993).
- M.A. Castro-Alamancos and I. Torres-Aleman, Long-term depression of glutamateinduced gamma-aminobutyric acid release in cerebellum by insulin-like growth factor I. *Proc Natl Acad Sci USA*, **90**, 7386-7390 (1993).

- B. Cheng and M.P. Mattson, IGF-I and IGF-II protect cultured hippocampal and septal neurons against calcium-mediated hypoglycemic damage. J Neurosci, 12, 1558-1566 (1992).
- C.M. Cheng, R.R. Reinhardt, W.H. Lee, G. Joncas, S.C. Patel and C.A. Bondy, Insulin-like growth factor 1 regulates developing brain glucose metabolism. *Proc Natl Acad Sci USA*, 97, 10236-10241 (2000).
- G. Clarke, R.A. Collins, B.R. Leavitt, D.F. Andrews, M.R. Hayden, C.J. Lumsden and R.R. McInnes, A one-hit model of cell death in inherited neuronal degenerations. *Nature*, 406, 195-199 (2000).
- A.M. Corse, M.M. Bilak, S.R. Bilak, M. Lehar, J.D. Rothstein and R.W. Kuncl, Preclinical testing of neuroprotective neurotrophic factors in a model of chronic motor neuron degeneration. *Neurobiol Dis*, 6, 335-346 (1999).
- A.J. D'Ercole, P. Ye and G. Gutierrez-Ospina, Use of transgenic mice for understanding the physiology of insulin-like growth factors. *Horm Res*, 45 Suppl 1, 5-7 (1996).
- R. Dantzer, G. Gheusi, R.W. Johnson and K.W. Kelley, Central administration of insulin-like growth factor-1 inhibits lipopolysaccharide-induced sickness behavior in mice. *Neuroreport*, **10**, 289-292 (1999).
- 29. F. de Pablo and E.J. de la Rosa, The developing CNS: a scenario for the action of proinsulin, insulin and insulin-like growth factors. *Trends Neurosci*, **18**, 143-150 (1995).
- K. DeLellis, S. Ingles, L. Kolonel, R. McKean-Cowdin, B. Henderson, F. Stanczyk and N.M. Probst-Hensch, "IGF1 genotype, mean plasma level and breast cancer risk in the Hawaii/Los Angeles multiethnic cohort. *Br J Cancer*, 88, 277-282 (2003).
- H. Dudek, S.R. Datta, T.F. Franke, M.J. Birnbaum, R. Yao, G.M. Cooper, R.A. Segal, D.R. Kaplan and M.E. Greenberg, Regulation of neuronal survival by the serine-threonine protein kinase Akt. *Science*, 275, 661-665 (1997).
- A.M. Fernandez, A.G. de la Vega and I. Torres-Aleman, Insulin-like growth factor I restores motor coordination in a rat model of cerebellar ataxia. *Proc Natl Acad Sci* USA, 95, 1253-1258 (1998).
- A.M. Fernandez, A.G. Gonzalez de la Vega, B. Planas and I. Torres-Aleman, Neuroprotective actions of peripherally administered insulin-like growth factor I in the injured olivo-cerebellar pathway. *Eur J Neurosci*, 11, 2019-2030 (1999).
- C. Gamboa and A. Ortega, Insulin-like growth factor-1 increases activity and surface levels of the GLAST subtype of glutamate transporter. *Neurochem Int*, 40, 397-403 (2002).
- E. Garcia-Galloway, C. Arango, S. Pons and I. Torres-Aleman, Glutamate excitotoxicity attenuates insulin-like growth factor-i prosurvival signaling. *Mol Cell Neurosci*, 24, 1027-1037 (2003).
- L.M. Garcia-Segura, J. Perez, S. Pons, M.T. Rejas and I. Torres-Aleman, Localization of insulin-like growth factor I (IGF-I)-like immunoreactivity in the developing and adult rat brain. *Brain Res*, 560, 167-174 (1991).
- C.M. Gillespie, A.L. Merkel and A.A. Martin, Effects of insulin-like growth factor-I and LR3IGF-I on regional blood flow in normal rats. *J Endocrinol*, 155, 351-358 (1997).
- D.L. Gonzalez, W. Buno, S. Pons, M.S. Garcia-Calderat, E. Garcia-Galloway and I. Torres-Aleman, Insulin-like growth factor I potentiates kainate receptors through a phosphatidylinositol 3-kinase dependent pathway. *Neuroreport*, **12**, 1293-1296 (2001).

- S. Grinspoon, K. Miller, D. Herzog, D. Clemmons and A. Klibanski, Effects of Recombinant Human Insulin-Like Growth Factor (IGF)-I and Estrogen Administration on IGF-I, IGF Binding Protein (IGFBP)-2, and IGFBP-3 in Anorexia Nervosa: A Randomized-Controlled Study. *J Clin Endocrinol Metab*, 88, 1142-1149 (2003).
- 40. J. Guan, L. Bennet, P.D. Gluckman and A.J. Gunn, Insulin-like growth factor-1 and post-ischemic brain injury. *Prog Neurobiol*, **70**, 443-462 (2003).
- K.M. Guthrie, T. Nguyen and C.M. Gall, Insulin-like growth factor-1 mRNA is increased in deafferented hippocampus: spatiotemporal correspondence of a trophic event with axon sprouting. *J Comp Neurol*, 352, 147-160 (1995).
- C. Hawkes and S. Kar, Insulin-like growth factor-II/mannose-6-phosphate receptor: widespread distribution in neurons of the central nervous system including those expressing cholinergic phenotype. J Comp Neurol, 458, 113-127 (2003).
- 43. S. Heck, F. Lezoualc'h, S. Engert and C. Behl, Insulin-like growth factor-1mediated neuroprotection against oxidative stress is associated with activation of nuclear factor kappaB. *J Biol Chem*, **274**, 9828-9835 (1999).
- 44. S. Humbert, E.A. Bryson, F.P. Cordelieres, N.C. Connors, S.R. Datta, S. Finkbeiner, M.E. Greenberg and F. Saudou, The IGF-1/Akt pathway is neuroprotective in Huntington's disease and involves Huntingtin phosphorylation by Akt. *Dev Cell*, **2**, 831-837 (2002).
- M.A. Hynes, P.J. Brooks, J.J. Van Wyk and P.K. Lund, Insulin-like growth factor II messenger ribonucleic acids are synthesized in the choroid plexus of the rat brain. *Mol Endocrinol*, 2, 47-54 (1988).
- 46. M. Kanje, A. Skottner, J. Sjoberg and G. Lundborg, Insulin-like growth factor I (IGF-I) stimulates regeneration of the rat sciatic nerve. *Brain Res*, **486**, 396-398 (1989).
- M. Kanzaki, Y.Q. Zhang, H. Mashima, L. Li, H. Shibata and I. Kojima, Translocation of a calcium-permeable cation channel induced by insulin- like growth factor-I. *Nat Cell Biol*, 1, 165-170 (1999).
- B.K. Kaspar, J. Llado, N. Sherkat, J.D. Rothstein and F.H. Gage, Retrograde viral delivery of IGF-1 prolongs survival in a mouse ALS model. *Science*, **301**, 839-842 (2003).
- 49. I. Kazanis, E. Bozas, H. Philippidis and F. Stylianopoulou, Neuroprotective effects of insulin-like growth factor-I (IGF-I) following a penetrating brain injury in rats. *Brain Res*, **991**, 34-45 (2003).
- W. Kelsch, S. Hormuzdi, E. Straube, A. Lewen, H. Monyer and U. Misgeld, Insulin-like growth factor 1 and a cytosolic tyrosine kinase activate chloride outward transport during maturation of hippocampal neurons. *J Neurosci*, 21, 8339-8347 (2001).
- 51. G. Kempermann, Why new neurons? Possible functions for adult hippocampal neurogenesis. *J Neurosc*, 22, 635-638 (2002).
- E.C. Lai, K.J. Felice, B.W. Festoff, M.J. Gawel, D.F. Gelinas, R. Kratz, M.F. Murphy, H.M. Natter, F.H. Norris and S.A. Rudnicki, Effect of recombinant human insulin-like growth factor-I on progression of ALS. A placebo-controlled study. The North America ALS/IGF-I Study Group. *Neurology*, 49, 1621-1630 (1997).
- 53. W.H. Lee, S. Javedan and C.A. Bondy, Coordinate expression of insulin-like growth factor system components by neurons and neuroglia during retinal and cerebellar development. *J Neurosci*, **12**, 4737-4744 (1992).
- 54. W.H. Lee, K.M. Michels and C.A. Bondy, Localization of insulin-like growth factor binding protein-2 messenger RNA during postnatal brain development:

correlation with insulin-like growth factors I and II. *Neuroscience*, 53, 251-265 (1993).

- 55. K.R. Leslie, S.B. Nelson and G.G. Turrigiano, Postsynaptic depolarization scales quantal amplitude in cortical pyramidal neurons. *J Neurosci*, **21**, RC170 (2001).
- M. Lijffijt, P.S. van Dam, J.L. Kenemans, H.P. Koppeschaar, W.R. de Vries, M.L. Drent, A. Wittenberg and C. Kemner, Somatotropic-axis deficiency affects brain substrates of selective attention in childhood-onset growth hormone deficient patients. *Neurosci Lett*, 353, 123-126 (2003).
- J.C. Liou, F.Z. Tsai and S.Y. Ho, Potentiation of quantal secretion by insulin-like growth factor-1 at developing motoneurons in Xenopus cell culture. J Physiol, (2003).
- S.B. Lupien, E.J. Bluhm and D.N. Ishii, Systemic insulin-like growth factor-I administration prevents cognitive impairment in diabetic rats, and brain IGF regulates learning/memory in normal adult rats. J. Neurosci Res, 74, 512-523 (2003).
- K.B. Mackay, S.A. Loddick, G.S. Naeve, A.M. Vana, G.M. Verge and A.C. Foster, Neuroprotective Effects of Insulin-Like Growth Factor-Binding Protein Ligand Inhibitors in Vitro and in Vivo. J Cereb Blood Flow Metab, 23, 1160-1167 (2003).
- Y.H. Man, J.W. Lin, W.H. Ju, G. Ahmadian, L. Liu, L.E. Becker, M. Sheng, and Y.T. Wang, Regulation of AMPA receptor-mediated synaptic transmission by clathrin- dependent receptor internalization. *Neuron*, 25, 649-662 (2000).
- S. Manes, E. Mira, C. Gomez-Mouton, R.A. Lacalle, P. Keller, J.P. Labrador and A. Martinez, Membrane raft microdomains mediate front-rear polarity in migrating cells. *EMBO J*, **18**, 6211-6220 (1999).
- 62. A.L. Markowska, M. Mooney and W.E. Sonntag, Insulin-like growth factor-1 ameliorates age-related behavioral deficits. *Neuroscience*, **87**, 559-569 (1998).
- 63. P.A. McKelvie, K.M. Rosen, H.C. Kinney and L. Villa-Komaroff, Insulin-like growth factor II expression in the developing human brain. *J Neuropathol Exp Neurol*, **51**, 464-471 (1992).
- 64. J.D. Mitchell, J.H. Wokke and G.D. Borasio, Recombinant human insulin-like growth factor I (rhIGF-I) for amyotrophic lateral sclerosis/motor neuron disease (Cochrane Review). *Cochrane Database Syst Rev*, CD002064 (2002).
- P. Mitchell, Cancelled IGF-1 trials bode ill for diabetic patients. *Lancet*, **350**, 1606 (1997).
- W. Ni, K. Rajkumar, J.I. Nagy and L.J. Murphy, Impaired brain development and reduced astrocyte response to injury in transgenic mice expressing IGF binding protein-1. *Brain Res*, 769, 97-107 (1997).
- M.P. Nieto-Bona, L.M. Garcia-Segura and I. Torres-Aleman, Transynaptic modulation by insulin-like growth factor I of dendritic spines in Purkinje cells. *Int J Dev Neurosci*, 15, 749-754 (1997).
- T. Niikura, Y. Hashimoto, T. Okamoto, Y. Abe, T. Yasukawa, M. Kawasumi, T. Hiraki, Y. Kita, K. Terashita, K. Kouyama and I. Nishimoto, Insulin-like growth factor I (IGF-I) protects cells from apoptosis by Alzheimer's V642I mutant amyloid precursor protein through IGF-I receptor in an IGF-binding protein-sensitive manner. *J Neurosci*, 21, 1902-1910 (2001).
- 69. R.J. Nudo, G.W. Milliken, W.M. Jenkins and M.M. Merzenich, Use-dependent alterations of movement representations in primary motor cortex of adult squirrel monkeys. *J Neurosci*, **16**, 785-807 (1996).

- A. Nunez, E. Carro, and I. Torres-Aleman, Insulin-like growth factor I modifies electrophysiological properties of rat brain stem neurons. *J Neurophysiol*, 89, 3008-3017 (2003).
- S. Pons and I. Torres-Aleman, Insulin-like Growth Factor-I Stimulates Dephosphorylation of Ikappa B through the Serine Phosphatase Calcineurin (Protein Phosphatase 2B). J Biol Chem, 275, 38620-38625 (2000).
- 72. B.E. Pulford, L.R. Whalen and D.N. Ishii, Peripherally administered insulin-like growth factor-I preserves hindlimb reflex and spinal cord noradrenergic circuitry following a central nervous system lesion in rats. *Exp Neurol*, **159**, 114-123 (1999).
- K. Sakagami, D.M. Wu and D.G. Puro, Physiology of rat retinal pericytes: modulation of ion channel activity by serum-derived molecules. *J Physiol*, 521 Pt 3, 637-650 (1999).
- A. Savchenko, T.W. Kraft, E. Molokanova and R.H. Kramer, Growth factors regulate phototransduction in retinal rods by modulating cyclic nucleotide-gated channels through dephosphorylation of a specific tyrosine residue. *Proc Natl Acad Sci USA*, 98, 5880-5885 (2001).
- R.E. Schmidt, D.A. Dorsa, L.N. Beaudet, S.B. Plurad, C.A. Parvin and M.S. Miller, Insulin-like growth factor I reverses experimental diabetic autonomic neuropathy. *Am. J. Pathol.* 155, 1651-1660 (1999).
- D. Seto, W.H. Zheng, A. McNicoll, B. Collier, R. Quirion and S. Kar, Insulin-like growth factor-I inhibits endogenous acetylcholine release from the rat hippocampal formation: possible involvement of GABA in mediating the effects. *Neuroscience*, 115, 603-612 (2002).
- U. Sonnewald, A.Y. Wang, S.B. Petersen, N. Westergaard, A. Schousboe, R. Erikson and A. Skottner, 13C NMR study of IGF-I- and insulin-effects on mitochondrial function in cultured brain cells. *Neuroreport*, 6, 878-880 (1995).
- W.E. Sonntag, S.A. Bennett, A.S. Khan, P.L. Thornton, X. Xu, R.L. Ingram and J.K. Brunso-Bechtold, Age and insulin-like growth factor-1 modulate N-methyl-Daspartate receptor subtype expression in rats. *Brain Res Bull*, 51, 331-338 (2000).
- 79. T. Standal, M. Borset, S. Lenhoff, F. Wisloff, B. Stordal, A. Sundan, A. Waage and C. Seidel, Serum insulin-like growth factor is not elevated in patients with multiple myeloma but is still a prognostic factor, *Blood*, (2002).
- K.A. Sullivan and E.L. Feldman, Immunohistochemical localization of insulin-like growth factor-II (IGF-II) and IGF-binding protein-2 during development in the rat brain. *Endocrinology*, 135, 540-547 (1994).
- 81. H. Thoenen, Neurotrophins and neuronal plasticity. Science, 270, 593-598 (1995).
- H. Thoenen and M. Sendtner, Neurotrophins: from enthusiastic expectations through sobering experiences to rational therapeutic approaches. *Nat Neurosci*, 5 Suppl, 1046-1050 (2002).
- 83. I. Torres-Aleman. Serum growth factors and neuroprotective surveillance. *Mol Neurobiol*, **21**, 153-160 (2000).
- 84. I. Torres-Aleman and A.M. Fernandez, The role of insulin-like growth factors in human neurodegeneration. (Springer Publishing Co., 1998), 131-149.
- 85. J.L. Trejo, E. Carro, E. Garcia-Galloway and I. Torres-Aleman, Role of insulin-like growth factor I signaling in neurodegenerative diseases. *J Mol Med* (2003).
- P.S. van Dam, A. Aleman, W.R. de Vries, J.B. Deijen, E.A. van der Veen, E.H. de Haan and H.P. Koppeschaar, Growth hormone, insulin-like growth factor I and cognitive function in adults. *Growth Horm IGF Res*, **10 Suppl B**, S69-S73 (2000).
- 87. B. Voutsinos-Porche, G. Bonvento, K. Tanaka, P. Steiner, E. Welker, J.Y. Chatton, P.J. Magistretti and L. Pellerin, Glial glutamate transporters mediate a functional

metabolic crosstalk between neurons and astrocytes in the mouse developing cortex. *Neuron*, **37**, 275-286 (2003).

- Q. Wan, Z.G. Xiong, H.Y. Man, C.A. Ackerley, J. Braunton, W.Y. Lu, L.E. Becker, J.F. MacDonald, and Y.T. Wang, Recruitment of functional GABA(A) receptors to postsynaptic domains by insulin. *Nature*, 388, 686-690 (1997).
- K. Woodson, J.A. Tangrea, M. Pollak, T.D. Copeland, P.R. Taylor, J. Virtamo, and D. Albanes, Serum insulin-like growth factor I: tumor marker or etiologic factor? A prospective study of prostate cancer among Finnish men. *Cancer Res*, 63, 3991-3994 (2003).
- A. Yamaguchi, M. Tamatani, H. Matsuzaki, K. Namikawa, H. Kiyama, M.P. Vitek, N. Mitsuda, and M. Tohyama, Akt activation protects hippocampal neurons from apoptosis by inhibiting transcriptional activity of p53. *J Biol Chem*, 276, 5256-5264 (2001).
- 91. H.X. Zhuang, C.K. Snyder, S.F. Pu, and D.N. Ishii, Insulin-like growth factors reverse or arrest diabetic neuropathy: effects on hyperalgesia and impaired nerve regeneration in rats. *Exp Neurol*, **140**, 198-205 (1996).
- C. Zona, M.T. Ciotti, and P. Calissano, Human recombinant IGF-I induces the functional expression of AMPA/kainate receptors in cerebellar granule cells. *Neurosci Lett*, 186, 75-78 (1995).

# Chapter 11

# AGING AND LIFE SPAN

Holly M. Brown-Borg

Department of Pharmacology, Physiology and Therapeutics, University of North Dakota School of Medicine and Health Sciences 501 N. Columbia Rd., Grand Forks, ND, USA 58203

Key Words: Body size; dwarf; glucose regulation; longevity; stress resistance; therapy; transgenic

## **1. INTRODUCTION**

The factors that regulate aging processes are poorly understood. However, the closely related molecules, growth hormone (GH) and insulinlike growth factor-I (IGF-I), and corresponding downstream signaling factors have been shown to play a major role in the regulation of aging and life span. This mechanism appears to be well conserved evolutionarily, as evidence for altered signaling of these pathways extends the life span of yeast to mammals. Genetic models of aging have provided extensive data implicating a role for these hormones in both premature and delayed aging. These models are under active investigation to relate the GH/IGF-I/insulin pathway to the physiological changes associated with aging.

#### 2. GH DECLINES WITH AGE

It is well documented that growth hormone levels begin to decline soon after the peripubertal period of rapid growth<sup>1</sup>. The progressive decline of GH

secretion with age has been termed the "somatopause" in humans. Reduced GH secretion results from a loss of the high-amplitude release that occurs nocturnally. Further studies indicate that hypothalamic growth hormone releasing hormone (GHRH) and somatostatin may become dysregulated with age<sup>2,3</sup>. Decreased plasma IGF-I levels parallel the decline in GH pulses<sup>4</sup>. The lower levels of plasma GH and IGF-I in the elderly are thought to cause many aspects of physical change in aging including decreased muscle mass, strength, skin thickness and bone mineral density and increased fat mass and overall loss of energy. These changes parallel those observed in adult GH deficiency. GH replacement successfully alleviates the symptoms of adult GH deficiency. The beneficial effects of GH therapy in aged individuals have also been reported along with both observed and potential adverse side effects including arthralgia, carpal tunnel syndrome, edema, weight gain, myalgia, hyperglycemia, cancer and colonic polyps<sup>5,6</sup>. Therefore, GH therapy in the aged is controversial, at best.

#### 3. GH DEFICIENCY AND LIFE SPAN: MAMMALS

Endocrine mutants have provided tremendous insight into the role of hormones in aging (Table 1). Several types of GH/IGF spontaneous mutants and genetically engineered mice have been studied and include the Ames dwarf, Snell dwarf, GH receptor/binding protein knockout, Little, GH receptor antagonist, IGF receptor knockdown, and GH transgenic mice. Each of these mutants is described in the context of life span followed by specific information regarding possible mechanisms of delayed and premature aging.

Ames dwarf mice have a primary pituitary deficiency resulting in the absence of GH, prolactin (PRL), and thyrotropin (TSH). These deficiencies result from a point mutation in the *prop-1* gene<sup>7</sup> that promotes appropriate differentiation of the pituitary cells into somatotrophs, lactotrophs, and thyrotrophs. Consequent to GH deficiency, these mice have undetectable levels of plasma IGF-I<sup>8</sup>. Ames dwarf mice live 49-64% longer than their wild-type siblings (males and females, respectively)<sup>9</sup>. Ames mice are one of the longest living endocrine mutants and serve as an example of delayed aging. They are characterized by their diminutive body size (one third of normal) and reduced fertility.

Phenotype	Yeast ( <i>Sch9</i> , <i>RAS</i> ) <sup>32,154</sup>	Worms ( <i>daf2</i> , <i>age-1</i> ) <sup>33-35</sup>	Flies (Chico, INR) <sup>38,39</sup>	Mice ( <i>prop-1</i> , <i>pit-1</i> , GHR/BP KO) <sup>9,10,12</sup>	Mice GH over- expressing <sup>44,45,179</sup>
GH/IGF1/ Insulin signaling	¥	¥	¥	¥	<b>↑</b>
Body size	¥	¥	¥	$\mathbf{\Psi}$	♠
Reproduction	¥	Ψ	¥	¥	¥
Glucose metabolism	¥	Ψ	¥	¥	↑
Stress	↑	↑	↑	↑	¥
resistance					
Longevity	Υ	↑	♠	Ϋ́	<b>4</b>

Table 1. Phenotypic characteristics of mutant yeast, worms, flies and mice.

Snell dwarf mice are phenotypically identical to Ames mice lacking GH, PRL, and thyrotropin owing to lack of appropriate pituitary differentiation because of a mutation in the transcription factor *pit-1* (downstream of *prop-1*). These mice also have significantly reduced plasma IGF-I levels and live significantly longer (40–45%) than normal littermates<sup>10</sup>.

More evidence that reduced GH/IGF-I leads to life extension is derived from GH receptor/binding protein knockout (GHR/BP KO) mice<sup>11</sup>. These mice have high plasma GH but significantly reduced IGF-I levels because of the lack of receptor binding in target tissues. The GHR/BP KO mice are smaller than wild-type mice, exhibit delayed puberty, and the average increase in life span is  $31-46\%^{12,13}$ .

Mice with a spontaneous mutation in the GH releasing hormone (GHRH) receptor were first described by Eicher and Beemer<sup>14</sup> and termed "Little" mice<sup>15,16</sup>. These mice are deficient in GH and smaller in size, but exhibit normal body proportions, reduced fertility, and a 25% increase in life span compared to wild type controls when fed a low-fat diet<sup>10</sup>.

Mice engineered to overexpress a GH antagonist (mutated bovine GH) phenotypically resemble "Little" mice but do not exhibit extended life spans compared to their normal counterparts<sup>13,17</sup>. In addition, these mice have greatly reduced plasma IGF-I levels.

Attempts at creating IGF-I knockout mice have usually resulted in embryonic lethality<sup>18,19</sup>, although more recently Liu and LeRoith<sup>20</sup> reported a

42% postnatal survival rate in mice created to lack IGF-I gene expression. These mice are severely growth retarded both *in utero* and postnatally (35% at birth and 65% as adults). No life span data have been reported in these mice. However, an IGF-I receptor knockdown mouse was recently shown to live 26% longer than normal mice in this line<sup>21</sup>. These mice exhibit a 50% reduction in IGF-I receptor expression, slightly smaller body weights, and normal fertility. In addition, a liver-specific IGF-I gene deficient mouse (LID) was created by Yakar and coworkers<sup>22</sup> but no data regarding life span of these mice are available at this time. These mice develop and grow normally but exhibit altered carbohydrate metabolism<sup>22,23</sup>.

The evidence is overwhelming and clearly demonstrates that reduced GH/IGF-I signaling via reductions in plasma GH/IGF-I hormone levels or altered hormone receptor interactions extends life span in rodents. Table 2 summarizes some of the general clinical features of GH deficiencies and corresponding mutations. A human correlate of the mutation in Ames dwarf mouse was described by Krzisnik and coworkers<sup>24</sup> and showed that some members of this human "dwarf" population (prop-1 gene mutation) also outlived normal individuals from the same population. These individuals were not extremely long lived (>100) thus it is postulated that the potential detrimental effects of GH/IGF-I deprivation may not be pronounced or even realized in a 2 to 3 year-old-mouse living in a protected laboratory environment versus a 100-year-old human<sup>25</sup>. Other populations of prop-1 mutant people have been reported and showed no significant increases in life span over that of the normal population<sup>26-28</sup>. In addition, GHR/BP gene knockout mice are the mouse correlate of Laron dwarfism in humans. This syndrome is caused by a mutation in the GH receptor and resembles clinical GH deficiency. Individuals with Laron syndrome exhibit high plasma GH levels, very low circulating IGF-I concentrations, short stature, obesity, and impaired physical development<sup>29</sup>. Regarding mental status, the older literature suggests that impairment is common; however, Kranzler and coworkers found normal intelligence in a homogeneous population exhibiting GHR deficiency<sup>29,30</sup>. No differences in mortality are reported between those with Laron dwarfism and the normal population. Robust data on these populations are unlikely because so few individuals are available for study (~250 individual cases have been identified worldwide). GH deficiency in humans has been associated with increased fat mass, decreased muscle and bone mass, insulin resistance, and reduced life span<sup>31</sup>. However, some **GH-deficient** hypopituitary patients are also deficient in adrenocorticotropin (ACTH), likely accounting for the observed increase in mortality of this population. Clinical presentation is often non-specific and may include fatigue, lack of energy, poor concentration, memory loss in adults, and short stature in children. Human defects in IGF-I production are

Phenotype/	 BDAD1 <sup>#</sup>	POU1F1 (pit-	CHDHDp	CU19	CHDd	
Mutant Gene	PROFI	1) <sup>a</sup>	GHKHK	GHI	GHK	
GH/IGF1/Insulin	_	-		-	-	
signaling	$\mathbf{\Psi}$	$\mathbf{\Psi}$	$\mathbf{\Psi}$	$\mathbf{\Psi}$	$\mathbf{\Psi}$	
Growth failure/						
Short stature	+	+	+	+	+	
Body fat	increased	increased	increased	increased	increased	
Reproduction						
Neonate	Microphallus	Microphallus	N/A	N/A	Small genitalia/gonads	
Adolescent	Puberty delayed	Puberty delayed	Puberty delayed	Puberty delayed	Puberty delayed	
Adult	Infertility	Infertility	Normal fertility		Normal fertility	
Glucose metabolism						
Neonate	Hypoglycemia	Hypoglycemia	Normal	Hypoglycemia	Hypoglycemia/	
Adult	Fasting	Fasting	-	-	hyperinsulinema	
	symptoms	symptoms			Fasting symptoms	
Longevity	Possibly	?	Normal	?	Normal <sup>29</sup>	
					Damma 131	
	increased				Decreased	
	Decreased <sup>183</sup>					

extremely rare, most probably as a result of intrauterine lethality<sup>20</sup>. In addition, multiple examples in invertebrates support and extend that found in mammalian systems.

*Table 2.* Phenotypic characteristics of GH/IGF-I deficiency/insensitivity in humans.<sup>183</sup> <sup>a</sup>GH deficiency due to hypothalamic/pituitary dysfunction. <sup>b</sup>GH releasing hormone receptor – isolated GH deficiency.<sup>c</sup>Isolated GH deficiency. <sup>d</sup>GH insensitivity.

## **3.1 GH deficiency and life span: Invertebrates**

Inactivation of the pathways that promote growth extend life span in yeast, worms, and flies. In prokaryotes such as yeast, down-regulation of glucose signaling by mutations in *Sch9*, *RAS2*, or *CyR1/PKA* genes extend life span up to  $300\%^{32}$ . The pathway becomes more complex with the evolutionary progression to nematodes (*C. elegans*) and flies (*D. melanogaster*) and includes insulin for glucose regulation and IGF-I-like pathways (Table 3). The *daf-2* gene encodes an insulin/IGF receptor, and when disrupted, significantly extends life span in nematodes<sup>33</sup>. Disruption of downstream components of the *daf-2* signaling pathway including *age-1* [a homolog of the mammalian phosphoinositide-3-kinase (PI3K)] and *Akt/PKB* also leads to life span extension. Elegant genetic experiments in the nematode have yielded great insight regarding the life history of this organism and mechanisms such as this that evolved to protect the worm from periods of low food availability. Some *daf-2* and *age-1* mutants can

live 150% longer than normal worms and have normal fertility and development<sup>34-36</sup>. Interestingly, restoration of daf-2 signaling in neurons alone returned the life span of *daf-2* mutants to wild-type values while genetic rescue of muscle and intestinal daf-2 signaling had no effect<sup>37</sup>. These data suggest that the nervous system is involved in regulation of longevity in the nematode. In fruit flies, mutation of the gene encoding the insulin-like receptor (INR) or of the gene encoding the insulin receptor substrate (chico) increases life span<sup>38,39</sup>. This extension in life span is likely related to juvenile hormone (JH) deficiency (INR mutants lack JH). Juvenile hormone is a growth factor known to regulate larval development and reproduction, suppress responses to external stress, and influence insect adult longevity<sup>40,41</sup>. Functionally, JH exhibits some homology to thyroid hormones in mammals. Treatment of long-lived mutant female flies with an analog of JH shortens life span<sup>38,42</sup>. Each of these studies involving mutants in organisms ranging from yeast to mammals strongly implicates the GH/IGF-I/insulin signaling pathway with aging and life span.

	Yeast	Worms	Flies	Mammals
Ligands				Growth hormone
	Glucose ↓	Insulin/IGF-I-like ↓	Insulin/IGF-I-like Chico ↓	IGF-I ♥
Receptors	Gpr1 ↓	DAF-2 ♥	INR ♥	IGF-I receptor ↓
Downstream signaling molecules	Cyr1 (cAMP) Sch9 PKA ↓	AGE-1 Akt/PKB ↓	PI3K Akt/PKB ✔	PI3K Akt/PKB ↓
Physiological outcome	Stress resistance Growth	Stress resistance Growth	Stress resistance Growth	Stress resistance Growth
Reduced pathway signaling	Extended longevity	Extended longevity	Extended longevity	Extended longevity

Table 3. Common longevity signaling molecules in different species.

#### 4. GH EXCESS AND LIFE SPAN

In contrast to these deficiencies in GH and IGF-I that result in delayed aging, GH excess can hasten the process of aging. Several GH transgenic rodent models have been developed over the last two decades. The mice were created to overexpress a GH transgene (rat, bovine, ovine, human) resulting in high to supraphysiological levels of GH in the plasma. Life

expectancy in these mice is reduced to less than 50% of their normal counterparts<sup>43-45</sup>. In addition to a shortened life, these animals exhibit insulin resistance and hyperinsulinemia<sup>47-48</sup> (but are not diabetic) and show several signs of premature aging. Relatively young mice (6-8 months of age) begin to show physical signs of aging including weight loss, scoliosis, and coat deterioration. Further examination reveals significant evidence of premature aging in mice overexpressing GH including reduced replicative potential<sup>43</sup>, reduced antioxidative defense capacity<sup>49-51</sup>, increased free radical processes<sup>52</sup>, reduced reproductive function and life span<sup>53-55</sup>, increased astrogliosis<sup>56</sup>, deficits in learning and memory<sup>57</sup>, and increased plasma corticosterone<sup>56,58</sup>. The primary cause of death in these models of GH overexpression is related to significant kidney pathologies (glomerular nephritis and sclerosis)<sup>59,60</sup>. It is most likely the excess GH, not the detrimental effects of heterologous GH expression, that is causative in the premature aging and significant reduction in life span of GH transgenics animals. Supporting evidence for this assertion is observed in mice expressing a transgene for GHRH (MT-GHRH)<sup>61</sup>. The MT-GHRH mice exhibit enlarged pituitary glands, significant elevations in homologous (mouse) plasma GH levels and considerably shortened life spans<sup>62</sup>. A reduction of life span is also observed following antisense GH transgene expression in rats<sup>63</sup>. Moreover, overexpression of IGF-II in smooth muscle cells of mice shortens life span significantly, especially in males<sup>64</sup>. Highdose GH treatment in rats has also proved to be toxic.65 Although the majority of reports indicate that GH treatment decreases life span, there are data in mice administered low or moderate doses of GH that resulted in either slightly increased life span or no effect on longevity, respectively<sup>66,67</sup>. In the study by Khansari and Gustad<sup>66</sup>, human GH was administered to mice, leading to the question of specificity as human GH has both lactotropic and somatotropic activity in rodents.

Pathological GH excess occurs in humans with acromegaly. Pituitary adenomas can release excessive GH, resulting in significantly increased incidence of diabetes, cardiovascular disease, and tumors<sup>68-72</sup>. In the normal population, these same diseases increase in incidence with age. Patients with acromegaly are treated with somatostatin analogs (octreotide) that act to decrease plasma GH concentrations and subsequently reduce IGF-I levels, resulting in clinical improvement<sup>73</sup>. High levels of IGF-I have been associated with increased risk of several diseases in humans including breast, lung, colorectal, and prostate cancer<sup>74-78</sup>. Similarly, IGF promotes tumors in mice<sup>79-81</sup>. The antiapoptotic effects of IGF-I, as well as its mitogenic activity are likely related to these observations. High normal plasma GH levels have been identified as an early mortality risk factor<sup>82</sup>. In one study, treatment of critically ill elderly patients with GH significantly increased mortality<sup>83</sup>.

Moreover, a recent study demonstrated that children treated with GH experience statistically significant increases in incidence of cancer as adults<sup>84</sup>. Therefore, excessive GH/IGF-I is detrimental and can result in significant disease and reduced life span in mammals.

## 5. MECHANISMS OF LIFE EXTENSION: GROWTH AND BODY SIZE

It has been proposed that during evolution, the common IGF/insulin pathway diverged into two, one to regulate cell division and growth and the other to control metabolism and partitioning of energy resources<sup>35,85</sup>. In nematodes, flies and mammals, these metabolic pathways regulate energy, reproductive activity, and stress responses such that when food is abundant, growth, sexual maturation, and reproduction dominate. When food is scarce, resources are used to favor survival and directed away from growth and reproduction to increase stress resistance and repair processes leading to delayed aging and longevity.

Potential mechanisms of longevity causally related to GH and IGF-I include body size, metabolism, reproduction, and resistance to stress. Physiological factors affected by deficiencies or excesses in GH/IGF/insulin signaling are significant. Somatic growth is driven by components of this pathway; therefore, body size differences are obvious in many of these mutants. Those mutants with reduced signaling of this pathway are significantly smaller than wild-type controls. In worms, flies, and rodents, the range of differences is related to the degree of suppression of the IGF-I pathway, in particular. IGF-I receptor knockdown mice are slightly smaller (8% in males) than wild-type mice while Ames and Snell dwarf mice are 66% smaller (one third the size of wild-type). The GHR/BP knockout mice are 40% the size of wild type siblings as adults. Flies with disrupted insulin receptors or insulin receptor substrate genes are also considered dwarfs. Furthermore, the fact that small body size is strongly associated with longevity also holds true for domestic dogs<sup>86</sup> and likely for humans<sup>87</sup>. The relationship between small body size and longevity is further supported by reports showing that mice selected for reduced body sizes live longer<sup>88</sup> and calorie restriction also reduces growth and adult body size<sup>89</sup>. Therefore, growth negatively influences life span in mammals<sup>90</sup> and likely other species. Growth and oxidative metabolism are causally linked by signal transduction pathways<sup>90</sup>. Low levels of reactive oxygen species stimulate cell proliferation and signal transduction<sup>91-93</sup>.

In addition to reductions in body size and weight, Ames dwarf mice exhibit the expected declines in lean muscle mass. However, these mice also

266

have lower bone mineral densities and lower percent body fat compared to age-matched wild-type mice<sup>94</sup>. Less body fat may appear paradoxical in view of the deficiency of GH as GH is a lipolytic factor. The absence of GH should result in increased fat mass. However, the low insulin levels may counter this effect by decreasing storage of this fuel. Finally, the brain accounts for a greater proportion of body weight in Ames dwarf than in corresponding wild-type mice.

# 5.1 Mechanisms of life span extension: glucose regulation

Hormonal regulation of metabolic pathways is a key factor in the determinants of longevity in several organisms. The actions of GH on glucose regulation are well known. This hormone is a diabetogenic factor in that it opposes the actions of insulin. GH elevates plasma glucose concentrations by stimulating gluconeogenesis and glycogenolysis and inhibiting glucose uptake at the tissue level. Therefore, elevated GH levels such as those observed in acromegalic patients and genetically engineered mice with elevated plasma GH levels exhibit hyperinsulinemia. hyperglycemia, and/or insulin resistance<sup>48,95</sup>. More than 50% of humans with supraphysiological GH levels become diabetic and develop micro- and macrovascular complications associated with hyperglycemia. In sharp contrast, GH-deficient animals have low blood insulin, low glucose, and increased insulin sensitivity<sup>96-99</sup>. Administration of a single dose of insulin induced a larger decrease in blood glucose levels in Ames and GHR/BP KO mice than in corresponding wild-type mice<sup>100</sup>. However, disposing of glucose was reduced in both types of mice, indicating a decreased ability to increase insulin secretion in response to an acute glucose challenge. Two different mechanisms may be involved. GHR/BP KO mice have increased numbers of insulin receptors while Ames mice have higher levels of IRS-1 and IRS-2, two downstream effectors of the insulin receptor. This evidence is supported by decreased numbers of pancreatic islets in the Ames mouse<sup>101</sup>. Caloric restriction, long known to increase life span in rodents<sup>102-105</sup>, flies<sup>106</sup>, and likely primates<sup>107</sup>, also reduces insulin levels in the blood. Despite normal blood glucose concentrations and clearance, liver specific IGF-I deficient mice<sup>22</sup> are hyperinsulinemic and exhibit increased insulin resistance due to GH hypersecretion and  $\beta$ -cell hyperplasia. Therefore, these mice are not likely to exhibit life span extension, as signaling of this pathway is not truly reduced<sup>23,108</sup>. Mechanistically, reducing circulating glucose delays aging by decreasing the accumulation and detrimental processes associated with glycation end products<sup>109,110</sup> and reducing metabolism (less fuel) and associated metabolic ROS generation.

Insulin sensitivity declines with age<sup>111</sup>. In addition, insulin sensitivity is specifically related to visceral fat stores; sensitivity increases with reduction in visceral fat<sup>112,113</sup>. A case in point is the FIRKO mouse that lacks expression of the insulin receptor specifically in fat tissue. These insulin receptor knockout animals have significantly reduced fat mass, are protected against age-related obesity and decreased insulin sensitivity. The decrease in fat mass occurs in the absence of reduced food intake. In addition, these mice do not develop diabetes or glucose intolerance and, importantly, live 18% longer than wild-type control mice<sup>114,115</sup>. Although studies of centenarians have not yet shown statistical differences in IGF-I levels. greatly enhanced insulin sensitivity is strongly correlated with longevity in this unique population<sup>116,117</sup>. Moreover, polymorphic variants of the IGF-I receptor and PI3K genes were shown to affect plasma IGF-I levels and human longevity, suggesting that life span control is evolutionarily conserved<sup>118</sup>. Low serum IGF binding protein 2 concentrations have also been found to be a strong indicator of overall physical function in elderly men<sup>119</sup>.

## 5.2 Mechanisms of life span extension - reproduction

GH and IGF-I are known to effect development of numerous organ symptoms. Therefore, lifelong reduction of this hormone produces some secondary endocrine alterations. For example, GHR/BP KO mice have decreased thyroid hormones, increased glucocorticoids, and decreased insulin and glucose as previously mentioned. Their phenotype is similar to Ames and Snell dwarf mice that lack thyroid hormone and exhibit increased corticosteroids and decreased blood insulin and glucose concentrations<sup>95</sup>.

Reproductive competence is significantly affected by the GH/IGF signaling pathway. The GHR/BP KO mice exhibit delayed puberty and most animals are fertile. However, the Ames and Snell dwarf mice have significant delays in sexual maturation; the females are infertile while males are considered subfertile<sup>120</sup>. The IGF-I receptor knockdown mice exhibit normal puberty and fertility although the males have not been examined thoroughly. Concomitant with lower plasma IGF-I concentrations, calorie-restricted rodents exhibit delayed puberty, reduced litter size, and lower fecundity<sup>121-123</sup>. The effects of caloric restriction (CR) on reproductive function result from alterations in neuroendocrine factors including IGF-I. When CR is confined to adulthood, life span is also extended although not to the same extent as that in young mice<sup>104</sup>. Although CR has been documented to extend life span in yeast, worms, rotifers, flies, spiders, fish, rodents, and likely nonhuman primates, there are some species that are not affected<sup>105,124</sup>. Flies with mutations in the insulin-like receptor or in the insulin receptor

substrate (IRS) exhibit infertility or reduced fecundity<sup>38,39</sup>. However, IRS heterozygotes show evidence of increased longevity without reductions in fecundity or body size<sup>39</sup>. Juvenile hormone in insects is likely a secondary longevity factor downstream of insulin/IGF-I that regulates reproduction and growth as mentioned earlier<sup>125</sup>. In *C. elegans*, reproductive ability is coupled to metabolism such that when conditions are favorable for reproduction, metabolic demand is high. Nutritional signals act through insulin/IGF-I and stimulate production of nuclear hormone receptors that promote reproduction (such as daf-9)<sup>126,127</sup>. Ablation of nematode germ line derived cells results in life span extension of 60% suggesting a major hormonal influence on life span<sup>128</sup>. Germ line cells and the somatic gonad may exert antagonistic actions on aging as the removal of both did not affect life span.

Reproductive function of GH transgenic mice is altered by the high plasma GH concentrations. Female GH transgenic mice exhibit early puberty and increased ovulation rates; however, fertility is reduced and the reproductive life span is severely shortened when compared to wild-type mice (degree of suppression appears to depend on level of circulating GH) <sup>58,129,130</sup>. Male mice with elevated GH levels are fertile, but exhibit alterations in sexual behavior, reproductive pituitary hormone secretion, and a reduced reproductive life span<sup>57,130,131</sup>.

## 5.3 Mechanisms of life span extension: Stress resistance

Resistance to oxidative stress is another factor significantly affected by altered signaling of the GH/IGF pathway and closely regarded in the aging field as a major player in life span determination. The free radical theory of aging proposes that endogenously generated reactive oxygen species (ROS) cause aging via damage to DNA, proteins, and lipids<sup>132,133</sup>. The effects of GH and IGF-I on oxidative metabolism and oxidative damage have been documented in numerous reports. GH is an anabolic factor that increases cellular metabolism. Increased metabolic activity (glucose oxidation and oxygen consumption) leads to increased oxidative phosphorylation and increased production of ROS as byproducts of metabolism. Rollo and coworkers<sup>52</sup> showed that GH overexpression increased superoxide radicals and oxidative damage to membrane lipids (lipid peroxidation). Perhaps adaptively, tissues from mice with elevated plasma GH exhibit significantly reduced levels of antioxidative enzymes including manganese superoxide dismutase (MnSOD; dismutates superoxide anion into H<sub>2</sub>O<sub>2</sub> and O<sub>2</sub>), copperzinc SOD (similar to MnSOD), catalase, and glutathione peroxidase [both of which reduce hydrogen peroxide (H<sub>2</sub>O<sub>2</sub>) to H<sub>2</sub>O and O<sub>2</sub> in cytosol and mitochondria, respectively]<sup>49-51</sup>. In addition, direct effects of GH and IGF-I in vitro strongly support the in vivo data showing that these two hormones directly down-regulate catalase activity, glutathione peroxidase activity, and protein and MnSOD protein in hepatocytes<sup>134</sup>. Other proteins hormones tested (luteinizing hormone and prolactin) had little if any effect on antioxidant enzymes, indicating specificity of GH and IGF-I<sup>134</sup>.

In stark contrast to suppressed antioxidative defense induced by GH, there are several reports demonstrating enhanced defense capacity of this system in GH deficiency. Ames dwarf mice exhibit elevated catalase levels (activity, protein, and/or mRNA) in liver, kidney, heart, and hypothalamic tissues<sup>49,50,135</sup>. Glutathione peroxidase (GPX) and superoxide dismutase (SOD) activities are also elevated in GH-deficient dwarf mice (unpublished data)<sup>49</sup>. Muscle tissue GPX activity is preserved in dwarf mice at several ages following both acute and chronic exercise while that from wild-type mice declines with age<sup>136</sup>. Interestingly, GH replacement in dwarf mice down-regulates catalase, glutathione peroxidase, and MnSOD proteins and activities in both young and adult animals<sup>137</sup>. Nonenzymatic antioxidant molecules are also elevated in dwarf mice. Metallothionein and glutathione both exhibit ROS scavenging abilities, and levels of these are significantly increased in multiple tissues from the dwarf mouse<sup>138,139</sup>.

In general, circulating GH/IGF-I status is correlated with antioxidative capacity; high levels of plasma GH suppress this defense mechanism while the absence or low levels of GH/IGF-I enhance the ability of an organism to counter oxidants. The signal transduction pathways used by GH, IGF-I, and insulin overlap and influence redox-regulated transcription factors<sup>140</sup>. The PI3K pathway is defective in GH transgenic animals while animals with reduced GH/IGF-I signaling (dwarf and calorie-restricted mice) exhibit increased PI3K sensitivity<sup>120,141,142</sup>. Signaling via the PI3K pathway is significantly affected by free radicals. Insulin resistance in diabetes and obesity is improved by antioxidants<sup>143-145</sup>. This point is further illustrated in p66<sup>shc</sup> mice with disrupted IGF-I/insulin signaling. The p66<sup>shc</sup> protein is an adaptor protein that is crucial to IGF-I/insulin receptor signaling. Disruption of this gene results in mice with enhanced resistance to oxidants and extended life span when compared to wild-type control mice<sup>146</sup>.

Oxidative damage to DNA, proteins, and lipids can occur when reactive oxidants are not appropriately countered by compounds of the antioxidant system. Mitochondrial oxidant production (liver  $H_2O_2$ ) is significantly lower in dwarf mice, possibly indicating decreased metabolic activity in the absence of GH<sup>147</sup>. Consequent to reduced ROS and elevated antioxidants, dwarf mice exhibit lower DNA and protein oxidative damage<sup>147</sup>. Importantly, Sanz and coworkers<sup>148</sup> have also shown that mitochondrial DNA damage in brain and heart tissues is dramatically lower in Ames mice when compared to wild-type control mice. Functionally, these GH-deficient mice out-survive their GH sufficient counterparts following administration

of the systemic oxidative stressor paraquat<sup>120</sup>. The IGF-I receptor knockdown mice challenged with paraquat also lived significantly longer than mice with normal levels of IGF-I receptors (wild-type)<sup>21</sup>.

Long-living invertebrate IGF/insulin mutants also exhibit enhanced stress resistance. Catalase and SOD proteins and activity levels are up-regulated in the *daf-2* and *age-1* mutant nematodes<sup>149-151</sup>. In addition, these investigators showed that these mutants consistently express resistance to oxygen,  $H_2O_2$ and paraquat<sup>150-152</sup>. Moreover, one of four *C. elegans* SOD genes (SOD3) is directly regulated by the IGF/insulin signaling pathway<sup>153</sup>. Reduced signaling of this growth and metabolic pathway in Drosophila enhances oxidative stress resistance by increasing the expression of SOD<sup>38,39</sup>. Genes that extend life span in yeast confer resistance to oxidative stress and heat shock<sup>32,154</sup>. Reduced GH/IGF/insulin signaling is also associated with increased resistance to other stressors including heat shock, UV, and gamma irradiation. Overall, the reported studies strongly support the notion that GH/IGF-I/insulin signaling pathways are intimately involved in the modulation of oxidative stress. The suppressive effect of GH on multiple components of the antioxidant system and consequent oxidative damage may be one mechanistic reason that levels of this hormone decline with aging.

GH-deficient dwarf mice and rats resist cancer development following administration of chemical carcinogens<sup>155-157</sup> and exhibit reduced growth of transplanted tumors<sup>158</sup>. Spontaneous tumor incidence appears to be delayed and the severity reduced in the hypopituitary dwarf mice<sup>10,159</sup>. Moreover, tumor growth in IGF-I deficient mice is reduced relative to control mice<sup>160</sup>. Cancer incidence is also lessened in calorie-restricted rodents.<sup>104</sup> In addition, GH/IGF-I deficient mice develop significantly less osteoarthritis than wild-type mice<sup>161</sup>. It has been postulated that stress resistance is coordinately increased (heat shock proteins, antioxidants, detoxification systems, metal chelators, and repair systems) and this up regulation results in multi-stress resistance to different stressors<sup>90,162,163</sup>. Recent work by Murakami and coworkers<sup>164</sup> exemplifies this idea well in showing that fibroblasts from the skin of long-living Snell dwarf mice are resistant to multiple forms of cellular stress including heat, paraquat, H<sub>2</sub>O<sub>2</sub>, UV light, and the toxic metal cadmium.

#### 6. GH THERAPY IN THE ELDERLY

GH therapy to reduce or reverse aging is becoming quite popular. A highly cited study in favor of this treatment was conducted by Rudman and coworkers<sup>5</sup>. In this study, elderly men with very low levels of IGF-I were treated for several months with recombinant human GH, resulting in

increased muscle mass, decreased fat mass, and increased bone density at one site (1.6%; lumbar vertebral). Since this time, other studies have confirmed these findings but also suggest that the benefits are neither clear nor consistent<sup>6</sup> as no parallel functional improvements (strength, endurance, mood, and mental status) were observed in this study as well as in others<sup>165,166</sup>. GH administration for anti-aging purposes in humans thus far has been confined to short-term studies using relatively small numbers of individuals. Despite the lack of consensus regarding the beneficial or detrimental effects of this somatotropic hormone, GH is being marketed as the "fountain of youth". GH therapy in healthy elderly individuals is controversial, and extreme caution should be exercised owing to unproven effects<sup>6,166,167</sup> potentially dangerous side and the enhanced and responsiveness this population of form of hormone to this supplementation<sup>168</sup>.

GH therapy has been reported to enhance cognitive function in elderly patients<sup>169,170</sup>. Animal studies provide support for these observations showing that GH improves brain vascularity, blood flow, and cognitive function<sup>171,172</sup>. IGF-I has also been observed to protect and stimulate the central nervous system in animal experiments<sup>173-175</sup>. However, an improvement in memory following GH treatment in young but not old rats possibly reflects a down-regulation of GH receptors in the old animals<sup>176</sup>. In contrast, there is also compelling evidence in animals suggesting that GH deficiency is associated with maintenance of cognitive function. Ames dwarf mice do not exhibit the age-related decline in cognitive function (including memory) and behavior that is observed in wild type control mice<sup>177</sup>. Similarly, GHR/BP knockout mice show no decline in cognitive function when age-matched wild-type littermates are significantly impaired<sup>178</sup>, suggesting that the absence of GH action enhances memory retention. Other than hypophysectomy, which decreases brain IGF-I mRNA, direct evidence of brain dysfunction in GH/IGF-I-deficient animal models is lacking<sup>180</sup>. Correlative data are available and show that deficits in the IGF-I axis occur in aged brain and that IGF-I levels are associated with cognitive function<sup>181,182</sup>. Experiments teasing out the effects of age versus the effects of low GH or IGF-I levels have yet to be done.

## 7. CONCLUSIONS

The natural age-related decline in plasma GH levels and the concomitant decrease in IGF-I that occurs in mammals is likely a protective mechanism to decrease metabolic activity and cellular division. High levels of either of these hormones throughout life could play a contributory role in pathological changes associated with aging such as increased oxidative damage and cancer. Decreasing IGF-I and insulin signaling without a concomitant decline (or increase) in GH concentrations may provide an optimal balance of hormones that would maintain muscle and bone mass yet enhance stress resistance and prevent many of the physical signs of aging. In addition, this type of treatment may result in decreased incidence of age-related diseases such as cancer, diabetes and cardiovascular disease. Reduced activity of the somatotropic axis is key in the quest to slow or delay aging and promote life extension.

## REFERENCES

- 1. E. Corpas, S.M. Harman and M.R. Blackman, Human growth hormone and human aging. *Endocrine Rev*, **14**, 20-39 (1993).
- 2. C.V. DeGennaro, M. Zoli, D. Cocchi, A. Maggi, P. Marrama and L.F. Agnati, Reduced growth hormone releasing factor (GHRF)-like immunoreactivity and GHRF gene expression in the hypothalamus of aged rats. *Peptides* **10**, 705-708 (1989).
- 3. C.V. Degennaro, V. Colonna, F. Fidone, D. Cocchi and E.E. Muller, Feedback effects of growth hormone on growth hormone-releasing hormone and somatostatin are not evident in aged rats. *Neurobiol Aging*, **14**, 503-507 (1993).
- 4. D. Rudman, M.H. Kutner, C.M. Rogers, M.F. Lubin, G.A. Fleming and R.P. Bain, Impaired growth hormone secretion in the adult population: relation to age and adiposity. *J Clin Invest* **67**, 1361-1369 (1981).
- D. Rudman, A.G. Felle, H.S. Nagraj, G.A. Gergans, P.Y. Lalitha, A.F. Goldberg, R.A. Schenkler, L. Cohn, I.W. Rudman and D.E. Mattson. Effects of human growth hormone in men over 60 years old. *New Engl J Med*, 323, 1-6 (1990).
- M.A. Papadakis, D. Grady, D. Black, L. Wells and C. Grunfeld, Growth hormone replacement in healthy older men improves body composition but not functional ability. *Ann Intern Med*, **124**, 708-716 (1996).
- M.W. Sornson, W. Wu, J.S. Dasen, S.E. Flynn, D.J. Norman, S.M. O'Connell, I. Gukovsky, C. Carriere, A.K. Ryan and A.P.Miller, Pituitary lineage determination by the prophet of pit-1 homeodomain factor defective in Ames dwarfism. *Nature*, 384, 327-333 (1996).
- 8. V. Chandrashekar and A. Bartke, Induction of endogenous insulin-like growth factor-I secretion alters the hypothalamic-pituitary-testicular function in growth hormone-deficient adult dwarf mice. *Biol Repro*, **48**, 544-551 (1993).
- 9. H.M. Brown-Borg, K.E. Borg, C.J. Meliska and A. Bartke, Dwarf mice and the aging process. *Nature*, **384**, 33 (1996).
- 10. K. Flurkey, J. Papconstantinou, R.A. Miller and D.A. Harrison, Lifespan extension and delayed immune and collagen aging in mutant mice with defects in growth hormone production. *Proc Natl Acad Sci*, **98**, 6736-6741 (2001).
- Y. Zhou, B.C. Xu, H.G. Maheshwari, L. He, M. Reed, M. Lozykowski, S. Okada, T.E. Wagner, L.A. Cataldo, K. Coschigano, G. Baumann and J.J. Kopchick, A mammalian model of Laron syndrome produced by targeted disruption of the mouse growth hormone receptor/binding protein gene (The Laron mouse). *Proc Natl Acad Sci* 94, 13215-13220 (1997).

- K.T. Coschigano, D. Clemmons, L.L. Bellush and J.J. Kopchick, Assessment of growth parameters and life span of GHR/BP gene-disrupted mice. *Endocrinology*, 141, 2608-2613 (2000).
- A. Bartke, V. Chandrashekar, F. Dominici, D. Turyn, B. Kinney, R. Steger and J.J. Kopchick, Insulin-like growth factor-1 (IGF-1) and aging: controversies and new insights. *Biogerontology*, 4, 1-8 (2003).
- 14. E.M. Eicher and W.G. Beamer, Inherited ateliotic dwarfism in mice. Characteristics of the mutation, little, on chromosome 6. *J Hered* **67**, 87-91 (1976).
- P. Godfrey, J. Rahal, W. Beamer, N.G. Copeland, N.A. Jenkins and K.E. Mayo, GHRH receptor of little mice contains a missense mutation in the extracellular domain that disrupts receptor function. *Nat Genet*, **199**, 227-232 (1993).
- S.C. Lin, C.R. Lin, I. Gukovsky, D.M. Simmons, L.W. Swanson and M.G. Rosenfeld, Molecular basis of the little mouse phenotype and implications for cell type specific growth. *Nature*, 364, 208-213 (1993).
- W.Y. Chen, D.C. Wight, B.V. Metha, T.E. Wagner and J.J. Kopchick, Glycine 119 of bovine growth hormone is critical for growth-promoting activity. *Mol Endocrinol*, 5, 1845-1852 (1991).
- J.P. Liu, J. Baker, A.S. Perkins, E.J. Robertson and A. Efstratiadis, Mice carrying null mutations of the genes encoding insulin-like growth factor 1 (Igf-1) and type 1 IGF receptor (Igf1r). *Cell*, **75**, 59-72 (1993).
- L. Powell-Braxton, P. Hollingshead, C. Warburton, M. Dowd, S. Pitts-Meek, D. Dalton, H. Gillett and T.A. Stewart, IGF-1 is required for normal embryonic growth in mice. *Genes Dev*, 7, 2609-2617 (1996).
- 20. J.L. Liu and D. LeRoith, Insulin-like growth factor 1 is essential for postnatal growth in response to growth hormone. *Endocrinology*, **140**, 5178-5184 (2001).
- M. Holzenberger, J. Dupont, B. Ducos, P. Leneuve, A. Geloen, P.C. Even, P. Cervera and Y. Le Bouc, IGF-1 receptor regulates lifespan and resistance to oxidative stress in mice. *Nature*, 421, 182-187 (2003).
- S. Yakar, J.L. Liu, B. Stannard, A. Butler, D. Accili, B. Sauer and D. LeRoith, Normal growth and development in the absence of hepatic insulin-like growth factor I. Proc Natl Acad Sci USA, 96, 7324-7329 (1999).
- S. Yakar, J.L. Liu, A.M. Fernandez, Y. Wu, A.V. Schally, J. Frystyk, S.D. Chernausek, W. Mejia and D. LeRoith, Liver-specific igf-1 gene deletion leads to muscle insulin insensitivity. *Diabetes*, 50, 1110-1118 (2001).
- 24. C. Krzisnik, Z. Kolacio, T. Battelino, M. Brown, J.S. Parks and Z. Laron, The 'little people' of the Island of Krk revisited. Etiology of hypopituitarism revealed. *J Endo Genet*, **1**, 9-19 (1999).
- 25. D.J. Davenport, Power to the people. Sci Aging Knowledge Environ, 50, ns8 (2003).
- 26. R.G. McArthur, K. Morgan, J.A. Phillips, 3<sup>rd</sup>, M. Bala and J. Klassen, The natural history of familial hypopituitarism. *Am J Med Genet*, **22**, 553-566 (1985).
- 27. A.L. Rosenbloom, J. Guevara-Aguirre, R.G. Rosenfeld and U. Francke, Growth hormone receptor deficiency in Ecuador. *J Clin Endocrinol Metab*, **84**, 4436-4443 (1999).
- J.A. Barreto-Filho, M.R. Alcantara, R. Salvatori, M.A. Barreto, A.C. Sousa, V. Bastos, A.H. Souza, R.M. Pereira, P.E. Clayton, M.S. Gill and M.H. Aguiar-Oliveira, Familial isolated growth hormone deficiency is associated with increase systolic blood pressure, central obesity and dyslipidemia. *J Clin Endocrinol Metab*, 87, 2018-2023 (2002).
- 29. Z. Laron, Natural history of the classical form of primary growth hormone (GH) resistance (Laron syndrome). *J Pediatr Endocrinol Metab*, **12**, 231-249 (1999).

- J.H. Kranzler, A.L. Rosenbloom, V. Martinez and J. Guevara-Aguirre, Normal intelligence with severe insulin-like growth factor I deficiency due to growth hormone receptor deficiency: a controlled study in a genetically homogeneous population. J Clin Endocrinol Metab, 83, 1953-1958 (1998).
- B.A. Bengtsson, G. Johannsson, S.M. Shalet, H. Simpson and P.H. Sonken, Treatment of growth hormone deficiency in adults. J Clin Endocrinol Metab, 85, 933-942 (2000).
- 32. P. Fabrizio, F. Pozza, S.D. Pletcher, C.M. Gendron and V.D. Longo, Regulation of longevity and stress resistance by Sch9 in yeast. *Science*, **292**, 288-290 (2001).
- K.D. Kimura, H.A. Tissenbaum, Y. Liu and G. Ruvkun, daf-2, an insulin receptorlike gene that regulates longevity and diapause in Caenorhabditis elegans. *Science*, 277, 942-946 (1997).
- D. Gems, A.J. Sutton, M.L. Sundermeyer, P.S. Albert, K.V. King, M.L. Edgley, P.L. Larsen and D.L. Riddle, Two pleiotropic classes of daf-2 mutation affect larval arrest, adult behavior, reproduction and longevity in Caenorhabditis elegans. *Genetics*, 150, 120-155 (1998).
- 35. L. Guarente and C. Kenyon, Genetic pathways that regulate ageing in model organisms. *Nature*, **408**, 255-262 (2000).
- A. Dillin, A.L. Hsu, N. Arantes-Oliveira, J. Lehrer-Graiwer, H. Hsin, A.G. Fraser, R.S. Kamath, J. Ahringer and C. Kenyon, Rates of behavior and aging specified by mitochondrial function during development. *Science*, 298, 2398-2401(2000).
- 37. C.A. Wolkow, K.D.Kimura, M.-S. Lee and G. Ruvkun, Regulation of C. elegans life-span by insulinlike signaling in the nervous system. *Science*, **290**, 147-150 (2000).
- M. Tatar, A. Kopelman, D. Epstein, M.P. Tu, C.M. Yin and R.S. Garofalo, A mutant drosophila insulin receptor homolog that extends life-span and impairs neuroendocrine function. *Science*, 292, 107-110 (2001).
- D.J. Clancy, D. Gems, L.G. Harshman, S. Oldham, H. Stocker, E. Hafen, S.J. Leevers and L. Partridge, Extension of life-span by loss of CHICO, a *Drosophila* insulin receptor substrate protein. *Science*, 292, 104-106 (2001).
- 40. M. Tatar, N. Priest and S. Chien, Negligible senescence during reproductive diapause in *D. melanogaster. Am Nat*, **158**, 248-258 (2001).
- 41. W.S. Herman and M. Tatar, Juvenile hormone regulation of longevity in the migratory monarch butterfly. *Proc R Soc Lond*, **268**, 2509-2514 (2001).
- 42. H. Dingle and R. Winchell, Juvenile hormone as a mediator of plasticity in insect life history. *Arch Insect Biochem Physiol*, **35**, 359-373 (1997).
- 43. W.R. Pendergrass, Y. Li, D. Jiang and N.S. Wolf, Decrease in cellular replicative potential in giant mice transfected with the bovine growth hormone gene correlates to shortened life span. *J Cell Physiol*, **156**, 96-103 (1993).
- 44. E. Wolf, E. Kahnt, J. Ehrlein, W. Hermanns, G. Brem and R. Wanke, Effects of longterm elevated serum levels of growth hormone on life expectancy of mice: lessons from transgenic animals models. *Mech Ageing Dev*, **68**, 71-87 (1993).
- 45. M. Cecim, A. Bartke, Y.S. Yun and T.E. Wagner, Expression of human, but not bovine growth hormone genes promotes development of mammary tumors in transgenic mice. *Transgenics*, **1**, 431-437 (1994).
- L.J. Kajiura and C.D. Rollo, A mass budget for transgenic "supermice" engineered with extra rat growth hormone genes: evidence for energetic limitation. *Can J Zool*, 72, 1010-1017 (1994).

- 47. A. Balbis, J.M. Dellacha, R.S. Calandra, A. Bartke and D. Turyn, Down regulation of masked and unmasked insulin receptors in the liver of transgenic mice expressing bovine growth hormone gene. *Life Sci*, **51**, 771-778 (1992).
- A. Balbis, A. Bartke and D. Turyn, Overexpression of bovine growth hormone in transgenic mice is associated with changes in hepatic insulin receptors and in their kinase activity. *Life Sci*, 59, 1362-1371 (1996).
- 49. H.M. Brown-Borg, A.M. Bode and A. Bartke, Antioxidative mechanisms and plasma growth hormone levels: Potential relationship in the aging process. *Endocrine*, **11**, 41-48 (1999).
- 50. H.M. Brown-Borg and S.G Rakoczy, Catalase expression in delayed and premature aging mouse models. *Exp Gerontol*, **35**, 199-212 (2000).
- 51. S. Hauck and A. Bartke, Free radical defenses in the liver and kidney of human growth hormone transgenic mice: possible mechanisms of early mortality. *J Gerontol Bio Sci*, **56A**, B153-162 (2001).
- 52. C.D. Rollo, J. Carlson and M. Sawada, Accelerated aging of giant transgenic mice is associated with elevated free radical processes. *Can J Zool*, **74**, 606-620 (1996).
- M. Cecim, C. Fadden, J. Kerr, R. Steger and A. Bartke, Infertility in transgenic mice overexpressing the bovine growth hormone gene: disruption of the neuroendocrine control of prolactin secretion during pregnancy. *Biol Repro*, 52, 1187-1192 (1995).
- A. Bartke, A.G. Amador, C.J. Meliska, J.S. Yun and T.E. Wagner, Neuroendocrine function, sexual behavior and fertility in transgenic mice expressing growth hormone genes. In: *Current Advances in Andrology*, G.M.H. Waites, J. Frick and G.W.H. Baker, eds., Proceedings of the VIth International Congress of Andrology, Salzburg, pp. 295-301 (1997).
- 55. C.D. Rollo, J. Rintoul and L.J. Kajiura, Lifetime reproduction of giant transgenic mice: the energetic stress paradigm. *Can J Zool*, **78**, 1336-1345 (1997).
- 56. D.B. Miller, A. Bartke and J.P. O'Callaghan, Increased glial fibrillary acidic protein (GFAP) levels in the brains of transgenic mice expressing the bovine growth hormone (bGH) gene. *Exp Gerontol*, **30**, 383-400 (1995).
- 57. C.J. Meliska, P.A. Burke, A. Bartke and R.A. Jensen, Inhibitory avoidance and appetitive learning in aged normal mice: comparison with transgenic mice having elevated plasma growth hormone levels. *Neurobiol Learn Mem*, **68**, 1-12 (1997).
- M. Cecim, P.K. Ghosh, A.I. Esquifino, T. Began, T.E. Wagner, J.S. Yun and A. Bartke, Elevated corticosterone levels in transgenic mice expressing human or bovine growth hormone genes. *Neuroendocrinology*, 53, 313-316 (1991).
- 59. R. Wanke, E. Wolf, W. Hermanns, S. Folger, T. Buchmuller and G. Brem, The GHtransgenic mouse as an experimental model for growth research: clinical and pathological studies. *Horm Res*, **37**, 74-87 (1992).
- C.W. Yang, L.J. Striker, C. Pesce, W.Y. Chen, E.P. Paten, S. Elliot, T. Doi, J.J. Kopchick and G.E. Striker, Glomerulosclerosis and body growth are mediated by different portions of bovine growth hormone: studies in transgenic mice. *Lab Invest*, 68, 62-70 (1993).
- 61. K.E. Mayo, R.E. Hammer, L.W. Swanson, R.L. Brinster, M.G. Rosenfeld and R.M. Evans, Dramatic pituitary hyperplasia in transgenic mice expressing a human growth hormone releasing factor gene. *Mol Endocrinol*, **2**, 606-612 (1988).
- A. Bartke, H.M. Brown-Borg, A.M. Bode, J. Carlson, W. S. Hunter and R.T. Bronson, Does growth hormone prevent or accelerate aging? *Exp Gerontol*, 33, 675-687 (1998).
- 63. I. Shimokawa, Y. Higami, M. Utsuyama, T. Tuchiya, T. Komatsu, T. Chiba and H. Yamaza, Life span extension by reduction in growth hormone-insulin-like growth factor-1 axis in a transgenic rat model. *Am J Pathology*, **160**, 2259-2265 (2002).

276

- S. Zaina, L. Pettersson, A.B. Thomsen, C.-M. Chai, Z. Qi, J. Thyberg and J. Nilsson, Shortened life span, bradycardia, and hypotension in mice with targeted expression of an Igf2 transgene in smooth muscle cells. *Endocrinology*, 144, 2695-2703 (2003).
- 65. M.D. Groesbeck, A.F. Parlow and W.H. Daughaday, Stimulation of supranormal growth in pubertal, adult plateaued, and hypophysectomized female rats by large doses of rat growth hormone: physiological effects and adverse consequences. *Endocrinology*, **120**, 1963-1975 (1987).
- 66. D.N. Khansari and T. Gustad, Effects of long-term, low-dose growth hormone therapy on immune function and life expectancy of mice. *Mech Ageing Dev*, **57**, 87-100 (1991).
- D.N. Kalu, P.B. Orhii, C. Chen, D.Y. Lee, G.B. Hubbard, S. Lee and Y. Olatunjibello, Aged-rodent models of long-term growth hormone therapy lack of deleterious effect on longevity. *J Gerontol Biol Sci Med Sci*, 53, B452-B463 (1998).
- L. Sacca, A. Cittadini and S. Fazio, Growth hormone and the heart. Endocrine Rev, 15, 555-573 (1994).
- A. Colao, A. Balzano, D. Ferone, N. Panza, G. Grande, P. Marzullo, A. Bove, G. Iodice, B. Merola and G. Lombardi, Increased prevalence of colonic polyps and altered lymphocyte subset pattern in the colonic lamina propria in acromegaly. *Clin Endocrinol*, 47, 23-28 (1997).
- P.J. Jenkins, P.D. Fairclough, T. Richards, D.G. Lowe, J. Monson, A. Grossman, J.A. Wass and M. Besser, Acromegaly, colonic polyps and carcinoma. *Clin Endocrinol*, 47, 17-22 (1997).
- 71. S.M. Orme, R.H. McNally, R.A. Cartwright and P.E. Belchetz, Mortality and cancer incidence in acromegaly: a retrospective cohort study. United Kingdom Acromegaly Study Group. *J Clin Endocrinol Metab*, **70**, (1998).
- 72. B.A. Bengtsson, S. Eden, I. Ernest, A. Oden and B. Sjogren, Epidemiology and long-term survival in acromegaly. *Acta Med Scand*, **223**, 327-335 (2000).
- 73. R.D. Utiger, Treatment of acromegaly, New Engl J Med, 342, 1210-1211 (2000).
- 74. H. Yu and T. Rohan, Role of the insulin-like growth factor family in cancer development and progression. J Natl Cancer Inst, 92, 1472-1489 (2000).
- J.M. Chan, M.J. Stampfer, E. Giovannucci, P.H. Gann, J. Ma, P. Wilkinson, C.H. Hennekens and M. Pollak, Plasma insulin-like growth factor-1 and prostate cancer risk: a prospective study. *Science*, 279, 563-566 (1998).
- A. Wolk, C.S. Mantzoros, S.O. Andersson, R. Bergstrom, L.B. Signorello, P. Lagiou, H.O. Adami and D. Trichopoulos, Insulin-like growth factor 1 and prostate cancer risk: a population-based, case-control study. *J Natl Cancer Inst*, **90**, 911-915 (1998).
- S.E. Hankinson, W.C. Willett, G.A. Colditz, D.J. Hunter, D.S. Michaud, B. Deroo, B. Rosner, F.E. Speizer and M. Pollak, Circulating concentrations of IGF-1 and the risk of breast cancer. *Lancet*, 351, 1593-1596 (1998).
- J. Ma, M. Pollack, E. Giovanucci, J.M. Chan, X. Tao, C. Hennekens and M.J. Stampfer, A prospective study of plasma levels of IGF-1, IGFBP-3, and colon cancer risk among men. *J Natl Cancer Inst*, **91**, 620-625 (1999).
- 79. S.T. Ng, J. Zhou, O.O. Adesanya, J. Wang, D. LeRoith and C.A. Bondy, Growth hormone induces mammary gland hyperplasia in aging primates. *Nat Med*, **3**, 1141-1144 (1997).
- C.E. Rogler, D. Yang, L. Rossetti, J. Donohoe, E. Alt, C.J. Chang, R. Rosenfeld, K. Neely and R. Hintz, Altered body composition and increased frequency of diverse malignancies in insulin-like growth factor-II transgenic mice. *J Biol Chem* 269, 13779-13784 (1994).
- 81. A.S. Bates, A.J. Evans, P. Jones and R.N. Clayton, Assessment of GH status in acromegaly using serum growth hormone, serum insulin-like growth factor-1 and urinary growth hormone excretion. *Clin Endocrinol (Oxf)*, **42**, 417-423 (1995).

- P. Maison, B. Balkau, D. Simon, P. Chanson, G. Rosselin and E. Eschwege, Growth hormone as a risk for premature mortality in healthy subjects: data from the Paris prospective study. *British Med J*, **316**, 1132-1133 (1998).
- 83. J. Takala, E. Ruokonen, N.R. Webster, M.S. Nielsen, D.F. Zandstra, G. Vundelinckx and C.J. Hinds, Increased mortality associated with growth hormone treatment in critically ill adults. *N Eng J Med*, **341**, 785-792 (1999).
- 84. A.J. Swerdlow, C.D. Higgins, P. Adlard and M.A. Preece, Risk of cancer in patients treated with human pituitary growth hormone in the UK, 1959-85: a cohort study. *Lancet*, **360**, 273-277 (2002).
- 85. C.E. Finch and G. Ruvkun, The genetics of aging. Ann Rev Genom Hum Genet, 2, 435-462 (2001).
- 86. G.J. Patronek, D.J. Waters and L.T. Glickman, Comparative longevity of pet dogs and humans: Implications for gerontology research. *J Gerontol*, **52A**, B171-178 (1997).
- 87. T.T. Samaras, H. Elrick and L.H. Storms, Is height related to longevity? *Life Sci*, **72**, 1781-1802 (2003).
- 88. R.C. Roberts, The lifetime growth and reproduction of selected strains of mice. *Heredity*, **16**, 369-381 (1961).
- P.H. Duffy,R.J. Feuers and R.W. Hart, Effect of chronic caloric restriction on the circadian regulation of physiological and behavioral variable in old male B6C3F1 mice. Chronobiol Int 7:291-303 (1990).
- 90. C.D. Rollo, Growth negatively impacts life span of mammals. *Evol Develop*, **55**, 55-61 (2002).
- 91. R.H. Burdon, Superoxide and hydrogen peroxide in relation to mammalian cell proliferation. *Free Radic Biol Med*, **18**, 775-794 (1995).
- 92. H.M. Lander, An essential role for free radicals and derived species in signal transduction. *FASEB J*, **11**,118-124 (1997).
- 93. G.N. Rao, K.A. Katki, N.R. Madamanchi, Y. Wu and M.J. Birrer, JunB forms the majority of AP-1 complex and is a target for redox regulation by receptor tyrosine kinase and G protein-coupled receptor agonist in smooth muscle cells. *J Biol Chem*, 274, 6003-6010 (1999).
- 94. M.L. Heiman, F.C. Tinsley, J.A. Mattison, S. Hauck and A. Bartke, Body composition of prolactin-, growth hormone, and thyrotropin-deficient Ames dwarf mice. *Endocrine*, **20**, 149-154 (2003).
- 95. F.P. Dominici, A. Balbis, A. Bartke and D. Turyn, Role of hyperinsulinemia on hepatic insulin binding and insulin receptor autophosphorylation in the presence of high growth hormone (GH) levels in transgenic mice expressing GH gene. J Endocrinol, 159, 1-25 (1998).
- 96. K.E. Borg, H.M. Brown-Borg and A. Bartke, Assessment of the primary adrenal cortical and pancreatic hormone basal levels in relation to plasma glucose and age in the unstressed Ames dwarf mouse. *Proc Soc Biol Med*, **210**, 126-133 (1995).
- F.P. Dominici, G. Arostegui Diaz, A. Bartke, J.J. Kopchick and D. Turyn, Compensatory alterations of insulin signal transduction in liver of growth hormone receptor knockout mice. *J Endocrinol*, 166, 579-590 (2000).
- F.P. Dominici, D.P. Argentino, A. Bartke and D. Turyn, The dwarf mutation decreases high dose insulin responses in skeletal muscle, the opposite of effects in liver. *Mech Ageing Dev*, **124**, 819-827 (2003).
- S. Hauck, W.S. Hunter, N. Danilovich, J.J. Kopchick, A. Bartke, Reduced levels of thyroid hormones, insulin and glucose, and lower body core temperature in the growth hormone receptor/binding protein knockout mouse. *Exp Biol Med*, 226, 552-558 (2001).

- 100. K.T. Coschigano, M.E. Riders, LL. Bellush and J.J. Kopchick, Glucose metabolism in growth hormone receptor/binding protein gene disrupted mice. *Endocrine Soc Ann Mtg*, San Diego, CA, abstract P3-537, p. 553 (1999).
- 101. J.A. Parsons, A. Bartke and R.L. Sorenson, Number and size of islets of Langerhans in pregnant, human growth hormone-expressing transgenic, and pituitary dwarf mice: effect of lactogenic hormones. *Endocrinology*, **136**, 2013-2021 (1995).
- 102. C.M. McCay, M. Crowell and L. Maynard, The effect of retarded growth upon the length of life and upon ultimate size. *J Nutrition*, **10**, 63-79 (1935).
- 103. C.M. McCay, G. Sperling, and L.L. Barnes, Growth, aging, chronic diseases and lifespan in rats, *Arch. Biochem. Biophys.* **2**:469-479 (1947).
- R. Weindruch and R.L. Walford, Dietary restriction in mice beginning at 1 year of age: effect on life-span and spontaneous cancer incidence. *Science*, 215, 1415-1418 (1982).
- 105. R. Weindruch and R.L. Walford, The Retardation of Aging and Disease by Dietary Restriction, Charles C. Thomas, Springfield, IL (1988).
- 106. T. Chapman and L. Partridge, Female fitness in Drosophila melanogaster: an interaction between the effect of nutrition and of encounter rate with males. *Proc R Soc Lond Ser B*, **263**, 755-759 (1996).
- 107. M.A. Lane, D.J. Baer, W.V. Rumpler, R. Weindruch, D.K. Ingram, E.M. Tilmont, R.G. Cutler and G.S. Roth, Calorie restriction lowers body temperature in rhesus monkeys, consistent with a postulated anti-aging mechanism in rodents. *Proc Natl Acad Sci USA*, **93**, 4159-4164 (1996).
- 108. R. Yu, S. Yakar, Y.L. Liu, Y. Lu, D. LeRoith, D. Miao and J.L. Liu, Liver-specific IGF-1 gene deficient mice exhibit accelerated diabetes in response to streptozotocin, associate with early onset of insulin resistance. *Mol Cell Endocrinol*, 204, 31-42 (2003).
- K.M. Reiser, Nonenzymatic glycation of collagen in aging and diabetes. Soc Exp Biol Med Minireview, pp. 23-34 (1998).
- 110. J.W. Baynes and V.M. Monnier, The Maillard Reaction in Aging, Diabetes and Nutrition. Alan R. Liss, New York (1989).
- 111. F.S. Facchini, N. Hua, F. Abbasi and G.M. Reaven, Insulin resistance as a predictor of age-related diseases. *J Clin Endocrinol Metab*, **86**, 3574-3578 (2001).
- N. Barzilai, S. Banerjee, M. Hawkins, W. Chen and L. Rossetti, Caloric restriction reverses hepatic insulin resistance in aging rats by decreasing visceral fat. J Clin Invest, 201, 1353-1361 (1998).
- 113. N. Barzilai and G. Gupta, Revisiting the role of fat mass in the life extension induced by caloric restriction. *J Gerontol Biol Sci*, **54**, B89-B96 (1999).
- M. Bluher, M.D Michael, O.D. Peroni, K. Ueki, N. Carter, B.B. Kahn and C.R. Kahn, Adipose tissue selective insulin receptor knockout protects against obesity and obesity-related glucose intolerance. *Dev Cell*, 3, 25-38 (2002).
- 115. M. Bluher, B.B. Kahn and C.R. Kahn, Extended longevity in mice lacking the insulin receptor in adipose tissue. *Science* **299**, 572-574 (2003).
- G. Paolisso, A. Gambardella, S. Ammendola, A. D'Amore, V. Balbi, M. Varricchio and F. D'Onofrio, Glucose tolerance and insulin action in healthy centenarians. *Am J Physiol*, 270, E890-E894 (1996).
- 117. G. Paolisso, M. Barbieri, M. R. Rizzo, C. Carella, M. Rotondi, M. Bonafe, C. Franceschi, G. Rose and G. De Benedictis, Low insulin resistance and preserved beta-cell function contribute to human longevity but are not associated with TH-INS genes. *Exp Gerontol*, 37, 149-156 (2001).
- 118. M. Bonafe, M. Barbieri, F. Marchegiani, F. Olivieri, E. Ragno, C. Giampieri, E. Mugianesi, M. Centurelli, C. Franceschi, and G. Paolisso, Polymorphic variants of insulin-like growth factor I (IGF-I) receptor and phosphoinositide 3-kinase genes

affect IGF-I plasma levels and human longevity: cues for an evolutionarily conserved mechanism of life span control. *J Clin Endocrinol Metab*, **88**, 3299-3304 (2003).

- 119. A.W. van den Beld, W.F. Blum, H.A. Pols, D.E. Grobbee and S.W. Lamberts, Serum insulin-like growth factor binding protein-2 levels as an indicator of functional ability in elderly men. *Eur J Endocrinol*, **148**, 627-634 (2003).
- 120. A. Bartke, Delayed aging in Ames dwarf mice. Relationships to endocrine function and body size. In: *Results and Problems in Cell Differentiation: The Molecular Genetics of Aging*, S. Hekimi, ed, Vol 29. Springer-Verlag, Berlin, pp. 181-202 (2000).
- 121. R. Weindruch and S. S. Sohal, Caloric intake and aging. *N Engl J Med*, **337**, 986-994 (1997).
- 122. J. Meites, Anti-aging interventions and their neuroendocrine aspects in mammals, In: *Neurobiology and Neuroendocrinology of Ageing*, R. Falvo, A. Bartke, E. Giacobini, eds., Cambridge, J Reprod Fert, pp. 1-9 (1993).
- 123. A.M. Holehan and B.J. Merry, The control of puberty in dietary restricted female rat. *Mech Ageing Dev*, **32**, 179-191 (1985).
- 124. J.R. Carey, P. Liedo, L. Harshan, Y. Zhang, H.G. Muller, L. Partridge and J.L. Wang, Life history response of Mediterranean fruit flies to dietary restriction. Aging Cell, 1, 140-148 (2002).
- 125. M. Tatar, A. Bartke and A. Antebi, The endocrine regulation of aging by insulin-like signals. *Science*, **299**, 1346-1351 (2003).
- 126. B. Gerisch, C. Weitzel, C. Kober-Eisermann, V. Rottiers and A. Antebi, A hormonal signaling pathway influencing C. elegans metabolism, reproductive development. and life span. *Dev Cell*, **1**, 841-851 (2001).
- 127. K. Jia, P.S. Albert and D.L. Riddle, DAF-9, a cytochrome P450 regulating C. elegans larval development and adult longevity. *Development*, **129**, 221-231 (2002).
- 128. H. Hsin and C. Kenyon, Signals from the reproductive system regulate the lifespan of C. elegans. *Nature*, **399**, 362-366 (1999).
- 129. E.M. Naar, A. Bartke, S.S. Majumdar, F.C. Buonomo, J.S. Yun and T.E. Wagner, Fertility of transgenic female mice expressing bovine growth hormone or human growth hormone variant genes. *Biol Reprod*, **45**, 178-187 (1991).
- A. Bartke, M. Cecim, K. Tang, R.W. Steger, V. Chandrashekar and D. Turyn, Neuroendocrine and reproductive consequences of overexpression of growth hormone in transgenic mice. *Proc Soc Exp Biol Med*, 206, 345-359 (1994).
- 131. V. Chandrashekar, A. Bartke and T.E. Wagner, Endogenous human growth hormone (GH) modulates the effect of gonadotropin-releasing hormone on pituitary function and the gonadotropin response to the negative feedback effect of testosterone in adult transgenic mice bearing human GH gene. *Endocrinology*, 123, 2717-2722 (1988).
- 132. D. Harman, Aging. A theory based on free radical and radiation chemistry. J Gerontology, 11, 298-300 (1956).
- 133. D. Harman, Free radicals in aging. *Mol Cell Biochem*, 84, 155-161 (1988).
- 134. H.M. Brown-Borg, S.G. Rakoczy, M.A. Romanick and M.A. Kennedy, Effects of growth hormone and insulin like growth factor-1 on hepatocyte antioxidative enzymes. *Exp Biol Med*, **227**, 94-104 (2002).
- 135. S. Hauck and A. Bartke, Effects of growth hormone on hypothalamic catalase and CuZn superoxide dismutase. *Free Rad Biol Med*, **28**, 970-978 (2000).
- M.A. Romanick, S.G. Rakoczy and H.M. Brown-Borg, Long-lived Ames dwarf mouse exhibits increased antioxidant defense in skeletal muscle. *Mech Ageing Dev*, 125, 269-81 (2004).
- 137. H.M. Brown-Borg and S.G. Rakoczy, Growth hormone administration to long-living dwarf mice alters multiple components of the antioxidative defense system. *Mech Ageing Dev*, **124**, 1013-1024 (2003).

- 138. M.M. Meyer, J.C. Swinscoe, H.M. Brown-Borg and E.C. Carlson, Increased glomerular metallothionein accompanies reduced glomerular basement membrane thickening in the Ames dwarf mouse model of delayed aging, Experimental Biology, San Diego, CA, Abstract # 217.1 (2003).
- 139. H.M. Brown-Borg, S.G. Rakoczy, M.A. Kennedy and M.A. Romanick, Relationship between plasma growth hormone, antioxidants and oxidative damage in premature and delayed aging mice. 83<sup>rd</sup> Annual Meeting of the Endocrine Society, p. 237 (2001).
- 140. M. Meyer, R. Schreck, J.M. Muller and P.A. Baeuerle, Redox control of gene expression by eukaryotic transcription factors NF-6B, AP-1, and SRF/TCF. In: *Oxidative Stress, Cell Activation and Viral Infection*, C. Pasquier, R. Y. Olivier, C. Auclair and L. Packer, Birkhauser Verlag, Basel, pp. 217-235 (1994).
- F.P. Dominici, D. Cifone, A. Bartke and D. Turyn, Alterations in the early steps of the insulin-signaling system in skeletal muscle of GH-transgenic mice. *Am J Physiol*, 277, E447-454 (1999).
- 142. T. Parr, Insulin exposure and unifying aging. Gerontology, 45, 121-135 (1999).
- 143. A.S. Blair, E. Hajduch, G.J. Litherland and H.S. Hundal, Regulation of glucose transport and glycogen synthesis in L6 muscle cells during oxidative stress. *J Biol Chem*, 274, 36293-36299 (1999).
- 144. L.L. Hansen, Y. Ikeda, G.S. Olsen, A.K. Busch and L. Mosthaf, Insulin signaling is inhibited by micromolar concentrations of  $H_2O_2$ . *J Biol Chem*, **274**, 36293-36299 (1999).
- 145. A. Tirosh, R. Potashnik, N. Bashan and A. Rudich, Oxidative stress disrupts insulininduced cellular redistribution of insulin receptor sustrate-1 and phosphatidylinositol 3-kinase in 3T3-L1 adipocytes. *J Biol Chem*, **274**, 10595-10602 (1999).
- 146. E. Migliaccio, M. Giorgio, S. Mele, G. Pelicci, P. Reboldi, P.P. Pandolfi, L. Lanfrancone and P.G. Pelicci, The p66<sup>shc</sup> adaptor protein controls oxidative stress response and life span in mammals. *Nature*, **402**, 309-313 (1999).
- 147. H.M. Brown-Borg, W.T. Johnson, S.G. Rakoczy, M.A. Kennedy and M.A. Romanick, Mitochondrial oxidant production and oxidative damage in Ames dwarf mice. J Am Aging Assoc, 24, 85-96 (2001).
- 148. A. Sanz, A. Bartke and G. Barja, Long-lived Ames dwarf mice: Oxidative damage to mitochondrial DNA in heart and brain. *J Amer Aging Assoc*, **25**, 119-122 (2002).
- J.R. Vanfleteren, Oxidative stress and ageing in *Caenorhabditis elegans*. Biochec J, 292, 605-608 (1993).
- 150. P.L. Larsen, Aging and resistance to oxidative damage in *Caenorhabditis elegans*. Proc Natl Acad Sci USA, **90**, 8905-8909 (1993).
- 151. J.R. Vanfleteren and A. De Vreese, The gerontogenes *age-1* and *daf-2* determine metabolic rate potential in aging *Caenorhabditis elegans*. *FASEB J*, **9**, 1355-1361 (1995).
- 152. H. Adachi, Y. Fujiwara and N. Ishii, Effects of oxygen on protein carbonyl and aging in *Caenorhabditis elegans* mutants with long (age-1) and short (mev-1) life spans. J *Gerontol Biol Sci*, **53**, B240-B244 (1998).
- 153. Y. Honda and S. Honda, The *daf-2* gene network for longevity regulates oxidative stress resistance and MN-superoxide dismutase gene expression in *Caenorhabditis elegans. FASEB J*, **13**, 1385-1393 (1999).
- 154. P. Fabrizio, L.L. Liou, V.N. Moy, A. Diaspro, J. Selverstone Valentine, E.B. Gralla and V.D. Longo, SOD2 functions downstream of Sch9 to extend longevity in yeast. *Genetics*, **163**, 35-46 (2003).
- F. Bielschowsky and M. Bielschowsky, Carcinogenesis in the pituitary of dwarf mouse. The response to dimethylbenzanthracene applied to the skin. Br J Cancer, 15, 257-262 (1961).

- 156. M.M. Ramsey, R.L. Ingram, A.B. Cashion, A.H. Ng, J.M. Cline, A.F. Parlow and W.E. Sonntag, Growth hormone-deficient dwarf animals are resistant to dimethylbenzanthracine (DMBA)-induced mammary carcinogenesis. *Endocrinology*, **143**, 4139-4142 (2002).
- 157. H.W. Chen, H. Meier, H.-J. Heininger and R.J. Huebner, Tumorigenesis in strain DW/J mice and induction by prolactin of the group-specific antigen of endogenous C-type RNA tumor virus. *J Natl Cancer Inst*, **49**, 1145-1154 (1972).
- 158. E.G. Rennels, D.M. Anigstein and L. Anigstein, A cumulative study of the growth of sarcoma 180 in anterior pituitary dwarf mice. *Tex Rep Biol Med*, 23, 776-781 (1965).
- Y. Ikeno, R.T. Bronson, G.B. Hubbard, S. Lee, A. Bartke, Delayed occurrence of fatal neoplastic diseases in Ames dwarf mice: correlation to extended longevity. J Gerontol, 58A, 291-296 (2003).
- 160. Yang, 1996,
- 161. R. Silberberg, Articular aging and osteoarthritis in dwarf mice. *Pathologia et Microbiologia*, **38**, 417-430 (1972).
- 162. S.M. Jazwinski, Longevity, genes and aging. Science, 273, 54 (1996).
- 163. G.M. Martin, S.N. Austad and T.E. Johnson, Genetic analysis of ageing: role of oxidative damage and environmental stresses. *Nat Genetic*, **13**, 25-34 (1996,).
- 164. S. Murakami, A. Salmon and R. A. Miller, Multiplex stress resistance in cells from long-lived dwarf mice. *FASEB J*, **17**, 1565-1566 (2003).
- 165. K.H. Lange, J.L. Andersen, N. Beyer, F. Isaksson, B. Larsson, M.H. Rasmussen, A. Juul, J. Bulow and M. Kjaer, GH administration changes myosin heavy chain isoforms in skeletal muscle but does not augment muscle strength or hypertrophy, either alone or combined with resistance exercise training in healthy elderly men. J Clin Endocrinol Metab, 87, 513-523 (2002).
- 166. M.R. Blackman, J.D. Sorkin, T. Munzer, M.F. Bellantoni, J. Busby-Whitehead, T.E. Stevens, J. Jayme, K.G. O'Conner, C. Christmas, J.D. Tobin, K.J Stewart, E. Cottrell, C. St. Clair, K.M. Pabst and S.M. Harman, Growth hormone and sex steroid administration in health aged women and men. J Am Med Assoc, 288, 2282-2292 ().
- 167. K. von Werder, The somatopause is no indication for growth hormone therapy. J Endocrinol Invest, 22(Suppl 5), 137-141 (1999).
- A.A. Toogood and S.M. Shalet, Growth hormone replacement therapy in the elderly with hypothalamic-pituitary disease: a dose-finding study. J *Clin Endocrinol Metab*, 84, B1-B6 (1999).
- 169. A.R. Hoffman, S.A. Lieberman and G.P. Ceda, Growth hormone therapy in the elderly: implications for the aging brain. *Psychoneuroendocrinology*, **17**, 327-333 (1992).
- 170. A. Aleman, H.J.J. de Vries, E.H.F. de Haan, H.J.J. Verhaar, M.M. Samson and H.P. Koppeschaar, Age-sensitive cognitive function, growth hormone and insulin-like growth factor 1 plasma levels in healthy older men. *Neuropsychobiology*, **41**, 73-78 (2000).
- 171. W.E. Sonntag, C. Lynch, P. Thornton, A. Kahn, S. Bennett and R. Ingram, The effects of growth hormone and IGF-1 deficiency on cerebrovascular and brain aging. *J Anat*, **197**, 575-585 (2000).
- A.L. Markowska, M. Mooney and W.E. Sonntag, Insulin-like growth factor-1 (IGF-1) ameliorates age-related behavioral deficit. *Neuroscience*, 87, 559-569 (1998).
- 173. L.M. Garcia-Segura, G.P. Cardona-Gomez, J.A. Chowen and I. Azcoitia, Insulin-like growth factor-1 receptors and estrogen receptors interact in the promotion of neuronal survival and neuroprotection. *J Neurocytol*, 29, 425-437 (2000).
- 174. L.M. Frago, C. Paneda, S.K. Dickson, A.K. Hewson, J. Argente and J.A. Chowen, Growth hormone (GH) and GH-releasing peptide-6 increase brain insulin-like growth factor-1 expression and activate intracellular signaling pathways involved in neuroprotection, *Endocrinology*, **143**, 4113-4122 (2002).

282

- 175. E. Carro, J.L. Trejo, S. Busiguina and I. Torres-Aleman, Circulating insulin-like growth factor I mediates the protective effects of physical exercise against brain insults of different etiology and anatomy. *J Neurosci*, **21**, 5678-5684 (2001).
- 176. S. Schneider-Rivas, S. Rivas-Arancibia, F. Vazquez-Pereyra, R. Vazquez-Sandoval and G. Borgonio-Perez, Modulation of long-term memory and extinction responses induced by growth hormone (GH) and growth hormone releasing hormone (GHRH) in rats. *Life Sci*, **56**, PL433-441 (1995).
- 177. B.A. Kinney, C.J. Meliska, R.W. Steger and A. Bartke, Evidence that Ames dwarf mice age differently from their normal siblings in behavioral and learning and memory parameters. *Horm Behav*, **39**, 277-285 (2001).
- 178. B.A. Kinney, K.T. Coschigano, J.J. Kopchick, R.W. Steger and A. Bartke, Evidence that age-induced decline in memory retention is delayed in growth hormone resistant GH-R-KO (Laron) mice. *Physiol Behav*, **72**, 653-660 (2001).
- R.W. Steger, A. Bartke and M. Cecim, Premature ageing in transgenic mice expressing different growth hormone genes. J Reprod Fertil, 46(Supple), 61-75 (1993).
- M.A. Hynes, J.J. van Wyk, P.J. Brooks, A.J. D'Ercole, M. Jansen and P.K. Lund, Growth hormone dependence of somatomedin-c/insulin-like growth factor I and insulin-like growth factor II messenger ribonucleic acids. *Mol Endocrinol*, 1, 233-242 (1987).
- W.E. Sonntag, J.K. Brunso-Bechtold and D.R. Riddle, Age-related decreases in growth hormone and insulin-like growth factor (IGF-1): Implications for brain aging. 4, 311-329 (2001).
- P.S. van Dam and A. Aleman, Insulin-like growth factor-I, cognition and brain aging. *Eur J Pharmacol*, 490, 87-95 (2004).
- 183. E.O. Reiter and R.G. Rosenfeld, Normal and aberrant growth, In: *Williams Textbook* of *Endocrinology*, P.R. Larsen, H.M. Kronenberg, S. Melmed and K.S. Polonsky, eds., Tenth Edition, Saunders, Philadelphia, pp. 1003-1114 (2003).

# Chapter 12

# INTERACTIONS OF INSULIN-LIKE GROWTH FACTOR-I AND ESTROGEN IN THE BRAIN

Pablo Mendez<sup>1</sup>, Gloria Patricia Cardona-Gomez<sup>2</sup> and Luis Miguel Garcia-Segura<sup>1</sup> <sup>1</sup>Instituto Cajal, C.S.I.C., E-28002 Madrid, Spain; <sup>2</sup>Group of Neuroscience, School of Medicine, University of Antioquia, PO Box 1226 Medellin, Colombia

Key words: Estradiol; gonadotrophins; IGF-I; neuronal development; neuroprotection; synaptic plasticity.

## **1. INTRODUCTION**

Insulin-like growth factor exerts different effects on the development and plasticity of the nervous system and is a potent neuroprotectant (see Chapters 8 and 10). As in other organs, the actions of IGF-I in the nervous system may be affected by the interaction with other molecules. In the brain, these include neurotransmitters, neuromodulators, hormones and other growth factors. For instance, growth hormone regulates IGF-I expression in the brain in an anatomically specific manner<sup>1</sup> and leukemia inhibitory factor has been shown to regulate IGF-I levels in peripheral nerves<sup>2</sup>. The cross-talk between IGF-I intracellular signaling with the signaling pathways of other factors may also play an important role in the actions of IGF-I in the nervous system. This is the case of the interaction of IGF-I and nerve growth factor (NGF) in peripheral nerves, where the phosphatidylinositol-3 kinase (PI3K)-Akt-glycogen synthase kinase 3 (GSK3) pathway seems to underlie the synergism of these two factors on axonal growth<sup>3</sup>. Neurotransmitters, such as glutamate, may affect the actions of IGF-I in the brain by interfering with IGF-I receptor signaling. Recently it has been shown that glutamate at high concentrations induces loss of sensitivity to IGF-I by phosphorylating the IGF-I receptor docking protein insulin-receptor-substrate (IRS)-1 in Ser(307) through a pathway involving activation of protein kinase (PK) A and PKC<sup>4</sup>.

One of the best characterized factors that affect IGF-I's actions in the nervous system is the ovarian hormone estradiol. Evidence has accumulated over the past years indicating a close interdependence between the actions of IGF-I and estradiol in the brain. Both factors have in common the duality of being hormones, as well as locally produced neuromodulators, and they exert similar pleiotropic actions in the developing and adult brain. During the development of the nervous system, IGF-I and estradiol promote the differentiation and survival of specific neuronal populations. In the adult brain, IGF-I and estradiol act as neuromodulators, regulate synaptic plasticity, are involved in the response of neural tissue to injury, regulate neurogenesis, and protect neurons against different neurodegenerative stimuli<sup>5</sup>.

IGF-I and its receptors are expressed at high levels by neurons and glia in the developing rodent brain. However, the expression of IGF-I is dramatically reduced in many brain areas, excluding the hypothalamus, by the time of puberty. In contrast, expression of the IGF-I receptor is maintained in neurons and glia in the adult brain<sup>6</sup>, suggesting that under physiological conditions the actions of peripheral IGF-I are predominant in adult life<sup>7,8</sup>.

Estradiol is mainly produced in the ovary and reaches the brain from the circulation. In addition, estradiol is formed in the brain from androgen precursors present in plasma and probably also from androgen precursors locally synthesized from cholesterol. Local production of estradiol is important in the developing brain, as inferred from the high level of expression of aromatase, the enzyme that converts testosterone into estradiol<sup>9</sup>. In the adult rodent brain, the level of aromatase expression is substantially reduced, suggesting a decrease in the local formation of estradiol. Interestingly, both IGF-I and estradiol synthesis are induced in damaged brain regions following brain injury, suggesting a possible role for the local production of these factors in brain recovery<sup>10,11</sup>.

In this chapter we focus on the functional implications and the molecular mechanisms involved in the interaction of IGF-I and estradiol in the brain.
# 2. INTERACTION OF IGF-I AND ESTROGEN IN NEUROENDOCRINE AND REPRODUCTIVE EVENTS IN THE BRAIN

The nervous system is a target for the ovarian hormone estradiol. This hormone regulates brain development and function, acting on neurons, synapses, and glial cells<sup>12</sup>. For many years, the actions of estradiol in the brain were thought to be restricted to areas involved in neuroendocrine regulation and the control of sexual behavior. Today it is evident that estradiol exerts a broad spectrum of actions in many brain areas that are not directly related to reproduction<sup>12-15</sup>.

One of the main mechanisms of action of estradiol in the brain, as in other organs, is the activation of nuclear estrogen receptors. These receptors belong to the steroid/thyroid nuclear receptor superfamily, and once activated they homo- or heterodimerize, interact with DNA, and recruit a cohort of transcriptional cofactors to the regulatory regions of target genes. The two mammalian estrogen receptors cloned to date, denominated  $\alpha$  and  $\beta$ , are widely distributed throughout the central nervous system (Table 1). In addition, estradiol exerts rapid membrane effects on neural cells, modulating ion channels, neurotransmitter transporters, intracellular calcium levels, and phosphorylation of different kinases<sup>5,12-15</sup>.

As an endocrine signal, IGF-I represents a link between the growth and reproductive axes, and the interaction between IGF-I and estradiol in the brain is of particular physiological relevance for the regulation of growth, sexual maturation, and adult neuroendocrine function. IGF-I regulates gonadotrophins by acting at the level of both the hypothalamus and pituitary<sup>16-18</sup>. IGF-I from peripheral origin reaches the rat hypothalamus, where it appears to be a signal related with the initiation of puberty<sup>17</sup>. Coincident with the onset of puberty, IGF-I mRNA and protein levels increase dramatically in the rat mediobasal hypothalamus and median eminence<sup>19,20</sup>. Decreased IGF-I plasma levels in growth hormone deficient mice, as well as in growth hormone receptor deficient mice, are associated with reproductive deficits, a delay in female puberty, and alterations in the functioning of the hypothalamic-pituitary-gonadal axis. In addition, deletion of insulin receptor substrate-2 (IRS-2), a component of the insulin/IGF-I receptor signaling cascade, causes infertility in females<sup>19</sup>.

Interactions between estrogen and IGF-I have been reported in brain areas involved in neuroendocrine control. One such area is the hypothalamic arcuate nucleus, a key center for neuroendocrine regulation of the growth and reproductive axes. This nucleus shows estrogen-induced and estrous cycle-related plastic changes in synaptic connectivity and glial-neuronal interactions in female rodents<sup>20</sup>. During the preovulatory and ovulatory phases of the estrous cycle, there is a transient disconnection of axo-somatic  $\gamma$ -aminobutyric acid-(GABA)ergic synapses on the arcuate somas of adult female rats. This synaptic remodeling is induced by estradiol and is linked to plastic modifications in astroglial processes<sup>20</sup>.

During the estrous cycle, estradiol increases IGF-I accumulation in the arcuate nucleus<sup>21</sup>. IGF-I levels increase in the afternoon of proestrus and remain high in the morning of estrus. This effect may be in mediated in part by the increased expression of insulin-like growth factor binding protein-2 by hypothalamic tanycytes, specialized glial cells that express IGF-I receptors and accumulate IGF-I from the cerebrospinal fluid<sup>22</sup>. Furthermore, results from in vivo experiments using intracerebroventricular infusion of specific receptor antagonists have shown that both estrogen receptors and IGF-I receptors are involved in the induction of synaptic and glial plastic modifications in the arcuate nucleus that take place during the estrous cycle and are linked to luteinizing hormone release<sup>23</sup>. Under normal conditions, the number of synaptic inputs on arcuate neuronal somas of cycling female rats decreases between the morning of proestrus and the morning of estrus. This phasic synaptic and glial remodeling is blocked when the selective and specific IGF-I receptor antagonist JB1 is administered in the lateral cerebral ventricle to neutralize the local actions of IGF-I. In contrast, JB1 did not affect the number of synapses in proestrus rats, suggesting that IGF-I receptor activation is necessary for the estrogen-dependent synaptic plastic changes, but not for the normal maintenance of synaptic inputs<sup>24</sup>.

Arcuate neurons express IGF-I receptors<sup>6</sup> and may, therefore, be a direct target for the actions of JB1. This antagonist may affect pre- and/or postsynaptic mechanisms, since ultrastructural studies have shown that IGF-I receptors are present both in axo-somatic presynaptic terminals, as well as in neuronal somas of the rat arcuate nucleus<sup>6</sup>. Arcuate astrocytes are another possible cellular target for the IGF-I receptor antagonist, since they also express IGF-I receptors<sup>6</sup>. Arcuate astrocytes appear to be directly involved in the regulation of synaptic inputs to arcuate neurons<sup>20</sup>. Synaptic disconnection of arcuate neurons in estrous females is accompanied by increased expression of glial fibrillary acidic protein (GFAP) and by an increase in the ensheathment of neuronal surfaces by astrocytic processes<sup>20</sup>. Interestingly, studies on hypothalamic tissue fragments from ovariectomized rats have shown that IGF-I receptor activation is needed for the induction of GFAP changes by estrogen in the arcuate nucleus<sup>25</sup>.

Another alternative mechanism involved in the blocking effect of JB1 on arcuate nucleus synaptic plasticity may be the regulation of local IGF-I levels as a consequence of the antagonistic effect of JB1 on IGF-I receptors in tanycytes. As mentioned previously, tanycytes, a specialized form of glial cells present in the walls of the third ventricle and in the median eminence, may play an important role in the regulation of local IGF-I levels in the arcuate nucleus. Tanycytes express IGF-I receptors, are able to accumulate IGF-I from extrahypothalamic sources, and show changes during the estrous cycle in IGF-I immunoreactivity and IGF-I accumulation. Therefore, tanycytes could represent a possible site of action for JB1 in the inhibition of synaptic plasticity. Indeed, administration of JB1 in the rat lateral cerebral ventricle is able to block the accumulation of IGF-I by arcuate nucleus tanycytes<sup>26</sup>.

Quesada and Etgen<sup>27,28</sup> have reported several functional interactions between estrogen and IGF-I receptor in neuroendocrine brain areas. These authors have shown that estradiol treatment induces alpha(1B)-adrenoceptor expression in the hypothalamus and preoptic area and that this hormonal effect is blocked by intracerebroventricular infusion of the IGF-I receptor antagonist JB1. Furthermore, estradiol potentiates the effects of IGF-I on alpha(1)-adrenergic receptor activation in the preoptic area and hypothalamus. Blockade of the IGF-I receptor during estrogen priming blocks estrogen-induced luteinizing hormone release and partially inhibits hormone-dependent reproductive behavior. Quesada and Etgen have proposed that the interaction of estrogen with IGF-I receptor may help coordinate the timing of ovulation with the expression of sexual receptivity<sup>28</sup>.

# 3. INTERACTION OF IGF-I AND ESTRADIOL ON NEURONAL DEVELOPMENT

The interactions of estrogen and IGF-I are also relevant for brain development. The interaction between these two factors contributes to the development of structural sex differences in the brain through regulation of the survival and differentiation of developing neurons in brain areas involved in the regulation of neuroendocrine events and reproduction<sup>29</sup>. Several studies have demonstrated an interdependence of estrogen receptors and the IGF-I receptor in promoting the survival and differentiation of developing hypothalamic neurons<sup>30,31</sup>. Both estradiol and IGF-I promote neuronal survival and differentiation in primary neuronal cultures grown in a defined medium deprived of serum and hormones. In these cultures, the induction of neuronal survival and differentiation by IGF-I is prevented by inhibiting the synthesis of estrogen receptor  $\alpha$  by the addition of a specific antisense oligonucleotide<sup>31</sup>. The effect of IGF-I can also be prevented by the estrogen receptor antagonist ICI 182,780<sup>31</sup>. In turn, the promotion of neuronal survival and differentiation by estradiol is prevented by blocking the synthesis of IGF-I in the cultures by using a specific IGF-I antisense

oligonucleotide<sup>31</sup>, as well as by the pharmacological blockade of the MAPK and the PI3K signaling pathways, both used by the IGF-I receptor<sup>32</sup>.

# 4. INTERACTION OF IGF-I AND ESTRADIOL ON THE RESPONSE TO BRAIN INJURY AND ADULT NEUROGENESIS

Estradiol and IGF-I may also interact in areas of the brain that are not directly involved in the control of neuroendocrine events and reproduction, as has been shown in studies assessing the neuroprotective effects of these two factors. IGF-I and estradiol have neuroprotective properties and prevent neuronal cell death in different experimental models of neurodegenerative diseases<sup>33,34</sup>. The interaction of IGF-I and estradiol in neuroprotection has been assessed in ovariectomized rats, using systemic administration of kainic acid to induce degeneration of hippocampal hilar neurons<sup>35</sup>, an experimental model of excitotoxic cell death. Both systemic administration of estradiol and intracerebroventricular infusion of IGF-I prevent hilar neuronal loss induced by kainic acid. The neuroprotective effect of estradiol is blocked by intracerebroventricular infusion of the IGF-I receptor antagonist JB1, while the neuroprotective effect of IGF-I is blocked by the intracerebroventricular infusion of the estrogen receptor antagonist ICI 182,780<sup>35</sup>. Similar results have been obtained after the unilateral infusion of 6-hydroxdopamine into the medial forebrain bundle to lesion the nigrostriatal dopaminergic pathway<sup>36</sup>, a model of Parkinson's disease. Pretreatment with estrogen or IGF-I significantly prevents the loss of substantia nigra compacta neurons and the related motor disturbances. Blockage of IGF-I receptors by intracerebroventricular JB-1 attenuates the neuroprotective effects of both estrogen and IGF-I. These findings suggest that the neuroprotective actions of estradiol and IGF-I after brain injury depend on the coactivation of both estrogen and IGF-I receptors.

Another functional outcome of the interaction of IGF-I with estrogen in the brain is the regulation of adult neurogenesis. Neural precursors located in the subgranular zone of the dentate gyrus of the hippocampus proliferate in adult rodents. The newly generated neurons are functional, integrate in hippocampal circuits and may be involved in certain forms of hippocampusdependent learning. IGF-I and estradiol are among the molecules that have been identified as modulators of adult neurogenesis<sup>8,37</sup>. A recent study has shown that intracerebroventricular administration of the estrogen receptor antagonist ICI 182,780 blocks IGF-I-induced neurogenesis in adult ovariectomized rats<sup>38</sup>, suggesting that estrogen receptors are involved in the action of IGF-I on adult hippocampal neurogenesis (Fig. 1). This interaction of IGF-I with estrogen receptors may occur directly in the proliferating cells, since they express receptors for both IGF-I and estrogen<sup>38</sup>.

# 5. MECHANISMS OF INTERACTION BETWEEN IGF-I AND ESTRADIOL IN THE BRAIN

Analysis of the distribution of the IGF-I receptor and of the two known forms of estrogen receptors ( $\alpha$  and  $\beta$ ) in the rat brain reveals that most neural cells expressing IGF-I receptor also express estrogen receptors<sup>39</sup>. Colocalization of estrogen receptor  $\alpha$  and IGF-I receptor immunoreactivity is observed, by confocal microscopy, in neurons in the preoptic area and hypothalamus. IGF-I receptor immunoreactivity is detected in the cytoplasm and estrogen receptor  $\alpha$  immunoreactivity is located in the cell nucleus. Some neurons in the ventromedial nucleus and the arcuate nucleus are immunoreactive for both receptors. Many neurons in the hypothalamus and preoptic area show estrogen receptor  $\alpha$  immunoreactivity only; however, most IGF-I receptor immunoreactive neurons are also estrogen receptor  $\alpha$ immunoreactive<sup>39</sup>.

In the hippocampal formation, colocalization of estrogen receptor  $\alpha$  and the IGF-I receptor is seen in many pyramidal neurons in the lateral portion of the CA1 field. The number of double labeled cells is lower in the CA2 layer and no double labeling is observed in the CA3 or the dentate gyrus<sup>39</sup>.

The cerebral cortex presents abundant immunohistochemical colocalization of IGF-I receptor and estrogen receptor  $\alpha$ . There is abundant colocalization in layers III and IV from the frontoparietal cortex motor area and to a lesser extent in layer VI. Similarly, many neurons show colocalization of estrogen receptor  $\alpha$  and IGF-I receptor in the primary olfactory cortex, whereas colocalization is less abundant in the posterior cingulate cortex and scarce in the frontoparietal cortex somatosensory area<sup>39</sup>.

Estrogen receptor  $\beta$  is expressed in almost all IGF-I receptor immunoreactive neurons in the preoptic area and the hypothalamus. Double labeled cells are abundant in the preoptic area and in the supraoptic, paraventricular and arcuate nuclei. In most neurons the labeling for estrogen receptor  $\beta$  is observed in the cell nucleus; however, cytoplasm labeling is faint, but consistent, in cells with estrogen receptor  $\beta$  in nuclei. Colocalization of estrogen receptor  $\beta$  and IGF-I receptor is observed in the cytoplasm. In magnocellular neurons of the supraoptic and paraventricular nuclei, estrogen receptor  $\beta$  immunoreactivity is predominantly located in the cell nucleus and there is little colocalization of estrogen receptor  $\beta$  and IGF-I receptor immunoreactivity in the cytoplasm<sup>39</sup>.



*Figure 1.* Total number of BrdU-labeled cells (A) and total number of BrdU-labeled neurons (B) in the granule cell layer and subgranular zone of the dentate gyrus of ovariectomized rats. Animals received a daily intraperitoneal injection of 10 mg/ml BrdU (5-bromo-2-deoxyuridine; Sigma, St. Louis, MO.) in 0.9% NaCl solution during six consecutive days (daily dose: 50 mg/kg b.w.). Animals were killed 24 h after the last injection of BrdU. Data are expressed as means  $\pm$  S.E.M. E2, Animals treated with 17- $\beta$  estradiol; IGF-I, animals treated with insulin-like growth factor-I; ICI, animals treated with the estrogen receptor antagonist ICI 182,780. The number of animals in each experimental group was: control (n = 5); estradiol (n = 6); IGF-I (n = 5); IGF-I + ICI (n = 6); IGF-I + E2 + ICI (n = 6); ICI (n = 7). Asterisks = significant differences (\*p < 0.05, \*\*p < 0.01, \*\*\*p < 0.001) versus control values.

Many pyramidal neurons of the CA3 layer from the hippocampus are immunoreactive for both estrogen receptor  $\beta$  and IGF-I receptor, with overlapping colocalization in the cytoplasm. In the dentate gyrus there is cytoplasmic colocalization of both receptors in many interneurons. The colocalization of estrogen receptor  $\beta$  and IGF-I receptor is low in the rest of the hippocampal fields<sup>39</sup>.

	<b>Colocalization of IGF-I</b>	receptors with
Area studied	ER alpha	ER beta
PREOPTIC AREA	N	N,A
Medial preoptic nucleus	-	N,A
HYPOTHALAMIC NUCLEI		
Supraoptic & paraventricular	-	N,A
Ventromedial & Arcuate	N	N,A
HIPPOCAMPUS		
CA1	N	А
CA2	N	N,A
CA3	-	N,A
Dentate gyrus	-	N,A
CEREBRAL CORTEX		
Primary olfactory cortex	N	N,A
Posterior cingulate cortex	Ν	N,A
Frontoparietal cortex		
Motor area		
Layer I	-	N,A
Layers II-V	N	N,A
Layer VI	Ν	Ν
Somatosensory area		
Layers I-VI	Ν	N,A
CEREBELLAR CORTEX		
Molecular layer	-	А
Purkinje cell layer	-	Ν
Granule cell layer	-	А
White matter	-	А

*Table 1.* Coexpression of IGF-I receptor and estrogen receptors in selected brain areas. A, Astrocytes; N, neurons. Based on ref. 39.

Estrogen receptor  $\beta$  and IGF-I receptor are colocalized in the frontoparietal cortex motor area from layer II to layer VI. The co-expression is more abundant in layers II and III. There is scarce colocalization of estrogen receptor  $\beta$  and IGF-I receptor in the posterior cingulate region and the frontoparietal cortex somatosensory area. A high proportion of neurons show coexpression of estrogen receptor  $\beta$  and IGF-I receptor  $\beta$  and IGF-I receptor in the proportion of neurons show coexpression of estrogen receptor  $\beta$  and IGF-I receptor in the primary

olfactory cortex. Cytoplasmic coexpression of the estrogen receptor  $\beta$  and IGF-I receptor is observed in the Purkinje cells of the cerebellum<sup>39</sup>.

Colocalization of immunoreactivity for estrogen receptor  $\beta$  and IGF-I receptor is observed in glial cells with the morphological appearance of astrocytes in all brain areas. These glial cells are particularly abundant in the hilus of the dentate gyrus and in the white matter of the cerebellum<sup>39</sup>.

Table 1 is a summary of the distribution of immunoreactive cells in the preotic area, the hypothalamus, the cerebral cortex, the hippocampus and the cerebellar cortex, in which colocalization of IGF-I and estrogen receptors has been observed.

The abundant coexpression of estrogen receptors with IGF-I receptors in neurons and glia in the brain indicates that interactions of the intracellular signaling pathways of IGF-I receptors and estrogen receptors are possible in many brain cells. In different cell lines, including neuroblastoma cells, IGF-I may activate estrogen receptors in the absence of estradiol<sup>40</sup>. Whether or not this is also valid for the brain *in vivo* is unknown. However, both *in vitro* and *in vivo* studies have shown that in the brain estradiol may activate the two main signal transduction cascades coupled to the IGF-I receptor: the PI3K and the MAPK signaling pathways.

In primary cultures of astroglia and in explants of cerebral cortex, estradiol induces the rapid activation of ERK<sup>41,42</sup>. The estradiol-induced activation of ERK may be detected in vivo after systemic administration of the hormone<sup>43</sup>. Estradiol also induces the phosphorylation of Akt in dissociated neurons from the rat cerebral cortex and the hippocampus, in cerebral cortical explants and in the adult rat brain in vivo<sup>43,44</sup>. Furthermore. IGF-I and estradiol act synergistically to increase Akt activity in the rat brain<sup>44</sup>. Figure 2 shows the result of an experiment in which IGF-I was infused with an osmotic minipump into the lateral cerebral ventricle of ovariectomized rats to determine if a sustained increase in intracerebral IGF-I levels affects the activation of Akt/PKB by estrogen in the hippocampus. Animals were infused for 6 days with IGF-I and then injected with estradiol and killed 24 h later. A significant increase in the phosphorylation of Akt/PKB was observed in rats injected with estradiol alone compared to animals treated with vehicle. The infusion of IGF-I in the lateral cerebral ventricle was, by itself, unable to increase the phosphorylation of Akt/PKB. However. the hippocampus of animals that received the intracerebroventricular infusion of IGF-I showed a striking increase in Akt/PKB phosphorylation in response to estradiol compared to animals infused with vehicle. This synergistic effect of IGF-I and estradiol was not observed for ERK1 and ERK2<sup>44</sup>.



*Figure 2.* Levels of phosphorylated Akt (pAkt) in the hippocampus of ovariectomized rats killed 24 h after systemic administration of estradiol (E), after 7 days of intracerebroventricular infusion of IGF-I (I) or 24 h after systemic administration of estradiol and 7 days after intracerebroventricular infusion of IGF-I (EI). Levels of basal Akt were used as a control for protein loading. C, Control rats injected with vehicles. Data are normalized to control values and represent the mean  $\pm$  SEM of arbitrary densitometric units. Samples from 6 rats were analyzed for each experimental group. Asterisks represent significant differences (p < 0.05) versus control values.

Glycogen synthase kinase 3 (GSK3), downstream of Akt, may also be a point of interaction of estrogen and IGF-I signaling. Physiological phosphorylation of microtubule-associated proteins by GSK3B may be involved in the regulation of microtubule dynamics, neuritic growth, synaptogenesis and synaptic plasticity<sup>45</sup>. However, under pathological conditions, GSK3β may be responsible for the hyperphosphorylation of tau in Alzheimer's disease<sup>46</sup> and its inhibition is associated with the activation of survival pathways in neurons47. Interestingly, estradiol regulates the activity of GSK3 and decreases the phosphorylation of tau in the rat hippocampus in vivo<sup>48</sup>. Furthermore, estradiol increases the association of tau with phosphorylated GSK3B, with the p85 subunit of the phosphatidylinositol 3kinase and with  $\beta$ -catenin, another substrate of GSK3<sup>48</sup>. These observations are coherent with results showing that systemic administration of estradiol to adult ovariectomized rats results in a transient increase in tyrosine phosphorylation of the IGF-I receptor, in a transient interaction of the IGF-I receptor with the estrogen receptor  $\alpha$ , but not  $\beta$ , and in an enhanced interaction of estrogen receptor  $\alpha$  with the p85 subunit of PI3K in the brain  $(Fig. 3)^{49}$ .



Figure 3. (A) Estrogen treatment in vivo activates IGF-IR and induces the association between IGF-I receptor (IGF-IR) and estrogen receptor a (ER $\alpha$ ) in the hypothalamus of ovariectomized rats. Animals were killed at the indicated periods of time after acute estrogen treatment. IGF-I receptor was immunoprecipitated from total lysates. The immunocomplexes were separated using sodium dodecyl sulfate-polyacrylamide gel electrophoresis (SDS-PAGE), and probed with an antiphosphotyrosine antibody (p-Tyr, upper gel), an antibody against estrogen receptor a (middle gel). To verify equal loading, membranes were probed with an antibody against IGF-I receptor (lower gel). The histogram shows the densitometric analysis of the Western blots. Values are the percentages of control levels and are represented as means  $\pm$  S.E.M. from three different animals. Data were analyzed by ANOVA followed by the Bonferroni's test. Asterisks indicate significant differences (p < 0.05) versus the values of control animals injected with vehicle (C). (B) Estrogen treatment induces the association between IGF-I receptor and estrogen receptor  $\alpha$ , but not estrogen receptor  $\beta$  (ER  $\beta$ ) in the hypothalamus of ovariectomized rats. Animals were treated with estrogen for 1 h, 3 h or with vehicle. Hypothalamic lysates were immunoprecipitated with IGF-I receptor antibody and probed with estrogen receptor  $\alpha$ , estrogen receptor  $\beta$  and IGF-I receptor antibodies. The ability of these antibodies to recognize the antigens in total hypothalamic lysates (Lys) is also shown. (C) Parallel immunoprecipitations were performed using a nonimmune (n.i.) serum to verify the specificity of the bands detected by Western blotting.

Together these findings suggest that estrogen receptor  $\alpha$  may affect IGF-I actions in the brain by a direct interaction with some of the components of IGF-I signaling, such as the IGF-I receptor, PI3K and GSK3 $\beta$  (Fig. 4).

# 5.1 Functional Implications of the Interactions in the Signaling of IGF-I Receptor and Estrogen Receptorα in the Brain

The interaction of estrogen receptor- $\alpha$  with the signaling pathways of IGF-I receptor in the brain may explain the interaction of estradiol and IGF-I in the regulation of different neural events. ERK and Akt may be involved in the interaction of IGF-I and estradiol in the regulation of neuronal differentiation, synaptic function, synaptic remodeling, neuroprotection, and sexual behavior<sup>1,43,50</sup>. The synergistic interaction of IGF-I and estradiol in the phosphorylation of Akt may be critical for the regulation of glucose transport and metabolism<sup>51,52</sup> and for neuroprotective actions. Akt regulates several transcription factors that may be involved in the control of neuronal survival, such as CREB,<sup>53</sup> NF-kappaB<sup>54</sup>, and several members of the forkhead family<sup>55-57</sup>. In addition, activation of Akt results in the phosphorylation of the Bcl-2 family member Bad and this may suppress Bad-induced cell death<sup>58,59</sup>. Furthermore, Akt activation enhances Bcl-2 promoter activity<sup>53</sup> and both IGF-I and estrogen induce Bcl-2 expression in neurons. Interestingly, IGF-I receptor activation is necessary for the induction of Bcl-2 by estradiol in the adult brain<sup>43</sup>. Also downstream of Akt, IGF-I and estradiol may interact on the regulation of microtubule dynamics, neuritic growth, synaptogenesis, synaptic plasticity, and neuronal survival, acting on GSK3B and its substrates,  $\beta$ -catenin, and tau<sup>48</sup>.



*Figure 4.* Proposed model for the interaction of estradiol and IGF-I in the brain. Estradiol interacts with IGF-I at different levels, regulating the activity and association of different proteins involved in the cellular response to IGF-I. Estradiol regulates the activity of protein kinases, increasing IGF-IR (1) and Akt (3) phosphorylation. E2 decreases GSK3  $\beta$  activity (4), which in turn decreases tau phosphorilation (6). In addition, ER $\alpha$  interacts with GSK3  $\beta$  (4) and estradiol treatment modulates the interaction between Er $\alpha$ , IGF-IR (1), and p85 (2). The association of p85 (5) and  $\beta$  catenin (7) with tau is also increased on estradiol treatment in the brain.

## 6. CONCLUDING REMARKS

The studies reviewed in this chapter indicate that the mechanisms of intracellular signaling of IGF-I and estradiol are intimately associated in the nervous system. Estrogen receptor  $\alpha$  appears to be part of the signaling mechanism of IGF-I in the brain. In turn, estradiol activates IGF-I receptor signaling in neurons and glia. The interaction between IGF-I and estradiol appears to be relevant during brain development and in adulthood and participate in the regulation of neuronal differentiation, neuronal survival, synaptic connectivity, neural plasticity, neuroendocrine regulation, and behavior. The coupling of the signaling pathways of IGF-I and estradiol in neural cells may act as a coincidence signal detector to timely coordinate the endocrine and paracrine effects of both factors.

Finally, it should be noted that the crosstalk between the signaling of estradiol and IGF-I are not a peculiarity of neural tissue, since both factors interact in other cellular systems as well.<sup>60-62</sup> Among the most studied examples are breast cancer cells. Both factors play a role in the development and proliferation of breast cancer cells and show synergistic interactions in the activation of cell cycle components, resulting in increased proliferation rates.<sup>63,64</sup> The cross-talk between the receptors for estradiol and IGF-I in breast cancer cells show many similarities to that observed in neural cells.<sup>63,65-67</sup> Therefore, it is possible that the interactions, reviewed in this chapter, between IGF-I and estradiol may represent a general mechanism operating in different tissues.

## ACKNOWLEDGEMENTS

The authors acknowledge financial support from Ministerio de Ciencia y Tecnología, Spain (SAF 2002-00652), Comunidad de Madrid (08.5/0002.2/2003) and the Commission of the European Communities, specific RTD programme "Quality of Life and Management of Living Resources" (QLK6-CT-2000-00179).

#### REFERENCES

- L.M. Frago, C. Pañeda, S.L. Dickson, A.K. Hewson, J. Argente and J.A. Chowen, Growth hormone (GH) and GH-releasing peptide-6 increase brain insulin-like growth factor-I expression and activate intracellular signaling pathways involved in neuroprotection. *Endocrinology*, 143, 4113-4122 (2002).
- 2. F. de Pablo, L.R. Banner and P.H. Patterson, IGF-I expression is decreased in LIFdeficient mice after peripheral nerve injury. *NeuroReport*, **11**, 1365-1368 (2000).
- D.M. Jones, B.A Tucker, M. Rahimtula and K.M. Mearow, The synergistic effects of NGF and IGF-1 on neurite growth in adult sensory neurons: convergence on the PI 3kinase signaling pathway. *J Neurochem*, 86, 1116-1128 (2003).
- 4. E. Garcia-Galloway, C. Arango, S. Pons and I. Torres-Aleman, Glutamate excitotoxicity attenuates insulin-like growth factor-I prosurvival signaling. *Mol Cell Neurosci* 24, 1027-1037 (2003).
- 5. G.P. Cardona-Gomez, P. Mendez, L.L. DonCarlos, I. Azcoitia and L.M. Garcia-Segura, Interactions of estrogens and insulin-like growth factor-I in the brain: implications for neuroprotection, *Brain Res Rev*, **37**, 320-334 (2001).
- 6. L.M. Garcia-Segura, J.R. Rodriguez and I. Torres-Aleman, Localization of the insulinlike growth factor I receptor in the cerebellum and hypothalamus of adult rats: an electron microscopic study, J. Neurocytol. **26**, 479-490 (1997).
- 7. I. Torres-Aleman, Insulin-like growth factors as mediators of functional plasticity in the adult brain. *Horm Metab Res*, **31**, 114-119 (1999).

- 8. J.L. Trejo, E. Carro and I. Torres-Aleman, Circulating insulin-like growth factor I mediates exercise-induced increases in the number of new neurons in the adult hippocampus, J. Neurosci. 21, 1628-1634 (2001).
- E.D. Lephart, A review of brain aromatase cytochrome P450. Brain Res Brain Res Rev, 22, 1-26 (1996).
- L.M. Garcia-Segura, A. Wozniak, I. Azcoitia, J.R. Rodriguez, R.E. Hutchison and J.B. Hutchison, Aromatase expression by astrocytes after brain injury: implications for local estrogen formation in brain repair. *Neuroscience*, **89**, 567-578 (1999).
- 11. J. Guan, L. Bennet, P.D. Gluckman and A.J. Gunn, Insulin-like growth factor-1 and post-ischemic brain injury. *Prog Neurobiol*, **70**, 443-462 (2003).
- 12. J.A. Chowen, I. Azcoitia, G.P. Cardona-Gomez and L.M. Garcia-Segura, Sex steroids and the brain: lessons from animal studies. *J Pediatr Endocrinol Metab*, **13**,1045-1066 (2000).
- 13. B. McEwen, Estrogen actions throughout the brain, *Recent Prog Horm Res*, **57**, 357-384 (2002).
- P.M. Wise, D.B. Dubal, M.E. Wilson, S.W. Rau, M. Bottner and K.L. Rosewell, Estradiol is a protective factor in the adult and aging brain: understanding of mechanisms derived from in vivo and in vitro studies. *Brain Res Brain Res Rev*, 37, 313-319 (2001).
- 15. D.G. Stein, Brain damage, sex hormones and recovery: a new role for progesterone and estrogen? *Trends Neurosci*, **24**, 386-391 (2001).
- J.P. Bourguignon, A. Gerard, M.L. Alvarez-Gonzalez and P. Franchimont, Acute suppression of gonadotrophin-releasing hormone secretion by insulin-like growth factor I and subproducts: an age-dependent endocrine effect. *Neuroendocrinology*, 58, 525-530 (1993).
- 17. J.K. Hiney, V. Srivastava, C.L. Nyberg, S.R. Ojeda and W.L. Dees, Insulin-like growth factor I of peripheral origin acts centrally to accelerate the initiation of female puberty. *Endocrinology*, **137**, 3717-3728 (1996).
- F. Pazos, F. Sanchez-Franco, J. Balsa, J. Lopez-Fernandez, J. Escalada and L. Cacicedo, Regulation of gonadal and somatotropic axis by chronic intraventricular infusion of insulin-like growth factor 1 antibody at the initiation of puberty in male rats. *Neuroendocrinology*, **69**, 408-416 (1999).
- 19. D.J. Burks, J.F. de Mora, M. Schubert, D.J. Withers, M.G. Myers, H.H. Towery, S.L. Altamuro, C.L. Flint and M.F. White, IRS-2 pathways integrate female reproduction and energy homeostasis. *Nature*, **407**, 377-382 (2000).
- 20. B.H. Miller and A.C. Gore, Alterations in hypothalamic insulin-like growth factor-I and its association with gonadotrophin releasing hormone neurones during reproductive development and ageing. *J Neuroendocrinol*, **13**, 728-736 (2001).
- L.M. Garcia-Segura, J.A. Chowen, A. Parducz and F. Naftolin, Gonadal hormones as promoters of structural synaptic plasticity: cellular mechanisms. *Prog Neurobiol*, 44, 279-307 (1994).
- G.P. Cardona-Gomez, J.A. Chowen and L.M. Garcia-Segura, Estradiol and progesterone regulate the expression of insulin-like growth factor-I receptor and insulin-like growth factor binding protein-2 in the hypothalamus of adult female rats. J Neurobiol, 43, 269-281 (2000).
- 23. G.P. Cardona-Gomez, J.L. Trejo, A.M. Fernandez and L.M. Garcia-Segura, Estrogen receptors and insulin-like growth factor-I receptors mediate estrogen-dependent synaptic plasticity. *NeuroReport*, **11**, 1735-1738 (2000).

- M.C. Fernandez-Galaz, F. Naftolin and L.M. Garcia-Segura, Phasic synaptic remodelling of the arcuate nucelus during the estrous cycle depends on insulin-like growth factor-I receptor activation. *J Neurosci Res*, 55, 286-292 (1999).
- M.C. Fernandez-Galaz, E. Morschl, J.A. Chowen, I. Torres-Aleman, F. Naftolin and L.M. Garcia-Segura, Role of astroglia and insulin-like growth factor-I in gonadal hormone-dependent synaptic plasticity. *Brain Res Bull*, 44, 525-531 (1997).
- M.C. Fernandez-Galaz, I. Torres-Aleman and L.M. Garcia-Segura, Endocrinedependent accumulation of IGF-I by hypothalamic glia. *NeuroReport*, 8, 373-377 (1996).
- 27. A. Quesada and A.M. Etgen, Insulin-like growth factor-1 regulation of alpha(1)adrenergic receptor signaling is estradiol dependent in the preoptic area and hypothalamus of female rats. *Endocrinology*, **142**, 599-607 (2001).
- 28. A. Quesada and A.M. Etgen, Functional interactions between estrogen and insulin-like growth factor-I in the regulation of alpha 1B-adrenoceptors and female reproductive function. *J Neurosci*, **22**, 2401-2048 (2002).
- 29. H.F. Carrer and M.J. Cambiasso, Sexual differentiation of the brain: genes, estrogen, and neurotrophic factors. *Cell Mol Neurobiol*, **22**, 479-500 (2002).
- C.D. Toran-Allerand, L. Ellis and K.H. Pfenninger, Estrogen and insulin synergism in neurite growth enhancement in vitro: mediation of steroid effects by interactions with growth factors? *Brain Res*, 469, 87-100 (1988).
- M. Dueñas, I. Torres-Aleman, F. Naftolin and L.M. Garcia-Segura, Interaction of insulin-like growth factor-1 and estradiol signaling pathways on hypothalamic neuronal differentiation. *Neuroscience*, 74, 531-539 (1996).
- 32. L.M. Garcia-Segura, G.P. Cardona-Gomez, J.A. Chowen and I. Azcoitia, Insulin-like growth factor-I receptors and estrogen receptors interact in the promotion of neuronal survival and neuroprotection. *J Neurocytol*, **29**, 425-437 (2000).
- 33. E. Carro, J.L. Trejo, A. Nunez and I. Torres-Aleman, Brain repair and neuroprotection by serum insulin-like growth factor I. *Mol Neurobiol*, **27**,153-162 (2003).
- 34. L.M. Garcia-Segura, I. Azcoitia and L.L. DonCarlos, Neuroprotection by estradiol. *Prog Neurobiol*, **63**, 29-60 (2001).
- 35. I. Azcoitia, A. Sierra and L.M. Garcia-Segura, Neuroprotective effects of estradiol in the adult rat hippocampus: interaction with insulin-like growth factor-I signalling, *J Neurosci Res*, **58**, 815-822 (1999).
- A. Quesada and P.E. Micevych, Estrogen interacts with the IGF-1 system to protect nigrostriatal dopamine and maintain motoric behavior after 6-hydroxdopamine lesions. *J Neurosci Res*, **75**, 107-116 (2004).
- 37. P. Tanapat, N.B. Hastings, A.J. Reeves and E. Gould, Estrogen stimulates a transient increase in the number of new neurons in the dentate gyrus of the adult female rat. J Neurosci, 19, 5792-5801 (1999).
- M. Perez-Martin, I. Azcoitia, J.L. Trejo, A. Sierra and L.M. Garcia-Segura, An antagonist of estrogen receptors blocks the induction of adult neurogenesis by insulin-like growth factor-I in the dentate gyrus of adult female rat. *Eur J Neurosci*, 18, 923-930 (2003).
- G.P. Cardona-Gomez, L. DonCarlos and L.M. Garcia-Segura, Insulin-like growth factor I receptors and estrogen receptors colocalize in female rat brain. *Neuroscience*, 99, 751-760 (2000).
- Z.Q. Ma, S. Santagati, C. Patrone, G. Pollio, E. Vegeto and A. Maggi, Insulin-like growth factors activate estrogen receptor to control the growth and differentiation of the human neuroblastoma cell line SK-ER3. *Mol Endocrinol*, 8, 910-918 (1994).

- 41. C.D. Toran-Allerand, M. Singh and G. Sétáló, Novel mechanisms of estrogen action in the brain: new players in an old story. *Front Neuroendocrinol*, **20**, 97-121 (1999).
- 42. T. Ivanova, M. Karolczak and C. Beyer, Estrogen stimulates the mitogen-activated protein kinase pathway in midbrain astroglia. *Brain Res*, **889**, 264-269 (2001).
- G.P. Cardona-Gomez, P. Mendez, L.L. DonCarlos, I. Azcoitia and L.M. Garcia-Segura, Interactions of estrogen and insulin-like growth factor-I in the brain: molecular mechanisms and functional implications. J Steroid Biochem Mol Biol, 83, 211-217 (2003).
- 44. G.P. Cardona-Gomez, P. Mendez and L.M. Garcia-Segura, Synergistic interaction of estradiol and insulin-like growth factor-I in the activation of PI3K/Akt signaling in the adult rat hypothalamus. *Brain Res Mol Brain Res*, **107**, 80-88 (2002).
- 45. A.C. Hall, F.R. Lucas and P.C. Salinas, Axonal remodeling and synaptic differentiation in the cerebellum is regulated by WNT-7a signaling. *Cell*, **100**, 525-535 (2000).
- 46. S. Lovestone, C. H. Reynolds, D. Latimer, D. R. Davis, B. H. Anderton, J.M. Gallo, D. Hanger, S. Mulot, B. Marquardt, S. Stabel, J. R. Woodgett and C.C.J. Miller, Alzheimer's disease-like phosphorylation of the microtubule-associated protein tau by glycogen synthase kinase-3 in transfected mammalian cells. *Curr Biol*, 4, 1077-1086 (1994).
- D.A. Cross, D.R. Alessi, P. Cohen, M. Andjelkovich and B.A. Hemmings. Inhibition of glycogen synthase kinase-3 by insulin mediated by protein kinase B. *Nature*, **378**, 785-789 (1995).
- P. Cardona-Gomez, M. Perez, J. Avila, L.M. Garcia-Segura and F. Wandosell, Estradiol inhibits GSK3 and regulates interaction of estrogen receptors, GSK3 and beta-catenin, in the hippocampus, *Mol Cell Neurosci*, 25, 363-373 (2004).
- P. Mendez, I. Azcoitia and L.M. Garcia-Segura, Estrogen receptor alpha forms estrogen-dependent multimolecular complexes with insulin-like growth factor receptor and phosphatidylinositol 3-kinase in the adult rat brain. *Brain Res Mol Brain Res*, 112, 170-176 (2003).
- 50. A.M. Etgen and M. Acosta-Martinez, Participation of growth factor signal transduction pathways in estradiol facilitation of female reproductive behavior. *Endocrinology*, **144**, 3828-3835 (2003).
- 51. S.A. Summers, V.P. Yin, E.L. Whiteman, L.A. Garza, H. Cho, R.L. Tuttle, and M.J. Birnbaum, Signaling pathways mediating insulin-stimulated glucose transport. *Ann NY Acad Sci*, **892**, 169-186 (1999).
- 52. C.M. Cheng, R.R. Reinhardt, W.H. Lee, G. Joncas, S.C. Patel and C.A. Bondy, Insulinlike growth factor 1 regulates developing brain glucose metabolism. *Proc Natl Acad Sci USA*, **97**, 10236-10241 (2000).
- 53. S. Pugazhenthi, A. Nesterova, C. Sable, K.A. Heidenreich, L.M. Boxer, L.E. Heasley and J.E. Reusch, Akt/protein kinase B up-regulates Bcl-2 expression through cAMP-response element-binding protein. *J Biol Chem*, **275**, 10761-10766 (2000).
- 54. L.P. Kane, V.S. Shapiro, D. Stokoe and A. Weiss, Induction of NF-kappaB by the Akt/PKB kinase. *Curr Biol*, 9, 601-604 (1999).
- 55. A. Brunet, A. Bonni, M.J. Zigmond, M.Z. Lin, P. Juo, L.S. Hu, M.J. Anderson, K.C. Arden, J. Blenis and M.E. Greenberg, Akt promotes cell survival by phosphorylating and inhibiting a Forkhead transcription factor. *Cell*, **96**, 857-868 (1999).
- 56. G.J. Kops, N.D. de Ruiter, A.M. De Vries-Smits, D.R. Powell, J.L. Bos and B.M. Burgering, Direct control of the Forkhead transcription factor AFX by protein kinase B. *Nature*, **398**, 630-634 (1999).
- 57. E.D. Tang, G. Nunez, F.G. Barr and K.L. Guan, Negative regulation of the forkhead transcription factor FKHR by Akt. *J Biol Chem*, **274**, 16741-16746 (1999).

- L. del Peso, M. Gonzalez-Garcia, C. Page, R. Herrera and G. Nunez, Interleukin-3induced phosphorylation of BAD through the protein kinase Akt. *Science*, 278, 687-689 (1997).
- 59. S.R. Datta, H. Dudek, X. Tao, S. Masters, H. Fu, Y. Gotoh and M.E. Greenberg, Akt phosphorylation of BAD couples survival signals to cell-intrinsic machinery. *Cell*, **91**, 231-241 (1997).
- 60. A. Ghahary, S. Chakrabarti and L.J. Murphy, Localization of the sites of synthesis and action of insulin-like growth factor-I in the rat uterus. *Mol Endocrinol*, **4**, 191–195 (1990).
- 61. T. Sato, G. Wang, M.P. Hardy, T. Kurita, G.R. Cunha and P.S. Cooke, Role of systemic and local IGF-I in the effects of estrogen on growth and epithelial proliferation of mouse uterus. *Endocrinology*, **143**, 2673-2679 (2002).
- D.M. Klotz, S.C. Hewitt, P. Ciana, M. Raviscioni, J.K. Lindzey, J. Foley, A. Maggi, R.P. DiAugustine and K.S. Korach, Requirement of estrogen receptor-alpha in insulinlike growth factor-1 (IGF-1)-induced uterine responses and in vivo evidence for IGF-1/estrogen receptor cross-talk. *J Biol Chem*, 277, 8531-8537 (2002).
- 63. I.H. Hamelers and P.H. Steenbergh, Interactions between estrogen and insulin-like growth factor signaling pathways in human breast tumor cells. *Endocr Relat Cancer*, **10**, 331-345 (2003).
- 64. I.H. Hamelers, R.F. van Schaik, H.A. van Teeffelen, J.S. Sussenbach and P.H. Steenbergh, Synergistic proliferative action of insulin-like growth factor I and 17 betaestradiol in MCF-7S breast tumor cells. *Exp Cell Res*, **273**, 107-117 (2002).
- 65. Dupont and D. Le Roith, Insulin-like growth factor 1 and oestradiol promote cell proliferation of MCF-7 breast cancer cells: new insights into their synergistic effects. *Mol Pathol*, **54**, 149-154 (2001).
- 66. M.B. Martin and A. Stoica, Insulin-like growth factor-I and estrogen interactions in breast cancer. *J Nutr*, **132**, 3799S-3801S (2002).
- 67. R.X. Song, C.J. Barnes, Z. Zhang, Y. Bao, R. Kumar and R.J. Santen, The role of Shc and insulin-like growth factor 1 receptor in mediating the translocation of estrogen receptor alpha to the plasma membrane. *Proc Natl Acad Sci USA*, **101**, 2076-2081 (2004).

# Chapter 13

# CANCER

#### Adda Grimberg

The Children's Hospital of Philadelphia; University of Pennsylvania School of Medicine; University of Pennsylvania Cancer Center. Philadelphia, PA 19104-4318

**Key words:** Apoptosis; autocrine; cancer; endocrine; epidemiologic studies; malignant transformation; mitogenesis; paracrine; risk factors.

#### **1. INTRODUCTION**

The growth hormone (GH)/insulin-like growth factor (IGF) axis is the principal endocrine system controlling somatic growth. Hormones are defined as molecules that are released by endocrine glands into the circulation for delivery to target tissues, where they exert their effects via specific receptor binding and post-receptor signaling. Hence. endocrinologists have traditionally measured circulating hormone levels as a major marker of endocrine system action. Epidemiologic data have associated cancer risk with changes in circulating levels of several GH/IGF axis components. The reproducibility of these findings has focused a lot of attention recently on the possible role the GH/IGF axis may contribute to cancer development and progression.

There are two important shortcomings with this approach. First, associations can never prove causation, because the direction of causation remains unknown and there may be confounders. Second, as unexpectedly demonstrated by liver-specific *igf1* gene-deleted (LID) mice, endocrine IGF is but part of the story. Despite a 75% reduction in circulating IGF-I concentrations, LID mice had normal postnatal growth<sup>1</sup>. Just as the relative contributions of endocrine versus local (autocrine or paracrine) GH/IGF axis

components are being questioned in normal growth, so too is their significance in the neoplastic process.

Thus, there are two levels of possible GH/IGF contributions, each supported by different types of data. Endocrine action is investigated in epidemiologic studies and some *in vivo* animal tumor models, whereas autocrine/paracrine actions are studied in *in vitro* experiments of cellular signaling and some *in vivo* animal models. The relative significance of and the relationships between these two levels in carcinogenesis remain unclear, but data are accumulating for both. Currently, there is no strong evidence that GH or IGF-I causes cancer, but the data suggest a possible promoting role for preexisting lesions.

## 2. ENDOCRINE GH/IGF AXIS

#### 2.1 GH

Epidemiologic data associating circulating GH levels with cancer risk have come from various patient populations, which serve as human models of high or low GH levels, and more recently, variations within the normal population. Animal models have also contributed to this field.

#### 2.1.1 GH Deficiency

Concern for a possible carcinogenic effect of GH was raised in 1988 by a letter to Lancet, in which the authors reported a higher incidence of leukemia in their GH recipients than children of the general population<sup>2</sup>. Subsequent reanalyses identified the greater presence of leukemic risk factors, such as prior neoplasms and radiation therapy, among GH-deficient children as a confounder<sup>3,4</sup>. Many patients on recombinant human GH (rhGH) therapy are enrolled in post-marketing surveillance studies, including the National Cooperative Growth Study, KABI International Growth Study, and others, which serve as powerful research tools that complement the safety monitoring by the Food and Drug Administration's MEDWATCH system. A review of 12,209 patients in one such study, representing more than 51,000 patient-years at risk, revealed 10 new cases of extracranial neoplasms, which did not differ from the expected incidence derived from the National Cancer Institute's SEER (Surveillance, Epidemiology, and End Results) Program<sup>5</sup>. Additional data are continuously being accumulated, and the current tally exceeds 200,000 patient-years at risk.

306

A recent retrospective cohort study renewed the scare following years of reassuring experience. In the United Kingdom, 1848 patients treated with cadaveric pituitary-derived GH from 1959 to 1985, were followed up for cancer incidence through 1995 and mortality through 2000. The authors found a higher incidence of colorectal cancer (n = 2) and higher rates of mortality from colorectal cancer (n = 2) and Hodgkin's disease (n = 2) compared to the general population<sup>6</sup>. The small numbers make inference difficult, as does the switch from cadaveric pituitary-derived GH to rhGH in 1985. Implications for clinical rhGH therapy will be addressed in Section 16.5.1, but this study certainly underscores the need for long-term surveillance of GH recipients into adulthood, when the incidence of cancer normally rises.

Regarding untreated GH deficiency, a retrospective review of 333 patients consecutively diagnosed between 1956 and 1987 revealed 3 male deaths from malignancy (expected 10.1) and 4 female deaths (expected 4.1)<sup>7</sup>. But the major finding of this study was a shortened life expectancy, associated with an increased mortality from cardiovascular disease.

#### 2.1.2 Acromegaly

Acromegaly, a condition of pathologic GH excess most commonly caused by a GH-secreting pituitary adenoma, has been studied as a natural model of GH effects on carcinogenesis. Because the underlying pituitary tumor is benign (adenoma), and stereotactic radiation therapy is reserved for surgery- and medication-resistant cases, changes in cancer development in these patients are generally related to the GH excess. The conclusions are controversial (for an excellent point/counterpoint, see ref. 8 and 9), for two main reasons. First, acromegaly is rare (prevalence of 4 to 6 per million), so many studies are uncontrolled, retrospective, small series reports. It is difficult to accurately determine the true cancer incidence among acromegalics from such data, and ascertainment bias may be affecting the results. Second, prior to therapeutic advances, premature mortality (primarily from cardiovascular disease) prevented acromegalic individuals from reaching older ages, when cancer risk is greatest. Thus, meta-analyses comparing early versus more recent reports may show different results, and the appropriateness of the selected control populations may be questionable. By virtue of dying younger than the control population, the acromegalic cancer risk may be under-appreciated. To resolve the debate, multicenter acromegaly registries are being compiled in the United States and United Kingdom.

Data thus far suggest, if anything, an effect of acromegaly on colorectal neoplasia. Still, findings of an increased incidence of colon polyps and

adenocarcinoma have been inconsistent<sup>8,9</sup>. Again, experimental design issues, including control reference data and examination techniques (flexible sigmoidoscopy versus full-length colonoscopy), may be confounding the analyses. It is generally agreed that colorectal neoplasia behaves differently in this patient group. Acromegalics have increased length of total colon and sigmoid loop, mucosal hypertrophy, prolonged colon transit times, and higher levels of deoxycholic acid reflecting increased transit times and intraluminal bacterial activity<sup>10,11</sup>. In acromegalics, colonic lesions tend to be more right-sided, adenomas tend to be larger and more dysplastic, and polyps more often occur multiply<sup>12</sup>.

Colon cancer in acromegaly is an excellent example of how epidemiologic associations are insufficient to prove causation. Acromegaly is the clinical model of GH excess. Yet IGF-I is also pathologically increased in acromegaly. So is it the GH or the IGF-I? Intestinal length and mucosal mass were increased in transgenic mice that overexpressed either GH and IGF-I or IGF-I alone (with GH suppression); increased crypt cell proliferation was observed only in the IGF-I transgenics, whereas GH excess stimulated more intestinal epithelial cell differentiation<sup>13-16</sup>. Furthermore, does the IGF-I excess promote colon cancer directly, through the mechanisms described later in this chapter, or is there a confounder? For instance, excessive IGF-I may lead to increased intestinal length, which prolongs transit time and thereby leads to greater deoxycholic acid levels<sup>17</sup>. Deoxycholic acid has been implicated in colonic mucosal proliferation<sup>18</sup> and colon cancer cell migration<sup>19</sup>, and has been shown to stimulate proteasomal degradation of the tumor suppressor p53<sup>20</sup> and at high concentrations, inhibit expression of the tumor suppressor  $BRCAI^{21}$ .

The possibility of confounding has been similarly raised for prostate cancer in acromegaly. Accentuated and early benign prostatic hyperplasia has been reported in acromegaly<sup>22</sup>. This may lead to an ascertainment bias, as these men may be more likely to seek medical attention for prostate-related symptomatology and be more likely to have detection of subclinical prostate cancer<sup>23</sup>. The overall association of acromegaly with prostate cancer risk is still unclear, owing to the aforementioned effects of early mortality on study results. Thus, we await the outcomes of the multicenter acromegaly registries and additional mechanistic studies.

#### 2.1.3 Normal GH variability

Recently, the consequences of subtle variations in circulating hormone levels within the normal population, rather than larger pathologic changes, have become the interest of epidemiologic studies. A T-to-A polymorphism

at position 1663 of the human *GH1* gene has been identified as a cause of such variability in GH and IGF-I levels. Data from two case-control studies in Hawaii are summarized in Table 1. The first was a population-based study of 535 case patients with colorectal adenocarcinoma and 650 control-subjects; the second, a sigmoidoscopy screening-based study of 139 case patients with adenoma and 202 control subjects<sup>24</sup>. They showed that the *A* allele was associated with a lower ratio of plasma IGF-I/IGFBP-3 and a lower risk of colorectal cancer in the total population; the reduction in colorectal cancer risk was evident in the Caucasian and Native Hawaiian subgroups of the population, but not in the Japanese. Thus, additional ethnic genetic and/or environmental factors, such as diet or exercise, can modify the effects of GH variability on cancer risk.

	T/T	T/A	A/A	P (trend)
IGF-I (ng/ml)	172.3	175.0	152.3	0.05
IGFBP-3	2859	3070	2792	0.05
(ng/ml)				
Height (cm)	170.0	169.5	170.0	0.78
Overall				
OR colorectal	1.00	0.75	0.62	0.006
cancer		[0.58–0.99]	[0.43-0.90]	
OR adenoma	1.00	0.76	0.62	0.17
		[0.46–1.24]	[0.31 - 1.22]	
OR colorectal				
cancer				
Caucasian	1.0	0.85	0.44	0.05
		[0.50–1.43]	[0.21-0.93]	
Native	1.0	0.39	0.20	0.003
Hawaiian		[0.16-0.94]	[0.07-0.59]	
Japanese	1.0	0.79	0.85	0.34
		[0.56-1.12]	[0.53-1.36]	

*Table 1.* Summary of data examining the association between T-to-A polymorphism and risk of colorectal neopalasia in Hawaiian populations. Based on data from ref. 24. OR = adjusted odds ratio; [] = 95% confidence intervals.

#### 2.1.4 Animal models

Animal models allow independent manipulation of both hormonal levels and environmental factors such as carcinogen exposure to test relationships between the two. Mammary carcinogenesis was studied in Spontaneous Dwarf rats (SDR), which are Sprague Dawley rats rendered GH deficient by a *GH* gene mutation<sup>25</sup>. When exposed to the direct-acting carcinogen *N*methyl-*N*-nitrosourea, SDR exhibited lower mammary tumor incidence (3 of 15 rats) and lower tumor number (average 0.2 tumors/rat) compared to similarly exposed wild-type rats (10 of 10 rats and 5.3 tumors/rat, respectively). The indirect-acting carcinogen 7,12-dimethylbenz[*a*] anthracine (DMBA) produced an average 0.21 tumors per SDR/SDR rat versus 4 tumors per wild type or heterozygous rat. Yet these rats also demonstrated an important role for GH in normal mammary gland development. SDRs had less alveolar development, but normal ductal branching; GH infusion induced epithelial cell proliferation and alveolar development similar to that of the wild-type rats.

Mammary carcinogenesis was also studied in dwarf rats (dw) of the Lewis strain, who harbor a recessive mutation specifically affecting GH synthesis<sup>26</sup>. Heterozygous dw/+ rats are normal size; homozygous dw/dw rats have 10% circulating GH concentrations and 50% IGF-I concentrations of their heterozygous littermates. dw/dw and their dw/+ littermates were treated with DMBA on day of life 50 and then 6 weeks of daily injections: dw/+ received normal saline, whereas the dw/dw rats were randomized among saline and low- and high-dose porcine GH. None of the dw/dw saline rats developed mammary tumors by 27 weeks after DMBA; 70% of dw/+ rats had tumors. GH treatment dose-dependently increased mammary tumorigenesis in the dw/dw rats (83% of low-dose and 100% of high-dose). These studies indicate the GH/IGF-I deficiency confers resistance to carcinogen-induced tumor formation.

## 2.2 IGF-I

#### 2.2.1 Epidemiologic studies

Multiple large case-control studies have found positive associations between high circulating IGF-I concentrations and increased risk for different types of cancer. The striking features of these studies are the reproducibility across cancer types and the significant effects of variations in IGF-I concentrations that fall within the normal range, *i.e.* the "high" IGF-I levels in these studies are not pathologically high from disease states, but rather the highest quartile or quintile of the general population compared to the lowest quartile or quintile. These studies are extensively discussed elsewhere<sup>23,27</sup>, and summarized in Table 2<sup>28-50</sup>.

Cancer	IGF axis component	Study design	Results	Ref.
Prostate				
	IGF-I	Case (152)-	RR for highest quartile:	28
	IGFBP-3	control	IGF-I = 4.3 [1.8-10.6]	
		(152)	IGFBP-3 = 0.41 [0.17-1.0]	
	IGF-I	Case (210)-	OR per 100 ng/ml increment of IGF-I:	29
	IGFBP-3	control	1.51 [1.0-2.26] overall and 2.93 [1.43-5.97]	
		(224)	for age less than 70 yrs	
			No association with serum [IGFBP-3]	
	IGF-I	Case (149)-	Increased risk with increasing [IGF-I] (p =	30
	IGFBP-3	control	0.02) and increasing [IGFBP-3] ( $p = 0.03$ )	
		(298)	overall; for men age <59 yrs, association	
			stronger for high [IGF-I] $(p = 0.01)$ , but	
			[IGFBP-3] lost ( $p = 0.44$ )	
	IGF-I	Meta-	OR for highest quartile IGF-I similar to that	31
		analysis	for highest quartile of testosterone	
	IGF-I	Meta-	OR for high IGF-I: 1.47 [1.23-1.77]	32
	IGFBP-3	analysis	OR for high IGFBP-3: 1.26 [1.03-1.54]	
	IGF-I	Case (530)-	For advanced-stage prostate cancer:	33
	IGFBP-3	control	RR for highest quartile $IGF-I = 5.1$ [2.0-	
		(534)	[13.2], high [BP-3] = 0.2 [0.1-0.6].	
			RR for combination high IGF-I/low BP-3 =	
			9.5 [1.9-48.4]	
			Neither was predictive of early-stage	
			cancer, nor associated with Gleason score.	
	IGFBP-3	Pre-op in	[IGFBP-3] lowest in men with bone mets.	34
	IGFBP-2	120 men	[IGFBP-2] higher in cancer patients than	
		with	healthy men and declined with	
		localized	prostatectomy. Among cancer patients,	
		disease	[IGFBP-2] was lower with advanced	
		(post-op in	disease and larger tumor volume.	
		51 of them)		
		VS. 44 beelthy meen		
		healthy men		
		vs. 19 noual		
		hone mets		
Gastro-		bone mets		
intestinal				
Colo-	IGF-I	case (193) -	<b>RR</b> for highest quintile	35
rectal	IGFRP-3	control	IGF-I = 2.51 [1.15-5.46]	55
cancer	IGF-II	(318)	IGFBP-3 = 0.28 [0.012-0.66]	
vanvei		men only	IGF-II not associated with risk.	
	IGFBP-	case $(102)$	OR for highest quintile	36
	1.2.3	control	[GFBP-1 = 0.48[0.23-1.0]	20
	1,2,2	(200)	IGFBP-2 = 0.38[0.015-0.094]	
		women only	IGFBP-3 = 2.46[1.09-5.57]	
	IGF-II	case (92)-	IGF-II elevated in Dukes A and B, but not	37

2	1	$\gamma$
э	I	4

Cancer	IGF axis	Study	Results	Ref.
	component	design		
	IGFBP-2	control (57)	advanced disease.	
			IGFBP-2 related to tumor burden and fell	
<b>.</b>		(50)	with curative resection	
Colo-	IGF-II	case (52) –	OR for adenoma per SD change:	38
rectal		control	IGF-II = 3.05[2.04-4.57]	
adenoma	105 I	(293)	IGFBP-2 =2.26[1.53-3.44]	•
Gastric	IGF-I	26 cases	All 26 had IGF-I over the normal limits	39
adenocar	IGFBP-3	pre-op and	pre-op that significantly decreased post-op	
- cinoma		post-op	but not to normal.	
		days 14 and	IGFBP-3 also high pre-op.	
T		50.		
Lung	ICE I	anaa (204)	OP for highest quartil-	40
	IUF-I ICERP 2	case(204) -	OK for highest quartile $I = 2.0611 \pm 10.2561$	40
	IGFBP-3	(218)	IGF-1 = 2.00[1.19-3.30] ICEPP 2 = 0.48[0.35.0.02]	
	10 <b>F</b> -11	(210)	IOFDF-3 = 0.46[0.23-0.92]	
	ICE I	0000 (02)	No associations found	41
	IGF-I	case(95) =	no associations found.	41
	10707-	(186)		
	1,2,5	(100) Women		
		only		
	IGF-I	case(230) =	IGF-I not associated	42
	IGEBP-3	control	OR highest quartile of IGEBP-3:	72
	101.01.5	(659)	All cases: $0.50[0.25-1.02]$	
		Men only	Fyer smokers $(n = 184)$ : $0.41[0.018-0.92]$	
GYN		inten enny		
Breast	IGF-I	case (397) –	Post-menonausal: no association.	43
cancer		control	Premenopausal: RR for top tertile	
		(620)	AII = 2.33[1.06-5.16]	
		()	Age < 50  yrs = 4.58[1.75 - 12.0]	
			Adjusting for IGFBP-3 raised the RR's All	
			= 2.88[1.21-6.85]	
			Age < 50  yrs = 7.28[2.40-22.0]	
	IGF-I	case (63) –	IGF-I levels similar.	44
	IGFBP-1,3	control (27)	OR for high:	
			IGFBP-3 = 0.18[0.05-0.55]	
			IGFBP-1 = 0.21[0.07-0.68]	
	IGF-I	case (40)	OR for high free $IGF-I = 6.31[1.03-38.72]$ .	45
	IGFBP-3	control (40)	High total IGF-I/intact IGFBP-3 ratio, OR	
			= 3.35[1.08-10.36]	
	IGF-I	case (149) -	No associations found.	46
	IGFBP-	control		
	1,2,3	(333)		
		only post-		
		menopausal.		
	Number of	Case (53) –	19 repeats: $OR = 2.8/[1.16-7.06]$	47
	CA repeats	control (53)	Both 19 repeats and high plasma IGF-I: OR	

Cancer	IGF axis component	Study design	Results	Ref.
	in IGF-I		= 5.12[1.42-18.5]	
	gene			
Ovarian	IGF-I	Case (132)	All: no association.	48
cancer		-	Age $< 55$ yrs: top tertile OR = $4.97[1.22$ -	
		control	20.2]	
		(263)		
Endo-	IGF-I	Case (84) -	IGF-I: inverse association	49
metrial	IGF-II	control (84)	IGF-II: positive association	
cancer				
Bladder	IGF-I	Case (154)	OR for highest quartile	50
cancer	IGFBP-3	_	IGF-I = 3.10[1.43-6.70]	
		control	IGFBP-3 = 0.38[0.19-0.78]	
		(154)	IGF-I/IGFBP-3 molar ratio = 4.30[1.99-	
			9.28]	

*Table 2.* Summary of studies associating cancer risk with circulating levels of IGF and IGFBPs. \* numbers in parentheses refer to sample sizes for the case-control studies. [] = 95% confidence intervals for odds ratios and relative risks.

#### 2.2.2 Animal models

As described in the introduction, LID mice serve as the classic model to distinguish endocrine from autocrine/paracrine IGF-I, and have circulating IGF-I concentrations 25% of normal<sup>1</sup>. Six weeks of treatment with recombinant human IGF-I or saline further manipulated the circulating IGF-I concentrations such that control + IGF-I had the highest IGF-I levels, LID + saline had the lowest, and control + saline and LID + IGF-I fell inbetween. When mouse colon adenocarcinoma cells were transplanted onto the surface of the cecum of these animals, the aggressiveness of tumor behavior paralleled the IGF-I levels. Control + IGF-I mice had the greatest frequency of tumor growth, greatest mean tumor weight, greatest tumor vessel count, greatest frequency of hepatic metastases, and greatest numbers of metastases per liver; LID + saline mice had the lowest by all parameters, and the other mice were intermediary<sup>51</sup>.

Athymic nude mice were injected with fibroblasts that contained either normal (16,000/cell) or experimentally increased (190,000/cell) density of the type 1 IGF receptor (IGF-1R). Systemic IGF-I treatment did not change tumor development in the mice injected with normal fibroblasts. However, for the mice with high IGF-1R fibroblasts, systemic IGF-I treatment decreased tumor latency, increased fibrosarcoma growth, and increased mitogenesis<sup>52</sup>.

Circulating IGF-I had an effect on cancer behavior in both models, but it is important to note that both also involved local changes in the cell's growth regulatory mechanisms (preexisting adenocarcinoma or almost 12-fold overexpression of IGF-1R). Thus, the current evidence may support a permissive effect of circulating IGF-I on existing cancers, but not a causal role in the creation of cancer.

## 2.3 IGFBP-3

Many of the same case-control studies showing positive associations of cancer risk with high IGF-I concentrations showed inverse associations of cancer risk with high IGFBP-3 concentrations. This suggests that theoretically the greatest risk of developing cancer comes from having high IGF-I coupled with low IGFBP-3 levels. GH raises both. The IGFBP-3 data are also summarized in Table 2<sup>28-50</sup>.

## 3. AUTOCRINE/PARACRINE GH/IGF AXIS

Many studies have demonstrated that local perturbations in the GH/IGF axis can enhance cell survival and proliferation, and thereby promote cancer. These changes can occur in the cancerous cell itself or in the supporting stroma. Several caveats must be kept in mind. Foremost, multiple genetic changes are required for the creation and progression of each and every cancer, so the GH/IGF axis must be considered within that greater context. Second, none of the changes are universal; they are frequently cancer type-specific and even within a cancer type, stage-specific. Finally, an exhaustive review of all the published changes would exceed the scope of this chapter. What follows, then, is a selective review of illustrative examples demonstrating pathophysiologic principles by which GH/IGF axis changes may contribute to cancer.

#### 3.1 GH

Although the pituitary gland is the main source of GH in the body, abnormal augmentation of local GH production can increase proliferation of the local cells. For example, expression of GH mRNA identical to pituitary GH mRNA has been shown in the ductal luminal epithelial and myoepithelial cells and scattered stromal fibroblasts of normal human mammary glands. This expression was increased in both the epithelial and supporting stromal compartments of three progressive proliferative disorders: benign fibroadenoma, preinvasive intraductal carcinoma, and invasive ductal carcinoma with lymph node metastases<sup>53</sup>. Normal, proliferative, and neoplastic lesions of the breast were also found to express

314

the GH receptor (GHR), as did the stroma to a lesser degree. GHR expression levels varied greatly among individuals, but did not correlate with lesion histology<sup>54</sup>. When MCF-7 human breast cancer cells were stably transfected with either the gene for human GH or a translation-deficient mutated *GH* gene, GH expression increased cellular proliferation and was synergistic with trophic factors such as IGF-I<sup>55</sup>. Mechanistically, GH caused transcriptional repression of the p53-regulated placental transforming growth factor- $\beta$  (PTGF- $\beta$ ); reduced PTGF- $\beta$  in turn led to decreased Smad-mediated transcription which resulted in decreased cell cycle arrest and apoptosis<sup>55</sup>.

#### 3.2 IGFs

Like GH, overexpression of the IGFs may enhance cell survival and proliferation. For example, IGF-I overexpression was found in 31 of 50 thyroid adenomas and 38 of 53 thyroid carcinomas examined, and correlated with carcinoma tumor diameter but not patient age, gender, or tumor stage<sup>56</sup>. IGF-I was not increased in sporadic adrenocortical carcinomas, but the amount of IGF-II protein in the malignant tumors far exceeded that in benign tumors or normal adrenal tissue<sup>57</sup>.

The single-copy six-exon *IGF-1* gene is transcribed from two promoters (P1 and P2), located 5' to exons 1 and 2, respectively, resulting in RNAs with different 5' leader sequences; alternative RNA splicing and differential polyadenylation yield multiple mature transcripts<sup>58</sup>. Normally, local IGF-I expression can be increased by GH<sup>59</sup>, estrogen<sup>60</sup>, cAMP<sup>61</sup>, and transforming growth factor- $\beta$  (TGF- $\beta$ )<sup>62</sup>, and decreased by glucocorticoids<sup>63</sup>. In contrast, the most common mechanism of IGF-II overexpression involves loss of imprinting; IGF-II is normally expressed from the paternal allele only and is under reciprocal regulation with the maternally expressed tumor suppressor, H19<sup>64,65</sup>.

As proof of principle, two sets of IGF-I transgenic mice created by DiGiovanni *et al.* supported the hypothesis that local IGF-I overexpression can enhance cellular proliferation and contribute to neoplasia. Transgenic mice overexpressing IGF-I in the basal epithelial cells of the prostate developed prostatic hyperplasia by 2 to 3 months of age, and atypical hyperplasia and prostatic intraepithelial neoplasia by 6 to 7 months. Well-differentiated adenocarcinomas were found in mice starting at age 6 months, and two of the older mice developed less differentiated (small cell) carcinomas. Of all the mice aged 6 months or greater, 50% had prostate tumors<sup>66</sup>. Transgenic mice overexpressing IGF-I in the basal cells of the epidermis had morphologic changes of their skin and ears, epidermal hyperplasia, hyperkeratosis, and about half the older mice developed squamous papillomas, some of which converted into carcinomas. Increased

skin proliferation was indicated by increased labeling index. IGF-I overexpression also increased susceptibility to carcinogens, as carcinogen-induced papilloma development in the IGF over-expressing mice was seven fold greater than in their nontransgenic littermates<sup>67</sup>.

## **3.3 IGF receptors**

Overexpression of the IGF receptor can also augment cellular proliferation, and when concomitant with IGF overexpression, forms an effective autocrine loop for self-stimulated growth. The growth-promoting actions of both IGF-I and IGF-II are mediated by the type 1 IGF receptor (IGF-1R), an  $\alpha_2\beta_2$  heterotetrameric tyrosine kinase receptor closely related to the insulin receptor (IR). Not only are IGF-1R and IR structurally homologous, but they also share common signaling pathways, ligand cross-reactivity, and can form hybrid receptors. (Table 3)<sup>68-73</sup>.

Receptor	Structure	Ligands*	Characteristics	Function
IGF-1R	Transmembrane $\alpha_2\beta_2$	IGF-I	Ubiquitous;	cell survival,
	tyrosine kinase	IGF-II	important for	mitogenesis,
			normal growth	transformation
IGF-2R	Identical to mannose-	IGF-II	Soluble receptor	clear IGF-II from
	6-phosphate receptor			the circulation
IR-A	Transmembrane $\alpha_2\beta_2$	IGF-II	Predominant IR	fetal growth,
	tyrosine kinase	insulin	isoform in fetal	metabolism
			tissues;	
			shorter (12 amino	
			acids omitted from	
			α-subunit by	
			skipping exon 11)	
IR-B	Transmembrane $\alpha_2\beta_2$	insulin	Longer IR isoform;	glucose storage
	tyrosine kinase		found in	and oxidation,
			metabolically	lipid storage,
			responsive adult	protein synthesis,
			tissues (fat, liver,	regulation of
** *	<b>—</b> 1 0	105.1	muscle)	gene expression
Hybrids	Transmembrane $\alpha_2\beta_2$	IGF-I	Random assembly	cell survival,
	tyrosine kinase		of an IR	mitogenesis,
			hemireceptor	transformation
			combined with an	
			IGF-1R	
			hemireceptor	

Table 3. Summary of IGF and insulin receptors. \*Ligands listed refer to high-affinity binding only.

Overexpression of IGF-1R, IR, and hybrid receptors have all been found in cancers, such as cancers of the breast<sup>74</sup> and thyroid<sup>75</sup>. In fact, IGF-I-

316

stimulated growth of thyroid papillary carcinoma cells in culture was attenuated by antibodies specifically targeting either IGF-1R or hybrid receptors<sup>75</sup>. Thus, the hybrid receptors are not merely structural errors from receptor over-expression, but serve growth-promoting functions in the malignant cells.

#### **3.4 IGF/Receptor signaling**

Ligand binding to the IGF-1R (and IR) causes autocatalytic phosphorylation of the receptor's tyrosine kinase domain, which also phosphorylates additional IGF-1R (and IR) tyrosine residues important for the recruitment of adapter molecules like IRS and Shc. These in turn activate kinase cascades, primarily the PI3 kinase/Akt (PKB) pathway and the MAP kinase pathway, respectively, and lead to signal transduction to the nucleus and mitochondrion<sup>76</sup>.



*Figure 1*. Activation of Akt by IGF/IGF-1R signaling stimulates cell survival. (Modified from ref. 76).

Enhanced IGF/IGF-1R signaling can contribute to each of the four stages of cancer progression<sup>76</sup>. In the first stage, malignant transformation, a cell must acquire the ability to both advantageously proliferate and escape the

body's protective mechanisms, *i.e.* cell cycle arrests that regulate cell growth and apoptosis, or cell suicide, that aborts any aberrant cells. Stimulation by growth factors, like IGF-I, is required for cell cycle entry and progression to the late  $G_1$  phase restriction point, beyond which the cell is committed to completing a round of cell division. IGF-I stimulates the G<sub>1</sub> phase progression by inducing cyclin D1 for assembly with cyclin-dependent kinase (CDK)4<sup>77,78</sup>. IGF-I can further stimulate survival and proliferation through multiple branch points of its signaling pathways, one of which is summarized here and in Figures 1 and 2. As shown in Figure 1, IGF-1R binding by IGF leads to phosphorylation and activation of Akt via PI3 kinase<sup>79</sup>. Activated Akt, itself a kinase, phosphorylates multiple substrates including the Bcl-family member, Bad, and the forkhead transcription factor, FKHRL1. When phosphorylated, Bad and FKHRL1 are sequestered in the cytoplasm by binding to 14-3-3 proteins. Akt also phosphorylates caspase 9, directly inhibiting its function, and IKKa, thereby activating NFkB for nuclear localization and transcriptional activation of survival genes such as *c-myc*. Conversely, the consequences of Akt inactivation, when IGF/IGF-1R signaling is absent, are shown in Figure 2. Unphosphorylated Bad localizes to the mitochondrion, where it binds  $Bcl-X_L$  and releases cytochrome c to the apoptosome, thereby activating caspase 9. Similarly, unphosphorylated FKHRL1 localizes to the nucleus, where it transcriptionally activates numerous genes including IGFBP-1 and FasL. FasL is the ligand (hence, the name) of Fas, the membrane-bound death receptor that activates caspase 8 via the adapter molecule, FADD. Thus, both the mitochondrial and cytoplasmic caspase cascades are activated and converge on activating the execution caspases to complete apoptosis. In short, activation of Akt by IGF/IGF-1R accomplishes both, stimulating cell survival while avoiding apoptosis76,80,81.

The second stage of cancer progression involves additional adaptations that enable continued growth of the clonally expanded, transformed cell as a bulky tumor, wherein nutrient delivery may become restrictive. As shown in  $colon^{82,83}$ ,  $lung^{84}$ , and thyroid cancers<sup>85</sup>, IGF-I induces the angiogenesis agent, vascular endothelial growth factor (VEGF), via increased synthesis of the HIF1 $\alpha$  transcription factor. Colon cancer cells harboring a dominant-negative truncated IGF-1R, when injected into nude mice, developed smaller tumors, reduced VEGF expression, lower tumor vessel count, and decreased pericyte coverage of endothelial cells<sup>86</sup>.



Figure 2. IGF-1R and Akt inactivation lead to apoptosis. (Modified from ref. 76.)

IGF-1R signaling further contributes to cell motility and cell environment interactions that are important for local invasion and the metastatic process. Adherence junctions connect epithelial cells into a normally growing sheet and consist of a core (transmembrane E-cadherin plus cytoplasmic  $\alpha$ -,  $\beta$ -, and  $\gamma$ -catenins) that is coupled to microfilaments via  $\alpha$ -catenin, either directly or indirectly through  $\alpha$ -actinin and vinculin<sup>87</sup>. In MCF-7 breast cancer cells, IGF-1R signaling was shown to promote cell separation, via disassembly of the adherence junctions and redistribution of  $\alpha$ -actinin, actin, and fascin into motile apicolateral spikes, as well as to cell migration, via reassembly of stress fibers, development of long membrane protrusions, and stimulation of myosin light chain kinase<sup>88</sup>. IGF-1R signaling also relocalizes integrins to the leading edge of migrating cells. Conversely, integrins, heterodimers that bind extracellular matrix molecules and transduce signals to the intracellular environment, can modulate IGF-1R signaling when they are activated by their own ligand binding<sup>89</sup>. While inducing cellular changes necessary for motility, IGF-1R signaling can help create a suitable microenvironment for the migrating cell by inducing expression of proteases like cathepsin D<sup>90</sup>, matrix metalloproteinases<sup>91,92</sup>, and urokinase plasminogen activator<sup>93</sup>. These proteases not only dissolve basement membranes in the path of the migrating cell, but they can also cleave IGFBP-3, thereby releasing any bound IGF in the microenvironment for further cell stimulation<sup>94</sup>. Transfection of IGF-1R increased cell spreading on fibronectin, colony formation in soft agar, and metastatic behavior of M-27 Lewis lung carcinoma cells, which are usually poorly invasive and express low numbers of IGF-1R; use of site-directed mutants enabled further delineation of the various IGF-1R functions to different subsets of the normal tyrosine phosphorylation sites<sup>95</sup>.

During the last stage, many cancers acquire resistance to therapeutic agents designed to kill rapidly proliferating cells. Conditions of increased IGF signaling led to increased resistance of a variety of cancer cell lines to such agents in *in vitro* experiments. These include resistance of breast<sup>96</sup> and colorectal cancer<sup>97</sup> cells to irradiation; breast cancer cells to herceptin<sup>98</sup>, doxorubicin, and taxol<sup>99</sup>; colorectal cancer cells to 5-fluorouracil<sup>97</sup>; small cell lung cancer cells to etoposide<sup>100</sup>; pancreatic cancer cells to COX-2 inhibitors<sup>101</sup>; rhabdomyosarcoma cells to all-trans-retinoic acid (ATRA)<sup>104</sup>; and multiple myeloma<sup>105</sup> and thyroid cancer cells<sup>106</sup> to Apo2L/tumor necrosis factor-related apoptosis-inducing ligand (TRAIL).

## 3.5 IGFBPs

Another major mechanism for enhanced cellular survival and proliferation is the reduction of IGFBPs, either through decreased expression or increased proteolysis. Because the binding affinity of IGF for the IGFBPs exceeds that for IGF-1R, IGFBPs tend to reduce IGF/IGF-1R signaling through competitive binding. This has been demonstrated by numerous experiments in different cell types using Des-(1-3)-IGF-I, an IGF-I analog that binds IGF-1R and stimulates DNA synthesis but cannot bind IGFBP- $3^{107-109}$ . As the principal endocrine IGFBP, IGFBP-3 is the best studied and will be the focus of this discussion.

In addition to inhibiting growth by preventing IGF from binding IGF-1R, IGFBP-3 can also inhibit growth and induce apoptosis in an IGFindependent manner<sup>68</sup>. This has been demonstrated by experiments using IGF-free systems,  $IGF-1R^{-\prime}$  cells and rhIGFBP-3 fragments with negligible binding affinity for IGF-I<sup>68,110,111</sup>. IGF-independent actions of IGFBP-3 are presumed to occur via IGFBP-3-specific receptors, which were first detected on the surface of breast cancer cells<sup>112,113</sup> and involve the midregion of the IGFBP-3 molecule<sup>114</sup>. The exact signal transduction pathway of IGFBP-3mediated apoptosis is still unclear, but proposed mechanisms include: direct inhibition of IGF-1R<sup>115</sup>, increased intracellular calcium<sup>116</sup>, nuclear

translocation and RXR-binding<sup>117-120</sup>, changes in BcL family members<sup>121</sup>, and mitochondrial actions<sup>122</sup>.

Thus, a cell can increase its IGF bioavailability, enhance its growth and reduce apoptosis by decreasing the amount of IGFBP-3<sup>110</sup>. Changes in levels of the other IGFBPs, both up and down, have also been found in a variety of cancers, including prostate<sup>94</sup>, lung<sup>123</sup>, and adrenocortical tumors<sup>57</sup>. Although generally thought to be growth inhibitory, IGFBPs have been shown to be growth stimulatory under certain experimental conditions. By sequestering IGF from the IGF-1R, the IGFBPs may allow local accumulation of IGF without down-regulation of the IGF-1R; when this local pool is simultaneously released through IGFBP proteolysis into low-affinity IGFbinding fragments, a greater amount of IGF can bind the IGF-1R for a larger growth stimulus than had the IGF accumulated in a bioavailable form all along<sup>68</sup>. Hence, the literature contains conflicting reports of increased IGFBP levels as growth inhibitory or growth stimulatory<sup>124</sup>, and further studies are needed. The growth stimulatory effect of IGF presentation is illustrated in Figure 1, and the growth inhibition, via inhibition of IGF-1R signaling or via IGFBP-3-receptor-mediated pathways, is illustrated in Figure 2.

#### **3.6 IGFBP proteases**

As mentioned, IGFBP proteases play a critical role in modulating the local bioavailability of IGF for binding IGF-1R; IGF binding affinity is greatest for the intact IGFBP, less for IGF-1R, and less still for IGFBP fragments. Prostate-specific antigen (PSA), whose serum level correlates with prostate cancer volume, was the first IGFBP protease biochemically identified.<sup>125,126</sup> The current list falls into three major categories: kallikreinlike serine proteases, matrix metalloproteinases, and cathepsins<sup>68</sup>. IGFBP proteolysis can be enhanced through increased protease expression by through acidic cancers protease activation by the tumor's or microenvironment<sup>68,127</sup>.

Interestingly, IGFBP proteolysis also occurs at both the local and endocrine levels. Although the PSA elevation in prostate cancer patients' serum negatively correlates with the circulating level of intact IGFBP-3 (and positively with circulating IGFBP-2 levels)<sup>128</sup>, the pattern of IGFBP-3 cleavage fragments in serum differs from that created by seminal PSA, so additional protease(s) must be involved<sup>68</sup>. The frequency of supranormal plasma IGFBP-3 proteolysis was also found to increase with increasing stage of primary breast cancers<sup>129</sup>.

## 4. INDIRECT CONTRIBUTIONS

Cancer is a multigenetic phenomenon, and changes in the GH/IGF axis must be considered within the larger context of cellular signaling. Primary changes in the GH/IGF axis may impact the activities of other signaling pathways, and conversely, primary changes in other pathways may modify GH/IGF axis function.

## 4.1 GH

Although the somatomedin hypothesis identified IGF-I as the principal signaling target of GH, GH has been found to induce other downstream signaling molecules as well. These include: hepatocyte growth factor in the liver<sup>130</sup>, epidermal growth factor (EGF) in the kidney<sup>131</sup>, basic fibroblast growth factor in chondrocytes<sup>132</sup>, interleukin-6 in osteoblasts<sup>133</sup>, bone morphogenetic proteins 2 and 4 in fibroblasts<sup>134</sup>, interleukin-1 $\alpha$  and 1 $\beta$  in the thymus<sup>135</sup>, and preadipocyte factor-1 in beta cells<sup>136</sup>. Whether GH contributes to the neoplastic process through downstream signaling molecules outside the IGF axis is yet unknown. In addition, GH was shown to induce phosphorylation of the EGF receptor (EGFR)<sup>137</sup>; because phosphorylation slowed the rate of EGF-induced EGFR intracellular redistribution and degradation, GH cotreatment potentiated EGF-induced EGFR signaling<sup>138</sup>. EGFR over-expression is one of the most common genetic lesions found in esophageal cancer<sup>139</sup>, and IGFBP-3 induction was recently identified as a consequence of this over-expression<sup>140</sup>. Cross-talk between the GH/IGF axis and EGF/EGFR pathways in the pathogenesis of esophageal cancer remains to be elucidated.

## 4.2 IGF-IGF-1R signaling

IGF/IGF-1R signaling may contribute to cancer through interactions with oncogenes and other mitogens. For example, although IGF-1R was insufficient for growth in soft agar, it was required for malignant transformation by the SV40 large T antigen; stable transfection with SV40 large T antigen led to colony formation by  $IGF-1R^{+/+}$  but not  $IGF-1R^{-/-}$  cells<sup>141</sup>. Signal cooperation has also been shown between IGF-1 and hepatocyte growth factor-scatter factor (HGF-SF) in hepatocellular carcinoma<sup>142</sup> and granulocyte-monocyte-colony-stimulating factor (GM-CSF) in acute myeloid leukemic cells<sup>143</sup>. Likewise, the IGF axis can interact with sex steroid systems to affect cancer development and behavior. This has

been studied most for estrogen in breast cancer and androgens in prostate cancer. The interactions are bi-directional and frequently stage specific. Estrogen and IGF-I were synergistic in stimulating MCF-7 breast cancer cell proliferation; the combination enhanced G<sub>1</sub> phase progression through complementary regulation of p21, cyclin D1-CDK4 and cyclin E-CDK2 complexes<sup>144,145</sup>. IGF1R overexpression has contributed to continued growth of estrogen-deprived breast cancer cells<sup>146</sup> and androgen-independent prostate cancer cells<sup>147</sup>. However, IGF-1R expression is lost as prostate cancers progress to metastases, an effect likely mediated by WT-1(Wilm's tumor gene product)<sup>148</sup>. One of the proposed mechanisms by which tamoxifen is beneficial for breast cancer treatment, beyond its estrogen inhibitory activity, is its reduction of serum IGF-I concentrations<sup>149,150</sup>.

Many tumor suppressors, on the other hand, function at least in part by inhibiting IGF action. This may occur at the transcriptional or functional levels. Transcription of IGF-1R is repressed by WT-1<sup>151</sup> and the tumor suppressor p53<sup>152</sup>. p53 also represses transcription of IGF-II<sup>153</sup>, but stimulates transcription of IGFBP-3<sup>154</sup> (see Section 4.3). Other tumor suppressors can inhibit IGF action at the signaling level, without affecting transcription levels. PTEN (phosphatase and tensin homolog) is a phosphatase that dephosphorylates Akt, thereby inhibiting one of IGF's major signaling pathways<sup>155,156</sup>. Germline *PTEN* mutations have been detected in Cowden syndrome, characterized by multiorgan hamartomas and increased cancer risk, and allelic losses and somatic mutations at the PTEN locus are frequently found in cancers, especially in later stages as they become more aggressive<sup>157,158</sup>. The von Hippel Lindau gene product (VHL) leads to reduced VEGF production via ubiquitin-mediated degradation of HIF-1 $\alpha^{159}$ . VHL was shown to also directly interact with protein kinase C- $\delta$ , causing its dissociation from IGF-1R and inhibition of IGF-mediated invasiveness<sup>160</sup>. VHL is important in the pathogenesis of renal cell carcinoma, and its germline mutation causes a dominantly inherited familial cancer syndrome<sup>161</sup>.

### 4.3 IGFBP-3

As mentioned in the previous section, IGFBP-3 is transcriptionally induced by the tumor suppressor  $p53^{154}$ . In fact, IGFBP-3 was shown to mediate p53-induced apoptosis during serum starvation<sup>162</sup>, and multiple *p53* mutants that lost the ability to induce IGFBP-3 and Bax, but not p21, were unable to induce apoptosis<sup>163,164</sup>. IGFBP-3 is also induced by cytokines, retinoic acid, DNA damage (both irradiation and drug-induced), and hypoxia<sup>110,165</sup>. By inducing IGFBP-3 and repressing both IGF-1R/IGF-II,
p53 switches the IGF axis balance from growth stimulation to growth inhibition and apoptosis. p53 is the most frequently mutated gene among all human cancers, and its germline mutation is associated with Li-Fraumeni Syndrome, another dominant familial cancer syndrome<sup>110</sup>.

Conversely, oncogenes can affect the IGF axis by inhibiting IGFBP-3 function. For example, oncogenic H-*ras* caused resistance to the growth-inhibitory effects of IGFBP-3 in breast cancer cells<sup>166</sup>. E7, a product of human papillomaviruses associated with cervical cancer, was shown to bind IGFBP-3, leading to IGFBP-3 proteolysis and inhibition of IGFBP-3-induced apoptosis<sup>167</sup>.

#### 5. CLINICAL IMPLICATIONS

#### 5.1 rhGH treatment

Clinical practice with growth-promoting therapies has already been affected by the concern of possible cancer-promoting effects of GH and IGF-I. GH deficiency frequently develops as a consequence of irradiation or chemotherapy, and for intracranial neoplasms, from the brain tumor itself or from tumor resection<sup>168</sup>. Because the greatest risk of cancer recurrence is in the first year after treatment, current practice in many centers is to defer rhGH therapy until the patient is at least one year tumor-free. Yet when carefully studied, the evidence indicated that rhGH did not increase recurrence rates<sup>169</sup>.

rhGH is currently accepted as a safe therapy, but there are two important considerations as we look to the future. First, rhGH therapy is evolving from limited physiologic replacement to increasingly pharmacologic use, in terms of escalating doses and additional indications<sup>170</sup>. Its safety profile must be reevaluated in this new context. To guide continued safety, IGF-I levels in all rhGH recipients should be closely monitored and rhGH dose titrated to avoid supraphysiologic IGF-I levels.

Second, the follow-up study of the United Kingdom cadaveric pituitary GH recipients bears mentioning<sup>6</sup>. The significance of the results for current rhGH recipients is unclear. It is difficult to generalize from a sample size of two cases, and change in product makes comparisons dubious. Dosing schedule for rhGH is different than that for pituitary GH, and infectious risks have been eliminated by the switch to recombinant technology<sup>171</sup>. Just as Creutzfeldt-Jakob disease painfully exposed the incomplete purity of the pituitary GH, perhaps viral oncogenes were also transmitted? Yet the study pointed out a very important concept: because cancer incidence is greatest in

older ages, ongoing surveillance of rhGH recipients into late adulthood is needed to fully assess the long-term safety.

#### 5.2 Cancer prevention

The ultimate goal of understanding how the GH/IGF axis may contribute to cancer is to devise ways of manipulating the system to prevent or treat cancer. Calorie restriction has long been recognized as an effective means of reducing circulating IGF-I levels. In cancer-susceptible mouse models, calorie restriction has been shown to not only lower IGF-I levels, but also to delay spontaneous tumor development and suppress carcinogen-induced tumor progression<sup>172,173</sup>. For example, calorie restriction of p53haploinsufficient mice suppressed progression of p-cresidine induced bladder tumors, but not when the circulating IGF-I levels were restored by pump infusion<sup>174</sup>. In human epidemiologic studies, overweight has been the most reproducible cancer risk factor, so now the focus has shifted onto more prevention-oriented studies; physical exercise has been shown protective for breast and colon cancers<sup>175</sup>, and several studies are trying to dissect out specific dietary components as particularly risk-affecting<sup>176-178</sup>. The burgeoning obesity epidemic throws urgency and keen interest upon this area of investigation, and opens up a new line of query as well: since the insulin-IR system is so closely related to the IGF axis and hyperinsulinemia is a frequent complication of obesity, the possible contributions of insulin to cancer remain to be explored.

#### 5.3 Cancer treatment

Forays into GH/IGF axis modulation as a means of treating cancer have fallen into two general categories: dampening the whole GH axis and more specifically targeting IGF/IGF-1R signaling. Table 4 summarizes the data from experiments using upstream targets<sup>179-184</sup>. To date, approaches for specifically inhibiting IGF-1R signaling include adenoviral dominant negative IGF-1R<sup>185</sup>, IGF-1R antibodies<sup>186</sup>, IGF-1R antisense<sup>187</sup>, IGF-1R signals, and IGF-1 antisense<sup>189</sup>. The clinical efficacy of IGF-1R inhibition is yet unknown, as are the potential toxicities from inhibiting normal IGF-1R or cross-reactivity with IR. Thus this construes a hot area for on-going research.

GH antagonist	Model system	Effect	Ref		
GH-R antag	Meningiomas	Deer tumor volume and tumor weight	170		
(pegvisomant)	xenografted into	Serum [ICE.]] and [BP.3] decr: [BP.1]			
	athymic mice:	and [BP-4] incr			
	animals treated X Sucks	No difference in tumor [IGE-1]			
-+Transgenic	DMBA induced	Transgenia mice smaller			
mice with GH-		deer UCE II. Deer tumer incidence			
antag	mammary tumors	deci [IGF-I]. Deci tumor incidence.			
GH-RH antag	Caki-I (renal adenoCA)	Mice: decr tumor volume; decr tumor	181		
(MZ-4-71)	Tissue cx and	weight. Decr serum [GH] and [IGF-I].			
	xenografted into	Also decr [IGF-I] in liver and tumor;			
	nude mice	decr [IGF-II] in tumor.			
		Cx: inhibit growth, but only at high			
		concentrations.			
GH-RH antags	HT-29 (colon CA):	Mice: decr tumor volume and weight by	182		
(MZ-4-71,	Tissue cx and	all 3 agents.			
MZ5-156 and	xenografted into	Decr cell prolif, incr apoptosis;			
JV-1-36)	nude mice	decr tumor IGF-II.			
		No change in serum [IGF-I] or [IGF-II].			
		Cx: MZ-5-156 dose-dependently decr			
		IGF-II and cell prolif.			
GH-RH antag	PC-3 (prostate CA):	Mice: tumor volume decr 49% by JV-1-	183		
(JV-1-38);	Tissue cx and	38, 30% (NS) by RC-160, 63% by			
somatostatin	xenografted into	combination.			
analog	nude mice	Decr serum [IGF-I] by RC-160 only;			
(RC-160)		decr tumor IGF-II mRNA by both; decr			
		VEGF by both.			
		Cx: cell prolif inhibited by JV-1-38, not			
		RC-160 alone but moreso in			
		combination.			
GH-RH antag	MNNG/HOS	Mice: tumor volume and weight decr for			
(JV-1-38)	(osteosarcoma) and	both, incr tumor doubling time.			
	SK-ES-1 (Ewing's	Decr [IGF-I] in serum and liver mRNA;			
	sarcoma): Tissue cx	Decr [IGF-II] and IGF-II mRNA in both			
	and xenografted into	tumors.			
	nude mice	Cx: inhibited prolif of both cell lines.			

*Table 4.* GH antagonism to reduce IGF-I and inhibit cancer.Antag, antagonist; cx, tissue culture; decr, decrease; incr, increase; prolif, proliferation.

#### ACKNOWLEDGEMENTS

I would like to thank the National Institutes of Health for financial support (Grant K08 DK64352).

#### Cancer

#### REFERENCES

- S. Yakar, J.L. Liu, B. Stannard, A. Butler, D. Accili, B. Sauer and D. LeRoith, Normal growth and development in the absence of hepatic insulin-like growth factor I. *Proc Natl Acad Sci USA*, 96,7324-7329 (1999).
- 2. S. Watanabe, Y. Tsunematsu, J. Fujimoto, *et al*, Leukemia in patients treated with growth hormone. *Lancet*, **1**,1159-1160 (1988).
- 3. S.L. Blethen, Complications of growth hormone therapy in children. *Curr Opin Pediat*, **7**, 466-471 (1995).
- D.B. Allen, Safety of human growth hormone therapy: current topics. J Pediat, 128, S8-S13 (1996).
- G.A. Tuffli, A. Johanson, A.C. Rundle and D.B. Allen, Lack of increased risk for extracranial, nonleukemic neoplasms in recipients of recominant deoxyribonucleic acid growth hormone. *J Clin Endocrinol Metab*, **80**,1416-1422 (1995).
- 6. A.J. Swerdlow, C.D. Higgins, P. Adlard and M.A. Preece, Risk of cancer in patients treated with human pituitary growth hormone in the UK, 1959-85: a cohort study. *Lancet*, **360**, 273-277 (2002).
- T. Rosen and B.A. Bengtsson, Premature mortality due to cardiovascular disease in hypopituitarism. *Lancet*, 336, 285-288 (1990).
- S. Melmed, Acromegaly and cancer: not a problem? J Clin Endocrinol Metab, 86, 2929-2934 (2001).
- P.J. Jenkins and M. Besser, Clinical perspective: acromegaly and cancer: a problem. J Clin Endocrinol Metab, 86, 2935-2941 (2001).
- 10. P.J. Jenkins, Acromegaly and colon cancer. *Growth Horm Igf Res*, **10**, S35-S36 (2000).
- 11. P.J. Jenkins, M.J. Veysey, L.A. Thomas, et al. Increased serum deoxycholic acid levels in acromegalic patients with colorectal neoplasia. *J Enocrinol*, **152**, 74 (1997).
- A.G. Renehan, P. Bhaskar, J.E. Painter, S.T. O'Dwyer, N. Haboubi, J. Varma, S.G. Bal, and S.M. Shalet, The prevalence and characteristics of colorectal neoplasia in acromegaly. *J Clin Endocrinol Metab*, 85,3417-3424 (2000).
- L.S. Mathews, R.E. Hammer, R.R. Behringer, A.J. D'Ercole, G.I. Bell, R.L. Brinster and R.D. Palmiter. Growth enhancement of transgenic mice expressing human insulin-like growth factor I. *Endocrinology*, **123**,2827-2833 (1988).
- C.J. Quaife, L.S. Mathews, C.A. Pinkert, R.E. Hammer, R.L. Brinster and R.D. Palmiter, Histopathology associated with elevated levels of growth hormone and insulin-like growth factor I in transgenic mice. *Endocrinology*, **124**, 40-48 (1989).
- M.H. Ulshen, R.H. Dowling, C.R. Fuller, E.M. Zimmermann and P.K. Lund, Enhanced growth of small bowel in transgenic mice overexpressing bovine growth hormone. *Gastroenterology*, **104**,973-980 (1993).
- K. Ohneda, M.H. Ulshen, C.R. Fuller, A.J. D'Ercole and P.K. Lund. Enhanced growth of small bowel in transgenic mice expressing human insulin-like growth factor I. *Gastroenterology*, **112**,444-454 (1997).
- M.J. Veysey, L.A. Thomas, A.I. Mallet, P.J. Jenkins, G.M. Besser, G.M. Murphy and R.H. Dowling, Colonic transit influences deoxycholic acid kinetics. *Gastroenterology* 121, 812-822 (2001).
- T. Ochsenkuhn, E. Bayerdorffer, A. Meining, M. Schinkel, C. Thiede, V. Nussler, M. Sackmann, R. Hatz, A. Neubauer and G. Paumgartner, Colonic mucosal proliferation is related to serum deoxycholic acid levels. *Cancer*, 85, 1664-1669 (1999).

- 19. V. Milovic, I.C. Teller, G.M. Murphy, W.F. Caspary and J. Stein, Deoxycholic acid stimulates migration in colon cancer cells. *Eur J Gastroenterol Hepatol*, **13**, 945-949 (2001).
- D Qiao, S.V. Gaitonde, W. Qi and J.D. Martinez, Deoxycholic acid suppresses p53 by stimulating proteasome-mediated p53 protein degradation. *Carcinogenesi*, 22, 957-964 (2001).
- D.F. Romagnolo, R.B. Chirnomas, J. Ku, B.D. Jeffy, C.M. Payne, H. Holubec, L. Ramsey, H. Bernstein, C. Bernstein, K. Kunke, A. Bhattacharyya, J. Warneke and H. Garewal, Deoxycholate, an endogenous tumor promoter and DNA damaging agent, modulates BRCA-1 expression in apoptosis-sensitive epithelial cells: loss of BRCA-1 expression in colonic adenocarcinomas. *Nutri Cancer*, 46, 82-92 (2003).
- A. Colao, P. Marzullo, D. Ferone, S. Spiezia, G. Cerbone, V. Marino, A. Di Sarno,
  B. Merola and G. Lombardi, Prostatic hyperplasia: an unknown feature of acromegaly. *J Clin Endocrinol Metab*, 83,775-779 (1998).
- A. Grimberg and P. Cohen, GH and prostate cancer: Guilty by association? J Endocrinol Invest, 22, 64-67 (1999).
- L. Le Marchand, T. Donlon, A. Seifried, R. Kaaks, S. Rinaldi and L.R. Wilkens, Association of a common polymorphism in the human GH1 gene with colorectal neoplasia. J Natl Cancer Inst, 94, 454-460 (2002).
- 25. S.M. Swanson and T.G. Unterman, The growth hormone-deficient spontaneous dwarf rat is resistant to chemically induced mammary carcinogenesis. *Carcinogenesis*, **23**, 977-982 (2002).
- M.M. Ramsey, R.L. Ingram, A.B. Cashion, A.H. Ng, J.M. Cline, A.F. Parlow and W.E. Sonntag, Growth hormone-deficient dwarf animals are resistant to dimethylbenzanthracine (DMBA)-induced mammary carcinogenesis. *Endocrinology*, 143, 4139-4142 (2002).
- M. Pollak, Insulin-like growth factor physiology and cancer risk. *Europ J Cancer*, 36, 1224-1228 (2000).
- J.M Chan, M.J. Stampfer, E. Giovannucci, P.H. Gann, J. Ma, P. Wilkinson, C.H. Henneken, and M. Pollak, Plasma insulin-like growth factor-I and prostate cancer risk: a prospective study. *Science*, 279, 563-566 (1998).
- A. Wolk, C.S. Mantzoros, S.O. Andersson, R. Bergstrom, L.B. Signorello, P. Lagiou, H.O. Adami and D. Trichopoulos, Insulin-like growth factor 1 and prostate cancer risk: a population-based, case-control study. *J Natl Cancer Inst*, **90**, 911-915 (1998).
- P. Stattin, A. Bylund, S. Rinaldi, C. Biessy, H. Dechaud, U.H. Stenman, L. Egevad, E. Riboli, G. Hallmans and R. Kaaks, Plasma insulin-like growth factor-I, insulinlike growth factor-binding proteins, and prostate cancer risk: a prospective study. J Natl Cancer Instit, 92,1910-1917 (2000).
- 31. T. Shaneyfelt, R. Husein, G. Bubley and C.S. Mantzoros, Hormonal predictors of prostate cancer: a meta-analysis. *J Clin Oncol* **18**, 847-853 (2000).
- 32. R. Shi, H.J. Berkel and H. Yu, Insulin-like growth factor-I and prostate cancer: a meta-analysis. *Brit J Cancer*, **85**, 991-996 (2001).
- 33. J.M. Chan, M.J. Stampfer, J. Ma, P. Gann, J.M. Gaziano, M. Pollak and E. Giovannucci, Insulin-like growth factor-I (IGF-I) and IGF binding protein-3 as predictors of advanced-stage prostate cancer. *J Natl Cancer Inst*, **94**, 1099-1106 (2002).
- S.F. Shariat, D.J. Lamb, M.W. Kattan, C. Nguyen, J. Kim, J. Beck, T.M. Wheeler and K.M. Slawin, Association of preoperative plasma levels of insulin-like growth

328

factor I and insulin-like growth factor binding proteins-2 and -3 with prostate cancer invasion, progression, and metastasis. *J Clin Oncol*, **20**, 833-841 (2002).

- J. Ma, M. Pollak, E. Giovannucci, J.M. Chan, Y. Tao, C.H. Hennekens, and M.J. Stampfer, Prospective study of colorectal cancer risk in men and plasma levels of insulin-like growth factor (IGF)-I and IGF-binding protein-3. J Natl Cancer Inst, 91, 620-625 (1999).
- R. Kaaks, P. Toniolo, A. Akhmedkhanov, A. Lukanova, C. Biessy, H. Dechaud, S. Rinaldi, A. Zeleniuch-Jacquotte, R.E. Shore and E. Riboli, Serum C-peptide, insulin-like growth factor (IGF)-I, IGF-binding proteins, and colorectal cancer risk in women. J Natl Cancer Inst, 92, 1592-1600 (2000).
- A.G. Renehan, J. Jones, C.S. Potten, S.M. Shalet and S.T. O'Dwyer, Elevated serum insulin-like growth factor (IGF)-II and IGF binding protein-2 in patients with colorectal cancer. *Brit J Cancer*, 83, 1344-1350 (2000).
- A.G. Renehan, J.E. Painter, D. O'Halloran, W.S. Atkin, C.S. Potten, S.T. O'Dwyer and S.M. Shalet, Circulating insulin-like growth factor II and colorectal adenomas. *J Clin Endocrinol Metab*, 85, 3402-3408 (2000).
- C.M. Franciosi, M.G. Piacentini, M. Conti, F. Romano, F. Musco, R. Caprotti, F. Rovelli and F. Uggeri, IGF-1 and IGF-1BP3 in gastric adenocarcinoma. Preliminary study. *Hepato-Gastroenterol*, **50**, 297-300 (2003).
- H. Yu, M.R. Spitz, J. Mistry, J. Gu, W.K. Hong and X. Wu, Plasma levels of insulin-like growth factor-I and lung cancer risk: a case-control analysis. J Natl Cancer Inst, 91, 151-156 (1999).
- A. Lukanova, P. Toniolo, A. Akhmedkhanov, C. Biessy, N.J. Haley, R.E. Shore, E. Riboli, S. Rinaldi and R. Kaaks, A prospective study of insulin-like growth factor-I, IGF-binding proteins-1, -2 and -3 and lung cancer risk in women. *Int J Cancer*, 92, 888-892 (2001).
- S.J. London, J.M. Yuan, G.S. Travlos, Y.T. Gao, R.E. Wilson, R.K. Ross and M.C. Yu, Insulin-like growth factor I, IGF-binding protein 3, and lung cancer risk in a prospective study of men in China. J Natl Cancer Inst, 94, 749-754 (2002).
- S.E. Hankinson, W.C. Willett, G.A. Colditz, D.J. Hunter, D.S. Michaud, B. Deroo, B. Rosner, F.E. Speizer and M. Pollak, Circulating concentrations of insulin-like growth factor-I and risk of breast cancer. *Lancet*, **351**, 1393-1396 (1998).
- 44. E.H. Ng, C.Y. Ji, P.H. Tan, V. Lin, K.C. Soo and K.O. Lee, Altered serum levels of insulin-like growth-factor binding proteins in breast cancer patients. *Ann Surg Onco*, **5**,194-201 (1998).
- B.D. Li, M.J. Khosravi, H.J. Berkel, A. Diamandi, M.A. Dayton, M. Smith and H. Yu, Free insulin-like growth factor-I and breast cancer risk. *Int J Cancer*, **91**, 736-739 (2001).
- 46. L. Keinan-Boker, H.B. Bueno De Mesquita, R. Kaaks, C.H. Van Gils, P.A. Van Noord, S. Rinaldi, E. Riboli, J.C. Seidell, D.E. Grobbee and P.H. Peeters, Circulating levels of insulin-like growth factor I, its binding proteins -1,-2, -3, C-peptide and risk of postmenopausal breast cancer. *Int J Cancer*, **106**, 90-95 (2003).
- 47. H. Yu, B.D. Li, M. Smith, R. Shi, H.J. Berkel and I. Kato, Polymorphic CA repeats in the IGF-I gene and breast cancer. *Breast Cancer Res Treat*, **70**, 117-122 (2001).
- A. Lukanova, E. Lundin, P. Toniolo, A. Micheli, A. Akhmedkhanov, S. Rinaldi, P. Muti, P. Lenner, C. Biessy, V. Krogh, A. Zeleniuch-Jacquotte, F. Berrino, G. Hallmans, E. Riboli, and R. Kaaks, Circulating levels of insulin-like growth factor-I and risk of ovarian cancer. *Internatl J Cancer*, 101, 549-554 (2002).

- E. Petridou, P. Koukoulomatis, D.M. Alexe, Z. Voulgaris, E. Spanos and D. Trichopoulos, Endometrial cancer and the IGF system: a case-control study in Greece. Oncology, 64, 341-345 (2003).
- H. Zhao, H.B. Grossman, M.R. Spitz, S.P. Lerner, K. Zhang and X. Wu, Plasma levels of insulin-like growth factor-1 and binding protein-3, and their association with bladder cancer risk. *J Urol*, 169, 714-717 (2003).
- Y. Wu, S. Yakar, L. Zhao, L. Hennighausen and D. LeRoith, Circulating insulinlike growth factor-I levels regulate colon cancer growth and metastasis. *Cancer Res*, 62, 1030-1035 (2002).
- A.A. Butler, V.A. Blakesley, V. Poulaki, M. Tsokos, T.L. Wood, D. LeRoith and V. Poulaki, Stimulation of tumor growth by recombinant human insulin-like growth factor-I (IGF-I) is dependent on the dose and the level of IGF-I receptor expression. *Cancer Res*, 58, 3021-3027 (1998).
- M. Raccurt, P.E. Lobie, E. Moudilou, T. Garcia-Caballero, L. Frappart, G. Morel and H.C. Mertani, High stromal and epithelial human gh gene expression is associated with proliferative disorders of the mammary gland. *J Endocrinol*, **175**, 307-318 (2002).
- H.C. Mertani, T. Garcia-Caballero, A. Lambert, F. Gerard, C. Palayer, J.M. Boutin, B.K. Vonderhaar, M.J. Waters, P.E. Lobie and G. Morel, Cellular expression of growth hormone and prolactin receptors in human breast disorders. *Int J Cancer*, 79, 202-211 (1998).
- R. Graichen, D. Liu, Y. Sun, K.O. Lee and P.E. Lobie, Autocrine human growth hormone inhibits placental transforming growth factor-beta gene transcription to prevent apoptosis and allow cell cycle progression of human mammary carcinoma cells. J Biol Chem, 277, 26662-26672 (2002).
- E. Maiorano, A. Ciampolillo, G. Viale, P. Maisonneuve, A. Ambrosi, V. Triggiani,
  E. Marra and E. Perlino, Insulin-like growth factor 1 expression in thyroid tumors. *App Immunohistochem Molec Morphol*, 8, 110-119 (2000).
- N. Boulle, A. Logie, C. Gicquel, L. Perin and Y. Le Bouc, Increased levels of insulin-like growth factor II (IGF-II) and IGF-binding protein-2 are associated with malignancy in sporadic adrenocortical tumors. *J Clin Endocrinol Metab*, 83, 1713-1720 (1998).
- 58. P. Rotwein, Structure, evolution, expression and regulation of insulin-like growth factors I and II. *Growth Factors*, **5**, 3-18 (1991).
- Y. Kamai, S. Mikawa, K. Endo, H. Sakai and T. Komano, Regulation of insulin-like growth factor-I expression in mouse preadipocyte Ob1771 cells. *J Biol Chem*, 271, 9883-9886 (1996).
- M. Kassem, R. Okazaki, S.A. Harris, T.C. Spelsberg, C.A. Conover and B.L. Riggs, Estrogen effects on insulin-like growth factor gene expression in a human osteoblastic cell line with high levels of estrogen receptor. *Calcif Tissue Int*, 62, 60-66 (1998).
- 61. T.L. McCarthy, M. Centrella and E. Canalis, Cyclic AMP induces insulin-like growth factor I synthesis in osteoblast-enriched cultures. *J Biol Chem*, **265**, 15353-15356 (1990).
- R. Okazaki, S.K. Durham, B.L. Riggs and C.A. Conover, Transforming growth factor-beta and forskolin increase all classes of insulin-like growth factor-I transcripts in normal human osteoblast-like cells. *Biochem Biophys Res Com*, 207, 963-970 (1995).
- D. Swolin, C. Brantsing, G. Matejka and C. Ohlsson, Cortisol decreases IGF-I mRNA levels in human osteoblast-like cells. *J Endocrinol*, 149, 397-403 (1996).

#### Cancer

- 64. H. Nakagawa, R.B. Chadwick, P. Peltomaki, C. Plass, Y. Nakamura and A. de La Chapelle, Loss of imprinting of the insulin-like growth factor II gene occurs by biallelic methylation in a core region of H19-associated CTCF-binding sites in colorectal cancer. *Proc Natl Acad Sci USA*, **98**, 591-596 (2001).
- 65. C.R. Kaffer, A. Grinberg and K. Pfeifer, Regulatory mechanisms at the mouse Igf2/H19 locus. *Mol Cell Biol*, **21**, 8189-8196 (2001).
- 66. J. DiGiovanni, K. Kiguchi, A. Frijhoff, E. Wilker, D.K. Bol, L. Beltran, S. Moats, A. Ramirez, J. Jorcano and C. Conti, Deregulated expression of insulin-like growth factor 1 in prostate epithelium leads to neoplasia in transgenic mice. *Proc Natl Acad Sci USA*, **97**, 3455-3460 (2000).
- J. DiGiovanni, D.K. Bol, E. Wilker, L. Beltran, S. Carbajal, S. Moats, A. Ramirez, J. Jorcano and K. Kiguchi, Constitutive expression of insulin-like growth factor-1 in epidermal basal cells of transgenic mice leads to spontaneous tumor promotion. *Cancer Res*, 60, 1561-1570 (2000).
- A. Grimberg and P. Cohen, The Role of insulin-like growth factors and their binding proteins in growth control and carcinogenesis. J Cell Physiol, 183, 1-9 (2000).
- A. Ullrich, A. Gray, A.W. Tam, T. Yang-Feng, M. Tsubokawa, C. Collins, W. Henzel, T. Le Bon, S. Kathuria and E. Chen, Insulin-like growth factor I receptor primary structure: comparison with insulin receptor suggests structural determinants that define functional specificity. *EMBO J*, 5, 2503-2512 (1986).
- T. Ludwig, J. Eggenschweiler, P. Fisher, A.J. D'ercole, M.L. Davenport and A. Efstratiadis, Mouse mutants lacking the type 2 IGF receptor (IGF2R) are rescued from perinatal lethality in Igf2 and Igf1r null backgrounds. *Dev Biol*, **177**, 517-535 (1996).
- F. Frasca, G. Pandini, P. Scalia, L. Sciacca, R. Mineo, A. Costantino, I.D. Goldfine, A. Belfiore and R. Vigneri, Insulin receptor isoform A, a newly recognized, highaffinity insulin-like growth factor II receptor in fetal and cancer cells. *Mol Cell Biol*, 19, 3278-3288 (1999).
- G. Pandini, E. Medico, E. Conte, L. Sciacca, R. Vigneri and A. Belfiore, Differential gene expression induced by insulin and insulin-like growth factor-II through the insulin receptor isoform A. *J Biol Chem*, 278, 42178–42189 (2003).
- 73. M.A. Soos, C.E. Field and K Siddle, Purified hybrid insulin/insulin-like growth factor-I receptors bind insulin-like growth factor-I, but not insulin, with high affinity. *Biochem J*, **290**, 419-426 (1993).
- G. Pandini, R. Vigneri, A. Costantino, F. Frasca, A. Ippolito, Y. Fujita-Yamaguchi, K. Siddle, I.D. Goldfine and A. Belfiore, Insulin and insulin-like growth factor-I (IGF-I) receptor overexpression in breast cancers leads to insulin/IGF-I hybrid receptor overexpression: evidence for a second mechanism of IGF-I signaling. *Clin Cancer Res*, 5, 1935-1944 (1999).
- V. Vella, L. Sciacca, G. Pandini, R. Mineo, S. Squatrito, R. Vigneri and A. Belfiore, The IGF system in thyroid cancer: new concepts. *Mol Pathol*, 54, 121-124 (2001).
- 76. A. Grimberg, Mechanisms by which IGF-I may promote cancer. *Cancer Biol Ther*, 2(6),630-5(2003).
- 77. J.A. Diehl, Cycling to cancer with cyclin D1. Cancer Biol Ther, 1, 226-231 (2002).
- 78. B. Dufourny, J. Alblas, H.A. van Teeffelen, F.M. van Schaik, B. van der Burg, P.H. Steenbergh and J.S. Sussenbach, Mitogenic signaling of insulin-like growth factor I in MCF-7 human breast cancer cells requires phosphatidylinositol 3-kinase and is independent of mitogen-activated protein kinase. J Biol Chem, 272, 31163-31171 (1997).

- 79. T.F. Franke, D.R. Kaplan, L.C. Cantley and A. Toker, Direct regulation of the Akt proto-oncogene product by phosphatidylinositol-3,4-bisphosphate. *Science*, **275**, 665-668 (1997).
- 80. S.R. Datta, A. Brunet and M.E. Greenberg, Cellular survival: a play in three Akts. *Genes Dev*, **13**, 2905-2927 (1999).
- T.F. Franke, C.P. Hornik, L. Segev, G.A. Shostak and C. Sugimoto, PI3K/Akt and apoptosis: size matters. *Oncogene*, 22, 8983-8998 (2003).
- R. Fukuda, K. Hirota, F. Fan, Y.D. Jung, L.M. Ellis and G.L Semenza, Insulin-like growth factor 1 induces hypoxia-inducible factor 1-mediated vascular endothelial growth factor expression, which is dependent on MAP kinase and phosphatidylinositol 3-kinase signaling in colon cancer cells. J Biol Chem, 277, 38205-38211 (2002).
- S.A. Bustin, S. Dorudi, S.M. Phillips, R.M. Feakins and P.J. Jenkins, Local expression of insulin-like growth factor-I affects angiogenesis in colorectal cancer. *Tumour Biol*, 23, 130-138 (2002).
- Y. Tang, D. Zhang, L. Fallavollita and P. Brodt, Vascular endothelial growth factor C expression and lymph node metastasis are regulated by the type I insulin-like growth factor receptor. *Cancer Res*, 63, 1166-1171 (2003).
- V. Poulaki, C.S. Mitsiades, C. McMullan, D. Sykoutri, G. Fanourakis, V Kotoula, S. Tseleni-Balafouta, D.A. Koutras and N. Mitsiades, Regulation of vascular endothelial growth factor expression by insulin-like growth factor I in thyroid carcinomas. *J Clin Endocrinol Metab*, 88, 5392-5398 (2003).
- N. Reinmuth, F. Fan, W. Liu, A.A. Parikh, O. Stoeltzing, Y.D. Jung, C.D. Bucana, R. Radinsky, G.E. Gallick and L.M. Ellis, Impact of insulin-like growth factor receptor-I function on angiogenesis, growth, and metastasis of colon cancer. *Lab Invest*, 182, 1377-1389 (2002).
- 87. B.M. Gumbiner, Regulation of cadherin adhesive activity. *J Cell Biol*, **148**, 399-404 (2000).
- M.A. Guvakova, J.C. Adams and D. Boettiger, Functional role of alpha-actinin, PI 3-kinase and MEK1/2 in insulin-like growth factor I receptor kinase regulated motility of human breast carcinoma cells. J Cell Sci, 115, 4149-4165 (2002).
- D.R. Clemmons and L.A. Maile, Minireview: Integral membrane proteins that function coordinately with the insulin-like growth factor I receptor to regulate intracellular signaling. *Endocrinology*, 144, 1664-1670 (2003).
- F. Wang, R. Duan, J. Chirgwin and S.H. Safe, Transcriptional activation of cathepsin D gene expression by growth factors. J Mol Endocrinol, 24, 193-202 (2000).
- A. Yoon and R.A. Hurta, Insulin like growth factor-1 selectively regulates the expression of matrix metalloproteinase-2 in malignant H-ras transformed cells. *Mol Cell Biochem*, 223, 1-6 (2001).
- D. Zhang and P. Brodt, Type 1 insulin-like growth factor regulates MT1-MMP synthesis and tumor invasion via PI 3-kinase/Akt signaling. Oncogene, 22, 974-982 (2003).
- S.E. Dunn, J.V. Torres, J.S. Oh, D.M. Cykert and J.C. Barrett, Up-regulation of urokinase-type plasminogen activator by insulin-like growth factor-I depends upon phosphatidylinositol-3 kinase and mitogen-activated protein kinase kinase. *Cancer Res*, 61, 1367-1374 (2001).
- 94. A. Grimberg, R. Rajah, H. Zhao and P. Cohen, "The prostatic IGF system: New levels of complexity, in: *Molecular Mechanisms to Regulate the Activities of*

Insulin-Like Growth Factors, (K. Takano, N. Hizuka and S.I. Takahashi, eds.) Elsevier Science B.V., Amsterdam, 1998, pp. 205-213.

- P. Brodt, L. Fallavollita, A.M. Khatib, A.A. Samani and D. Zhang, Cooperative regulation of the invasive and metastatic phenotypes by different domains of the type I insulin-like growth factor receptor beta subunit. J Biol Chem, 276, 33608-33615 (2001).
- A.B. Langeland Marthinsen, A. Dybdahl Wanderas, S. Lundgren, T. Strickert, E. Hundal and B.A. Graff, Effects of growth factors on growth and radiation sensitivity of the human breast cancer cell line T-47D. Oncol Rep, 9, 397-403 (2002).
- E.S. Perer, A.K. Madan, A. Shurin, E. Zakris, K. Romeguera, Y. Pang and D.J. Beech, Insulin-like growth factor I receptor antagonism augments response to chemoradiation therapy in colon cancer cells. J Surg Res, 94, 1-5 (2000).
- Y. Lu, X. Zi, Y. Zhao, D. Mascarenhas and M. Pollak, Insulin-like growth factor-I receptor signaling and resistance to trastuzumab (Herceptin). *J Natl Cancer Inst*, 93, 1852-1857 (2001).
- D.J. Beech, N. Parekh and Y. Pang, Insulin-like growth factor-I receptor antagonism results in increased cytotoxicity of breast cancer cells to doxorubicin and taxol. Oncol Rep, 8, 325-329 (2001).
- G.W. Krystal, G. Sulanke and J. Litz, Inhibition of phosphatidylinositol 3-kinase-Akt signaling blocks growth, promotes apoptosis, and enhances sensitivity of small cell lung cancer cells to chemotherapy. *Mol Cancer Ther*, 1, 913-922 (2002).
- R.J. Levitt and M. Pollak, Insulin-like growth factor-I antagonizes the antiproliferative effects of cyclooxygenase-2 inhibitors on BxPC-3 pancreatic cancer cells. *Cancer Res*, 62, 7372-7376 (2002).
- 102. K.N. Thimmaiah, J. Easton, S. Huang, K.A. Veverka, G.S. Germain, F.C. Harwood and P.J. Houghton, Insulin-like growth factor I-mediated protection from rapamycin-induced apoptosis is independent of Ras-Erk1-Erk2 and phosphatidylinositol 3'-kinase-Akt signaling pathways. *Cancer Res*, 63, 364-374 (2003).
- D.J. Beech, E. Perer, J. Helms, A. Gratzer and N. Deng, Insulin-like growth factor-I receptor activation blocks doxorubicin cytotoxicity in sarcoma cells. *Oncol Rep*, 10, 181-184 (2003).
- L.M. Neri, P. Borgatti, P.L. Tazzari, R. Bortul, A. Cappellini, G. Tabellini, A. Bellacosa, S. Capitani and A.M. Martelli, The phosphoinositide 3-kinase/AKT1 pathway involvement in drug and all-trans-retinoic acid resistance of leukemia cells. *Mol Cancer Res*, 1, 234-246 (2003).
- 105. C.S. Mitsiades, N. Mitsiades, V. Poulaki, R. Schlossman, M. Akiyama, D. Chauhan, T. Hideshima, S.P. Treon, N.C. Munshi, P.G. Richardson and K.C. Anderson, Activation of NF-kappaB and upregulation of intracellular anti-apoptotic proteins via the IGF-1/Akt signaling in human multiple myeloma cells: therapeutic implications. *Oncogene*, 21, 5673-5683 (2002).
- 106. V. Poulaki, C.S. Mitsiades, V. Kotoula, S. Tseleni-Balafouta, A. Ashkenazi, D.A. Koutras and N. Mitsiades, Regulation of Apo2L/tumor necrosis factor-related apoptosis-inducing ligand-induced apoptosis in thyroid carcinoma cells. *Amer J Pathol*, 161, 643-654 (2002).
- A. Barreca, P.G. Artini, A. Cesarone, M. Arvigo, G. D'Ambrogio, A.R. Genazzani,
  G. Giordano and M. Minuto, Interrelationships between follicle stimulating hormone and the growth hormone----insulin-like growth factor--IGF-binding

proteins axes in human granulosa cells in culture. J Endocrinol Invest, 19, 35-42 (1996).

- 108. Y.M. Li, D.H. Schacher, Q. Liu, S. Arkins, N. Rebeiz, R.H. McCusker Jr., R. Dantzer and K.W. Kelley, Regulation of myeloid growth and differentiation by the insulin-like growth factor I receptor. *Endocrinology*, **138**, 362-368 (1997).
- 109. M. Karas, M. Danilenko, D. Fishman, D. LeRoith. J. Levy and Y. Sharoni, Membrane-associated insulin-like growth factor-binding protein-3 inhibits insulinlike growth factor-I-induced insulin-like growth factor-I receptor signaling in ishikawa endometrial cancer cells. J Biol Chem, 272, 16514-16520 (1997).
- 110. A. Grimberg, p53 and IGFBP-3: Apoptosis and cancer protection. *Mol Genet Metab*, **70**, 85-98 (2000).
- 111. J. Hong, G. Zhang, F. Dong and M.M. Rechler, Insulin-like growth factor (IGF)binding protein-3 mutants that do not bind IGF-I or IGF-II stimulate apoptosis in human prostate cancer cells. *J Biol Chem*, **277**,10489-10497 (2002).
- 112. Y. Oh, H.L. Muller, G. Lamson and R.G. Rosenfeld, Insulin-like growth factor (IGF)-independent action of IGF-binding protein-3 in Hs578T human breast cancer cells. Cell surface binding and growth inhibition. *J Biol Chem*, **268**, 14964-14971 (1993).
- Y. Oh, H.L. Muller, H. Pham and R.G. Rosenfeld, Demonstration of receptors for insulin-like growth factor binding protein-3 on Hs578T human breast cancer cells. J Biol Chem, 268, 26045-26048 (1993).
- 114. Y. Yamanaka, J.L. Fowlkes, E.M. Wilson, R.G. Rosenfeld and Y. Oh, Characterization of insulin-like growth factor binding protein-3 (IGFBP-3) binding to human breast cancer cells: kinetics of IGFBP-3 binding and identification of receptor binding domain on the IGFBP-3 molecule. *Endocrinology*, **140**, 1319-1328 (1999).
- 115. J.M. Ricort and M. Binoux, Insulin-like growth factor (IGF) binding protein-3 inhibits type 1 IGF receptor activation independently of its IGF binding affinity. *Endocrinology*, **142**, 108-113 (2001).
- J.M. Ricort, A. Lombet, C. Lassarre and M. Binoux, Insulin-like growth factor binding protein-3 increases intracellular calcium concentrations in MCF-7 breast carcinoma cells. *FEBS Lett*, **527**, 293-297 (2002).
- 117. G. Jaques, K. Noll, B. Wegmann, S. Witten, E. Kogan, R.T. Radulescu and K. Havemann, Nuclear localization of insulin-like growth factor binding protein 3 in a lung cancer cell line. *Endocrinology* **138**, 1767-1770 (1997).
- L.J. Schedlich, T.F. Young, S.M. Firth and R.C. Baxter, Insulin-like growth factorbinding protein (IGFBP)-3 and IGFBP-5 share a common nuclear transport pathway in T47D human breast carcinoma cells. *J Biol Chem*, 273, 18347-18352 (1998).
- 119. L.J. Schedlich, S.L. Le Page, S.M. Firth, L.J. Briggs, D.A. Jans and R.C. Baxter, Nuclear import of insulin-like growth factor-binding protein-3 and -5 is mediated by the importin beta subunit. *J Biol Chem*, **275**, 23462-23470 (2000).
- 120. B. Liu, H.Y. Lee, S.A. Weinzimer, D.R. Powell, J.L. Clifford, J.M. Kurie and P. Cohen, Direct functional interactions between insulin-like growth factor-binding protein-3 and retinoid X receptor-alpha regulate transcriptional signaling and apoptosis. J Biol Chem, 275, 33607-33613 (2000).
- 121. A.J. Butt, S.M. Firth, M.A. King and R.C. Baxter, Insulin-like growth factorbinding protein-3 modulates expression of Bax and Bcl-2 and potentiates p53independent radiation-induced apoptosis in human breast cancer cells. *J Biol Chem*, 275, 39174-39181 (2000).

#### Cancer

- 122. C.M. Perks, C. McCaig and J.M. Holly, Differential insulin-like growth factor (IGF)-independent interactions of IGF binding protein-3 and IGF binding protein-5 on apoptosis in human breast cancer cells. Involvement of the mitochondria. J Cell Biochem, 80, 248-258 (2000).
- 123. C. Coleman and A. Grimberg, Alterations in the expression of insulin-like growth factors and their binding proteins in lung cancer, in: *Methods in Molecular Medicine: Lung Cancer Volume 1 - Molecular Pathology Methods and Reviews*, (B. Driscoll, ed.) The Humana Press Inc., Totowa, 2002, pp.167-186.
- 124. A. Hoeflich, R. Reisinger, H. Lahm, W. Kiess, W.F. Blum, H.J. Kolb, M.M. Weber and E. Wolf, Insulin-like growth factor-binding protein 2 in tumorigenesis: protector or promoter? *Cancer Res*, 61, 8601-8610 (2001).
- 125. P. Cohen, H.C. Graves, D.M. Peehl, M. Kamarei, L.C. Giudice and R.G. Rosenfeld, Prostate-specific antigen (PSA) is an insulin-like growth factor binding protein-3 protease found in seminal plasma. J Clin Endocrinol Metab, 75, 104610-53 (1992).
- E. Okabe, J. Kajihara, Y. Usami and K. Hirano, The cleavage site specificity of human prostate specific antigen for insulin-like growth factor binding protein-3. *FEBS Lett*, 447, 87-90 (1999).
- 127. S.E. Nunn, D.M. Peehl and P. Cohen, Acid-activated insulin-like growth factor binding protein protease activity of cathepsin D in normal and malignant prostatic epithelial cells and seminal plasma. *J Cell Physiol* **171**, 196-204 (1997).
- 128. H. Kanety, Y. Madjar, Y. Dagan, J. Levi, M.Z. Papa, C. Pariente, B. Goldwasser and A. Karasik, Serum insulin-like growth factor-binding protein-2 (IGFBP-2) is increased and IGFBP-3 is decreased in patients with prostate cancer: correlation with serum prostate-specific antigen. J Clin Endocrinol Metab, 77, 229-233 (1993).
- 129. S.I. Helle, S. Geisler, T. Aas, T. Paulsen, J.M. Holly and P.E. Lonning, Plasma insulin-like growth factor binding protein-3 proteolysis is increased in primary breast cancer. *Brit J Cancer*, **85**, 74-77 (2001).
- S. Ekberg, M. Luther, T. Nakamura and J.O. Jansson, Growth hormone promotes early initiation of hepatocyte growth factor gene expression in the liver of hypophysectomized rats after partial hepatectomy. *J Endocrinol*, **135**, 59-67 (1992).
- 131. S.A. Rogers, J. Rasmussen, S.B. Miller and M.R. Hammerman, Effects of growth hormone on rat renal epidermal growth factor expression. *Amer J Physiol*, **267**, F208-F214 (1994).
- 132. T. Izumi, J. Shida, S. Jingushi, T. Hotokebuchi and Y. Sugioka, Administration of growth hormone modulates the gene expression of basic fibroblast growth factor in rat costal cartilage, both in vivo and in vitro. *Mol Cell Endocrinol*, **112**, 95-99 (1995).
- D. Swolin and C. Ohlsson, Growth hormone increases interleukin-6 produced by human osteoblast-like cells. J Clin Endocrinol Metab, 81, 4329-4333 (1996).
- 134. H. Li, P.M. Bartold, C.Z. Zhang, R.W. Clarkson, W.G. Young and M.J. Waters, Growth hormone and insulin-like growth factor I induce bone morphogenetic proteins 2 and 4: a mediator role in bone and tooth formation? *Endocrinology*, **139**, 3855-3862 (1998).
- Y.H. Tseng, M.A. Kessler and L.A. Schuler, Regulation of interleukin (IL)-1alpha, IL-1beta, and IL-6 expression by growth hormone and prolactin in bovine thymic stromal cells. *Mol Cell Endocrinol*, **128**, 117-127 (1997).
- 136. B.N. Friedrichsen, C. Carlsson, A. Moldrup, B. Michelsen, C.H. Jensen, B. Teisner and J.H. Nielsen, Expression, biosynthesis and release of preadipocyte factor-1/ delta-like protein/fetal antigen-1 in pancreatic beta-cells: possible physiological implications. J Endocrinol, 176, 257-266 (2003).

- 137. T. Yamauchi, K. Ueki, K. Tobe, H. Tamemoto, N. Sekine, M. Wada, M. Honjo, M. Takahashi, T. Takahashi, H. Hirai, T. Tushima, Y. Akanuma, T. Fujita, I. Komuro, Y. Yazaki and T. Kadowaki, Tyrosine phosphorylation of the EGF receptor by the kinase Jak2 is induced by growth hormone. *Nature*, **390**, 91-96 (1997).
- 138. Y. Huang, S.O. Kim, J. Jiang and S.J Frank, Growth hormone-induced phosphorylation of epidermal growth factor (EGF) receptor in 3T3-F442A cells. Modulation of EGF-induced trafficking and signaling. J Biol Chem, 278, 18902-18913 (2003).
- J. Okano, L. Snyder and A.K. Rustgi, Genetic alterations in esophageal cancer. *Meth Mol Biol*, 222, 131-145 (2003).
- 140. M. Takaoka, H. Harada, C. Andl, K. Oyama, Y. Naomoto, K.L. Dempsey, A.J. Kline-Szanto, W.S. El Deiry, A. Grimberg, and H. Nakagawa, Epidermal growth factor receptor regulates aberrant expression of insulin-like growth factor-binding protein 3. *Cancer Res*, 64, 7711-7723 (2004).
- 141. C. Sell, M. Rubini, R. Rubin, J.P. Liu, A. Efstratiadis and R. Baserga, Simian virus 40 large tumor antigen is unable to transform mouse embryonic fibroblasts lacking type 1 insulin-like growth factor receptor. *Proc Natl Acad Sci USA*, **90**, 11217-11221 (1993).
- 142. J.A. Price, S.J. Kovach, T. Johnson, L.G. Koniaris, P.A. Cahill, J.V. Sitzmann and I.H. McKillop, Insulin-like growth factor I is a comitogen for hepatocyte growth factor in a rat model of hepatocellular carcinoma. *Hepatology*, **36**, 1089-1097 (2002).
- D. Oksenberg, B.S. Dieckmann and P.L. Greenberg, Functional interactions between colony-stimulating factors and the insulin family hormones for human myeloid leukemic cells. *Cancer Res*, 50, 6471-6477 (1990).
- 144. A. Lai, B. Sarcevic, O.W. Prall and R.L. Sutherland, Insulin/insulin-like growth factor-I and estrogen cooperate to stimulate cyclin E-Cdk2 activation and cell Cycle progression in MCF-7 breast cancer cells through differential regulation of cyclin E and p21(WAF1/Cip1). J Biol Chem, 276, 25823-25833 (2001).
- 145. I.H. Hamelers, R.F. van Schaik, H.A. van Teeffelen, J.S. Sussenbach and P.H. Steenbergh, Synergistic proliferative action of insulin-like growth factor I and 17 beta-estradiol in MCF-7S breast tumor cells. *Exp Cell Res*, 273, 107-117 (2002).
- 146. R.L. Stephen, L.E. Shaw, C. Larsen, D. Corcoran and P.D. Darbre, Insulin-like growth factor receptor levels are regulated by cell density and by long term estrogen deprivation in MCF7 human breast cancer cells. *J Biol Chem*, **276**, 40080-40086 (2001).
- 147. T. Nickerson, F. Chang, D. Lorimer, S.P. Smeekens, C.L. Sawyers and M. Pollak, In vivo progression of LAPC-9 and LNCaP prostate cancer models to androgen independence is associated with increased expression of insulin-like growth factor I (IGF-I) and IGF-I receptor (IGF-IR). *Cancer Res*, **61**, 6276-6280 (2001).
- S.E. Damon, S.R. Plymate, J.M. Carroll, C.C. Sprenger, C. Dechsukhum, J.L. Ware and C.T. Roberts Jr., Transcriptional regulation of insulin-like growth factor-I receptor gene expression in prostate cancer cells. *Endocrinology*, **142**, 21-27 (2001).
- 149. M.J. Campbell, J.V. Woodside, J. Secker-Walker, A. Titcomb and A.J. Leathem, IGF status is altered by tamoxifen in patients with breast cancer. *Mol Pathol*, 54, 307-310 (2001).
- 150. B. Bonanni, H. Johansson, S. Gandini, A. Guerrieri-Gonzaga, R. Torrisi, M.T. Sandri, M. Cazzaniga, S. Mora, C. Robertson, E.A. Lien and A. Decensi, Effect of

336

low dose tamoxifen on the insulin-like growth factor system in healthy women. Breast Cancer Res Treat, 69, 21-27 (2001).

- 151. H. Werner, Z. Shen-Orr, F.J. Rauscher 3<sup>rd</sup>, J.F. Morris, C.T. Roberts Jr and D. LeRoith, Inhibition of cellular proliferation by the Wilms' tumor suppressor WT1 is associated with suppression of insulin-like growth factor I receptor gene expression. *Mol Cell Biol*, **15**, 3516-3522 (1995).
- 152. H. Werner, E. Karnieli, F.J. Rauscher and D. LeRoith, Wild-type and mutant p53 differentially regulate transcription of the insulin-like growth factor I receptor gene. *Proc Natl Acad Sci USA*, **93**, 8318-8323 (1996).
- 153. L. Zhang, F. Kashanchi, Q. Zhan, S. Zhan, J.N. Brady, A.J. Fornace, P. Seth and L.J. Helman, Regulation of insulin-like growth factor II P3 promotor by p53: a potential mechanism for tumorigenesis. *Cancer Res*, 56, 1367-1373 (1996).
- 154. L. Buckbinder, R. Talbott, S. Velasco-Miguel, I. Takenaka, B. Faha, B.R. Seizinger and N. Kley, Induction of the growth inhibitor IGF-binding protein 3 by p53. *Nature*, 377, 646-649 (1995).
- 155. C.M. van Golen, T.S. Schwab, K.M. Ignatoski, S.P. Ethier and E.L. Feldman, PTEN/MMAC1 overexpression decreases insulin-like growth factor-I-mediated protection from apoptosis in neuroblastoma cells. *Cell Growth Differ*, **12**, 371-378 (2001).
- 156. Y. Choi, J. Zhang, C. Murga, H. Yu, E. Koller, B.P. Monia, J.S. Gutkind and W. Li, PTEN, but not SHIP and SHIP2, suppresses the PI3K/Akt pathway and induces growth inhibition and apoptosis of myeloma cells. *Oncogene*, **21**, 5289-5300 (2002).
- 157. D. Bonneau and M. Longy, Mutations of the human PTEN gene. *Hum Mutat*, 16, 109-122 (2000).
- L.C. Cantley and B.G. Neel, New insights into tumor suppression: PTEN suppresses tumor formation by restraining the phosphoinositide 3-kinase/AKT pathway. *Proc Natl Acad Sci USA*, 96, 4240-4245 (1999).
- 159. W. Kim and W.G. Kaelin Jr, The von Hippel-Lindau tumor suppressor protein: new insights into oxygen sensing and cancer. *Curr Opin Genet Dev*, **13**, 55-60 (2003).
- 160. K. Datta, R. Nambudripad, S. Pal, M. Zhou, H.T. Cohen and D. Mukhopadhyay, Inhibition of insulin-like growth factor-I-mediated cell signaling by the von Hippel-Lindau gene product in renal cancer. *J Biol Chem*, 275, 20700-20706 (2000).
- 161. S.C. Clifford, M.E. Cockman, A.C. Smallwood, D.R. Mole, E.R. Woodward, P.H. Maxwell, P.J. Ratcliffe and E.R. Maher, Contrasting effects on HIF-1-alpha regulation by disease-causing pVHL mutations correlate with patterns of tumourigenesis in von Hippel-Lindau disease. *Hum Mol Genet*, 10, 1029-1038 (2001).
- 162. A. Grimberg, B. Liu, P. Bannerman, W.S. El-Deiry and P. Cohen, IGFBP-3 mediates p53-induced apoptosis during serum starvation. *Int J Oncol*, **21**, 327-335 (2002).
- P. Friedlander, Y. Haupt, C. Prives and M. Oren, A mutant p53 gene that discriminates between p53-responsive genes cannot induce apoptosis. *Mol Cell Biol*, 16, 4961-4971 (1996).
- R.L. Ludwig, S. Bates and K.H. Vousden, Differential activation of target cellular promoters by p53 mutants with impaired apoptotic function. *Mol Cell Bio*, 16, 4952-4960 (1996).
- 165. A. Grimberg, C.M. Coleman, T.F. Burns, B.P. Himelstein, R. David, C.J. Koch, P. Cohen and W.S. El-Deiry, "p53-dependent and p53-independent Induction of IGFBP-3 by DNA Damage and Hypoxia," *The 84<sup>nth</sup> Annual Meeting of The*

*Endocrine Society, San Francisco, CA.* The Endocrine Society Program and Abstracts, 2002, 226; paper submitted.

- 166. J.L. Martin and R.C. Baxter, Oncogenic ras causes resistance to the growth inhibitor insulin-like growth factor binding protein-3 (IGFBP-3) in breast cancer cells. J Biol Chem, 274, 16407-16411 (1999).
- 167. B. Mannhardt, S.A. Weinzimer, M. Wagner, M. Fiedler, P. Cohen, P. Jansen-Durr and W. Zwerschke, Human papillomavirus type 16 E7 oncoprotein binds and inactivates growth-inhibitory insulin-like growth factor binding protein 3. *Mol Cell Biol*, 20, 6483-6495 (2000).
- 168. T. Moshang, Jr. and A. Grimberg, The Effects of irradiation and chemotherapy on growth, *Endocrinol Metab Clin N Amer*, **25**, 731-741 (1996).
- 169. A.J. Swerdlow, R.E. Reddingius, C.D. Higgins, H.A. Spoudeas, K. Phipps, Z. Qiao, W.D. Ryder, M. Brada, R.D. Hayward, C.G. Brook, P.C. Hindmarsh and S.M. Shalet, Growth hormone treatment of children with brain tumors and risk of tumor recurrence. J Clin Endocrinol Metab, 85, 4444-4449 (2000).
- M. Henwood, A. Grimberg and T. Moshang, Jr, The Expanded spectrum of pediatric growth hormone use: Evidence for growth benefits, *Curr Opin Pediatr*, 14, 437-442 (2002).
- 171. A. Grimberg and P. Cohen, Optimizing growth hormone therapy in children, *Horm Res*, **48**, S11-5 (1997).
- 172. S.E. Dunn, F.W. Kari, J. French, J.R. Leininger, G. Travlos, R. Wilson and J.C. Barrett, Dietary restriction reduces insulin-like growth factor I levels, which modulates apoptosis, cell proliferation, and tumor progression in p53-deficient mice. *Cancer Res*, 57, 4667–4672 (1997).
- 173. S.D Hursting, J.A. Lavigne, D. Berrigan, S.N. Perkins and J.C. Barrett, Calorie restriction, aging, and cancer prevention: mechanisms of action and applicability to humans. *Ann Rev Med*, **54**, 131-152 (2003).
- 174. S.D Hursting, S.N. Perkins, J.M. Phang and J.C. Barrett, Diet and cancer prevention studies in p53-deficient mice. *J Nut*, **131**, 3092S-3094S (2001).
- 175. H. Vainio, R. Kaaks and F. Bianchini, Weight control and physical activity in cancer prevention: international evaluation of the evidence. *Eur J Cancer Prev*, **11**, S94-100 (2002).
- 176. J. Ma, E. Giovannucci, M. Pollak, J.M. Chan, J.M. Gaziano, W. Willett and M.J. Stampfer, Milk intake, circulating levels of insulin-like growth factor-I, and risk of colorectal cancer in men. J. Natl Cancer Inst, 93, 1330-1336 (2001).
- 177. M.D. Holmes, M. Pollak, W.C. Willett and S.E. Hankinson, Dietary correlates of plasma insulin-like growth factor I and insulin-like growth factor binding protein 3 concentrations. *Cancer Epidemiol Biomark Prev*, **11**, 852-861 (2002).
- 178. D. Gunnell, S.E. Oliver, T.J. Peters, J.L. Donovan, R. Persad, M. Maynard, D. Gillatt, A. Pearce, F.C. Hamdy, D.E. Neal and J.M. Holly, Are diet-prostate cancer associations mediated by the IGF axis? A cross-sectional analysis of diet, IGF-I and IGFBP-3 in healthy middle-aged men. *Brit J Cancer*, 88, 1682-1686 (2003).
- 179. I.E. McCutcheon, A. Flyvbjerg, H. Hill, J. Li, W.F. Bennett, J.A. Scarlett and K.E. Friend, Antitumor activity of the growth hormone receptor antagonist pegvisomant against human meningiomas in nude mice. *J Neurosurg*, 94, 487-492 (2001).
- M. Pollak, M.J. Blouin, J.C. Zhang and J.J. Kopchick, Reduced mammary gland carcinogenesis in transgenic mice expressing a growth hormone antagonist. *Brit J Cancer*, 85, 428-430 (2001).

#### Cancer

- 181. A. Jungwirth, A.V. Schally, J. Pinski, K. Groot, P. Armatis and G. Halmos, Growth hormone-releasing hormone antagonist MZ-4-71 inhibits in vivo proliferation of Caki-I renal adenocarcinoma. *Proc Natl Acad Sci USA*, **94**, 5810-5813 (1997).
- 182. K. Szepeshazi, A.V. Schally, K. Groot, P. Armatis, G. Halmos, F. Herbert, B. Szende, J.L. Varga and M. Zarandi, Antagonists of growth hormone-releasing hormone (GH-RH) inhibit IGF-II production and growth of HT-29 human colon cancers. *Brit J Cancer*, 82, 1724-1731 (2000).
- 183. A. Plonowski, A.V. Schally, M. Letsch, M. Krupa, F. Hebert, R. Busto, K. Groot and J.L. Varga, Inhibition of proliferation of PC-3 human prostate cancer by antagonists of growth hormone-releasing hormone: lack of correlation with the levels of serum IGF-I and expression of tumoral IGF-II and vascular endothelial growth factor. *Prostate*, **52**, 173-182 (2002).
- 184. R. Braczkowski, A.V. Schally, A. Plonowski, J.L. Varga, K. Groot, M. Krupa and P. Armatis, Inhibition of proliferation in human MNNG/HOS osteosarcoma and SK-ES-1 Ewing sarcoma cell lines in vitro and in vivo by antagonists of growth hormone-releasing hormone: effects on insulin-like growth factor II. *Cancer*, 95, 1735-1745 (2002).
- 185. C.T. Lee, K.H. Park, Y. Adachi, J.Y. Seol, C.G. Yoo, Y.W. Kim, S.K. Han, Y.S. Shim, K. Coffee, M.M. Dikov and D.P. Carbone, Recombinant adenoviruses expressing dominant negative insulin-like growth factor-I receptor demonstrate antitumor effects on lung cancer. *Cancer Gene Ther*, **10**, 57-63 (2003).
- 186. D. Sachdev, S.L. Li, J.S. Hartell, Y. Fujita-Yamaguchi, J.S. Miller and D. Yee, A chimeric humanized single-chain antibody against the type I insulin-like growth factor (IGF) receptor renders breast cancer cells refractory to the mitogenic effects of IGF-I. *Cancer Res*, **63**, 627-635 (2003).
- 187. K. Scotlandi, C. Maini, M.C. Manara, S. Benini, M. Serra, V. Cerisano, R. Strammiello, N. Baldini, P.L. Lollini, P. Nanni, G. Nicoletti and P. Picci, Effectiveness of insulin-like growth factor I receptor antisense strategy against Ewing's sarcoma cells. *Cancer Gene Ther*, 9, 296-307 (2002).
- 188. E.A. Bohula, A.J. Salisbury, M. Sohail, M.P. Playford, J. Riedemann, E.M. Southern and V.M. Macaulay, The efficacy of small interfering RNAs targeted to the type 1 insulin-like growth factor receptor (IGF1R) is influenced by secondary structure in the IGF1R transcript. *J Biol Chem*, **278**, 15991-15997 (2003).
- 189. A. Ly, J.C. Francois, L.C. Upegui-Gonzalez, B. Swiercz, C. Bedel, H.T. Duc, D. Bout and J. Trojan, Alterations in tumorigenicity of embryonal carcinoma cells by IGF-I triple-helix induced changes in immunogenicity and apoptosis. *Life Sci*, 68, 307-319 (2000).

### Chapter 14

### GENETIC BASIS OF PROPORTIONAL SHORT STATURE

Ángel Campos-Barros, Karen E. Heath and Jesús Argente Hospital Infantil Universitario Niño Jesús. Department of Paediatric Endocrinology, Universidad Autónoma de Madrid, Spain

Key words: Gene; GH; IGF-I; short stature.

#### **1. INTRODUCTION**

The considerable progression in molecular biology and genetic techniques during the last two decades, together with the unravelling of the human genome in recent years, has notably contributed to expand the knowledge of the genetic bases of pathological growth in humans. Furthermore, these dramatic advances have also provided us with a host of new molecular techniques that allow a better and quicker diagnosis of molecular anomalies related with the pathophysiology of the human growth hormone/insulin-like growth factor-I (GH/IGF-I) axis. Hence, the number of genes whose alterations appear related to, and/or associated with the different manifestations and classifications of human pathological growth has increased considerably. In the present chapter we review the genetic and molecular basis of proportional undergrowth due to GH or IGF-I deficiency or resistance. Thus, we will examine the different genes related with the GH/IGF-I axis and their various alterations that are currently known to be implicated in the etiopathogenesis of proportional undergrowth.

# 2. MOLECULAR BASES OF PROPORTIONAL UNDERGROWTH

Intensive investigations during the last two decades have demonstrated the critical role of the GH/IGFs system in all phases of mammalian growth, from intrauterine growth through to childhood and puberty.

A deficiency in the insulin-like growth factors (IGFs), in particular IGF-I, may be the consequence of a primary or secondary IGF-I deficit, manifested by the absolute or relative absence of detectable IGF-I in the serum or plasma (Table 1). GH insufficiency due to defects in the synthesis and/or secretion of GH is also manifested by low or undetectable circulating levels of GH and IGF-I (Table 1).

Aetiology	Serum GH	Serum IGF-I	PHENOTYPE
	levels	levels	
GH deficiency	11	$\downarrow\downarrow$	Short stature,
GH resistance	$\rightarrow$ or $\uparrow$	$\downarrow\downarrow$	prominent forehead and
IGF-I deficiency	$\rightarrow$ or $\uparrow$	$\downarrow\downarrow$	underdeveloped nasal bridge
IGF-I resistance	$\rightarrow$ or $\uparrow$	$\rightarrow$ or $\uparrow$	

Table 1. Molecular determinants of proportional undergrowth.

Resistance to IGF-I due to defects in the IGF-I receptor or downstream the receptor would also result in a similar phenotype, but with normal or elevated IGF-I levels. Likewise, resistance to the actions of GH, due to defects in the GH receptor or downstream the post-receptor signaling pathway, result in a similar phenotype in the face of normal or elevated GH serum levels and depleted IGF-I concentrations.

A deficiency in IGF-I, regardless of its origin, results in a phenotype of short stature, prominent forehead and an underdeveloped nasal bridge (Fig. 1A). Birth length is normal or slightly reduced in the majority of children with IGF-I deficiency. However, postnatal growth is highly abnormal owing to progressive slowing of the growth velocity, especially after 6 months of age (Fig. 1B).



*Figure 1.* (A) The typical phenotype associated with a deficiency of IGF-I A (Adapted from Pérez-Jurado<sup>1</sup>). (B) Pathological and normal growth curves. The  $3^{rd}$  to  $95^{th}$  height percentiles are represented at each year of age. The pathological growth curve is indicated by the curve outside the percentile range. GH treatment was administered, as indicated by the arrow, and growth was restored to the normal range.

The complexity of the human GH axis regulation at the hypothalamus and pituitary gland, as well as in target cells and tissues, predicts the existence of a relatively large number of genes that could be potentially involved as the molecular basis of proportional undergrowth (Fig 2.).

### 3. GENETIC DETERMINANTS OF GH DEFICIENCY

The frequency and prevalence of GH/IGF-I deficiency is difficult to estimate, and may vary depending upon the different diagnostic criteria applied and the ethnical origin of the studied population. Previous studies estimated a prevalence of idiopathic GH deficiency of 1/3480<sup>2</sup>. Approximately 5% to 30% of patients with GH deficiency have an affected first degree relative, thus suggesting the possibility of an underlying genetic defect. Furthermore, the fact that only 20% of the sporadic cases of GH deficiency is the result of environmental factors or

hypothalamic/hypophyseal anatomical defects as revealed by magnetic resonance imaging, suggests also the possibility that a high proportion of the sporadic cases may be also caused by genetic defects<sup>3</sup>.

Proportional undergrowth with a genetic basis may be caused by many different genetic determinants among the genes involved in the regulation of the GH/IGF-I axis. The associated phenotypes and their molecular basis will constitute the main body of this chapter's content.



*Figure 2.* The complexity of the multilevel GH/IGF-I axis regulation. The chromosomal location of all involved genes is indicated. Level I: the hypothalamus with the releasing factor GHRH, ghrelin, and somatostatin (SS), that regulate pituitary GH secretion; level II: the anterior pituitary gland with the transcription factors involved in its embryonic development and cell lineage specification; level III: the liver, as the main organ for the synthesis of IGF-I, stimulated by GH and mediated by the GHR at the cell membrane and by the intracellular second messenger signaling cascade; level IV: the stomach, as the main producer of ghrelin; level V: the GHR and IGF-I receptors and downstream secondary elements in target tissues, *eg.*, the bone.

#### **3.1** Familiar isolated GH deficiency (IGHD)

Four different forms of IGHD differing in the degree of GH deficiency, molecular basis, mode of inheritance, and response to recombinant-derived GH treatment have been described (Table 2).

IA AR Absent Neonatal hypoglucemia; GH1 Homozygous consanguiniety; presence of different (17q22-24) deletions, missense	
antibodies against GH at the start of mutations that the GH treatment generate null allele	nse leles
IB AR Reduced Normal response to pharmacological <i>GHRH-R</i> Nonsense mutation GH stimulation tests; good response (7p14) and splice site to GH therapy homozygosity	ions:
GH1 Compound (17q22-24) heterozygosity of point mutations	of
II AD Reduced Normal response to pharmacological <i>GH1</i> Heterozygous <i>GH</i> stimulation tests; good response (17q22-24) mutations that affer to <i>GH</i> therapy sequence such as splice sites conserved residues	affect ences ites or lues
III XL Reduced Hypogammaglobulinemia; increase in BTK Splice site mutatio β-lymphocytes and immunoglobulins (Xq21.3-q22) after the start of treatment	itions

Table 2. Familiar Isolated GH Deficiency (IGHD).

#### **3.1.1 IGHD type IA (IGFD I; OMIM 262400)**

This is the most severe form and its incidence is unknown. Patients affected with IGHD IA usually have a normal or slightly reduced birth length and may present with hypoglycaemia during the neonatal period. However, in all of these subjects there is a very reduced growth rate after 6 months of age. The levels of circulating GH are undetectable, both in basal conditions and after all classes of provocative stimuli.

IGHD IA has an autosomal recessive pattern of inheritance and in the majority of the cases is due to homozygous deletions within the GH gene cluster encompassing the pituitary GH gene (GH1, OMIM +139250). The most frequent genetic defect (approximately 70% of the cases) detected in IGHD IA consists of a deletion of 6.7 kb, but other deletions of 7.6 kb, 7 kb, 45 kb and a double deletion in the GH gene cluster have been also described<sup>4</sup>\*.

As expected for an autosomal recessively transmitted disorder, a high incidence of consanguinity is observed in the families of patients with IGHD IA, which suggests that these patients inherited two identical mutated alleles. Interestingly, the heterozygous carriers of these deletions are not GH deficient, indicating that the presence of one single *GH1* allele is sufficient for maintaining GH production.

The human GH gene cluster is located on the long arm of chromosome 17 (17q23) and spans 66.5 kilobases (kb) (Fig. 3). It is comprised of five genes aligned in the same transcriptional orientation. From centromere to telomere the pituitary GH gene (GH1), chorionic somatomammotrophin pseudogene 1 (CSHP1), chorionic somatomammotrophin gene 1 (CSH1), placental GH gene (GH2), and gene 2 for chorionic somatomammotrophin (CSH2) are located. Each of these five genes consists of five exons and four short introns with a high degree of homology throughout their entire sequence (92% to 98%), suggesting that all of these genes arose from a single ancestral gene as a result of homologous, but unequal, recombinations that resulted in gene duplications. The presence of numerous Alu sequences in the cluster suggests that they could have been implicated in the recombination processes that gave rise to the actual GH gene cluster<sup>5</sup>. GH1 is expressed in the somatotrophs of the anterior pituitary and its major protein product is a pro-hormone of 217 amino acids that includes in the Nterminus a signal peptide necessary for its translocation into the endoplasmic reticulum (ER). Within the ER the cleavage of the 26-amino-acid signaling peptide occurs, giving origin to the biologically active form of GH, a 191amino-acid and 22-kDa protein. Alternative splicing of the primary RNA results in at least four different mRNA transcripts, the majority of which contain the whole coding sequence whereas 10% lack the first 45 bp of exon 3 and less than 5% lack the whole exon 3, or both exons 3 and  $4^6$ .



*Figure 3.* GH cluster on chromosome 17. The five genes, *i.e.*, the pituitary GH gene (GH1), the chorionic somatomammotrophin pseudogene 1 (CSHP1), the chorionic somatomammotrophin gene 1 (CSH1), the placental GH gene (GH2), and gene 2 for chorionic somatomammotrophin (CSH2) are oriented in the same direction along chromosome 17q22-24, and share a high degree (~90%) of homology.

In addition to deletions, other types of GH1 mutations such as point mutations and microdeletions associated with the severe phenotype of IGHD IA and undetectable plasma GH levels have been described<sup>7,8</sup>. All of these mutations generate null alleles of GH1, producing truncated proteins that are probably degraded intracellularly.

#### **3.1.2** IGFD type IB (IGHD IB; OMIM #173100)

IGHD IB is transmitted according to a Mendelian autosomal recessive pattern and is associated with low but detectable plasma GH levels after standard pharmacological stimuli. Other endocrine functions are not affected, and therefore the phenotype is less severe than in type IA.

The identification of a homozygous mutation in the mouse growth hormone releasing hormone receptor gene (ghrhr) as the molecular basis of the defect in the *Little* mouse, an animal model for autosomal recessive IGHD<sup>9</sup>, suggested that the human homologue, *GHRHR*, could be a good candidate to explain at least some cases of familiar IGHD. Indeed, since then, several mutations in *GHRHR* have been demonstrated as the main cause of IGHD IB<sup>10-17</sup>, whereas mutations in *GHRH* have been excluded<sup>18</sup>.

GH secretion is regulated by the interaction of GH releasing hormone (GHRH) and somatostatin, being released in 10 to 20 pulses in each 24-h cycle. The timing of GH pulses is primarily supervised by intermittent somatostatin withdrawal<sup>19</sup>, and the amplitude of GH pulses is driven by GHRH. Thus, GHRH acting through the GHRH receptor (GHRHR), plays a pivotal role in the regulation of GH synthesis and secretion in the pituitary. In humans the gene that codes for the GHRHR, *GHRHR*, has been located on chromosome  $7p14^{20}$ . Analysis of its genomic structure revealed the existence of 13 exons. The corresponding cDNA encodes a 423-amino-acid protein that has seven putative transmembrane domains, characteristic of G protein-coupled receptors. It is a member of the secretin family of G protein-coupled receptors and has 47%, 42%, 35%, and 28% identity with receptors for vasoactive intestinal peptide, secretin, calcitonin, and parathyroid hormone, respectively.

Splicing mutations in *GH1* have been also associated with IGHD IB. Cogan *et al.*, detected a G>C tranversion in the donor splice site of intron 4 of *GH1*<sup>21</sup>. The mutation activated a cryptic splice site 73 nucleotides upstream of the exon 4 splice donor resulting in the loss of amino acids 103 to 126 and in a reading frameshift that caused the incorporation of 94 novel amino acids. The authors suggested that the aberrant protein would not be secreted normally since the affected region, encoded by exons 4 and 5, is involved in the correct targeting of the protein to the secretory granules. More recently, Abdul-Latif *et al.*, reported a novel IVS4+5 G>C transversion in *GH1* of members of a highly inbred Bedouin kindred with the clinical features of type IB IGHD<sup>22</sup>. Through a similar mechanism, as described by Cogan *et al.*, the transversion does generate a different protein by means of activating a cryptic splice site<sup>21</sup>. Interestingly, 48.5% of heterozygous carriers presented with short stature (-1.7 SD) in spite of normal circulating levels of GH<sup>23</sup>.

#### **3.1.3** IGHD type II (IGHD II; OMIM #173100)

IGHD II presents clinical characteristics and diagnostic criteria similar as for patients with IGHD IB<sup>24</sup>, but it is associated with an autosomal dominant mode of inheritance. Genetic linkage studies demonstrated cosegregation of IGHD II with GH1 in the majority of the families, while GHRH was excluded by linkage analysis in all of the families studied<sup>18</sup>. Most of the IGHD II cases described, involved GH1 mutations that cause aberrant splicing of *GH1* transcripts<sup>8,25</sup>, affecting sequences necessary for normal mRNA processing ("splice enhancers") or the secondary structure of the heteronuclear RNA. All of these intronic mutations produce the same effect on the post-transcriptional processing of the RNA and result in the loss of exon 3 of GH1 in the mature mRNA. The resulting mutant GH protein is a previously described 17.5-kDa protein that lacks amino acids 32-71, including a cysteine residue. Although the dominant negative effect of these mutations has not yet been fully defined in humans, it has been shown that in transgenic mice, overproduction of the 17.5-kDa isoform, prevents maturation of the GH secretory vesicles, leading to somatotrophs loss and causing anterior pituitary hypoplasia<sup>25</sup>.

More recently, three heterozygous missense mutations of *GH1* have been described that result in a milder IGHD II phenotype than that caused by exon 3 skipping<sup>26,27</sup>. All three affect conserved residues, which apparently are critical for the molecular mechanism of GH secretion. Subtle mutations in *GH1* have been regarded as a comparatively rare cause of short stature. Miller *et al.* postulated that this bias might be due to the patient selection criteria usually adopted for *GH1* mutational screening rather than to the inherent mutational properties of the gene<sup>28</sup>. They screened *GH1* in 41 Caucasian individuals selected for short stature, reduced height velocity, and bone age delay; a group of 22 individuals with short stature and idiopathic GH deficiency, and a group of 154 controls. Twenty-four novel *GH1* mutations were identified. Signal transduction activity was assessed using *in vitro* expressed variants coding for 13 of the novel missense mutations. Reduced ability to activate the postreceptor JAK/STAT signaling pathway (see Sections 3 and 3.1 of this chapter) was found in six of these alterations.

A total of 15 novel *GH1* mutations were considered to be of probable phenotypic significance. Thus these lesions are more prevalent than previously recognized and although most of them may be insufficient on their own to account for the observed clinical phenotype, they are, nevertheless, likely to play a contributory role in the etiology of short stature.

#### 3.1.4 IGHD type II (OMIM #307200)

Only a few families with IGHD with an X-linked recessive pattern of inheritance have been described. All of the affected males presented with hypogammaglobulinemia<sup>29,30</sup>. Exogenous GH treatment to these patients increased  $\beta$  lymphocytes and elevated plasma immunoglobulin levels<sup>31</sup>. Genetic analysis of various affected families suggested that the combination of X-linked agammaglobulinemia (XLA) and isolated GH deficiency could be due to an alteration of the *XLA* gene (Bruton tyrosine kinase, *BTK*) on Xq21.3–q22 and/or of a contiguous locus that is probably involved in the expression of GH<sup>32</sup>. However, point mutations in *BTK* have been demonstrated to be the sole cause of the immunodeficiency and GH deficit in at least two families<sup>33,34</sup>.

Other forms of X-linked IGHD probably exist and this could explain the preponderance of males in relation to females with IGHD. Current evidence<sup>32,35</sup> suggests that there are various loci on the X chromosome implicated in GH regulation: Xp22.3 and Xq13.3–q21.2.

#### **3.2** *GH1* promoter and locus control region (LCR)

The proximal promoter region of *GH1* is highly polymorphic containing at least 16 single nucleotide polymorphisms (SNPs)<sup>36,37</sup>. Fifteen SNPs were shown to be arranged in 40 different haplotypes<sup>38</sup> (Fig. 4A). Functional analysis showed that 12 more common haplotypes were associated with a significantly reduced level of reporter gene expression *in vitro* whereas 10 rarer haplotypes were associated with significantly increased level. Individual SNPs contributed to promoter strength in a highly interactive and non-additive fashion. Haplotype partitioning was successful in identifying six SNPs as major determinants of *GH1* expression. An association between adult height and the mean *in vitro* expression value corresponding to an individual's *GH1* promoter haplotype combination was noted, although only 3.3% of the variance of adult height was found to be explicable by this parameter<sup>38</sup>.

The expression of the human GH1 is also influenced by a locus control region (LCR) located between 14.5 kb and 32 kb upstream of  $GH1^{39}$ . The

LCR contains multiple DNase I hypersensitive sites and is required for the activation of the *GH* cluster genes in both pituitary and placenta<sup>40,41</sup>. Two DNase hypersensitive sites (I and II) containing binding sites for the pituitary-specific transcription factor POU1F1 are responsible for the high level of somatotroph-specific expression of the *GH1*<sup>42,45</sup>. Horan *et al.*, identified three SNPs within sites I and II of the LCR (Fig. 4B), and ascribed these to three distinct LCR haplotypes<sup>38</sup>. Functional analysis using a series of LCR-*GH1* proximal promoter constructs demonstrated that the LCR enhanced proximal promoter activity by 2.8-fold depending upon the proximal promoter haplotype, and that the activity of the proximal promoter haplotypes. Thus the genetic basis of inter-individual differences in *GH1* gene expression appears to be much more complex than initially thought.



*Figure 4.* Location of the 16 reported SNPs in the *GH1* promoter (A) and the three SNPs in the LCR (B). (A) The positions of the binding sites for the transcription factors: nuclear factor 1 (NF1), POU1F1, and vitamin D receptor (VDRE) as well as the location of the cAMP-responsive elements, the TATA box, the transcriptional start site (denoted by an arrow), and the translational initiation codon (ATG) are also shown. The hatched box represents exon 1. (B) Positions of the three SNPs in the LCR relative to the two DNase hypersensitive sites I and II, required for the pituitary-specific expression of *GH1*<sup>43</sup>. A, Adenine; C, cytosine; G, guanine; T, thymine.

## 3.3 Biologically inactive GH: Kowarski syndrome (OMIM # 262650)

Kowarski et al., described two unrelated boys with growth retardation, normal immunoreactive GH after stimulation, and low levels of IGF-I<sup>46</sup>. They deduced that the children had biologically inactive GH. The short stature of Kowarski syndrome is caused by biologically inactive GH (OMIM # 262650) and is characterized by lack of GH action despite high immunoassayable GH levels in serum and marked catch-up growth to exogenous GH administration. In 1996, Takahashi et al., reported a patient with short stature and low GH activity<sup>47</sup>. Isoelectric focusing of the proband's serum revealed the presence of an additional peak. Sequencing of GH1 revealed a heterozygous C>T transition that resulted in the amino acid alteration R77C. Further studies demonstrated that the child's GH not only could not activate the GH receptor (GHR) but also inhibited the action of wild type GH because of its greater affinity for GHR and GH binding proteins (GHBPs). Takashi et al., later described another case, where the girl's height was 3.6 SD below the mean for age and sex<sup>48</sup>. Bone age was delayed by 1.5 years. She had a prominent forehead and a hypoplastic nasal bridge with normal body proportions. Results of biochemical studies showed the lack of GH action despite high immunoassayable GH levels in serum and marked catch up growth to exogenous GH administration. The studies were compatible with the theory of the production of bioinactive GH, which prevented dimerization of the GH receptor and thus affecting the signal transduction of GH. The patient was shown to be heterozygous for an A>G substitution in exon 4 of GH1 resulting in the amino acid change, D112G.

In a recent report, 74 children with familial short stature were screened for the presence of mutations in  $GHI^{49}$ . The functional characterization of one novel variant, 1179M, identified in a Spanish patient, showed normal misfolding and secretion patterns as well as normal activation of STAT5, in the face of a reduced ERK activation (approximately 50%). It should be noted that the 1179M variant did not cosegregate with the short stature phenotype in the family, strongly suggesting that this variant is on its own insufficient to fully account for the observed phenotype.

### 3.4 Combined pituitary hormone deficiency (CPHD; OMIM # 262600)

Combined pituitary hormone deficiency, previously also known as panhypopituitarism, is characterized by a phenotype of proportional undergrowth in the presence of a deficiency in one or more of the trophic pituitary hormones [thyroid-stimulating hormone (TSH), prolactin (PRL), adrenocorticotropic hormone (ACTH), follicle-stimulating hormone (FSH) and luteinizing hormone (LH)] in addition to GH deficiency. At least five different loci responsible for this hereditary disease have been established (Table 3). All of them encode pituitary-specific transcription factors that are involved in the activation and control of the embryonic development and cell lineage specification of the anterior pituitary (Fig. 5).

Gene involved (chromosome)	Mode of inheritance	Affected pituitary bormones	Other clinical characteristics	Most frequent mutations	
POU1F1 (PIT1) (3p11)	AR or AD	GH, PRL, TSH	Short stature, possible hypoplasia of the pituitary gland; undetectable GH and PRL	DNA binding domain mutations (AR) or homeodomain mutations (AD)	
PROP1 (5q)	AD or AR	GH, PRL, TSH, LH, FSH (ACTH)*	Short stature, possible hypo- or hyperplasia of the pituitary gland; undetectable LH and FSH	DNA binding domain mutations	
<i>LHX3</i> (9q34)	AR	GH, PRL, TSH, LH, FSH	Short stature, "stiff neck"; possible hypo- or hyperplasia of the pituitary gland	Missense mutations at conserved residues; homozygous deletions	
LHX4	AD	GH, PRL, TSH. LH.	Short stature, defects in the sella turcica	Missense mutations at conserved residues	
(1q25)		FSH, ACTH	Cl	DNA kinding domain	
HESX1	AD	for the other	optical atrophy, septo-	mutations affecting	
(3p21.2-21.1)		pituitary hormones	optic dysplasia; anterior pituitary hypoplasia; diabetes insípida	conserved residues	

*Table 3.* Combined pituitary hormone deficiency (CPHD). \* Observed approximately in one third of affected patients.

## 3.4.1 POU1F1 (POU domain, class 1, transcription factor 1; OMIM + 173110)

Dwarf *Snell* and *Jackson* dwarf mice present an autosomal recessive phenotype associated with specific deficits in GH, PRL and, occasionally TSH. Both mutant mice are caused by different mutations in *Pit1* (Pituitary transcription factor 1). The similarity of the phenotype with some forms of CPHD in humans led to the analysis of the homologous gene, *i.e.*, *POU1F1* (POU domain, class 1, transcription factor 1, previously known as *PIT1*, pituitary specific transcription factor 1). The human *POU1F1* (Fig. 6) is located on chromosome 3p11 and consists of 6 exons and 5 introns<sup>55</sup>. It encodes a 290-amino-acid, 33-kDa homeobox protein that contains two specific DNA binding domains, the *POU*-specific domain and the *POU* homeodomain. Through the DNA binding domains, *POU1F1* binds to

specific sequences located in the promoter<sup>56</sup> and distal locus control region<sup>45</sup> of the human *GH1* (Fig. 4). Binding of POU1F1 as a dimer to the two *cis*-acting elements present in the promoter of the *GH1* seems to be essential for driving *GH1* expression in the pituitary. Mutations of the functional domains of *POU1F1* may be inherited as dominant or recessive traits. They are the cause of a combined deficiency of GH, TSH, and PRL because the specific activation of the *GH*, *PRL*, and  $\beta$ -*TSH* coding genes requires binding of POU1F1 to the cis-acting elements present in the promoter region of *GH1*, *PRL* and  $\beta$ -*TSH* coding genes<sup>6,57-59</sup>.



*Figure 5.* Transcription factors involved in the ontogenic development of the mouse anterior pituitary cell lineage. The finely tuned spatio-temporal expression pattern of pituitary specific transcription factors drives the cell-lineage specification during ontogenic development of the anterior pituitary. Hesx1 is first expressed and plays a role in the formation of the anterior neuroectoderm. Prop1 is expressed in somatotrophs, lactotrophs, and thyrotrophs during embryonic development and seems to be required for the extinction of Hesx1 expression<sup>50</sup> and to promote Pit1 expression, required to promote the expression of *GH1*, *PRL* and *β*-*TSH* coding genes. Lhx3 is expressed from early developmental stages in the mouse pituitary, and its expression between embryonic day 10.5 (e10.5) and e12.5 seems to be critical for the posterior differentiation of thyrotrophs, lactotrophs, somatotrophs, and gonadotrophs. Lhx4 expression becomes restricted to the future anterior lobe of the pituitary where it seems to be supportive, but not essential, for specification of gonadotroph cells and to the specification, expansion, and terminal differentiation of the other pituitary cell lineages. (Adapted from PE Mullis<sup>51</sup>).

Several recessive and at least three dominant mutations in the human POUIF1 have been identified in sporadic cases and in multiple families with CPHD<sup>50</sup> (Fig. 6), which cause distinct phenotypes depending on the type of mutation. Production of a truncated protein in homozygosity due to a R172X mutation is associated with a predominant TSH deficiency and congenital cretinism<sup>54,60</sup>. A missense mutation in the POU specific domain, R158P, either in homozygosity or heterozygosity in combination with a null allele, results in the absence of GH and PRL secretion, partial TSH deficiency, and a normal pituitary size<sup>54</sup>. Finally, a *de novo* heterozygotic missense mutation has been described in a patient with panhypopituitary dwarfism<sup>61</sup>. This mutation is located further down from the DNA binding domain (R271W) and the mutant protein appears to have a dominant negative effect that inhibits the DNA binding of the normal protein. The discovery of the same mutation in another Japanese family, in which various carriers had no pathological phenotype, revealed the monoallelic expression of POUIF1 through genomic imprinting in nonaffected carriers<sup>62</sup>.



*Figure 6.* Schematic representation of *POU1F1*. Exons 1–6 are shown as striped boxes for coding sequence and spotted boxes for noncoding regions. Mutations, noted at their locations, without an asterisk represent recessive mutations and those with an asterisk represent dominant mutations. Recessive mutations produce varying degrees of loss of DNA-binding and/or transcriptional activation functions. All dominant mutations affect the POU specific domain<sup>52</sup>. Mutation W261C is boxed as it represents the *Snell* dwarf mouse mutation and has not been found in humans<sup>53</sup>. A hash is located against the mutation E250X as it is due to both a missense mutation at codon 250 and a 1 <sup>1</sup>bp deletion at nucleotide 747, which causes a missense mutation at codon 249. DelPOU1F1 is a deletion of the entire gene<sup>54</sup>. The protein domains showing the transcriptional activation domain and the homeodomain are shown in the lower panel.

<sup>\*</sup> Amino acid abbreviations: A, alanine; C, cysteine; D, aspartic acid; E, glutamic acid; F, phenylalanine; G, glycine; H, histidine; I, isoleucine; L, leucine; M, methionine; P, proline; Q, glutamine; R, arginine; S, serine; T, threonine; W, tryptophan; X, termination.

## **3.4.2 PROP1 (Prophet of Pit1, paired-like homeodomain transcription factor; OMIM + 601538)**

The Ames dwarf mouse presents with a phenotype similar to that expressed by the Snell dwarf. This similarity helped to define another locus involved in dwarfism, *i.e.*, the *Prop1* locus. Localization of the *Prop1* in the Ames mouse by Sornson *et al.*<sup>53</sup>, led to the localization of the human ortholog on the long arm of chromosome 5 (5q). *PROP1* expression precedes and is a requisite for the expression of *POU1F1*. The human *PROP1* gene spans approximately 3 kb and consists of three exons encoding a 223 amino acid protein with a central DNA, paired-like, binding domain and a transcriptional activation domain (Fig. 7).



*Figure 7.* Schematic representation of *PROP1.* Coding sequences of exons 1–3 are represented by shaded areas, while noncoding sequences are shown as spotted areas. Mutations are noted at their corresponding locations. Mutation S83P represents the *Ames* mouse dwarf mutation and has not been found in humans<sup>53</sup>. The protein domains showing the  $\alpha$ -helices of the paired homeodomain and the transactivation domain are shown in the lower panel.

The transcription factor PROP1 is expressed in the somatotrophs, lactotrophs, and thyrotrophs during embryonic development. Its expression seems to be required for the extinction of *HESX1* expression<sup>50</sup>, another pituitary transcription factor involved in the etiopathogenesis of CPHD (see later). PROP1 binds as a dimer to promoter P3 elements (palindromic

TAAT(NNN)ATTA sequences) localized in the 5'-UTR of POUIF1. Several mutations in the human *PROP1* associated to the CPHD phenotype have been identified in the recent years<sup>50,63</sup> (Fig 7). The hormonal phenotype (Table 3) is variable, including deficiencies of GH, TSH, PRL, LH, and FSH. The hormone deficiencies emerge gradually during postnatal development in a typical order rather than being uniformly present at once $^{63}$ . In some cases, affected patients may not produce LH and FSH in sufficient amounts to enter puberty spontaneously, while in other cases the gonadotropin secretion gradually decreases with increasing age without affecting the spontaneous puberty switch-on<sup>64</sup>. ACTH may be also affected with increasing age<sup>65,66</sup>. Most of the recessively inherited cases of CPHD in Europe are due to *PROP1* mutations. The allelic variants described include two simple amino acid substitutions, R120C and F117I, and a deletion of two basepairs in exon 2 (301deIAG) that produces a frameshift reading error from codon 101 with an early stop codon appearing at 109<sup>64</sup>. The 301delAG mutation could be the most frequent cause of familial combined pituitary hormone deficiency<sup>67</sup>.

## 3.4.3 LHX3 (LIM-homeobox-3; OMIM \*600577) and LHX4 (LIM-homeobox-4; OMIM\*602146)

*LHX3* and *LHX4* encode two pituitary specific LIM-type homeodomain transcription factors that play a key role in the determination of the pituitary cell lineages. In the mouse, both *Lhx3* and *Lhx4* are expressed from early embryological stages in Rathke's pouch (Fig. 5). During subsequent developmental stages, *Lhx3* expression is detectable at high levels before the initial detection of the  $\alpha$ -glycoprotein subunit ( $\alpha$ -GSU), localized to the anterior and intermediate lobes of the pituitary gland and persists in the adult pituitary, where it is expected to still play a functional role during adult life.

Human *LHX3* spans 7.2 kb, consists of 7 exons and is located in the subtelomeric region of chromosome 9 (9q34.2-34.3, Ref 68). A homozygous missense mutation (Y116C) in the LIM2 domain of *LHX3* that impairs the transcriptional activation function, and an intragenic 23 amino acid deletion that includes the DNA binding domain, have been identified as the molecular cause of the disease in some CPHD patients<sup>69</sup>. In these cases, additionally to the combined hormonal deficit of GH, TSH, PRL, FSH, and LH, a rigid cervical spine leading to a limited head rotation was the clinical hallmark of the recessively inherited disorder<sup>69</sup>.

In contrast to the sustained expression pattern of Lhx3 during developmental of the mouse pituitary gland, Lhx4 expression becomes restricted to the future anterior lobe of the pituitary.  $Lhx3^{-1-}$  and  $Lhx4^{-1-}$  mice studies have shown that the formation of a definitive pouch is dependent of

both *Lhx3* and *Lhx4*<sup>70,71</sup>. During the next step, *i.e.*, the actual formation of the pituitary gland, the expression of *Lhx3* is essential. Later, together with *Lhx3*, *Lhx4* expression seems to be supportive, but not essential, for specification of gonadotroph cells and to the specification, expansion and terminal differentiation of the pituitary cell lineages.

*LHX4* is located on the long arm of chromosome 1 (1q25). It spans more than 45 kb, and is composed of 6 exons. The protein contains two LIM-like domains (encoded by exons 2 and 3, respectively) and one homeodomain, encoded by exons 4 and  $5^{72}$ . A heterozygous intronic point mutation in the splice acceptor preceding exon 5 (G>C) was recently identified<sup>72</sup> in a family with a phenotype of combined GH, TSH, and ACTH deficiency, with a small sella turcica, hypoplastic anterior hypophysis, and ectopic posterior hypophysis. The mutation in the splice site acceptor results in the activation of two cryptic splice-acceptor sites present in exon 5, giving origin to two different shorter transcripts of 1252 and 1247 amino acids (vs. 1264 in the wild-type), respectively. Both alternative transcripts lack conserved residues which are essential components of the homeodomain.

#### 3.4.4 HESX1 (homeobox 1 expressed in embryonic stem cells; OMIM: \*601802)

The Hess1 gene, originally named Rpx for Rathke pouch homeobox, was originally identified in mouse<sup>73</sup>. Dattani *et al.*, generated a *Hesx1* null mutant mouse and these mice exhibited variable anterior CNS defects and pituitary dysplasia<sup>74</sup>. Mutants have a reduced prosencephalon, either anophthalmia or microphthalmia, defective olfactory development, and bifurcations in the Rathke pouch. Neonates exhibit abnormalities in the corpus callosum, the anterior and hippocampal commissures, and the septum pellucidum. Thus the *Hesx1* knockout mouse phenotype is comparable to the phenotype shown in patients with septo-optic dysplasia (SOD; OMIM #182230), initially named as Morsier syndrome. The term septo-optic dysplasia was coined in 1956 by de Morsier<sup>75</sup> who pointed out the association of optic nerve hypoplasia and absence of the septum pellucidum. This syndrome is characterized by the triad of pituitary hypoplasia, optic nerve hypoplasia and agenesis of midline brain structures, including the corpus callosum and septum pellucidum (Fig. 8). The phenotype is highly variable, with 62% of affected individuals having hypopituitarism and 30% having all three manifestations<sup>76-78</sup>.



*Figure 8.* (A,B) Axial (T1) magnetic resonance image (MRI) of a patient with septo-optic dysplasia (SOD). Squared ventricular cavity due to the absence of the septum. (C) MRI of the medial line section of the sagittal T1. Ectopic posterior pituitary (indicated with an arrow) and hypoplasia of the anterior pituitary. (D) MRI of the coronal T1. Hypoplasia of the optic nerves (indicated with arrows).

Owing to this similarity, Dattani *et al.* screened the human homologue, *HESX1* (OMIM 601802) in familial SOD patients<sup>74</sup>. *HESXI* spans 1.7 kb and contains four coding exons. They demonstrated homozygosity for a missense mutation, R160C, in the homeodomain of *HESX1* in a brother and sister with agenesis of the corpus callosum and CPHD. The mutation destroyed the ability of HESX1 to bind target DNA. The protein has two functional domains: the engrailed homology repressor domain at the N-terminus, which functions as a promoter-specific repressor, and a homeodomain in the Cterminal, which functions as a transcription factor binding domain<sup>79</sup>. Different studies have since shown that *HESX1* mutations are rare in sporadic SOD patients and that there is incomplete penetrance<sup>79,80</sup>. The *HESX1* mutations found to date are listed in Table 4. The degree of pituitary deficiency varies from an isolated deficit in GH to situations of panhypopituitarism. Diabetes insipidus is present in more than half of the cases. The alteration appears to be in the hypothalamus. This syndrome is almost always sporadic, although some family cases have been reported which show an autosomal recessive pattern of inheritance.

Mutation (amino acid)	Mutation (nucleotide)	Exon	State	Familial/ Sporadic	Functional domain	Functional consequences	MRI	Ref.
Q6H	18G>C	1	Htz	S		Unknown	AP hypoplasia, ectopic PP	80
I26T	77T>C	1	Hz	F	Engrailed homology repressor domain (eh1)	Impaired ability to recruit mammalian Groucho homolog/transducin- like enhancer of split-1 (Gro/TLE1), a crucial corepressor for HESX1, leading to partial loss of repression	AP hypoplasia, ectopic PP, severely hypoplastic AP lobe	81
Fs104, stop codon at 105	306/307 Ins AG	2	Htz	S	Truncated mutant,lacking homeodomain	Lack homeodomain (DNA binding domain)	Left ON hypoplasia, AP hypoplasia, ectopic PP	82
N125S*	374A>G	3	Htz	S	Homeodomain	Slightly elevated DNA binding	·	79
R160C	478C>T	4	Hz	F	Homeodomain	Loss of <i>in vitro</i> DNA binding	Agenesis of CC, ON hypoplasia	74
\$170L	509C>T	4	Htz	F	RESQFL motif which is unique to HESXI.	Reduced DNA binding	Bilateral ON hypoplasia in one sibling and normal in other	80
					Immediately C-terminal to homeodomain		AP hypoplasia, undescended PP	79
T181A Fs175, HESX1 terminates at one aa further than Wt	541A>G 1684delG	4 4	Htz Htz	S S	-	Unknown Increased DNA binding which in turn causes increased repression of PROP1- dependent gene activity	AP hypoplasia Absent CC, thin ON, small AP gland and absence of posterior bright spot.	80 83

*Table 4.* Mutations in the human *HESX1* gene. AP, anterior pituitary; aa, amino acid; CC, corpus callosum; ON, optic nerve; PP, posterior pituitary. \*N125S mutation occurs at a high frequency in the Afro-Carribean population and may therefore reflect a population-specific polymorphism<sup>79</sup>.

# 4. GH INSENSITIVITY SYNDROME (LARON SYNDROME, GHIS; OMIM #262500)

In 1966, Laron described the cases of three siblings with the clinical and biochemical characteristics of GH deficiency, but with extremely elevated levels of circulating GH<sup>84</sup>. The syndrome was named Laron syndrome but it

is now also referred to as the syndrome of primary GH resistance or primary GH insensitivity. In the following 2 years, the same authors diagnosed 22 patients with a similar clinical picture. These patients were of eastern Jewish origin<sup>85</sup>.

Laron syndrome is a rare autosomal recessive condition<sup>86</sup> associated with postnatal growth failure leading to extreme short stature, midfacial hypoplasia, truncal obesity, and hypoglycemia. If these patients do not receive treatment with recombinant GH, the adult height remains between 119 and 142 cm in males and 108 and 136 cm in females<sup>87</sup>. Patients have increased GH levels, associated with a deficiency of IGF-I and IGFBP-3<sup>88,89</sup>. These features commonly result from dysfunction of the GH receptor (GHR) and consequent failure of signal transduction pathways.

In 1984, Laron syndrome was shown to be due to mutations in *GHR* resulting in a resistance to  $GH^{90}$ . The *GHR* is located on chromosome 5p13-p12<sup>91</sup>. It spans 87 kb and is comprised of 10 exons. *GHR* encodes a receptor of 620 amino acids and a signal sequence of 18 amino acids (encoded by exon 2). Exons 3 to 7 encode for the extracellular domain of 246 amino acids. Exon 8 corresponds to the transmembrane domain of 24 amino acids. Finally, exons 9 and 10 correspond to the intracellular domain made up of 350 amino acids (Fig. 9A).



*Figure 9.* (A) Schematic outline of the GH receptor showing the extracellular, transmembrane and cytoplasmic domains. (B) The post-receptor intracellular signaling cascade: 1, Activation of cytosolic Janus kinase 2 (JAK-2); 2, activation of the signal transducer and activator of transcription proteins (STAT), STAT1, STAT5, and the mitogen activating protein kinase (MAPK) pathway including extracellular regulated kinase 1 and 2 (ERK1/2), p38, and c-Jun N-terminal kinase (JNK); 3, transcriptional regulation of target genes.
To date, several patients have been diagnosed. Most patients with GHIS have missense, nonsense and splice site mutations in *GHR* exons 4–7, encoding the extracellular domain of the GHR<sup>88,89,92</sup> (Human Gene Mutation Database, <u>www.jgmd.org</u>). These mutations usually result in low or absent serum GH binding protein (GHBP) and thus, failure of GH to bind and signal at GH responsive tissues, associated to a deficiency of IGF-I and IGFBP-3<sup>89,93,94</sup>. The relatively few mutations described in the transmembrane or intracellular domain of the GHR result in truncated GHRs lacking most of the intracellular domain of the receptor (dominant negative mutations) or consist of compound heterozygous mutations involving the cytoplasmic domain<sup>95-99</sup>. A few deletions have also been reported<sup>99,100</sup>.

Recently, Laron reviewed the treatment with exogenous recombinant biosynthetic IGF-I and a follow-up of the 60 patients from the original Israeli cohort<sup>92</sup>. Although stimulation of linear growth is observed after the start of the treatment, its effects on growth velocity was not as intense as that observed in GH-deficient patients. Some of the undesired consequences of the exogenous IGF-I treatment could be avoided if the treatment would be initiated at birth or in infancy, but still nowadays it remains the exception with only a small number of children on time-limited clinical trials.

## 4.1 Defective post-receptor signal transduction

The GHR is a member of the family of cytokine receptors. GHR lacks intrinsic kinase activity and relies on the regulation and activation of cytosolic Janus kinase 2 (JAK-2). A single GH molecule binds two GHR molecules inducing receptor dimerization and hence activation<sup>101</sup>. Binding of GH to the GHR results in a conformational change in the receptor dimer and transphosphorylation of associated JAK-2<sup>102</sup>.

JAK-2 is constitutively associated with the receptor at the proline-rich Box1 site (amino acids 276–287), and it is thought that ligand binding may stabilize the preformed receptor-JAK-2 complex<sup>102,103</sup>. Box 2 (amino acids 325–338) of the GHR is though to be required for full activation of JAK-2 by GH<sup>103,104</sup>. After GH stimulation, phosphorylated JAK-2 has been reported to activate the signal transducer and activator of transcription proteins (STAT), STAT1, STAT5, as well as the mitogen activating protein kinase (MAPK) pathway, including extracellular regulated kinase 1 and 2 (ERK1/2), p38, and c-Jun N-terminal kinase (JNK<sup>105-107</sup>, Fig. 9B). Activated STAT5 is translocated to the nucleus where the transcription of GHresponsive genes such as *IGF-I* and *IGFBP-3* among others, is modulated<sup>108-110</sup> (Fig. 9B). STAT5 requires anchorage to tyrosine 469 and/or 516 of the GHR for activation<sup>111,112</sup>, whereas STAT1, STAT3, and MAPK are activated via JAK2, independently of any direct association with the GHR.

Recently, Kofoed et al., described a patient with the clinical and biochemical characteristics of GHIS but with no mutation in the GHR<sup>113</sup>. They found that signal transduction mediated by the GHR occurs by means of at least three well established pathways: the signaling module involving MAPK, ERK1, and 2, STAT, and the phosphatidylinositol-3 kinase B signaling module. After analyzing these signalling pathways in fibroblast cell lines derived from the patient, they found that signal transduction from the GHR to ERK1 and ERK2 was normal. However, GH-induced signalling through the STAT pathways (STAT1, STAT3, STAT5a, and STAT5b) was aberrant in the patient derived cells. STAT1 phosphorylation was increasingly induced (8-10 times) by GH in the patient's fibroblasts as compared to normal fibroblasts. The combined phenotype found in the patient, GHIS and immunodeficiency, was consistent with the presence of a defect in the JAK/STAT system. A missense mutation in STAT5b (OMIM \*604260) in homozygosis, A630P, was indeed identified. The parents were shown to be heterozygous for this change.

*STAT5b* is localized to chromosome 17q11.2 and consists of 19 exons. The domains of the encoded protein are represented schematically in Fig. 10. The A630P mutation occurs in exon 15 which encodes for the src homology (SH2) domain of STAT5b.



*Figure 10.* Domains of the STAT5b protein. The location of the mutation identified in the patient with GH insensitivity syndrome is marked in the SH2 domain<sup>113</sup>. Numbers indicate the amino acid positions.

How GHIS is classified is still debatable. Do two types of Laron syndrome, type 1 and type 2, caused by mutations in *GHR* and *STAT5* respectively, exist or is Laron syndrome distinguishable from GHIS as a result of mutations in *STAT5b*? To date only one patient has been found to carry a *STAT5b* mutation; thus further patients need to be screened until a detailed clinical comparison can be undertaken.

# 5. ANOMALIES IN THE IGF-I AND IGF-I RECEPTOR GENES

# 5.1 IGF-I gene (*IGF1*) anomalies

As outlined in section 1, GHIS can be also due to a primary resistance to the IGF-I caused by defects in the IGF-I receptor or downstream of the receptor, in the face of normal or even elevated IGF-I levels. The few cases of IGF-I resistance reported up to now are discussed below.

The human *IGF1* gene spans more than 80 kb and consists of six exons and five introns (Fig. 11). *IGF1* has been mapped to the human chromosome 12. As a result of alternative splicing of the primary gene transcript, different IGF-I mRNAs are produced that encode protein precursors with different C-termini, called E domains (Fig. 11). The human IGF-IA protein contains a 35-residue E domain, and the IGF-IB transcript a 77-amino-acid C-terminal extension<sup>118,119</sup>.



*Figure 11.* Structure and expression of the human *IGFI*. Structures of different human *IGFIA* and *-IB* transcripts are displayed<sup>114-118.</sup> Exons are numbered and indicated by boxes, with coding regions in black and noncoding segments in stripes. Promoters are shown by the letter P, transcription start sites by arrows and locations of differental polyadenylation at the 3' end by arrows in the gene and by boxes of different lengths in the mRNAs.

Woods *et al.*, described a male, son of consanguineous parents, with extreme short stature, in whom a homozygous deletion of exons 4 and 5 of *IGF1* (OMIM \*147440) was demonstrated<sup>120</sup>. The patient had marked mental retardation, small cranial perimeter, acromicria, hypogonadism, delayed motor development, sensorineural deafness (see also Chapter 9 of this book), hypoglycemia during infancy, elevated serum GH levels, and low serum IGF-I levels that did not increase with GH administration. In 2000, Woods *et al.*, reported the results of 1-year recombinant human IGF-I therapy on body composition, bone mineral density, insulin sensitivity, and linear growth in this patient<sup>120</sup>. They concluded that recombinant IGF-I therapy increased bone mineral density (BMD) by 17% and bone mineral apparent density by 7%, indicating that IGF-I has a greater effect on bone growth than bone mineralization. Bone turnover markers and growth velocity also increased on recombinant human IGF-I treatment.

Rasmussen *et al.*, considered *IGF1* and the IGF-I receptor gene (*IGF1R*) as candidates for low birth weight, insulin resistance and type 2 diabetes, but they found no sequence alterations resulting in amino acid substitutions in the two genes in 82 Danish families with type 2 diabetes<sup>121</sup>. Arends *et al.*, then hypothesized that minor genetic variation in *IGF1* could influence preand postnatal growth<sup>122</sup>. Three microsatellite markers located in *IGF1* were analyzed in 124 short children born small for gestational age and their parents. Two polymorphic markers showed association with IGF-I levels and short stature and small head circumference. Thus the authors concluded that since low IGF-I levels are associated with type 2 diabetes and cardiovascular disease, they proposed that the *IGF-I* may provide a link between low birth weight and such diseases in later life.

#### 5.2 The Pygmy phenotype

Pygmies show partial GH resistance, which has been suggested to be mediated either at the level of the GH receptor<sup>123,124</sup> or at the level of the IGF-I receptor<sup>125,126</sup>. The GHR has been implicated based on the low levels of GHBP in the plasma of pygmies from Africa and Papua-New Guinea<sup>123,124</sup>. The plasma level of GHBP, the circulating ectodomain of the GHR, is believed to reflect GHR concentrations in target tissues. The concept of IGF-I resistance has been based on the lack of a proliferative response of immortalized lymphocytes derived from African Efe pygmies to IGF-I, and decreased *IGFIR* expression and function in these cells<sup>125,126</sup>. Both GH resistance and IGF-I resistance may lead to short stature, and they are not mutually exclusive. Davila *et al.*, more recently studied two pygmy populations from the Philippines (Aeta and Mamanwa people) that are

unrelated to the African pygmies<sup>127</sup>. Serum GHBP and IGF-I levels were significantly decreased in both pygmy populations, compared to normal-statured Philippino controls. The results, together with previous observations in African and New Guinean pygmies, indicated that short stature is associated with low serum GHBP levels in pygmy populations of diverse origins and in different parts of the world. This strengthens the tentative postulate that the GHBP/GHR system plays an important role in the genetic and perhaps nutritional determination of adult stature in humans. Molecular genetic studies of *GHR* in various pygmy populations may shed further light on the mystery of pygmy short stature.

## 5.3 IGF-I resistance

There are very few references of patients with short stature that could be included in this category. The first clinical reported case was of a girl with severe growth retardation and elevated IGF-I levels<sup>128</sup>. Other similar cases have been described<sup>129,130</sup>. Further studies of fibroblasts derived from this patient reported by Heath-Monnig *et al.*, were undertaken<sup>129</sup>. They concluded that the resistance to IGF-I action in the patient's fibroblasts was caused by an abnormal production and/or cell association of IGF binding proteins<sup>131</sup>. However, a genetic basis could not been demonstrated in any of these cases.

In 1992, the human gene for the IGF-I receptor, *IGF1R*, was identified<sup>132</sup>, (OMIM \*147370). It contains 21 exons and spans approximately 100 kb on chromosome  $15q25-q26^{133}$ .

Approximately 10% of infants with intrauterine growth retardation remain small and the causes of their growth deficits are often unclear. Abuzzahab *et al.* screened for mutations in *IGF1R* in 42 patients with unexplained intrauterine growth retardation and subsequent short stature<sup>134</sup>. They identified a compound heterozygote who had two point mutations in exon 2 of *IGF1R*, that resulted in the amino acid substitution R108Q in one allele and K115N in the other allele. Fibroblasts cultured from the patient showed decreased IGF-I receptor function. The child also had signs of a nonverbal learning disorder, obsessive tendencies and social phobias. Experimental evidence that the IGFs are necessary for normal brain development and function suggest that these mutations may also affect brain growth and neurological development<sup>135</sup> (see also Chapter 8 in this book), the clinical features described in the patient of Abuzzahab *et al.*, may reflect the state of partial resistance to IGF-I<sup>134</sup>.

In a second cohort of 50 children with short stature who had elevated circulating IGF-I concentrations, Abuzzahab *et al.*, identified one boy with an *R59X* mutation that reduced the number of IGF-I receptors on fibroblast

cultures<sup>134</sup>. The boy also had microcephaly at birth that persisted, and he had mild retardation of motor development and speech. The patient displayed abnormal facial features, short fingers, clinodactily, wide-set nipples, and pectus excavatum. Both children had intrauterine growth retardation and poor postnatal growth but the phenotypes were not identical.

Thus mutations in IGF1R that lead to abnormalities in the function or number of IGF-I receptors may also retard intrauterine and subsequent growth in humans. The identification and understanding of further IGF1Rmutations and their associated phenotypes may provide insights into both growth retardation and the IGF system in humans.

# 5.4 IGF-I acid-labile subunit (IGFALS; OMIM +601489)

*In vivo*, IGF-I and IGF-II are always complexed to one of a family of six IGF-binding proteins, IGFBP1-6 (see Chapter 3 in this book). Prenatally, binary IGFBP–IGF complexes of 50 kDa predominate in the serum, with IGFBP-2 being the most frequently occurring IGFBP moiety. In juveniles and adults, however, 80–85% of serum IGFs are found in a 150-kDa ternary complex composed of one molecule each of IGF, IGFBP3, and a protein that is found only in the serum, the acid-labile subunit (ALS) (OMIM 601489)<sup>136</sup>. ALS is a glycoprotein that is mainly synthesized in the liver and is stimulated by GH<sup>137,138</sup>. This subunit stabilizes the IGF-I and IGFBP-3 complex, reduces the passage of IGF-I to the extravascular compartment and extends its life<sup>139</sup>.

The *IGFALS* gene, located on chromosome 16, consists of two exons. The mature ALS protein consists of 578 amino acids preceded by a 27 amino acid signal sequence<sup>140</sup>. The amino acid sequence of ALS is largely composed of 18–20 leucine-rich repeats of 24 amino acids which participate in protein–protein interactions.

An animal model where the *Igfals* gene had been inactivated (the *Igfals*knockout mouse) resulted in the offspring having a small growth deficit<sup>141</sup>. This modest phenotype was observed despite reductions of 62% and 88% in the concentrations of IGF-I and IGFBP-3, respectively. Therefore, the authors concluded that ALS was required for postnatal accumulation of IGF-I and IGFBP-3 but, consistent with findings supporting a predominant role for locally produced IGF-I, was not critical for growth.

Recently, a mutation was reported in *IGFALS* in a 17-year-old boy who had a delayed onset of puberty, slow pubertal progress, and yet minimal slowing of his linear growth<sup>142</sup>. ALS was undetectable in the serum before and after stimulation with GH. A frameshift mutation was due to a deletion of a guanine at nucleotide 1338 (1338delG) which resulted in the appearance

of a premature stop codon at position 120 (E35FsX120) of the precursor form of ALS. If the protein was expressed, secreted, and stable in the circulation, this mutant ALS protein would retain only seven amino-terminal amino acids after the cleavage of the signal peptide and would therefore lack the IGFBP-3 binding domain. The patient was thought to be homozygous for the mutation but parental DNA was not available. Although they were unable to prove that there was a relation between the ALS deficiency and the patient's clinical condition, many of the clinical and biochemical features were similar to the *Igfals* knockout mouse. Thus, it can be speculated that the lack of ALS might be involved in a subtle impairment of linear growth, a delay in the onset and slow progresses of puberty, and a certain degree of insulin resistance. The ALS deficiency caused by inactivation of the IGFALS gene is associated with a severe disruption in the GH-IGF-I axis. These recent findings support the hypothesis that the circulating total IGF-I level might not be the major mediator of the growth promoting actions of GH. The main roles of circulating IGF-I may be the feedback control of GH secretion and the regulation of carbohydrate metabolism through the facilitation of insulin action<sup>143</sup>. Further investigation may reveal whether this disorder represents a rare case of ALS deficiency caused by a rare molecular defect or whether mutations in IGFALS might be involved in other cases of delayed growth and pubertal development in children.

## 6. ALTERATIONS IN SEX CHROMOSOMES

# 6.1 Mutations in *SHOX* (Short stature homeoboxcontaining gene; OMIM \*312865)

In 1997, two groups isolated a candidate gene for short stature,  $SHOX^{144,145}$ . SHOX is located in the pseudoautosomal region (PAR1) of the short arm of the X and Y chromosomes<sup>144,145</sup> and thus it is present as two copies in both females and males. SHOX encodes a homeodomain protein that has been shown to act as a transcription factor. Rao *et al.*, demonstrated that an individual diagnosed with idiopathic short stature (ISS), carried a mutation in one copy of SHOX that cosegregated with the short stature phenotype in the family<sup>144</sup>. They also proposed that SHOX was also responsible for the short stature phenotype associated with Turner syndrome (45, XO). Shortly after, mutations and deletions in SHOX were identified in families with Léri-Weill dyschondrosteosis<sup>146,147</sup> (LWD, OMIM #127300), a skeletal dysplasia with short stature, which suggested that the same gene is also involved in the skeletal anomalies observed in Turner syndrome. It was

also shown that the more severe form of LWD, Langer mesomelic dysplasia (LMD) was due to the homozygosity of the *SHOX* defect<sup>147</sup> (OMIM #249700). Molecular cytogenetic studies showed that *SHOX* dosage was reduced in Turner syndrome, suggesting that haploinsufficiency of SHOX is responsible for not only the short stature in Turner syndrome but also for the skeletal abnormalities<sup>148,149</sup>.

SHOX spans 40 kb of genomic DNA in the 2.6 Mb PAR1 region, and is composed of six coding codons<sup>144,145</sup>. In Turner syndrome the majority of cases are the result of a lack of one X-chromosome (45, XO), thus lacking one copy of SHOX. In the other syndromes, the majority of defects found are also deletions of SHOX. Most of the single base pair mutations are nonsense mutations that cluster in the homeobox domain and the DNA binding domain of SHOX (SHOX database: http://www.shox.uni-hd.de/).

At the molecular level, there are two alternative splice variants of SHOX, SHOXa and SHOXb. SHOXa is the transcriptionally active form. It contains a SH3 binding site and sites for phosphorylation which are absent in SHOXb<sup>144,145</sup>. SHOXa is widely expressed in different tissues from embryogenesis to adolescence. In situ hybridization studies of SHOX expression during human embryogenesis have shown that expression occurs in the mid-limb regions; radius, ulna, elbow, wrist, tibia, fibula, knee, and ankle<sup>150</sup> and in the first and second pharyngeal arches. In contrast, the SHOXb isoform has a more restricted expression pattern, with the highest level in bone marrow fibroblasts, providing additional evidence for its role in linear growth<sup>144</sup>. Its temporal and spatial expression patterns have led to suggestions that SHOX acts as a key regulator of skeletal development in the middle portions of the long limbs. Recently SHOX was shown to be expressed in hypertrophic/apoptotic chondrocytes of the growth plate, strongly suggesting that the protein plays a direct role in regulating the differentiation of these cells<sup>151</sup>. They also demonstrated that SHOX expression leads to cell cycle arrest and apoptosis in osteogenic stable cell lines, primary oral fibroblasts, and primary chondrocytes. They provided evidence that mutations that delete the homeodomain of SHOX (ie., Cterminal truncations) failed to display these activities.

Currently, the only treatment for these disorders involving SHOX is GH therapy. Although growth velocity is increased in some patients, the underlying molecular mechanism remains unclear. One important question is: where does SHOX lie in the GH–IGF-I axis? Are there other treatments more suitable for this group of disorders? Further clinical and basic investigations will hopefully help to answer some of these points.

# 6.2 Panhypopituitarism

In a single family in which the males were deficient in GH, an in-frame duplication of 33 bp encoding for 11 alanines in a polyalanine tract of *SOX3* (OMIM \*313430) was identified<sup>152</sup>. The consequences of this mutation on the function of the protein are not known. The mammalian genome contains a family of genes that are related to *SRY*, the testis-determining gene. The homology is restricted to the region of *SRY* that encodes a DNA-binding motif of the *HMG*-box class (the DNA-binding domain is called *HMG* for high mobility group). These genes have been named SOX, for *SRY*-related *HMG*-box. *SOX3* is a single exon gene on the X chromosome<sup>152</sup> (Xq26.3). Conditional disruption of *Sox3* in mice now suggests that anterior pituitary development depends on *Sox3* expression in the overlying neural ectoderm, which establishes midline structures and regulates production of inductive BMP and FGF signals. Concentrations of pituitary GH, FSH, LH and TSH were lower in 2-month old *Sox3* mutant mice.

The hypopituitary phenotype of the *Sox3* null mice resembles that described for humans with an 11-mer polyalanine tract expansion in *SOX3*. Another group of individuals with X-linked hypopituitarism have duplications of Xp26 encompassing *SOX3* have mental retardation and hypopituitarism, and their phenotype is probably due to overexpression of this gene<sup>153</sup>. As these individuals do not have craniofacial defects, the polyalanine tract expansion is probably a loss-of-function mutation. Thus, if both gain- and loss-of-function mutations can lead to hypopituitarism, then pituitary development must be very sensitive to *SOX3* dosage. Analysis of *Sox3* gain-of-function mice and analysis of additional humans with *SOX3* alterations will be particularly informative in unravelling the developmental functions of this gene and its contribution to pituitary dysfunction in humans.

# 7. INTRAUTERINE GROWTH RETARDATION

Intrauterine growth retardation has multifactorial causes that are very complex, including nutritional problems, toxic agents, deficiencies confined to the placenta, chromosome anomalies and other genetic alterations. Primordial dwarfism constitutes a delay in growth of prenatal origin that continues during the postnatal period and that can be divided into two large groups, depending on whether it is associated or not with microcephaly. One specific form of primordial dwarfism without microcephaly is Silver-Russell syndrome (SRS; OMIM #180860). This syndrome is associated with a specific craniofacial phenotype, including triangular face, prominent ears, possible asymmetry of limbs, and clinodactily of the fifth finger. The cause of this syndrome is probably heterogeneous and the majority of the cases may actually be caused by dominant *de novo* mutations.

The first real indication of the molecular bases of this syndrome was the observation of uniparental maternal disomy of chromosome 7 in an individual homozygous for a mutation that causes cystic fibrosis where only the mother was a carrier<sup>154</sup>. This patient, as well as others described afterwards, had intrauterine growth retardation and postnatal growth retardation that was not justified by the cystic fibrosis. Uniparental maternal disomy of chromosome 7 was later found in 10% of the patients with primordial dwarfism belonging to the group of SRS<sup>155</sup>. Hence, the results indicated the existence of one or more genes on chromosome 7 that regulate growth and undergo gamete imprinting. These could be either genes that stimulate growth and are expressed only by the chromosome of paternal origin or growth inhibitory genes that are expressed only by the maternal chromosome. The finding of one familial case of SRS, a mother and daughter with a tandem duplication in the chromosomal 7p13-p11.2 region, has allowed the critical region of chromosome 7 to be defined to a region that contains the genes for IGFBP1, IGFBP3, and growth factor receptorbinding protein 10  $(GRB10)^{156}$ . Subsequently, another patient with a similar duplication has been described. Molecular characterization of this patient demonstrated that the above mentioned genes were included and the duplicated region was of maternal origin<sup>157</sup>. Therefore, there must be one or more growth inhibiting genes in this region of chromosome 7 that are exclusively expressed from the maternal chromosome. Increased expression of this gene (or genes) either by maternal uniparental disomy or maternal duplications of the critical region will cause growth retardation. The GRB10 gene is a possible candidate for several reasons: (1) It is located in the critical region of the described duplications. (2) The homologue in mice (the meg1/Grb10 gene located on chromosome 11 is expressed exclusively by the maternal chromosome. (3) It is probably implicated in the growth defects that are observed in mice with duplication of chromosome 11 and reciprocal deficiency<sup>158</sup>. The known function of the GRB10 protein is also suggestive of a role in growth control. GRB10 binds the insulin receptor and IGF-IR via an SH2 domain and inhibits the tyrosine kinase activity associated with the receptor that is implicated with the growth promoting actions of insulin, IGF-I and IGF-II<sup>159</sup>. Yoshihashi et al., performed mutation analysis of the GRB10 gene in 58 unrelated patients with SRS and identified a P95S substitution within the N-terminal domain in two of the patients<sup>160</sup>. Evidence creating uncertainty about the role of *GRB10* in SRS has been reported<sup>161-163</sup>. Doubt was cast by the absence of mutations detected by sequencing GRB10 in classic SRS patients, in whom major structural chromosomal

abnormalities and maternal uniparental disomy had previously been excluded.

#### 8. FINAL REMARKS

The many advances in molecular biology and genetics over the last decade have facilitated the studies of the GH/IGF-I axis in humans and mice yielding great insights into pituitary development and control. The continued identification and characterization of novel factors involved in this complex growth cascade will lead to a better understanding of the GH/IGF-I axis and may provide new therapies for patients affected for these disorders.

All of the classic Mendelian patterns of inheritance have been demonstrated human undergrowth (autosomal, in X-linked. pseudoautosomal dominant). In fact, mutations in most of these genes have been implicated as the cause of abnormal and altered growth patterns from intrauterine life to childhood and puberty. In addition to the genetic component in familial cases, a high proportion of the sporadic cases may be also due to genetic abnormalities. The physiological bases of the GH/IGF-I axis are very complex as revealed by the increasing number of different genes that have been related with the etiopathogenesis of proportional short stature. Indeed, either GH deficiency or GH resistance, as well as IGF-I deficiency and IGF-I resistance can all produce a similar human phenotype due to the absolute or relative lack of IGF-I. The genes involved include the pituitary transcription factors POU1F1, PROP1, LHX3, LHX4, and HESX1, and genes directly or indirectly related with GH expression, release and signal transduction like GH1, GHRHR, GHR, STAT5b, IGF1, IGF1R, IGFALS, SHOX, BTK, and SOX3.

Furthermore, some potential candidates including *IGFBP-3* and *GRB10*, in particular in patients with proportional undergrowth of prenatal origin should be considered. Hence, the molecular studies in patients with prenatal abnormalities in growth provide a very exciting field of research at present and in the near future. In addition, the molecular bases of short stature due to gene abnormalities on both the X and Y chromosomes, is now under intense investigation, since the mechanisms that regulate growth are not well understood or remain completely unknown.

Therefore, the clinician should further investigate children with a phenotype of IGF-I deficiency, not only from an auxological, biochemical and radiological point of view, but also using appropriate molecular techniques.

In conclusion, the advances in molecular biology and genetics are rapidly unravelling the complex pathology of human growth. The number of disorders with a known molecular genetic basis is dramatically increasing, and the list will continue to grow in the near future. Therefore, it is relevant for physicians and scientists to be aware of all advances and novel techniques available for achieving a better diagnosis, appropriate family counselling and patient care. Although our knowledge has advanced rapidly in this field, an international consensus or clinical guidelines for a specific molecular diagnosis of proportional short stature remains to be established.

#### REFERENCES

- 1. L.A. Pérez Jurado, J. Argente, J. Pozo, V. Barrios, M. Hernández and U. Francke, Molecular diagnosis and endocrine evaluation of a patient with a homozygous 7.0 kilobase (kb) deletion of the growth hormone gene cluster: Achievement of an outstanding response to biosynthetic GH therapy and evidence indicating that growth hormone-binding hormone is not regulated by GH. *J Pediatr Endocrinol Metab*, **10**, 185-190 (1997).
- R. Lindsay, M. Feldkamp, D. Harris, J. Robertson and M. Rallison, Utah Growth Study: growth standards and the prevalence of growth hormone deficiency. *J Pediatr*, **125**, 29-35 (1994).
- 3. J. Argente, L.A. Pérez-Jurado and J.F. Sotos, Molecular basis of pathological growth. J Endocrine Genet, 1, 179-210 (2000).
- 4. P.E. Mullis, A. Akinci, C. Kanaka and C.G. Brook, Prevalence of human growth hormone-1 gene deletions among patients with isolated growth hormone deficiency from different populations. *Ped Res*, **31**, 532-534 (1992).
- 5. H. Hirt, J. Kimelman, M.J. Birnbau, E.Y. Chen, P.H. Seeburg, N.L. Eberhardt and A. Barta, The human growth hormone gene locus: Structure, evolution and allelic variations. *DNA*, **6**, 59-70 (1987).
- 6. G. Binder, Isolated growth hormone deficiency and the GH-1 gene: update 2002. *Horm Res*, **58**, S2-6 (2002).
- J.D. Cogan, J.A. Phillips III, N.A. Sakati, F. Frisch, E. Schoeber and D. Milner, Heterogeneous growth hormone (GH) gene mutations in familial GH deficiency. J Clin Endocrinol Metab, 76, 1224-1228 (1993).
- 8. A.M. Procter, J.A. Phillips III and D.N. Cooper, The molecular genetics of growth hormone deficiency. *Hum Genet*, **103**, 255-272 (1998).
- S.C. Lin, C.R. Lin, I. Gukovsky, A.J. Lusis, P.E. Sawchenko and M.G. Rosenfeld, Molecular basis of the little mouse phenotype and implications for cell type-specific growth. *Nature*, 364, 208-213 (1993).
- 10. M.P. Wajnrajch, J.M. Gertner, M.D. Harbison, S.C. Chua and R.L. Leibel, Nonsense mutation in the human growth hormone-releasing hormone receptor causes growth failure analogous to the little (lit) mouse. *Nat Genet*, **12**, 88-90 (1996).
- 11. M.P. Wajnrajch, J.M. Gertner, A.S. Sokoloff, I. Ten, M.D. Harbison, I. Netchine, H.G. Maheshwari, D.B. Goldstein, S. Amselem, G. Baumann and L.R. Leibel, Haplotype analysis of the growth hormone releasing hormone receptor locus in three apprently unrelated kindreds from the indian subcontinent with the identical mutation in the GHRH receptor. *Am J Med Genet*, **120A**, 77-83 (2003).
- 12. R. Salvatori, C.Y. Hayashida, M.H. Aguiar-Oliveira, J.A. Phillips III, A.H.O. Souza, R.G. Gondo, S.P.A. Toledo, M.M. Conceicao, M. Prince, H.G. Maheshwari,

G.Baumann, M.A. Levine, Familial dwarfism due to a novel mutation of the growth hormone-releasing hormone receptor gene. *J Clin Endocr Metab*, **84**, 917-923 (1999).

- 13. R. Salvatori, M.H. Aguiar-Oliveira, L.V. Monte, L. Hedges, N.L. Santos, R.M. Pereira and J.A. Phillips, Detection of a recurring mutation in the human growth hormone-releasing hormone receptor gene. *Clin Endocrinol (Oxf)*, **57**, 77-80 (2002).
- R. Salvatori, X. Fan, P.E. Mullis, A. Haile and M.A. Levine, Decreased expression of the GHRH receptor gene due to a mutation in a Pit-1 binding site. *Mol Endocrinol*, 16, 450-458 (2002).
- 15. R. Salvatori, X. Fan, J.D. Veldhuis and R. Couch, Serum GH response to pharmacological stimuli and physical exercise in two siblings with two new inactivating mutations in the GH-releasing hormone receptor gene. *Eur J Endocrinol*, **147**, 591-596 (2002).
- H.A. Oliveira, R. Salvatori, M.P. Krauss, C.R. Oliveira, P.R. Silva and M.H. Aguiar-Oliveira, Magnetic resonance imaging study of pituitary morphology in subjects homozygous and heterozygous for a null mutation of the GHRH receptor gene. *Eur J Endocrinol*, 148, 427-432 (2003).
- M. Carakushansky, A.J., Whatmore, P.E. Clayton, S.M. Shalet, H.K. Gleeson, D.A. Price, M.A. Levine and R. Salvatori, A new missense mutation in the growth hormonereleasing hormone receptor gene in familial isolated GH deficiency. *Eur J Endocrinol*, 148, 25-30 (2003).
- L.A. Pérez Jurado, J.A. Phillips III and U. Francke, Exclusion of growth hormonereleasing hormone gene mutations in familial isolated growth hormone deficiency by linkage and single strand conformation analysis. *J Clin Endocrinol Metab*, **78**, 622-628 (1994).
- F. Roelfsema, N.R. Biermasz, R.G. Veldman, J.D. Veldhuis, M. Frolich, W.H. Stovkis-Brantsma and J.M. Wit, Growth hormone (GH) secretion in patients with an inactivating defect of the GH-releasing hormone (GHRH) receptor is pulsatile: evidence for a role for non-GHRH inputs into the generation of GH pulses. J Clin Endocrinol Metab, 86, 2459-2464 (2001).
- 20. B.D. Gaylinn, C. von Kap-Herr, W. Golden and M.O. Thorner, Assignment of the human growth hormone-releasing hormone receptor gene (GHRHR) to 7p14 by in situ hybridization. *Genomics*, **19**, 193-195 (1994).
- 21. J.D. Cogan, J.A. Phillips III, S.S. Schenkman, R.D.G. Milner and N. Sakati, Familial growth hormone deficiency: a model of dominant and recessive mutations affecting a monomeric protein. *J Clin Endocrinol Metab*, **79**, 1261-1265 (1994).
- H. Abdul-Latif, E. Leiberman, M.R. Brown, R. Carmi and J.S. Parks, Growth hormone deficiency type IB caused by cryptic splicing of the *GH-1* gene. J Pediatr Endocrinol Metab, 13, 21-28 (2000).
- 23. E. Leiberman, D. Pesler, R. Parvari, K. Elbedour, H. Abdul-Latif, M.R. Brown, J.S. Parks and R. Carmi, Short stature in carriers of recessive mutation causing familial isolated growth hormone deficiency. *Am J Med Genet*, **90**, 188-192 (2000).
- J.A. Phillips III, Inherited defects in growth hormone synthesis and action. (Scriver CR, Beaudet AL, Sly WS, Valle D, eds.) The metabolic basis of inherited disease. 6<sup>a</sup> ed. New York: McGraw-Hill, 1989; 1965-1983.
- 25. R.C.C. Ryther, L.M. McGuinness, J.A. Phillips III, C.T. Moseley, C.B. Magoulas, I.C.A.F. Robinson and J.G. Patton, Disruption of exon definition produces a dominant negative growth hormone isoform that causes somatotroph death and IGHD II. *Hum Genet*, **113**, 140-148 (2003).

- G. Binder, E. Keller, M. Mix, G.G. Massa, W.H. Stokvis-Brantsma, J.M.Wit and M.B. Ranke, Isolated GH deficiency with dominant inheritance: new mutations, new insights. *J Clin Endocrinol Metab*, 86, 3877-3881 (2001).
- 27. J. Deladoey, P. Stocker and P.E. Mullis, Autosomal dominant GH deficiency due to an Arg 183His GH-1 gene mutation: clinical and molecular evidence of impaired regulated GH secretion. *J Clin Endocrinol Metab*, **86**, 3941-3947 (2001).
- D.S. Miller, M.D. Lewis, M. Horan, V. Newsway, T.E. Easter, J.W. Gregory, L. Fryklund, M. Norin, E.C. Crowne, S.J. Davies, P. Edwards, J. Kirk, K. Waldron, P.J. Smith, J.A. Phillips III, M.F. Scanlon, M. Krawczak, D.N. Cooper and A.M. Proctor. Novel mutations of the growth hormone 1 (*GH1*) gene disclosed by modulation of the clinical selection criteria for individuals with short stature. *Hum Mutat*, 21, 424-440 (2003).
- 29. T.A. Fleisher, R.M. White, S. Broder, S.P., Nissley, R.M. Blaese, J.J. Mulvihill, G.Y. Olive and T.A. Waldmann, X-linked hypogammaglobulinemia and isolate growth hormone deficiency. *N Engl J Med*, **302**, 1429-1434 (1980).
- M.E. Conley, A.W. Burks, H.G. Herrog and J.M. Puck, Molecular analysis of X-linked agammaglobulinemia with growth hormone deficiency. *J Pediatr*, 119, 392-397 (1992).
- 31. R. Rapaport, J. Oleske, H. Ahdieh, S. Solomon and S. Delfaus, Suppression of immune function in growth hormone deficient children during treatment with human growth hormone. *J Pediatr*, **109**, 434-439 (1993).
- 32. T. Ogata, C. Petit, G. Rappold, N. Matsuo, T. Matsumoto and P. Goodfellow, Chromosomal localization of (a) pseudoautosomal growth gene(s). *J Med Genet*, **29**, 624-628 (1992).
- B. Duriez, P. Duquesnoy, F. Dastot, P. Bougneres, S. Amselem and M. Goossens, An exon-skipping mutation in the btk gene of a patient with X.linked agammaglobulinemia and isolated growth hormone deficiency. *FEBS Lett*, **346**, 165-170 (1994).
- 34. K. Abo, H. Nishio, M.J. Lee, D. Tsuzuki, T. Takahashi, S. Yoshida, T. Nakajima, M. Matsuo and K. Sumino, A novel single basepair insertion in exon 6 of the Bruton's tyrosine kinase (Btk) gene from a Japanese X-linked agammaglobulinemia patient with growth hormone insufficiency. *Hum Mutat*, **11**, 336 (1998).
- 35. Y. Yokohama, K. Narahara, K. Tsuji, et al. Growth hormone deficiency and empty sella syndrome in a boy with dup(X)(q13.3-q21.2). *Am J Med Genet*, **42**, 660-664 (1992).
- 36. M. Giordano, C. Marchetti, E. Chiorboli, G. Bona and P.M. Richiardi, Evidence for gene conversion in the generation of extensive polymorphism in the promoter of the growth hormone gene. *Hum Genet*, **100**, 249-255 (1997).
- 37. J.K. Wagner, A. Eblé, J.D. Cogan, M.A. Prince, J.A. Phillips and P.E. Mullis, Allelic variations in the human growth hormone-1 gene promoter of growth hormone-deficient patients and normal controls. *Eur J Endocrinol*, **137**, 474-481 (1997).
- M. Horan, D.S. Miller, J. Hedderich, G. Lewis, V. Newsah, N. Mo, L. Fryklund, A.M. Procter, M. Krawczak and D.N. Cooper, Human growth hormone 1 (*GH1*) gene expression: Complex haplotype-dependent influence of polymorphic variation in the proximal promoter and locus control region. *Hum Mutat*, 21, 408-423 (2003).
- B.K. Jones, B.R. Monks, S.A. Liebhaber and N.E. Cooke, The human growth hormone gene is regulated by a multi-component locus control region. *Mol Cell Biol*, 15, 7010-7021 (1995).
- 40. T. Su, S.A. Liebhaber and N.E. Cooke, The human growth hormone gene cluster locus control region supports position-independent pituitary- and placenta-specific expression in the transgenic mouse. *J Biol Chem*, **275**, 7902-7909 (2000).

- 41. Y. Ho, E. Elefant, N. Cooke and S. Liebhaber, A defined locus control region determinant links chromatin domain acetylation with long-range gene activation. *Mol Cell*, **9**, 291-302 (2002).
- 42. F.P. Lemaigre, B. Peers, D.A. Lafontaine, M. May-Hartert, G.G. Rousseau, A. Belayew and J.A. Martial, Pituitary-specific factor binding to the human prolactin, growth hormone and placental lactogen genes. *DNA*, 8, 149-159 (1989).
- 43. Y. Jin, R.M. Surabhi, F. Fresnoza, A. Lytras and P.A. Cattin, A role for A/T-rich sequences and Pit-1/GHF-1 in a distal enhancer located in the human growth hormone locus control region with preferential pituitary activity in culture and transgenic mice. *Mol Endocrinol*, **13**, 1249-1266 (1999).
- 44. B.M. Shewchuk, S.L. Asa, N.E. Cooke and S.A. Liebhaber, Pit-1 binding sites at the somatotrope-specific DNase 1 hypersensitive sites I, II of the human growth locus control region are essential for in vivo hGH-N gene activation. *J Biol Chem*, 274, 35725-35733 (1999).
- 45. B.M. Shewchuk, S.A. Liebhaber and N.E. Cooke, Specification of unique Pit-1 activity in the hGH locus control region. *Proc Natl Acad Sci USA*, **99**, 11784-11789 (2002).
- 46. A.A. Kowarski, J. Schneider, E. Ben-Galim, V.V. Weldon and W.H. Daughaday, Growth failure with normal serum RIA-GH and low somatomedin activity: somatomedin restoration and growth acceleration after exogenous GH. J Clin Endocrinol Metab, 47, 461-464 (1978).
- 47. Y. Takahashi, H. Kaji, Y. Okinura, K. Goji, H. Abe and K. Chihara, Short stature caused by a mutant growth hormone with an antagonistic effect. *New Engl J Med*, **334**, 432-436 (1996).
- Y. Takahashi, H. Shirono, O. Arisaka, K. Takahashi, T. Yagi, J. Koga, H. Kaji, Y. Okinura, H. Abe, T. Tanaka and K. Chihara K. Biologically inactive growth hormone caused by amino acid substitution. *J Clin Invest*, 100, 1159-1165 (1997).
- 49. M.D. Lewis, M. Horan, D.S. Millar, V. Newsway, T.E. Easter, L. Fryklund, J.W. Gregory, M. Norin, C.J. Del Valle. J.P. López-Siguero, R. Cañete, L.F. Lopez-Canti, N. Díaz-Torrado, R. Espino, A. Ulied, M.F. Scanlon, A.M. Procter and D.N. Cooper, A novel dysfunctional growth hormone variant (Ile179Met) exhibits a decreased ability to activate the extracellular signal-regulated kinase pathway. *J Clin Endocrinol Metab*, **89**, 1068-1075 (2004).
- 50. J.S. Parks, M.R. Brown, D.L. Hurley, C.J. Phelps and M.P. Wajnrajch, Heritable disorders of pituitary development. *J Clin Endocrinol Metab*, **84**, 4362-4370 (1999).
- 51. P.E. Mullis, Transcription factors in pituitary gland development and their clinical impact on phenotype. *Horm Res*, **54**, 107-119 (2000).
- 52. B.I. Hendriks-Stegeman, K.D. Augustijn, B. Bakker, P. Holthuizen, P.C. van der Vliet and M. Jansen, Combined pituitary hormone deficiency caused by compound heterozygosity for two novel mutations in the POU domain of the Pit1/POU1F1 gene. J *Clin Endocrinol Metab*, **86**, 1545-1550 (2001).
- 53. M.W. Sornson, W. Wu, J.S. Dasen, S.E. Flynn, D.J. Norman, S.M. O'Connell, I. Gukovsky, C. Carriere, A.K. Ryan, A.P. Miller, L. Zuo, A.S. Gleiberman, B. Andersen, W.G. Beamer and M.G. Rosenfeld, Pituitary lineage determination by the Prophet of Pit-1 homeodomain factor defective in Ames dwarfism. *Nature*, 384, 327-333 (1996).
- 54. R.W. Pfaffle, G. DiMattia, J.S. Parks, M.R. Brown, J.M. Wit, M. Jansen, H. Van der Nat, J.L. Van en Brande, M.G. Rosenfeld and H.A. Ingraham, Mutation of the POU-specific domain of Pit-1 as a cause of hypopituitarism without pituitary hypoplasia. *Science*, **257**, 1118-1121 (1992).

- 55. K. Ohta, Y. Nobukuni, H. Mitsubuchi, T. Ohta, T. Tohma, Y. Jinno, F. Endo and I. Matsuda, Characterization of the gene encoding human pituitary-specific transcription factor, Pit-1. *Gene*, **122**, 387-388 (1992).
- L.E. Theill and M. Karin, Transcriptional control of GH expression and anterior pituitary development. *Endocr Rev*, 14, 670-689 (1993).
- M. Bodner, J.L. Castrillo, L. Theill, T. Deernick, M. Ellisman and M. Karin, The pituitary-specific transcription factor GHF-1 is a homeobox-containing protein. *Cell*, 55, 508-518 (1988).
- H.A. Ingraham, R. Chen, H.J. Mangalam, H.P. Elsholtz, S.E. Flynn, C.R. Lin, D.M. Simmons, L. Swanson and M.G. Rosenfeld, A tissue-specific transcription factor containing a homeodomain specifies a pituitary phenotype. *Cell*, 55, 519-529 (1988).
- 59. R.W. Pfäffle, C. Kim, B. Otten, J.M. Wit, U. Eiholzer, G. Heimann and J. Parks, Pit-1: Clinical aspects. *Horm Res*, **45**, S25-28 (1996).
- K. Tatsumi, K. Miyai, T. Notomi, K. Kaibe, N. Amino, Y. Mizuno and H. Kohno, Cretinism with combined hormone deficiency caused by a mutation in the PIT1 gene. *Nat Genet*, 1, 56-58 (1992).
- 61. S. Radowick, M. Nations, Y. Du, L.A. Berg, B.D. Weintraub and F.E. Wondirsford FE, A mutation in the POU-homeodomain of Pit-1 responsible for combined pituitary hormone deficiency. *Science*, **257**, 1115-1118 (1992).
- 62. N. Okamoto, Y. Wada, S. Ida, R. Koga, K. Ozono, H. Chiyo, A. Hayashi and K. Tatsumi, Monoallelic expression of normal mRNA in the PIT1 mutation heterozygotes with normal phenotype and biallelic expression in the abnormal phenotype. *Hum Mol Genet*, **3**, 1565-1568 (1994).
- 63. S. Mody, M.R. Brown and J.S. Parks, The spectrum of hypopituitarism caused by *PROP1* mutations. *Best Practice & Research in Clin Endocrinol Metab*, **16**, 421-431 (2002).
- W. Wu, J.D. Cogan, R.W. Pfäffle, J.S. Dasen, H. Frisch, M.O. O'Connell, S.E. Flynn, M.R. Brown, P.E. Mullis, J.S. Parks, J.A. Philips III and M.G. Rosenfeld, Mutations in PROP1 cause familial combined pituitary hormone deficiency. *Nat Genet*, 18, 147-149 (1998).
- 65. G. Agarwal, V. Bathia, S. Cook and P.Q. Thomas, Adrenocorticotropin deficiency in combined pituitary hormone deficiency patients homozygous for a novel PROP1 mutation. *J Clin Endocrinol Metab*, **85**, 4556-4561 (2000).
- 66. F. Pernasetti, S.P.A. Toled, V.V. Vasilyev, C.Y. Hayashida, J.D. Cogan, C. Ferrari, D.M. Lourenco and P.M. Mellon, Impaired adrenocorticotropin-adrenal axis in combined pituitary hormone deficiency caused by a two base pair deletion (301-302delAG) in the Prophet of Pit-1 gene. J Clin Endocrinol Metab, 85, 390-397 (2000).
- J.D. Cogan, W. Wu, J.A. Phillips III, J.P.A. Ivo, A. Agapito, O.V. Fofanova, M.G.F. Osorio, I. Birgan, A. Moreno and B.B. Mendonca, The Prop1 2-base pair deletion is a common cause of combined pituitary hormone deficiency. *J Clin Endocrinol Metab*, 83, 3346-3349 (1998).
- 68. S. Schmitt, A. Biason-Lauber, D. Betts and E.J. Schoenle, Genomic structure, chromosomal localisation, and expression pattern of the human LIM-homeobox3 (LHX3) gene. *Biochem Biophys Res Commun*, **274**, 49-56 (2000).
- I. Netchine, M.L. Sobrier, H. Krude, D. Schnabel, M. Maghnie, E. Marcos, B. Duriez, V. Cacheux, A. Moers, M. Goossens, A. Grüters and S. Amselem, Mutations in LHX3 result in a new syndrome revealed by combined pituitary hormone deficiency. *Nat Genet*, 25, 182-186 (2000).

- H.Z. Sheng, A.B. Zhadanov, B. Mosinger Jr, T. Fujii, S. Bertuzzi, A. Gringberg, E.J. Lee. S.P. Huang, K.A. Mahon and H. Westphal, Specification of pituitary cell lineages by LIM homeobox gene Lhx3. *Science*, 272, 1004-1007 (1996).
- H.Z. Sheng, K. Moriyama, T. Yamashita, H. Li, S.S. Potter, K.A. Mahon and H. Westphal, Multistep control of pituitary organogenesis. *Science*, 278, 1809-1812 (1997).
- K. Machinis, J. Pantel, I. Netchine, J. Léger, O.J.A. Camand, M.L. Sobrier, F. Dastot-Le Moal, P. Duquesnoy, M. Abibtol, P. Czernichow and S. Amselem, Syndromic short stature in patients with a germline mutation in the LIM homeobox LHX4. Am J Hum Genet, 69, 961-968 (2001).
- 73. E. Hermesz, S. Mackem and K.A. Mahon, Rpx: a novel anterior-restricted homeobox gene progressively activated in the prechordal plate, anterior neural plate, and Rathke's pouch of the mouse embryo. *Development*, **122**, 41-52 (1996).
- M.T. Dattani, J.P. Martiez-Barbara, P.Q. Thomas, J.M. Brickman, R. Gupta, I.L. Martensson, H. Toresson, M. Fox, J.K.H. Wales, P.C. Hindmarsh, S. Krauss, R.S.P. Beddington and I.C.A.F. Robinson, Mutations in the homeobox gene HESX1/Hesx1 associated with septo-optic dysplasia in human and mouse. *Nat Genet*, **19**, 125-133 (1998).
- 75. G. de Morsier, Etudes sur les dysraphies cranio-encephaliques: III: Agenesie du septum lucidum avec malformation du tractus optique: La dysplasie septo-optique. *Schweizer* Arch. Neurol. Psychiatr, **77**, 267-292 (1956).
- S.A. Arslanian, W.E. Rothfus, T.P. Foley Jr and D.J. Becker, Hormonal, metabolic, and neuroradiologic abnormalities associated with septo-optic dysplasia. *Acta Endocrinol* (*Copenh*), **107**, 282-288 (1984).
- 77. N. Izenberg, M. Rosenblum and J.S. Parks, The endocrine spectrum of septo-optic dysplasia. *Clin Pediatr (Phila)*, **23**, 632-636 (1984).
- U. Roessmann, Septo-optic dysplais or DeMorsier syndrome. J Clin Neuroophthalmol, 9, 156-159 (1989).
- J.M. Brickman, M. Clements, R. Tyrell, D. McNay, K. Woods, J. Warner, A. Stewart, R.S. Beddington and M. Dattani, Molecular effects of novel mutations in Hesx1/HESX1 associated with human pituitary disorders. *Development*, **128**, 5189-5199 (2001).
- P.Q. Thomas, M.T. Dattani, J.M. Brickman, D. McNay, G. Warne, M. Zacharin, F. Cameron, J.K. Hurst, K. Woods, D. Dunger, R. Stanhope, S. Forrest, I.C.A.F. Robinson and R.S.P. Beddington, Heterozygous *HESX1* mutations associated with isolated congenital pituitary hypoplasia and septo-optic dysplasia. *Hum Mol Genet*, **10**, 39-45 (2001).
- L.R. Carvalho, K.S. Woods, B.B. Mendonca, N. Marcal, A.L. Zamparini, S. Stifani, J.M. Brickman, I.J. Arnhold and M.T. Dattani, A homozygous mutation in *HESX1* is associated with evolving hypopituitarism due to impaired repressor-corepressor interaction. J Clin Invest, 112, 1192-1201 (2003).
- T. Tajima, T. Hattorri, T. Nakajima, K. Okuhara, K. Sato, S. Abe, J. Nakae and K. Fujieda, Sporadic heterozygous frameshift mutation in HESX1 causing pituitary and optic nerve hypoplasia and combined pituitary hormone deficiency in a Japanese patient. J Clin Endocrinol Metab, 88, 45-50 (2003).
- 83. R.N. Cohen, L.E. Cohen, D. Botero, C. Yu, A. Sagar, M. Jurkiewicz and S. Radovick, Enhanced repression by HESX1 as a cause of hypopituitarism and septooptic dysplasia. *J Clin Endocrinol Metab*, **8**, 4832-4839 (2003).

- Z. Laron, A. Pertzelan and S. Mannheimer, Genetic pituitary dwarfism with high serum concentration of growth hormone. A new inborn error of metabolism? *Isr J Med Sci*, 2, 152-155 (1966).
- 85. Z. Laron, A. Pertzelan and M. Karp, Pituitary dwarfism with high serum levels of growth hormone. *Isr J Med Sci*, 4, 883-894 (1968).
- A. Pertzelan, A. Adam and Z. Laron, Genetic aspects of pituitary dwarfism due to absence of biological activity of growth hormone. *Isr J Med Sci*, 4, 895-900 (1968).
- Z. Laron, P. Lilos and B. Klinger, Growth curves for Laron syndrome. *Arch Dis Child*, 68, 768-770 (1993).
- R.G. Rosenfeld, A.L. Rosenbloom and J. Guevara-Aguirre, Growth hormone (GH) insensitivity due to primary GH receptor deficiency. *Endocr Rev*, 15, 369-390 (1994).
- K.A. Woods, F. Dastot, M.A. Perece, A.J. Clark, M.C. Postel-Vinay, P.G. Chatelain, M.B. Ranke, R.G. Rosenfeld, S. Amselem and M.O. Savage, Phenotype: genotype relationships in growth hormone insensitivity syndrome. *J Clin Endocrinol Metab*, 82, 3529-3535 (1997).
- 90. R. Eshet, Z. Laron, A. Pertzelan and M. Dintzman, Defect of human growth hormone in the liver of two patients with Laron type dwarfism. *Isr J Med Sci*, **20**, 8-11 (1984).
- 91. D.E. Barton, B.E. Foellmer, W.I. Woods and U. Francke, Chromosome mapping of the growth hormone receptor gene in man and mouse. *Cytogenet Cell Genet*, **50**, 137-141 (1989).
- 92. Z. Laron, Laron syndrome (primary growth hormone resistance or insensitivity): the personal experience 1958-2003. *J Clin Endocrinol Metab*, **89**, 1031-1044 (2004).
- J. Wojcik, M.A. Berg, N. Esposito, M.E. Geffner, N. Sakati, E.O. Reiter, S. Dower, U. Francke, M.C. Postel-Vinay and J. Finidori, Four contiguous amino acid substitutions, identified in patients with Laron syndrome, differently affect the binding affinity and intracellular trafficking of the growth hormone receptor. *J Clin Endocrinol Metab*, 83, 4481-4489 (1998).
- 94. J. Gent, M. Van den Eijnden, P. Van Kerkhof and G.J. Strous, Dimerization and signal transduction of the growth hormone receptor. *Mol Endocrinol*, **17**, 967-975 (2003).
- 95. K.A. Woods, N.C. Fraser, M.C. Postel-Vinay, M.O. Savage and A.J. Clark, A homozygous splice site mutation affecting the intracellular domain of the growth hormone (GH) receptor resulting in Laron syndrome with elevated GH-binding protein. J Clin Endocrinol Metab, 81, 1686-1690 (1996).
- 96. H. Kaji, O. Nose, H. Tajiri, Y. Takahashi, K. Iida, T. Takahashi, Y. Okimura, H. Abe and K. Chihara, Novel compound heterozygous mutations of growth hormone (GH) receptor gene in a patient with GH insensitivity syndrome. *J Clin Endocrinol Metab*, 82, 3705-3709 (1997).
- R.M. Ayling, R. Ross, P. Towner, S. Von Laue, J. Finidori, S. Moutoussamy, C.R. Buchanan, P.E. Clayton and M.R. Norman, A dominant-negative mutation of the growth hormone receptor causes familial short stature. *Nat Genet*, 16, 13-14 (1997).
- 98. K. Iida, Y. Takahashi, H. Kaji, O. Nose, Y. Okimura, H. Abe and K. Chihara, Growth hormone (GH) insensitivity syndrome with high serum GH-binding protein levels caused by a heterozygous splice site mutation of the GH receptor gene producing a lack of intracellular domain. J Clin Endocrinol Metab, 83, 531-537 (1998).
- J.M. Gastier, M.A. Berg, P. Vesterhus, E.O. Reiter U. Francke, Diverse deletions in the growth hormone receptor gene cause growth hormone insensitivity syndrome. *Hum Mutat*, 16, 323-333 (2000).
- 100. P.J. Godowski, D.W. Leung, L.R. Meacham, J.P. Galgani, R. Hellmiss, R. Keret, P.S. Rotwein, J.S. Parks, Z. Laron and W.I. Wood, Characterization of the human growth

hormone receptor gene and demonstration of a partial gene deletion in two patients with Laron-type dwarfism. *Proc Natl Acad Sci USA*, **86**, 8083-8087 (1989).

- 101. J.A. Wells, Binding in the growth hormone receptor complex. *Proc Natl Acad Sci USA*, **93**, 1-6 (1996).
- L.S. Argetsinger, G.S. Campbell, X. Yang, B.A. Witthuhn, O. Silvennoinen, J.N. Ihle and C. Carter-Su. Identification of JAK2 as a growth hormone receptor-associated tyrosine kinase. *Cell*, 74, 237-244 (1993).
- P. Colosi, K. Wong, S.R. Leong and W.I. Wood, Mutational analysis of the intracellular domain of the human growth hormone receptor. J Biol Chem, 268, 12617-12623 (1993).
- U.D. Wang and W.I. Wood, Amino acids of the human growth hormone receptor that are required for proliferation and Jak-STAT signalling. *Mol Endocrinol*, 9, 303-311 (1995).
- 105. L.A. Winston and P.J. Bertics, Growth hormone stimulates the tyrosine phosphorylation of 42- and 45- kDa ERK-related proteins. *J Biol Chem*, **267**, 4747-4751 (1992).
- 106. T. Zhu and P.E. Lobie, Janus kinase 2-dependent activation of p38 mitogen-activated protein kinase by growth hormone. Resultant transcriptional activation of ATF-2 and CHOP, cytoskeletal re-organization and mitogenesis. *J Biol Chem*, **2375**, 2103-2114 (2000).
- 107. T. Zhu, E.I. Goh, D. LeRoith and P.E. Lobie, Growth hormone stimulates the formation of a multiprotein signalling complex involving p130(Cas) and CrkII. Resultant activation of c-Jun N-terminal kinase/stress-activated protein kinase (JNK/SAPK). J Biol Chem, 273, 33861-33875 (1998).
- 108. J. Herrington and C. Carter-Su, Signaling pathways activated by the growth hormone receptor. *Trends Endocrinol Metab*, **12**, 252-257 (2001).
- 109. W.H. Daughaday, Are there direct, non-IGF-I-mediated effects of hGH? in Lessons from Laron Syndrome (LS) 1966-1992. In: *Pediatric and Adolescent Endocrinology*, vol. 24, Z. Laron and J.S. Parks, Eds, Basel-New York (1993).
- P.A. Kelly, L. Goujon, A. Sotiropoulos, H. Dinerstein, N. Esposito, M. Edery, J. Finidori and M.C. Postel-Vinay, The GH receptor and signal transduction. *Horm Res*, 42, 133-139 (1994).
- 111. A. Sotiropoulos, S. Moutoussamy, F. Renaudie, M. Clauss, C. Kayser, F. Gouilleux, P.A. Kelly and J. Finidori, Differential activation of Stat3 and Stat5 by distinct regions of the growth hormone receptor. *Mol Endocrinol*, **10**, 998-1009 (1996).
- 112. A. Milward, L. Metherell, M. Mamara, M.J. Barahona, I.R. Wilkinson, C. Camacho-Hübner, M.O. Savage, C.M. Bidlingmaier, A.J.L. Clark, R.J.M. Ross and S.M. Webb, Growth hormone (GH) insensitivity syndrome due to a GH receptor truncated after Box1, resulting in isolated failure of STAT5 signal transduction. J Clin Endocrinol Metab, 89, 1259-1266 (2004).
- 113. E.M. Kofoed, V. Hwa, B. Little, K.A. Woods, C.K. Buckway, J. Tsubaki, K.L. Pratt, L. Bezrodnik, H. Jasper, A. Tepper, J.J. Heinrich and R.G. Rosenfeld, Growth hormone insensitivity associated with a STAT5b mutation. *N Eng J Med*, **349**, 1139-1147 (2003).
- 114. S.L. Chew, P. Lavender, A.J. Clark and R.J. Ross, An alternatively spliced human insulin-like growth factor-I transcript with hepatic tissue expression that diverts away from the mitogenic IBE1 peptide. *Endocrinology*, **136**, 1939-1944 (1995).
- 115. M. Jansen, F.M. van Schaik, A.T. Ricker, B. Bullock, D.E. Woods, K.H. Gabbay, A.L. Nussbaum, J.S. Sussenbach and J.L. Van den Brande, Sequence of cDNA encoding human insulin-like growth factor I Precursor. *Nature*, **306**, 609-611 (1983).

- L.B. Rall, J. Scott and G.I. Bell, Human insulin-like growth factor I and II messenger RNA: isolation of complementary DNA and analysis of expression. *Meth Enzymol*, 146, 239-248 (1987).
- 117. P.H. Steenbergh, A.M. Koonen-Reemst, C.B. Cleutjens and J.S. Sussenbach, Complete nucleotide sequence of the high molecular weight human IGF-I mRNA. *Biochem Biophys Res Commun*, **175**, 507-514 (1997).
- 118. P. Rotwein, Two insulin-like growth factor I messenger RNAs are expressed in human liver. *Proc Natl Acad Sci USA*, **83**, 77-81 (1986).
- 119. P. Rotwein, Molecular Biology of IGF-I and IGF-II. In: *The IGF system: Molecular Biology, Physiology and clinical applications.* R.G. Rosenfeld and C.T. Roberts Jr. Eds, Humana Press, Totowa NJ, (1999).
- 120. K.A. Woods, C. Camacho-Hubner, R.N. Bergma, D. Barter, A.J. Clark and M.O. Savage, Effects of insulin-like growth factor I (IGF-I) therapy on body composition and insulin resistance in IGF-I gene deletion. *J Clin Endocrinol Metab*, **85**, 1407-1411 (2000).
- 121. S.K. Rasmussen, C. Lautier, L. Hansen, S.M. Echwald, T. Hansen, C.T. Ekstrom, S.A. Urhammer, K. Borch-Johnsen, F. Grigorescu, R.J. Smith and O. Pedersen, Studies of the variability of the genes encoding the insulin like growth factor I receptor and its ligand in relation to type 2 diabetes mellitus. *J Clin Endocrinol Metab*, **85**, 1606-1610 (2000).
- 122. A. Arends, H.J.J. Verhaar, E.H.F. De Haan, W.R. De Vries, M.M. Samson, M.L. Drent, E.A. Van Der Veen and H.P.F. Koppeschaar, Polymorphism in the IGF-1 gene: clinical relevance for short children born small for gestational age (SGA). *J Clin Endocrinol Metab*, 87, 2720-2724 (2002).
- 123. G. Bauman, M.A. Shaw and T.J. Merimee, Low levels of high-affinity growth hormone-binding protein in African pygmies. *N Engl J Med*, **320**, 1705-1709 (1989).
- 124. T.J. Merimee, G. Baumann and W. Daughaday, Growth hormone-binding protein: II. Studies in pygmies and normal statured subjects. *J Clin Endocrinol Metab*, **71**, 1183-1188 (1990).
- 125. Y. Hattori, J.C. Vera, C.I. Rivas, N. Bersch, R.C. Bailey, M.E. Geffner and D.W. Golde, Decreased insulin-like growth factor I receptor expression and function in immortalized African Pygmy T cells. J Clin Endocrinol Metab, 81, 2257-2263 (1996).
- 126. A.B. Cortez, C. Van Dop, R.C. Bailey, N. Bersch, M. Scott, D.W. Golde and M.E. Geffner, IGF-I resistance in virus transformed β-lymphocytes from African Efe pygmies. *Biochem Mol Med*, 58, 31-36 (1996).
- 127. N. Davila, B.T. Shea, K. Omoto, M. Mercado, S. Misawa and G. Baumann, Growth hormone binding protein, insulin-like growth factor-I and short stature in two pygmy populations from the Philippines. *J Pediatr Endocrinol Metab*, **15**, 269-276 (2002).
- 128. J.R. Bierich, H. Moeller, M.B. Ranke and R.G. Rosenfeld, Pseudopituitary dwarfism due to resistance to somatomedin: A new syndrome. *Eur J Pediat*, **142**, 186-188 (1984).
- E. Heath-Monnig, H.J. Wohltmann, B. Mills-Dunlap W.H. Daughaday, Measurement of insulin-like growth factor I (IGF-I) responsiveness of fibroblasts of children with short stature: identification of a patient with IGF-I resistance. *J Clin Endocrinol Metab*, 64, 501-507 (1987).
- T. Momoi, C. Yamanaka, M. Kobayashi, T. Haruta, H. Sasaki, T. Yorifuji, M. Kaji and H. Mikawa, Short stature with normal growth hormone and elevated IGF-I. *Eur J Pediatr*, 151, 321-325 (1992).
- 131. S.E. Tollefsen, E. Heath-Monnig, M.A. Cascieri, M.L. Bayne and W.H. Daughaday, Endogenous insulin-like growth factor (IGF) binding proteins cause IGF-1 resistance in

cultured fibroblasts from a patient with short stature. J Clin Invest, 87, 1241-1250 (1991).

- 132. A.M. Abbott, R. Bueno, M.T. Pedrini, J.M. Murray and R.J. Smith, Insulin-like growth factor I receptor gene structure. *J Biol Chem*, **267**, 10759-10763 (1992).
- U. Francke, T.L. Yang.Feng, J.E. Brissenden and A. Ullrich, Chromosomal mapping of genes involved in growth control. *Cold Spring Harbor Symp Quant Biol*, **51**, 855-866 (1986).
- 134. M.J. Abuzzahab, A. Schneider, A. Goddard, F. Grigorescu, C. Lautier, E. Keller, W. Kiess, J. Klammt, J. Kratzsch, D. Osgood, R. Pfaffle, K. Raile, B. Seidel, R.J. Smith and S.D. Cherfnausek, IGF-I receptor mutations resulting in intrauterine and postnatal growth retardation. *N Eng J Med*, **349**, 2211-2222 (2003).
- 135. A.J. D'Ercole, P. Ye, A.S. Calikoglu and G. Guitierez-Ospina, The role of the insulinlike growth factors in the central nervous system. *Mol Neurobiol*, **13**, 227-255 (1996).
- 136. R.C. Baxter, Circulating levels and molecular distribution of the acid-labile a-subunit of the high molecular weight insulin-like growth factor-binding protein complex. *J Clin Endocrinol Metab*, **70**, 1347-1353 (1990).
- 137. R.C. Baxter, J.L. Martin V.A. Beniac, High molecular weight insulin-like growth factor binding protein complex: purification and properties of the acid-labile subunit from human serum. *J Biol Chem*, **264**, 11843-11848 (1989).
- 138. G.T. Ooi, F.J. Cohen, L.Y. Tseng, M.M. Rechler and Y.R. Boisclair, Growth hormone stimulates transcription of the gene encoding the acid-labile subunit (ALS) of the circulating insulin-like growth factor-binding protein complex and ALS promoter activity in rat liver. *Mol Endocrinol*, **11**, 997-1007 (1997).
- H-P. Guler, J. Zapf, C. Schmid and E.R. Froesch, Insulin-like growth factors I and II in healthy man: estimations of half lives and production rates. *Acta Endocrinol (Copenh)*, 121, 753-758 (1989).
- 140. S.R. Leong, R.C. Baxter, T. Camerato, J. Dai and W.I. Wood, Structure and functional expression of the acid-labile subunit of the insulin-like growth factor-binding protein complex. *Mol Endocrinol*, **6**, 870-876 (1992).
- 141. I. Ueki, G.T. Ooi, M.L. Tremblay, K. Hurst, L.A. Bach and Y.R. Boisclair, Inactivation of the acid labile subunit gene in mice results in mild retardation of postnatal growth despite profound disruptions in the circulating insulin-like growth factor system. *Proc Nat Acad Sci USA*, **97**, 6868-6873 (2000).
- 142. H.M. Domené, S.V. Bengolea, A.S. Martínez, M.G. Ropelato, P. Pennisi, P. Scaglia, J.J. Heinrich and H.G. Jasper, Deficiency of the circulation insulin-like growth factor system associated with inactivation of the acid-labile subunit gene. N Eng J Med, 350, 570-577 (2004).
- 143. A.A. Butler and D. LeRoith, Minireview: tissue-specific versus generalized gene targeting of the igf1 and igf1r genes and their roles in insulin-like growth factor physiology. *Endocrinology*, **142**, 1685-1688 (2001).
- 144. E. Rao, B. Weiss, M. Fukami, A. Rump, B. Niesler, A. Mertz, K. Muroya, G. Binder, S. Kirsch, M. Winkelmann, G. Nordsiek, U. Heinrich, M.H. Breuning, M.B. Ranke, A. Rosenthal, T. Ogata and G.A. Rappold, Pseudoautosomal deletions encompassing a novel homeobox gene cause growth failure in idiopathic short stature and Turner syndrome. *Nat Genet*, 16, 54-63 (1997).
- 145. J.W. Ellison, Z. Wardak, M.F. Young, P.G. Robey, M. Laig-Webster and W. Chiong, PHOG, a candidate gene for involvement in the short stature of Turner syndrome. *Hum Mol Genet*, 6, 1341-1347 (1997).

- 146. V. Belin, C. Cusin, G. Viot, D. Girlich, A. Toutain, A. Moncla, M. Vekemans, M. Le Merrer, A. Munnich and V. Cormier-Daire, SHOX mutations in dyschondrosteosis (Léri-Weill syndrome). *Nat Genet*, **19**, 67-69 (1998).
- 147. D.J. Shears, H.J. Vassal, F.R. Goodman, R.W. Palmer, W. Reardon, A. Superti-Furga, P.J. Scambler and R.M. Winter, Mutation and deletion of the pseudoautosomal gene SHOX cause Leri-Weill dyschondrosteosis. *Nat Genet*, **19**, 70-73 (1998).
- 148. T. Kosho, K. Muroya, T. Nagai, M. Fujimoto, S. Yokoya, H. Sakamoto, T. Hirano, H. Teraskai, H. Ohashi, G. Nishimura, S. Sato, N. Matsuo and T. Ogata, Skeletal features and growth patterns in 14 patients with haploinsufficiency of SHOX: implications for the development of Turner syndrome. *J Clin Endocrinol Metab*, 84, 4613-4621 (1999).
- 149. M. Adachi, K. Tachibana, Y. Asakura, K. Muroya, T. Ogata, Del (X)(p21.1) in a mother and two daughters: genotype-phenotype correlation of Turner features. *Hum Genet*, **106**, 306-10 (2000).
- 150. M. Clement-Jones, S. Schiller, E. Rao, R.J. Blaschke, A. Zuniga, R. Zeller, S.C. Robson, G. Binder, I. Glass, T. Strachan, S. Lindsay and G.A. Rappold, The short stature homeobox gene SHOX is involved in skeletal abnormalities in Turner syndrome. *Hum Mol Genet*, 9, 695-702 (2000).
- 151. A. Marchini, T. Marttila, A. Winter, S. Caldeira, I. Malanchi, R. Blaschke, B. Hacker, E. Rao, M. Karperien, J.M. Wit, W. Richter, M. Tommasinio and G. Rappold, The short stature homeodomain protein SHOX induces cellular growth arrest and apoptosis and is expressed in human growth plate chondrocytes. J Biol Chem, 279, 37103-37114 (2004).
- 152. F. Laumonnier, N. Ronce, B.C.J. Hamel, P. Thomas, J. Lespinasse, M. Raynaud, C. Paringaux, H. van Bokhoven, V. Kalscheuer, J.P. Fryns, J. Chelly, C. Moraine and S. Briault, Transcription factor SOX3 is involved in X-linked mental retardation with growth hormone deficiency. *Am J Hum Genet*, **71**, 1450-1455 (2002).
- 153. N.M. Solomon, S. Nouri, G.L. Warne, M. Lagerstrom-Fermer, S.M. Forrest and P.Q. Thomas, Increased gene dosage at Xq26-q27 is associated with X-linked hypopituitarism. *Genomics*, **79**, 553-559 (2002).
- 154. J.E. Spence, R.G. Perciaccante, G.M. Greig, H.F. Willard, D.H. Ledbetter, J.F. Hejtmancik, M.S. Pollack, W.E. O'Brien and A.L. Beaudet, Uniparental disomy as a mechanism for human genetic disease. *Am J Hum Genet*, **42**, 217-226 (1988).
- 155. D. Kotzot, S. Schmitt, F. Bernasconi, W.P. Robinson, I.W. Lurie, H. Ilyina, K. Mehes, B.C.J. Hamel, B.J. Otten, M. Hergersberg, E. Werder, E. Schoenle and A. Schinzel, Uniparental disomy 7 in Silver-Russell syndrome and primordial growth retardation. *Hum Mol Genet*, 4, 583-587 (1995).
- 156. C.A. Joyce, A. Sharp, J.M. Walker, H. Bullman and I.K. Temple, Duplication of 7p12.1-p13, including GRB10 and IGFBP1, in a mother and daughter with features of Silver-Russell syndrome. *Hum Genet*, **105**, 273-280 (1999).
- 157. D. Monk, E.L. Wakeling, V. Proud, M. Hitchins, S.N. Abu-Amero, P. Stanier, M.A. Preece and G.E. Moore, Duplication of 7p11.2-p13, including GRB10, in Silver-Russell syndrome. *Am J Hum Genet*, **66**, 36-46 (2000).
- 158. N. Miyoshi, Y. Kuroiwa, T. Kohda, H. Shitara, H. Yonekawa, T. Kawabe, H. Hasegawa, S.C. Barton, M.A. Surani, T. Kaneko-Ishino and F. Ishino, Identification of the Meg1/Grb10 imprinted gene on mouse proximal chromosome 11, a candidate for the Silver-Russell syndrome gene. *Proc Nat Acad Sci USA*, **95**, 1102-1107 (1998).
- 159. W. He, D.W. Rose, J.M. Olefsky and T.A. Gustafson, Grb10 interacts differentially with the insulin receptor, insulin-like growth factor I receptor, and epidermal growth factor receptor via the Grb10 Src homology 2 (SH2) domain and a second novel domain located between the pleckstrin homology and SH2 domains. *J Biol Chem*, 273, 6860-6867 (1998).

- 160. H. Yoshihashi, K. Maeyama, R. Kosaki, T. Ogata, M. Tsukahara, Y. Goto, J. Hata, N. Matsuo, R.J. Smith and K. Kosaki, Imprinting of human GRB10 and its mutations in two patients with Russell-Silver syndrome. *Am J Hum Genet*, 67, 476-482 (2000).
- 161. K. Hannula, M. Lipsanen-Nyman, T. Kontiokari and J. Kere, A narrow segment of maternal uniparental disomy of chromosome 7q31-qter in Silver-Russell syndrome delimits a candidate gene region. *Am J Hum Genet*, 68, 247-253 (2001).
- 162. M.P. Hitchins, D. Monk, G.M. Bell, Z. Ali, M.A. Preece, P. Stanier and G.E. Moore, Maternal repression of the human GRB10 gene in the developing central nervous system; evaluation of the role for GRB10 in Silver-Russell syndrome. *Eur J Hum Genet*, 9, 82-90 (2001).
- 163. J.A. McCann, H. Zheng, A. Islam, C.G. Goodyer and C. Polychronakos, Evidence against GRB10 as the gene responsible for Silver-Russell syndrome. *Biochem Biophys Res Commun*, 286, 943-948 (2001).

# Chapter 15

# **FUTURE PERSPECTIVES:**

From stem cells and IGF biology to the clinic

#### Yvan Arsenijevic

Unit of Oculogenetics. Eye Hospital Jules Gonin, 15 av. de France, 1004 Lausanne, Switzerland

Key words: Cancer; cell differentiation; growth; regeneration; stem cell renewal; transplantation.

## **1. INTRODUCTION**

From the previous chapters, it appears that insulin-like growth factors (IGFs) control the growth and the development of different organs showing the role of IGFs as master regulators during organogenesis. These chapters review the actual knowledge of IGF functions and discuss their roles during the building of our body and its maintenance, as well as in different pathologies. Here, I would like to show how the study of stem cells can help to dissect some IGF functions and how the knowledge of stem cells and their regulators may help in making clinical applications possible in the long term. IGF-I has pleiotropic functions during embryogenesis and adulthood; therefore, it is necessary to develop systems that allow to dissect the function of IGF-I, cell by cell phenotype, and identify the mechanism of IGF-I in each of these cell systems to understand how organogenesis and body growth take place. In this chapter I present various concepts rather than provide a review of the literature. Nonetheless, representative data and reviews will help to lead the reader through these concepts. The role of IGF-I in different stem cell systems is discussed in light of the concepts emerging from the study of stem cells. Indeed, when someone has to work with stem cells, he or she has to understand what a stem cell is and what the other cells are. The protocols that allow the study of stem cells also bring new tools for understanding the function of genetic and epigenetic factors, such as IGF-I. But first, what is a stem cell?

## 2. STEM CELL DEFINITIONS

All stem cells are derived from more primitive ones, the embryonic stem cells (ES cells) that are located in the inner cell mass of the blastocystes. The ES cells are pluripotent: they give rise to all cell phenotypes of the body<sup>1</sup>. It is believed that ES cells generate more specialized stem cells (organ-specific fetal stem cells) that will be the origin of each organ that composes the body<sup>2,3</sup>. During development, the fetal stem cells give rise to all cells that will form an organ: each organ has its specific stem cell that will initiate the formation of said organ. Such a phenomenon implies that the stem cells have to proliferate extensively. This is the most important characteristic of a stem cell: its enormous capacity to proliferate. The control of stem cell proliferation and renewal is discussed in detail later. Depending on the organ, a specific stem cell type can generate all the cell types of the organ. For instance, the neural stem cells generate successively all neural cells during brain formation, *i.e.*, neurons, astrocytes, and finally, oligodendrocytes. Thus, each neural stem cell is multipotent and can generate all the mentioned cell phenotypes<sup>4</sup>. Multipotentiality is the second characteristic. Nonetheless, in certain organs, such as the skin, you can find stem cells that produce only one cell type, for instance the melanocytes<sup>5</sup>. Organ-specific stem cells persist after birth and during adulthood, where they participate to renew a tissue (skin, bowels), to add new cells (neurons in the olfactory bulb) or to repair a damaged organ. When a bone is broken, or the skin cut, the local stem cells generate cells that will allow the bone to mend and the wounded skin to heal<sup>6</sup>. Tissue renewal and repair is the third characteristic of stem cells.

In summary, a stem cell is able to renew and generate the quantity of cells necessary to build the organ from where the stem cell is issued, to produce the specific cells of the organ, and participate to the repair of the wounded organ (Fig. 1).



Figure 1. Stem cells renew and produce all the cell phenotypes of an organ and participate in its repair. (A) In situ, stem cells (black circles) generate transient amplifying progenitors (white circles) that produce precursors (gray circles). Then, the precursors give rise to differentiated cells (white ellipse and gray rectangle). During adulthood, such phenomenon can occur throughout life (skin, small intestine epithelium) or the stem cell can be quiescent or have a slow turnover (brain). After injury, in many tissues the stem cells are activated and generate cells necessary for healing. (B) In vitro, a stem cell should be capable of generating progenitors and precursors, as well as itself (backwards arrow) showing by this property its renewal capacity. Differentiation induction of the precursors leads to the formation of the differentiated cells that compose the organ from which the stem cell is derived.

#### **3.** STEM CELL IDENTIFICATION

To work with stem cells, we would like to be able to precisely identify the stem cells in order to follow their proliferation and the generation of cells committed to a specific phenotype. However, only a few stem cells can be recognized by one or several markers, as is the case for the hematopoietic stem cells<sup>7,8</sup> or the embryonic stem cells<sup>1</sup>. In general, only clonal analysis may provide the means to identify that the cell growing in your dish is a stem cell. Clonal analysis means that a single cell is studied individually, without interference from the other cells. By understanding how one can characterize a stem cell, we will see that the same protocol

may be applied to study the function of genes or environmental factors in stem cells (growth factors, metabolites, oxygen, etc.).

The analysis of stem cell characteristics occurs in several steps. First, a piece of tissue is dissected from an organ in an area of interest, for instance, the germinative zone of the developing striatum in a newborn brain. Obviously, the dissection allows one to select the area of interest to be investigated, but this approach does not result in a pure cell population. In this blending of cell types, differentiated cells (such as neurons and glial cells), precursors,\* progenitors\*9\*, and stem cells are mixed together. The culture condition may help to enrich a specific population, but cellular heterogeneity remains. The first goal is to determine if a population of highly proliferative cells exists. The tissue is dissociated and cells are plated at clonal density in the presence of mitogens to determine which cells are proliferating. Only a small percentage of cells give rise to large colonies. These colonies can then be isolated and propagated to test their potential of proliferation. In fact, we expect that a stem cell can produce the number of the cells necessary to compose an organ<sup>10</sup>. However, when the stem cell divides, it will not necessarily give rise to another stem cell, but can also produce other cell types. In the case where the generation of two daughter stem cells occurs, we say that the stem cell has undergone a symmetric division, and an asymmetric division when one daughter stem cell and another cell type are generated (Fig. 2).



Figure 2. Stem cells, symmetric and asymmetric cell division. (A) One stem cell gives rise to two stem cells after a cell division: symmetric division (both cells are identical). In this case, we observe an expansion of the stem cell number. (B) One stem cell generates one stem cell and one progenitor or differentiated cell: asymmetric division (both cells are different). It is important to emphasize that the stem cell is always present, it renews itself. (C) One stem cell produces two non-stem cells. The stem cell is lost.

Progenitors are cells that divide for a long period of time, but with a limited renewal capacity; precursor cells undergo only few cell cycles and are restricted in their fate.

Thus, we can deduce the existence of a stem cell only retrospectively. Moreover, when the stem cell has started to form a colony, this colony is composed of a heterogeneous population of cells, maybe less heterogeneous than *in vivo*, but we have to assume that the stem cell has generated another cell phenotype unlike itself (Fig. 3). In consequence, to test the action of a gene or of a factor on stem cells, we need to target our study to this first cell that is at the origin of the colony. Nonetheless, culture conditions may help to enrich the colony in stem cells. For instance at birth, neural stem cells represent approximately 1% of the total cell population in the germinative area (only a small percentage of these cells can form large colonies that can be propagated<sup>11-13</sup>), but a 17-fold increase of the stem cell percentage can be achieved by cell culture in defined conditions (Zencak et al., submitted). How can we determine if a stem cell generates other stem cells? By subcloning the original colony it is possible to determine how many new clones are able to generate new large colonies with a great capacity of proliferation and differentiation: these new colonies are also derived from new stem cells (Fig. 3).

Parallel to the assessment of cell renewal potential, it is necessary to reveal whether the proliferating cell is able to generate the cells composing the organ. After cell expansion, to reveal the multipotentiality of the supposed stem cell, the derived progeny are induced to differentiate. The differentiation process requires conditions allowing the formation of all the cells composing the organ. For instance, for neural stem cells standard culture conditions allows the induction of the differentiation of neurons, astrocytes, and oligodendrocytes<sup>4,14-19</sup>. Depending on the stem cells and the phenotype to be generated, a unique culture condition is not sufficient to reveal the multipotentiality of the stem cell. In consequence, it is necessary to provide enough cells from a clone to test the same clone in different conditions to reveal the cell phenotypes.

Nonetheless after this *in vitro* characterization, the first thing to assess is to show that what we are studying *in vitro* is related to what happens *in vivo* (see Section 4.3).



Figure 3. Clonal analysis of stem cell action. (A) Under the stimulation of a factor X, a single stem cell gives rise to a large clone (clone generation I) containing progenitors and stem cells (see (C) for symbols). To know if this clone contains stem cells, it is necessary to isolate single cells from the clone and to stimulate its proliferation: a stem cell (black circle) will generate a new clone (clone generation II), whereas a precursor (white circle) will give rise only to a limited number of cells. Clones from generation II can be subsequently subcloned to test for the presence of a stem cell. Such procedure can be renewed several times (clone generation III and IV, and subsequent if necessary). A single clone can also be expanded to determine the expansion potency of the cell at the origin of the clone (bottom clone + graph). Each clone has to be tested to reveal its differentiation potential (arrow with gray symbols and white ellipse). (B) The actions of the factors X and Y are tested on two stem cells (supposed identical). Factor X favors the expansion of the stem cells (black circles), whereas factor Y stimulates the proliferation of the stem cells leading to the generation of progenitors, but without stem cells. Both clones are multipotent, but only the clones generated by factor X can renew. In conclusion, factor X stimulates the renewal of the tested stem cell, whereas factor Y does not. Using the progeny derived from the stem cell, different factors can also be tested to evaluate their capacity to induce cell differentiation. (C) Symbol nomenclature.

# 4. ADVANTAGES OF USING STEM CELLS

# 4.1 An unlimited source

One of the main advantages in using stem cells, is the large amount of cells that can be used for one study. For instance, the EGM and the EGL (the two germinative zones of the developing striatum in the brain) of 10 embryos at E14 give rise to around 20 million cells. These cells can be used directly (without induction of growth) to study neuronal differentiation for instance. If neural stem cells are isolated at this time, 3 weeks later, equivalent to the period of gestation, the same amount of cells can produce 2.5 billion striatal cells, corresponding to 1200 embryos. The advantage of such system is obvious. Moreover, these cells can be passed during months and frozen anytime to be used at a more opportune time. Nonetheless, it is necessary to control or not whether expanded cells maintain a normal karyotype and remain dependent on the presence of exogenous mitogens to stimulate their proliferation. In parallel, the main problem is to characterize the system and show that what you are looking at *in vitro* with these cells is relevant to what happens during in vivo development (see later). Nonetheless, the capacity for renovation is huge for skin stem cells, neural stem cells, and mesenchymal stem cells, for instance. Such cell culture systems may help make it possible to perform large scale screening of the function of hundreds or thousands of molecules or to screen gene expression in a given situation as described earlier. Other advantages reside in the possibility to generate a large cell number for transplantation studies (see Section 9).

# 4.2 Controls of cell state: drug and gene function screening

What is the advantage of using stem cells in comparison to primary cells from fetal tissue, for instance? By manipulating the stem cells *in vitro*, different cell populations can be separately studied, from the stem cell to the differentiated cell. Each step of cell development can be investigated using defined culture conditions that can help to synchronize cells in a specific cell stage. In consequence, the role and the mechanism of action of intrinsic and extrinsic factors in a population of cells can be more easily determined.

To study the function of a gene or the action of a factor specifically on stem cells, two approaches have to be considered. Either the possibility exists to isolate certain stem cells such as hematopoietic stem cells<sup>20</sup> or muscle stem cells<sup>21,22</sup> by specific markers, or it is first necessary to develop

a method to study the characteristics of the stem cell. In the first case, the analysis is facilitated by the fact that the cell population is homogeneous. In the second case, the use of clonal analysis, as described in Section 3 and in Fig. 3, could be the solution. In both cases, to reveal whether a factor enhances stem cell division, for instance, stem cell subcloning protocols allow the investigator to analyze whether the tested factor produces more new colonies when added to the incubation media during the proliferation of the first clone (Fig. 3B). To prove that this factor acts directly on stem cells, it is necessary to work at clonal density to prevent the influence of the neighboring cells. To test an inhibitor, a similar approach can be undertaken. In this case, we expect that no new colony will be formed.

An example of the utility of this approach is that several research groups think that glial scar formation during spinal cord injury impairs the regrowth of cortico-spinal fibers that control motricity. Certain researchers believed that the scar formation is derived from activated stem cells in the spinal cord stimulated by the injury<sup>23</sup>. To show that stem cells and the glial scar have a deleterious action, one of the goals is to prevent neural stem cell proliferation. The protocol of clonal analysis described earlier may help to identify an inhibitor of spinal cord stem cell proliferation. The screening of molecules preventing such proliferation appears to be of prime importance and the methods described here have already been used to study the actions of factors that stimulate stem cell proliferation<sup>11,13,24-30</sup> and may serve to identify therapeutic agents.

Other domains also take advantage of stem cell investigation. Studies of hematopoiesis and muscle cell differentiation processes are very well documented, because the culture conditions of these cell systems allow the precise identification of the different stages of development of each cell investigated. This approach leads to the determination of the function and the mechanism of action of different factors or genes on stem cells during hematopoiesis<sup>31</sup> and myogenesis<sup>32</sup>. Nonetheless, important progress has also been recently obtained in the study of stem cell based neurogenesis<sup>33-35</sup>, bone formation<sup>36</sup>, skin epithelium formation<sup>6</sup>, and so forth. The role of IGFs in these systems is discussed in the sections that follow.

The study of the induction of differentiation may also be approached by similar protocols. Indeed, several stem cell culture conditions produced an enriched population of nondifferentiated cells, allowing the study of factors generating differentiated cells, *i.e.*, the passage of a nondifferentiated into a specific cell phenotype. The appearance of differentiated cells in the culture could be due to different actions of factors in the medium, one committing the cells to the neuronal fate<sup>16</sup>, the other inducing the differentiation<sup>14,15</sup>, and finally the third one acting as a survival factor. Using such approach with stem cells, it was shown that IGFs control the differentiation of a multitude

of different cell types. From this book, as well as from different articles and reviews, it is evident that the IGF family controls the cell differentiation process in bone<sup>36,37</sup>; muscle<sup>38</sup>, blood<sup>39,40</sup>, brain<sup>15,27,28,41</sup>, heart<sup>42</sup>, gut<sup>43</sup>, and so forth. It appears that IGF-I has a pleiotropic function during the organogenesis of many tissues. Other factors, such as fibroblast growth factor (FGF)-2, have similar pleiotropic functions during cell proliferation and commitment. *In vitro* cell differentiation studies based on stem cell cultures system will surely help to dissect the function and the respective mechanism of each factor during development. The understanding of how one molecule can control different cell functions just after a cell division or cell commitment will surely help to understand how we can manipulate these systems in pathogenic conditions or for tissue regeneration.

# 4.3 *In vitro* stem cells reinitialize different steps of development

In vitro stem cell studies have an interest in developmental biology only if the stem cell behavior *in vitro* reflects a function that can or does happen during embryogenesis. Fortunately, several studies show similarities between cell fate induction *in vitro* and what occurs naturally *in vivo*. Some examples are described below for different stem cell systems, but we need to always keep in mind that we never know to which extent we "push" the system *in vitro*. Indeed, the concentration of several factors added in the growth medium are often supraphysiologic [*e.g.*, epidermal growth factor (EGF), insulin]. In this context, it is important to try to work with physiological concentrations of the growth factor to better understand the role of each factor avoiding the unknown nonspecific activation that high factor concentration can produce. The replacement of high concentrations of insulin by physiological levels of IGF-I and/or insulin and/or IGF-II should allow us to better understand the role of each of these factors and to identify whether their actions are sufficient for replacing insulin action.

Several studies with neural stem cells have shown that different steps of in vitro neural stem cell based neurogenesis parallel the steps observed during embryogenesis. For instance, in vitro neural stem cells first acquire the capacity to respond to FGF-2 stimulation, and then to EGF<sup>3,13,24</sup>. These experiments were confirmed by similar observations in studies of neural stem cells derived from EGF and FGF-2 knockout mice<sup>13</sup>. Zhu et al.<sup>19</sup> have shown that neural stem cells, isolated at different times of development, generate preferentially the cell phenotype produced during the time when these stem cells were isolated: indeed, stem cells isolated early produced isolated birth produced more neurons, and NSC after more

oligodendrocytes, as observed *in vivo*. The most convincing demonstration that NSCs maintain the capacity *in vitro* to mimic normal development was shown by transplantation studies. When NSCs are transplanted into a developing brain, cells generated from the neural stem cells can differentiate into various neuronal and glial phenotypes<sup>44-48</sup> and can integrate into a neuronal circuitry<sup>49</sup>.

Similar results were obtained using hematopoietic stem cells. The study of HSCs is facilitated by the possibility of isolating stem cells depending on the expression of specific markers expressed on the cell surface<sup>8</sup>. The most convincing demonstration of renewal, multipotentiality, and the maintenance of a normal HSC program was achieved by the transplantation of a single HSC in mortally irradiated mice. The HSC was capable of homing the bone marrow and repopulating all the blood cells<sup>50</sup>. This procedure can be repeated several times and shows the fantastic potential of the HSC cells. In fact, for researchers in all other stem cell fields, this protocol is the gold standard to demonstrate the stem cell identity and is the goal that each researcher would like to reach for his or her own specific field of research.

The ideal condition of testing stem cells would be to compare the engrafting potential of stem cells isolated directly from the body and after cell culture expansion. But this approach is often impossible owing to a lack of stem cell markers in several cell systems. Nonetheless, several stem cells can have an *in vitro* behavior that corresponds to their normal function *in vivo*: muscle stem cells can differentiate and fuse to form active contractile skeletal muscle cells<sup>32</sup>, cultured mesenchymal stem cells can participate in cartilage repair<sup>51</sup>, limbal stem cell of the cornea can be expanded *in vitro* to form a cornea for transplantation<sup>52</sup>, and epithelial skin stem cells can reconstitute a dermis for a person burned over a large area<sup>6</sup>.

# 5. STEM CELL RENEWAL AND IGFS

Stem cell renewal is due to different factors and genes that control several functions of the stem cell. Indeed, to survive, for long term renewal, a stem cell needs to remain nondifferentiated, proliferate, and to protect against the loss of the chromosome extremities (a degradation process occurring at each cell division<sup>53,54</sup>) by maintaining the presence of telomeres. Moreover, a stem cell has to be quiescent when necessary. It appears that self-renewal implicates a multitude of controls in a stem cell to proliferate adequately. Here, several examples of the role of IGFs on different stem cells systems are described to show the importance of IGFs in the control of stem cell renewal.

The first studies implicating IGFs in sustaining long term cell proliferation were undertaken on fibroblasts and placenta cells. In absence of serum, IGF-I has little effect on fibroblast division. However, in the presence of human hypopituitary serum, IGF-I strongly stimulates fibroblast proliferation. This effect is reinforced by dexamethasone<sup>55</sup>. Interestingly, GH can directly stimulate fibroblast division via the induction of IGF-I secretion that acts in an autocrine/paracrine mode<sup>56</sup>. This study supports the idea that GH can directly target certain cells without acting through the liver<sup>37</sup>.

Recent studies have shown that the IGF family, including the IGFBPs, stimulate the proliferation of primitive hematopoietic progenitor cells and stem cells. It is important to keep in mind that hematopoietic stem cell renewal was demonstrated by in vivo transplantation, because no factors allowed their expansion in vitro. In this context, any induction of cell division in hematopoietic stem cells is an important advance for understanding the cell renewal mechanism in these cells. Interestingly, during development CD3 liver cells secrete IGF-II when hematopoietic stem cells reside in this organ<sup>57</sup>. Moreover, coculture of CD3 cells with hematopoietic stem cells supports the expansion of the stem cells. The use of antibodies to block IGF-II prevents the supporting effect of liver cells showing that IGF-II may participate in the control of hematopoietic stem cell proliferation. A supporting effect of cells surrounding the primitive hematopoietic CD34<sup>+</sup>CD38<sup>-</sup> cells was also identified in human endothelial cells<sup>58</sup>. Interestingly, in this paradigm IGFBP-3 was shown to stimulate the proliferation of CD34<sup>+</sup>CD38<sup>-</sup> cells.

A role of the IGF family in the hematopoiesis control of early progenitors was also identified in myeloid progenitor cells in which IGF-I stimulated DNA synthesis and reduced apoptosis<sup>59</sup>. Enhanced proliferation of myeloid cell lines by IGF-I was also documented<sup>60</sup>. Interestingly, GH itself can have a direct action on B cells of normal subjects or on leukemic cell lines<sup>61</sup>. Moreover, GH is synthesized by these cells and can act via an autocrine mode. Taken together, these data show that different members of the growth hormone-insulin-like growth factor axis control early stages of hematopoiesis. IGF-I also inhibits apoptosis of ervthroblasts and megakarvoblasts, and has a slight effect on their proliferation<sup>39</sup>. These data and those presented in various chapters of this book show that IGF related proteins act on proliferation and apoptosis of different blood cells. Like in other systems, several factors control the renewal of the hematopoietic stem cells<sup>62,63</sup>. In this context, the IGF family may play an important role shared with these factors.

Using muscle stem cells, Deasy *et al.*<sup>22</sup> have shown that IGF-I, in the presence of serum and chick embryo extract, stimulates proliferation. IGF-I

was also shown to stimulate by two-fold the proliferation of porcine embryonic myogenic cells<sup>64</sup>. These cells also synthesize IGFBP-3, suggesting a regulatory role of this protein. Indeed, overexpression of IGFBP-3 by baculovirus prevents IGF-I induced proliferation. These examples, and those presented in Chapter 10, reveal that the production of muscle cells, that compose a great portion of our body, is also regulated by the IGF family.

IGF-I also plays an important role in the control of neural stem cell proliferation. Several factors have been shown to stimulate or control neural stem cell division, such as EGF<sup>4</sup>, FGF-2<sup>13,18,65</sup>, cystatin-C<sup>26</sup>, sonic hedgehog<sup>66,67</sup>, ciliary neurotrophic factor (CNTF)<sup>25</sup>, leukemia inhibitory factor (LIF)<sup>68</sup>, and probably delta<sup>69</sup>. The exact link between these factors is still unclear. Nonetheless, the *in vitro* study of neurogenesis based on stem cells allows us to start to identify the function of IGF-I in neural stem cells. It was shown that the mitogens EGF and FGF-2 cannot induce neural stem cell proliferation without the presence of IGF-I<sup>11</sup>. Moreover, this study revealed that FGF-2 acts as a survival factor, whereas IGF-I or EGF did not have this action alone, but only in combination.

It appears that each factor has a specific role in the control of neural stem cell proliferation. A collaboration between these factors was also described for stem cells derived from the olfactory bulb<sup>28</sup>. Nonetheless, the fact that mice knocked out for the *IgfI* or the *IgfIR* genes have a smaller brain, but still have a brain, shows that IGF-I is not necessary to maintain the population of neural stem cells, but may act on the neural stem cell number. It would be interesting to quantify the number of neural stem cells in these knockout versus wild type mice, in order to define whether endogenous IGF-I controls the neural stem cell number during development. Interestingly, in the absence of IGF-I (*IgfI* knockout mice) neural stem cells of the olfactory bulb maintain the capacity to proliferate normally in the presence of EGF, FGF-2, and high concentrations of insulin<sup>28</sup>. In consequence, other factors may have stimulated neural stem cell proliferation during the development of the IgfI knocked out brain such as those listed above, or pro-insulin which was detected in olfactory bulb progenitor cells<sup>28</sup>. Moreover, pro-insulin was shown to stimulate the proliferation of these progenitors<sup>28</sup> suggesting that it may play a role in the control of the neural stem cell number. On the other hand, it was also shown that injections of high doses of insulin  $(2 \mu g)$  into the newborn chick eye stimulate the proliferation of retinal progenitor cells<sup>70</sup>. In this case, it remains to determine whether this dose of insulin acts via the insulin or the IGF-I receptors.

Mammary gland cell proliferation is also controlled by IGFs and the IGFBPs<sup>71,72</sup>. Indeed, transgenic mice expressing IGF-II have a decreased

number of apoptotic cells in the mammary epithelial cells<sup>73</sup>. Moreover, the involution of the mammary gland after lactation is inhibited by the overexpression of IGF-I in transgenic mice<sup>74</sup>. IGF-I is expressed in the terminal end buds, which builds the gland during puberty<sup>75</sup> and allows the maintenance of the proliferation of the mammary epithelial cells *in vitro*<sup>76</sup>. These data show that IGF-I is a major actor in mammary gland morphogenesis.

IGF-I can also act on the most primitive cell of the body, the pluripotent zygote. *In vitro* IGF-I enhances the survival of rabbit and mouse embryos<sup>77,78</sup>. Moreover, IGF-I promotes rabbit blastocyst formation<sup>77</sup>. The establishment of an embryonic stem cell line (derived from the inner cell mass of blastocysts) is facilitated when IGF-I is present. Such effect is reinforced when IGFBP-1 is added<sup>79</sup>. Embryonic stem cells usually need serum or a feeder layer of cells to grow. In a recent study, it was shown that monkey embryonic stem cell proliferation can be induced by using a serum-free medium composed of defined factors, such as IGF-I, transforming growth factor (TGF) $\alpha$ , FGF-1, estradiol, and progesterone<sup>80</sup>.

Taken together, it seems that IGF-I acts throughout all developmental stages and on numerous stem cells, if not all. Several other *in vitro* stem cell systems use high insulin concentration in the medium to stimulate cell proliferation. In fact, it remains to be determined what is the function of insulin in these cultures and whether insulin acts through the IGF-I receptors or through the insulin receptor. Indeed, it is also possible that insulin plays a direct role on stem cells as suggested by the study of proinsulin on the proliferation of olfactory bulb progenitors<sup>28</sup>. When all stem cells have been investigated to determine the potential role of IGFs, we will have an overall idea of the role of IGFs on stem cells. Do IGFs have an identical role between all stem cells or do they control stem cell renewal with specific function depending on the stem cell type that they target? This understanding will help to design strategies to specifically manipulate a stem cell population either to promote tissue repair (see Section 10) or to prevent tumor formation (see Section 8).

## 6. IGF-I, STEM CELLS, AND BODY GROWTH

Knock-out of the *IGF-I* gene or deletion of the *IGF-II* gene<sup>81,82</sup> results in a reduction in the size of the transgenic animals. Moreover, the combination of two gene deletions provokes a more severe growth deficit. In all transgenic animals, the gross morphology of the majority of the organs seems to be affected in the same proportional manner (even if certain cell types are more affected by this deficiency). Indeed, the size of the legs, the
head, including the brain and the eyes, and the chest are reduced in the same proportion. The heart and the liver also reside normally in the body. In view of the redundant actions of IGFs on different stem cells, one hypothesis of the control of body size by IGFs could be an action of IGFs on stem cells, with a subsequent possibility that IGFs also control cell proliferation of the down stream precursors and their differentiation.

The growth of several organs is organized by the generation of columns of committed and differentiated cells derived from the stem cells<sup>83</sup> (Fig. 4) or cells derived from stem cells organized in clusters. This approach is simplistic because it is known that in certain organs, such as the brain, several cells migrate extensively before reaching their target<sup>84</sup>. Nonetheless, the model proposed here allows one to draw a general hypothesis that can be further ameliorated or denied.

Growth in columns occurs in the skin<sup>83</sup>, the small intestine<sup>85</sup>, several regions of the brain (for instance the cortex, the retina), the bones<sup>36</sup>, and so forth. The difference in size between the human cortex and the mouse cortex is not related to the thickness of the tissue, but to its extension. Moreover, the overexpression of  $\beta$ -catenin in mouse cortical neural progenitor cells leads to an expansion of neural progenitors and, consequently, to a dramatically increased surface of the cortex with an approximately normal pattern of neuron location<sup>86</sup>. This study (and those cited within) show that the number of cortical columns is related to the number of progenitor cells. Also, when we compare certain organs between a short and a tall person, the skin and the cortex, for instance, one can easily determine the origin (tall vs. short) of the organ if you consider the surface of the organ, whereas the cell layers existing in these organs will be similar in cell number and cell size.

These observations show that several organs grow by adding successive columns of cells laterally, rather than by multiplying the number of cells in the thickness of the tissue. Obviously, for organs like muscles, the size of the muscle fibers will be increased in tall people and the papers discussed in this book demonstrate clearly that GH and IGF-I also control the cell size. But here, I would like to point out that growth regulation can also occur in a horizontal/lateral plan. If IGF-I controls the survival or the proliferation of stem cells, an increased concentration of IGF-I would results in a greater pool of stem cells. If a future organ contains more stem cells and the environment favors the normal development of these cells, we can expect that this germinative zone will give rise to more columns of cells, rendering this organ larger in comparison to an organ that has fewer stem cells (Fig. 4). This could be possible only if the environment provides factors allowing the normal development of the stem cells.



*Figure 4.* Stem cells and the control of organ growth size. (A) At low IGF concentrations, stem cells do not expand, but generate the progenitor cells necessary to build the organ. (B and C) At high IGF concentrations, stem cells start to divide and expand, resulting in an increased number of cells in the horizontal plan. (D and E) Each new stem cell generates new progenitors that will add new column of cells increasing the lateral growth of the organ.

This hypothesis would explain the difference in body size between individuals and animals with different levels of IGF-I ranging from zero to physiological concentrations. Supposing that IGF-I has a direct action on stem cells (we supposed that all stem cells can be influenced by IGF-I; nonetheless other experiments are needed to evaluate if it is really the case) and on several other committed cells as evidenced by the pleiotropic actions of IGFs on precursor proliferation and cell differentiation, the cells generated in columns will survive and remain to build the organ. Increased organ growth can also be stimulated through stem cells to produce an increased number of progenitors, but not via a higher number of stem cells. The results would be a larger column of cells that generate more differentiated cells. For example, the increased size of a bone is believed to result from the following mechanism. GH stimulates the local production of IGF-I from the growth plate, leading to the acceleration of stem cell and progenitor proliferation<sup>36,87</sup> and the expansion of the proliferating cells<sup>88</sup>. Then, IGFs and IGFBPs participate in the differentiation of the chondrocytes (for review see refs. 36,88). To prove such a theory, implicating stem cells and IGFs in the control of organ size, a larger organ should have more stem cells than a smaller one, with the percentage of stem cells in the studied organ remaining the same. Such experiments remain to be undertaken, as does the identification of the exact role of IGFs on different stem cells.

The importance of IGF in the generation and survival of stem cells and the derived progeny leads one to ask the question of the source of IGFs and how the concentration of IGF influences the number of stem cells. Recent studies on transgenic mice that fail to express IGF-I in the liver have shown that hepatic IGF-I is not responsible for the normal development and growth of an animal<sup>89,90</sup>. Indeed, these transgenic mice have an identical growth compared to wild-type animals. Only the spleen shows a reduced size. A new theory proposes that the growth is in part due to the production of IGF-I by an autocrine/paracrine mode of several cell types contained in different organs of the body<sup>37</sup>. Indeed, numerous studies revealed that stem cells or differentiated cells produce IGF-I and/or IGF-II in an autocrine manner to sustain their proliferation differentiation (this book or and references<sup>11,27,37,41,71,91,92</sup>. Supposing that stem cells secrete a low amount of IGF-I, this concentration will influence only the fate of said stem cell. In this case, IGF-I has an autocrine action, whereas higher amounts of IGF-I will also reach cells located in the neighborhood, provoking autocrine and paracrine actions of IGF-I. (The paracrine action of IGF-I on body growth and the respective role of growth hormone is developed in details in ref. 37).

The IGF-I released from all the stem cells in a region will have an important cumulative effect, potentiating the actions of IGF-I leading to the proliferation of stem cells. Supporting this hypothesis, the number of stem cells entering the cell cycle *in vitro* is IGF-I dose dependent and neural stem cells secrete IGF-I<sup>93</sup>. A small elevation in IGF-I concentration, between 0.3 nM and 1 nM, produces a significant increase in proliferating stem cell number. Consequently, in a local environment the stem cells will be exposed to the same amount of IGF-I throughout the said area, leading to coordinate growth of the organs (Fig. 4). The stem cells will either expand (*e.g.*, brain, skin) or generate more precursors (bone). The arrest of cell proliferation will be controlled by the maturation of different cell types and circulating hormones (for instance see ref. 36).

The dissection of such a mechanism should help to target the cells that are necessary to adequately stimulate body growth. Until this is known, the respective roles of GH, IGFs, and insulin in the control of body growth will be continuously under debate. The understanding of which cell types are the target of these factors will surely help to dissect these mechanisms and help to further our understanding of normal and pathological growth processes.

## 7. IGFS AND CANCER STEM CELLS

Stem cells have an almost unlimited capacity to renew, as cancer cells do. The difference between these two cells resides in the fact that cancer cells have lost their capacity to exit the cell cycle under the influence of normal environmental stimuli. How immortalization leads to oncogenesis is still obscure. Reya *et al.*<sup>94</sup> described that several cancer cells share various characteristics of stem cells and named the cells at the origin of cancer, cancer stem cells. They consider a cancerous tumor to be an abnormal organ that is built with cancer stem cells, but also with precursors and differentiated cells. In fact, the cancer stem cell can renew itself as does as a stem cell and generate all the cells composing an organ. It appears that during the proliferation process of the cancer stem cells, showing that the proliferation of a cancer stem cell is not blocked to an irreversible process of cell division.

By understanding how normal stem cells and cancer stem cells renew, we may understand how to interfere with cancer cell division. Certain mutations lead to oncogenesis, however how this process of oncogenesis occurs after the mutation remains unsolved in many cases. Many genes can be upregulated by just one mutation, resulting in a new homeostasis of cell cycle regulation. Like stem cells, cancer stem cell division is controlled by several intrinsic and extrinsic factors, each one playing a specific role. We have previously seen that IGFs control the proliferation of several stem cells and other precursors. It is therefore not surprising to observe that IGF-I contributes to the development of numerous tumors (see Chapter 13) such as leukemia<sup>40</sup>, meduloblastoma<sup>95</sup>, breast cancer<sup>71</sup>, hepatocellular carcinoma<sup>96</sup>, and in several other cancers reviewed in ref. 97. It would be important to determine whether IGF-I stimulates the proliferation of the cancer stem cell or whether it acts on downstream progenitors. Such knowledge will help to further understand cancer stem cell renewal and how we can attempt to target specific mechanisms of the cancer stem cells by specific agents to prevent their proliferation. Indeed, if chemotherapy acts only on cancer progenitors and not on cancer stem cells, we would succeed in decreasing the size of the tumor, but could expect a second event of tumor formation, because the cancer stem cells were not eliminated. In this context, the dissection of the IGF-I role on tumor formation is of prime importance.

## 8. CELL TRANSPLANTATION

Cell transplantation is already used with recognized success in medicine to repopulate the hematopoietic system<sup>31</sup>. This approach also seems to be promising for certain organs that have fully or partially lost their function, like the pancreas<sup>98</sup> and the brain<sup>99-101</sup>. All encouraging data harvested from experiments done on these organs, with some success in the clinic, were obtained with cells derived directly from fetal tissues. One of the major problems is the quantity of cells that can be generated. For instance, depending on the hospital center, it is necessary to have one to five fetuses to transplant mesencephalic cells in one hemisphere of one Parkinson patient<sup>102</sup>. This approach has shown some success and some limitations (discussed in ref. 103).

One of these limits is obviously the number of cells that are available to transplant thousands of patients (number necessary for a region as large as Switzerland) and another limitation is the lack of external control of cell fate after transplantation. Indeed, in certain rare cases, the graft developed into a cyst<sup>104</sup> or was too efficient and produced aberrant motor behavior such as dyskenesia<sup>103</sup>. It appears that we need a cell source that can be both greatly amplified and genetically modified to eliminate uncontrolled cells and to prevent side effects. Stem cells appear to be a source of choice if we understand how to generate the desired cell phenotype that we want to transplant.

Knowing IGF's functions in the different stem cell systems that could be used for cell transplantation is of prime importance for the development of such therapy. Indeed, if IGFs are necessary for stem cell renewal and the differentiation of various cells, as described in Sections 5 and 6, the understanding of their mechanisms of action may help to generate the adequate cell phenotype for a defined transplantation paradigm. Moreover, recent studies have shown that the combination of IGF-I delivery with hematopoietic stem cell transplantation enhances graft integration<sup>105</sup> showing that IGF-I can also play a role in the dynamic process of cell engrafting.

# 9. **REGENERATION**

Comprehension of how a stem cell can be stimulated *in vitro* and how the environment drives the related progeny to adopt a specific cell fate opens new approaches for tissue regeneration *in vivo*. Indeed, almost every organ contains stem cells during adulthood that can be recruited to repair a wounded tissue area. In consequence, it is potentially possible to mobilize the endogenous stem cells by exogenous factors to rebuild a lesioned area. This approach has had some important and encouraging success in different animal models, with the most recent impressive result being the regeneration of skeletal muscles of old mice by Notch activation<sup>106</sup>. Interestingly, in this study it was shown that the degeneration of muscle during aging was due to a depletion of the Notch ligand, delta, in satellite cells and not to an intrinsic impairment of these cells, bringing new hope for other tissue regeneration.

In this context, IGF-I was also shown to be synthesized in injured skeletal muscles<sup>107</sup>. Moreover after motoneuron axotomy in transgenic mice expressing IGF-I, the regeneration of muscle is enhanced in comparison to wild type animals<sup>108</sup>. The number of satellite cells and of Sca-1 multipotent stem cells derived from the bone marrow located in skeletal muscles is increased in these IGF-I-transgenic animals, suggesting a better regeneration process. In addition, the gene transfer of IGF-I cDNA by viral vectors into the muscle of young mice, or by transgenesis with a specific IGF-I expression in muscles, prevents the loss of muscular function due to aging<sup>109,110</sup>. Indeed at old age, IGF-I treated (during youth) animals maintain muscular strength and mass at a level comparable to young mice.

Concerning peripheral neurons, several studies have shown that ectopic IGF-I adjunction leads to improvement of neuron regeneration after nerve lesions (for review see Ishii et al., 1994). In the same study using motoneuron axotomy, IGF-I transgenic mice show a quicker recovery and better nerve conduction velocity in comparison to controls<sup>108</sup>. In the central nervous system, it was shown that IGF-I stimulates neurogenesis in the hippocampal region<sup>111</sup> and that other factors can mobilize neural stem cells residing around the lateral ventricles in the adult mouse brain like EGF<sup>112</sup>. This mobilization leads to the generation of new neurons<sup>111,112</sup>. Interestingly, the infusion of a high dose of insulin with FGF-2 in the chick eye induces the formation of new neurons in the periphery of the retina<sup>70</sup>. These experiments show that different regions of the CNS can generate new neurons. The combination of factors controlling stem cell renewal and generation of neurons, such as IGF-I, should help to reconstitute a damaged area of brain after stroke or ischemia or during a process that induces neural degeneration such as Parkinson's and Huntington's diseases.

As seen above with muscle cells, a denervated tissue is often more sensitive to degenerative processes. Such a phenomenon also occurs in the rat cornea when the trigeminal nerve is sectioned<sup>113</sup>. Indeed, when the corneal epithelium is removed 2 weeks after trigeminal denervation, the healing is delayed and incomplete in comparison to control animals<sup>114</sup>. It was previously shown that IGF-I and substance P synergize *in vivo* to enhance the closure of the corneal wound<sup>115</sup>. This positive effect was also observed when both substances were administrated on the wounded denervated

cornea<sup>114</sup>. These experiments demonstrate that IGF-I can accelerate and restore normal healing after corneal injury.

An increased healing rate was also observed in fractured bones in the presence of IGF-I<sup>116,117</sup>. Studies of the potential intracellular pathways used by IGF-I revealed that IRS-1 is involved in such phenomenon<sup>118</sup>. Indeed, deletion of the *IRS-1* gene leads to impairment of fracture healing, even 10 days post-surgery, whereas bone was well healed after 3 weeks in wild type animals. Overexpression of IRS-1 in the lesioned area of bone in *IRS-1*<sup>-/-</sup> mice resulted in a recovery of bone healing. Understanding of the mechanisms mediating IGF-I's action on bone healing, as described here, will certainly help to design precise strategies to assure union of fractured bones in problematic lesions.

Burns are a dramatic injury that affect skin, but also organs located deeper in the body, such as muscles. IGF-I was shown to improve tissue recovery after thermal injury<sup>119</sup>; however, IGF-I has side effects in such treatments, probably due to the high concentrations of IGF-I used. In order to bypass the problem of high concentration of IGF-I, Jeschke *et al.* transferred IGF-I cDNA to burned rats by subcutaneous injections of liposomes containing IGF-I cDNA<sup>119</sup>. Interestingly, IGF-I expression induced not only skin reepithelialization, but also preserved muscle protein content, as well as body weight. These experiments show that IGF-I also has a beneficial role on skin and muscle recovery after thermal injury, but also suggest that its action is tightly linked and limited by the local concentration of IGF-I in the wounded area.

Taken together, these data show that GH and IGF related factors play a role not only during development, but can also help to recapitulate certain stages of this development in injured tissue in order to regenerate this tissue, giving new hope for regenerative medicine.

# **10. CONCLUDING REMARKS**

The respective roles of GH and IGFs are still unclear for several cell systems, but with the availability of stem cells for almost all organs, the dissection of the functions and the mechanism of these growth factors will surely be solved, creating great hopes for the development of regenerative medicine or to target cancer stem cells as described above. Such knowledge will also help to adequately stimulate growth in children with severe growth deficiency without excessively stimulating the whole body. This is of great importance, because several studies have shown that the loss of GHR (GHR deficient mice) or of IGF-I functions (*IGF-I-R*<sup>+/-</sup> mice) leads to an increased lifespan<sup>120,121</sup> probably due to a decreased susceptibility to oxidative damage

generated in stressful conditions<sup>121</sup>. Sustaining these data, the overexpression of GH reduces lifespan<sup>122</sup>, suggesting that inadequate dosages of GH might have long-term side effects. The understanding of the GH-IGF axis is not only of great interest for medicine, but also allows us to contemplate the limits of the body's lifespan and the balance between the benefits and side effects of the actions of IGF.

## ACKNOWLEDGEMENTS

I would like to thank Brigitte Angénieux, Dusan Zencak, and Faten Merhi-Soussi for helpful discussions, and Muriel Jaquet for her assistance in the writing of this manuscript. Our work is supported by the Swiss National Science Foundation, the French Association against Myopathies, the Foundation ProVisu, and the Foundation Velux.

## REFERENCES

- 1. P.J. Donovan and J. Gearhart, The end of the beginning for pluripotent stem cells. *Nature*, **414**, 92-97 (2001).
- 2. G.Q. Daley, From embryos to embryoid bodies: generating blood from embryonic stem cells. *Ann NY Acad Sci*, 996, 122-131 (2003).
- 3. V. Tropepe, S. Hitoshi, C. Sirard, T.W. Mak, J. Rossant and K D van der, Direct neural fate specification from embryonic stem cells: a primitive mammalian neural stem cell stage acquired through a default mechanism. *Neuron*, **30**, 65-78 (2001).
- 4. B.A. Reynolds, W. Tetzlaff and S.A. Weiss, Multipotent EGF-responsive striatal embryonic progenitor cell produces neurons and astrocytes. *J Neurosci*, **12**, 4565-4574, (1992).
- E.K. Nishimura, S.A. Jordan, H. Oshima, H. Yoshida, M. Osawa, M. Moriyama, I.J. Jackson, Y. Barrandon, Y. Miyachi and S. Nishikawa, Dominant role of the niche in melanocyte stem-cell fate determination. *Nature*, 416, 854-860, (2002).
- L. Alonso and E. Fuchs, Stem cells of the skin epithelium. *Proc Natl Acad Sci USA*, 100 Suppl 1, 11830-11835, (2003).
- 7. M. Engelhardt, M. Lubbert and Y. Guo, CD34(+) or CD34(-): which is the more primitive? *Leukemia*, 16, 1603-1608 (2002).
- J.L. Christensen and I.L. Weissman, Flk-2 is a marker in hematopoietic stem cell differentiation: a simple method to isolate long-term stem cells. *Proc Natl Acad Sci* USA, 98, 14541-14546, (2001).
- 9. R.M. Seaberg and K.D. van der, Stem and progenitor cells: the premature desertion of rigorous definitions. *Trends Neurosci*, **26**, 125-131 (2003).
- 10. R. McKay, Stem cells in the central nervous system. Science, 276, 66-71 (1997).
- 11. Y. Arsenijevic, S. Weiss, B. Schneider and P. Aebischer, Insulin-like growth factor-I is necessary for neural stem cell proliferation and demonstrates distinct actions of

epidermal growth factor and fibroblast growth factor-2. *J Neurosci*, **21**, 7194-7202, 2001.

- B.A. Reynolds and S. Weiss, Clonal and population analyses demonstrate that an EGF-responsive mammalian embryonic CNS precursor is a stem cell. *Dev Biol*, 175, 1-13 (1996).
- V. Tropepe, M. Sibilia, B.G. Ciruna, J. Rossant, E.F. Wagner and K.D. van der, Distinct neural stem cells proliferate in response to EGF and FGF in the developing mouse telencephalon. *Dev Biol*, 208, 166-188 (1999).
- S. Ahmed, B.A. Reynolds and S. Weiss, BDNF enhances the differentiation but not the survival of CNS stem cell- derived neuronal precursors. *J Neurosci*, 15, 5765-5778 (1995).
- 15. Y. Arsenijevic and S. Weiss, Insulin-like growth factor-I is a differentiation factor for postmitotic CNS stem cell-derived neuronal precursors: distinct actions from those of brain-derived neurotrophic factor. *J Neurosci*, **18**, 2118-2128 (1998).
- 16. K.K. Johe, T. G. Hazel, T. Muller, M.M. Dugich-Djordjevic and R.D. McKay, Single factors direct the differentiation of stem cells from the fetal and adult central nervous system. *Genes Dev*, **10**, 3129-3140 (1996).
- A. Kalyani, K. Hobson and M.S. Rao, Neuroepithelial stem cells from the embryonic spinal cord: isolation, characterization, and clonal analysis. *Dev Biol*, 186, 202-223 (1997).
- 18. T.D. Palmer, J. Takahashi and F.H. Gage, The adult rat hippocampus contains primordial neural stem cells. *Mol Cell Neurosci*, **8**, 389-404 (1997).
- G. Zhu, M..F. Mehler, P.C. Mabie and J.A. Kessler, Developmental changes in progenitor cell responsiveness to cytokines. *J Neurosci Res*, 56, 131-145 (1999).
- 20. Y. Guo, M. Lubbert and M. Engelhardt, Stem Cells, 21: 15-20 (2003).
- A. Baroffio, M. Hamann, L. Bernheim, M.L. Bochaton-Piallat, G. Gabbiani and C.R. Bader, Identification of self-renewing myoblasts in the progeny of single human muscle satellite cells. *Differentiation*, **60**, 47-57 (1996).
- 22. B.M. Deasy, Z. Qu-Peterson, J.S. Greenberger and J. Huard, Mechanisms of muscle stem cell expansion with cytokines. *Stem Cells*, **20**, 50-60 (2002).
- 23. J. Frisen, C. B. Johansson, C. Torok, M. Risling and U. Lendahl, Rapid, widespread, and longlasting induction of nestin contributes to the generation of glial scar tissue after CNS injury. *J Cell Biol*, **131**, 453-464 (1995).
- 24. F. Ciccolini and C.N. Svendsen, Fibroblast growth factor 2 (FGF-2) promotes acquisition of epidermal growth factor (EGF) responsiveness in mouse striatal precursor cells: identification of neural precursors responding to both EGF and FGF-2. *J Neurosci*, **18**, 7869-7880 (1998).
- 25. T. Shimazaki, T. Shingo and S. Weiss, The ciliary neurotrophic factor/leukemia inhibitory factor/gp130 receptor complex operates in the maintenance of mammalian forebrain neural stem cells. *J Neurosci*, **21**, 7642-7653, (2001).
- P. Taupin, J. Ray, W.H. Fischer, S.T. Suhr, K. Hakansson, A. Grubb and F.H. Gage, FGF-2-responsive neural stem cell proliferation requires CCg, a novel autocrine/paracrine cofactor. *Neuron*, 28, 385-397 (2000).
- H. Toda, M.Tsuji, I. Nakano, K. Kobuke, T. Hayashi, H. Kasahara, J. Takahashi, A. Mizoguchi, T. Houtani, T. Sugimoto, N. Hashimoto, T.D. Palmer, T. Honjo and K. Tashiro, Stem cell-derived neural stem/progenitor cell supporting factor is an autocrine/paracrine survival factor for adult neural stem/progenitor cells. *J Biol Chem*, 278, 35491-35500 (2003).
- 28. C. Vicario-Abejon, M.J. Yusta-Boyo, C. Fernandez-Moreno and F. de Pablo, Locally born olfactory bulb stem cells proliferate in response to insulin-related

factors and require endogenous insulin-like growth factor-I for differentiation into neurons and glia. *J Neurosci*, **23**, 895-906 (2003).

- 29. G. Zhu, M.F. Mehler, J. Zhao, Y.S. Yu and J.A. Kessler, Sonic hedgehog and BMP2 exert opposing actions on proliferation and differentiation of embryonic neural progenitor cells. *Dev Biol*, **215**, 118-129 (1999).
- A.V. Molofsky, R. Pardal, T. Iwashita, I.K. Park, M.F. Clarke and S.J. Morrison, Bmi-1 dependence distinguishes neural stem cell self-renewal from progenitor proliferation. *Nature*, 425, 962-967 (2003).
- M. Kondo, A.J. Wagers, M.G. Manz, S.S. Prohaska, D.C. Scherer, G.F. Beilhack, J.A. Shizuru and I.L. Weissman, Biology of hematopoietic stem cells and progenitors: implications for clinical application. *Annu Rev Immunol*, 21, 759-806 (2003).
- J.C. Chen and D.J. Goldhamer, Skeletal muscle stem cells. *Reprod Biol Endocrinol*, 1, 101 (2003).
- 33. D.J. Anderson, Stem cells and transcription factors in the development of the mammalian neural crest. *FASEB J*, **8**, 707-713 (1994).
- Y. Arsenijevic, Mammalian neural stem-cell renewal: nature versus nurture. *Mol Neurobiol*, 27, 73-98 (2003).
- 35. M.F. Mehler and J.A. Kessler, Progenitor cell biology: implications for neural regeneration. *Arch Neurol*, **56**, 780-784 (1999).
- H. Robson, T. Siebler, S.M. Shalet and G.R. Williams, Interactions between GH, IGF-I, glucocorticoids, and thyroid hormones during skeletal growth. *Pediatr Res*, 52, 137-147 (2002).
- 37. D. Le Roith, C. Bondy, S. Yakar, J.L. Liu and A. Butler, The somatomedin hypothesis: 2001. *Endocr Rev*, 22, 53-74 (2001).
- R.A. Frost and C.H. Lang, Regulation of insulin-like growth factor-I in skeletal muscle and muscle cells. *Minerva Endocrinol*, 28, 53-73 (2003).
- 39. M. Majka, A. Janowska-Wieczorek, J. Ratajczak, K. Ehrenman Z. Pietrzkowski, M.A. Kowalska, A.M. Gewirtz, S.G. Emerson and M.Z. Ratajczak, Numerous growth factors, cytokines, and chemokines are secreted by human CD34(+) cells, myeloblasts, erythroblasts, and megakaryoblasts and regulate normal hematopoiesis in an autocrine/paracrine manner. *Blood*, 97, 3075-3085 (2001).
- 40. W. Zumkeller, The insulin-like growth factor system in hematopoietic cells. *Leuk Lymphoma*, **43**, 487-491 (2002).
- T. Shimazaki, Y. Arsenijevic, A.K. Ryan, M.G. Rosenfeld and S.A. Weiss, role for the POU-III transcription factor Brn-4 in the regulation of striatal neuron precursor differentiation. *EMBO J*, 18 444-456 (1999).
- G. Condorelli, A. Drusco, G. Stassi, A. Bellacosa, R. Roncarati, G. Iaccarino, M.A. Russo, Y. Gu, N. Dalton, C. Chung, M.V. Latronico, C. Napoli, J. Sadoshima, C.M. Croce and J. Ross Jr, Akt induces enhanced myocardial contractility and cell size in vivo in transgenic mice. *Proc Natl Acad Sci USA*, **99**, 12333-12338 (2002).
- 43. R.S. MacDonald, The role of insulin-like growth factors in small intestinal cell growth and development. *Horm Metab Res*, **31**, 103-113 (1999).
- O. Brustle, U. Maskos and R. D. McKay, Host-guided migration allows targeted introduction of neurons into the embryonic brain. *Neuron*, 15, 1275-1285 (1995).
- O. Brustle, K.N. Jones, R.D. Learish, K. Karram, K. Choudhary, O. Wiestler, I.D. Duncan and R.D. McKay, Embryonic stem cell-derived glial precursors: a source of myelinating transplants. *Science*, 285, 754-756, 1999.
- 46. J.D. Flax, S. Aurora, C. Yang, C. Simonin, A.M. Wills, L.L. Billinghurst, M. Jendoubi, R.L. Sidman, J.H. Wolfe, S.U. Kim, and E.Y. Snyder, Engraftable human

neural stem cells respond to developmental cues, replace neurons, and express foreign genes. *Nat Biotechnol*, **16**, 1033-1039, 1998.

- 47. C.M. Rosario, B.D. Yandava, B. Kosaras, D. Zurakowski, R.L. Sidman and E.Y. Snyder, Differentiation of engrafted multipotent neural progenitors towards replacement of missing granule neurons in meander tail cerebellum may help determine the locus of mutant gene action. *Development*, **124**, 4213-4224 (1997).
- 48. J.O. Suhonen, D.A. Peterson J. Ray and F.H. Gage, Differentiation of adult hippocampus-derived progenitors into olfactory neurons in vivo. *Nature*, **383**, 624-627 (1996).
- 49. J.M. Auerbach, M.V. Eiden and R.D. McKay, Transplanted CNS stem cells form functional synapses in vivo. *Eur J Neurosci*, **12**, 1696-1704 (2000).
- 50. M. Osawa, K. Hanada, H. Hamada and H. Nakauchi, Long-term lymphohematopoietic reconstitution by a single CD34-low/negative hematopoietic stem cell. *Science*, **273**, 242-245 (1996).
- 51. O. Schultz, M. Sittinger, T. Haeupl and G.R. Burmester, Emerging strategies of bone and joint repair. *Arthritis Res*, **2**, 433-436 (2000).
- P. Rama, S. Bonini, A. Lambiase, O. Golisano, P. Paterna, M. De Luca and G. Pellegrini, Autologous fibrin-cultured limbal stem cells permanently restore the corneal surface of patients with total limbal stem cell deficiency. *Transplantation*, **72**, 1478-1485 (2001).
- M.A. Blasco, H.W. Lee, M.P. Hande, E. Samper, P.M. Lansdorp, R.A. DePinho and C.W. Greider, Telomere shortening and tumor formation by mouse cells lacking telomerase RNA. *Cell*, 91, 25-34 (1997).
- 54. C.W. Greider, Telomeres, telomerase and senescence. *Bioessays*, **12**, 363-369 (1990).
- 55. C.A. Conover, L.A. Dollar, R.L. Hintz, and R.G. Rosenfeld, Insulin-like growth factor I/somatomedin-C (IGF-I/SM-C) and glucocorticoids synergistically regulate mitosis in competent human fibroblasts. *J Cell Physiol*, **116**, 191-197 (1983).
- J.J. Cook, K.M. Haynes and G.A. Werther, Mitogenic effects of growth hormone in cultured human fibroblasts. Evidence for action via local insulin-like growth factor I production. J Clin Invest, 81: 206-212 (1988).
- 57. C.C. Zhang and H.F. Lodish, Insulin-like growth factor 2 expressed in a novel fetal liver cell population is a growth factor for hematopoietic stem cells. *Blood*, **103**, 2513-2521 (2004).
- L.Q. Liu, M. Sposato, H.Y. Liu, T. Vaudrain, M.J. Shi, K. Rider, Z. Stevens, J. Visser, H.K. Deng and M. Kraus, Functional cloning of IGFBP-3 from human microvascular endothelial cells reveals its novel role in promoting proliferation of primitive CD34+CD38- hematopoietic cells in vitro. *Oncol Res*, 13, 359-371 (2003).
- K.W. Kelley, W.A. Meier, C. Minshall, D.H. Schacher, Q. Liu, R. VanHoy, W. Burgess and R. Dantzer, Insulin growth factor-I inhibits apoptosis in hematopoietic progenitor cells. Implications in thymic aging. *Ann NY Acad Sci*, 840, 518-524, (1998).
- L.M. Wang, A.D. Keegan, W. Li, G.E. Lienhard, S. Pacini, J.S. Gutkind, M.G. Myers, Jr, X.J. Sun, M.F. White and S.A. Aaronson, Common elements in interleukin 4 and insulin signaling pathways in factor-dependent hematopoietic cells. *Proc Natl Acad Sci USA*, **90**, 4032-4036 (1993).
- 61. N. Hattori, T. Saito, T. Yagyu, B.H. Jiang, K. Kitagawa and C. Inagaki, GH, GH receptor, GH secretagogue receptor, and ghrelin expression in human T cells, B cells, and neutrophils. *J Clin Endocrinol Metab*, **86**, 4284-4291 (2001).

- B. Varnum-Finney, L. Xu, C. Brashem-Stein, C. Nourigat, D. Flowers, S. Bakkour, W.S. Pear and I.D. Bernstein, Pluripotent, cytokine-dependent, hematopoietic stem cells are immortalized by constitutive Notch1 signaling. *Nat Med*, 6, 1278-1281 (2000).
- C. Schaniel, M. Gottar E. Roosnek F. Melchers and A.G. Rolink, Extensive in vivo self-renewal, long-term reconstitution capacity, and hematopoietic multipotency of Pax5-deficient precursor B-cell clones. *Blood*, **99**, 2760-2766, 2002.
- M.S. Pampusch, E. Kamanga-Sollo, M.E. White, M.R. Hathaway and W.R. Dayton, Effect of recombinant porcine IGF-binding protein-3 on proliferation of embryonic porcine myogenic cell cultures in the presence and absence of IGF-I. *J Endocrinol*, 176, 227-235, 2003.
- 65. A. Gritti, L. Cova, E.A. Parati, R. Galli and A.L. Vescovi, Basic fibroblast growth factor supports the proliferation of epidermal growth factor-generated neuronal precursor cells of the adult mouse CNS. *Neurosci Lett*, **185**, 151-154, 1995.
- D.H. Rowitch, B. Jacques, S.M. Lee, J.D. Flax, E.Y. Snyder and A.P. McMahon, Sonic hedgehog regulates proliferation and inhibits differentiation of CNS precursor cells. *J Neurosci*, **19**, 8954-8965 (1999).
- 67. R.J. Wechsler-Reya and M.P. Scott, Control of neuronal precursor proliferation in the cerebellum by Sonic Hedgehog. *Neuron*, 22, 103-114 (1999).
- M.K. Carpenter, X. Cui, Z.Y. Hu, J. Jackson, S. Sherman, A. Seiger and L.U. Wahlberg, In vitro expansion of a multipotent population of human neural progenitor cells. *Exp Neurol*, **158**, 265-278 (1999).
- S. Hitoshi, T. Alexson, V. Tropepe, D. Donoviel, A.J. Elia, J.S. Nye, R.A. Conlon, T.W. Mak, A. Bernstein and K.D. van der, Notch pathway molecules are essential for the maintenance, but not the generation, of mammalian neural stem cells. *Genes Dev*, 16, 846-858 (2002).
- A.J. Fischer, B.D. Dierks and T.A. Reh, Exogenous growth factors induce the production of ganglion cells at the retinal margin. *Development*, **129**, 2283-2291 (2002).
- E. Marshman and C.H. Streuli, Insulin-like growth factors and insulin-like growth factor binding proteins in mammary gland function. *Breast Cancer Res*, 4, 231-239 (2002).
- 72. D.J. Flint, J. Beattie and G.J. Allan, Modulation of the actions of IGFs by IGFBP-5 in the mammary gland. *Horm Metab Res*, **35**, 809-815 (2003).
- R.A. Moorehead, C.V. Hojilla, B.I. De, G.A. Wood, J.E. Fata, E.D. Adamson, K.L. Watson, D.R. Edwards and R. Khokha, Insulin-like growth factor-II regulates PTEN expression in the mammary gland. *J Biol Chem*, 278, 50422-50427 (2003).
- S. Neuenschwander, A. Schwartz, T.L. Wood, C.T. Roberts Jr, L. Hennighausen and D. LeRoith, Involution of the lactating mammary gland is inhibited by the IGF system in a transgenic mouse model. J Clin Invest, 97, 2225-2232 (1996).
- 75. W. Ruan and D.L. Kleinberg, Insulin-like growth factor I is essential for terminal end bud formation and ductal morphogenesis during mammary development. *Endocrinology*, **140**, 5075-5081 (1999).
- 76. S. Deeks, J. Richards and S. Nandi, Maintenance of normal rat mammary epithelial cells by insulin and insulin-like growth factor 1. *Exp Cell Res*, **174**, 448-460 (1988).
- A. Herrler, C.A. Krusche and H.M. Beier, Insulin and insulin-like growth factor-I promote rabbit blastocyst development and prevent apoptosis. *Biol Reprod*, 59, 1302-1310 (1998).

- A.D. Lighten, G.E. Moore, R.M. Winston and K. Hardy, Routine addition of human insulin-like growth factor-I ligand could benefit clinical in-vitro fertilization culture. *Hum Reprod*, 13, 3144-3150 (1998).
- T.C. Lin, J.M. Yen, K.B. Gong, T.T. Hsu and L.R. Chen, IGF-1/IGFBP-1 increases blastocyst formation and total blastocyst cell number in mouse embryo culture and facilitates the establishment of a stem-cell line. *BMC Cell Biol*, 4, 14 (2003).
- 80. Y. Pei, J. Ma, X. Zhang and W. Ji, Serum-free culture of rhesus monkey embryonic stem cells. *Arch Androl*, **49**, 331-342 (2003).
- 81. J. Baker, J.P. Liu, E.J. Robertson and A. Efstratiadis, Role of insulin-like growth factors in embryonic and postnatal growth. *Cell*, **75**, 73-82 (1993).
- J.P. Liu, J. Baker, A.S. Perkins, E.J. Robertson and A. Efstratiadis, Mice carrying null mutations of the genes encoding insulin-like growth factor I (Igf-1) and type 1 IGF receptor (Igf1r). *Cell*, **75**, 59-72 (1993).
- 83. E. Fuchs and J.A. Segre, Stem cells: a new lease on life. Cell, 100, 143-155 (2000).
- C.B. Reid, S.F. Tavazoie and C.A. Walsh, Clonal dispersion and evidence for asymmetric cell division in ferret cortex. *Development*, 124, 2441-2450 (1997).
- 85. C.S. Potten and M. Loeffler, Stem cells: attributes, cycles, spirals, pitfalls and uncertainties. Lessons for and from the crypt. *Development*, **110**, 1001-1020 (1990).
- 86. A. Chenn and C.A. Walsh, Regulation of cerebral cortical size by control of cell cycle exit in neural precursors. *Science*, **297**, 365-369 (2002).
- 87. E.B. Hunziker, J. Wagner and J. Zapf, Differential effects of insulin-like growth factor I and growth hormone on developmental stages of rat growth plate chondrocytes in vivo. *J Clin Invest*, **93**, 1078-1086 (1994).
- C. Ohlsson, B.A. Bengtsson, O.G. Isaksson, T.T. Andreassen and M.C. Slootweg, Growth hormone and bone. *Endocr Rev*, **19**, 55-79 (1998).
- K. Sjogren, J.L. Liu, K. Blad, S. Skrtic, O. Vidal, V. Wallenius, D. LeRoith, J. Tornell, O.G. Isaksson, J.O. Jansson and C. Ohlsson, Liver-derived insulin-like growth factor I (IGF-I) is the principal source of IGF-I in blood but is not required for postnatal body growth in mice. *Proc Natl Acad Sci USA*, 96, 7088-7092 (1999).
- S. Yakar, J.L. Liu, B. Stannard, A. Butler, D. Accili, B. Sauer and D. LeRoith, Normal growth and development in the absence of hepatic insulin-like growth factor I. *Proc Natl Acad Sci USA*, **96**, 7324-7329 (1999).
- S.E. Tollefsen, J.L. Sadow and P. Rotwein, Coordinate expression of insulin-like growth factor II and its receptor during muscle differentiation. *Proc Natl Acad Sci* USA, 86, 1543-1547 (1989).
- S.E. Tollefsen, R. Lajara, R.H. McCusker, D.R. Clemmons and P. Rotwein, Insulinlike growth factors (IGF) in muscle development. Expression of IGF-I, the IGF-I receptor, and an IGF binding protein during myoblast differentiation. *J Biol Chem*, 264, 13810-13817 (1989).
- Y. Arsenijevic, J.G. Villemure, J.F. Brunet, J.J. Bloch, N. Deglon, C. Kostic, A. Zurn and P. Aebischer, Isolation of multipotent neural precursors residing in the cortex of the adult human brain. *Exp Neurol*, **170**, 48-62 (2001).
- 94. T. Reya, S.J. Morrison, M.F. Clarke and I.L. Weissman, Stem cells, cancer, and cancer stem cells. *Nature*, **414**, 105-111 (2001).
- L. Del Valle, S. Enam, A. Lassak, J.Y. Wang, S. Croul, K. Khalili and K. Reiss, Insulin-like growth factor I receptor activity in human medulloblastomas. *Clin Cancer Res*, 8, 1822-1830 (2002).
- J.G. Scharf and T. Braulke, The role of the IGF axis in hepatocarcinogenesis. *Horm Metab Res*, 35, 685-693 (2003).

- H.M. Khandwala, I.E. McCutcheon, A. Flyvbjerg and K.E. Friend, The effects of insulin-like growth factors on tumorigenesis and neoplastic growth. *Endocr Rev*, 21, 215-244 (2000).
- 98. D.B. Kaufman and W.L. Lowe Jr. Clinical islet transplantation. *Curr Diab Rep*, 3, 344-350 (2003).
- 99. A. Bjorklund and O. Lindvall, Cell replacement therapies for central nervous system disorders. *Nat Neurosci*, **3**, 537-544 (2000).
- 100. A.C. Bachoud-Levi, P. Remy, J.P. Nguyen, P. Brugieres, J.P. Lefaucheur, C. Bourdet, S. Baudic, V. Gaura, P. Maison, B. Haddad, M.F. Boisse, T. Grandmougin, R. Jeny, P. Bartolomeo, B.G. Dalla, J.D. Degos, F. Lisovoski, A.M. Ergis, E. Pailhous, P. Cesaro, P. Hantraye, and M. Peschanski, Motor and cognitive improvements in patients with Huntington's disease after neural transplantation. *Lancet*, 356, 1975-1979 (2000).
- 101. C.R. Freed, M.A. Leehey, M. Zawada, K. Bjugstad, L. Thompson, and R.E. Breeze, Do patients with Parkinson's disease benefit from embryonic dopamine cell transplantation? *J Neurol*, **250**, 11144-11146 (2003).
- 102. P. Brundin, J. Karlsson, M. Emgard, G.S. Schierle, O. Hansson, A. Petersen and R.F. Castilho, Improving the survival of grafted dopaminergic neurons: a review over current approaches. *Cell Transplant*, 9, 179-195 (2000).
- 103. A. Bjorklund, S.B. Dunnett, P. Brundin, A.J. Stoessl, C.R. Freed, R.E. Breeze, M. Levivier, M. Peschanski, L. Studer and R. Barker, Neural transplantation for the treatment of Parkinson's disease. *Lancet Neurol*, 2, 437-445 (2003).
- 104. A.N. Mamelak, F.A. Eggerding, DS. Oh, E. Wilson, R.L. Davis, R. Spitzer, J.A. Hay and W.L. Caton III, Fatal cyst formation after fetal mesencephalic allograft transplant for Parkinson's disease. *J Neurosurg*, **89**, 592-598 (1998).
- 105. E. Montecino-Rodriguez, R. Clark and K. Dorshkind, Effects of insulin-like growth factor administration and bone marrow transplantation on thymopoiesis in aged mice. *Endocrinology*, **139**, 4120-4126 (1998).
- 106. I.M. Conboy, M.J. Conboy, G.M. Smythe and T.A. Rando, Notch-mediated restoration of regenerative potential to aged muscle. *Science*, **302**, 1575-1577 (2003).
- 107. M. Hill, A. Wernig and G. Goldspink, Muscle satellite (stem) cell activation during local tissue injury and repair. *J Anat*, **203**, 89-99 (2003).
- 108. E.D. Rabinovsky, E. Gelir, S. Gelir, H. Lui, M. Kattash, F.J. DeMayo, S.M. Shenaq and R.J. Schwartz, Targeted expression of IGF-1 transgene to skeletal muscle accelerates muscle and motor neuron regeneration. *FASEB J*, 17, 53-55 (2003).
- 109. E.R. Barton-Davis, D.I. Shoturma, A. Musaro, N. Rosenthal and H.L. Sweeney, Viral mediated expression of insulin-like growth factor I blocks the aging-related loss of skeletal muscle function. *Proc Natl Acad Sci USA*, 95, 15603-15607 (1998).
- 110. A. Musaro, K. McCullagh, A. Paul, L. Houghton, G. Dobrowolny, M. Molinaro, E.R. Barton H.L. Sweeney and N. Rosenthal, Localized Igf-1 transgene expression sustains hypertrophy and regeneration in senescent skeletal muscle. *Nat Genet*, 27, 195-200 (2001).
- 111. M.A. Aberg, N.D. Aberg, H. Hedbacker, J. Oscarsson and P.S. Eriksson, Peripheral infusion of IGF-I selectively induces neurogenesis in the adult rat hippocampus. J *Neurosci*, 20, 2896-2903 (2000).
- 112. C.G. Craig, V. Tropepe, C.M. Morshead, B.A. Reynolds, S. Weiss and K.D. van der, In vivo growth factor expansion of endogenous subependymal neural precursor cell populations in the adult mouse brain. *J Neurosci*, **16**, 2649-2658 (1996).

- 113. R.W. Beuerman and B. Schimmelpfennig, Sensory denervation of the rabbit cornea affects epithelial properties. *Exp Neurol*, **69**, 196-201 (1980).
- 114. T. Nagano, M. Nakamura, K. Nakata, T. Yamaguchi, K. Takase, A. Okahara, T. Ikuse and T. Nishida, Effects of substance P and IGF-1 in corneal epithelial barrier function and wound healing in a rat model of neurotrophic keratopathy. *Invest Ophthalmol Vis Sci*, 44, 3810-3815 (2003).
- 115. M. Nakamura, K. Ofuji T. Chikama and T. Nishida, Combined effects of substance P and insulin-like growth factor-1 on corneal epithelial wound closure of rabbit in vivo. *Curr Eye Res*, 16, 275-278 (1997).
- 116. T.A. Linkhart, S. Mohan and D.J. Baylink, Growth factors for bone growth and repair: IGF, TGF beta and BMP. *Bone*, **19**, 1S-12S (1996).
- 117. G. Schmidmaier, B. Wildemann, T. Gabelein, J. Heeger, F. Kandziora, N. P. Haas and M. Raschke, Synergistic effect of IGF-I and TGF-beta1 on fracture healing in rats: single versus combined application of IGF-I and TGF-beta1. *Acta Orthop Scand*, **74**, 604-610 (2003).
- 118. T. Shimoaka, S. Kamekura, H. Chikuda, K. Hoshi, U.I. Chung, T. Akune, Z. Maruyama, T. Komori, M. Matsumoto, W. Ogawa, Y. Terauchi, T. Kadowaki, K. Nakamura and H. Kawaguchi, Impairment of bone healing by insulin receptor substrate-1 deficiency. *J Biol Chem* 279, 15314-15322 (2004).
- 119. M.G. Jeschke, R.E. Barrow, H.K. Hawkins, K. Yang, R.L. Hayes, B.J. Lichtenbelt, J.R. Perez-Polo and D.N. Herndon, IGF-I gene transfer in thermally injured rats. *Gene Ther*, 6, 1015-1020 (1999).
- 120. K.T. Coschigano, A.N. Holland, M.E. Riders, E.O. List, A. Flyvbjerg and J.J. Kopchick, Deletion, but not antagonism, of the mouse growth hormone receptor results in severely decreased body weights, insulin, and insulin-like growth factor I levels and increased life span. *Endocrinology*, **144**, 3799-3810 (2003).
- 121. M. Holzenberger, J. Dupont, B. Ducos, P. Leneuve, A. Geloen, P.C. Even, P. Cervera and Y. Le Bouc, IGF-1 receptor regulates lifespan and resistance to oxidative stress in mice. *Nature*, **421**, 182-187 (2003).
- 122. Z. Laron, Effects of growth hormone and insulin-like growth factor 1 deficiency on ageing and longevity. *Novartis Found Symp*, **242**, 125-137 (2002).

# INDEX

Abnormal eye movement, 225 Acid-labile subunit, 56, 67, 69-76 hepatic insulin-like growth factor-I expression, 76 inactivation of, 75-76 overexpression, ubiquitous, 76 Acromegaly, 265 Aging, 259-284 acromegaly, 265 cognitive function, enhancement of, 272 diabetogenic factor, 267 elderly, growth hormone therapy in, 271-272 free radical theory of aging, 269 growth hormone overexpression, 265 hyperinsulinemia, 265 insulin resistance, 265 insulin sensitivity, 268 Laron syndrome, 262 oxidative damage, 269 oxidative metabolism, 269 reproduction, 268-269 reproductive competence, 268-269 reproductive function, 269 stress resistance, 269 271 Alzheimer amyloidosis, 249 Amyloid-beta injury, 205 Amyotrophic lateral sclerosis, 225 Androgen, ovarian insulin-like growth factor system, 98-101 hyperandrogenism, 99 polycystic ovaries, pathogenesis of, 99-101 polycystic ovary syndrome, development of, 99-101 primordial follicle development, promotion of, 99 Anterior pituitary, secretion from, 3 Anti-insulin-like, 9-10 Apnea, sleep, 159 Apoptosis, 167-220, 305-340 Astrocytes, 187-220 insulin-like growth factor actions on, 202 Auditory sensory system, vestibular systems, 230-233

Autocrine function, 132-135, 305-340 Autoimmune encephalomyelitis, 205 Beta amyloid, 249 Binding protein, 1–26 biological functions of, 6 insulin-like growth factor, 14 Bladder cancer, 313 Blood-brain barrier, 188 Bone marrow, 167-186 Brain, estrogen, interactions of insulin-like growth factor-I cerebellar cortex, 293 cerebral cortex, 293 dentate gyrus, 293 frontoparietal cortex, 293 granule cell layer, 293 hippocampal formation, 291 hippocampus, 293 hypothalamic nuclei, 293 immunohistochemical colocalization, 291-294 insulin-like growth factor-I signaling, estrogen, interaction between, 295 medial preoptic nucleus, 293 molecular layer, 293 neuroprotection, 290 posterior cingulate cortex, 293 preoptic area, 293 primary olfactory cortex, 293 purkinje cell layer, 293 Brain-derived neurotrophic factor, 199 Brain development, 187-220 astrocytes, microglia, insulin-like growth factor actions on, 202 blood-brain barrier, 188 brain-derived neurotrophic factor, 199 central nervous system, 194-195, 202-203 epidermal growth factor, 195 fibroblast growth factor, 195 insulin-like growth factor binding proteins, 192-194 myelination, 200-202 insulin-like growth factor actions on, 199-202

#### 414

neural stem cells, 195-198 neurons, insulin-like growth factor actions on, 198-199 neurite growth, 199 neurogenesis, 198-199 synaptogenesis, 199 oligodendrocyte lineage, 199-200 oligodendrocytes, insulin-like growth factor actions on, 199-202 Brain function, 243-258 Brain injury, 187-220 Brain senescence, 249 Breast cancer, 312 Burns, 404 C-terminal domains, 56 Caenorhabditis elegans model, 36, 41, 43-45 Dauer arrest, 44 neurosecretory system, 43-44 Cancer, 305-340, 385 animal models, 309-310, 313-314 autocrine/paracrine growth hormone/insulin-like growth factor axis, 314-321 bladder, 313 breast, 312 clinical implications, 324-326 colon, 308 colorectal adenoma, 312 colorectal neoplasia, 307 decreased expression, 320 endocrine growth hormone/insulin-like growth factor axis, 306-314 endometrial, 313 epidemiologic studies, 311-313 gastric adenocarcinoma, 312 gastro-intestinal colorectal, 311 growth hormone deficiency, 306-307 malignant transformation, 317-318 normal growth hormone variability, 308-309 prevention, 325 prostate, 308, 311 proteolysis, 320 treatment, 325-326 tumor suppressors, 323 Cardiac output, 149-166 Cardiomyocytes, 149-166 Cardiovascular system, 149-166 calcium, hypertrophy-signaling, 150 cardiac hypertrophy, 149-150, 154 genes involved in, 154

cardiac output, 150-151 factors determining, 151-152 afterload, 152 left ventricular myocardial mass, 152 myocardial contractility, 152 preload, 151-152 cardiac remodeling, 152 Frank-Starling mechanism, 152 cardiomyocytes, 154 cardiovascular system, structure, 149-152 contractility, 155-156 endothelial dysfunction, defined, 151 extracellular matrix, 154 growth hormone deficiency, 156-157 hypertrophy, defined, 149 intracellular Ca-transients, 155-156 myocardium, 149-150 myofilament Ca-sensitivity and, myosin isoform shift, 156 nitric oxide, 151 peripheral resistance, 150-151, 153 structure, 149-152 transgenic mice, 153 vascular endothelial function, 150-151 vasodilators, 153 Cell adhesion, motility, 13 Cell differentiation, 187-220, 385 Cell survival, protective effect on, 13-14 Cell transplantation, 402 Central nervous system, 194-195, 202-203 Cerebellar ataxia, 249 Cerebellar cortex, 293 Cerebellar deafferentation, 205 Cerebral artery occlusion, 205 Cerebral cortex, 293 Chemokines, 167-186 CHICO, 42, 45 Chronic obstructive pulmonary disease, 159 Chronic renal failure, 68 Circulating insulin-like growth factor-I, reduction of, 33-34 Classical insulin-like growth factor-I receptor knockout mice, 35-36 Clonal analysis, 387-388 Cognate receptors, 125 Cognitive function, enhancement of, 272 Colon cancer, 308 Colorectal adenoma, 312 Colorectal neoplasia, 307 Conditional gene targeting of insulin-like growth factor-IR, 37-38 Conditional knockouts. 29-30 Cre-loxP recombination system, 29

local vs. systemic hormonal effects, 33-35 Conventional knockouts, 28-29 chimeric adults, 28 COPD. See Chronic obstructive pulmonary disease Cornelia de Lange syndrome, 69 Creillox recombination, 36 Crosstalk mechanism, 131 Cryogenics, 205 Crystallographic structure, receptor complexes, 6 Cuprizone, 205 Cytokines, 167–186 Cytoskeletal remodeling, 133 Deafness, 225 Deletions, insulin receptor substrate signaling proteins, phenotypes resulting from. 42 Dentate gyrus, 293 colchicine injection, 205 Diabetes, 132-133 Diabetogenic factor, 267 Discovery, date of, 2 Drosophila, 36, 42 human insulin receptors, contrasted, 45 insulin receptor signaling, 45-46 Duchenne muscular dystrophy, 68 Dwarf, 259-284 Dyslipidemia, 42 Dysregulated tau, 42 Ectrolytic lesion to hippocampus experimental, 205 Elderly, growth hormone therapy in, 271-272 Endocrine system, 10-11, 305-340 Endometrial cancer, 313 Endometrium, 91-116 Energy homeostasis, disorders of, 41 Epidermal growth factor, 195 Epithelial cells, 167-186 Erythropoiesis, stimulation of, 173 Estradiol, 285-304 Estrogen, interactions of insulin-like growth factor-I, in brain, 285-304 cerebellar cortex, 293 cerebral cortex, 293 dentate gyrus, 293 estradiol, 286-287 frontoparietal cortex, 293 gonadotrophins, 287 granule cell layer, 293

hippocampal formation, 291 hippocampus, 293 hypothalamic arcuate nucleus, 287-288 hypothalamic nuclei, 293 immunohistochemical colocalization, 291-294 insulin-like growth factor-I signaling, estrogen, interaction between, 295 medial preoptic nucleus, 293 molecular layer, 293 neuroprotection, 290 posterior cingulate cortex, 293 preoptic area, 293 primary olfactory cortex, 293 purkinje cell layer, 293 synaptic remodeling, 288-289 Experimental models, 27-54 Caenorhabditis elegans, 36, 41, 43-45 Dauer arrest, 44 neurosecretory system, 43-44 CHICO, 42, 45 circulating insulin-like growth factor-I, reduction of, 33-34 classical insulin-like growth factor-I receptor knockout mice, 35-36 conditional gene targeting of insulin-like growth factor-IR, 37-38 conditional knockouts, 29-30 Cre-loxP recombination system, 29 local vs. systemic hormonal effects, 33-35 conventional knockouts, 28-29 chimeric adults, 28 Creillox recombination, 36-37 deficiency, mouse models, 31-38 ablation of, 32 homozygous disruption of, 32 null mutants, 32 deletions, insulin receptor substrate signaling proteins, phenotypes resulting from, 42 Drosophila, 36, 42 human insulin receptors, contrasted, 45 insulin receptor signaling, 45-46 dyslipidemia, 42 dysregulated tau, 42 energy homeostasis, disorders of, 41 future developments, 46 genetic ablation, substrate signaling proteins, 38-43 glucose intolerance, 42 growth retardation in, 35 hepatic insulin resistance, 42

### hyperglycemia, 42 hyperinsulinemia, 42 hyperleptinemia, 42 hyperphagia, 42 hypertension, 42 impaired adipocyte differentiation, 42 infertility, 42 insulin-like growth factor-IR deficiency, perinatal lethality caused by, 36 insulin receptor substrate-mediated signal transduction, 37 insulin receptor substrate proteins, 38 insulin resistance in skeletal muscle, 42

knockouts, 39-40 signaling proteins, intercrossing of, 40 mammalian brain, development, 42 neuroendocrine phenotypes, 41 neuroendocrine tissues, 40-43 neuroprotection, 34 null neonates, low survival rate of, 33 perinatal lethality, 37 phosphorylation, 42 plasma membrane proteins, 35 RNA interference, 30-31 amplification activity, 31 Caenorhabditis elegans, 30-31 Drosophila, 30 signaling cascades, intracellular, 38 signaling system, evolutionarily conserved elements, 43 somatomedin hypothesis, 33 technology, 28-31 tissue-specific knockouts, 37-38 Extracellular matrix, 154, 167-186 Eye movement, abnormal, 225 Female reproductive system, 91-116 androgen, ovarian insulin-like growth

androgen, ovarian insulin-like growth factor system, 98–101 insulin-like growth factorPBs, 102–106 ovary, 92–106 follicle-stimulating hormone, 93 growth hormone, 92–93 insulin-like growth factor-I receptor, 93–98 reproduction, energy costs of, 110 uterus, 106–110 insulin-like growth factor-I as "estromedin," 106–109 insulin-like growth factor system in primate uterus, 109–110 Fetal growth, 15 Fibroblast growth factor, 195 Fractured bones, 404 Free radical theory of aging, 269 Frontoparietal cortex, 293 Gastric adenocarcinoma, 312 Gastro-intestinal colorectal, 311 Gene targeting, 27-54, 341-384 Genetic ablation, substrate signaling proteins, 38-43 Glia, 187-220 Glucose intolerance, 42 Glucose regulation, 259-284 Gonadotrophin-releasing hormone neurons, 2.2.8Gonadotrophins, 285-304 Granule cell layer, 293 Growth hormone-releasing hormone, 1-26 Growth hormones aging, 259-284 binding proteins, 55-90 brain development, 187-220 cancer, 305-340 cardiovascular system, 149-166 estrogen, in brain, 285-304 female reproductive system, 91-116 future developments, 385-412 hematopoiesis, 167-186 insulin-like growth factor-I, experimental models, 27-54 neuronal plasticity, 243-258 physiology of, 1-26 proportional short stature, genetic basis of, 341-384 respiratory systems, 149-166 sensory organ development, 221-242 skeletal muscle, 117-148 Growth-promoting polypeptides, 55 Gustatory system, 229-230 taste buds differentiation, 229 tongue development, 229 Hearing loss, 225 Heart failure, 158 Hematopoiesis, 167-186 autocrine/paracrine way, 178 erythropoiesis, stimulation of, 173 growth hormone/ICF-I circuits, 177-178 thymocyte differentiation, 178 hematopoietic microenvironments, 167 - 172bone marrow, 168-169

thymic epithelial cells, 170

### 416

thymopoiesis, 171 thymus, 169-172 immunodeficiency, 180 intracellular signaling pathways, 179 migration of lymphocytes, influence on, 176 migratory capacity of peripheral cells, 176 modulating apoptosis, role in, 175 multiple myeloma cells, chemoattractant for, 175 thymic hormone secretion, modulation of, 173 thymocytes, proliferation of, 173 transgenic mice, 174 Hepatic insulin resistance, 42 HESX1, 357-359 High-affinity binding proteins, 125-126 Hippocampal formation, 291 Hippocampus, 293 electrolytic lesion, 205 Homologous protein family, 55-56 Homologous recombination techniques, 128-129 Huntingtin's disease, 249 Hyperglycemia, 42 Hyperinsulinemia, 42, 265 Hyperleptinemia, 42 Hyperphagia, 42 Hypertension, 42, 157-158 Hypertrophy, cardiac, 154 genes involved in, 154 Hypodactyly, 68-69 Hypoglycemia, 249 Hypopituitary patients, 156 Hypothalamic control, growth hormone secretion, 3-4 Hypothalamic nuclei, 293 Hypoxia, 249 Idiopathic late onset cerebellar ataxia, 225 Immunodeficiency, 180 Immunohistochemical colocalization, 291-294 Impaired adipocyte differentiation, 42 Infertility, 42 Inner ear, 221-242 Insulin-like growth factor, 167-186 aging, 259–284 binding proteins, 55-90 brain development, 187-220 cancer, 305-340 cardiovascular system, 149-166

estrogen, in brain, 285-304

female reproductive system, 91-116 future developments, 385-412 hematopoiesis, 167-186 insulin-like growth factor-I, experimental models, 27-54 neuronal plasticity, 243-258 physiology of, 1-26 proportional short stature, genetic basis of, 341-384 respiratory systems, 149-166 sensory organ development, 221-242 skeletal muscle, 117-148 Insulin-like growth factor binding protein, 1, 55-116, 187-220 acid-labile subunit, 64-65 binding sites for, 62 expression, 58-65 inhibitory role of, 60 regulators of, 61-62 ternary complex, formation of, 64-65 tissue expression, 59-62, 64 Insulin-like growth factor binding protein-1, 58-60, 65-67, 70-72 deficiency of, 70 endogenous fetal overexpression, 72 impaired glucose tolerance, 72 overexpression of, 70-72 Insulin-like growth factor binding protein-2, 60-61,73 importance of tissue concentration of inhibitory, 73 ubiquitous overexpression of, 73 Insulin-like growth factor binding protein-3, 61-62, 67 Insulin-like growth factor binding protein-4, 62-63,74 selective overexpression, 74 targeted overexpression, 74 Insulin-like growth factor binding protein-5, 63, 74-75 targeted overexpression, 75 transgenic mice, 75 ubiquitous overexpression, 74-75 Insulin-like growth factor binding protein-6, 63-64.75 Insulin-like growth factor receptor, 91-116 Krause's corpuscles, 230 Laron's syndrome, 225, 262 Leber congenital amaurosis syndrome, 225, 227 Leprechaunism, 66

#### 418

Life span, 259-284 acromegaly, 265 cognitive function, enhancement of, 272 diabetogenic factor, 267 elderly, growth hormone therapy in, 271-272 free radical theory of aging, 269 growth hormone overexpression, 265 hyperinsulinemia, 265 insulin resistance, 265 insulin sensitivity, 268 Laron syndrome, 262 oxidative damage, 269 oxidative metabolism, 269 reproduction, 268-269 reproductive competence, 268-269 reproductive function, 269 stress resistance, 269 271 LIM-homeobox-3, 356-357 LIM-homeobox-4, 356-357 Lipogenesis, 9 Lipolytic action, 122 Long term cell proliferation, sustaining of, 395-396 Longevity, 259-284 Malignant transformation, 305-340. See also Cancer Mammalian brain, development, regeneration of, 42 Mammary gland cell proliferation, 396-397 Mechano-growth factor, 133 Medial preoptic nucleus, 293 Meissner's corpuscles, 230 Mendelian autosomal recessive pattern, 347 Metabolism, 122 Microglia, insulin-like growth factor actions on, 202 Mitogenesis, 305-340 Mitogenic peptides, 11 Molecular layer, 293 Morsier syndrome, 357 Motor neuron diseases, 249 Mouse models, deficiency, 31-38 ablation of, 32 homozygous disruption of, 32 null mutants, 32 Multiple myeloma cells, chemoattractant for, 175 Muscle, 117–148 growth hormone releasing hormone, 119 Myelination, 200-202

insulin-like growth factor actions on, 199-202 Myoblasts, 117-148 Myocardial infarction, 158 Myocardium, 149-166 Myofiber, 117-148 Myofilament Ca-sensitivity, 156 Myogenesis, stimulation of, 121 Myosin isoform shift, 156 Myotube, 117–148 N-terminal domains, 56 Negative feedback action, 4 Nerve cell plasticity, 243-258 Neural plasticity, 245-248 cell numbers, 245 connectivity, 245 defined, 244 neuronal excitability, modulation of, 246-247 neurotransmitters, release of, 245-246 postsynaptic neurotransmitter receptors, 246 synaptic remodeling, 247-248 synaptic strength, 245 Neural stem cells, 195-198 Neuro-repair, 243-258 Neurodegenerative diseases, 243-258 Neuroendocrine phenotypes, 41 Neuroendocrine tissues, 40–43 Neurogenesis, 221-242 Neuronal development, 285-304 Neurons, insulin-like growth factor actions on, 198–199 neurite growth, synaptogenesis, 199 neurogenesis, 198-199 Neuroprotection, 34, 248-251, 285-304 adult neurogenesis, 248 defined, 245 functionally overlapping cell populations, 248 hazardous substances, 250 loss of function, 248-249 synaptic contacts, 250-251 tonic protective actions, 249-251 Neurotoxic drugs, 249 Neurotrophic factors, 221-242 Notch activation, 403 Null insulin receptor, 66-67 Null neonates, low survival rate of, 33

Obstructive sleep apnea syndrome, 159 Olfactory function, reduction of, 225

Olfactory system, 228-229 GnRH neurons, 228 gonadotrophin-releasing hormone neurons, 228 olfactory neurons, 228 sense neurons, 228 Oligodendrocytes, 187-220 insulin-like growth factor actions on, 199-202 lineage, 199-200 Oocyte, 91–116 Optic nerve hypoplasia, 225 Otitis media, 225 Ovary, 91-116 deficiency of, 92-93 follicle-stimulating hormone, 93 folliculogenesis, 93-94 gonadotropin responsiveness, 95-96 growth hormone, 92-93 increased energy requirements, 93 insulin-like growth factor-I receptor, 93-98 insulin-like growth factors, 93-98 nonhuman primate, 96 Oxidative stress, 249 Pacinian's corpuscles, 230 Panhypopituitarism, 369 Paracrine, 305-340 Perinatal lethality, 37 Peripheral resistance, 149-166 Phosphorylation, 42 Physiology, growth hormone/insulin-like growth factor axis, 1-26 actions of growth hormone, 8-10 anterior pituitary, secretion from, 3 anti-insulin-like, 9-10 binding protein, 1, 5-6 biological functions of, 6 insulin-like growth factor, 14 as carrier proteins, 14 cell adhesion, motility, 13 cell survival, protective effect on, 13-14 class I cytokine receptor superfamily, 4-5 crystallographic structure, receptor complexes, 6 discovery, date of, 2 endocrine actions of, 10-11 fetal growth, 15 future development, 15 growth hormone receptor, 1, 4-8 growth hormone-releasing hormone, 3-4 in human fetus, 2

hypothalamic control, growth hormone secretion, 3-4 insulin-like growth factor binding proteins, 1 insulin-like growth factor-I receptor, 11 - 14intracellular signaling cascade, 7, 12 isolation, growth hormone, 4 ligands for, 12 lipogenesis, 9 mitogenic peptides, 11 negative feedback action, 4 physiological roles, 15 physiology, 10-15 postnatal growth, 15 promotion of cell differentiation, 13 in rats, 2-3 somatostatin, 3-4 stimulation of protein synthesis, 9 sulfation factor, 10 transformed cell phenotype, 12 Pituitary dysfunction, 225 Plasma membrane proteins, 35 Pleitropic hormone, 121 Pluripotent zygote, 397 Posterior cingulate cortex, 293 Postnatal growth, 15 Posttranslational modification, 57 POU1F1, 352-354 Prader-Willi syndrome, 159 Preoptic area, 293 Primary olfactory cortex, 293 Primordial dwarfism, 369 Proliferation, stimulation of, 131 Promotion of cell differentiation, 13 PROP1, 355-356 Proportional short stature, genetic basis, 341-384 Prostate cancer, 308, 311 Proteolysis, role of, 57 Purkinje cell layer, 293

#### Quinolinic acid, 205

Rabson-Mendenhall syndrome, 66 Receptor signal transduction, 27–54 Recombinant virus, 135–136 impaired regenerative capacity, 134 insulin resistance, 134 Reproduction, 268–269 energy costs of, 110 reproductive competence, 268–269 reproductive function, 269

Reproductive system, female, 91-116 androgen, ovarian insulin-like growth factor system, 98-101 insulin-like growth factorPBs, 102-106 ovary, 92-106 follicle-stimulating hormone, 93 growth hormone, 92-93 insulin-like growth factor-I receptor, 93-98 insulin-like growth factors, 93-98 reproduction, energy costs of, 110 uterus, 106-110 insulin-like growth factor-I as "estromedin," 106-109 insulin-like growth factor system in primate uterus, 109-110 Respiratory system, 159 chronic obstructive pulmonary disease, 159 obstructive sleep apnea syndrome, 159 Prader-Willi syndrome, 159 Retina, 221-242 vascularization of, 227 Retinitis pigmentosa, 225 RNA interference, 27-54 amplification activity, 31 Caenorhabditis elegans, 30-31 Drosophila, 30 Ruffini's corpuscles, 230 Sensorineural hearing loss, 225 Sensorineural organs, development of, 222-225 Sensorineural system, 221-242 Sensory organ development, 221-242 auditory, vestibular systems, 230-233 cell proliferation, 233 fibroblast growth factors, 231 gustatory system, 229-230 taste buds differentiation, 229 tongue development, 229 olfactory system, 228-229 GnRH neurons, 228 gonadotrophin-releasing hormone neurons, 228 olfactory neurons, 228 sense neurons, 228 placodial developmental origin, 223 sensorineural organs, development of, 222 - 225signaling mechanisms, 233 tactile system, 230 Krause's corpuscles, 230

Meissner's corpuscles, 230 Pacinian's corpuscles, 230 Ruffini's corpuscles, 230 visual system, 226-227 growth hormone immunoreactivity, 226 Leber congenital amaurosis syndrome, 227retina, vascularization of, 227 septo-optic dysplasia, 227 Sensory receptors, 221-242 Septo-optic dysplasia, 225, 227 Signal transduction pathways, 119 Signaling cascades, intracellular, 38 Silver-Russell syndrome, 369-370 Single-chain polypeptide, 127 Single-transmembrane receptor, 119 Skeletal muscle, 117-148 autocrine function, 132-135 cognate receptors, 125 crosstalk mechanism, 131 cytoskeletal remodeling, 133 diabetes, 132-133 growth hormone-binding protein, 119 growth hormone deficiency, 121 growth hormone releasing hormone, 119 growth hormone system, 118-124 high-affinity binding proteins, 125-126 homologous recombination techniques, 128-129 insulin-like growth factor-I effects on cell growth, 12-130 effects on skeletal muscle metabolism, 130 insulin-like growth factor-I in regeneration, aging, 133-135 insulin resistance, 132 ligands, 125-126 lipolytic action, 122 mechano-growth factor, 133 metabolism, 122 muscle growth, 121-122 myogenesis, stimulation of, 121 physiological role, 121-124 pleitropic hormone, 121 proliferation, stimulation of, 131 recombinant virus, 135-136 impaired regenerative capacity, 134 insulin resistance, 134 signal transduction, 119-120 signal transduction pathways, 119 single-chain polypeptide, 127 single-transmembrane receptor, 119

terminal differentiation, regulation of, 131-132 transgenic mice, 129 Sleep apnea syndrome, 159 Sleep disorders, 149-166 Somatomedin hypothesis, 33 Somatostatin, 1-26 Stem cells advantages of using, 391-394 identification, 387-390 renewal, 385, 394-397 Steriodogenesis, 91-116 Stimulation of protein synthesis, 9 Stress resistance, 259-284, 269 271 Structural aspects, human insulin-like growth factor binding proteins, 58 Sulfation factor, 10 Synaptic contacts, 243-258 Synaptic plasticity, 285-304 Tactile system, 230 Krause's corpuscles, 230 Meissner's corpuscles, 230 Pacinian's corpuscles, 230 Ruffini's corpuscles, 230 Taste buds differentiation, 229 Terminal differentiation, regulation of, 131-132 Thymic epithelial cells, 167-186 Thymic nurse cells, 167–186 Thymocytes, 167-186 proliferation of, 173 Thymopoiesis, 171 Thymulin, 167-186 Thymus, 167-186 Tissue-specific knockouts, 37-38 Tongue development, 229 Transformed cell phenotype, 12 Transgenes, 27-90, 187-220, 259-284 Transplantation, 385 Tumor suppressors, 323 Turner's syndrome, 225, 367–368 Usher's syndrome, 225

Uterus, 91–116 athymic female hosts, 107 expression of, 109–110

hypoplasia, 106-107 insulin-like growth factor system, 106-110 muscle of, 109 regulation of, 109 uterine growth, elucidation of estrogen-insulin-like growth factor-I interaction in, 107-109 Vasodilators, 153 Vertebrate sensory organ development, 221-242 auditory, vestibular systems, 230-233 cell proliferation, 233 fibroblast growth factors, 231 gustatory system, 229-230 taste buds differentiation, 229 tongue development, 229 olfactory system, 228-229 GnRH neurons, 228 gonadotrophin-releasing hormone neurons, 228 olfactory neurons, 228 sense neurons, 228 placodial developmental origin, 223 sensorineural organs, development of, 222-225 signaling mechanisms, 233 tactile system, 230 Krause's corpuscles, 230 Meissner's corpuscles, 230 Pacinian's corpuscles, 230 Ruffini's corpuscles, 230 visual system, 226-227 growth hormone immunoreactivity, 226 Leber congenital amaurosis syndrome, 227 retina, vascularization of, 227 septo-optic dysplasia, 227 Vestibular system, 230-233 Visual system, 225-227 growth hormone immunoreactivity, 226 Leber congenital amaurosis syndrome, 227 retina, vascularization of, 227 septo-optic dysplasia, 227

X-linked recessive patterns, 349